

Programa de Doctorado de Materiales

GENERATION AND CLASSIFICATION OF ADS SOLUTIONS AND THEIR CFT INTERPRETATION

TESIS DOCTORAL

Jesús Montero Aragón Julio 2018



Universidad de Oviedo Universidá d'Uviéu University of Oviedo

Programa de Doctorado en Materiales

GENERATION AND CLASSIFICATION OF ADS SOLUTIONS AND THEIR CFT INTERPRETATION

TESIS DOCTORAL

Directora de tesis: Yolanda Lozano Gómez



Vicerrectoráu d'Organización Académica Vice-rectorate for Academic Organization

RESUMEN DEL CONTENIDO DE TESIS DOCTORAL

1 Título de la Tesis				
Español: Generación y clasificación de soluciones AdS y su interpretación como CFT.	Inglés: Generation and classification of AdS solutions and their CFT interpretation.			
2 Autor				
Nombre: Jesús Montero Aragón	DNI:			
Programa de Doctorado: Materiales				
Órgano responsable: Departamento de Ciencia de los Materiales e Ingenieria Metalúrgica				

RESUMEN (en español)

En esta tesis se tratará el papel de T-dualidad no abeliana como técnica para generar soluciones en supergravedad de tipo II. Nos centraremos en el problema abierto de proporcionar a los nuevos *backgrounds* AdS, de forma consistente, teorías de campos súperconformes, las cuales pueden ser vistas como puntos fijos del flujo de renormalización de *quivers* lineales de rango creciente. Se propondrá así mismo una interesante relación entre estos *quivers* y la descripción mediante T-dualidad abeliana de las teorías originales en términos de *quivers* circulares.

Se revisará por otra parte el potencial de T-dualidad no abeliana para motivar, sondar o poner a prueba clasificaciones de soluciones supersimétricas. Encontraremos nuevos *backgrounds* para distintos espacios AdS y cantidades de supersimetría preservada, incluyendo una solución N=2 AdS4 explícita en teoría M con flujo puramente magnético, la cual resulta relevante para nuevas dualidades de SCFT en tres dimensiones. Nuestros esfuerzos nos conducirán así mismo a nuevos candidatos AdS3xS2 para límites *near-horizon* de agujeros negros en dimensiones altas, motivando así una extensión de la clase conocida de estas geometrías.

Por último, se intentará conseguir una clasificación más amplia de soluciones Minkowski supersimétricas de tipo II en dimensiones bajas, en las cuales se permitirá que la simetría R se encuentre manifiesta geométricamente. Además de recuperar diversas geometrías *near-horizon* AdS, pondremos de manifiesto la unicidad de la geometría N=4 AdS4xS3 de tipo II. Por otra parte, se descubrirá una nueva clase de soluciones Mink3, que incluye un *background* puramente NS. Se espera que estos resultados resulten de interés para las compactaciones con flujo y la extensión de holografía a teorías no conformes, aparte de la clasificación de geometrías AdS en dimensiones más altas.



Vicerrectoráu d'Organización Académica Vice-rectorate for Academic Organization

RESUMEN (en Inglés)

In this thesis we will discuss non-Abelian T-duality as a solution generating technique in type II supergravity. We will focus on the open problem of providing newly generated AdS brackgrounds with consistent dual superconformal field theories, which can be seen as renormalization group fixed points of linear quivers of increasing rank. An interesting relation between these quivers and the Abelian T-dual description of the original theories in terms of circular quivers will be pointed out.

We will review the potential of non-Abelian T-duality to motivate, probe or challenge classifications of supersymmetric solutions. We will find new backgrounds for different AdS spaces and amounts of preserved supersymmetry, including an explicit N=2 AdS4 solution in M-theory with purely magnetic flux, of relevance for new dualities of three-dimensional SCFTs. Our endeavours will also lead us to new AdS3xS2 candidates to higher-dimensional black hole near-horizon limits, while motivating an extension of the known class of such geometries.

Finally, a broader classification of type II supersymmetric lower-dimensional Minkowski solutions will be attempted, in which the R-symmetry is allowed to be realized geometrically. Apart from recovering several known AdS near-horizon geometries, we will report on the uniqueness of the type II N=4 AdS4xS3 geometry. Besides, a new class of Mink3 solutions will be uncovered, which includes a pure NS background. These results are expected to be of interest for flux compactifications and the extension of holography to non-conformal theories, other than the classification of AdS geometries in higher dimensions.

Generación y clasificación de soluciones AdS y su interpretación como CFT

Dedicado a mis abuelos, por su apoyo y cariño a lo largo de toda mi vida.

Abstract

In this thesis we will discuss non-Abelian T-duality as a solution generating technique in type II supergravity. We will focus on the open problem of providing newly generated AdS backgrounds with consistent dual superconformal field theories, which can be seen as renormalization group fixed points of linear quivers of increasing rank. An interesting relation between these quivers and the Abelian T-dual description of the original theories in terms of circular quivers will be pointed out.

We will review the potential of non-Abelian T-duality to motivate, probe or challenge classifications of supersymmetric solutions. We will find new backgrounds for different AdS spaces and amounts of preserved supersymmetry, including an explicit $\mathcal{N} = 2 \text{ AdS}_4$ solution in M-theory with purely magnetic flux, of relevance for new dualities of three-dimensional SCFTs. Our endeavours will also lead us to new $\text{AdS}_3 \times S^2$ candidates to higherdimensional black hole near-horizon limits, while motivating an extension of the known class of such geometries.

Finally, a broader classification of type II supersymmetric lower-dimensional Minkowski solutions will be attempted, in which the R-symmetry is allowed to be realized geometrically. Apart from recovering several known AdS nearhorizon geometries, we will report on the uniqueness of the type II $\mathcal{N} = 4$ AdS₄ × S³ geometry. Besides, a new class of Mink₃ solutions will be uncovered, which includes a pure NS background. These results are expected to be of interest for flux compactifications and the extension of holography to non-conformal theories, other than the classification of AdS geometries in higher dimensions.

Resumen

En esta tesis se tratará el papel de T-dualidad no abeliana como técnica para generar soluciones en supergravedad de tipo II. Nos centraremos en el problema abierto de proporcionar a los nuevos *backgrounds* AdS, de forma consistente, teorías de campos súperconformes, las cuales pueden ser vistas como puntos fijos del flujo de renormalización de *quivers* lineales de rango creciente. Se propondrá así mismo una interesante relación entre estos *quivers* y la descripción mediante T-dualidad abeliana de las teorías originales en términos de *quivers* circulares.

Se revisará por otra parte el potencial de T-dualidad no abeliana para motivar, sondar o poner a prueba clasificaciones de soluciones supersimétricas. Encontraremos nuevos *backgrounds* para distintos espacios AdS y cantidades de supersimetría preservada, incluyendo una solución $\mathcal{N} = 2$ AdS₄ explícita en teoría M con flujo puramente magnético, la cual resulta relevante para nuevas dualidades de SCFT en tres dimensiones. Nuestros esfuerzos nos conducirán así mismo a nuevos candidatos AdS₃ × S² para límites *near-horizon* de agujeros negros en dimensiones altas, motivando así una extensión de la clase conocida de estas geometrías.

Por último, se intentará conseguir una clasificación más amplia de soluciones Minkowski supersimétricas de tipo II en dimensiones bajas, en las cuales se permitirá que la simetría R se encuentre manifiesta geométricamente. Además de recuperar diversas geometrías *near-horizon* AdS, pondremos de manifiesto la unicidad de la geometría $\mathcal{N} = 4 \text{ AdS}_4 \times S^3$ de tipo II. Por otra parte, se descubrirá una nueva clase de soluciones Mink₃, que incluye un *background* puramente NS. Se espera que estos resultados resulten de interés para las compactaciones con flujo y la extensión de holografía a teorías no conformes, aparte de la clasificación de geometrías AdS en dimensiones más altas.

Contents

Pr	Preface I					
Ac	AcknowledgementsIIIImpact factorsV					
In						
1	Introduction.1Non-Abelian T-duality and holography.2Classification of the new solutions	1 1 7				
2	Worldsheet formalism of non-Abelian T-duality2.1 Transformation of the NS-NS sector	11 11 13				
3	Articles 8.1 New $AdS_3 \times S^2$ T-duals with $\mathcal{N} = (0, 4)$ supersymmetry 8.2 $\mathcal{N} = 2 \text{ AdS}_4$ solution in M-theory with purely magnetic flux . 8.3 Large superconformal near-horizons from M-theory 8.4 3D $\mathcal{N} = 4$ linear quivers and non-Abelian T-duals 8.5 The AdS_5 non-Abelian T-dual of KW as a $\mathcal{N} = 1$ linear quiver 8.6 Mink ₃ × S ³ solutions of type II supergravity	17 18 55 88 96 132 174				
4	Results	225				
5	5 Conclusions 23					
Co	clusiones	233				
Bi	liography	237				



Preface

This thesis has been submitted to the University of Oviedo, as a partial fulfillment of the requirements to obtain the PhD degree. The work presented here has been developed during the years 2014-2018 under the supervision of Prof. Yolanda Lozano in the High Energy Physics Theory Group at the Department of Physics of the same university.

Thesis objectives

In this work we intend to study and classify solutions of type II supergravity, providing a holographic interpretation thereof. In particular we aim to:

- Using non-Abelian T-duality, generate AdS solutions in type II supergravity that are hard to reach employing other methods available in the literature.
- Regarding the existence, or not, of a CFT associated with the new AdS solutions, provide an interpretation to the non-compact direction arising under non-Abelian T-duality.
- Improve our understanding of this transformation, in order to eventually elucidate whether it can be thought of as a string theory symmetry or not.
- Using pure spinors and G-structure formalism, propose an extension of the known classes of supersymmetric solutions, motivated by the new backgrounds found using non-Abelian T-duality.
- Extend to lower dimensions the existing results on the classification of supersymmetric type II geometries with a Minkowski external space.

The structure of this thesis, in the collection-of-papers format, is as follows: We start with the introduction in section 1, followed by a review the worldsheet formalism of non-Abelian T-duality in section 2, that includes the transformation rules for both the NS-NS and RR-sectors. A copy of the papers on which this thesis is based is given in section 3. Next, a technical summary of the results is provided in section 4, followed by a more general discussion and conclusions in section 5. Last, a list of references for this text is supplied, apart from the ones included in the copies of the articles.

We remark that, following requirements of the PhD programme, the abstract and conclusions are written both in English and Spanish.

Acknowledgements

This work was possible thanks to the 4-year-long salary provided by the Spanish Government through the *Ayuda predoctoral* fellowship BES-2013-064815, linked to national grant FPA2012-35043-C02-02. This latter project also provided direct travel support in several occasions, besides the associated travel grant EEBB-I-17-12390. We were also lucky to enjoy support for a number of sojourns coming from: the ERC through the COST Action MP1210 "The String Theory Universe", including a fellowship of the STSM programme, the Asturian Regional Government grant FC-GRUPIN-14-08, and the Oviedo U. and Banco Santander fellowship *Ayudas de movilidad*.

I enjoyed help of vital importance for the successful outcome of my PhD from a long list of individuals. First, I strongly appreciate the guidance, patience and support (even for the tough decisions) of my advisor, Prof. Yolanda Lozano. Without her, this work would not have been possible. Besides, the collaborator who certainly has born me the most was Niall Thomas Macpherson, from whom I learnt pretty much stuff, fundamental for what I was able to do so far. I am also grateful to my other collaborators, who were also of most importance for the performance achieved this years: Prof. Carlos Núñez, Eoin Colgáin, Georgios Itsios, Özgür Kelekci, Miok Park and Daniël Prins. It was a pleasure to enjoy many discussions and the atmosphere at the HEP-TH Group at Oviedo.

I am also happy to thank many research institutions and scientists around Europe and the US for their hospitality during several sojourns, workshops and seminars. The list includes (not exhaustively) the Physics departments of the following universities: Milano-Bicocca, Stony Brook, Swansea, KU Leuven, SISSA, King's College London, Oxford, IPhT CEA/Saclay, IPN Lyon, etc.

A number of friends coloured my experience in Oviedo, helping me to realize what lies beyond Physics and work, and allowing me to enjoy the wonderful landscape and possibilities of Asturias, "*paraíso natural*". This includes the cheerful and sporty people of the *Grupo de Montaña Uniovi*, a number of its members I also joined for numerous friendly running sessions and races. Being honest, these people mostly contributed to my taste of *sidra*, *cachopo* and *espcichas*. I thank Alessandro Pini for getting me in touch with these people. Not less friendly or active were to me the fellow PhD students of the Geology Depart. at Oviedo, who I couldn't have met without the warm reception of my friend Pablo Valenzuela. I feel fortunate I could count on the unconditional support and encouragement of my family for so many years. They always made it easy for me to keep in mind that pursuing a degree in higher education was possible and worth doing. They celebrated my successes, and cheered me up during tougher periods. The most outstanding example are my grandparents, whose affection and all kinds of support I enjoyed my whole lifetime. Not to mention my parents, who granted me easy early years at university, despite our limited resources. My gratitude goes to my aunts, uncles, and cousins, for providing infinitely-many joyful moments.

Finally, thank you, Cris, for being *always* by my side...

Impact factors

The impact factors of the scientific journals where the following articles have been published are included below. Source: Journal Citation Reports.

- 1. N. T. Macpherson, J. Montero, D. Prins, " $Mink_3 \times S^3$ solutions of type II supergravity", Nuclear Physics B (2018).
- 2. G. Itsios, Y. Lozano, J. Montero, C. Núñez, "The AdS_5 non-Abelian T-dual of Klebanov-Witten as a $\mathcal{N} = 1$ linear quiver from M5-branes", Journal of High Energy Physics **09**, 038 (2017).
- Y. Lozano, N. T. Macpherson, J. Montero, C. Núñez, "Three-dimensional *N* = 4 linear quivers and non-Abelian T-duals", Journal of High En-ergy Physics 11, 133 (2016).
- O. Kelekci, Y. Lozano, J. Montero, E. Ó. Colgáin and M. Park, "Large superconformal near-horizons from M-theory", Physical Review D 93, 086010 (2016).
- 5. Y. Lozano, N T. Macpherson, J Montero, " $\mathcal{N} = 2$ supersymmetric AdS4 solution in M-theory with purely magnetic flux", Journal of High Energy Physics 10, 4 (2015).
- 6. Y. Lozano, N. T. Macpherson, J. Montero, E. Ó. Colgáin, "New $AdS_3 \times S^2$ T-duals with $\mathcal{N} = (0, 4)$ supersymmetry", Journal of High Energy Physics 8, 121 (2015).

Journal	Year	Impact Factor	Area
Nuclear Physics B	2016	3.678	Physics, Particles and Fields
Physical Review D	2016	4.557	Astronomy, Astrophysics, Physics, Particles and Fields
JHEP	2016	6.063	Physics, Particles and Fields

1. Introduction

We start providing some context and motivation for the use of non-Abelian T-duality to generate AdS solutions of supergravity, reviewing the main advantages and issues of this technique, with emphasis on the holographic interpretation of the newly generated solutions. We will continue this introduction in section 1.2, where the focus will change to the classification of the novel AdS geometries, still in the holographic setting. We will end up motivating the use of Killing spinor techniques to build up classes of non-AdS solutions.

1.1 Non-Abelian T-duality and holography

In recent years, the gauge/gravity or holographic correspondence has proven to be a powerful spin-off of String Theory, permitting the study of stronglycoupled quantum field theories, even when a Lagrangian description is not available, by means of a dual weakly-coupled semiclassical theory of gravity. This duality was originally formulated as an AdS/CFT correspondence [1], relating a string theory on an Anti de-Sitter (AdS) space to a quantum conformal field theory (CFT) on flat spacetime. In particular, the Maldacena conjecture connected type IIB supergravity (the low-energy limit of string theory) on $AdS_5 \times S^5$ to the strong coupling limit of $\mathcal{N} = 4$ super Yang-Mills (SYM) in four dimensions.

By means of the AdS/CFT correspondence, several dualities arising in string theory have become common tools in the field theoretical understanding of CFTs, translating the action e.g. of T- or S-duality relating different AdS vacua to the dual field theories.

Many interesting applications of holography, including the study of QCD phenomena and condensed matter systems, require models to be scale-dependent and show minimal or no preserved supersymmetry, as this has not been observed yet. This urged for an extension of the AdS/CFT correspondence to non-conformal theories, giving rise to a more general gauge/gravity duality.

In recent years, the systematic study of supersymmetric solutions with

non-trivial fluxes in the low-energy limit of string and M-theory has proven relevant in the context of holography as well as in phenomenologically relevant string compactifications. The geometries of interest are usually a warped product of either a Minkowski or AdS external space with an internal manifold which, in the latter case, is taken to be compact in order to allow for a holographic interpretation. Indeed, many AdS/CFT pairs have been constructed from solutions generated in supergravity.

In this thesis we will focus on the use of non-Abelian T-duality (NATD) as a solution generating technique in type II supergravity. We will uncover new supersymmetric solutions in both type IIA/IIB, starting from seed solutions of type IIB/IIA. These are chosen carefully: they have a well known holographic dual field theory, but also they are expected to lead us to new backgrounds, allowing us e.g. to provide explicit examples to known classifications, when these are hard to reach using other methods. Furthermore, if we start with a 10D gravity solution with AdS external space and apply NATD to the isometries of the internal space, the AdS factor will be preserved and we may wonder what kind of dual CFT could be associated to the new solution. Based on computations relying on the backgrounds fields (i.e. the metric, dilaton, Kalb-Ramond two-form and RR fluxes), we may try to deduce some properties of this putative CFT.

The T-duality formalism¹ relies on a non-linear sigma-model (NLSM) describing the propagation of a string in 10D target space, for which an isometry group G is assumed. For Abelian T-duality, G = U(1), otherwise G is a non-Abelian isometry group. In order to get the dual action, the theory is minimally coupled to G, gauging the isometry but enforcing a flat connection with the addition to the action of a Lagrange multiplier term. Integrating out this G-connection, one is led to the dual NLSM, describing the propagation of a string in a different target space. For the usual U(1) T-duality, the Buscher rules [2] allow us to relate algebraically the metric and B-field between the original 10D background and its T-dual, reading them off from the coefficients of the embedding scalars. An important point is that the coordinate for the original isometric direction is replaced, after integrating out the G-connection, by the corresponding Lagrange multiplier, which lives in the Lie algebra of G.

The transformation of the RR-sector under T-duality was derived in [3], matching the field contents of both reductions to nine dimensions of two T-dual theories in type II. This permitted to apply T-duality to an initial solution with RR fluxes turned on and get a T-dual background which still was a type II supergravity solution.

 $^{^1\}mathrm{We}$ introduce the worldsheet formalism of non-Abelian T-duality in section 2.

Introduction

The generalization of Abelian T-duality (ATD) to non-Abelian isometries at the level of the string sigma model was known since the nineties [4, 5], allowing for the transformation of the NS-NS sector of a given 10D target space. However, it has not been until this decade that the work of [6] taught us how to transform the RR sector accordingly, extending the construction made for the Abelian case by [7]. This relied on the spinor representation of T-duality, which can be derived from the Lorentz transformation relating the vielbeins for left and right movers of the dual target space. The transformation rules for the RR sector then follow writing the fluxes as bispinors.

The discovery of [6] led to the recent works uncovering a plethora of new backgrounds, for different AdS dimensionalities and amounts of preserved supersymmetry. In fact, NATD permits to reach solutions that are difficult to be found using other methods, and without e.g. needing to tackle with the PDEs of some class of solutions. A prototypical example is the explicit AdS₆ solution in type IIB supergravity² found in [9] applying NATD to the type IIA Brandhuber-Oz solution [10], which permitted to get hints of the possible dual 5D fixed point theories [11]. A rich class of solutions constructed using NATD included AdS₅ backgrounds with a Sasaki-Einstein internal space. These geometries were extensively investigated [12, 13, 14], supplying a rich phenomenology for the possible SCFTs and RG flows of the new supersymmetric backgrounds. However, a full description of the dual field theories (e.g. including a superpotential) was typically missing in these examples.

Despite its power as a solution generator, NATD presents several, long standing questions that hinder the holographic interpretation of the new backgrounds, see e.g. [15, 16]. Even if the sigma model computation of the NS-NS sector of the new background is straight-forward, the procedure is not known to be invertible, unlike its Abelian counterpart, and it fails to provide all the global information about the new geometry, this latter issue being one of the main points we deal with in this thesis.

In the ATD case, it was possible to derive a condition to recover the U(1) group manifold from the Lagrange multiplier living in its Lie algebra. This was achieved requiring the cancellation of holonomies of the U(1) gauge connection around non-trivial cycles of the string worldsheet (whose number n is given by the power $\mathcal{O}(g_s^n)$ of the string coupling constant perturbative series). This killed two birds with one stone: on the one hand, it showed ATD maps circles to circles (even if the radius thereof is inverted), and on the other hand, it proved that it was a symmetry to all orders in perturbative string theory.

²Another example of an explicit AdS_6 IIB solution being the Abelian T-dual of Brandhuber-Oz, found in [8].

However, in the non-Abelian case, non-commutativity of the gauge connection gives rise to terms apparently impeding the cancellation of the unwanted holonomies. As an immediate consequence, NATD can only be proven to be a transformation between string NLSMs at tree level in string perturbation theory (i.e. for spherical worldsheets). As a by-product, we also have lost the argument that allowed in the Abelian case to keep a compact space after dualization: all that is left as dual coordinates by NATD live in the Lie algebra of the isometry group, which is obviously non-compact by its construction as a vector space. In addition to this, regarding corrections in the string length, the conformal symmetry of the string NLSM is only proven to survive the NATD transformation at first order in α' (as opposed to its Abelian counterpart, shown to work at all orders). These issues have an overall and clear output:

NATD is not proven to be a string theory symmetry, but just a transformation, or map, between type II supergravity solutions.

It is precisely this fact what allows us to claim that the backgrounds generated under its action can be associated to different vacua of string theory, and are not just physically equivalent realizations of the same vacuum, as opposed to its Abelian counterpart [17, 18].

As a drawback, we currently lack a well-defined representation of NATD on the field theory side. The putative CFT duals to the new backgrounds are only guaranteed to exist in the strong coupling regime, as the NATD transformation is not known to survive α' or 1/N corrections. Still, we may infer some properties of this would-be CFT from the new background, and even *define* a quiver gauge theory, from which the CFT would arise as an RG flow fixed point. This will be our approach in this thesis.

We will focus on NATD applied with respect to an SU(2) isometry group acting without isotropy³, and dualize along the directions of its group manifold realization in 10D target space: a (possibly squashed) 3-sphere. Note, however, that other incarnations of NATD as a solution generator are known in the literature, see e.g. [19] or [20]. Our choice of NATD is known to provide a map between SUGRA solutions [21] ⁴ and usually allows for some (if not all) SUSY to be preserved, choosing appropriately the dualization directions⁵, whereas NATD on coset spaces leads in known examples to non-

³The isometry group is chosen such that there are no fixed points under its action.

⁴NATD on non-semisimple groups has been shown to lead to solutions of generalized supergravity for some Bianchi spaces, see [22] and the references therein.

⁵Upon dualization on a SU(2) isometry, it was shown in [21] that the vanishing of Kosmann spinorial Lie derivative for the Killing spinors along the isometry directions in

supersymmetric solutions [19]. Besides, an SU(2) isometry is the simplest generalization of the Abelian case. The use of a larger isometry group would require a more constrained initial background, allowing less room for possible seed solutions.

From the discussion above, we expect the group manifold coordinates of SU(2), e.g. Euler angles realizing the 3-sphere in target space, to get mapped to elements of the uncompact $\mathfrak{su}(2)$. Still, an adequate parametrization of the dual coordinates may be required in order to realize any left-over global symmetries. Indeed, upon dualization on a round 3-sphere with SO(4) isometry, an SU(2) subgroup is expected to survive in the dual background⁶. Using spherical coordinates to parametrize the NAT-dual directions, this symmetry is realized explicitly at the level of the metric as a warped $M_3 \simeq \mathbf{R}^+ \times S^2$ space.

Regarding compactness, we seem to get a 3-sphere substituted for this M_3 space. Usually, one can run along its "radial" direction $\rho \in \mathbf{R}^+$ without finding any hint of the 10D supergravity description breaking down (e.g. singularities, large curvature or dilaton). In the simplest cases, the metric on M_3 clearly interpolates between \mathbf{R}^3 for small ρ , and a cylinder for large ρ , thus describing a space with the shape of an infinite cigar. At least in this cases, we could claim the global properties of the new background are perfectly determined, M_3 being uncompact and thus having the new background an internal space of infinite volume.

However, if one is interested in the holographic interpretation of the newly found AdS solution, an infinite internal space would lead to an inconsistent dual SCFT, with an infinite holographic central charge or free energy. We would then need to discard this solution as suitable for holography. Instead, in this thesis we settle for an alternative perspective: the fact that the dual coordinates live in a Lie algebra, and no group manifold can be assigned to them from the information provided by NATD alone, is just a hint that, apart from the residual isometries⁷, the global properties of the dual manifold cannot be inferred from the transformation itself. In this second approach, the internal space is thought to be compact, yet globally unknown.

Assuming the new solutions have indeed a compact internal space, compatible with a would-be consistent dual field theory, but not fully determined,

the original background was a necessary and also sufficient condition for SUSY to be preserved.

⁶For a full $SO(4) \approx SU(2) \times SU(2)$ isometry, the string worldsheet theory is a principal chiral model exhibiting the to-be-dualized SU(2) isometry, but also an additional spectator SU(2), which can be realized as a residual isometry after NATD in appropriate coordinates.

⁷Any global symmetries already present in the original target space, and independent of the dualization directions, will be also manifest in the new background.

gives rise to one of the most important open questions in the study of NATD: *How* do we get the piece of extra information needed to determine the new AdS/CFT pair? We hope to make a contribution towards a definite answer to this question in this work.

Our approach in [23, 24], see sections 3.4 and 3.5, followed the ideas developed in [25], also extended and applied in other recent works [26, 27, 28, 29].

In [25] the non-Abelian T-dual of $AdS_5 \times S^5$ already found in the seminal paper [6] is investigated, using its embedding in the $\mathcal{N} = 2$ Gaiotto-Maldacena geometries [30], in order to regularize the field theory and *complete* the background. GM geometries are the gravity duals of theories living in the worldvolume of D4-NS5 intersections. The embedding of the NATD background in this class provided a strong support to the brane set-up proposed for the solution (let us recall that the mapping of brane intersections under NATD is not known). In particular, they manage to use the detailed AdS/CFT dictionary known for this class of solutions to propose a consistent 4D $\mathcal{N} = 2$ long quiver for which a *complete* gravity dual solution (i.e. with finite internal volume) can be determined. Then, the NATD gravity solution arises as an asymptotic, local realization of this completed solution⁸.

The first extension of these ideas was done in [23], in which a solution generated with NATD is embedded in the class [32, 33] of type IIB $\mathcal{N} = 4$ $\mathrm{AdS}_4 \times S^2 \times S^2$ backgrounds. A consistent field theory proposal was achieved for the new background employing the holographic dictionary of [34] between these geometries and the $T^{\hat{\rho}}_{\rho}(SU(N))$ Gaiotto-Witten theories [35] living on the worldvolume of a D3-D5-NS5 system⁹.

Later, it was claimed in [24] that the NATD solution obtained from the gravity dual of the Klebanov-Witten model [36] can be seen as the near-horizon limit of a D4-NS5-NS5' orthogonal intersection, whose worldvolume dynamics is given at the conformal fixed point by the $\mathcal{N} = 1$ mass deformation of the $\mathcal{N} = 2$ theories proposed in [25]. This is an extension to the non-Abelian case of the relation between the Abelian T-duals of the $\mathcal{N} = 2$ AdS₅ × S^5/\mathbb{Z}_2 and the $\mathcal{N} = 1$ Klebanov-Witten solutions [37].

More generally, the rough idea is to use the AdS/CFT correspondence to provide the missing information from string theory about the NATD gravity solutions. A (consistent) field theory proposal is employed to complete the background, even possibly smoothing out its singularities. In the works enclosed in this thesis, our approach to this problem is the following:

⁸Indeed, the completed gravity solution arises as a regular superposition of Maldacena-Núñez geometries [31], which yields the (singular) NATD solution in a certain limit.

 $^{^{9}\}mathrm{A}$ brief introduction to these theories and their holographic duals is given in [23], included in section 3.4 .

- For the new AdS solutions generated using NATD, we systematically study their properties, computing the quantized charges associated to RR-fluxes, the holographic central charge/free energy and providing an analysis of the preserved supersymmetry.
- We propose a Hanany-Witten brane set-up based on the quantized charges.
- We use insights from four- and three-dimensional SCFTs with wellknown gravity duals to propose a consistent linear quiver, regularizing the former brane set-up by a motivated addition of flavour branes. It will be conjectured that this field theory is dual to a regular background with finite internal space, from which the non-Abelian T-dual solution arises in a certain limit.

Other than generating and holographically interpreting supergravity solutions generated with NATD, it is also interesting to consider how the new solutions relate to known classes of supersymmetric geometries. In some cases, this might even lead to the extension of such classifications. Applied techniques can then be utilized to construct other classes of geometries, even if not directly related to NATD backgrounds.

1.2 Classification of the new solutions

Killing spinor techniques, such as the pure spinor formalism and G-structures [38, 39], have been extensively used in the literature to find supersymmetric classes of AdS solutions of supergravity with different amounts of preserved supersymmetry, even if originally devised for $\mathcal{N} = 1$ backgrounds¹⁰. Using a decomposition Ansatz for the Killing spinors into external and internal components, the dilatino and gravitino variations are reduced to constrains on the internal manifold. The Killing spinor equations can then be rephrased, in terms of polyforms, as conditions for a certain reduction of the structure group of the corresponding internal manifold.

Typical examples in the NATD literature are gravity backgrounds with RR-fluxes dual to $\mathcal{N} = 1$ theories (see e.g. [41, 42]), in which internal spinors defined in a 4- or 6-dimensional manifold give rise to a SU(3)-structure for that space. This is the simplest generalization of Calabi-Yau manifolds, permitting string compactifications for backgrounds supported with either Kalb-Ramond or RR-fluxes, in which the non-vanishing torsion classes (e.g. the

¹⁰String compactifications of flux backgrounds, some of them of phenomenological interest, were a motivation for these methods (see [40] for a review).

non-closure of the holomorphic 3-form) deforming the Calabi-Yau are sourced by the non-trivial fluxes.

The effect of NATD on the Killing spinors is well-known¹¹. The RRfluxes transform under the spinor representation of NATD [6], which can be derived from the Lorentz transformation relating the frame of left and right movers. Applying the aforementioned decomposition to the Killing spinors, NATD has the effect of rotating one internal spinor w.r.t. the other, typically reducing the structure group [44, 42]. This general pattern helps identifying the class of solutions the new background should belong to.

The G-structure formalism was used in [43] to embed a new $\mathcal{N} = 2 \text{ AdS}_4$ background, generated with NATD, in 11D. This solution has no electric flux and should therefore fit into the classification of [45, 46]. This solution is the second possible candidate to provide a holographic realization of the 3d-3d correspondence between three-dimensional cycles wrapped by M5-branes, and the 3D $\mathcal{N} = 2$ CFTs resulting from the twisted compactification of the corresponding 6D (2,0) theory. The first explicit solution within this class was the uplift [47, 48] from 7D gauged supergravity of the Pernici-Sezgin solution [49], constructed in the 80s.

Considering $AdS_3 \times S^2$ solutions dual to 2D $\mathcal{N} = (0,4)$ SCFTs, a celebrated example is the $AdS_3 \times S^2 \times CY_3$ near-horizon of the Maldacena-Strominger-Witten 4D black-hole [50], which results from M-theory compactified on $CY_3 \times S^1$. The dual $\mathcal{N} = (0,4)$ CFT, arising from M5-branes wrapping a 4-cycle inside the CY_3 , allows for the counting of black hole microstates (the Bekenstein-Hawking entropy and subleading corrections to it) in terms of the central charge of the CFT [51, 52]. This solution has a small superconformal symmetry¹², with an SU(2) R-symmetry, but other realizations of the algebra are known to exist, what suggested that a larger class of such black hole near-horizons should be available. Addressing this point, $\mathcal{N} = (0, 4) \operatorname{AdS}_3 \times S^2$ solutions with $SU(2) \times SU(2)$ R-symmetry, compatible with a *large* superconformal algebra, are found in [54]. These lie outside the known class of solutions engineered in [55] and therefore motivated its extension to include the new backgrounds. This was done in [56], where the aforementioned decomposition method was applied to the Killing spinors, in order to get the most general Ansatz compatible with a SU(2)-structure internal manifold¹³.

¹¹For a review of SU(3) and SU(2) structures on six-dimensional internal manifolds, and details of the effect of NATD on them, see the appendices of [43], section 3.2, and the references therein.

¹²The different possible 2D $\mathcal{N} = 4$ superconformal algebras are introduced e.g. in [53].

¹³The minimal SU(2) R-symmetry of these solutions was assumed to be realized geometrically as the SU(2) isometry from the S^2 factor in the $AdS_3 \times S^2$ Ansatz. The large

An important example of allowing the R-symmetry to be manifest as an isometry for the bosonic supergravity fields is the comprehensive classification work done for type IIB $AdS_4 \times S^2 \times S^2$ solutions [32, 33]. This was motivated by the search for holographic duals to three dimensional $\mathcal{N} = 4$ CFTs arising from 4D SYM with a planar defect. The dual SCFTs, fixed points of the $T^{\hat{\rho}}_{\rho}(SU(N))$ theories of Gaiotto and Witten [35], admit a global $SO(2,3) \times SO(4)$ symmetry, with the $SO(4) \simeq SO(3) \times SO(3)$ R-symmetry holographically related to an isometry of the bulk geometry. This demanded that the internal manifold should be written as $S^2 \times S^2 \times \Sigma_2$ with a Riemann surface Σ_2 , which was in fact highly restricted, even allowing to explicitly solve the Killing spinor equations without resorting to the geometrical techniques outlined above. As a result, a full classification of type IIB $\mathcal{N} = 4$ $AdS_4 \times S^2 \times S^2 \times \Sigma_2$ backgrounds, arising in the near-horizon limit of D3-D5-NS5 brane intersections, was given in terms of a pair of harmonic functions defined on the Riemann surface Σ_2 , which was either an infinite strip or an annulus. This permitted a field theory interpretation in terms of, respectively, linear or circular quiver realizations of the Gaiotto-Witten theories [34, 57].

As we mentioned before, this is the class of AdS/CFT pairs used in [23], see section 3.4, to embed a new type IIB solution generated with NATD from the $\mathcal{N} = 4 \text{ AdS}_4 \times S^3 \times S^2$ geometry, arising in the near-horizon limit of a D2-D6 brane system¹⁴. This non-Abelian T-dual background could be associated this way to a regularised dual linear quiver. Interestingly, the Abelian T-dual of this same IIA seed solution was already recovered in [57] in a certain limit of a circular brane set-up, dual to a circular-quiver version of the Gaiotto-Witten theories.

Turning our attention to gravity backgrounds beyond conformal solutions, which can be phenomenologically relevant to flux compactifications, a large number of attempts have been made to construct/classify solutions with external Minkowski space and minimal supersymmetry, see [59, 60, 38, 61] or [62, 63, 64, 65] for recent results. The relatively low amount of (super)isometries entails significant difficulties to solve the reduced Killing spinor equations. Progress was made in [66] for backgrounds containing a Mink₄ factor, assuming $\mathcal{N} = 2$ supersymmetry and a geometrically-realised

superconformal symmetry can then arise when an extra $SU(2)_R$ factor emerges from the Killing spinor bilinears derived in the analysis.

¹⁴This geometry can be obtained from the $\mathcal{N} = 8 \text{ AdS}_4 \times S^7 / \mathbb{Z}_k$ solution of 11D supergravity, upon reduction to type IIA on the Hopf fibre (where the \mathbb{Z}_k orbifold is applied) of one of the two 3-spheres inside the S^7 . This is in contrast to the IIA $\mathcal{N} = 6$ gravity dual of ABJM [58], also associated to a D2-D6 system, and arising from the reduction of the same 11D solution, but on the \mathbb{Z}_k -orbifolded circle $S^1 \hookrightarrow S^7 \to \mathbb{CP}^3$.

SU(2) R-symmetry, which allowed to recover AdS solutions¹⁵.

Inspired by this example, we tackled the problem of classifying type II solutions with a $Mink_3 \times S^3$ factor in [67], in which the three-sphere allows for the geometrization of a would-be SO(4) R-symmetry, instead of the $S^2 \times S^2$ factor assumed in [32] and subsequent works. In fact, our initial goal was to find gravity duals of three-dimensional $\mathcal{N} = 4$ SCFTs by examining $AdS_4 \times S^3$ type II backgrounds, but this approach seemed too restrictive and allowing for a more general Mink₃ factor turned out to be more convenient.

Besides, it is worth highlighting that even if the assumption of the S^3 factor will lead to solutions with $\mathcal{N} > 1$, our classification could still be of interest for phenomenological applications, as we expect to be able to reduce supersymmetry through a deformation of the 3-sphere.

The structure of this thesis, in the collection-of-papers format, is as follows: In section 2, we review the worldsheet formalism of non-Abelian Tduality, deriving the transformation rules for both the NS-NS and RR-sectors. A copy of the papers on which this thesis is based is given in section 3. Next, a technical summary of the results is provided in section 4, followed by a more general discussion and conclusions in section 5. Last, a list of references for this text is supplied, apart from the ones included in the copies of the articles.

We remark that, following requirements of the PhD programme, the abstract and conclusions of this thesis are written both in English and Spanish.

$$ds^{2} = e^{2A} ds^{2} (Mink_{d}) + ds^{2}_{10-d} = e^{2B} \left(e^{2\rho} ds^{2} (Mink_{d}) + d\rho^{2} \right) + ds^{2}_{9-d},$$

¹⁵Even if optional for supersymmetry, the R-symmetry is part of the superconformal algebra required for the AdS factor. Remark that a 10D solution with a Mink_d factor can be regarded as a AdS_p solution with p > d, in particular as AdS_{d+1}:

where A and B are warp factors depending only on the internal manifolds M_{10-d} and M_{9-d} , respectively.

2. Worldsheet formalism of non-Abelian T-duality

We present below a brief introduction to the formulation of non-Abelian T-duality (NATD) for the string sigma model. This follows closely and summarizes the detailed review given in [12], complemented by [41] and [68].

2.1 Transformation of the NS-NS sector

Consider a 10D target space with an SU(2) isometry and metric written as:

$$ds^{2} = G_{\mu\nu}(x)dx^{\mu}dx^{\nu} + 2G_{\mu i}(x)dx^{\mu}L^{i} + G_{ij}(x)L^{i}L^{j} , \qquad (2.1)$$

where $\mu = 1, 2, ..., 7$ run over the transverse-space directions and $L^i = -i \operatorname{Tr}(t^i g^{-1} dg)$, for i = 1, 2, 3 are the Maurer-Cartan forms, with t^i the SU(2) generators. The group element $g \in SU(2)$ depends on the target space isometry directions, realizing the SU(2) group manifold. We can choose to parametrize g such that the corresponding coordinates are the Euler angles:

$$g = e^{\frac{i}{2}\tau_3\phi} e^{\frac{i}{2}\tau_2\theta} e^{\frac{i}{2}\tau_3\psi} , \qquad (2.2)$$

where $\theta \in [0, \pi]$, $\psi \in [0, 4\pi]$, $\phi \in [0, 2\pi]$ and the $\tau_i = \sqrt{2} t_i$ are the Pauli matrices, as SU(2) generators. On the other hand, the Kalb-Ramond field¹ can be written as:

$$B = B_{\mu\nu}(x)dx^{\mu} \wedge dx^{\nu} + B_{\mu i}(x)dx^{\mu} \wedge L^{i} + \frac{1}{2}B_{ij}(x)L^{i} \wedge L^{j} .$$
 (2.3)

Remark that all the metric and B field coefficients do only depend, by construction, on the transversal directions, same as for the dilaton $\Phi(x)$. The only fields depending on the SU(2) directions are the L^i forms.

¹For the sake of generality, we take an unspecified B in this exposition. Remark, however, that B = 0 for the seed solution in all examples considered in this thesis.

The Lagrangian density for the NLSM describing the propagation of a string in the above 10D target space is given by:

$$\mathcal{L} = Q_{\mu\nu}\partial_{+}X^{\mu}\partial_{-}X^{\nu} + Q_{\mu i}\partial_{+}X^{\mu}L^{i}_{-} + Q_{i\mu}L^{i}_{+}\partial_{-}X^{\mu} + E_{ij}L^{i}_{+}L^{j}_{-} , \qquad (2.4)$$

where we have introduced the embedding scalars $\{X^{\mu}\}$ for the transverse directions², $L^{i}_{\pm} = -i \operatorname{Tr}(t^{i}g^{-1}\partial_{\pm}g)$ and the coefficients

$$Q_{\mu\nu} = G_{\mu\nu} + B_{\mu\nu}, \qquad Q_{\mu i} = G_{\mu i} + B_{\mu i}, \qquad Q_{i\mu} = G_{i\mu} + B_{i\mu},$$

$$E_{ij} = G_{ij} + B_{ij}, \qquad (2.5)$$

from which metric and B-field are read-off. In order to perform the NATD procedure, we minimally couple the theory to the isometry group SU(2), substituting partial for covariant derivatives,

$$\partial_{\pm}g \longrightarrow D_{\pm}g = \partial_{\pm}g - A_{\pm}g , \qquad (2.6)$$

where A_{\pm} is the SU(2) connection. We enforce it to be flat by adding the Lagrange multiplier term $-i \operatorname{Tr}(vF_{\pm})$, where $F_{\pm} = \partial_{+}A_{-} - \partial_{-}A_{+} - [A_{+}, A_{-}]$ is the curvature for A_{\pm} . The resulting action is then invariant under

$$g \to h^{-1}g$$
, $v \to h^{-1}vh$, $A_{\pm} \to h^{-1}A_{\pm}h - h^{-1}\partial_{\pm}h$, (2.7)

for $h \in SU(2)$. If we now integrate out the SU(2) connection (i.e. find its equations of motion and substitute it back into the action) the dual action will be reached. However, the new theory will still depend on the spectator fields X^{μ} , the Euler angles and also the Lagrange multipliers v_i . In order to preserve the number of d.o.f., we gauge fix the SU(2) isometry by setting $g = \mathbb{I}$, i.e. in eq. (2.2) we take $\theta = \psi = \phi = 0$. With this choice of gauge, the original coordinates get substituted by the Lagrange multipliers v_i , so that the dual sigma model does only depend on them and on the spectator fields X^{μ} .

We remark that other gauge fixings are possible (even mixing original and dual coordinates), that might be useful e.g. in realizing some residual isometries. The different choices are locally related by diffeomorphisms in the NATD target space, as shown in section 2.2.3 of [12]. A general gauge fixing is also discussed in appendix B of that paper.

After the $g = \mathbb{I}$ gauge fixing and a partial integration on the Lagrange multiplier term, the EOMs of the gauge connection yield:

$$A^{i}_{+} = i M^{-1}_{ji} (\partial_{+} v_{j} + Q_{\mu j} \partial_{+} X^{\mu}), \qquad (2.8)$$
$$A^{i}_{-} = -i M^{-1}_{ij} (\partial_{-} v_{j} - Q_{j\mu} \partial_{-} X^{\mu}),$$

²These are spectator fields under NATD.

where we introduced the matrix M = E + f, with $f_{ij} = f_{ij}{}^k v_k$ and $f_{ij}{}^k$ the structure constants of SU(2). Substituting back A^i_{\pm} into the gauged action, we get the dual NLSM:

$$\widehat{\mathcal{L}} = Q_{\mu\nu}\partial_{+}X^{\mu}\partial_{-}X^{\nu} + (\partial_{+}v_{i} + \partial_{+}X^{\mu}Q_{\mu i})M_{ij}^{-1}(\partial_{-}v_{j} - Q_{j\mu}\partial_{-}X^{\mu}) .$$
(2.9)

The metric and B-field of the new target space can now be read-off as the symmetric and anstisymmetric part of the above coefficients w.r.t. the kinetic terms of the new embedding scalars $\{X^{\mu}, v^i\}$. This yields a generalization of the Buscher rules [2] to the non-Abelian case:

$$\widehat{Q}_{\mu\nu} = Q_{\mu\nu} - Q_{\mu i} M_{ij}^{-1} Q_{j\nu} , \qquad \widehat{E}_{ij} = M_{ij}^{-1} ,
\widehat{Q}_{\mu i} = Q_{\mu j} M_{ji}^{-1} , \qquad \widehat{Q}_{i\mu} = -M_{ij}^{-1} Q_{j\mu} .$$
(2.10)

From the path integral derivation of the former, one sees the dilaton suffers a 1-loop shift:

$$\widehat{\Phi}(x,v) = \Phi(x) - \frac{1}{2}\log(\det M).$$
(2.11)

Correspondingly, the transformation of the worldsheet derivatives is

$$L^{i}_{+} = -(M^{-1})_{ji} \left(\partial_{+} v_{j} + Q_{\mu j} \partial_{+} X^{\mu}\right), \qquad (2.12)$$
$$L^{i}_{-} = (M^{-1})_{ij} \left(\partial_{-} v_{j} - Q_{j\mu} \partial_{-} X^{\mu}\right),$$

while $\partial_{\pm} X^{\mu}$ remain invariant. This represents a canonical transformation between the pair of T-dual sigma models [69, 70].

For the examples considered in this thesis B = 0 in the seed solution, so that Q = G and the above (2.10) simplifies to:

$$\hat{G}_{\mu\nu} = G_{\mu\nu} - M_{ij}^{-1} G_{\mu i} G_{\nu j}, \qquad \hat{G}_{ij} = \frac{1}{2} M_{(ij)}^{-1}, \qquad \hat{G}_{i\mu} = -\frac{1}{2} M_{[ij]}^{-1} G_{j\mu}$$
$$\hat{B}_{i\mu} = -\frac{1}{2} M_{(ij)}^{-1} G_{j\mu}, \qquad \hat{B}_{ij} = \frac{1}{2} M_{[ij]}^{-1}, \qquad (2.13)$$

where now the matrix M reduces to $M_{ij} = G_{ij} + \sqrt{2} \epsilon_{ijk} v_k$. Remark that, in general, Kalb-Ramond 2-form components are generated for the dual back-ground, even if the original seed solution had none.

2.2 Transformation of spinors and the RR sector

In order to determine how the RR sector transforms under NATD, we first need to derive the action on the spinors of the Lorentz transformation relating left and right movers. We start rewriting the metric of the original target space in terms of the frame fields

$$ds^{2} = \eta_{AB}e^{A}e^{B} + e^{a}e^{a}, \qquad (2.14)$$

where the vielbein is chosen to be

$$e^{A} = e^{A}_{\mu} dx^{\mu} , \qquad e^{a} = \kappa^{a}_{\ j} L^{j} + \lambda^{a}_{\mu} dx^{\mu} , \qquad (2.15)$$

with A = 1, 2, ..., 7 and a = 1, 2, 3. Left and right movers transform differently under NATD, as given in (2.12), leading to two different sets of frame fields after the transformation:

$$\hat{e}_{+} = -\kappa M^{-T} (dv + Q^{T} dX) + \lambda dX ,$$

$$\hat{e}_{-} = \kappa M^{-1} (dv - Q dX) + \lambda dX .$$
(2.16)

However, these frame fields still describe the same geometry, and must therefore be related by a Lorentz transformation, $\hat{e}_{+} = \Lambda \hat{e}_{-}$, which can be found to be

$$\Lambda = -\kappa M^{-T} M \kappa^{-1} = -\kappa^{-T} M M^{-T} \kappa^{T} .$$
 (2.17)

The action Ω induced by this Lorentz transformation on spinors is then given by:

$$\Omega^{-1}\Gamma^a\Omega = \Lambda^a{}_b\Gamma^b \,. \tag{2.18}$$

For NATD performed on a freely acting SU(2) isometry, Ω reads³:

$$\Omega = \Gamma_{11} \frac{-\Gamma_{123} + \zeta_a \Gamma^a}{\sqrt{1 + \zeta_a \zeta^a}} , \qquad (2.19)$$

where Γ_{11} is the product of all ten gamma matrices, with $(\Gamma_{11})^2 = \mathbb{I}$, and $\zeta^a = \kappa^a{}_i v^i / (\det \kappa)$ (for $B_{ij} = 0$). Let us remark that Ω leaves invariant the gamma matrices Γ^A corresponding to the transverse directions.

We are now ready to introduce the action of NATD on the RR sector. Employing the democratic formalism [71], RR fields are combined with their Hodge duals⁴ to form the following bispinors:

³A general form of Ω for an isometry group acting without isotropy is given in [19].

⁴Higher *p*-forms are related to the lower ones by $F_p = (-1)^{\left\lfloor \frac{p}{2} \right\rfloor} \star F_{10-p}$, with the Hodge dual given by:

$$(\star F_p)_{\mu_{p+1} \cdot \mu_D} = \frac{1}{p!} \sqrt{|g|} \epsilon_{\mu_1 \dots \mu_D} F_p^{\mu_1 \dots \mu_p}$$

where $\not{F}_p = \frac{1}{p!} \Gamma_{\mu_1...m_p} F_p^{\mu_1\mu_2...m_p}$ using the Clifford map. Remark that F_{2n+1} odd-degree fluxes correspond to type IIB and F_{2n} even-degree fluxes to type IIA. The action of NATD on the RR sector, from a type IIB to a IIA solution, is then given by:

$$\widehat{P} = P \cdot \Omega^{-1} \,, \tag{2.21}$$

where Ω is given in eq. (2.19). If starting from a type IIA to a IIB solution instead, the role of P and \hat{P} is exchanged. Thanks to this relation, given the RR sector of a seed solution, we can determine the fluxes needed for the T-dual background to be also a solution of type II supergravity.

In general, fluxes along the duality directions are mapped as $F_p \rightarrow \hat{F}_{p-1}, \hat{F}_{p-3}$ under NATD, while the map works the other way round for fluxes not along the duality directions, $F_p \rightarrow \hat{F}_{p+1}, \hat{F}_{p+3}$. This is a generalization of the Abelian case, in which $F_p \rightarrow \hat{F}_{p\pm 1}$, and underlies the change in the dimensionality of the D-branes under NATD, mapping *p*-branes to $(p \pm 1)$ and $(p \pm 3)$ -branes⁵. A review of the "recipes" relating the original and Tdual fluxes for the simple case of NATD on a round S^3 can be found in the appendix B of our [23], see section 3.4, where also the Abelian case [3] is reviewed. The more general prescription for NATD applied on a squashed S^3 is treated in the appendices of [12, 68].

Knowledge of the spinor representation Ω of NATD also allows us to transform the ten-dimensional MW Killing spinors, say ϵ_1 and ϵ_2 , of a type II supergravity solution to get the spinors $\hat{\epsilon}_1$ and $\hat{\epsilon}_2$ associated to the T-dual solution:

$$\hat{\epsilon}_1 = \epsilon_1, \qquad \hat{\epsilon}_2 = \Omega \epsilon_2.$$

In the examples considered, where NATD is always applied to the internal space, the above transformation reduces to a rotation of one of the Killing spinors w.r.t. the other. This has the effect of reducing the structure group of the internal space. A fast review of the transformation of pure spinors and G-structures under NATD is given in the appendices of the enclosed work [43], in section 3.2. An extended introduction can be also found in [41].

⁵Remark, however, that this does not imply that the number of independent charges is doubled under NATD. Sometimes, some of the would-be branes are found not to be BPS, see e.g. [54], included in section 3.1. In other examples, some branes dielectrically expand into others of higher dimensionality, and are interpreted to actually describe the same d.o.f.s. This is discussed in the results section 4.
3. Articles

We present below a copy of the papers on which this thesis is based.

Contents

3.1	New $AdS_3 \times S^2$ T-duals with $\mathcal{N} = (0, 4)$ supersymmetry
3.2	$\mathcal{N} = 2 \text{ AdS}_4$ solution in M-theory with purely magnetic flux
3.3	Large superconformal near-horizons from M-theory 88
3.4	3D $\mathcal{N} = 4$ linear quivers and non-Abelian T-duals 96
3.5	The AdS_5 non-Abelian T-dual of KW as a $\mathcal{N} = 1$ linear quiver
3.6	${ m Mink}_3 imes S^3$ solutions of type II supergravity $\ .\ .\ 174$

3.1 New $AdS_3 \times S^2$ T-duals with $\mathcal{N} = (0, 4)$ supersymmetry



Published for SISSA by 2 Springer

RECEIVED: July 21, 2015 ACCEPTED: August 4, 2015 PUBLISHED: August 25, 2015

19

New $AdS_3 imes S^2$ T-duals with $\mathcal{N} = (0,4)$ supersymmetry

Yolanda Lozano,^a Niall T. Macpherson,^b Jesús Montero^a and Eoin Ó Colgáin^{c,d,e}

^aDepartamento de Física, Universidad de Oviedo,

33007 Oviedo, Spain

- ^bDipartimento di Fisica, Università di Milano-Bicocca and INFN Sezione di Milano-Bicocca, I-20126 Milano, Italy
- ^cC.N. Yang Institute for Theoretical Physics, SUNY Stony Brook, NY 11794-3840, U.S.A.
- ^dDepartment of Mathematics, University of Surrey, Guildford GU2 7XH, U.K.

^eKavli Institute for Theoretical Physics China, Institute for Theoretical Physics, Chinese Academy of Sciences, Beijing 100190, China

E-mail: ylozano@uniovi.es, niall.macpherson@mib.infn.it, monteroaragon@uniovi.es, eoin.ocolgain@stonybrook.edu

ABSTRACT: It is well known that Hopf-fibre T-duality and uplift takes the D1-D5 nearhorizon into a class of $AdS_3 \times S^2$ geometries in 11D where the internal space is a Calabi-Yau three-fold. Moreover, supersymmetry dictates that Calabi-Yau is the only permissible SU(3)-structure manifold. Generalising this duality chain to non-Abelian isometries, a strong parallel exists, resulting in the first explicit example of a class of $AdS_3 \times S^2$ geometries with SU(2)-structure. Furthermore, the non-Abelian T-dual of $AdS_3 \times S^3 \times S^3 \times S^1$ results in a new supersymmetric $AdS_3 \times S^2$ geometry, which falls outside of all known classifications. We explore the basic properties of the holographic duals associated to the new backgrounds. We compute the central charges and show that they are compatible with a large $\mathcal{N} = 4$ superconformal algebra in the infra-red.

KEYWORDS: Supersymmetry and Duality, AdS-CFT Correspondence

ARXIV EPRINT: 1507.02659

OPEN ACCESS, © The Authors. Article funded by $SCOAP^3$.

Contents

2 $AdS_3 \times S^2$ geometries in 11D with SU(2)-structure43The $AdS_3 \times S^3 \times S^3 \times S^1$ background with pure RR flux73.1Quantized charges83.2Instantons93.3Baryon vertices and 't Hooft monopoles103.4Central charge104Non-Abelian T-dual $AdS_3 \times S^3 \times S^2$ solution in IIA104.1Background104.2Large gauge transformations124.3Quantized charges134.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27	1	Introduction	1	
3The $AdS_3 \times S^3 \times S^3 \times S^1$ background with pure RR flux73.1Quantized charges83.2Instantons93.3Baryon vertices and 't Hooft monopoles103.4Central charge104Non-Abelian T-dual $AdS_3 \times S^3 \times S^2$ solution in IIA104.1Background104.2Large gauge transformations124.3Quantized charges124.3Quantized charges134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27	2	$AdS_3 \times S^2$ geometries in 11D with SU(2)-structure	4	
3.1Quantized charges83.2Instantons93.3Baryon vertices and 't Hooft monopoles103.4Central charge104Non-Abelian T-dual $AdS_3 \times S^3 \times S^2$ solution in IIA104.1Background104.2Large gauge transformations124.3Quantized charges124.3Quantized charges134.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27	3	The $AdS_3 imes S^3 imes S^3 imes S^1$ background with pure RR flux	7	
3.2Instantons93.3Baryon vertices and 't Hooft monopoles103.4Central charge104Non-Abelian T-dual $AdS_3 \times S^3 \times S^2$ solution in IIA104.1Background104.2Large gauge transformations124.3Quantized charges124.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		3.1 Quantized charges	8	
3.3Baryon vertices and 't Hooft monopoles103.4Central charge104Non-Abelian T-dual $AdS_3 \times S^3 \times S^2$ solution in IIA104.1Background104.2Large gauge transformations124.3Quantized charges124.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		3.2 Instantons	9	C .
3.4Central charge104Non-Abelian T-dual $AdS_3 \times S^3 \times S^2$ solution in IIA104.1Background104.2Large gauge transformations124.3Quantized charges124.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		3.3 Baryon vertices and 't Hooft monopoles	10	H
4Non-Abelian T-dual $AdS_3 \times S^3 \times S^2$ solution in IIA104.1Background104.2Large gauge transformations124.3Quantized charges124.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		3.4 Central charge	10	는
4.1Background104.2Large gauge transformations124.3Quantized charges124.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27	4	Non-Abelian T-dual $AdS_3 imes S^3 imes S^2$ solution in IIA	10	님
4.2Large gauge transformations124.3Quantized charges124.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		4.1 Background	10	.0
4.3Quantized charges124.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		4.2 Large gauge transformations	12	\bigcirc
4.3.1Color branes134.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		4.3 Quantized charges	12	\odot
4.3.2Flavor branes134.3.3A possible brane intersection?144.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		4.3.1 Color branes	13	\frown
4.3.3 A possible brane intersection?144.4 Instantons154.5 Baryon vertices and t'Hooft monopoles154.6 Central charge165 Example in new class of $AdS_3 \times S^2$ geometries in 11D176 A new IIB $AdS_3 \times S^2 \times S^2$ solution197 Comments on supersymmetry218 Conclusions26A Hopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		4.3.2 Flavor branes	13	\mathbb{N}
4.4Instantons154.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		4.3.3 A possible brane intersection?	14	\bigcirc
4.5Baryon vertices and t'Hooft monopoles154.6Central charge165Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		4.4 Instantons	15	\square
4.6 Central charge165 Example in new class of $AdS_3 \times S^2$ geometries in 11D176 A new IIB $AdS_3 \times S^2 \times S^2$ solution197 Comments on supersymmetry218 Conclusions26A Hopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		4.5 Baryon vertices and t'Hooft monopoles	15	
5Example in new class of $AdS_3 \times S^2$ geometries in 11D176A new IIB $AdS_3 \times S^2 \times S^2$ solution197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27		4.6 Central charge	16	OI
6A new IIB $AdS_3 \times S^2 \times S^2$ solution19197Comments on supersymmetry218Conclusions26AHopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27	5 Example in new class of $AdS_3 \times S^2$ geometries in 11D		17	
7 Comments on supersymmetry218 Conclusions26A Hopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27	$6 \hspace{0.1 cm} ext{A new IIB} \hspace{0.1 cm} AdS_3 imes S^2 imes S^2 \hspace{0.1 cm} ext{solution}$		19	\mathbb{N}
8 Conclusions26A Hopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27	7 Comments on supersymmetry		21	
A Hopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$ 27	8 Conclusions			
	A Hopf-fibre T-duality for $AdS_3 imes S^3 imes S^3 imes S^1$			

1 Introduction

It is not surprising that supersymmetric $AdS_3 \times S^2$ solutions to 11D supergravity [1, 2] bear a striking resemblance to their $AdS_5 \times S^2$ counterparts [3]; obvious cosmetic differences, such as supersymmetry and G-structures,¹ are ultimately tied to dimensionality. In common, we note that both spacetimes possess manifest SU(2) isometries, dual to the R-symmetries of

¹Killing spinors, or supersymmetry variations, transform as a doublet under the SU(2) R-symmetry and are tensored with the Killing spinors of AdS_{d+1} , which have $2^{\frac{d}{2}}$ complex components.

the respective 2D $\mathcal{N} = (0, 4)$ [4, 5] and 4D $\mathcal{N} = 2$ [6, 7] SCFTs, and that supersymmetric geometries are in one-to-one correspondence with second-order PDEs. For the $\frac{1}{2}$ -BPS bubbling geometries of Lin, Lunin and Maldacena (LLM), one famously encounters the 3D continuous Toda equation [3], while a similar local analysis in [2] has revealed a 5D analogue for $\frac{1}{4}$ -BPS geometries:

$$y \,\partial_y \left(y^{-1} \partial_y J \right) = \mathrm{d}_4 \left(J \cdot \mathrm{d}_4 \mathrm{sech}^2 \zeta \right), \tag{1.1}$$

where the internal space exhibits SU(2)-structure.² Above ζ is a scalar depending on the 5D coordinates (y, x_i) , J is the Kähler-form of the 4D base and d₄ denotes the pull-back of the derivative to the base. The 4D base corresponds to an *almost Calabi-Yau* two-fold [8].

Finding explicit supersymmetric geometries is thus equivalent, at least locally, to solving these PDEs. Despite the difficulties, we have witnessed a growing number of $AdS_5 \times S^2$ geometries, and associated Toda solutions; starting with early constructions from gauged supergravity [9], through examples found directly in 11D [10],³ recently a large number of solutions have been constructed by exploiting an added isometry and a connection to electrostatics [7, 12–14]. More recently, exotic solutions without an electrostatic, or with only an emergent electrostatic description have been found [15, 16]. Relevant to this current work, it is noteworthy that the SU(2) non-Abelian T-dual of $AdS_5 \times S^5$ also corresponds to a solution in this class [17].

In contrast, little is known about solutions to (1.1). Given the current literature, if we eliminate geometries exhibiting more supersymmetry, which one can disguise as $AdS_3 \times S^2$ (see section 4 of [2]), there is no known $\frac{1}{4}$ -BPS geometry that solves (1.1). In this paper, after uplift to 11D, we identify the non-Abelian T-dual of $AdS_3 \times S^3 \times CY_2$ [17] as the first example in this class. Admittedly, this example solves (1.1) in the most trivial way, since $\partial_y J = d_4 \zeta = 0$. That being said, it should be borne in mind that the linear supersymmetry conditions are satisfied non-trivially. It is worth appreciating an obvious parallel to Abelian T-duality, where the uplifted geometry is an example of an SU(3)-structure manifold, namely Calabi-Yau.

Before proceeding, we touch upon the generality of (1.1). It is not clear if all supersymmetric $\frac{1}{4}$ -BPS $AdS_3 \times S^2$ solutions in 11D with SU(2)-structure satisfy (1.1). Indeed, the analysis of LLM made the simplifying assumption that there are no $AdS_5 \times S^2$ geometries with purely magnetic flux. Similarly, [2] precluded both purely electric and magnetic fluxes [2], a choice that is supported by AdS-limits of wrapped M5-brane geometries [1, 18]. For LLM, it can be explicitly shown that extra fluxes are inconsistent with supersymmetry [19]⁴ and an attempt at a more general analysis for $AdS_3 \times S^2$ geometries appeared in [22], which derives the supersymmetry conditions in all generality, but unfortunately fails to constrain the fluxes greatly. Using these conditions, one can show that the existence of a single chiral spinor internally, corresponding to SU(3)-structure,

 $^{^2\}mathrm{SU}(2)\text{-structure}$ in 6D is equivalent to two canonical $\mathrm{SU}(3)\text{-structures}.$

³The 11D solution can be dimensionally reduced and T-dualised, where it becomes a quotient of $AdS_5 \times S^5$. This provides no contradiction with a no-go result for $\frac{1}{2}$ -BPS AdS₅ in IIB ref. [11].

⁴Generalising the Killing spinor ansatz [20] allows one to also describe maximally supersymmetric 11D solutions or $\frac{1}{2}$ -BPS pp-waves, such as [21].

implies Calabi-Yau.⁵ For SU(2)-structure manifolds, we note that the non-Abelian T-dual of $AdS_3 \times S^3 \times CY_2$ fits neatly into the classification of [2]. In contrast, the non-Abelian T-dual of $AdS_3 \times S^3 \times S^3 \times S^1$ preserves the same supersymmetry, $\mathcal{N} = (0, 4)$ in 2D, yet falls outside this class, thus motivating future work to extract the more general class [24].

Non-Abelian T-duality has revealed itself as a powerful tool to construct explicit AdS solutions that seemed unreachable by other means. In this work we present some further examples. Interesting solutions generated this way⁶ are the only explicit AdS_6 solution to Type IIB supergravity constructed in [37]⁷ and the recent $\mathcal{N} = 2 AdS_4$ solution to M-theory with purely magnetic flux constructed in [42], which provides the only such explicit solution besides the Pernici-Sezgin background derived in the eighties [43]. Both these solutions may play an important role as gravity duals of, respectively, 5d fixed point theories arising from Type IIB brane configurations, probably from 7-branes as in [44] (see [45]), and of 3d SCFTs arising from M5-branes wrapped on 3d manifolds in the context of the 3d-3d correspondence [46]. In turn, the new AdS_3 backgrounds that we construct in this paper may provide the holographic duals of new 2D large $\mathcal{N} = (0, 4)$ field theories arising from D-brane intersections. Other AdS_3 backgrounds dual to $\mathcal{N} = (0, 2)$ 2D field theories haven been constructed recently in [28] (see also [47]) by compactifying on a 2D manifold the Klebanov-Witten background, combined with Abelian and non-Abelian T-dualities.

An essential difference with respect to its Abelian counterpart, is that non-Abelian T-duality has not been proved to be a symmetry of String Theory. In the context of the AdS/CFT correspondence one could thus expect new AdS backgrounds from known ones with very different dual CFTs. Furthermore, these CFTs are only guaranteed to exist in the strong coupling regime, since there is no reason to expect that the transformation should survive α' or 1/N corrections.

Even if the understanding of the CFT interpretation of the transformation is today very preliminary, some results point indeed in these directions. The non-Abelian T-dual of the $AdS_5 \times S^5$ background constructed in [17] has been shown for instance to belong to the family of $\mathcal{N} = 2$ Gaiotto-Maldacena geometries [7], proposed as duals of the, intrinsically strongly coupled, T_N Gaiotto theories [6]. Similarly, the non-Abelian T-dual of the $AdS_5 \times$ $T^{1,1}$ background [48] gives rise to an AdS_5 background [49] that belongs to the general class of $\mathcal{N} = 1$ solutions in [50, 51], whose dual CFTs generalize the so-called Sicilian quivers of [52], and are the $\mathcal{N} = 1$ analogues of the $\mathcal{N} = 2$ solutions in [6].

⁵A small caveat here is that one of the 6D spinors ϵ_{\pm} was assumed to be chiral, however the supersymmetry constraints on scalar bilinears are strong enough to ensure $\epsilon_{-} = -i\epsilon_{+}$. The Calabi-Yau conditions $dJ = d\Omega = 0$ then follow. We thank D. Tsimpis for raising this loop-hole.

 $^{^6\}mathrm{See}$ also [25-28] for further AdS solutions and [29-36] for a more varied sample of the NAT duality literature.

⁷Supersymmetry imposes severe constraints to the existence of AdS_6 solutions in ten and eleven dimensions [38, 39]. Prior to [37] the only known explicit solution to Type II supergravities was the Brandhuber and Oz background [41], which was shown to be the only possible such solution in (massive) IIA in [38]. Later [39] proved the non-existence of AdS_6 solutions in M-theory and derived the PDEs that such solutions must satisfy in Type IIB (see also [40]), to which the example in [37], constructed from the Brandhuber and Oz solution via non-Abelian T-duality, provides the only known explicit solution (besides the Abelian T-dual).

Some works have tried to explore in more depth the CFT realization of AdS backgrounds generated through non-Abelian T-duality in different dimensions [25, 45]–[28]. Its interplay with supersymmetry [53] and phenomenological properties of the dual CFTs, such as the type of branes generating the geometry, the behavior of universal quantities such as the free energy, or the entanglement entropy, the realization of baryon vertices, instantons, giant gravitons, are by now quite systematized (see [54]). Very recently, we have witnessed as well an exciting and novel application in the exchange of particles with vortices [55]. In this paper we will analyze some of these properties in the 2D holographic duals to the new AdS_3 backgrounds that we generate. We will see that they fit in the general picture observed in other dimensions.

Perhaps the most puzzling obstacle towards a precise CFT interpretation of non-Abelian T-duality is the fact that even if the group used to construct the non-Abelian T-dual background is compact, the original coordinates transforming under this group are replaced in the dual by coordinates living in its Lie algebra. Non-compact internal directions are thus generated, which are hard to interpret in the CFT. We will also encounter this problem for the backgrounds generated in this paper.

The paper is organized as follows. In section 2 we present the first explicit example of an $AdS_3 \times S^2$ geometry belonging to the general class of solutions [2]. This is constructed by uplifting the non-Abelian T-dual of $AdS_3 \times S^3 \times CY_2$ derived in [17] to 11D. In section 3 we recall the basic properties of the $AdS_3 \times S^3 \times S^3 \times S^1$ background that will be the basis of the new solutions that we present in sections 4, 5 and 6. In section 4 we construct the non-Abelian T-dual of this background with respect to a freely acting SU(2) on one of the S^3 . By exploring the solution we derive some properties of the associated dual CFT such as the central charge and the type of color and flavor branes from which it may arise. We suggest a possible explicit realization in terms of intersecting branes. In section 5 we construct one further solution through Abelian T-duality plus uplift to 11D from the previous one and show that it provides an explicit example of an $AdS_3 \times S^2$ geometry in 11D belonging to a new class that is beyond the ansatz in [2]. In section 6 we present a new $AdS_3 \times S^2 \times S^2$ solution to Type IIB obtained by further dualizing the solution in section 3 with respect to a freely acting SU(2) on the remaining S^3 . By analyzing the same brane configurations we argue that the field theory dual shares some common properties with the CFT dual to the original $AdS_3 \times S^3 \times S^3 \times S^1$ background but in a less symmetric fashion. In section 7 we analyze in detail the supersymmetries preserved by the different solutions that we construct. We show that the solutions constructed through non-Abelian T-duality from the $AdS_3 \times S^3 \times S^3 \times S^1$ background exhibit large $\mathcal{N} = (0, 4)$ supersymmetry. This is supported by the analysis of the central charges performed in sections 4 and 6. Section 8 contains our conclusions. Finally, in the appendix we study in detail the effect of Hopf-fibre T-duality in the $AdS_3 \times S^3 \times S^3 \times S^1$ background to further support our claims in the text concerning the isometry supergroup of our solutions.

2 $AdS_3 \times S^2$ geometries in 11D with SU(2)-structure

In this section we demonstrate that the non-Abelian T-dual of the D1-D5 near-horizon, a solution that was originally written down in [17], uplifts to 11D, where it provides the

first explicit example of a $\frac{1}{4}$ -BPS $AdS_3 \times S^2$ geometry with an internal SU(2)-structure manifold. We recall that this class has appeared in a series of classifications [1, 2, 18, 22], yet until now, not a single explicit example in this class was known. It is indeed pleasing to recognise that the chain of dualities that generates this new example is no more than a simple non-Abelian generalisation of a well-known mapping from the $AdS_3 \times S^3 \times CY_2$ geometry of Type IIB supergravity into the 11D supergravity class $AdS_3 \times S^2 \times CY_3$.⁸ It is worth noting that until relatively recently [17] (also [49]), the workings of this new mapping, which is made possible through non-Abelian T-duality, were also unknown.

We begin by reviewing the classification of ref. [22], which has an advantage over other approaches [1], since it uses local techniques and is thus guaranteed to capture all supersymmetric solutions. Moreover, this work also extends the ansatz of ref. [2] and dispenses with the need for an analytic continuation from $S^3 \times S^2$ to $AdS_3 \times S^2$. Based on symmetries, the general form for a supersymmetric spacetime of this type may be expressed as

$$ds_{11}^2 = e^{2\lambda} \left[\frac{1}{m^2} ds^2 (AdS_3) + e^{2A} ds^2 (S^2) + ds_6^2 \right],$$

$$G_4 = \operatorname{Vol}(AdS_3) \wedge \mathcal{A} + \operatorname{Vol}(S^2) \wedge \mathcal{H} + \mathcal{G},$$
(2.1)

where λ , A denote warp-factors depending on the coordinates of the 6D internal space and \mathcal{A}, \mathcal{H} and \mathcal{G} correspond to one, two and four-forms, respectively, with legs on the internal space. The constant m denotes the inverse radius of AdS_3 . The supersymmetry conditions, which are given in terms of differential conditions on spinor bilinears, further built from two a priori independent 6D spinors ϵ_{\pm} , can be found in [22].

Setting $\mathcal{A} = \mathcal{G} = 0$, one finds that only a particular linear combination, $\tilde{\epsilon} = \epsilon_+ \pm i\gamma_7\epsilon_$ appears in the effective 6D Killing spinor equations, allowing one to recover the work of [2]. In this simplifying case one can show that the internal space must be of the form [2, 22]

$$ds_6^2 = g_{ij} dx^i dx^j + e^{-6\lambda} \sec^2 \zeta dy^2 + \cos^2 \zeta (d\psi + P)^2$$
(2.2)

with P a one-form connection on the 4D base with metric g_{ij} . The SU(2)-structure is then specified by 2 one-forms, $K^1 \equiv \cos \zeta (d\psi + P)$, $K^2 \equiv e^{-3\lambda} \sec \zeta dy$, the Kähler-form, J, and the complex two-form, Ω , on the base.

The remaining two-form appearing in the field strength, G_4 , is fully determined by supersymmetry,

$$\mathcal{H} = -yJ - \frac{1}{2m}\partial_y(y\sin^2\zeta)dy \wedge (d\psi + P) -\frac{y}{m}\cos\zeta\sin\zeta d_4\zeta \wedge (d\psi + P) + \frac{y\cos^2\zeta}{2m}dP.$$
(2.3)

⁸See appendix B of ref. [56] for a concrete realisation of the (Abelian) duality chain.

The above class of geometries is subject to the supersymmetry conditions:

$$2my = e^{3\lambda} \sin \zeta,$$

$$e^{A} = \frac{\sin \zeta}{2m},$$

$$d(e^{3\lambda} \cos \zeta \Omega) = 0,$$

$$2md(e^{3\lambda+2A}J) = d_{4}P \wedge dy.$$
(2.4)

Details of how (1.1) is implied by these conditions can be found in [2].

In order to identify a solution in this class, we start by recalling the non-Abelian T-dual of $AdS_3 \times S^3 \times T^4$ [17], which provides a solution to massive IIA supergravity,

$$ds_{IIA}^{2} = ds^{2}(AdS_{3}) + d\rho^{2} + \frac{\rho^{2}}{1+\rho^{2}}ds^{2}(S^{2}) + ds^{2}(T^{4}),$$

$$B_{2} = \frac{\rho^{3}}{1+\rho^{2}}vol(S^{2}), \qquad \Phi = -\frac{1}{2}\ln(1+\rho^{2}),$$

$$m = 1, \qquad F_{2} = \frac{\rho^{3}}{1+\rho^{2}}vol(S^{2}),$$

$$F_{4} = vol(AdS_{3}) \wedge \rho d\rho + vol(T^{4}), \qquad (2.5)$$

where following [17], we have suppressed factors associated to radii for simplicity. As a consequence, the AdS_3 metric is normalised so that $R_{\mu\nu} = -\frac{1}{2}g_{\mu\nu}$, whereas S^2 is canonically normalised to unit radius.

We next perform two T-dualities along the T^4 , the coordinates of which we label, $x_1, \ldots x_4$. Performing T-dualities with respect to x_1 and x_2 , we can replace the Romans' mass, m = 1, with higher-dimensional forms, while leaving the NS sector unaltered. In addition to the NS two-form, the geometry is then supported by the following potentials from the RR sector,

$$C_{1} = \frac{1}{2} (x_{1} dx_{2} - x_{2} dx_{1} + x_{3} dx_{4} - x_{4} dx_{3}),$$

$$C_{3} = \frac{\rho^{3}}{1 + \rho^{2}} \operatorname{vol}(S^{2}) \wedge C_{1}.$$
(2.6)

We note that $dC_1 = J$, where J is the Kähler form on T^4 and the Bianchi for F_4 , namely $dF_4 = H_3 \wedge F_2$ is satisfied in a trivial way since $F_4 = dC_3 = B_2 \wedge J$. We can now uplift the solution on a circle to 11D by considering the standard Kaluza-Klein ansatz,

$$ds_{11}^{2} = (1+\rho^{2})^{\frac{1}{3}} \left[ds^{2} (AdS_{3}) + \frac{\rho^{2}}{1+\rho^{2}} ds^{2} (S^{2}) + d\rho^{2} + ds^{2} (T^{4}) \right] + (1+\rho^{2})^{-\frac{2}{3}} Dz^{2},$$

$$G_{4} = \operatorname{vol}(S^{2}) \left[\frac{\rho^{3}}{1+\rho^{2}} J + \frac{\rho^{2} (\rho^{2}+3)}{(1+\rho^{2})^{2}} d\rho \wedge Dz \right],$$
(2.7)

where we have defined $Dz \equiv dz + C_1$.

Adopting m = 2, so that normalisations for AdS_3 agree, and up to an overall sign in \mathcal{H} , which can be accommodated through the sign flip $\rho \leftrightarrow -\rho$, we find that the supersymmetry conditions (2.4) are satisfied once one identifies accordingly

$$y = \rho, \qquad e^{\lambda} = (1+\rho^2)^{\frac{1}{6}}, \qquad e^A = \frac{\rho}{(1+\rho^2)^{\frac{1}{2}}}, \qquad P = C_1,$$

$$J = dx_1 \wedge dx_2 + dx_3 \wedge dx_4,$$

$$\Omega = (dx_1 + idx_2) \wedge (dx_3 + idx_4).$$
(2.8)

Thus the non-Abelian T-dual plus 11D uplift of the D1-D5 near horizon fits in the classifications [1, 2, 18, 22]. It is easy to see that one can replace T^4 with K3 and the construction still holds. It is also easy to see that the above solution can be derived on the assumption that the base is Calabi-Yau and that λ, ζ only depend on y. Indeed, this is a requirement of the 6D SU(2)-structure manifold to be a complex manifold [22]. In this case, the supersymmetry conditions imply $e^{3\lambda} \cos \zeta$ is a constant. We can then solve for λ, ζ and A giving us the above solution.

Another interesting feature of the 11D solution is that in performing the classification exercise using Killing spinor bilinears [2, 22], one finds a U(1) isometry that emerges from the analysis for free. Often this U(1) corresponds to an R-symmetry, for example [3, 23], but in this setting, the relevant superconformal symmetry in 2D either corresponds to small superconformal symmetry with R-symmetry SU(2), or large superconformal symmetry with R-symmetry SU(2) × SU(2). There appears to be no place for a U(1) R-symmetry and it is an interesting feature of solutions fitting into the class of [2] that the U(1) is the M-theory circle and the Killing spinors are uncharged with respect to this direction.⁹

In the rest of this paper, we study non-Abelian T-duals of another well-known $\frac{1}{2}$ -BPS AdS_3 solution with $\mathcal{N} = (4, 4)$ supersymmetry, namely $AdS_3 \times S^3 \times S^3 \times S^1$, where we will find a new supersymmetric solution that does not fit into the class in [2].

3 The $AdS_3 \times S^3 \times S^3 \times S^1$ background with pure RR flux

In this section we recall the basic properties of the $AdS_3 \times S^3 \times S^3 \times S^1$ background [57]–[58], which will be the basis of our study in the following sections.

The $AdS_3 \times S^3 \times S^3 \times S^1$ background is a half-BPS solution of Type II string theory supported by NS5-brane and string flux. In this paper we will be interested in its realization in Type IIB where it is supported by D5 and D1-brane fluxes [59]. This description arises after compactifying on a circle the $AdS_3 \times S^3 \times S^3 \times \mathbb{R}$ near horizon geometry of a D1-D5-D5' system where the two stacks of D5-branes are orthogonal and intersect only along the line of the D1-branes [57, 60, 61]. How to implement the S^1 compactification has remained unclear (see [59]), and it has only been argued recently [62] that the \mathbb{R} instead of the S^1

⁹In 11D one can identify the two projection conditions to verify that supersymmetry is not enhanced. From CY_2 directions, we inherit $\Gamma^{6789}\eta = -\eta$, the rotation on the 11D spinor becomes $\eta = \exp[-\frac{1}{2}\tan^{-1}\left(\frac{1}{\rho}\right)\Gamma^{\chi\xi z}]\tilde{\eta}$. One finds the additional projector, $\Gamma^{\rho z 67}\tilde{\eta} = -\tilde{\eta}$, thus confirming that the 11D solution is indeed $\frac{1}{4}$ -BPS.

factor arising in the near horizon limit could just be an artefact of the smearing of the D1-branes on the transverse directions prior to taking the limit. This reference has also provided the explicit $\mathcal{N} = (4, 4)$ CFT realization conjectured in [58, 59, 63] for the field theory dual. This CFT arises as the infrared fixed point of the $\mathcal{N} = (0, 4)$ gauge theory living on the D1-D5-D5' intersecting D-branes.

The $AdS_3 \times S^3_+ \times S^3_- \times \mathbb{R}$ metric is given by

$$ds_{IIB}^2 = L^2 ds^2 (AdS_3) + R_+^2 ds^2 (S_+^3) + R_-^2 ds^2 (S_-^3) + dx^2$$
(3.1)

with

$$ds^{2}(AdS_{3}) = r^{2}(-dt^{2} + dx_{1}^{2}) + \frac{dr^{2}}{r^{2}}$$
(3.2)

in Poincaré coordinates. Plus, the background is supported by a single non trivial RR flux

$$F_3 = 2L^2 \operatorname{Vol}(AdS_3) + 2R_+^2 \operatorname{Vol}(S_+^3) + 2R_-^2 \operatorname{Vol}(S_-^3),$$
(3.3)

with Hodge dual

$$F_{7} = \left\{ 2L^{3} \operatorname{Vol}(AdS_{3}) \land \left(-\frac{R_{+}^{3}}{R_{-}} \operatorname{Vol}(S_{+}^{3}) + \frac{R_{-}^{3}}{R_{+}} \operatorname{Vol}(S_{-}^{3}) \right) + \frac{2R_{+}^{3}R_{-}^{3}}{L} \operatorname{Vol}(S_{+}^{3}) \land \operatorname{Vol}(S_{-}^{3}) \right\} \land \mathrm{d}x$$
(3.4)

We take $g_s = 1$ such that the dilaton is zero and Einstein's equations are satisfied only when

$$\frac{1}{L^2} = \frac{1}{R_+^2} + \frac{1}{R_-^2}.$$
(3.5)

This background has a large invariance under $SO(4)^+ \times SO(4)^-$ spatial rotations. Of these $SU(2)_R^+ \times SU(2)_R^-$ correspond to the R-symmetry group of the $\mathcal{N} = (0, 4)$ field theory living at the D1-D5-D5' intersection, and $SU(2)_L^+ \times SU(2)_L^-$ to a global symmetry. The field theory has gauge group $U(N_1)$, with N_1 the number of D1-branes, and a global symmetry $SU(N_5^+) \times SU(N_5^-)$, with N_5^+ and N_5^- the number of D5 and D5' branes. The two R-symmetries give rise to two current algebras at levels depending on the background charges, and to a large $\mathcal{N} = (4, 4)$ superconformal symmetry in the infra-red [58, 59, 62, 63].

The study of the supergravity solution allows to derive properties of the dual field theory that we will be able to mimic after the non-Abelian T-duality transformation. In the next subsections we analyze the quantized charges, some brane configurations such as baryon vertices and 't Hooft monopoles, and the central charge associated to the $AdS_3 \times S^3 \times S^3 \times S^1$ background.

3.1 Quantized charges

The F_7 and F_3 fluxes generate D1 and D5-brane charges given by:

$$N_1 = \frac{1}{2\pi\kappa_{10}^2 T_1} \int (-F_7) = \frac{R_+^3 R_-^3 \delta x}{8L\pi^2} , \qquad (3.6)$$

where δx is the length of the x-direction interval, which should be chosen such that N_1 is quantized, and

$$N_5^+ = \frac{1}{2\pi\kappa_{10}^2 T_5} \int_{S_-^3} (-F_3) = R_-^2, \qquad N_5^- = \frac{1}{2\pi\kappa_{10}^2 T_5} \int_{S_+^3} (-F_3) = R_+^2, \qquad (3.7)$$

which should also be quantized. Accordingly, one can find D1 and D5 BPS solutions. The D1 are extended along the $\{t, x_1\}$ directions and couple to the potential

$$C_2 = L^2 r^2 \mathrm{d}t \wedge \mathrm{d}x_1 \,. \tag{3.8}$$

Changing coordinates to

$$\begin{cases} r = r_{+} r_{-} \\ x = \frac{R_{+}^{2}}{\sqrt{R_{+}^{2} + R_{-}^{2}}} \log r_{+} - \frac{R_{-}^{2}}{\sqrt{R_{+}^{2} + R_{-}^{2}}} \log r_{-}, \end{cases}$$
(3.9)

the metric becomes the near horizon limit of the intersecting D1-D5-D5' configuration [57, 60, 61]:

$$N_5^+ D5 : 012345$$

$$N_5^- D5' : 016789$$

$$N_1 D1 : 01 (3.10)$$

with $dx_2^2 + \cdots + dx_5^2 = dr_+^2 + r_+^2 ds^2(S_+^3)$, $dx_6^2 + \cdots + dx_9^2 = dr_-^2 + r_-^2 ds^2(S_-^3)$, with the D1-branes smeared on these directions:

$$ds_{IIB}^2 = L^2 r_+^2 r_-^2 (-dt^2 + dx_1^2) + R_+^2 \frac{dr_+^2}{r_+^2} + R_-^2 \frac{dr_-^2}{r_-^2} + R_+^2 ds^2 (S_+^3) + R_-^2 ds^2 (S_-^3) . \quad (3.11)$$

The BPS D5-branes are then found lying on the $(t, x_1, r_+, S^3_+), (t, x_1, r_-, S^3_-)$ directions.

The 2D $\mathcal{N} = (0,4)$ gauge theory living on the worldvolume of the D1-branes and intersecting D5-branes has been identified recently in [62]. A key role is played by the chiral fermions of the D5-D5' strings that lie at the intersection. Quite remarkably the central charge of the $\mathcal{N} = (4,4)$ CFT to which this theory flows in the infra-red has been shown to coincide with the central charge of the supergravity solution, that we review in subsection 3.4.

3.2 Instantons

The previous configuration of D5, D5' branes joined in a single manifold, where the D1branes lie, admits a Higgs branch where the D1-branes are realized as instantons in the D5-branes [58]. One can indeed compute the quadratic fluctuations of the D5-branes to obtain the effective YM coupling:

$$S_{\rm fluc}^{\rm D5} = -\int \frac{1}{g_{\rm D5}^2} F_{\mu\nu}^2 \qquad \text{with} \qquad \frac{1}{g_{\rm D5}^2} = \frac{L^2 r_+^2 r_-^2}{4(2\pi)^3}$$
(3.12)

and check that the DBI action of the D1-branes satisfies

$$S_{\rm DBI}^{\rm D1} = -\int \frac{16\pi^2}{g_{\rm D5}^2} \tag{3.13}$$

as expected for an instantonic brane.

3.3 Baryon vertices and 't Hooft monopoles

A D7-brane wrapped on $S^3_+ \times S^3_- \times S^1$ realizes a baryon vertex in the $AdS_3 \times S^3_+ \times S^3_- \times S^1$ geometry, since it develops a tadpole of N_1 units, as it is inferred from its CS action:

$$S_{\rm CS}^{\rm D7} = 2\pi T_7 \int C_6 \wedge F = -2\pi T_7 \int_{S_+^3 \times S_-^3 \times S^1} F_7 \int dt A_t = -N_1 \int dt A_t , \qquad (3.14)$$

where δx is taken to satisfy that N_1 is an integer as in (3.6).

Similarly, there are two t'Hooft monopoles associated to the ranks of the two flavor groups that are realized in the $AdS_3 \times S^3_+ \times S^3_- \times S^1$ background as D3-branes wrapping the S^3_{\pm} . The corresponding Chern-Simons terms show that these branes have tadpoles of N_5^{\pm} units that should be cancelled with the addition of these numbers of fundamental strings:

$$S_{\rm CS}^{\rm D3^{\pm}} = -2\pi T_3 \int_{S_{\pm}^3} F_3 \int dt A_t = N_5^{\mp} \int dt A_t \ . \tag{3.15}$$

3.4 Central charge

The central charge associated to the $AdS_3 \times S^3_+ \times S^3_- \times S^1$ background can be computed using the Brown-Henneaux formula [64], giving [58, 59]:

$$c = 2N_1 \frac{N_5^+ N_5^-}{N_5^+ + N_5^-}.$$
(3.16)

This expression agrees with the central charge for a large $\mathcal{N} = (4, 4)$ CFT with affine $\mathrm{SU}(2)^{\pm}$ current algebras at levels k^{\pm} : $c = 2k^{+}k^{-}/(k^{+} + k^{-})$ [65], with $k^{\pm} = N_1 N_5^{\pm}$. A strong check of the validity of the $\mathcal{N} = (0, 4)$ field theory in the D1-branes proposed in [62] is that it correctly reproduces (3.16) at the infrared fixed point (see also [58]).

4 Non-Abelian T-dual $AdS_3 \times S^3 \times S^2$ solution in IIA

In this section we dualize the $AdS_3 \times S^3_+ \times S^3_- \times S^1$ solution with respect to the $SU(2)_L^$ acting on the S^3_- . This dualization was reported in [53] to produce a new AdS_3 solution preserving 16 supercharges. As we shall demonstrate in section 7 [53] overlooked an extra implied condition and the preserved supersymmetry is in fact 8 supercharges. The solution thus preserves large $\mathcal{N} = (0, 4)$ supersymmetry in 2D. In this section we present a detailed study of the geometry and infer some properties about the field theory interpretation of this solution.

4.1 Background

Applying the general rules in [66] (see also [53]) we find a dual metric

$$ds_{IIA}^2 = L^2 ds^2 (AdS_3) + R_+^2 ds^2 (S_+^3) + \frac{4}{R_-^2} \left(d\rho^2 + \frac{R_-^6 \rho^2}{64\Delta} \left(d\chi^2 + \sin^2 \chi d\xi^2 \right) \right) + dx^2, \quad (4.1)$$

where

$$\Delta = \frac{R_{-}^{6} + 16R_{-}^{2}\rho^{2}}{64}.$$
(4.2)

The dual dilaton is given by

$$e^{-2\Phi} = \Delta, \tag{4.3}$$

while the NS 2-form is simply

$$B_2 = \frac{R_-^2 \rho^3}{4\Delta} \operatorname{Vol}(S^2) \tag{4.4}$$

where S^2 refers to the 2-sphere parametrised by $0 \le \chi \le \pi$, $0 \le \xi < 2\pi$ in (4.1). The dual RR-sector is given by

$$\begin{split} m &= \frac{R_{-}^{2}}{4}, \\ \hat{F}_{2} &= 0, \\ \hat{F}_{4} &= -\frac{R_{-}^{3}}{4LR_{+}} \left(L^{4} \text{Vol}(AdS_{3}) + R_{+}^{4} \text{Vol}(S_{+}^{3}) \right) \wedge dx \\ &\quad + 2\rho \left(L^{2} \text{Vol}(AdS_{3}) + R_{+}^{2} \text{Vol}(S_{+}^{3}) \right) \wedge d\rho, \\ \hat{F}_{6} &= -2L^{2} \rho^{2} \text{Vol}(AdS_{3}) \wedge \text{Vol}(S^{2}) \wedge d\rho - 2R_{+}^{2} \rho^{2} \text{Vol}(S_{+}^{3}) \wedge \text{Vol}(S^{2}) \wedge d\rho, \\ \hat{F}_{8} &= -\frac{2L^{3} R_{+}^{3} \rho}{R_{-}} \text{Vol}(AdS_{3}) \wedge \text{Vol}(S_{+}^{3}) \wedge dx \wedge d\rho, \\ \hat{F}_{10} &= \frac{2L^{3} R_{+}^{3} \rho^{2}}{R_{-}} \text{Vol}(AdS_{3}) \wedge \text{Vol}(S_{+}^{3}) \wedge \text{Vol}(S^{2}) \wedge dx \wedge d\rho. \end{split}$$

Here $\hat{F} = Fe^{-B_2}$ and $F_p = dC_{p-1} - H_3 \wedge C_{p-3}$. Page charges will be computed from these \hat{F} according to $d * \hat{F} = *j^{\text{Page}}$.

Applying the results in [66] this background is guaranteed to satisfy the (massive) IIA supergravity equations of motion. Given that the S^3 on which we have dualized has constant radius the non-Abelian T-dual solution is also automatically non-singular. An open problem though is the range of the new coordinate ρ , which as a result of the non-Abelian T-duality transformation lives in \mathbb{R}^+ .

The generation of non-compact directions under non-Abelian T-duality is indeed a generic feature that does not occur under its Abelian counterpart. In the last case the extension of the transformation beyond tree level in string perturbation theory determines uniquely the global properties of the, in principle non-compact, coordinate that replaces the dualized U(1) direction. How to extend non-Abelian T-duality beyond tree level is however a long standing open problem (see [67] for more details), and as a result we are lacking a general mechanism that allows to compactify the new coordinates. For freely acting SU(2) examples we need to account in particular for the presence of the non-compact ρ direction in the dual internal geometry, which poses a problem to its CFT interpretation, where we can expect operators with continuous conformal dimensions. Note that in the $AdS_3 \times S^2 \times S^1$ duals under consideration in this paper one cannot hope that the same mechanism that should be at work for compactifying the \mathbb{R} factor arising in the original $AdS_3 \times S^3 \times S^3 \times S^3 \times \mathbb{R}$ geometry should be applicable. As argued in [62], the \mathbb{R} instead of the S^1 factor arising in the near horizon limit could be due to the smearing of the D1-branes on the transverse directions prior to taking the limit, and could presumably be avoided with a supergravity solution describing localized branes. This is not directly applicable to our situation because ρ is not an isometric direction.

Previous approaches in the recent non-Abelian T-duality literature have tried to infer global properties through imposing consistency to the dual CFT [45, 54]. We will also follow this approach in this paper. We should start noticing that the new AdS_3 metric described by (4.1) is perfectly regular for all $\rho \in [0, \infty)$, with the 3d space replacing the S_-^3 in the original background becoming \mathbb{R}^3 for small ρ and $\mathbb{R} \times S^2$ for large ρ . As shown in [45, 54] the definition of large gauge transformations in the dual geometry can give however non-trivial information about its global properties.

4.2 Large gauge transformations

The relevance of large gauge transformations is linked to the existence of non-trivial 2-cycles in the geometry, where

$$\frac{1}{4\pi^2} \left| \int_{2-\text{cycle}} B_2 \right| \in [0,1) \,. \tag{4.5}$$

In our non-singular metric we can only guarantee the existence of a non-trivial S^2 for large ρ . For finite ρ and given the absence of any global information, we will resort to the most general situation in which the cycle remains non-trivial and we need to care about large gauge transformations. We will see that consistency of the CFT in this most general situation will lead to the condition of vanishing large gauge transformations, which is compatible with the original situation in which the two-cycle may in fact be trivial at finite ρ .

Assuming the existence of a non-trivial two-cycle at finite ρ , the ρ dependence of B_2 in (4.4) implies that large gauge transformations must be defined such that (4.5) is satisfied as we move in this direction. This implies that for $\rho \in [\rho_n, \rho_{n+1}]$ with ρ_n determined by $16\rho_n^3/(R_-^4 + 16\rho_n^2) = n\pi$, B_2 must be given by

$$B_2 = \left(\frac{R_-^2 \rho^3}{4\Delta} - n\pi\right) \operatorname{Vol}(S^2) \,. \tag{4.6}$$

The fluxes from which the Page charges are computed then change in the different intervals to

$$F_2 \rightarrow F_2 + n\pi F_0 \operatorname{Vol}(S^2)$$

$$\hat{F}_6 \rightarrow \hat{F}_6 + n\pi \hat{F}_4 \wedge \operatorname{Vol}(S^2),$$
(4.7)

which will affect the values of the Page charges that we compute next.

4.3 Quantized charges

The transformation of the RR fluxes under non-Abelian T-duality implies that the D1 color branes of the original background transform into D2-branes extended on $\{t, x_1, \rho\}$ and D4branes on $\{t, x_1, \rho, S^2\}$. Analogously, the D5 flavor branes wrapped on the S_-^3 are mapped into D2 and D4 branes wrapped on $\{t, x_1, r_-\}$ and $\{t, x_1, r_-, S^2\}$ respectively, and the D5 transverse to the S_-^3 are transformed into D6 and D8 branes wrapped on $\{t, x_1, r_+, S_+^3, \rho\}$ and $\{t, x_1, r_+, S_+^3, \rho, S^2\}$, respectively. We show in this section that there are quantized charges in the non-Abelian T-dual background that can be associated to these branes.

4.3.1 Color branes

It is possible to define N_2 and N_4 quantized charges in the dual background that should be associated to D2 and D4 color branes:

$$N_4 = \frac{1}{2\pi\kappa_{10}^2 T_4} \int_{S_+^3 \times S^1} \hat{F}_4 = \frac{R_+^3 R_-^3}{16\pi L} \,\delta x \;, \tag{4.8}$$

$$N_2 = \frac{1}{2\pi\kappa_{10}^2 T_2} \int_{S^3_+ \times S^2 \times S^1} \hat{F}_6 = nN_4 , \qquad (4.9)$$

where n is the parameter labeling large gauge transformations. (4.9) is the value of the D2 charge in the $\rho \in [\rho_n, \rho_{n+1}]$ interval. Note that, as it seems to be quite generic under non-Abelian T-duality, the condition imposed on the geometry by (4.8) is different, and in fact incompatible, from the one that the original background satisfied, given by (3.6). A re-quantization must thus be done in the new background.

Let us now analyze the condition (4.9). We first see that for zero n the charge associated to the D2-branes vanishes. Second, as we change interval, N_2 undergoes a transformation, $N_2 \rightarrow N_2 - N_4$, that is very reminiscent of Seiberg duality [68]. This was proposed in [54] as a way to relate the CFTs dual to the solution as we move in ρ . As stressed in [28]¹⁰ this cannot be however the full story since there is a change in the number of degrees of freedom as we move in ρ . This is explicit in the holographic free energies. The precise realization in the CFT of the running of ρ remains at the very heart of our full understanding of the interplay between non-Abelian T-duality and AdS/CFT. We hope we will be able to report progress in this direction in future publications.

For the particular background considered in this paper it is only possible to find BPS color and flavor branes when n = 0. In particular, color branes are D4-branes wrapped on the $\{t, x_1, \rho, S^2\}$ directions. Thus, we will take the view that ρ is restricted to the fundamental region $[0, \rho_1]$, with ρ_1 satisfying $16\rho_1^3/(R_-^4 + 16\rho_1^2) = \pi$. Choosing to end the geometry at a regular point presents however other problems, now for the geometry, where extra localized sources should be included. It was proposed in [28] that at these transition points new gauge groups would be added to the CFT through an "unhiggsing" mechanism not associated to an energy scale. Given that this mechanism relies in the existence of large gauge transformations it does not seem applicable to our background. A full understanding of the "unhiggsing" mechanism and its precise realization in the absence of an energy scale remains as an interesting open problem.

4.3.2 Flavor branes

Let us now examine flavor branes in the dual background. We find the following quantized charges in the dual background that should be associated to flavor branes:

$$N_8^f = 2\pi F_0 = \frac{\pi}{2} R_-^2, \qquad N_6^f = \frac{1}{2\pi \kappa_{10}^2 T_6} \int_{S^2} \hat{F}_2 = n N_8^f, \qquad (4.10)$$

$$N_4^f = \frac{1}{2\pi\kappa_{10}^2 T_4} \int_{S_+^3} \int_{\rho_n}^{\rho_{n+1}} \mathrm{d}\rho \,\hat{F}_4 \,, \qquad N_2^f = \frac{1}{2\pi\kappa_{10}^2 T_2} \left| \int_{S_+^3 \times S^2} \int_{\rho_n}^{\rho_{n+1}} \mathrm{d}\rho \,\hat{F}_6 \right| \,. \tag{4.11}$$

¹⁰We would also like to acknowledge fruitful conversations with D. Rodríguez-Gómez on this issue.

Here we have made explicit the interval on which the ρ direction has to be integrated and we have not restricted ourselves to vanishing large gauge transformations.

The first two charges in (4.10) correspond to the D8 and D6 flavor branes that originate on the N_5^+ D5-branes of the original background. Thus, our expectation is to find BPS D8 wrapped on $\{t, x_1, r_+, S_+^3, \rho, S^2\}$ and BPS D6 wrapped on $\{t, x_1, r_+, S_+^3, \rho\}$. However, as for the color branes, we also find that the D6 are never BPS unless $R_- = 0$ and that the D8 (anti-D8 in our conventions) are BPS only in the absence of large gauge transformations. This is again suggestive of a dual background where large gauge transformations are not possible. In the absence of these the D5 flavor branes give rise to just D8 flavor branes in the dual background.

The D5' branes of the original background give rise in turn to D4-branes wrapped on $\{t, x_1, r_-, S^2\}$ and D2-branes wrapped on $\{t, x_1, r_-\}$, which turn out to be BPS only when located at $\rho = 0$. In this position however both the DBI and CS actions of the D4 vanish, leaving just D2-branes as candidate flavor branes.

4.3.3 A possible brane intersection?

Summarizing, we have found that there are only BPS color and flavor branes in the absence of large gauge transformations, in which case there is only one color or flavor brane in the non-Abelian T-dual background associated to each color or flavor brane of the original theory. Note that this is essentially different from previous examples in the literature (for instance [45, 54]) where both types of color and flavor branes were guaranteed to exist for all n. We argue in the conclusions that this could be explained by the absence of non-trivial 2-cycles in our particular dual geometry.¹¹

We have shown that the D1-branes are replaced by D4-branes wrapped on $\{t, x_1, \rho, S^2\}$ and the D5 and D5' flavor branes are replaced by D8-branes wrapped on $\{t, x_1, r_+, S^3_+, \rho, S^2\}$ and D2-branes wrapped on $\{t, x_1, r_-\}$, respectively. This is summarized pictorially as



Here we have also indicated the brane that turns out not to occur as a BPS configuration even if expected a priori from the analysis of the fluxes.

Note that precisely a $D1 \rightarrow D4$, $D5 \rightarrow D8$, $D5' \rightarrow D2$ map is what one would have obtained after (Abelian) T-dualizing the D1, D5, D5' system along three directions transverse to the D1 and the D5 and longitudinal to the D5'. This suggests a dual geometry coming out as the near horizon limit of the brane intersection:

$N_8^f \ D8:$	012345789	
$N_{2}^{f} D2:$	016	
$N_4 D4:$	01789	(4.12)

¹¹In the examples in [45, 54] non-trivial S^2 were guaranteed to exist due to the presence of singularities.

In this brane intersection the SO(4)⁺ × SO(4)⁻ symmetry of the original field theory is replaced by a SO(4)⁺ × SU(2) symmetry. Of this, SU(2)⁺_R × SU(2)_R would correspond to the R-symmetry group of a large $\mathcal{N} = (0, 4)$ field theory living at the intersection, and the remaining SU(2)⁺_L to a global symmetry. This is consistent with the central charge computation in subsection 4.6 and with the supersymmetry analysis in section 7 (see also the appendix). The field theory would moreover have gauge group U(N₄) and a global symmetry SU(N^f₈) × SU(N^f₂). Some field theory configurations that we present next are compatible with this brane realization.

4.4 Instantons

A very similar calculation to the one in subsection 3.2 shows that the D4 color branes can be realized as instantons in the D8 flavor branes. In this case

$$S_{\rm fluc}^{\rm D8} = -\int \frac{1}{g_{\rm D8}^2} F_{\mu\nu}^2 \qquad \text{with} \qquad \frac{1}{g_{\rm D8}^2} = \frac{L^2 r_+^2 r_-^2 \rho^2}{(2\pi)^6}$$
(4.13)

and the DBI action of the D4-branes satisfies

$$S_{\rm DBI}^{\rm D4} = -\int \frac{16\pi^2}{g_{\rm D8}^2} \,, \tag{4.14}$$

as expected for an instantonic brane.

4.5 Baryon vertices and t'Hooft monopoles

The original D7-brane baryon vertex configuration is mapped after the duality into a D4-brane wrapped on $S^3_+ \times S^1$ and a D6-brane wrapped on $S^3_+ \times S^1 \times S^2$. The second one however has vanishing tadpole charge in the absence of large gauge transformations, given that

$$S_{\rm CS}^{\rm D6} = 2\pi T_6 \int (C_5 - B_2 \wedge C_3) \wedge F = -2\pi T_6 \int_{S_+^3 \times S^1 \times S^2} \hat{F}_6 \int dt A_t$$

= $-nN_4 \int dt A_t$. (4.15)

For the D4 wrapped on $S^3_+ \times S^1$ we find

$$S_{\rm CS}^{\rm D4} = -2\pi T_4 \int_{S_+^3 \times S^1} \hat{F}_4 \int dt A_t = -N_4 \int dt A_t \ . \tag{4.16}$$

As a result, there is one candidate for baryon vertex in the non-Abelian T-dual background, realized as a D4-brane wrapped on $S^3_+ \times S^1$.

Similarly, in the original background we had $D3^{\pm}$ -branes wrapped on S_{\pm}^3 t'Hooft monopoles whose tadpole charges were given by the ranks of the flavor groups. The $D3^+$ is mapped after the duality into a D4 wrapped on $\{S_{+}^3, \rho\}$ and a D6 wrapped on $\{S_{+}^3, \rho, S^2\}$ with tadpole charges

$$S_{\rm CS}^{\rm D4} = -2\pi T_4 \int_{S_+^3} \int_{\rho_n}^{\rho_{n+1}} \mathrm{d}\rho \hat{F}_4 \int \mathrm{d}t A_t = N_4^f \int \mathrm{d}t A_t \,, \tag{4.17}$$

and

$$S_{\rm CS}^{\rm D6} = -2\pi T_6 \int_{S_+^3 \times S^2} \int_{\rho_n}^{\rho_{n+1}} \mathrm{d}\rho \hat{F}_6 \int \mathrm{d}t A_t = -N_2^f \int \mathrm{d}t A_t \;. \tag{4.18}$$

Given that N_4^f is not associated to a BPS D4-brane in the absence of large gauge transformations it is sensible to also not associate to it a 't Hooft monopole configuration. The D6-brane thus remains as the candidate 't Hooft monopole, with tadpole charge given by the charge of the D2 flavor brane.

The $D3^-$ 't Hooft monopole of the original background is in turn mapped into a D0brane and a D2-brane wrapped on the S^2 . We indeed find that these branes have tadpoles with charges

$$S_{\rm CS}^{\rm D0} = -2\pi T_0 \, m \int dt A_t = -N_8 \int dt A_t \,, \qquad (4.19)$$

and

$$S_{\rm CS}^{\rm D2} = -2\pi T_2 \int_{S^2} \hat{F}_2 \int dt A_t = -nN_8 \int dt A_t \ . \tag{4.20}$$

Clearly the second brane does not carry any tadpole charge in the absence of large gauge transformations. Thus, only the D0-brane remains as candidate 't Hooft monopole, with tadpole charge given by the charge of the D8 flavor brane.

Consistently with our previous results we find two 't Hooft monopole configurations in the dual background whose tadpole charges are given by the charges of the two D2 and D8 dual flavor branes.

4.6 Central charge

Finally in this section we compute the central charge of the dual supergravity solution. We show that as in the original theory it is possible to define two R-symmetry currents from which

$$c = 2\frac{k^+k^-}{k^++k^-} \tag{4.21}$$

as in [65]. We take the general expressions in [70], to which the reader is referred for more details.

Rewriting the original IIB metric as

$$ds_{str}^{2} = \alpha(r)\beta(r)dr^{2} + \alpha(r)dx_{1,1}^{2} + g_{ij}dy^{i}dy^{j}, \qquad (4.22)$$

we read off

$$\alpha = L^2 r^2, \quad \beta = \frac{1}{r^4}.$$
 (4.23)

Substituting these in the expressions for the internal volume 12 and r-dependent quantity κ we obtain

$$V_{\rm int} = \int d^7 y \, e^{-2\Phi} \, \sqrt{\det(g_{ij})} = 4\pi^4 R_+^3 R_-^3 \delta x \tag{4.24}$$

$$\kappa = V_{\rm int}^2 \, \alpha(r) = V_{\rm int}^2 Lr.$$

¹²We have generalized these as in [54] to account for the y-dependent dilaton in the dual background.

The central charge of the original theory can then be computed as

$$c \sim \beta^{d/2} \kappa^{3d/2} (\kappa')^{-d} \tag{4.25}$$

where d = 1 in our case and $\kappa' \equiv d\kappa/dr$, to obtain

$$c = \frac{1}{(2\pi)^2} L R_+^3 R_-^3 \delta x = 2L^2 N_1 = 2N_1 \frac{N_5^+ N_5^-}{N_5^+ + N_5^-}, \qquad (4.26)$$

where we have substituted δx from (3.6), $L^2 = N_5^+ N_5^- / (N_5^+ + N_5^-)$, and have fixed the normalization factor in (4.25) to agree with the central charge computed in [65], with $k^{\pm} = N_1 N_5^{\pm}$.

Similarly for the non-Abelian T-dual solution we find

$$\tilde{V}_{\text{int}} = \int \mathrm{d}^7 y \, e^{-2\Phi} \sqrt{\det(\tilde{g}_{ij})} = \frac{1}{3} \pi^6 R_+^3 R_-^3 \delta x \,, \qquad (4.27)$$

from where, taking the same normalization factor as in (4.26),

$$\tilde{c} = \frac{1}{48} L R_+^3 R_-^3 \,\delta x = \frac{\pi}{3} L^2 N_4 = 2N_4 \frac{N_2^f N_8^f}{3N_2^f + N_8^f} \,. \tag{4.28}$$

Note that it is not possible to bring the dual central charge into the form (4.21) unless we change the normalization factor. Indeed, the change in the internal volume produced by the non-Abelian T-duality transformation translates generically into central charges differing by constant factors (see for instance [45, 49]). Still, up to this normalization factor, the central charge is of the form (4.21), with two levels that depend differently on the products of color and flavor charges. We denote these by $k^+ = 3N_4N_2^f$, $k^- = N_4N_8^f$. Note that consistently with the form of the dual geometry, the $+ \leftrightarrow -$ symmetry of the original background has now disappeared. It would be interesting to understand the field theory origin of the values for the two levels that we obtain. The central charge is thus compatible with a large $\mathcal{N} = (0, 4)$ superconformal theory dual to our solution.

5 Example in new class of $AdS_3 \times S^2$ geometries in 11D

In this section, following section 2, we manipulate the massive IIA solution of the previous section by performing two Abelian T-dualities, in the process rendering it as a solution to massless IIA supergravity. We will then be in a position to uplift the solution to 11D supergravity. As we detail in section 7, while not entirely obvious, there are indeed two manifest global U(1) isometries, namely the overall transverse x-direction and the remaining Hopf-fibre, which becomes a global symmetry after the initial T-duality.

Performing the T-duality on the x-direction, the NS sector is unchanged, while the T-dual RR sector becomes

$$F_{1} = \frac{R_{-}^{2}}{4} dx,$$

$$F_{3} = \frac{4R_{-}^{2} \rho^{3}}{16\rho^{2} + R_{-}^{4}} \sin \chi d\chi \wedge d\xi \wedge dx - \frac{R_{-}^{3}}{4R_{+}L} [L^{4} \text{Vol}(AdS_{3}) + R_{+}^{4} \text{Vol}(S_{+}^{3})],$$

$$F_{5} = [2L^{2} \operatorname{Vol}(AdS_{3}) + 2R_{+}^{2} \operatorname{Vol}(S_{+}^{3})] \wedge \rho \, \mathrm{d}\rho \wedge \mathrm{d}x - \frac{4R_{-}^{3} \rho^{3}}{L \, R_{+}(16\rho^{2} + R_{-}^{4})} [L^{4} \operatorname{Vol}(AdS_{3}) + R_{+}^{4} \operatorname{Vol}(S_{+}^{3})] \sin \chi \mathrm{d}\chi \wedge \mathrm{d}\xi.$$
(5.1)

We can further T-dualise on the Hopf-fibre direction, which we parametrise through the coordinate ψ , to get the massless IIA solution:

$$d\hat{s}^{2} = L^{2}ds^{2}(AdS_{3}) + \frac{R_{+}^{2}}{4}(d\theta^{2} + \sin^{2}\theta d\phi^{2}) + dx^{2} + \frac{4}{R_{+}^{2}}d\psi^{2} + \frac{4}{R_{-}^{2}}d\rho^{2} + \frac{4R_{-}^{2}\rho^{2}}{16\rho^{2} + R_{-}^{4}}(d\chi^{2} + \sin^{2}\chi d\xi^{2}),$$

$$\hat{\rho} = \frac{16\rho^{3}}{16\rho^{3}} = h_{+} h_$$

$$B = \frac{1}{16\rho^2 + R_-^4} \sin \chi d\chi \wedge d\xi + \cos \theta d\phi \wedge d\psi,$$

$$e^{-2\hat{\Phi}} = \frac{R_-^2 R_+^2}{256} (16\rho^2 + R_-^4),$$

$$F_2 = -\frac{R_-^2}{4} dx \wedge d\psi - \frac{R_-^3 R_+^3}{32L} \sin \theta d\theta \wedge d\phi,$$

$$F_4 = -\frac{4R_-^2 \rho^3}{16\rho^2 + R_-^4} \sin \chi d\chi \wedge d\xi \wedge dx \wedge d\psi + \frac{R_-^3 L^3}{4R_+} \operatorname{vol}(AdS_3) \wedge d\psi,$$

$$+\frac{\rho R_+^2}{4} \sin \theta d\theta \wedge d\phi \wedge d\rho \wedge dx - \frac{R_-^3 R_+^3 \rho^3}{2L(16\rho^2 + R_-^4)} \sin \theta d\theta \wedge d\phi \wedge \sin \chi d\chi \wedge d\xi.$$

Uplifting to 11D, we get:

$$ds_{11}^{2} = e^{2\lambda} \left[L^{2} ds^{2} (AdS_{3}) + e^{2A} (d\chi^{2} + \sin^{2}\chi d\xi^{2}) + ds_{6}^{2} \right],$$

$$G_{4} = -\frac{4R_{-}^{2}\rho^{3}}{16\rho^{2} + R_{-}^{4}} \sin\chi d\chi \wedge d\xi \wedge dx \wedge d\psi + \frac{R_{-}^{3}L^{3}}{4R_{+}} \operatorname{vol}(AdS_{3}) \wedge d\psi,$$

$$+ \frac{R_{+}^{2}}{4} \sin\theta d\theta \wedge d\phi \wedge \rho d\rho \wedge dx - \frac{R_{-}^{3}R_{+}^{3}\rho^{3}}{2L(16\rho^{2} + R_{-}^{4})} \sin\theta d\theta \wedge d\phi \wedge \sin\chi d\chi \wedge d\xi$$

$$+ \left[\frac{16\rho^{2}(16\rho^{2} + 3R_{-}^{4})}{(16\rho^{2} + R_{-}^{4})^{2}} d\rho \wedge \sin\chi d\chi \wedge d\xi - \sin\theta d\theta \wedge d\phi \wedge d\psi \right] \wedge \mathrm{D}z, \qquad (5.2)$$

where we have defined

$$e^{2\lambda} = e^{-\frac{2}{3}\hat{\Phi}}, \qquad e^{2A} = \frac{4R_{-}^{2}\rho^{2}}{16\rho^{2} + R_{-}^{4}},$$

$$ds_{6}^{2} = \frac{R_{+}^{2}}{4}(d\theta^{2} + \sin^{2}\theta d\phi^{2}) + dx^{2} + \frac{4}{R_{+}^{2}}d\psi^{2} + \frac{4}{R_{-}^{2}}d\rho^{2} + \frac{256}{R_{-}^{2}R_{+}^{2}(16\rho^{2} + R_{-}^{4})}Dz^{2},$$

$$Dz \equiv dz + C_{1},$$

$$C_{1} = -\frac{R_{-}^{2}}{8}(xd\psi - \psi dx) + \frac{R_{-}^{3}R_{+}^{3}}{32L}\cos\theta d\phi.$$
(5.3)

One can check that the Bianchi identity and the equations of motion are satisfied. As we argue in section 7, this uplifted geometry is expected to be $\frac{1}{4}$ -BPS. What is particularly

interesting about this uplift is that the internal manifold exhibits SU(2)-structure, yet it is beyond the scope of the ansatz in [2], since \mathcal{A} and \mathcal{G} in (2.1) are clearly non-zero. This opens up the possibility that we can read off the relation between the 6D Killing spinors appearing in the more general classification [22], feed them into supersymmetry conditions and identify a more general class of supersymmetric $AdS_3 \times S^2$ solutions in 11D supergravity with SU(2)-structure manifolds. One can then use the supersymmetry conditions to find further explicit solutions, some of which may be, in contrast to non-Abelian T-duals, compact. We hope to report on this in future work [24].

6 A new IIB $AdS_3 \times S^2 \times S^2$ solution

In this section we dualize once more the $AdS_3 \times S^3 \times S^2$ solution of section 4 with respect to the SU(2)⁺_L acting on the S^3_+ . We show that this dualization produces a new AdS_3 solution, this time in Type IIB. As we discuss in section 7, and further in appendix A, the new solution we generate will be $\frac{1}{4}$ -BPS and still preserve $\mathcal{N} = (0, 4)$ supersymmetry in 2D.¹³

The new background is given by

$$ds_{IIB}^{2} = L^{2}ds^{2}(AdS_{3}) + dx^{2} + \frac{4}{R_{+}^{2}} \left(d\rho_{+}^{2} + \frac{R_{+}^{6}\rho_{+}^{2}}{64\Delta_{+}} (d\chi_{+}^{2} + \sin^{2}\chi_{+}d\xi_{+}^{2}) \right) + \frac{4}{R_{-}^{2}} \left(d\rho_{-}^{2} + \frac{R_{-}^{6}\rho_{-}^{2}}{64\Delta_{-}} (d\chi_{-}^{2} + \sin^{2}\chi_{-}d\xi_{-}^{2}) \right),$$
(6.1)

where we have introduced $(\rho_{-}, \chi_{-}, \xi_{-})$ to equal our previous (ρ, χ, ξ) after the first dualization on S^{3}_{-} , and $(\rho_{+}, \chi_{+}, \xi_{+})$ to denote the new coordinates arising after the second dualization on S^{3}_{+} . Δ_{\pm} are given by

$$\Delta_{\pm} = \frac{R_{\pm}^6 + 16R_{\pm}^2\rho_{\pm}^2}{64}.$$
(6.2)

The corresponding dilaton is just

$$e^{-2\Phi} = \Delta_+ \Delta_-, \tag{6.3}$$

and the new NS-NS 2-form is given by

$$B_2 = \frac{R_+^2 \rho_+^3}{4\Delta_+} \operatorname{Vol}(S_+^2) + \frac{R_-^2 \rho_-^3}{4\Delta_-} \operatorname{Vol}(S_-^2)$$
(6.4)

where S_{\pm}^2 are the 2-spheres parameterized by (χ_{\pm}, ξ_{\pm}) , respectively. The dual RR sector is given by¹⁴

$$\hat{F}_{1} = \frac{R_{-}^{3}R_{+}^{3}}{32L}dx + \frac{1}{4}R_{-}^{2}\rho_{+}d\rho_{+} - \frac{1}{4}R_{+}^{2}\rho_{-}d\rho_{-},$$
$$\hat{F}_{3} = \frac{1}{4}R_{+}^{2}\rho_{-}^{2}d\rho_{-}\wedge \operatorname{Vol}(S_{-}^{2}) - \frac{1}{4}R_{-}^{2}\rho_{+}^{2}d\rho_{+}\wedge \operatorname{Vol}(S_{+}^{2})$$

 $^{^{13}}$ The only subtlety here would appear to be the correct identification of the global SU(2) with respect to which one T-dualises.

 $^{^{14}\}mathrm{Note}$ that these are the fluxes associated to the Page charges.

$$\bar{F}_{5} = 2L^{2}\rho_{-}\rho_{+}\operatorname{Vol}(AdS_{3}) \wedge d\rho_{-} \wedge d\rho_{+} \\
- \frac{L^{3}}{4}\operatorname{Vol}(AdS_{3}) \wedge dx \wedge \left(\frac{R_{+}^{3}}{R_{-}}\rho_{-} d\rho_{-} + \frac{R_{-}^{3}}{R_{+}}\rho_{+} d\rho_{+}\right), \\
\bar{F}_{7} = \frac{L^{3}}{4}\operatorname{Vol}(AdS_{3}) \wedge dx \wedge \left(\frac{R_{+}^{3}}{R_{-}}\rho_{-}^{2}d\rho_{-} \wedge \operatorname{Vol}(S_{-}^{2}) + \frac{R_{-}^{3}}{R_{+}}\rho_{+}^{2}d\rho_{+} \wedge \operatorname{Vol}(S_{+}^{2})\right), \\
- 2L^{2}\rho_{-}\rho_{+}\operatorname{Vol}(AdS_{3}) \wedge d\rho_{-} \wedge d\rho_{+} \wedge \left(\rho_{-}\operatorname{Vol}(S_{-}^{2}) + \rho_{+}\operatorname{Vol}(S_{+}^{2})\right), \\
\bar{F}_{9} = 2L^{2}\rho_{-}^{2}\rho_{+}^{2}\operatorname{Vol}(AdS_{3}) \wedge d\rho_{-} \wedge \operatorname{Vol}(S_{-}^{2}) \wedge d\rho_{+} \wedge \operatorname{Vol}(S_{+}^{2}).$$
(6.5)

This solution satisfies the IIB equations of motion and preserves eight supersymmetries. As our previous massive AdS_3 solution, it is perfectly regular, with the range of the new \mathbb{R}^+ direction, ρ_+ , also to be determined. As we did after the first dualization, we link the running of both non-compact directions ρ_{\pm} to large gauge transformations in this background. The ranges of these coordinates must then be divided in $[\rho_{\pm(n_{\pm})}, \rho_{\pm(n_{\pm}+1)}]$ intervals in which large gauge transformations with n_{\pm} parameters on the non-trivial S_{\pm}^2 cycles ensure that B_2 lies in the fundamental region.

The field theory analysis that can be made from this supergravity solution follows very closely the one we made for the previous massive AdS_3 solution, so we will omit the details. As in that case each of the brane configurations that we described in section 2 is mapped to a single brane configuration in the dual for $n_{\pm} = 0$, and no dual configurations exist otherwise unless $R_{-} = R_{+} = 0$. For $n_{\pm} = 0$ we find the brane configurations:

- Color branes: D7 on $\{t, x_1, \rho_-, S^2_-, \rho_+, S^2_+\}$
- Flavor branes: D5 on $\{t, x_1, r_-, \rho_+, S_+^2\}$ (at $\rho_- = 0$)

D5' on
$$\{t, x_1, r_+, \rho_-, S_-^2\}$$
 (at $\rho_+ = 0$)

This can be summarized pictorially as



where we have crossed out the branes not occurring as BPS configurations but expected a priori from the analysis of the fluxes. The charges of the surviving BPS D7, D5 and D5' are:

$$N_7 = +\frac{1}{2\kappa_{10}^2 T_7} \int_{S^1} \hat{F}_1 = \frac{R_+^3 R_-^3}{32L} \delta x$$
(6.6)

$$N_5^+ = -\frac{1}{2\kappa_{10}^2 T_5} \int_{S_-^2} \int_0^{\rho_{-(1)}} \mathrm{d}\rho_- \hat{F}_3 \tag{6.7}$$

$$N_5^- = +\frac{1}{2\kappa_{10}^2 T_5} \int_{S_+^2} \int_0^{\rho_{+(1)}} \mathrm{d}\rho_+ \,\hat{F}_3 \tag{6.8}$$

where once again δx is the hand-set length of the x-direction. $\rho_{\pm(1)}$ satisfy $16\rho_{\pm(1)}^3/(R_{\pm}^4 + 16\rho_{\pm(1)}^2) = \pi$. Hence, a candidate brane intersection is:

 $N_5^+ D5: 013456$ $N_5^- D5': 012789$ $N_7 D7: 01345789$ (6.9)

which realizes the $\mathrm{SU}(2)^+ \times \mathrm{SU}(2)^-$ symmetries of the background. As shown in section 7 (see also the appendix) these correspond to R-symmetries in the dual theory. Thus, the dual field theory is still a large $\mathcal{N} = (0, 4)$ SCFT. The field theory living at the intersection would have gauge group $\mathrm{U}(N_7)$ and a global symmetry $\mathrm{SU}(N_5^+) \times \mathrm{SU}(N_5^-)$. Consistently with this picture we also have:

- Baryon vertices: D1 on $\{t, S^1\}$ with tadpole charge N_7
- 't Hooft monopoles: D3[±] on $\{t, \rho_{\pm}, S_{\pm}^2\}$ with tadpole charge N_5^{\mp}
- Central charge:

$$c = \frac{2}{3}N_7 \frac{N_5^+ N_5^-}{N_5^+ + N_5^-} \tag{6.10}$$

This form for the central charge agrees with a large $\mathcal{N} = (0, 4)$ dual CFT with affine $\mathrm{SU}(2)^{\pm}$ current algebras at levels $k^{\pm} = N_7 N_5^{\pm}$, even if with a different overall factor compared to [65]. This is consistent with the supersymmetry analysis. Together with the analysis of brane configurations this suggests a dual field theory in which D7 branes substitute the D1-branes of the original field theory dual to the $AdS_3 \times S_4^3 \times S_-^3 \times S^1$ solution. In this theory the global $\mathrm{SU}(2)_+^L \times \mathrm{SU}(2)_-^L$ symmetries have disappeared. It would be interesting to see if one can indeed derive these properties from the brane intersection given by (6.9).

7 Comments on supersymmetry

In this section we comment on the number of supersymmetries the various solutions to 10D Type II supergravity preserve. To make the text self-contained, we start by recalling our supersymmetry conventions [66, 71]. The fermionic supersymmetry variations for Type IIA and Type IIB supergravity are respectively

$$\delta\lambda = \frac{1}{2} \partial \!\!\!/ \Phi \eta - \frac{1}{24} I \!\!\!/_3 \sigma_3 \eta + \frac{1}{8} e^{\Phi} \left[5m\sigma_1 + \frac{3}{2} I \!\!\!/_2 i \sigma_2 + \frac{1}{24} I \!\!\!/_4 \sigma_1 \right] \eta,$$

$$\delta\Psi_\mu = \nabla_\mu \eta - \frac{1}{8} H_{3\,\mu\nu\rho} \Gamma^{\nu\rho} \sigma_3 + \frac{1}{8} e^{\Phi} \left[m\sigma_1 + \frac{1}{2} I \!\!\!/_2 i \sigma_2 + \frac{1}{24} I \!\!\!/_4 \sigma_1 \right] \Gamma_\mu \eta, \qquad (7.1)$$

and

$$\delta\lambda = \frac{1}{2}\partial \!\!\!/ \Phi \eta - \frac{1}{24} \not \!\!\!/ H_3 \sigma_3 \eta + \frac{1}{2} e^{\Phi} \left[\not \!\!\!/ F_1 i \sigma_2 + \frac{1}{12} \not \!\!\!/ F_3 \sigma_1 \right] \eta,$$

$$\delta\Psi_{\mu} = \nabla_{\mu} \eta - \frac{1}{8} H_3 {}_{\mu\nu\rho} \Gamma^{\nu\rho} \sigma_3 - \frac{1}{8} e^{\Phi} \left[\not \!\!\!/ F_1 i \sigma_2 + \frac{1}{6} \not \!\!\!/ F_3 \sigma_1 + \frac{1}{240} \not \!\!\!/ F_5 i \sigma_2 \right] \Gamma_{\mu} \eta, \qquad (7.2)$$

where λ denotes the dilatinos, Ψ_{μ} the gravitinos and η is a Majorana-Weyl spinor

$$\eta = \begin{pmatrix} \epsilon_+\\ \epsilon_- \end{pmatrix}. \tag{7.3}$$

The supersymmetry preserved by the non-Abelian T-dual of $AdS_3 \times S^3 \times CY_2$ is welldocumented [17, 66] and analysis leads to the conclusion that half the supersymmetry is broken in the transformation. Therefore, for the geometries exhibited in section 2, all solutions preserve eight supersymmetries, or $\mathcal{N} = (0, 4)$ supersymmetry in 2D. We have noted that the 11D uplift fits into the classification of [2] and further demonstrated that supersymmetry is not enhanced beyond $\frac{1}{4}$ -BPS in 11D, thus providing the first concrete example in the class of [2].

For the geometry $AdS_3 \times S^3 \times S^3 \times S^1$, supersymmetry breaking is not a foregone conclusion. To see why this may be the case, we recall that the geometry $AdS_3 \times S^3 \times$ $S^3 \times S^1$ possesses an SU(2) × SU(2) R-symmetry, yet is manifestly SO(4) × SO(4)-invariant. Therefore, it could be expected that a judicious choice of the T-duality SU(2) factor would result in a geometry preserving the same amount of supersymmetry as the original solution. This intuition is based on ref. [37], where T-duality with respect to a global SU(2) isometry generated a surprising new supersymmetric AdS_6 solution to IIB supergravity.

Here we correct statements in the literature¹⁵ and show that picking out a left or rightacting SU(2) isometry from one of the three-spheres leads to broken supersymmetry in an analogous fashion to $AdS_3 \times S^3 \times CY_2$ non-Abelian T-duals. For completeness, we do this in two ways, uncovering a consistent picture.

Firstly, and most easily, we can import the findings of ref. [66]. We recall for spacetimes with SO(4) isometry - with generalisations to SU(2) isometry [53] - that supersymmetry breaking is encoded in a single condition, namely (3.11) of ref. [66],

$$\left[-\frac{1}{2R_{-}}\Gamma^{\chi\xi}\sigma_{3} - \frac{1}{4R_{-}}\Gamma^{\chi\xi\rho}i\sigma_{2} - \frac{1}{4}\left(\frac{1}{L}\Gamma^{012} + \frac{1}{R_{+}}\Gamma^{678}\right)\sigma_{1}\right]\tilde{\eta} = 0,$$
(7.4)

where, assuming we T-dualise from the IIB form for the geometry, $\tilde{\eta}$ is related by a factor to the Killing spinor of IIA supergravity η ,

$$\tilde{\eta} \equiv e^{-X} \eta = \exp\left(\frac{1}{2} \tan^{-1}\left(\frac{R_{-}^2}{4\rho}\right) \Gamma^{\chi\xi} \sigma_3\right) \eta.$$
(7.5)

Using (3.5), we can rewrite these conditions as:

$$\begin{bmatrix} -\frac{1}{R_{-}}\Gamma^{\chi\xi} + \frac{1}{L}\Gamma^{012\rho} + \frac{1}{R_{+}}\Gamma^{678\rho} \end{bmatrix} \tilde{\epsilon}_{+} = 0, \\ \begin{bmatrix} \frac{1}{R_{-}}\Gamma^{\rho\chi\xi} - \frac{1}{L}\Gamma^{012} - \frac{1}{R_{+}}\Gamma^{678} \end{bmatrix} \tilde{\epsilon}_{-} = 0,$$
(7.6)

 $^{^{15}}$ It was initially reported in ref. [53] that an application of non-Abelian T-duality to an SU(2) factor in one of the SO(4) isometries resulted in a T-dual preserving sixteen supersymmetries. The analysis of ref. [53] failed to take account of an additional condition, which breaks supersymmetry to eight.

and further using (3.14) of ref. [66], which in this case reads,

$$\tilde{\epsilon}_{+} = \Gamma^{\rho} \epsilon_{+}, \quad \tilde{\epsilon}_{-} = -\epsilon_{-}, \quad \Gamma^{\rho\chi\xi} = -\Gamma^{345}, \tag{7.7}$$

we recover what turns out to be the original projection condition of the IIB geometry (3.1)

$$\left[\frac{1}{L}\Gamma^{012} + \frac{1}{R_{+}}\Gamma^{345} + \frac{1}{R_{-}}\Gamma^{678}\right]\eta = 0.$$
(7.8)

We observe that squaring this expression, we recover (3.5). We note also that in the process of redefining the spinors, the chirality of $\tilde{\epsilon}_+$ is flipped so that it now corresponds to a Killing spinor of Type IIB supergravity. On its own, this projection condition would suggest the background is $\frac{1}{2}$ -BPS, however we also find that the following identification is also implied

$$\epsilon_+ = \epsilon_-. \tag{7.9}$$

This constitutes an additional condition, which breaks supersymmetry to $\frac{1}{4}$ -BPS, or eight supersymmetries.

To develop a better understanding of what has just happened, it is also useful to explicitly work out the Killing spinors for the original solution (3.1). Following a calculation similar to ref. [61], except translated into our conventions, and making use of the projection condition (7.8), which falls out from the analysis, we can determine the precise form of the Killing spinors in their original IIB setting:

$$\epsilon_{+} = \left[r^{\frac{1}{2}} + r^{-\frac{1}{2}} (t \Gamma_{0}^{2} + x \Gamma_{1}^{2}) \right] (\alpha_{1} + \Omega \beta_{1}) + r^{-\frac{1}{2}} (\Omega \alpha_{2} + \beta_{2}),$$

$$\epsilon_{-} = \left[r^{\frac{1}{2}} + r^{-\frac{1}{2}} (t \Gamma_{0}^{2} + x \Gamma_{1}^{2}) \right] (\alpha_{1} - \Omega \beta_{1}) - r^{-\frac{1}{2}} (\Omega \alpha_{2} - \beta_{2}),$$
(7.10)

where we have defined the constant spinors, $\Gamma^{01}\alpha_i = \alpha_i$, $\Gamma^{01}\beta_i = -\beta_i$, and the matrix,

$$\Omega = e^{-\frac{1}{2}\psi_1\Gamma^{34}} e^{-\frac{1}{2}\theta_1\Gamma^{53}} e^{-\frac{1}{2}\phi_1\Gamma^{34}} e^{-\frac{1}{2}\psi_2\Gamma^{67}} e^{-\frac{1}{2}\theta_2\Gamma^{86}} e^{-\frac{1}{2}\phi_2\Gamma^{67}}, \tag{7.11}$$

where the angular dependence follows for the explicit form of left-invariant one-forms τ_{α} , which satisfy $d\tau_{\alpha} = \frac{1}{2} \epsilon_{\alpha\beta\gamma} \tau^{\beta} \wedge \tau^{\gamma}$. The existence of Poincaré supersymmetries of both chirality with respect to Γ^{01} indicates that supersymmetry is $\mathcal{N} = (4, 4)$ in 2D.

We now can appreciate that the identification (7.9) is a by-product of the fact that all Killing spinors with angular dependence get projected out under the non-Abelian T-duality. It is worth noting that a single Hopf-fibre T-duality also results in the same supersymmetry breaking, although one can consider a linear combination of the Hopf-fibres, which preserves additional supersymmetries [72].¹⁶

This final observation that angular dependence gets projected out presents us with a small puzzle. Namely, how can the loss of angular dependence be reconciled with $\mathcal{N} = (0, 4)$ supersymmetry, which requires, at a very least, the geometric realisation of an associated

¹⁶That the T-dual geometry in this special case must preserve twelve supersymmetries, and not the generic eight, can be most easily seen by resorting to the Kosmann spinorial-Lie-derivative [73]. One can then use the powerful result in ref. [53] that the supersymmetries uncharged under the T-duality direction are preserved.

SU(2) R-symmetry? To answer this question, we need to recall that an SU(2) transformation on a round three-sphere results in a residual S^2 factor in the metric. This then is one candidate SU(2) R-symmetry. As we shall appreciate later, the Killing spinors of the non-Abelian T-dual also have dependence on SU(2)_R of the remaining three-sphere. This suggests the presence of large $\mathcal{N} = (0, 4)$ supersymmetry where the corresponding isometry group is $D(2, 1|\gamma) \times SL(2, \mathbb{R}) \times SU(2)$, which as we explain in the appendix, is analogous to the Abelian T-dual, i.e. the geometry $AdS_3 \times S^3 \times S^2 \times T^2$.

In a bid to make this work self-contained, we now explicitly check that the residual S^2 becomes the SU(2) R-symmetry, that the remaining SO(4) has an SU(2)_L global symmetry and that supersymmetry is indeed $\mathcal{N} = (0, 4)$, as claimed. To do so, we solve the Killing spinor equations for IIA supergravity in the T-dual geometry.

We begin by introducing a frame for the remaining three-sphere,

$$ds^{2}(S_{+}^{3}) = \frac{1}{4} [(d\psi + \cos\theta d\phi)^{2} + d\theta^{2} + \sin^{2}\theta d\phi^{2}].$$
(7.12)

We next introduce the natural dreibein, $e^6 = \frac{R_+}{2} (d\psi + \cos\theta d\phi)$, $e^7 = \frac{R_+}{2} d\theta$, $e^8 = \frac{R_+}{2} \sin\theta d\phi$ and reverse the overall sign of the RR sector relative to (4.5), so we can import results from ref. [66], where expressions are given in terms of spherical coordinates, which are best suited to the current example. We note that $H_3 = dB_2$ has no legs along the ψ -direction, so the gravitino variation in this direction simply reads:

$$e^{-X}\delta\Psi_{6} = \frac{2}{R_{+}}\partial_{\psi}\tilde{\eta} + \frac{1}{2R_{+}}\Gamma^{78}\tilde{\eta} + \frac{e^{-2X}}{R_{-}\sqrt{16\rho^{2} + R_{-}^{4}}} \left[-\frac{R_{-}^{2}}{4}\sigma_{1} - \rho\Gamma^{\chi\xi}i\sigma_{2} -\rho\left(\frac{R_{-}}{L}\Gamma^{012\rho} + \frac{R_{-}}{R_{+}}\Gamma^{678\rho}\right)\sigma_{1} - \frac{R_{-}^{3}}{4}\left(\frac{1}{R_{+}}\Gamma^{012\rho} + \frac{1}{L}\Gamma^{678\rho}\right)\sigma_{1}\right]\Gamma_{6}\tilde{\eta}, (7.13)$$

where we have multiplied by the matrix e^{-X} and redefined the original Killing spinor as in (7.5). The Killing spinor $\tilde{\eta}$ is a IIA spinor satisfying the projection conditions

$$\left(\frac{R_{-}}{L}\Gamma^{012\chi\xi}i\sigma_{2} + \frac{R_{-}}{R_{+}}\Gamma^{678\chi\xi}i\sigma_{2}\right)\tilde{\eta} = \tilde{\eta},$$

$$\Gamma^{\rho}\sigma_{1}\tilde{\eta} = -\tilde{\eta},$$
(7.14)

hopefully making it obvious, through the appearance of two projection conditions, that the number of preserved supersymmetries is eight.

Bearing in mind that $\tilde{\eta}$ is comprised of Majorana-Weyl spinors of opposite chirality, we can dualise gamma matrices as follows

$$\Gamma^{012x}\sigma_1\tilde{\eta} = \Gamma^{678\rho\chi\xi}i\sigma_2\tilde{\eta}, \quad \Gamma^{678x}\sigma_1\tilde{\eta} = \Gamma^{012\rho\chi\xi}i\sigma_2\tilde{\eta}.$$
(7.15)

Then using the above expressions, one can rewrite (7.13) as

$$e^{-X}\delta\Psi_{6} = \frac{2}{R_{+}}\partial_{\psi}\tilde{\eta} + \frac{1}{2R_{+}}\Gamma^{78}\tilde{\eta} + \frac{e^{-2X}}{R_{+}\sqrt{16\rho^{2} + R_{-}^{4}}}\Gamma^{78}\left[-2\rho + \frac{R_{-}^{2}}{2}\Gamma^{\chi\xi}\sigma_{3}\right]\tilde{\eta}.$$
 (7.16)

Finally, we insert the expression for e^{-2X} ,

$$e^{-2X} = \frac{1}{\sqrt{16\rho^2 + R_-^4}} (4\rho + R_-^2 \Gamma^{\chi\xi} \sigma_3), \qquad (7.17)$$

to reach the conclusion that $\partial_{\psi}\eta = \partial_{\psi}(e^X\tilde{\eta}) = 0$, so after an SU(2) transformation the Killing spinors are independent of the Hopf-fibre, meaning that we can Abelian T-dualise later with respect to this direction. Similar calculations for the θ and ϕ -directions show that the Killing spinors also do not depend on these. We therefore see in an explicit fashion that the second three-sphere is now comprised of a global SU(2)_L symmetry, yet with the Killing spinors still dependent on SU(2)_R. In analogy with the Abelian case, we have an $D(2|1, \gamma) \times SL(2, \mathbb{R}) \times SU(2)$ symmetry algebra.

To extract the R-symmetry dependence on the residual S^2 , we consider $e^{-X}\delta\Psi_{\alpha}$, where $\alpha \in \{\chi, \xi\}$. We find

$$e^{-X}\delta\Psi_{\chi} = \frac{\sqrt{16\rho^{2} + R_{-}^{4}}}{2R_{-}\rho}\partial_{\chi}\tilde{\eta} + \frac{e^{-2X}}{(16\rho^{2} + R_{-}^{4})}\left(-\frac{R_{-}^{5}}{4\rho}\Gamma^{\chi}\sigma_{1} + \frac{(16\rho^{2} + 3R_{-}^{4})}{2R_{-}}\Gamma^{\xi}i\sigma_{2}\right)\tilde{\eta} + \frac{1}{R_{-}\sqrt{16\rho^{2} + R_{-}^{4}}}\left(-\frac{R_{-}^{2}}{2}\Gamma^{\chi}\sigma_{1} + 2\rho\Gamma^{\xi}i\sigma_{2}\right)\tilde{\eta}, \\ = \frac{\sqrt{16\rho^{2} + R_{-}^{4}}}{2R_{-}\rho}\left(\partial_{\chi}\tilde{\eta} + \frac{1}{2}\Gamma^{\xi}i\sigma_{2}\right)\tilde{\eta},$$
(7.18)

where in the second line we have expanded e^{-2X} . A similar calculation for $e^{-X}\delta\Psi_{\xi}$, after simplifications leads to

$$e^{-X}\delta\Psi_{\xi} = \frac{\sqrt{16\rho^2 + R_{-}^4}}{2R_{-}\rho} \left(\frac{1}{\sin\chi}\partial_{\xi}\tilde{\eta} + \frac{1}{2}\frac{\cos\chi}{\sin\chi}\Gamma^{\xi\chi} - \frac{1}{2}\Gamma^{\chi}i\sigma_2\right)\tilde{\eta}.$$
 (7.19)

Up to the inclusion of the Killing spinors for AdS_3 , we can then write the explicit form for the IIA Killing spinor

$$\eta = e^X e^{-\frac{1}{2}\chi\Gamma^{\xi}i\sigma_2} e^{-\frac{1}{2}\xi\Gamma^{\chi\xi}}\tilde{\eta}_{AdS_3}$$
(7.20)

where $\tilde{\eta}_{AdS_3}$ denotes the Killing spinors for AdS_3 ,

$$\nabla_{\mu}\tilde{\eta} = \frac{1}{2}\gamma_{3}\Gamma_{\mu}\tilde{\eta},\tag{7.21}$$

where we have defined $\gamma_3 \equiv \Gamma^{012}$. A calculation similar to appendix A, then shows that supersymmetry is indeed $\mathcal{N} = (0, 4)$. Similar calculations to above show that the dilatino variation vanishes. Again these results are all expected and follow from the analysis presented in [66], and more generally [53].

To go from the massive IIA solution of section 4 to the massless solution in section 5, we perform two T-dualities with respect to both the overall transverse direction x and the Hopf-fibre of the remaining three-sphere. As we have argued, both correspond to global U(1) isometries and it is expected that supersymmetry will be preserved. As one further final check that this is indeed the case, we record some of the gravitino variations after these two Abelian T-dualities. The gravitino variations in the x-direction and ψ -direction, notably those featuring in the T-duality, are respectively

$$e^{-X}\delta\Psi_x = \frac{1}{4L}\Gamma^{\theta\phi\chi\xi} \left[-\frac{L}{R_-}\Gamma^{\theta\phi\chi\psi} + \frac{L}{R_+}\Gamma^{\rho\chi\xi x} - 1 \right] \sigma_1\tilde{\eta}, \tag{7.22}$$

and

$$e^{-X}\delta\Psi_{\psi} = \frac{1}{2R_{+}}\Gamma^{\theta\phi}\sigma_{3}\left[\Gamma^{\rho x\psi}i\sigma_{2}+1\right]\tilde{\eta} + \frac{1}{4L}\Gamma^{\psi\theta\phi\chi\xi}\left[-\frac{L}{R_{-}}\Gamma^{\theta\phi x\psi} + \frac{L}{R_{+}}\Gamma^{\rho\chi\xi x} - 1\right]\sigma_{1}\tilde{\eta},$$
(7.23)

leading to good, commuting projection conditions. Furthermore, up to a redefinition in $\tilde{\epsilon}_+$, namely $\tilde{\epsilon}_+ \to -\Gamma^{x\psi}\tilde{\epsilon}_+$, with ϵ_- unchanged so it maintains its chirality, these projection conditions can be mapped back to (7.14), so we see that they are consistent. Yet again, by analogy with the Abelian T-duals discussed in the appendix, the isometry group for this geometry is expected to be $D(2|1,\gamma) \times \mathrm{SL}(2,\mathbb{R})$.

8 Conclusions

Non-Abelian T-duality is a symmetry of the equations of motion of type II supergravity. This has been shown explicitly for SO(4)-invariant spacetimes via dimensional reduction [66], results of which featured prominently in this current work. For spacetimes with less symmetry, e. g. the class of Bianchi IX spacetimes with SU(2) isometry, partial results exist [53, 74], but given the number of examples explored to date, it is safe to assume that the non-Abelian T-duality procedure with RR fluxes outlined in [17], and generalised to larger non-Abelian groups in [75], will take Type II supergravity solutions into each other.

We have made use of this solution-generating property in this paper to provide sample geometries for a class of $\frac{1}{4}$ -BPS $AdS_3 \times S^2$ spacetimes in 11D supergravity, where the internal space is an SU(2)-structure manifold. Despite a number of studies asserting that the class exists [1, 18, 22], most notably the classification in [2], there was no explicit example known. Not only have we demonstrated that the non-Abelian T-dual of the wellknown geometry $AdS_3 \times S^3 \times CY_2$ provides an example in this class, we have exhibited non-Abelian T-duals of a related geometry, $AdS_3 \times S^3 \times S^3 \times S^1$, which fall outside this class. This suggests that the general supersymmetry conditions of ref. [22] can be mined further to extract a larger class of supersymmetric solutions based on SU(2)-structure manifolds, thus extending the [2] class. It may be hoped that the non-Abelian T-duals, despite being manifestly non-compact, may serve to identify compact solutions via ansatz when the full class of supersymmetric $AdS_3 \times S^2$ solutions of 11D supergravity are identified.

On a related note, the $\frac{1}{4}$ -BPS AdS_3 solutions we generate involve a Romans' mass. Therefore, they will serve as a test of an ongoing program of work classifying the AdS solutions of massive IIA supergravity [38, 76, 77]. Furthermore, it may be interesting to consider non-Abelian T-duals of general $AdS_3 \times S^3 \times S^3 \times \Sigma_2$ solutions to 11D supergravity [78], where Σ_2 is a Riemann surface.

We have discussed some properties of the field theories associated to the $AdS_3 \times S^3 \times S^2$ and $AdS_3 \times S^2 \times S^2$ backgrounds that we construct with an aim at testing the general ideas on the CFT interpretation of non-Abelian T-duals in [54] (see also [45]). We have seen that as in previous examples there seems to be a doubling of charges after the transformation. In our AdS_3 cases however the branes responsible for the extra charges turn out to be supersymmetric only in the absence of large gauge transformations, in which case the extra charges vanish. The absence of large gauge transformations can be explained in turn either by the non-existence of non-trivial 2-cycles in the dual geometry at finite ρ , or, else, by a geometry terminating at a regular point. As in [45] the termination of the geometry at a regular point is intimately related to the depletion of the rank of one of the gauge groups.

An important piece of information about the CFT duals to the new solutions comes from the analysis of their central charges. We have shown that as in the original theory it is possible to define two R-symmetry currents from which the central charges exhibit the expected $c \sim k^+k^-/(k^+ + k^-)$ behaviour for a large $\mathcal{N} = (0, 4)$ superconformal algebra, in full agreement with the supersymmetry properties of the solutions.

Acknowledgments

We would like to thank Özgür Kelekci, Nakwoo Kim, Andrea Prinsloo, Martin Roček, Diego Rodríguez-Gómez and Dimitrios Tsimpis for useful discussions. The work of Y.L., N.T.M. and J.M. has been partially supported by the COST Action MP1210 "The String Theory Universe". Y.L. and J.M. are partially supported by the Spanish Ministry of Economy and Competitiveness grant FPA2012-35043-C02-02. J.M. is supported by the FPI fellowship BES-2013-064815 linked to the previous project. He is grateful for the warm hospitality extended by the theoretical physics group at Milano-Bicocca U. where part of this work was done. N.T.M is supported by INFN and by the European Research Council under the European Union's Seventh Framework Program (FP/2007-2013) - ERC Grant Agreement n. 307286 (XD-STRING). E. Ó C is grateful to KIAS for hospitality during the final stages of this project. E. Ó C. is supported by the Marie Curie award PIOF-2012-328625 T-DUALITIES.

A Hopf-fibre T-duality for $AdS_3 \times S^3 \times S^3 \times S^1$

To support claims in the text concerning the isometry supergroup for non-Abelian Tduals, here we present simpler Hopf-fibre T-duals in an analogous fashion. Abelian Hopffibre T-duals of the related IIB geometry with small superconformal symmetry, namely $AdS_3 \times S^3 \times CY_2$, were considered in [79]. There it was noted that supersymmetry can be preserved completely. Here we explicitly show that this is not the case when one starts with a geometry with large superconformal symmetry. Moreover, following [79], we could extend our analysis here to geometries supported by both NS and RR fields, where Tduality results not in $S^1 \times S^2$, but in (squashed) Lens spaces, S^3/\mathbb{Z}_p , however we focus on the simplest case with just RR fields.

Starting from $AdS_3 \times S^3 \times S^3 \times S^1$ (3.1), it is known that Abelian T-duality on a given Hopf-fibre will produce a $\frac{1}{4}$ -BPS $AdS_3 \times S^3 \times S^2 \times T^2$ geometry, where the corresponding supergroup is $D(2|1,\gamma) \times SL(2,\mathbb{R}) \times SU(2)$ [60, 61]. Here γ is a real parameter equating to the ratio of the radii of the three-sphere and two-sphere.¹⁷ Recalling that the bosonic subgroup of the supergroup $D(2|1, \gamma)$ is $SL(2, \mathbb{R}) \times SU(2) \times SU(2)$, we recognise that the symmetries simply correspond to the isometries of $AdS_3 \times S^3 \times S^2$.

Assuming we begin in Type IIB with the solution (3.1), the geometry resulting from a Hopf-fibre T-duality may be written as

$$ds^{2} = L^{2}ds^{2}_{AdS_{3}} + \frac{4}{R_{+}^{2}}d\psi^{2} + \frac{R_{+}^{2}}{4}(d\theta^{2} + \sin^{2}\theta d\phi^{2}) + R_{-}^{2}ds^{2}_{S_{-}^{3}} + dx^{2},$$

$$B_{2} = \cos\theta d\phi \wedge d\psi, \quad e^{\Phi} = \frac{2}{R_{+}},$$

$$F_{2} = -\frac{R_{+}^{2}}{4}\sin\theta d\theta \wedge d\phi,$$

$$F_{4} = \left[2L^{2}\text{Vol}(AdS_{3}) + 2R_{-}^{2}\text{Vol}(S_{-}^{3})\right] \wedge d\psi.$$
(A.1)

As with the original geometry, the Bianchis and the equations of motion are trivially satisfied. Plugging this solution into the dilatino variation, one can extract two commuting projection conditions,

$$\left(-\frac{1}{R_{+}}\Gamma^{\theta\phi}i\sigma_{2} + \frac{1}{L}\Gamma^{012\psi}\sigma_{1} + \frac{1}{R_{-}}\Gamma^{678\psi}\sigma_{1}\right)\eta = (\Gamma^{\psi}\sigma_{1} - 1)\eta = 0,$$
(A.2)

confirming that we now have eight preserved supersymmetries versus the original sixteen. As a consistency check, we observe that squaring the first projection condition, we recover the constraint on the radii (3.5). Solving for the Killing spinor along the internal directions, we find

$$\eta = e^{-\frac{1}{2}\theta\Gamma^{\phi}i\sigma_2}e^{\frac{1}{2}\phi\Gamma^{\theta\phi}}\tilde{\eta},\tag{A.3}$$

where $\tilde{\eta}$ denotes the Killing spinor for AdS_3 . Employing the left-invariant one-forms for the inert three-sphere, we see that angular dependence drops out, so we have an $SU(2)_L$ global symmetry, just as we witnessed in the non-Abelian case. As a direct consequence, the Killing spinors are independent of the Hopf-fibre and we can perform a further Abelian T-duality. It is an interesting feature of this geometry that uplifting on the M-theory circle to 11D, we recover the $AdS_3 \times S^3 \times S^3 \times T^2$ geometry in 11D, so that supersymmetry is restored to $\frac{1}{2}$ -BPS.¹⁸ This uplift should be contrasted with the more trivial T-duality on the *x*-direction and uplift, which leads to the same upstairs solution.

It is instructive to perform another Hopf-fibre T-duality, thus mirroring the combination of non-Abelian transformations in section 6. Doing so with respect to the ψ_2 -direction, we get

$$\begin{split} \mathrm{d}s^2 &= L^2 \mathrm{d}s^2_{AdS_3} + \frac{4}{R_+^2} \mathrm{d}\psi_1^2 + \frac{R_+^2}{4} (\mathrm{d}\theta_1^2 + \sin^2\theta_1 \mathrm{d}\phi_1^2) + \frac{4}{R_-^2} \mathrm{d}\psi_2^2 \\ &+ \frac{R_-^2}{4} (\mathrm{d}\theta_2^2 + \sin^2\theta_2 \mathrm{d}\phi_2^2) + \mathrm{d}x^2, \end{split}$$

¹⁷Prior to T-duality, this is just the ratio of the radii of the three-spheres.

 $^{^{18}\}mathrm{This}$ is the reverse of the dimensional reduction considered in ref. [61].

$$B_{2} = \cos \theta_{1} \mathrm{d}\phi_{1} \wedge \mathrm{d}\psi_{1} + \cos \theta_{2} \mathrm{d}\phi_{2} \wedge \mathrm{d}\psi_{2}, \quad e^{\Phi} = \frac{4}{R_{+}R_{-}},$$

$$F_{3} = \frac{R_{+}^{2}}{4} \sin \theta_{1} \mathrm{d}\theta_{1} \wedge \mathrm{d}\phi_{1} \wedge \mathrm{d}\psi_{2} - \frac{R_{-}^{2}}{4} \sin \theta_{2} \mathrm{d}\theta_{2} \wedge \mathrm{d}\phi_{2} \wedge \mathrm{d}\psi_{1},$$

$$F_{5} = (1 + *_{10}) \left[-2L^{2} \mathrm{Vol}(AdS_{3}) \wedge \mathrm{d}\psi_{1} \wedge \mathrm{d}\psi_{2}\right], \qquad (A.4)$$

where we have added subscripts to distinguish the angular coordinates. We note that the NS sector is even under an exchange of angular coordinates, whereas the RR sector is odd. We now check the remaining supersymmetry. From the dilatino variation, we get the projection condition:

$$\Gamma^{\psi_1\psi_2}i\sigma_2\eta = -\eta. \tag{A.5}$$

From the gravitino variations along the x, ψ_1 and ψ_2 directions, we get the additional projection

$$\left[\frac{L}{R_{+}}\Gamma^{012\theta_{1}\phi_{1}\psi_{2}} - \frac{L}{R_{-}}\Gamma^{012\theta_{2}\phi_{2}\psi_{1}}\right]\sigma_{1}\eta = -\eta.$$
(A.6)

One can check that the two projection conditions we have indeed commute, so supersymmetry is not broken further.

We can once again solve for angular dependence, getting

$$\eta = e^{-\frac{1}{2}\theta_1 \Gamma^{\phi_1 \psi_1} \sigma_3} e^{-\frac{1}{2}\phi_1 \Gamma^{\phi_1 \theta_1}} e^{-\frac{1}{2}\theta_2 \Gamma^{\phi_2 \psi_2} \sigma_3} e^{-\frac{1}{2}\phi_2 \Gamma^{\phi_2 \theta_2}} \tilde{\eta}, \tag{A.7}$$

where $\tilde{\eta}$ is expected to be the Killing spinor for AdS_3 . Indeed, one can check that the remaining equation is just the Killing spinor equation for AdS_3 , $\nabla_{\mu}\eta = \frac{1}{2}\gamma_3\Gamma_{\mu}\eta$, where we have defined $\gamma_3 \equiv \Gamma^{012}$. Solving the AdS_3 Killing spinor equation, we find

$$\tilde{\eta} = \left(r^{\frac{1}{2}} + r^{-\frac{1}{2}}(t\Gamma_0^2 + x_1\Gamma_1^2)\right)\tilde{\eta}_+ + r^{-\frac{1}{2}}\tilde{\eta}_-,\tag{A.8}$$

where $\tilde{\eta}_{\pm}$ are constant spinors subject to (A.5) and (A.6) satisfying $\Gamma^{01}\tilde{\eta}_{\pm} = \pm \tilde{\eta}_{\pm}$. We clearly see that the preserved supersymmetry is $\mathcal{N} = (0, 4)$, since the Killing spinors separate into the usual Poincaré and superconformal Killing spinors, each with a different chirality. The same conclusion can be drawn for the non-Abelian T-dual in section 6. However, in contrast to the usual small superconformal symmetry, we appear to have $SU(2) \times SU(2)$ R-symmetry, which is suggested from the angular dependence of the Killing spinors.

To further check the R-symmetry, we can also analyse the isometry algebra in our conventions in 10D, following a procedure outlined in ref. [61]. The first step is to identify the corresponding generic 10D Killing vector field, whose existence is always guaranteed for supersymmetric geometries. Using the Killing spinor equations presented in section 7, standard arguments show that

$$V = \frac{1}{2} \left(\bar{\epsilon}_{+} \Gamma^{M} \epsilon_{+} + \bar{\epsilon}_{-} \Gamma^{M} \epsilon_{-} \right) \partial_{M}$$
(A.9)

is always Killing. Note, we define $\bar{\epsilon} \equiv \epsilon^{\dagger} \Gamma^{0}$, with $(\Gamma^{0})^{\dagger} = -\Gamma^{0}$ and $(\Gamma^{i})^{\dagger} = \Gamma^{i}, i = 1, \dots, 9$. From (A.5), we have $\epsilon_{-} = \Gamma^{\psi_{1}\psi_{2}}\epsilon_{+}$, and as a result,

$$V^{M} = \frac{1}{2} \left(\bar{\epsilon}_{+} \Gamma^{M} \epsilon_{+} - \bar{\epsilon}_{+} \Gamma^{\psi_{1}\psi_{2}} \Gamma^{M} \Gamma^{\psi_{1}\psi_{2}} \epsilon_{+} \right).$$
(A.10)

We immediately recognise that $V^{\psi_i} = 0$, which is as expected, since these components of the vector field drop out when we reduce on a Hopf-fibre from 11D [61]. Thus, $V^M = \bar{\epsilon}_+ \Gamma^M \epsilon_+$, only depends on one of the Majorana-Weyl spinors.

It is then a simple exercise to determine the internal components of V,

$$V_{\text{int}} = \frac{2}{R_{+}} \bar{\epsilon}_{+} \Gamma^{\theta_{1}} \epsilon_{-} \xi_{1}^{+} + \frac{2}{R_{+}} \bar{\epsilon}_{+} \Gamma^{\phi_{1}} \epsilon_{-} \xi_{2}^{+} + \frac{2}{R_{-}} \bar{\epsilon}_{+} \Gamma^{\theta_{2}} \epsilon_{-} \xi_{1}^{-} + \frac{2}{R_{-}} \bar{\epsilon}_{+} \Gamma^{\phi_{2}} \epsilon_{-} \xi_{2}^{-} - \frac{2}{R_{+}} \bar{\epsilon}_{+} \Gamma^{\psi_{1}} \epsilon_{-} \xi_{3}^{+} - \frac{2}{R_{-}} \bar{\epsilon}_{+} \Gamma^{\psi_{2}} \epsilon_{-} \xi_{3}^{-} + (+ \longleftrightarrow -), \qquad (A.11)$$

where we have relabeled $\tilde{\epsilon}$ simply ϵ for convenience and have defined the following twosphere Killing vectors:

$$\xi_1^+ = \cos\phi_1\partial_{\theta_1} - \sin\phi_1\cot\theta_1\partial_{\phi_1}, \quad \xi_2^+ = \sin\phi_1\partial_{\theta_1} + \cos\phi_1\cot\theta_1\partial_{\phi_1}, \quad \xi_3^+ = \partial_{\phi_1},$$

with ξ_i^- similarly defined in terms of the coordinates (θ_2, ϕ_2) . Note the Killing vectors satisfy the expected SU(2) commutation relations, $[\xi_i^{(+)}, \xi_j^{(+)}] = -\epsilon_{ijk}\xi_k^{(+)}$, etc.

The subscripts on ϵ refer to chirality with respect to Γ^{01} . It is straightforward to show that other combinations of spinors cannot contribute to these vector bilinears. We note that since we have eight supersymmetries, there is *a priori* no relation between say $\bar{\epsilon}_{+}\Gamma^{\theta_{1}}\epsilon_{-}$ and $\bar{\epsilon}_{+}\Gamma^{\theta_{2}}\epsilon_{-}$ etc., so the SU(2) symmetries should be viewed as being independent. This suggests an $\mathcal{N} = (0, 4)$ SCFT with SU(2) × SU(2) R-symmetry.

Evaluating the external AdS_3 components of the Killing vector V, we find

$$V_{\text{ext}} = (\bar{\epsilon}_{+}\Gamma^{2}\epsilon_{-} + \bar{\epsilon}_{-}\Gamma^{2}\epsilon_{+})(M_{01} + D) + \bar{\epsilon}_{-}\Gamma^{0}\epsilon_{-}(P_{0} - P_{1}) + \bar{\epsilon}_{+}\Gamma^{0}\epsilon_{+}(K_{0} + K_{1}),$$

where now ϵ denotes the constant chiral spinors appearing in the expression for the AdS_3 Killing spinor (A.8), and we have defined the AdS_3 Killing vectors in Poincaré patch as

$$P_{0} = \partial_{t}, \qquad P_{1} = -\partial_{x},$$

$$M_{01} = x \partial_{t} + t \partial_{x}, \qquad D = r \partial_{r} + t \partial_{t} + x \partial_{x},$$

$$K_{0} = (t^{2} + x^{2} + r^{2})\partial_{t} + 2t(r\partial_{r} + x\partial_{x}),$$

$$K_{1} = (t^{2} + x^{2} - r^{2})\partial_{x} + 2x(r\partial_{r} + t\partial_{t}).$$
(A.12)

These satisfy the usual conformal algebra:

$$[M_{\mu\nu}, P_{\rho}] = -(\eta_{\mu\rho}P_{\nu} - \eta_{\nu\rho}P_{\mu}), \qquad [M_{\mu\nu}, K_{\rho}] = -(\eta_{\mu\rho}K_{\nu} - \eta_{\nu\rho}K_{\mu}), [M_{\mu\nu}, D] = 0, \quad [D, P_{\mu}] = -P_{\mu}, \qquad [D, K_{\mu}] = K_{\mu}, [P_{\mu}, K_{\nu}] = 2M_{\mu\nu} - 2\eta_{\mu\nu}D,$$
(A.13)

with $\mu, \nu = 0, 1$.

Recalling the bosonic subgroup of $D(2|1, \gamma)$, we come to the conclusion that after two Hopf-fibre T-dualities, the isometry supergroup of the $AdS_3 \times S^3 \times S^3$ geometry, namely $D(2|1, \gamma) \times D(2|1, \gamma)$ becomes simply $D(2|1, \gamma) \times SL(2, \mathbb{R})$, where γ is the ratio of the two-sphere radii. Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- [1] J.P. Gauntlett, O.A.P. Mac Conamhna, T. Mateos and D. Waldram, AdS spacetimes from wrapped M5 branes, JHEP 11 (2006) 053 [hep-th/0605146] [INSPIRE].
- [2] H. Kim, K.K. Kim and N. Kim, 1/4-BPS M-theory bubbles with SO(3) × SO(4) symmetry, JHEP 08 (2007) 050 [arXiv:0706.2042] [INSPIRE].
- [3] H. Lin, O. Lunin and J.M. Maldacena, Bubbling AdS space and 1/2 BPS geometries, JHEP 10 (2004) 025 [hep-th/0409174] [INSPIRE].
- [4] J.M. Maldacena, A. Strominger and E. Witten, Black hole entropy in M-theory, JHEP 12 (1997) 002 [hep-th/9711053] [INSPIRE].
- [5] N. Gaddam, A. Gnecchi, S. Vandoren and O. Varela, Rholography, black holes and Scherk-Schwarz, JHEP 06 (2015) 058 [arXiv:1412.7325] [INSPIRE].
- [6] D. Gaiotto, N = 2 dualities, JHEP 08 (2012) 034 [arXiv:0904.2715] [INSPIRE].
- [7] D. Gaiotto and J. Maldacena, The gravity duals of N = 2 superconformal field theories, JHEP 10 (2012) 189 [arXiv:0904.4466] [INSPIRE].
- [8] D. Joyce, Lectures on special Lagrangian geometry, math/0111111 [INSPIRE].
- [9] J.M. Maldacena and C. Núñez, Supergravity description of field theories on curved manifolds and a no go theorem, Int. J. Mod. Phys. A 16 (2001) 822 [hep-th/0007018] [INSPIRE].
- [10] A. Fayyazuddin and D.J. Smith, Localized intersections of M5-branes and four-dimensional superconformal field theories, JHEP 04 (1999) 030 [hep-th/9902210] [INSPIRE].
- [11] E. Ó Colgáin and B. Stefanski, Jr., A search for $AdS_5 \times S^2$ IIB supergravity solutions dual to N = 2 SCFTs, JHEP 10 (2011) 061 [arXiv:1107.5763] [INSPIRE].
- [12] R.A. Reid-Edwards and B. Stefański Jr., On Type IIA geometries dual to N = 2 SCFTs, Nucl. Phys. B 849 (2011) 549 [arXiv:1011.0216] [INSPIRE].
- [13] A. Donos and J. Simon, The electrostatic view on M-theory LLM geometries, JHEP 01 (2011) 067 [arXiv:1010.3101] [INSPIRE].
- [14] O. Aharony, L. Berdichevsky and M. Berkooz, 4D N = 2 superconformal linear quivers with type IIA duals, JHEP 08 (2012) 131 [arXiv:1206.5916] [INSPIRE].
- [15] P.M. Petropoulos, K. Sfetsos and K. Siampos, Gravity duals of $\mathcal{N}=2$ superconformal field theories with no electrostatic description, JHEP 11 (2013) 118 [arXiv:1308.6583] [INSPIRE].
- [16] P.M. Petropoulos, K. Sfetsos and K. Siampos, Gravity duals of $\mathcal{N} = 2$ SCFTs and asymptotic emergence of the electrostatic description, JHEP 09 (2014) 057 [arXiv:1406.0853] [INSPIRE].
- [17] K. Sfetsos and D.C. Thompson, On non-abelian T-dual geometries with Ramond fluxes, Nucl. Phys. B 846 (2011) 21 [arXiv:1012.1320] [INSPIRE].
- [18] O.A.P. Mac Conamhna and E. Ó Colgáin, Supersymmetric wrapped membranes, AdS₂ spaces and bubbling geometries, JHEP 03 (2007) 115 [hep-th/0612196] [INSPIRE].

- [19] E. Ó Colgáin, J.-B. Wu and H. Yavartanoo, On the generality of the LLM geometries in M-theory, JHEP 04 (2011) 002 [arXiv:1010.5982] [INSPIRE].
- [20] E. Ó Colgáin, Beyond LLM in M-theory, JHEP 12 (2012) 023 [arXiv:1208.5979] [INSPIRE].
- [21] T. Ortiz, H. Samtleben and D. Tsimpis, Matrix model holography, JHEP 12 (2014) 096 [arXiv:1410.0487] [INSPIRE].
- [22] E. Ó Colgáin, J.-B. Wu and H. Yavartanoo, Supersymmetric $AdS_3 \times S^2$ M-theory geometries with fluxes, JHEP **08** (2010) 114 [arXiv:1005.4527] [INSPIRE].
- [23] J.P. Gauntlett, D. Martelli, J. Sparks and D. Waldram, Supersymmetric AdS₅ solutions of M-theory, Class. Quant. Grav. 21 (2004) 4335 [hep-th/0402153] [INSPIRE].
- [24] O. Kelekci, E. Ó Colgáin and J. Montero, work in progress.
- [25] G. Itsios, C. Núñez, K. Sfetsos and D.C. Thompson, On non-abelian T-duality and new N = 1 backgrounds, Phys. Lett. B 721 (2013) 342 [arXiv:1212.4840] [INSPIRE].
- [26] K. Sfetsos and D.C. Thompson, New $\mathcal{N} = 1$ supersymmetric AdS₅ backgrounds in type IIA supergravity, JHEP **11** (2014) 006 [arXiv:1408.6545] [INSPIRE].
- [27] N.T. Macpherson, C. Núñez, L.A. Pando Zayas, V.G.J. Rodgers and C.A. Whiting, Type IIB supergravity solutions with AdS₅ from Abelian and non-Abelian T dualities, JHEP 02 (2015) 040 [arXiv:1410.2650] [INSPIRE].
- [28] Y. Bea et al., Compactifications of the Klebanov-Witten CFT and new AdS₃ backgrounds, JHEP 05 (2015) 062 [arXiv:1503.07527] [INSPIRE].
- [29] A. Barranco, J. Gaillard, N.T. Macpherson, C. Núñez and D.C. Thompson, G-structures and flavouring non-Abelian T-duality, JHEP 08 (2013) 018 [arXiv:1305.7229] [INSPIRE].
- [30] N.T. Macpherson, Non-Abelian T-duality, G₂-structure rotation and holographic duals of N = 1 Chern-Simons theories, JHEP 11 (2013) 137 [arXiv:1310.1609] [INSPIRE].
- [31] J. Gaillard, N.T. Macpherson, C. Núñez and D.C. Thompson, Dualising the baryonic branch: dynamic SU(2) and confining backgrounds in IIA, Nucl. Phys. B 884 (2014) 696 [arXiv:1312.4945] [INSPIRE].
- [32] E. Caceres, N.T. Macpherson and C. Núñez, New type IIB backgrounds and aspects of their field theory duals, JHEP 08 (2014) 107 [arXiv:1402.3294] [INSPIRE].
- [33] K.S. Kooner and S. Zacarías, Non-abelian T-dualizing the resolved conifold with regular and fractional D3-branes, arXiv:1411.7433 [INSPIRE].
- [34] S. Zacarías, Semiclassical strings and non-abelian T-duality, Phys. Lett. B 737 (2014) 90 [arXiv:1401.7618] [INSPIRE].
- [35] P.M. Pradhan, Oscillating strings and non-abelian T-dual Klebanov-Witten background, Phys. Rev. D 90 (2014) 046003 [arXiv:1406.2152] [INSPIRE].
- [36] E. Gevorgyan and G. Sarkissian, Defects, non-abelian T-duality and the Fourier-Mukai transform of the Ramond-Ramond fields, JHEP 03 (2014) 035 [arXiv:1310.1264] [INSPIRE].
- [37] Y. Lozano, E. Ó Colgáin, D. Rodriguez-Gomez and K. Sfetsos, Supersymmetric AdS₆ via T duality, Phys. Rev. Lett. 110 (2013) 231601 [arXiv:1212.1043] [INSPIRE].
- [38] A. Passias, A note on supersymmetric AdS₆ solutions of massive type IIA supergravity, JHEP 01 (2013) 113 [arXiv:1209.3267] [INSPIRE].
- [39] F. Apruzzi, M. Fazzi, A. Passias, D. Rosa and A. Tomasiello, AdS₆ solutions of type-II supergravity, JHEP 11 (2014) 099 [Erratum ibid. 1505 (2015) 012] [arXiv:1406.0852] [INSPIRE].
- [40] H. Kim, N. Kim and M. Suh, Supersymmetric AdS₆ solutions of type IIB supergravity, arXiv:1506.05480 [INSPIRE].
- [41] A. Brandhuber and Y. Oz, The D4-D8 brane system and five-dimensional fixed points, Phys. Lett. B 460 (1999) 307 [hep-th/9905148] [INSPIRE].
- [42] Y. Lozano, N.T. Macpherson and J. Montero, $A \mathcal{N} = 2$ supersymmetric AdS_4 solution in *M*-theory with purely magnetic flux, arXiv:1507.02660 [INSPIRE].
- [43] M. Pernici and E. Sezgin, Spontaneous compactification of seven-dimensional supergravity theories, Class. Quant. Grav. 2 (1985) 673 [INSPIRE].
- [44] O. DeWolfe, A. Hanany, A. Iqbal and E. Katz, Five-branes, seven-branes and five-dimensional E(n) field theories, JHEP 03 (1999) 006 [hep-th/9902179] [INSPIRE].
- [45] Y. Lozano, E. Ó Colgáin and D. Rodriguez-Gomez, Hints of 5D fixed point theories from non-abelian T-duality, JHEP 05 (2014) 009 [arXiv:1311.4842] [INSPIRE].
- [46] T. Dimofte, D. Gaiotto and S. Gukov, Gauge theories labelled by three-manifolds, Commun. Math. Phys. 325 (2014) 367 [arXiv:1108.4389] [INSPIRE].
- [47] T.R. Araujo and H. Nastase, $\mathcal{N} = 1$ SUSY backgrounds with an AdS factor from non-Abelian T duality, Phys. Rev. D 91 (2015) 126015 [arXiv:1503.00553] [INSPIRE].
- [48] I.R. Klebanov and E. Witten, Superconformal field theory on three-branes at a Calabi-Yau singularity, Nucl. Phys. B 536 (1998) 199 [hep-th/9807080] [INSPIRE].
- [49] G. Itsios, C. Núñez, K. Sfetsos and D.C. Thompson, Non-Abelian T-duality and the AdS/CFT correspondence:new N = 1 backgrounds, Nucl. Phys. B 873 (2013) 1 [arXiv:1301.6755] [INSPIRE].
- [50] I. Bah, C. Beem, N. Bobev and B. Wecht, AdS/CFT dual pairs from M5-branes on Riemann surfaces, Phys. Rev. D 85 (2012) 121901 [arXiv:1112.5487] [INSPIRE].
- [51] I. Bah, C. Beem, N. Bobev and B. Wecht, Four-dimensional SCFTs from M5-branes, JHEP 06 (2012) 005 [arXiv:1203.0303] [INSPIRE].
- [52] F. Benini, Y. Tachikawa and B. Wecht, Sicilian gauge theories and N = 1 dualities, JHEP 01 (2010) 088 [arXiv:0909.1327] [INSPIRE].
- [53] O. Kelekci, Y. Lozano, N.T. Macpherson and E. Ó Colgáin, Supersymmetry and non-Abelian T-duality in type-II supergravity, Class. Quant. Grav. 32 (2015) 035014 [arXiv:1409.7406] [INSPIRE].
- [54] Y. Lozano and N.T. Macpherson, A new AdS₄/CFT₃ dual with extended SUSY and a spectral flow, JHEP 11 (2014) 115 [arXiv:1408.0912] [INSPIRE].
- [55] J. Murugan and H. Nastase, A nonabelian particle-vortex duality, arXiv:1506.04090 [INSPIRE].
- [56] P. Karndumri and E. Ó Colgáin, 3D Supergravity from wrapped D3-branes, JHEP 10 (2013) 094 [arXiv:1307.2086] [INSPIRE].
- [57] P.M. Cowdall and P.K. Townsend, Gauged supergravity vacua from intersecting branes, *Phys. Lett.* B 429 (1998) 281 [Erratum ibid. B 434 (1998) 458] [hep-th/9801165] [INSPIRE].

- [58] S. Gukov, E. Martinec, G.W. Moore and A. Strominger, The search for a holographic dual to $AdS_3 \times S^3 \times S^3 \times S^1$, Adv. Theor. Math. Phys. 9 (2005) 435 [hep-th/0403090] [INSPIRE].
- [59] J. de Boer, A. Pasquinucci and K. Skenderis, AdS/CFT dualities involving large 2 D N = 4 superconformal symmetry, Adv. Theor. Math. Phys. 3 (1999) 577 [hep-th/9904073]
 [INSPIRE].
- [60] H.J. Boonstra, B. Peeters and K. Skenderis, Brane intersections, Anti-de Sitter space-times and dual superconformal theories, Nucl. Phys. B 533 (1998) 127 [hep-th/9803231] [INSPIRE].
- [61] J.P. Gauntlett, R.C. Myers and P.K. Townsend, Supersymmetry of rotating branes, Phys. Rev. D 59 (1998) 025001 [hep-th/9809065] [INSPIRE].
- [62] D. Tong, The holographic dual of $AdS_3 \times S^3 \times S^3 \times S^1$, JHEP **04** (2014) 193 [arXiv:1402.5135] [INSPIRE].
- [63] S. Elitzur, O. Feinerman, A. Giveon and D. Tsabar, String theory on $AdS_3 \times S^3 \times S^3 \times S^1$, Phys. Lett. B 449 (1999) 180 [hep-th/9811245] [INSPIRE].
- [64] J.D. Brown and M. Henneaux, Central charges in the canonical realization of asymptotic symmetries: an example from three-dimensional gravity, Commun. Math. Phys. 104 (1986) 207 [INSPIRE].
- [65] A. Sevrin, W. Troost and A. Van Proeyen, Superconformal algebras in two-dimensions with N = 4, Phys. Lett. **B 208** (1988) 447 [INSPIRE].
- [66] G. Itsios, Y. Lozano, E. O Colgáin and K. Sfetsos, Non-abelian T-duality and consistent truncations in type-II supergravity, JHEP 08 (2012) 132 [arXiv:1205.2274] [INSPIRE].
- [67] E. Alvarez, L. Álvarez-Gaumé, J.L.F. Barbon and Y. Lozano, Some global aspects of duality in string theory, Nucl. Phys. B 415 (1994) 71 [hep-th/9309039] [INSPIRE].
- [68] N. Seiberg, Electric-magnetic duality in supersymmetric nonAbelian gauge theories, Nucl. Phys. B 435 (1995) 129 [hep-th/9411149] [INSPIRE].
- [69] I.R. Klebanov and M.J. Strassler, Supergravity and a confining gauge theory: duality cascades and χ_{SB} resolution of naked singularities, JHEP **08** (2000) 052 [hep-th/0007191] [INSPIRE].
- [70] I.R. Klebanov, D. Kutasov and A. Murugan, Entanglement as a probe of confinement, Nucl. Phys. B 796 (2008) 274 [arXiv:0709.2140] [INSPIRE].
- [71] S.F. Hassan, T duality, space-time spinors and RR fields in curved backgrounds, Nucl. Phys. B 568 (2000) 145 [hep-th/9907152] [INSPIRE].
- [72] A. Donos and J.P. Gauntlett, Flowing from AdS₅ to AdS₃ with T^{1,1}, JHEP 08 (2014) 006 [arXiv:1404.7133] [INSPIRE].
- [73] Y. Kosmann, A note on Lie-Lorentz derivatives, Annali Mat. Pura Appl. 91 (1972) 317.
- [74] J. Jeong, O. Kelekci and E. Ó Colgáin, An alternative IIB embedding of F(4) gauged supergravity, JHEP 05 (2013) 079 [arXiv:1302.2105] [INSPIRE].
- [75] Y. Lozano, E. Ó Colgáin, K. Sfetsos and D.C. Thompson, Non-abelian T-duality, Ramond fields and coset geometries, JHEP 06 (2011) 106 [arXiv:1104.5196] [INSPIRE].
- [76] F. Apruzzi, M. Fazzi, D. Rosa and A. Tomasiello, All AdS₇ solutions of type-II supergravity, JHEP 04 (2014) 064 [arXiv:1309.2949] [INSPIRE].

- [77] F. Apruzzi, M. Fazzi, A. Passias and A. Tomasiello, Supersymmetric AdS₅ solutions of massive IIA supergravity, JHEP 06 (2015) 195 [arXiv:1502.06620] [INSPIRE].
- [78] C. Bachas, E. D'Hoker, J. Estes and D. Krym, *M*-theory solutions invariant under $D(2,1;\gamma) \oplus D(2,1;\gamma)$, Fortsch. Phys. **62** (2014) 207 [arXiv:1312.5477].
- [79] M.J. Duff, H. Lü and C.N. Pope, AdS₃ × S³ (un)twisted and squashed and an O(2,2,Z) multiplet of dyonic strings, Nucl. Phys. B 544 (1999) 145 [hep-th/9807173] [INSPIRE].

3.2 $\mathcal{N} = 2$ supersymmetric AdS_4 solution in M-theory with purely magnetic flux



56

Published for SISSA by 2 Springer

RECEIVED: July 21, 2015 ACCEPTED: August 28, 2015 Published: October 1, 2015

A $\mathcal{N}=2$ supersymmetric AdS_4 solution in M-theory with purely magnetic flux

Yolanda Lozano,^a Niall T. Macpherson^b and Jesús Montero^a

^aDepartment of Physics, University of Oviedo, Avda. Calvo Sotelo 18, Oviedo, 33007 Spain ^bDipartimento di Fisica, Università di Milano-Bicocca, and INFN, sezione di Milano-Bicocca, Milano, I-20126 Italy *E-mail:* ylozano@uniovi.es, niall.macpherson@mib.infn.it, monteroaragon@uniovi.es

ABSTRACT: We find a new $\mathcal{N} = 2 A dS_4$ solution in M-theory supported by purely magnetic flux via a sequence of abelian and non-abelian T-dualities. This provides the second known example in this class besides the uplift of the Pernici and Sezgin solution to 7d gauged supergravity constructed in the eighties. We compute the free energy of the solution, and show that it scales as $N^{3/2}$. It is intriguing that even though the natural holographic interpretation is in terms of M5-branes wrapped on a special Lagrangian 3-cycle, this solution does not exhibit the expected N^3 behavior.

KEYWORDS: Supersymmetry and Duality, AdS-CFT Correspondence

ARXIV EPRINT: 1507.02660

doi:10.1007/JHEP10(2015)004

OPEN ACCESS, © The Authors. Article funded by $SCOAP^3$.

Contents

1	Introduction	1	
2	The IIB NAT dual AdS_4 solution	3	
3	The IIA NAT-T dual AdS_4 solution	5	
	3.1 Supersymmetry	6	
	3.2 Properties of the CFT	8	\Box
4	The purely magnetic AdS_4 solution in M-theory	10	H
	4.1 Fluxes	10	E
	4.2 Geometry and local SU(2) structure	11	Ю
	4.3 Properties of the CFT	13	
	4.4 Free energy	15	\square
5	Conclusions	16	
5 A	Conclusions Some details on the NAT and T duality transformations	16 17	(2)
5 A	Conclusions Some details on the NAT and T duality transformations A.1 The IIB NAT duality	16 17 18	(20)
5 A	Conclusions Some details on the NAT and T duality transformations A.1 The IIB NAT duality A.2 The IIA NAT-T duality	 16 17 18 19)(201
5 A	Conclusions Some details on the NAT and T duality transformations A.1 The IIB NAT duality A.2 The IIA NAT-T duality	 16 17 18 19 	(2015
5 A B	Conclusions Some details on the NAT and T duality transformations A.1 The IIB NAT duality A.2 The IIA NAT-T duality Type-II G-structure conditions for AdS_4 solutions	 16 17 18 19 20 	(2015)
5 A B C	Conclusions Some details on the NAT and T duality transformations A.1 The IIB NAT duality A.2 The IIA NAT-T duality Type-II G-structure conditions for AdS_4 solutions Detailed supersymmetry analysis	 16 17 18 19 20 22 	(2015)0
5 A B C	Conclusions Some details on the NAT and T duality transformations A.1 The IIB NAT duality A.2 The IIA NAT-T duality Type-II G-structure conditions for AdS_4 solutions Detailed supersymmetry analysis C.1 A SU(2) T-duality invariant Killing spinor on $AdS_4 \times \mathbb{CP}^3$	 16 17 18 19 20 22 22 	(2015)00
5 A B C	Conclusions Some details on the NAT and T duality transformations A.1 The IIB NAT duality A.2 The IIA NAT-T duality Type-II G-structure conditions for AdS_4 solutions Detailed supersymmetry analysis C.1 A SU(2) T-duality invariant Killing spinor on $AdS_4 \times \mathbb{CP}^3$ C.2 A U(1) of SU(3)-structures on \mathbb{CP}^3	 16 17 18 19 20 22 22 23)(2015)004
5 A B C	Conclusions Some details on the NAT and T duality transformations A.1 The IIB NAT duality A.2 The IIA NAT-T duality Type-II G-structure conditions for AdS_4 solutions Detailed supersymmetry analysis C.1 A SU(2) T-duality invariant Killing spinor on $AdS_4 \times \mathbb{CP}^3$ C.2 A U(1) of SU(3)-structures on \mathbb{CP}^3 C.3 A U(1) of SU(2)-structures in the non-Abelian T-dual	 16 17 18 19 20 22 22 23 24 	(2015)004
5 A B C	ConclusionsSome details on the NAT and T duality transformationsA.1 The IIB NAT dualityA.2 The IIA NAT-T dualityType-II G-structure conditions for AdS_4 solutionsDetailed supersymmetry analysisC.1 A SU(2) T-duality invariant Killing spinor on $AdS_4 \times \mathbb{CP}^3$ C.2 A U(1) of SU(3)-structures on \mathbb{CP}^3 C.3 A U(1) of SU(2)-structures in the non-Abelian T-dualC.4 Killing spinors in M-theory	 16 17 18 19 20 22 23 24 26)(2015)004

1 Introduction

In recent years non-Abelian T duality (NAT duality) has been very successfully applied as a generator of new supergravity backgrounds that may have interesting applications in the context of the AdS/CFT correspondence [1]–[17]. While some of these backgrounds represent explicit new solutions to existing classifications [1, 4, 5, 12], some of them have been shown to fall outside known classifications [18] or to provide the only explicit solution to some set of PDEs [3].

A very inspiring example is the AdS_6 solution to Type IIB supergravity constructed in [3]. Supersymmetry is known to impose strong constraints on AdS_6 backgrounds [19,

57

20],¹ even if large classes of fixed point theories are known to exist in 5 dimensions [22–24] with expected AdS_6 duals. The only AdS_6/CFT_5 explicit example identified to date is the duality between the Brandhuber and Oz solution to massive Type IIA [25] (known to be the only possible IIA AdS_6 background [19]²) and the fixed point theory that arises from the D4/D8/O8 system in [22]. Yet, there are families of 5 and 7-brane webs giving rise to 5d fixed point theories [27–29] whose dual AdS_6 spaces remain to be identified. The solution in [3]³ provides a possible holographic dual to these theories.

The duality between 3d SCFTs arising from M5-branes wrapped on 3d manifolds and AdS_4 spaces is yet another example in which explicit AdS_4 solutions with the required properties are scarce. Remarkable progress has been achieved recently [30] through the construction of explicit $AdS_4 \times \Sigma_3 \times M_4$ solutions to massive IIA which are candidate duals to compactifications of the (1, 0) 6d CFTs living in NS5-D6-D8 systems [31] on a 3-manifold Σ_3 , which could eventually lead to a generalization of the 3d-3d correspondence [32–38] to $\mathcal{N} = 1$. The $\mathcal{N} = 2$ case is yet especially interesting, since with this number of supersymmetries the 3d-3d correspondence allows to associate a 3d $\mathcal{N} = 2$ SCFT to the 3d manifold on which the M5-branes are wrapped [32–38]. This field theory arises as a twisted compactification on the 3d Riemann surface of the (2, 0) CFT₆ living in the M5-branes.

However, to date only one $\mathcal{N} = 2 \ AdS_4$ explicit solution to M-theory is known that could provide the holographic dual to these compactifications. This solution is the uplift to eleven dimensions [40, 41] of the Pernici-Sezgin solution [42] to 7d gauged supergravity, that dates back to the 80's. This is of the form $AdS_4 \times M_7$ where M_7 is an S^4 -fibration over a hyperbolic manifold H_3 , on which the M5-branes are wrapped. The Pernici-Sezgin solution is the only explicit solution of the form $AdS_4 \times \Sigma_3 \times S^4$ in the general class of $\mathcal{N} = 2 \ AdS_4$ backgrounds obtained from M5-branes wrapping calibrated cycles in [43].

In this paper we construct a new $\mathcal{N} = 2 \ AdS_4$ solution to M-theory belonging to the general class of $\mathcal{N} = 2 \ AdS_4$ backgrounds derived in [43]. This class is defined by requiring that the Killing spinors satisfy the same projection conditions as the wrapped branes and that there is no electric flux. Yet the solutions need not describe in general M5-branes wrapped in 3d manifolds in the near horizon limit. Our solution seems to belong to this more general class.

We obtain our solution through non-Abelian T-duality on the $AdS_4 \times CP^3$ background dual to ABJM [44], followed by an Abelian T-duality and an uplift to eleven dimensions. The non-Abelian T-duality transformation is responsible for the breaking of the supersymmetries from $\mathcal{N} = 6$ to $\mathcal{N} = 2$. The detailed properties of the resulting $\mathcal{N} = 2 \ AdS_4$ solution to Type IIB were studied in [11]. This solution contains two U(1)'s, one of which can be further used to (Abelian) T-dualize back to Type IIA without breaking any of the supersymmetries. Finally, the solution is uplifted to eleven dimensions, where it can be shown to fulfill the conditions for 11d $\mathcal{N} = 2 \ AdS_4$ solutions with purely magnetic flux, derived in [43].⁴

¹See also [21].

²Variations of it such as orbifold solutions have also been constructed in [26].

 $^{^3\}mathrm{See}~[8]$ for a discussion of the properties of the associated CFT.

⁴A systematic study of the most general class of $\mathcal{N} = 2 AdS_4$ solutions of 11d supergravity, that includes the results in [43], was carried out in [45].

The paper is organized as follows. In section 2 we recall briefly the IIB solution constructed in [11] through non-Abelian T-duality acting on the $AdS_4 \times CP^3$ IIA background. In section 3 we construct its IIA Abelian T-dual, and discuss some properties of the associated dual CFT of relevance for the CFT interpretation of the 11d solution. Section 4 contains the uplift to M-theory. Here we discuss some properties of the CFT associated to the 11d solution, that are implied by the analysis of the supergravity solution as well as its IIA description. We compute the holographic central charge and show that, as expected, it coincides with the central charge of the IIB solution written in terms of the 11d charges. Thus, it scales with $N^{3/2}$, contrary to the expectation for M5-branes. We argue that the field theory analysis that we perform suggests that there should be Kaluza-Klein monopoles sourcing the background, and that M5-branes should only play a role in the presence of large gauge transformations (in a precise way that we define). This is intimately related to the existence of a non-compact direction inherited by the NAT duality transformation, which, as discussed at length in the NAT duality literature (see for instance [8, 11, 17]), represents the most puzzling obstacle towards a precise CFT interpretation of this transformation. Finally, in appendix A we present our conclusions. Here we discuss further our result for the free energy, as well as the view that we have taken to try to give a CFT meaning to the non-compact direction. We have relegated most of the technical details to three appendices. In appendix A we include some details of the derivation of both the NAT and T dual solutions presented in sections 2 and 3. These details are especially relevant for the supersymmetry analysis. In appendix B we review the G-structure conditions for preservation of supersymmetry of $AdS_4 \times M_6$ solutions to Type II supergravities. In appendix C we perform the detailed supersymmetry analysis of the solutions in IIA, IIB and M-theory.

2 The IIB NAT dual AdS_4 solution

This solution was constructed in [11], where some properties of the associated dual CFT were also analyzed. We refer the reader to this paper for more details. In this section we present the background for completeness. More technical properties of the derivation that will be useful for the study of the backgrounds constructed from this one in the following sections are presented in appendix A.

The background arises as the NAT dual of the $AdS_4 \times \mathbb{CP}^3$ background with respect to a freely acting SU(2) in the parameterization of the \mathbb{CP}^3 as a foliation in $T^{1,1} = S^2 \times S^3$:

$$ds^{2}(\mathbb{CP}^{3}) = d\zeta^{2} + \frac{1}{4} \left(\cos^{2} \zeta (d\theta_{1}^{2} + \sin^{2} \theta_{1} d\phi_{1}^{2}) + \sin^{2} \zeta (d\theta_{2}^{2} + \sin^{2} \theta_{2} d\phi_{2}^{2}) \right. \\ \left. + \sin^{2} \zeta \cos^{2} \zeta (d\psi + \cos \theta_{1} d\phi_{1} + \cos \theta_{2} d\phi_{2})^{2} \right) \\ = d\zeta^{2} + \frac{1}{4} \left(\cos^{2} \zeta (d\theta_{1}^{2} + \sin^{2} \theta_{1} d\phi_{1}^{2}) + \sin^{2} \zeta (\omega_{1}^{2} + \omega_{2}^{2}) + \sin^{2} \zeta \cos^{2} \zeta (\omega_{3} + \cos \theta_{1} d\phi_{1})^{2} \right)$$

$$(2.1)$$

where $0 \leq \zeta < \frac{\pi}{2}$, $0 \leq \theta_i < \pi$, $0 \leq \phi_i \leq 2\pi$, $0 \leq \psi \leq 4\pi$.

Dualising with respect to the SU(2) acting on the 3-sphere parameterized by (ψ,θ_2,ϕ_2) we obtain

$$d\tilde{s}^{2} = \frac{L^{2}}{4}ds^{2}(AdS_{4}) + L^{2}\left(d\zeta^{2} + \frac{1}{4}\cos^{2}\zeta\left(d\theta_{1}^{2} + \sin^{2}\theta_{1}d\phi_{1}^{2}\right)\right) + ds^{2}(M_{3}), \qquad (2.2)$$

where $ds^2(M_3)$ stands for the 3-dimensional metric:

$$ds^{2}(M_{3}) = \frac{1}{16 \det M} \Big[L^{4} \sin^{4} \zeta \Big(dr^{2} + r^{2} d\chi^{2} - \sin^{2} \zeta (\sin \chi dr + r \cos \chi d\chi)^{2} + r^{2} \cos^{2} \zeta \sin^{2} \chi (d\xi + \cos \theta_{1} d\phi_{1})^{2} \Big) + 16r^{2} dr^{2} \Big].$$
(2.3)

 $\det M$ is given by:

$$\det M = \frac{L^2}{64} \sin^2 \zeta \left(16r^2 (\sin^2 \chi + \cos^2 \chi \cos^2 \zeta) + L^4 \sin^4 \zeta \cos^2 \zeta \right).$$
(2.4)

Here (χ, ξ) parameterize the new 2-sphere arising through the NAT duality transformation, that we will denote by \tilde{S}^2 . r is the non-compact coordinate generated by the transformation, which lives in \mathbb{R}^+ . The presence of this non-compact direction is intimately related to the long-standing open problem of extending NAT duality beyond spherical world sheets. In the context of AdS/CFT applications this poses a problem to the CFT interpretation of AdS backgrounds generated through this transformation. Some ideas to provide a consistent interpretation have been proposed in [8, 11] (see also [17]), which we will partially use in this paper. The reader is referred to these papers for more details.

The dilaton reads in turn

$$e^{\phi} = \frac{L}{k} \frac{1}{\sqrt{\det M}} \,. \tag{2.5}$$

A B_2 field is also generated that reads:

,

$$B_{2} = \frac{L^{2} \sin^{2} \zeta}{64 \det M} \Big[-L^{4} r \cos^{2} \zeta \sin^{4} \zeta \cos \theta_{1} \sin \chi \, d\phi_{1} \wedge d\chi \\ -16 \, r^{2} \left(r (\cos^{2} \zeta \cos^{2} \chi + \sin^{2} \chi) \operatorname{Vol}(\tilde{S}^{2}) + \sin^{2} \zeta \sin^{2} \chi \cos \chi \, d\xi \wedge dr \right) \\ -\cos^{2} \zeta \cos \theta_{1} \cos \chi \left(L^{4} \sin^{4} \zeta + 16r^{2} \right) dr \wedge d\phi_{1} \Big].$$

$$(2.6)$$

Together with this we find the RR sector:

$$F_1 = \frac{\kappa}{2} \left(r \sin^2 \zeta \sin \chi \, d\chi - \sin^2 \zeta \cos \chi \, dr - r \sin 2\zeta \cos \chi \, d\zeta \right) \tag{2.7}$$

$$\hat{F}_3 = -\frac{3}{128}kL^4 \sin^3 2\zeta \, d\zeta \wedge \operatorname{Vol}(S_1^2) + \frac{k}{2} \Big(r \, dr \wedge (\cos^2 \zeta \operatorname{Vol}(S_1^2) \tag{2.8})$$

$$+ \sin 2\zeta \cos \theta_1 \, d\zeta \wedge d\phi_1 + \sin 2\zeta \sin^2 \chi \, d\zeta \wedge d\xi) - r^2 \sin 2\zeta \cos \chi \, d\zeta \wedge \operatorname{Vol}(S^2))$$

$$\hat{F}_5 = \frac{1}{64} k L^6 \sin^3 \zeta \cos \zeta \operatorname{Vol}(AdS_4) \wedge d\zeta - \frac{3}{8} k L^2 r \operatorname{Vol}(AdS_4) \wedge dr$$

$$+ \frac{k}{2} r^2 \Big(\cos^2 \zeta \operatorname{Vol}(S_1^2) + \sin 2\zeta \cos \theta_1 \, d\zeta \wedge d\phi_1 \Big) \wedge dr \wedge \operatorname{Vol}(\tilde{S}^2) ,$$

$$(2.9)$$

where $F_p = dC_{p-1} - H_3 \wedge C_{p-3}$ and $\hat{F} = F \wedge e^{-B_2}$ are the fluxes associated to the Page charges.

THEP10 (2015) 004

Note that after the dualisation a singularity has appeared at the fixed point $\zeta = 0$, where the squashed S^3 used to dualise shrinks to zero size. This singularity is associated to the component of the metric on the r-direction, and is always compensated with the singularity in the dilaton in computations of physical quantities such as gauge couplings, internal volumes, etc. We will see that it will be inherited by the IIA and M-theory solutions where physical quantities will however be perfectly well defined as well.

The IIA NAT-T dual AdS_4 solution 3

Following the steps in appendix A we get the following solution in Type IIA after dualizing the previous background along the ϕ_1 direction, that we will simply rename as ϕ^5

$$ds^{2} = \frac{L^{2}}{4}ds^{2}(AdS_{4}) + L^{2}d\zeta^{2} + \frac{L^{2}}{4}\cos^{2}\zeta d\theta^{2} + \sum_{i=1}^{4} \left(\mathcal{G}^{i}\right)^{2},$$
(3.1)

where

$$\begin{aligned} \mathcal{G}^{1} &= \frac{L}{2\sqrt{\Xi}} y_{1} \sin^{2} \zeta \cos \zeta \sin \theta d\xi, \\ \mathcal{G}^{2} &= -\frac{2}{L\sqrt{\Delta}\sqrt{\mathcal{Z}}} \left(\mathcal{Z} dy_{1} + y_{1} y_{2} dy_{2} \right), \\ \mathcal{G}^{3} &= -\frac{L}{2\sqrt{\mathcal{Z}}} \sin \zeta dy_{2}, \\ \mathcal{G}^{4} &= \frac{2}{L \cos \zeta \sqrt{\Delta}\sqrt{\Xi}} \left[\Delta d\phi - \sin^{2} \zeta \cos^{2} \zeta \cos \theta \left\{ y_{1} y_{2} dy_{1} + \left(y_{2}^{2} + \frac{L^{4}}{16} \sin^{4} \zeta \right) dy_{2} \right\} \right], \quad (3.2) \end{aligned}$$
 and we have defined

а

$$\Delta = \sin^2 \zeta \left(y_1^2 + \cos^2 \zeta y_2^2 + \frac{L^4}{16} \sin^4 \zeta \cos^2 \zeta \right), \qquad \Xi = \Delta \sin^2 \theta + y_1^2 \sin^4 \zeta \cos^2 \theta, \quad (3.3)$$
$$\mathcal{Z} = y_1^2 + \frac{L^4}{16} \sin^4 \zeta \cos^2 \zeta,$$

and

$$y_1 = r \sin \chi, \qquad y_2 = r \cos \chi, \qquad (3.4)$$

so that we have

$$64L^{2}\sum_{i=1}^{4} \left(\mathcal{G}^{i}\right)^{2} = \frac{1}{\Delta\Xi}\cos^{2}\zeta\left\{16\Delta\sec^{2}\zeta d\phi + \sin^{2}\zeta\cos\theta\left[L^{4}\sin^{4}\zeta(\cos\chi dr - r\sin\chi d\chi)\right.\right.\\ \left. + 16r^{2}\cos\chi dr\right]\right\}^{2} + \frac{16L^{4}}{\Xi}r^{2}\sin^{4}\zeta\cos^{2}\zeta\sin^{2}\theta\sin^{2}\chi\,d\xi^{2} \\ \left. + \frac{16L^{4}}{\mathcal{Z}}\sin^{2}\zeta(\cos\chi\,dr - r\sin\chi\,d\chi)^{2} \\ \left. + \frac{256}{\Delta\mathcal{Z}}\left[\frac{L^{4}}{16}\sin^{4}\zeta\cos^{2}\zeta(\sin\chi dr + r\cos\chi d\chi) + r^{2}\sin\chi dr\right]^{2}.$$

$$(3.5)$$

⁵Also $\theta_1 \equiv \theta$ and $S^2 \equiv \tilde{S}^2$.

The NS 2 form is given by

$$B_2 = \frac{r^2}{\Xi} \sin^2 \zeta \sin \chi \bigg[\sin^2 \zeta \sin \chi \big(\cos \theta d\xi \wedge d\phi - \cos \chi d\xi \wedge dr \big) - r \big(\cos^2 \zeta \sin^2 \theta + \sin^2 \zeta \sin^2 \chi \big) d\chi \wedge d\xi \bigg], \qquad (3.6)$$

while the dilaton is

$$e^{\Phi} = \frac{4}{kL\cos\zeta\sqrt{\Xi}}.$$
(3.7)

Notice that this blows up at $\zeta = 0$ indicating that the geometry is singular here, which is confirmed when one studies the curvature invariants.

The RR sector is given by

$$\hat{F}_2 = \frac{k}{16} \bigg[3L^4 \sin^3 \zeta \cos^3 \zeta \sin \theta d\zeta \wedge d\theta + 8r \sin 2\zeta \big(\cos \theta d\zeta \wedge dr - \cos \chi d\zeta \wedge d\phi \big) + 8r \cos^2 \zeta \sin \theta d\theta \wedge dr - 8 \sin^2 \zeta \big(\cos \chi dr \wedge d\phi + r \sin \chi d\phi \wedge d\chi \big) \bigg],$$
(3.8)

$$\hat{F}_4 = \frac{k}{2} r \cos \zeta \sin \chi \bigg[2 \sin \zeta \bigg(\sin \chi d\zeta \wedge d\xi \wedge dr \wedge d\phi + r \cos \theta d\zeta \wedge d\xi \wedge dr \wedge d\chi r \cos \chi d\zeta \wedge d\xi \wedge d\phi \wedge d\chi \bigg) + r \cos \zeta \sin \theta d\theta \wedge d\xi \wedge dr \wedge d\chi \bigg],$$
(3.9)

where the gauge invariant fluxes are expressed in terms of these via $\hat{F} = F \wedge e^{-B_2}$.

3.1 Supersymmetry

It was shown in [11, 13] that the NAT dual of ABJM preserves $\mathcal{N} = 2$ supersymmetry in 3d, which means the R-symmetry is U(1) in the dual CFT. The argument relies on a proof from [13]. In order to see this we must package all the dependence of the original geometry on the SU(2) isometry in a canonical frame⁶

$$e^{a+3} = e^{C_a(x)}(\omega_a + A_a(x)) \tag{3.10}$$

where each left-invariant one form ω_a appears only once, and x^{μ} are coordinates on some 7d base which fibers the squashed 3-sphere containing the SU(2) isometry. Then there is a bijective map between spinors independent of the SU(2) directions in this frame and those preserved by the NAT dual solution. The map acts on the 10 dimensional MW Killing spinors as

$$\hat{\epsilon}_1 = \epsilon_1, \qquad \hat{\epsilon}_2 = \Omega_{\mathrm{SU}(2)} \epsilon_2, \tag{3.11}$$

with the matrix⁷

$$\Omega_{\rm SU(2)} = \Gamma^{(10)} \frac{-e^{C_1 + C_2 + C_3} \Gamma^{456} + v_a e^{C_a} \Gamma^{a+3}}{\sqrt{e^{2(C_1 + C_2 + C_3)} + e^{2C_a} v_a^2}},\tag{3.12}$$

⁶We write e^{a+3} to match notation elsewhere where the canonical vielbeins are e^4, e^5, e^6 .

 $^{^7\}mathrm{This}$ expression originally appeared in [5], where it was conjectured to hold by analogy with the Abelian case.

where v_a are dual coordinates in the NAT dual geometry, which we are expressing elsewhere in terms of spherical or cylindrical polar coordinates $v_1 = y_1 \cos \xi$, $v_2 = y_2 \sin \xi$, $v_3 = y_2$.

In appendix C.1 we derive a spinor for ABJM independent of the SU(2) directions. This may be written in terms of 6 dimensional MW spinors on \mathbb{CP}^3 as in appendix B

$$\eta_{+}^{1} = e^{i\frac{3\pi}{4}}\eta_{+}, \qquad \eta_{+}^{2} = e^{-i\frac{3\pi}{4}}\eta_{+}$$
(3.13)

where $(\eta_{+}^{1,2})^* = \eta_{-}^{1,2}$ with the sign labeling chirality. It is possible to decompose the 6d spinors in terms of two in linearly independent parts $\eta_{+} = \pi_{+} + \tilde{\pi}_{+}$ obeying the projections of eq. (C.32). We can then make the coordinate dependence explicit as

$$\pi_{+} = e^{\zeta \gamma^{34}} \pi_{+}^{0}, \qquad \tilde{\pi}_{+} = e^{-\zeta \gamma^{34}} \tilde{\pi}_{+}^{0}, \qquad (3.14)$$

where we have introduced linearly independent constant spinors obeying the projections

$$\gamma^{1456}\pi^{0}_{+} = -\pi^{0}_{+}, \qquad \gamma^{2345}\pi^{0}_{+} = \pi^{0}_{+},$$

$$\gamma^{1456}\tilde{\pi}^{0}_{+} = -\tilde{\pi}^{0}_{+}, \qquad \gamma^{2345}\tilde{\pi}^{0}_{+} = \tilde{\pi}^{0}_{+}, \qquad (3.15)$$

in the frame of eq. (A.3). The 10d spinor is constructed as in eq. (B.3), but all dependence on the \mathbb{CP}^3 directions is in eq. (3.14), which is clearly independent of the SU(2) directions.

So $\mathcal{N} = 2$ is preserved under the NAT duality transformation. We show in appendices C.2, C.3 that the solution does this by mapping a U(1)'s worth of the SU(3)-structures supported by \mathbb{CP}^3 to a U(1)'s worth of dynamical SU(2)-structures defined on the dual internal space \hat{M}_6 . Of course only two of these dual objects are truly distinct: those that are defined in terms of the two linearly independent Killing spinors on \hat{M}_6 . These may be summed to give the internal part of the $\mathcal{N} = 2$ Killing spinor in IIB, namely

$$\hat{\eta}_{+}^{1} = e^{i\frac{3\pi}{4}} \left(e^{\zeta\gamma^{34}} \pi_{+}^{0} + e^{-\zeta\gamma^{34}} \tilde{\pi}_{+}^{0} \right),$$

$$\hat{\eta}_{+}^{2} = e^{i\frac{3\pi}{4}} \left(\kappa_{\parallel} \left(e^{\zeta\gamma^{34}} \pi_{+}^{0} + e^{-\zeta\gamma^{34}} \tilde{\pi}_{+}^{0} \right) + \kappa_{\perp} \mathcal{J} \left(e^{-\zeta\gamma^{34}} \pi_{-}^{0} + e^{\zeta\gamma^{34}} \tilde{\pi}_{-}^{0} \right) \right),$$
(3.16)

where κ_{\perp} and κ_{\parallel} satisfy $\kappa_{\perp}^2 + \kappa_{\parallel}^2 = 1$ and are given in eq. (C.27). The Matrix \mathcal{J} is defined as

$$\mathcal{J} = i \frac{e^{C_1 + C_2 + C_3} \cos 2\zeta \gamma^1 + e^{C_1} v_1 \gamma^4 + e^{C_2} v_2 \gamma^5 + e^{C_3} v_3 \gamma^6}{\sqrt{e^{2(C_1 + C_2 + C_3)} \cos^2 2\zeta + e^{2C_1} v_1^2 + e^{2C_2} v_2^2 + e^{2C_3} v_3^3}},$$
(3.17)

in the frame of eq. (A.3), where e^{C_a} are given in eq. (A.2). However for what follows it is only important that \mathcal{J} is independent of ϕ .

It turns out that the amount of preserved supersymmetry is left invariant when we perform an additional T-duality on ∂_{ϕ} . As proved in [55] (see also [13]), to see this it is sufficient to show that the Killing spinors are independent of ϕ in the canonical frame of T-duality, where ϕ only appears in the vielbein

$$e^{\phi} = e^{C(x)} (d\phi + A(x)).$$
 (3.18)

We get to such a frame by performing a SO(4) transformation \mathcal{R} in eq. (A.14), on the canonical NAT dual vielbeins in eq. (A.8). The action of this rotation on the 10d Majorana Killing spinor will be $\epsilon \to S\epsilon$, where

$$\mathcal{S}^{-1}\gamma^a \mathcal{S} = \mathcal{R}^a_{\ b}\gamma^b. \tag{3.19}$$

The matrix S will be complicated but will not depend on ϕ because \mathcal{R} does not, which is all that matters. This together with the fact that eq. (3.16) is ϕ independent ensures all supersymmetry must be preserved. Indeed in the next section (see also appendix C.4) we see that upon lifting to M-theory the NAT-T dual solution preserves $\mathcal{N} = 2$ supersymmetry in the form of a local SU(2) structure in 7d of the form given in [45].

3.2 Properties of the CFT

In this section we briefly discuss some properties of the CFT associated to the NAT-T dual solution. The discussion follows very closely the analysis in reference [11] for the IIB NAT solution. For this reason we will omit most of the explicit computations. The reader is referred to this reference for more details.

It was shown in [8, 11] that the presence of large gauge transformations in NAT dual backgrounds allows to constrain quite non-trivially their global properties. In our particular background (see [11]) it is easy to see that at the singularity $\zeta = 0$ the NS 2-form given by (3.6) reduces to

$$B_2|_{\zeta \sim 0} = -r \operatorname{Vol}(S^2) \,, \tag{3.20}$$

while the space spanned by (ζ, S^2) becomes conformal to a singular cone with boundary S^2 . Therefore large gauge transformations can be defined on this non-trivial 2-cycle, which must render

$$b = \frac{1}{4\pi^2} \left| \int_{S^2} B_2 \right| \tag{3.21}$$

in the fundamental region. For this, B_2 must transform into $B_2 \to B_2 + n\pi \text{Vol}(S^2)$ when $r \in [n\pi, (n+1)\pi]$.

In turn, the \hat{F}_3 and \hat{F}_5 field strengths lying on the ζ, θ, ϕ and ζ, θ, ϕ, S^2 directions of the NAT dual solution in [11] give rise after the T-duality to \hat{F}_2 and \hat{F}_4 field strengths lying on the ζ, θ and ζ, θ, S^2 directions, with the second one non-vanishing only in the presence of large gauge transformations. Accordingly, a Page charge associated to the ζ, θ components of \hat{F}_2 is generated in IIA from the quantization condition

$$\frac{1}{2\kappa_{10}^2} \int \hat{F}_p = T_{8-p} N_{8-p} \,. \tag{3.22}$$

As in [11] this charge is to be interpreted as the rank of the gauge group of the CFT dual to the solution in the $r \in [0, \pi]$ interval. We indeed get $N_6 = N_5$, with N_5 given by (3.18) in [11]. Specifically this fixes L to satisfy

$$kL^4 = 64\pi N_6. \tag{3.23}$$

Note that this means that color branes are now D6-branes spanned on the $\mathbb{R}_{1,2} \times M_1 \times S_{\phi}^1 \times S^2$ directions. One can indeed check that these branes are BPS when placed at the $\zeta = 0$ singularity. As in [11] the combination $e^{-\phi}\sqrt{g_{rr}}$ in the DBI action is non-singular, rendering well-defined color branes.

In the presence of large gauge transformations with parameter n there is also a nonvanishing D4-brane charge $N_4 = nN_6$, equal to the N_3 charge in [11]. Indeed one can check that D4-branes are also BPS when placed at $\zeta = 0$. These branes should also play a role as color branes for $n \neq 0$, that is, in the $r \in [n\pi, (n+1)\pi]$ intervals. The physical interpretation is that N_4 charge is created in the worldvolume of a D6 when it crosses nNS5. It is then plausible that the field theory dual to the solution in the $[n\pi, (n+1)\pi]$ intervals arises in a (D4, D6) bound state — NS5 intersection. A similar realization was suggested in [17] for AdS_3 duals.

The charge interpreted as level in ABJM is also doubled under the NAT duality transformation. As a result, after the new Abelian T-duality we get two charges, k_6 , k_4 , associated to the (r, θ) and (r, θ, S^2) components of \hat{F}_2 and \hat{F}_4 , respectively. They thus correspond to D6 and D4 branes or, equivalently, to D4-branes carrying both monopole and dipole charges. We may express the levels in terms of k and the number of large gauge transformations performed as

$$k_6 = k \frac{(2n+1)\pi}{4}, \qquad k_4 = k \frac{(3n+2)\pi}{12}.$$
 (3.24)

Finally, it is easy to check that particle-like brane configurations can be associated to each of the charges with an interpretation as either rank or level in the IIA background. These branes are in all cases D2 or D4 branes wrapped on different sub-manifolds of the internal space. In particular:

- Di-monopoles \leftrightarrow D2 on $M_1 \times S^1_{\phi}$, D4 on $M_1 \times S^1_{\phi} \times S^2$
- 't Hooft monopoles \leftrightarrow D2 on $\{M_1, \theta\}$, D4 on $\{M_1, \theta, S^2\}$
- Di-baryons \leftrightarrow D2 on $\{\zeta, S^1_{\phi}\}$, D4 on $\{\zeta, S^1_{\phi}, S^2\}$
- Baryon vertices \leftrightarrow D2 on $\{\zeta, \theta\}$, D4 on $\{\zeta, \theta, S^2\}$

As for the IIB AdS_4 solution (see [11] for the details) the di-monopoles and 't Hooft monopoles have to sit at $\zeta = 0$ while the di-baryons sit at r = 0.

The previous analysis suggests a dual CFT in the $r \in [0, \pi]$ region defined in terms of a $U(N_6)_{k_4} \times U(N_6)_{-k_4}$ quiver gauge theory with $\mathcal{N} = 2$ supersymmetry, sourced by D6-branes spanned on the $\mathbb{R}_{1,2} \times M_1 \times S_{\phi}^1 \times S^2$ directions.⁸ In turn, for $r \in [n\pi, (n+1)\pi]$ the gauge theory would arise from (D4, D6) - NS5 intersections. It was argued in [11] that invariance under large gauge transformations would imply that the seemingly different CFTs dual to the solution as the non-compact internal direction increases, could be related by some kind of duality, as in [39], with the essential difference that in this case the flow parameter would not be the energy scale but the non-compact internal direction. Reference [17] proposed an alternative mechanism which, applied to our solution, would imply that new $U(N_6) \times U(N_6)$ gauge groups would be created by some kind of un-higgsing mechanism, also not related to

⁸One can see (see [11]) that k_4 is the level associated to the D6 color branes.

an energy scale, every time a NS5-brane is crossed. It would be interesting to understand better these proposals for the dual CFT as r increases.

In any case, keeping in mind that there is no a priori reason to expect that the new geometry makes sense as a string theory background,⁹ we can just take these proposals as stringy inspired arguments in favor of the existence of a fundamental region in which the dual CFT would contain the same number of gauge groups as the original one.

Restricting ourselves to the $r \in [0, \pi]$ region, a candidate brane realization of the dual CFT would then be the T-dual of the brane picture proposed in [11] for the NAT dual of ABJM:

$$5_{2}^{2}: \qquad \times \times \times \times \times - z_{1} z_{2} - N_{6} D6: \qquad \times \times - x - x - x \times x - (3.25)$$
$$(5_{2}^{2}, k_{4} D4): \qquad \times \times - x \cos \theta - - - \sin \theta$$

where 5_2^2 denotes the IIA exotic brane that arises after a T-duality transformation along a worldvolume direction of the 5_2^2 exotic brane of the IIB configuration [11], and z_1 and z_2 denote the two special Killing directions of this brane [49, 50]. In our notation the $(5_2^2, k_4 \text{ D4})$ bound state would be extended along the 0124 and $x^5 \cos \theta + x^9 \sin \theta$ directions, and its relative orientation w.r.t. the 5_2^2 -brane in the 59 plane would depend on k_4 .

Note that the previous picture implies that in M-theory the corresponding AdS_4 geometry would be sourced in the fundamental region $r \in [0, \pi]$ by Kaluza-Klein monopoles, as we discuss in the next section.

4 The purely magnetic AdS_4 solution in M-theory

In this section we lift the solution of the previous section to M-theory and show that it falls into the general class of solutions with purely magnetic flux considered in [43]. The analysis of quantized charges suggests a dual CFT arising from Kaluza-Klein monopoles and M5-branes wrapped on the Taub-NUT direction of the monopoles. We compute the central charge and show that it scales with $(N_5 + N_6/2)^{3/2}$, where N_5 is the number of wrapped M5-branes and N_6 the number of Kaluza-Klein monopoles. This becomes simply $N_6^{3/2}$ in the fundamental region $r \in [0, \pi]$.

4.1 Fluxes

The RR potentials of the IIA solution are given by

$$C_1 = \frac{k}{16} \bigg(\cos^2 \zeta \cos \theta (3L^4 \sin^3 \zeta \cos \zeta d\zeta - 8rdr) - 8r \cos \chi \sin^2 \zeta d\phi \bigg), \qquad (4.1)$$

$$C_3 - B_2 \wedge C_1 = \frac{k}{2} r^2 \cos\zeta \sin\chi \bigg(\sin\zeta \sin\chi d\zeta \wedge d\xi \wedge d\phi - \cos\theta \cos\zeta d\xi \wedge dr \wedge d\chi \bigg).$$
(4.2)

 C_1 gives rise to the $g_{\mu z}/g_{zz}$ component of the 11d metric, where z denotes the eleventh direction. Given that there is a magnetic charge associated to C_1 in IIA, a quantized Taub-NUT charge arises in 11d. The brane that carries Taub-NUT charge is the Kaluza-Klein

 $^{^{9}}$ An essential difference with respect to its Abelian counterpart is that non-Abelian T-duality has not been proved to be a symmetry of String Theory (see [51]).

monopole, which is connected to the IIA D6-brane upon reduction along the eleventh, Taub-NUT direction. Since the IIA solution was sourced by D6-branes in the fundamental region $r \in [0, \pi]$, Kaluza-Klein monopoles should play the role of color branes in M-theory in this region. As we will discuss, BPS KK-monopoles spanned on the $\mathbb{R}_{1,2} \times M_1 \times S_{\phi}^1 \times S^2$ directions can indeed be constructed in 11d that give rise to the D6 color branes in IIA upon reduction.

 $(C_3 - B_2 \wedge C_1)$ gives rise in turn to the 3-form¹⁰

$$\hat{C}_3 = C_3 - i_k C_3 \wedge \frac{k_1}{k^2} \tag{4.3}$$

in 11d. Note that \hat{C}_3 has no components along the eleventh direction. This will be of relevance in our later discussion. The M-theory 4-form flux is then given by

$$G_4 = dC_3 = d\left(\hat{C}_3 + i_k C_3 \wedge \left(\frac{k_1}{k^2} + dz\right)\right) \tag{4.4}$$

which, as we can see, is purely magnetic. Therefore there will be no M2-branes sourcing the 11d solution.

As we have noted, \hat{C}_3 is by construction transverse to the eleventh direction. This potential couples in the worldvolume of M2-branes constrained to move in the space transverse to the Killing direction and in the worldvolume of M5-branes wrapped on this direction [46]. Moreover, its magnetic components are associated to wrapped M5-branes. Indeed one can show that these branes are BPS in the 11d background and are to be interpreted as color branes. They give rise upon reduction to the color D4-branes of the IIA background. Other field theory observables that we will be able to describe holographically will be constructed in terms of M2-branes transverse to the Killing direction or M5-branes wrapped on this direction.

4.2 Geometry and local SU(2) structure

In [45] it was shown that the most general $\mathcal{N} = 2$ preserving solution in M-theory with an AdS_4 factor supports an SU(2) structure in 7d. As the M-theory 4-form G_4 is purely magnetic it actually falls into the more constrained class of solutions originally considered in [43]. In this section we show that we can uplift the IIA solution to M-theory and fit it into this class of solutions.

The metric ansatz of [45] is of the form

$$ds_{11}^2 = e^{2\tilde{\Delta}} \bigg(ds^2 (AdS_4) + ds^2 (\mathcal{M}_7) \bigg)$$
(4.5)

where we have

$$e^{2\tilde{\Delta}} = L^2 e^{-\frac{2}{3}\Phi}, \qquad ds^2(\mathcal{M}_7) = \frac{1}{L^2} \left[ds^2(\mathcal{M}_6) + e^{2\Phi} \left(C_1 + dz \right)^2 \right],$$
 (4.6)

¹⁰Our notation is that $i_k C_3$ denotes the interior product of C_3 with the Killing vector $k^{\mu} = \delta_z^{\mu}$, that points on the eleventh direction, k_1 is the 1-form $k_1 = i_k g$ and k^2 the scalar $k^2 = i_k i_k g$, where g stands for the eleven dimensional metric.

so that $\operatorname{Ricci}(AdS_4) = -12 g(AdS_4)$ to match the conventions of [45]. The metric on \mathcal{M}_6 is mearly the internal part of the IIA metric in eq. (3.1).

It is possible to express the internal 7d metric in the form

$$ds^{2}(\mathcal{M}_{7}) = ds^{2}(\mathrm{SU}(2)) + E_{1}^{2} + E_{2}^{2} + E_{3}^{2}$$

$$(4.7)$$

where $ds^2(SU(2))$ is the metric on a 4 manifold supporting a canonical SU(2)-structure with associated real 2-form $J = J_3$ and holomorphic 2-form $\Omega = J_1 + iJ_2$, satisfying

$$J_3 \wedge J_3 = \frac{1}{2}\Omega \wedge \bar{\Omega}, \qquad J_3 \wedge \Omega = 0, \qquad \iota_{E_i} J_3 = \iota_{E_i} \Omega = 0.$$
(4.8)

Since G_4 is purely magnetic it is possible to define local coordinates such that [43]

$$E_1 = \frac{1}{4} e^{-3\tilde{\Delta}} \rho \, d\xi, \qquad E_2 = \frac{1}{4} e^{-3\tilde{\Delta}} \frac{d\rho}{\sqrt{1 - e^{-6\tilde{\Delta}} \rho^2}},\tag{4.9}$$

where ξ parametrizes the U(1) R-symmetry and ρ is defined through the associated Reeb vector $\tilde{\xi}$ as $|\tilde{\xi}|^2 = e^{-6\Delta}\rho^2$. Supersymmetry then requires that the SU(2) forms satisfy

$$d(e^{3\tilde{\Delta}}\sqrt{1-|\tilde{\xi}|^{2}E_{3}}) = e^{3\tilde{\Delta}}(2J_{3}-2|\tilde{\xi}|E_{2}\wedge E_{3}),$$

$$d(|\tilde{\xi}|^{2}e^{9\tilde{\Delta}}J_{2}\wedge E_{2}) = e^{3\tilde{\Delta}}|\tilde{\xi}|d(e^{6\tilde{\Delta}}J_{1}\wedge E_{3}),$$

$$d(e^{6\tilde{\Delta}}J_{1}\wedge E_{2}) = -e^{3\tilde{\Delta}}|\tilde{\xi}|d(e^{3\tilde{\Delta}}J_{2}\wedge E_{3}),$$
(4.10)

and the flux be given by

$$G_4 = \frac{1}{4} d\xi \wedge d \left(e^{3\tilde{\Delta}} \sqrt{1 - |\tilde{\xi}|^2} J_1 \right). \tag{4.11}$$

We find that the uplift of the IIA solution fits into the above parametrisation. All the forms are defined in terms of the internal M-theory Killing spinors derived in appendix C.4, one needs only to plug them into the bi-linears in appendix B of [45]. Performing these steps with some liberal application of Mathematica, we find the local coordinate

$$\rho = \frac{k}{8}L^4 y_1 \sin^2 2\zeta \sin \theta, \qquad (4.12)$$

and the solutions specific vielbein

$$E_{3} = -\frac{e^{-3\tilde{\Delta}}}{\sqrt{1 - e^{-6\tilde{\Delta}}\rho^{2}}} \left[\frac{kL^{2}}{256} \sin^{2} 2\zeta \left(L^{4} \sin^{2} 2\zeta \sin \theta d\theta + 64y_{2} d\phi - 64(y_{1} dy_{1} + y_{2} dy_{2}) \cos \theta \right) + L^{2} \cos 2\zeta (dz + C_{1}) \right],$$

$$(4.13)$$

where C_1 is the potential giving rise to the IIA RR 2-form, which may be found in eq. (4.1). To express the SU(2) forms we introduce the following orthonormal frame

$$e^{1} = \frac{1}{\sqrt{X_{1}}\sqrt{X_{2}}} \left(2X_{1}dy_{2} + 32y_{1}y_{2}\cos^{2}\zeta\sin\theta\left(\sin\theta dy_{1} + y_{1}\cos\theta d\theta\right) \right),$$
(4.14)

$$e^{2} = \frac{e^{-3\tilde{\Delta}}kL^{2}}{8\sqrt{X_{1}}\sqrt{X_{2}}} \left(\cos^{2}\zeta\cos\theta\left(X_{1}(16y_{2}^{2} + L^{4}\sin^{4}\zeta)dy_{2} - 256y_{1}^{3}y_{2}\sin^{2}\theta dy_{1} \right) - X_{2}\sin^{2}\zeta d\phi \right)$$
$$- \frac{32y_{2}e^{3\tilde{\Delta}}}{kL^{6}\sin^{2}\zeta\sqrt{X_{1}}\sqrt{X_{2}}} \left(16y_{2}\cos\theta dy_{2} - L^{4}\cos^{2}\zeta\sin^{2}\zeta\sin\theta d\theta \right),$$
$$e^{3} = \frac{e^{-3\tilde{\Delta}}kL^{4}\sin 2\zeta}{32\sqrt{X_{1}}\sqrt{1 - e^{-6\Delta}\rho^{2}}} \left(-X_{1}d\zeta + 4y_{1}\sin 2\zeta\cos 2\zeta\sin\theta\left(\sin\theta dy_{1} + y_{1}\cos\theta d\theta\right) \right),$$
$$e^{4} = \frac{\cos 2\zeta}{2L^{2}\sqrt{X_{1}}} \left(16y_{2}d\phi + L^{4}\cos^{2}\zeta\sin^{2}\zeta\sin\theta d\theta - 16\cos\theta\left(y_{1}dy_{1} + y_{2}dy_{2}\right) \right)$$
$$- \frac{kL^{6}\cos^{2}\zeta\sin^{2}\zeta\sqrt{X_{1}}e^{-3\tilde{\Delta}}}{8\sqrt{1 - e^{-6\Delta}\rho^{2}}} \left(dz + C_{1} \right),$$

where

$$X_{1} = 16y_{1}^{2}\cos^{2}\theta\sin^{2}\zeta + 16y_{2}^{2}\sin^{2}\theta\cos^{2}\zeta + L^{4}\sin^{2}\theta\sin^{4}\zeta\cos^{2}\zeta, \qquad (4.15)$$

$$X_{2} = 16y_{1}^{2}\cos^{2}\theta(16y_{1}^{2} + L^{4}\sin^{4}\zeta\cos^{2}\zeta) + L^{4}\cos^{2}\zeta\sin^{2}\zeta\sin^{2}\theta(16y_{1}^{2} + \cos^{2}\zeta(16y_{2}^{2} + L^{4}\sin^{4}\zeta)).$$

With respect to this basis we have

$$J = e^{1} \wedge e^{2} + e^{3} \wedge e^{4}, \qquad \Omega = e^{i\alpha}(e^{1} + ie^{2}) \wedge (e^{3} + ie^{4}), \tag{4.16}$$

where the phase α is defined through

$$\tan \alpha = -\frac{e^{3\tilde{\Delta}}}{kL^2 \cos \theta \, y_1^2}.\tag{4.17}$$

4.3 Properties of the CFT

As in the previous section, some properties of the CFT dual can be inferred by analyzing the 11d supergravity solution. The picture that arises is simply the 11d realization of the IIA picture described in subsection 3.2, apart from some subtleties that have to do with the existence of the special Taub-NUT direction. Indeed, all brane configurations that play a role in 11d will be either transverse to this direction or wrapped on it.

The non-trivial S^2 of the IIA geometry is also present in the 11d uplift. Therefore one can define large gauge transformations for the uplift of the B_2 field, which is the 11d 3-form potential with a component along the Taub-NUT direction, i_kC_3 . Thus, as in the IIA background, we need to divide r in intervals of length π in order to have i_kC_3 lying in the fundamental region. From here the discussion parallels exactly the IIA discussion.

In 11d we find quantized charges N_6 and $N_5 = nN_6$, equal to the N_6 and N_4 , respectively, in IIA. N_6 is associated to KK-monopoles and N_5 to M5-branes wrapped on

the Taub-NUT direction of the monopole. The interpretation is that M5-brane charge (with the M5-brane wrapped in the Taub-NUT direction of the monopole) is created in the worldvolume of the KK-monopole when it crosses M5-branes transverse to the Taub-NUT direction.¹¹ Using the worldvolume effective action that describes a KK-monopole in 11d [47, 48] one can easily check that it is BPS when placed at $\zeta = 0$. The calculation parallels the D6-brane calculation in IIA with the only difference that the action is now written in terms of eleven dimensional fields. Similarly an M5-brane wrapped on the Taub-NUT direction is also BPS at this location.

As in IIA, the charge interpreted as level in 11d is also doubled, and we get two values k_6 and k_5 equal to the k_6 and k_4 charges, respectively, in IIA. These are now associated to wrapped M5-branes carrying monopole and dipole charges.¹²

Similarly, we find particle-like brane configurations, which are either M2-branes transverse to the Taub-NUT direction or M5-branes wrapped on this direction. These branes are wrapped on the same sub-manifolds of the internal space as in IIA. Namely,

- Di-monopoles \leftrightarrow M2 on $M_1 \times S_{\phi}^1$, M5 on $M_1 \times S_{\phi}^1 \times S^2 \times S_z^1$
- 't Hooft monopoles \leftrightarrow M2 on $\{M_1, \theta\}$, M5 on $\{M_1, \theta, S^2, S_z^1\}$
- Di-baryons \leftrightarrow M2 on $\{\zeta, S_{\phi}^1\}$, M5 on $\{\zeta, S_{\phi}^1, S^2, S_z^1\}$
- Baryon vertices \leftrightarrow M2 on $\{\zeta, \theta\}$, M5 on $\{\zeta, \theta, S^2, S_z^1\}$

As for the IIB AdS_4 solution (see [11] for the details) the di-monopoles and 't Hooft monopoles have to sit at $\zeta = 0$ while the di-baryons sit at r = 0. In these derivations we have used the action that describes M2-branes transverse to the Taub-NUT direction of the monopole. In this action i_kC_3 couples in both the DBI and CS parts, so the M2-branes are sensitive to large gauge transformations. The details of this action can be found in [46].

Putting together this information, and in analogy with the IIA discussion, we expect a field theory in the $r \in [0, \pi]$ region described by a $U(N_6)_{k_5} \times U(N_6)_{-k_5}$ quiver with $\mathcal{N} = 2$ supersymmetry, sourced by KK-monopoles spanned on the $\mathbb{R}_{1,2} \times M_1 \times S^2 \times S_{\phi}^1$ directions, and with Taub-NUT direction z. A possible brane realization in the fundamental region $r \in [0, \pi]$ could be

$$5^{3}: \qquad \times \times \times \times \times - z_{1} z_{2} - z$$

$$N_{6} M_{6}: \qquad \times \times - \times - \times \times - z$$

$$(5^{3}, k_{4} M_{5}): \qquad \times \times - \times \cos \theta - - - \sin \theta \times$$

$$(4.18)$$

where z denotes the eleventh direction, the M6 is the Kaluza-Klein monopole with Taub-NUT direction z and 5^3 is the exotic brane that gives rise to the IIA 5^2_2 brane upon reduction [49, 50].

⁻¹¹Recall that in 11d $i_k C_3 \rightarrow i_k C_3 + n\pi \text{Vol}(S^2)$, and the M5 is magnetically charged with respect to this field.

 $^{^{12}\}mathrm{As}$ shown in [46], M5-branes wrapped on an isometric direction can carry KK-monopole charge, with the Taub-NUT direction equal to the isometric direction.

4.4 Free energy

We can now calculate the free energy on a 3-sphere in the CFT dual to the solution in M-theory. This is expressed in terms of the effective 4 dimensional Newton constant G_4 as

$$\mathcal{F}_{S^3} = \frac{\pi}{2G_4}.$$
(4.19)

One can determine G_4 via a dimensional reduction of supergravity on the internal space \mathcal{M}_7 , the result is

$$\frac{1}{16\pi G_4} = \frac{\pi}{2(2\pi)^9} \int_{\mathcal{M}_7} e^{9\tilde{\Delta}} \text{Vol}(\mathcal{M}_7), \qquad (4.20)$$

where we work in units such that $l_p = 1$. For the case at hand the relevant quantity is

$$e^{9\tilde{\Delta}} \operatorname{Vol}(\mathcal{M}_7) = \frac{k^2 L^6}{32} r^2 \sin^3 \zeta \cos^3 \zeta \sin \theta \sin \chi d\zeta \wedge d\theta \wedge d\phi \wedge dr \wedge d\chi \wedge d\xi \wedge dz.$$
(4.21)

Integrating this in the region $r \in [n\pi, (n+1)\pi]$, $z \in [0, 2\pi]$ and using eqs. (3.23), (3.24) we arrive at

$$\mathcal{F}_{S^3} = \frac{\sqrt{2\pi}}{36} \left(12 + \frac{N_6^2}{\left(N_5 + \frac{N_6}{2}\right)^2} \right) \sqrt{\kappa_6} \left(N_5 + \frac{N_6}{2}\right)^{3/2}.$$
(4.22)

This reproduces, as expected, the result in IIB, with $N_5, N_3 \to N_6, N_5$ [11]. Essentially we have $\mathcal{F}_{S^3} \sim \left(N_5 + \frac{N_6}{2}\right)^{3/2}$ which reproduces the $N^{3/2}$ behavior of ABJM. In particular, in the fundamental region $r \in [0, \pi]$, where $N_5 = 0$, we find

$$\mathcal{F}_{S^3} = \frac{\sqrt{2\pi}}{3^{3/2}} \sqrt{k_4} N_6^{3/2} \,. \tag{4.23}$$

This is not a surprising result, given that the dependence of the free energy in type II theories, like the central charge and entanglement entropy of the strip, depends on the internal directions only through the quantity

$$V_{\rm int} = \int_{\mathcal{M}_6} e^{-2\Phi} \operatorname{Vol}(\mathcal{M}_6), \qquad (4.24)$$

and is thus invariant under Abelian T-duality¹³ and uplift to 11d.

So, quite surprisingly, we have found an AdS_4 M-theory solution with purely magnetic flux that falls in the general classification of [43], that originates in M5-branes wrapped on calibrated 3-cycles, but whose free energy does not exhibit the expected N^3 behavior. We leave a further discussion on this issue for the conclusions.

¹³It is not invariant though under non-Abelian T-duality, because even if the integrand is invariant, the S^3 on which the dualisation is performed is transformed into an $M_1 \times S^2$ space, where M_1 is the space spanned by the *r*-direction, and thus the domain of integration changes. This is the reason why the prefactors in (4.23) are not the same as in ABJM.

5 Conclusions

In this work we have presented a new warped AdS_4 solution of M-theory preserving $\mathcal{N} = 2$ supersymmetry, giving the only known solution in this class other than the uplift of Pernici-Sezgin. A legitimate question to ask is whether this solution is truly distinct from Pernici-Sezgin, indeed this solution was generated by performing first a NAT then T duality on $AdS_4 \times \mathbb{CP}^3$, and some geometries derived via NAT duality have been shown to fall within the ansatz of previous solutions. This does not seem to be the case with this example: the quickest thing to note is that the free energy of Pernici-Sezgin scales as N^3 while this solution scales as $N^{3/2}$. Additionally the uplift of Pernici-Sezgin is everywhere non singular while the curvature invariants of this solution blow up in certain regions of parameter space. One might still wonder if this solution approximates Pernici-Sezgin at least locally away from the singularity, as was argued in [1] to be the case for the NAT dual of $AdS_5 \times S^5$ and the Gaiotto-Maldacena geometries [53]. This also does not seem to be the case. Sfetsos and Thompson were able to find an additional solution to the Gaiotto-Maldacena Toda equation which gave their solution. The differential equations giving rise to Pernici-Sezgin are more simple and are solved uniquely. So this solution is truly distinct.

In this work, following on [11], we have taken the view that the range of r is restricted to lie in a specific cell of length π after n large gauge transformations of B_2 . The reason is to ensure that $0 < |\int_{S^2} B_2| < 4\pi^2$, a restriction motivated by string theory. However this does present an issue for the geometry, we are choosing to end it at a regular point which would usually demand the inclusion of extra localized sources. From a purely geometric view point we might choose to take $0 < r < \infty$, however this would be very undesirable from an AdS/CFT perspective. A continuous r would lead to, among other things, a CFT dual with operators of continuous conformal dimension [8]. An attractive resolution to these issues is that the NAT duality generates a solution which approximates a better defined solution free of these pathologies. At any rate, regardless of these potential criticisms, it seems likely that one could use this work as a stepping stone to further populate the solution space of purely magnetic M-theory solutions.

Supersymmetric probe branes in the 11d uplift of the Pernici-Sezgin solution were considered in [52], with an aim at introducing punctures on the Riemann surface along the lines of [53]. The BPS configurations were shown to preserve two U(1)'s, one more than required by the R-symmetry of the 3d $\mathcal{N} = 2$ SCFT. This second U(1) corresponds to a global U(1) in the 3d field theory, and seems to play a key role in the 3d-3d correspondence [54]. It was argued in [52] that a large number of supersymmetric M5's would ultimately backreact on the Pernici-Sezgin geometry to produce a new AdS_4 solution with a U(1)² isometry. It would be interesting to show whether the AdS_4 solution obtained in this paper, containing a U(1)² isometry, could be related to this physical situation.

That the free energy of our purely magnetic AdS_4 solution scales like $\mathcal{F}_{S^3} \sim N^{3/2}$ rather than N^3 is a little puzzling. It was proved in [45] that the presence of M2 branes, whether accompanied by M5's or not, always gives rise to $N^{3/2}$ behavior. However we know that our solution cannot contain M2 branes, indeed it is not possible to accommodate M2 branes in a purely magnetic flux ansatz, so what are we to make of this apparent contradiction. Firstly it should be noted that, at least as far as the authors are aware, there is no proof $\mathcal{F}_{S^3} \sim N^3$ holds universally for all wrapped M5 brane solutions. However this seems like an inadequate evasion of a confusing result. More likely is that the solution we present does not correspond to wrapped M5 branes. Indeed, the ansatz taken in [43] to derive the purely magnetic solutions is defined by requiring the Killing spinors to satisfy the same projection conditions as the wrapped branes. Yet the solutions need not describe in general M5-branes wrapped in 3d manifolds in the near horizon limit. The metric we have obtained is rather complicated and it seems difficult to identify a 3-cycle in the internal geometry that such branes might wrap. This together with the fact that the free energy does not scale with N^3 is suggesting that this is indeed the case for our solution.

On the other hand, even if the CFT interpretation of the solution is yet very preliminary, we seem to have found that there are quantized charges associated to both KK-monopoles and M5-branes, with the first being the only sources of the geometry in the r-region that we have defined as the fundamental region. This is also suggestive of a geometry not originating from wrapped M5-branes.

Finally, let us comment on something slightly tangential. In the process of discussing the supersymmetry preserved by purely magnetic M-theory solutions we analised the Gstructure preserved by the NAT dual of ABJM. We showed in appendix C.3 that this IIB solution preserves a U(1)'s worth of dynamical SU(2)-structures in 6d. We note that, it is possible to take the intersection of two of these and define an identity structure. However, given that a complete systematic study of AdS_4 solutions to type II supergravity preserving $\mathcal{N} = 2$ supersymmetry is currently absent form the literature, we have not pursued this here. Even so we know that, as with the better studied AdS_5 , $\mathcal{N} = 1$ cases [61, 62], supersymmetry should be preserved in terms of either a local "SU(2)-structure" or "identity structure" on the internal co-dimensions of the isometry dual to the U(1) R-symmetry. The NAT dual of ABJM will certainly fall into the latter class.

Acknowledgments

We would like to thank Jerome Gauntlett, Dario Martelli, Achilleas Passias, Alessandro Tomasiello and Alberto Zaffaroni for useful discussions. This work has been partially supported by the COST Action MP1210 "The String Theory Universe". Y.L. and J.M. are partially supported by the Spanish Ministry of Economy and Competitiveness grant FPA2012-35043-C02-02. J.M. is supported by the grant BES-2013-064815 of the same institution. He is grateful for the warm hospitality extended by the theoretical physics group at Milano-Bicocca U. where part of this work was done. N.T.M. is supported by INFN and by the European Research Council under the European Union's Seventh Framework Program (FP/2007-2013) — ERC Grant Agreement n. 307286 (XD-STRING).

A Some details on the NAT and T duality transformations

In this appendix we give some details on the derivation of the solution in section 3. The starting point is the $AdS_4 \times \mathbb{CP}^3$ metric written as a Hopf fibration

$$ds^{2} = ds^{2}(M_{7}) + e^{2C_{a}}(\omega_{a} + A_{a})^{2}, \qquad (A.1)$$

where ω_a are SU(2) left-invariant 1-forms satisfying $d\omega_a = \frac{1}{2} \epsilon_{abc} \omega_b \wedge \omega_c$ and

$$ds^{2}(M_{7}) = \frac{L^{2}}{4} \left[ds^{2}(AdS_{4}) + 4d\zeta^{2} + \cos^{2}\zeta \left(d\theta^{2} + \cos^{2}\theta d\phi^{2} \right) \right],$$

$$e^{2C_{1}} = e^{2C_{2}} = \frac{L^{2}}{4} \sin^{2}\zeta, \qquad e^{2C_{3}} = \frac{L^{2}}{4} \sin^{2}\zeta \cos^{2}\zeta,$$

$$A_{1} = A_{2} = 0, \qquad A_{3} = \cos\theta d\phi, \qquad (A.2)$$

where the AdS radius is 1. Specifically we introduce the vielbeins

$$e^{x^{\mu}} = \frac{L}{2}\rho \,dx^{\mu}, \qquad e^{\rho} = \frac{L}{2\rho}d\rho, \qquad e^{1} = Ld\zeta, \qquad e^{2} = \frac{L}{2}\cos\zeta d\theta, \qquad e^{3} = \frac{L}{2}\cos\zeta\sin\theta d\phi,$$
$$e^{4} = \frac{L}{2}\sin\zeta\omega_{1}, \qquad e^{5} = \frac{L}{2}\sin\zeta\omega_{2}, \qquad e^{6} = \frac{L}{2}\sin\zeta\cos\zeta(\omega_{3} + \cos\theta d\phi). \tag{A.3}$$

The dilaton of this solution is constant and set to $e^{\Phi} = \frac{k}{L}$, while the non trivial fluxes are

$$F_{2} = G_{2} + J_{1}^{a} \wedge (\omega_{a} + A_{a}) + \frac{1}{2} \epsilon_{abc} K_{0}^{a} (\omega_{b} + A_{b}) \wedge (\omega_{c} + A_{c}),$$

$$F_{4} = G_{4} + K_{3}^{a} \wedge (\omega_{a} + A_{a}) + \frac{1}{2} \epsilon_{abc} M_{2}^{a} (\omega_{b} + A_{b}) \wedge (\omega_{c} + A_{c})$$

$$+ N_{1} (\omega_{1} + A_{1}) \wedge (\omega_{2} + A_{2}) \wedge (\omega_{3} + A_{3}), \qquad (A.4)$$

where the only non zero components are

$$G_2 = -\frac{k}{2}\cos^2\zeta\sin\theta d\theta \wedge d\phi, \qquad J_1^3 = -k\sin\zeta\cos\zeta d\zeta, \qquad K_0^3 = -\frac{k}{2}\sin^2\zeta,$$

$$G_4 = \frac{3kL^2}{8}\operatorname{Vol}(AdS_4). \tag{A.5}$$

A.1 The IIB NAT duality

Expressing the solution in this manner allows one to simply read off the result of performing a NAT duality transformation on ω_a using [13]. The dual metric is given by

$$d\hat{s}^2 = ds^2(M_7) + \sum_{a=1}^3 \hat{e}^{a+3}$$
(A.6)

We have introduced cylindrical polars for the dual coordinates

$$v_1 = y_1 \cos \xi, \qquad v_2 = y_1 \sin \xi, \qquad v_3 = y_2,$$
 (A.7)

and choose to express the dual canonical vielbeins \hat{e} in a way that makes the residual U(1) isometry given by ∂_{ξ} explicit

$$\cos \xi \hat{e}^{4} + \sin \xi \hat{e}^{5} = -\frac{1}{8L\Delta} \sin \zeta \bigg[4y_{1}y_{2} \big(4dy_{2} + L^{2} \sin^{2} \zeta \cos^{2} \zeta (d\xi + \cos \theta d\phi) \big) + dy_{1} \big(16y_{1}^{2} + L^{4} \sin^{4} \zeta \cos^{2} \zeta \big) \bigg],$$

$$\cos \xi \hat{e}^{5} - \sin \xi \hat{e}^{4} = -\frac{1}{8L\Delta} \sin \zeta \left[4y_{1} dy_{2} + \cos^{2} \zeta \left(-4y_{2} dy_{1} + L^{2} y_{1} \sin^{2} \zeta (d\xi + \cos \theta d\phi) \right) \right],$$
$$\hat{e}^{6} = -\frac{1}{8L\Delta} \sin \zeta \cos \zeta \left[16y_{1} y_{2} dy_{1} + dy_{2} (16y_{2}^{2} + L^{4} \sin^{4} \zeta) -4L^{2} y_{1}^{2} \sin^{2} \zeta (d\xi + \cos \theta d\phi) \right]$$
$$\hat{e}^{a} = e^{a}, \qquad a = x^{\mu}, \rho, 1, 2, 3,$$
(A.8)

where we define

$$\Delta = \sin^2 \zeta \left(y_1^2 + \cos^2 \zeta y_2^2 + \frac{L^4}{16} \sin^4 \zeta \cos^2 \zeta \right).$$
 (A.9)

A NS two form is generated

$$B_2 = \frac{1}{\Delta} y_1 \sin^2 \zeta \left(y_1 dy_2 - y_2 \cos^2 \zeta dy_1 \right) \wedge d\xi$$
$$- \frac{1}{\Delta} \sin^2 \zeta \cos^2 \zeta \cos \theta \left(y_1 y_2 dy_1 + \left(y_2^2 + \frac{L^4}{16} \sin^4 \zeta \right) dy_2 \right) \wedge d\phi$$

while the dilaton becomes 14

$$e^{-2\hat{\Phi}} = \frac{L^2}{4} \Delta e^{-2\Phi}.$$
 (A.10)

The solution also has all possible RR forms turned on. These can be found in [11] where this solution was originally derived.

A.2 The IIA NAT-T duality

We would now like to perform a T-duality on the global U(1) corresponding to ∂_{ϕ} . To do this we can once more make use of the results of [13] (see [55] for the original derivation). In order to do this we need to express the metric and B_2 as

$$d\hat{s}^{2} = d\hat{s}^{2}(M_{9}) + e^{2C}(d\phi + A_{1})^{2},$$

$$B_{2} = B + B_{1} \wedge d\phi.$$
(A.11)

Clearly B_2 is already in this form, while the same can be achieved for the metric with a rotation of the vielbein basis $\hat{e} \to \mathcal{R}\hat{e}$, giving

$$e^{2C} = \frac{\Xi}{4\Delta} L^2 \cos^2 \zeta, \qquad A_1 = \frac{y_1^2 \sin^4 \zeta \cos \theta}{\Xi} d\xi, \qquad (A.12)$$

where

$$\Xi = \Delta \sin^2 \theta + y_1^2 \sin^4 \zeta \cos^2 \theta. \tag{A.13}$$

¹⁴Notice that, for simplicity in other expressions, we are extracting a factor of $\frac{L^2}{4}$ with respect to the definition of Δ in [13].

A rotation that achieves this is

76

1

$$\mathcal{R} = \begin{pmatrix} -\frac{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2}} \sin \zeta \cos \theta}{\sqrt{\Xi_{0}}} & \frac{\sin \theta (\sin \xi - \zeta_{3} \cos \xi)}{\sqrt{\Xi_{0}}} & -\frac{\sin \theta (\zeta_{3} \sin \xi + \cos \xi)}{\sqrt{\Xi_{0}}} & \frac{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2}} \sin \theta}{\sqrt{\Xi_{0}}} \\ 0 & \frac{\zeta_{3} \sin \xi + (\zeta_{1}^{2} + \zeta_{2}^{2} + 1) \cos \xi}{\sqrt{\Delta_{0}} \sqrt{\zeta_{1}^{2} + \zeta_{2}^{2} + 1}} & \frac{(\zeta_{1}^{2} + \zeta_{2}^{2} + 1) \sin \xi - \zeta_{3} \cos \xi}{\sqrt{\Delta_{0}} \sqrt{\zeta_{1}^{2} + \zeta_{2}^{2} + 1}} & \frac{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2}} \zeta_{3}}{\sqrt{\Delta_{0}} \sqrt{\zeta_{1}^{2} + \zeta_{2}^{2} + 1}} \\ 0 & -\frac{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2}} \sin \xi}{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2} + 1}} & \frac{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2} + 1}}{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2} + 1}} & \frac{1}{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2} + 1}} \\ -\frac{\sqrt{\Delta_{0}} \sin \theta}{\sqrt{\Xi_{0}}} & \frac{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2}} \sin \zeta \cos \theta (\zeta_{3} \cos \xi - \sin \xi)}{\sqrt{\Delta_{0}} \sqrt{\Xi_{0}}} & \frac{\sqrt{\zeta_{1}^{2} + \zeta_{2}^{2}} \sin \zeta \cos \theta (\zeta_{3} \sin \xi + \cos \xi)}{\sqrt{\Delta_{0}} \sqrt{\Xi_{0}}} & -\frac{(\zeta_{1}^{2} + \zeta_{2}^{2}) \sin \zeta \cos \theta}{\sqrt{\Delta_{0}} \sqrt{\Xi_{0}}} \end{pmatrix}$$
(A.14)

which acts on 2456. We have introduced the following expressions

$$\Delta_0 = 1 + \zeta_1^2 + \zeta_2^2 + \zeta_3^2, \qquad \Xi_0 = \sin^2 \theta \Delta_0 + \sin^2 \zeta \cos^2 \theta \left(\zeta_1^2 + \zeta_2^2\right), \qquad \zeta_a = v_a e^{\sum_{b \neq a} C_a}.$$
(A.15)

With the rotated vielbein basis, we may give the RR forms in [11] in terms of them and then use [13, 55] to read off the T-dual solution, getting then the results in section 3.

B Type-II G-structure conditions for AdS_4 solutions

In this appendix we review the G-structure conditions for supersymmetric $AdS_4 \times M_6$ solutions, which is a slight modification¹⁵ of what may be found in [56, 57], but with notation more akin to [59, 63]. The metric can be cast in the form

$$ds^{2} = e^{2A} ds^{2} (A dS_{4}) + ds^{2} (M_{6}), \tag{B.1}$$

where the AdS radius is 1 and the dilaton has support only in M_6 . The fluxes have the same direct product structure, which in terms of the RR polyform we may express as

$$F = F_{\text{int}} + e^{4A} \text{Vol}(AdS_4) \wedge \tilde{F}.$$
(B.2)

We use a real representation of the 10d gamma matrices¹⁶ in which the Dirac and ordinary conjugates coincide. A 4+6 split is performed on the 10d MW Killing spinors where $\epsilon = (\epsilon_1, \epsilon_2)^T$ and $\Gamma^{(10)} \epsilon = \sigma_3 \epsilon$ so that we can write

$$\epsilon_1 = e^{A/2} \left(\zeta_+ \otimes \eta_+^1 + \zeta_- \otimes \eta_-^1 \right),$$

$$\epsilon_2 = e^{A/2} \left(\zeta_+ \otimes \eta_\pm^2 + \zeta_- \otimes \eta_+^2 \right),$$
(B.3)

where \pm labels chirality in 4 and 6 dimensions, so that the upper/lower signs should be taken in IIA/IIB and $(\eta_{+})^{*} = \eta_{-}$ and we take the internal, $\eta^{1,2}$ spinor to have unit norm.

Preservation of supersymmetry may be expressed in terms of differential conditions on two pure spinors

$$\Psi_{\pm} = 8 \eta_{\pm}^1 \otimes \eta_{\pm}^{2\dagger}. \tag{B.4}$$

¹⁵Specifically with the normalization of the internal spinor.

¹⁶Specifically for the AdS_4 directions $\Gamma^{\mu} = \hat{\gamma}^{\mu} \otimes 1$, while on \mathbb{CP}^3 we define $\Gamma^i = \gamma^{(4)} \otimes \gamma^i$, where $\hat{\gamma}^{\mu}$ and γ^a are representations of the gamma matrices in 3+1 and 6 dimensions respectively. We define $\Gamma^{(10)} = \gamma^{(4)} \otimes \gamma^{(7)}$, where $\gamma^{(4)} = i\hat{\gamma}^{tx^1x^2r}$ and $\gamma^{(7)} = -i\gamma_{123456}$.

77

These conditions are given by

$$(d-H) \wedge \left(e^{3A-\Phi}\Psi_{\pm}\right) = -2e^{2A-\Phi} \operatorname{Re}\Psi_{\pm}, \tag{B.5}$$
$$(d-H) \wedge \left(e^{4A-\Phi}\Psi_{\pm}\right) = -3e^{3A-\Phi} \operatorname{Im}\Psi_{\pm} + e^{4A}\tilde{F},$$

where once more the upper/lower signs should be taken in IIA/IIB and

$$e^{4A}\tilde{F} = \iota_{\text{Vol}(AdS_4)}F.$$
(B.6)

The G-structure on M_6 can either be an SU(3), when η^1_+ and η^2_+ are globally parallel, or SU(2) when they are not. Using a Fierz identity and the Clifford map it is possible to express Φ_{\pm} as polyforms. In the SU(3)-structure case we can write this in terms of a complex 2-form J and a holomorphic 3-form Ω_{hol} as

$$\Psi_{+} = e^{-i\theta_{+}}e^{-iJ}, \qquad \Psi_{-} = e^{i\theta_{-}}\Omega_{\text{hol}}, \tag{B.7}$$

where the forms may be expressed in terms of the internal spinors as

$$J_{ab} = -i\eta_+^{\dagger}\gamma_{ab}\eta_+, \qquad \left(\Omega_{\rm hol}\right)_{abc} = -\eta_-^{\dagger}\gamma_{abc}\eta_+, \tag{B.8}$$

where

$$\eta_{+}^{1} = e^{i\alpha_{1}}\eta, \qquad \eta_{+}^{2} = e^{i\alpha_{2}}\eta, \qquad \eta_{+}^{\dagger}\eta_{+} = 1, \qquad \theta_{\pm} = \alpha_{1} \mp \alpha_{2},$$
(B.9)

and the forms obey $J \wedge J \wedge J = \frac{3i}{4}\Omega_{\text{hol}} \wedge \overline{\Omega}_{\text{hol}}, J \wedge \Omega_{\text{hol}} = 0.$

For the $\mathrm{SU}(2)$ -structure case the internal spinor may be expressed as

$$\eta_{+}^{1} = e^{i\alpha_{1}}\eta_{+}, \qquad \eta_{+}^{2} = e^{i\alpha_{2}}(\kappa_{\parallel}\eta_{+} + \kappa_{\perp}\chi_{+})$$
 (B.10)

where $\chi^{\dagger}_{+}\eta_{+} = 0$ and $\kappa^{2}_{\parallel} + \kappa^{2}_{\perp} = 1$. The pure spinors may then be expressed in terms of a holomorphic 1-form z, a real 2-form j and a holomorphic 2-form ω_{hol} as

$$\Psi_{+} = i e^{i\theta_{+}} e^{\frac{1}{2}z \wedge \bar{z}} \wedge \left(\kappa_{\parallel} e^{-ij} - i\kappa_{\perp}\omega_{\rm hol}\right),$$

$$\Psi_{-} = e^{i\theta_{-}} z \wedge \left(\kappa_{\perp} e^{-ij} + i\kappa_{\parallel}\omega_{\rm hol}\right).$$
 (B.11)

The various forms may be extracted from the spinor via

$$z_a = -i\eta_-^{\dagger}\gamma_a\chi_+, \qquad j_{ab} = \frac{1}{2} \left(-i\eta_+^{\dagger}\gamma_{ab}\eta_+ + i\chi_+^{\dagger}\gamma_{ab}\chi_+ \right), \qquad (\omega_{\rm hol})_{ab} = i\eta_-^{\dagger}\gamma_{ab}\chi_-, \tag{B.12}$$

and obey the conditions,

$$j \wedge j = \frac{1}{2}\omega_{\text{hol}} \wedge \bar{\omega}_{\text{hol}}, \qquad j \wedge \omega_{\text{hol}}, \qquad \omega_{\text{hol}} \wedge \omega_{\text{hol}} = 0, \qquad \iota_z \omega_{\text{hol}} = \iota_z j = 0.$$
 (B.13)

Finally it should be noted that the above conditions are actually the conditions for $\mathcal{N} = 1$ in 3d. We will be concerned with $\mathcal{N} = 2$ supersymmetry which implies a CFT dual with U(1) R-symmetry. This will manifest itself in the fact that there should be a U(1)'s worth of pure spinors satisfying eq. (B.5), two of which are independent.¹⁷

¹⁷In the sense that they can be constructed from two sets of linearly independent internal spinors (η_1, η_2) and $(\tilde{\eta}_1, \tilde{\eta}_2)$.

C Detailed supersymmetry analysis

In this appendix we shall look at how the Killing spinors are transformed under the sequence of dualities we perform to reach the M-theory solution of section 4.2. We shall begin by identifying a set of spinors on \mathbb{CP}^3 that are uncharged under the SU(2) on which the NAT-duality is performed.

C.1 A SU(2) T-duality invariant Killing spinor on $AdS_4 \times \mathbb{CP}^3$

We express the metric of ABJM in terms of the vielbein basis of eq. (A.3). Supersymmetry is preserved in type IIA when the variations of the dilatino and gravitino vanish. For ABJM which has a constant dilaton and zero Romans mass these constraints are

where $D_{\mu}\epsilon = (\partial_{\mu} + \frac{1}{4}\omega_{\mu,ab}\Gamma^{ab})\epsilon$. Specifically we have

and

$$\begin{split} \omega^{x^{\mu}\rho} &= \frac{2}{L}e^{x^{\mu}}, \qquad \omega^{12} = -\omega^{36} = \frac{\tan\zeta}{L}e^{2}, \qquad \omega^{13} = \omega^{26} = \frac{\tan\zeta}{L}e^{3} \\ \omega^{14} &= -\omega^{56} = -\frac{\cot\zeta}{L}e^{4}, \qquad \omega^{15} = \omega^{46} = -\frac{\cot\zeta}{L}e^{5}, \qquad \omega^{16} = -\frac{2\cot2\zeta}{L}e^{6}, \end{split}$$
(C.3)

$$\omega^{23} = \frac{1}{L} (-2\cot\theta_1 \sec\zeta e^3 + \tan\zeta e^6), \qquad \omega^{45} = \frac{1}{L} (-2\cot\theta_1 \sec\zeta e^3 + (\cot\zeta + 2\tan\zeta)e^6).$$

Inserting the fluxes into the variation of the dilatino and manipulating leads to

$$\left(\Gamma^{2345} + \Gamma^{16}(\Gamma^{23} + \Gamma^{45})\right)\epsilon = \epsilon.$$
(C.4)

This constraint preserves a maximum of 24 real supercharges, however one finds that such a Killing spinor, which also solves the gravitino variation, must depend on the SU(2) directions [11]. Here we take a different approach and impose the projection

$$\Gamma^{2345}\epsilon = \epsilon, \tag{C.5}$$

which preserves only half the supercharges. Turning attention to the gravitino variation, one finds that the components along the AdS_4 directions give

$$D_{\mu}\epsilon + \frac{1}{L}\Gamma_{AdS_4}\Gamma_{\mu}(\sigma_1)\epsilon, \qquad (C.6)$$

which is a standard Killing spinor equation for AdS_4 which one can solve without any constraint.

Using the projection or eq. (C.5) it is possible to show that the gravitino variation along the \mathbb{CP}^3 directions reduce to a differential equation and an additional projection

$$\partial_{\zeta} \epsilon + \Gamma^6(i\sigma_2)\epsilon = 0, \tag{C.7}$$

$$\Gamma^{1456}\epsilon = -\bigg(\cos 2\zeta + \sin 2\zeta\Gamma^6(i\sigma_2)\bigg)\epsilon.$$
(C.8)

These are solved by

$$\epsilon = e^{-\Gamma^6(i\sigma_2)}\epsilon_0 \tag{C.9}$$

where ϵ_0 is a spinor which depends only on the AdS_4 coordinates and obeys

$$\Gamma^{2345}\epsilon_0 = -\Gamma^{1456}\epsilon_0 = \epsilon_0.$$
 (C.10)

Thus we have found a Killing spinor preserving 8 real supercharges which gives $\mathcal{N} = 2$ supersymmetry in 3d. This is the most general spinor which is independent of the SU(2) directions (in the prefered frame) and so [13] informs us that 8 supercharges are preserved under a SU(2) NAT duality transformation.

As the solution is a direct product and we know that there are 4 independent Killing spinors preserved by AdS_4 , we must have 2 preserved on \mathbb{CP}^3 . On the other hand the AdS_4 factor and supersymmetry preserved by the spinor imply that we are describing a subsector of ABJM with U(1) R-symmetry. The Killing spinors should be invariant under the action of this U(1). Indeed we can impose an additional projection

$$P_{\alpha}\epsilon = \Gamma^6(i\sigma_2)\epsilon, \qquad P_{\alpha} = \Gamma^3(-\cos\alpha\Gamma^4 + \sin\alpha\Gamma^5)$$
 (C.11)

where α is a constant which parametrizes the U(1). Notice that if one defines two spinors such that $P_{\alpha_1}\chi_{\alpha_1} = \Gamma^6(i\sigma_2)\chi_{\alpha_1}$ and $P_{\alpha_2}\chi_{\alpha_2} = \Gamma^6(i\sigma_2)\chi_{\alpha_2}$ hold, then we have $\chi^{\dagger}_{\alpha_1}\chi_{\alpha_2} = 0$ when $\alpha_1 - \alpha_2 = \pi$, so we are still describing $\mathcal{N} = 2$ supersymmetry.

C.2 A U(1) of SU(3)-structures on \mathbb{CP}^3

We know the 6d Killing spinors of ABJM define an SU(3)-structure [58], so the internal spinors η^1_+ and η^2_+ must match up to a phase. Specifically we define

$$\eta_{+}^{1} = e^{i\frac{\theta_{0}}{2}}\eta_{+}, \qquad \eta_{+}^{2} = e^{-i\frac{\theta_{0}}{2}}\eta_{+}.$$
(C.12)

The projective constraints in 6d become

$$\gamma^{1456}\eta_{+} = -\left(\cos 2\zeta + \hat{P}_{\alpha}\sin 2\zeta\right)\eta_{+}, \qquad \gamma^{2345}\eta_{+} = \eta_{+}, \qquad \hat{P}_{\alpha}\eta_{+} = \gamma^{6}\eta_{-}, \qquad (C.13)$$

where $\hat{P}_{\alpha} = \gamma^3 (-\cos \alpha \gamma^4 + \sin \alpha \gamma^5)$. These are still a little complicated, to get to a canonical frame we first rotate in γ^4 , γ^5 , and then γ^3 , γ^4 such that $\hat{P}_{\alpha} = -\tilde{\gamma}^{34}$ and $\gamma^{1456}\eta_+ = -\eta_+$. This leads to new vielbeins which we express in terms of eq. (A.3) as

$$\begin{split} \tilde{e}^a &= e^a, \qquad a = 1, 2, 6, \\ \tilde{e}^3 &= \cos 2\zeta e^3 + \sin 2\zeta (-\cos \alpha e^4 + \sin \alpha e^5), \\ \tilde{e}^4 &= \sin 2\zeta e^3 + \cos 2\zeta (\cos \alpha e^4 - \sin \alpha e^5), \\ \tilde{e}^5 &= \sin \alpha e^4 + \cos \alpha e^5. \end{split}$$
(C.14)

With respect to this basis we have

$$\tilde{\gamma}^{16}\eta_{+} = \tilde{\gamma}^{32}\eta_{+} = \tilde{\gamma}^{45}\eta_{+} = +i\eta_{+}, \qquad \tilde{\gamma}^{346}\eta_{+} = -\eta_{-}, \tag{C.15}$$

and so the SU(3)-structures are given by the forms

$$J_{\alpha} = \tilde{e}^1 \wedge \tilde{e}^6 + \tilde{e}^3 \wedge \tilde{e}^2 + \tilde{e}^4 \wedge \tilde{e}^5,$$

$$\Omega_{hol,\alpha} = -i(\tilde{e}^1 + i\tilde{e}^6) \wedge (\tilde{e}^3 + i\tilde{e}^2) \wedge (\tilde{e}^4 + i\tilde{e}^5).$$
(C.16)

The forms satisfy eq. (B.5) for any constant α provided

$$\theta_{+} = \theta_{0} = \frac{3\pi}{2}, \qquad \theta_{-} = 0, \qquad e^{2A} = \frac{L^{2}}{4}.$$
(C.17)

One should note that if we take $(J_0, \Omega_{hol,0})$ we can generate the whole U(1) again by sending $\psi \to \psi - \alpha$, inside the left invariant 1-forms ω_i . This is what we expect since the isometry ∂_{ψ} gives the geometric realisation of the U(1) subgroup of the R-symmetry of ABJM.

C.3 A U(1) of SU(2)-structures in the non-Abelian T-dual

We would now like to find the G-structure and Killing spinors of the geometry after performing the SU(2) isometry non-Abelian T-duality. Fortunately we can exploit a map for the SU(2) transformation of the pure spinors that was proposed in [6]

$$\hat{\Psi}_{\pm} = \Psi_{\mp} \, \Omega_{\mathrm{SU}(2)}^{-1},$$
 (C.18)

where in general, in the frame of eq. (A.3)

$$\Omega_{\rm SU(2)} = \frac{1}{\sqrt{1+\zeta_a^2}} \Gamma^{(10)} \bigg(-\Gamma_{456} + \sum_{a=1}^3 \zeta_a \Gamma^{a+3} \bigg), \tag{C.19}$$

for ζ_a defined as in [5], which for our parametrisation of ABJM specifically is

$$\zeta_1 = \frac{4}{L^2 \cos \zeta \sin^2 \zeta} y_1 \cos \xi, \qquad \zeta_2 = \frac{4}{L^2 \cos \zeta \sin^2 \zeta} y_1 \sin \xi, \qquad \zeta_3 = \frac{4}{L^2 \sin^2 \zeta} y_2. \quad (C.20)$$

Although eq. (C.18) will give us the pure spinors in type IIB, it is still instructive to study the MW Killing spinors. The action of the NAT duality transformation on this is given by [13]

$$\hat{\epsilon}_1 = \epsilon_1, \qquad \hat{\epsilon}_2 = \Omega_{\mathrm{SU}(2)} \,\epsilon_2, \tag{C.21}$$

which corresponds to the following 6d spinors

$$\hat{\eta}_{+}^{1} = e^{i\frac{3\pi}{4}}\eta_{+},$$
(C.22)
$$\hat{\eta}_{+}^{2} = -i e^{-i\frac{3\pi}{4}} \bigg[i \frac{\cos 2\zeta \tilde{\gamma}^{1} + \tilde{\zeta}_{1} \tilde{\gamma}^{4} + \tilde{\zeta}_{2} \tilde{\gamma}^{5} + \zeta_{3} \tilde{\gamma}^{6}}{\sqrt{1 + \zeta_{a} \zeta_{a}}} \eta_{-} + \frac{\sin 2\zeta}{\sqrt{1 + \zeta_{a} \zeta_{a}}} \eta_{+} \bigg],$$

with the spinors on AdS_4 unchanged. Here we use the frame of eq. (C.14), but with dual vielbeins, have made use of the projections and defined

$$\tilde{\zeta}_1 = \frac{4}{L^2 \cos \zeta \sin^2 \zeta} y_1 \cos(\xi + \alpha), \qquad \tilde{\zeta}_2 = \frac{4}{L^2 \cos \zeta \sin^2 \zeta} y_1 \sin(\xi + \alpha). \tag{C.23}$$

Here we see that α only appears in the combination $\xi + \alpha$, which indicates that ∂_{ξ} plays the role of the U(1) R-symmetry in the NAT dual solution, indeed this can be confirmed by computing the Kosmann derivative along ∂_{ξ} .

The spinors in eq. (C.22) actually define a dynamical SU(2) structure, which means $\hat{\eta}^1_+$ and $\hat{\eta}^2_+$ are not globally parallel and the angle between them is point dependent. We can simplify the expression for $\hat{\eta}^2_+$ considerably with further rotations of the vielbein basis. There is an optimum frame, in which all components of the SU(2)-structure are relatively simple. The vielbeins are given in this frame by

$$\begin{split} \hat{e}^{1} &= \frac{1}{4L^{3} \sin^{3} \zeta \cos \zeta \sqrt{\Delta_{q}}} \left(L^{4} \sin^{2} \zeta \sin 4\zeta - 32(y_{1}dy_{1} + y_{2}dy_{2}) \right), \end{split} \tag{C.24} \\ \hat{e}^{2} &= \frac{1}{4L^{3} \sin^{3} \zeta \cos \zeta \sqrt{\Delta_{p}} \sqrt{\Delta_{q}}} \left[\cos \zeta \left(\sin 2\zeta \left(L^{4} \sin^{2} \zeta \Delta_{p} d\theta - 32y_{1}^{2} \sin 2(\xi + \alpha) \sin \theta d\phi \right) \right. \\ &- 128y_{1}y_{2} \cos(\xi + \alpha) d\zeta \right) - 64y_{1} \cos(\xi + \alpha) \sin \zeta \cos 2\zeta dy_{2} \right], \end{aligned} \\ \hat{e}^{3} &= \frac{1}{2L^{3} \sin^{2} \zeta \cos \zeta \sqrt{\Delta_{0}} \sqrt{\Delta_{p}}} \left[\cos^{2} \zeta \left(32y_{1}y_{2} \cos(\xi + \alpha) d\xi + 32y_{2} \sin(\xi + \alpha) dy_{1} \right. \\ &+ \left(32y_{1}y_{2} \cos(\xi + \alpha) \cos \theta + L^{4} \Delta_{0} \cos 2\zeta \sin^{2} \zeta \sin \theta \right) d\phi \right) - 32y_{1} \sin(\xi + \alpha) dy_{2} \right], \end{aligned} \\ \hat{e}^{4} &= \frac{2}{L^{5} \sin^{3} \zeta \cos \zeta \sqrt{\Delta_{0}} \sqrt{\Delta_{p}} \sqrt{\Delta_{q}}} \left[\cos \zeta \left(y_{1} \sin(\xi + \alpha) \left(64y_{1}^{2} \cos^{2}(\xi + \alpha) + L^{4} \sin^{2} \zeta \Delta_{p} \right) d\xi \right. \\ &+ \cos(\xi + \alpha) \left(64y_{1}^{2} \sin^{2}(\xi + \alpha) dy_{1} + 64y_{1}y_{2} dy_{2} - L^{4} \Delta_{p} \sin^{2} \zeta dy_{1} \right) \\ &+ L^{4}y_{1} \Delta_{q} \sin(\xi + \alpha) \sin^{2} \zeta \cos \theta d\phi \right) - 2L^{4}y_{1} \Delta_{0} \cos(\xi + \alpha) \cos 2\zeta \sin \zeta d\zeta \right], \end{aligned} \\ \hat{e}^{5} &= -\frac{2}{L \sin^{2} \zeta \cos \zeta \sqrt{\Delta_{p}}} \left[2y_{1} \sin(\xi + \alpha) d\zeta - y_{2} \sin 2\zeta \cos \zeta^{2} \sin \theta d\phi \\ &+ \frac{1}{4} \sin 4\zeta \left(\sin(\xi + \alpha) dy_{1} + y_{1} \cos(\xi + \alpha) (d\xi + \cos \theta d\phi) \right) \right] \right] \\ \hat{e}^{6} &= -\frac{2}{L \sin^{2} \zeta \cos \zeta \sqrt{\Delta_{q}}} \left[\cos 2\zeta \sin \zeta dy_{2} \\ &+ 2 \cos \zeta \left(y_{2} d\zeta + y_{1} \cos \zeta \sin \zeta (\cos(\xi + \alpha) d\theta + \sin(\xi + \alpha) \sin \theta d\phi) \right) \right], \end{split}$$

where

$$\Delta_0 = 1 + \zeta_a^2, \qquad \Delta_q = \cos^2 2\zeta + \zeta_a^2, \qquad \Delta_p = \Delta_q - \sin^2 2\tilde{\zeta}\zeta_1^2. \tag{C.25}$$

In this basis the action of NAT duality on the 6d spinors is simply

$$\hat{\eta}^{1} = e^{i\frac{\theta_{0}}{2}}\eta_{+}, \qquad \hat{\eta}^{2}_{+} = -e^{-i\frac{\theta_{0}}{2}} \left[\kappa_{\parallel}\eta_{+} + i\kappa_{\perp}\hat{\gamma}^{1}\eta_{-}\right]$$
(C.26)

where

82

$$\kappa_{\parallel} = \frac{\sin 2\zeta}{\sqrt{1 + \zeta_a \zeta_a}}, \qquad \kappa_{\perp} = \sqrt{\frac{\cos 2\zeta + \zeta_a \zeta_a}{1 + \zeta_a \zeta_a}} \tag{C.27}$$

and $\kappa_{\parallel}^2 + \kappa_{\perp}^2 = 1$. The projections the original spinor obeys are most succinctly expressed as

$$\hat{\gamma}_{2345}\eta_+ = \eta_+, \qquad \hat{\gamma}_{1456}\eta_+ = -(\kappa_\perp - \kappa_\parallel \hat{\gamma}^{34})\eta_+,$$
 (C.28)

in the basis where $\gamma^{(7)}\eta_+ = \eta_+$ as before. The U(1)'s worth of SU(2)-structure is given by the following forms

$$z_{\alpha} = \hat{e}^{1} + i\hat{e}^{6},$$

$$j_{\alpha} = (\kappa_{\perp}\hat{e}^{3} - \kappa_{\parallel}\hat{e}^{4}) \wedge \hat{e}^{2} + (\kappa_{\perp}\hat{e}^{4} + \kappa_{\parallel}\hat{e}^{3}) \wedge \hat{e}^{5},$$

$$\omega_{hol,\alpha} = -i((\kappa_{\perp}\hat{e}^{3} - \kappa_{\parallel}\hat{e}^{4}) + i\hat{e}^{2}) \wedge ((\kappa_{\perp}\hat{e}^{4} + \kappa_{\parallel}\hat{e}^{3}) + i\hat{e}^{5}),$$
(C.29)

which satisfy the supersymmetry conditions of eq. (B.5) for any constant α provided

$$\theta_{+} = 0, \qquad \theta_{-} = \theta_{0} = \frac{3\pi}{2}, \qquad e^{2A} = \frac{L^{2}}{4}.$$
(C.30)

We could take the intersection of the two linearly independent SU(2) structures defined for $\alpha = 0$ and $\alpha = \pi$, and define an identity structure. However, the supersymmetry conditions of such an object are absent from the literature at present and deriving them is outside the scope of this work.

C.4 Killing spinors in M-theory

ŀ

Before we can define the M-theory Killing spinor, we must first derive the MW Killing spinors in type IIA after an additional T-duality is performed. As we want to make contact with [45] we need to work with the two linearly independent spinors in 6d. These are given by

$$\pi_{+}^{1} = e^{i\frac{\theta_{0}}{2}}\pi_{+}, \qquad \pi_{+}^{2} = e^{-i\frac{\theta_{0}}{2}}\pi_{+}, \qquad \tilde{\pi}_{+}^{1} = e^{i\frac{\theta_{0}}{2}}\tilde{\pi}_{+}, \qquad \tilde{\pi}_{+}^{2} = e^{-i\frac{\theta_{0}}{2}}\tilde{\pi}_{+}, \qquad \theta_{0} = \frac{3\pi}{2}$$
(C.31)

and are such that

$$\gamma^{1456}\pi_{+} = -\left(\cos 2\zeta - \sin 2\zeta\gamma^{34}\right)\pi_{+}, \qquad \gamma^{2345}\pi_{+} = \pi_{+}, \qquad \gamma^{246}\pi_{+} = -\pi_{-}$$
$$\gamma^{1456}\tilde{\pi}_{+} = -\left(\cos 2\zeta + \sin 2\zeta\gamma^{34}\right)\tilde{\pi}_{+}, \qquad \gamma^{2345}\tilde{\pi}_{+} = \tilde{\pi}_{+}, \qquad \gamma^{246}\tilde{\pi}_{+} = \tilde{\pi}_{-}, \qquad (C.32)$$

in the frame of eq. (A.3). The independent 10d MW spinors $\epsilon_{1,2}$ and $\tilde{\epsilon}_{1,2}$ are then constructed in the obvious way from eq. (B.3), with $\eta \to \pi$ and using the same spinors on AdS_4 . We must act on these spinors first with $\Omega_{SU(2)}$, which in this frame is as in eq. (C.19), then with $\Omega_{U(1)}$, which gives the transformation of the spinor under the Abelian T-duality [55]. In the frame of eqs. (A.3), (A.8) this is most succinctly expressed as

$$\begin{split} \Omega_{\mathrm{U}(1)} &= \frac{1}{\sqrt{\Delta_0}\sqrt{\Xi_0}} \sin \zeta \cos \theta \Gamma^{(10)} \bigg[\big(-\zeta_2 + \zeta_1 \zeta_3 \big) \Gamma^4 + \big(\zeta_1 + \zeta_2 \zeta_3 \big) \Gamma^5 - \big(\zeta_1^2 + \zeta_2^2 \big) \Gamma^6 \bigg] \\ &- \frac{\sqrt{\Delta_0}}{\sqrt{\Xi_0}} \sin \theta \Gamma^{(10)} \Gamma^3. \end{split}$$

We take the 10d MW Killing spinors in IIA after the NAT-T duality transformation to be

$$\hat{\hat{\epsilon}}_1 = \epsilon_1, \qquad \hat{\hat{\epsilon}}_2 = \Omega_{\mathrm{U}(1)} \Omega_{\mathrm{SU}(2)} \epsilon_2,$$
 (C.33)

with an equivalent expression with $\epsilon \to \tilde{\epsilon},$ which means that the new 6d Killing spinors are given by

$$\hat{\pi}^{1}_{+} = e^{i\frac{\theta_{0}}{2}}\pi_{+}, \qquad \hat{\pi}^{2}_{+} = -e^{-i\frac{\theta_{0}}{2}} \Big(\hat{\kappa}_{\parallel}\pi_{+} + \hat{\kappa}_{\perp}\mathcal{F}\pi_{-} \Big), \\ \hat{\pi}^{1}_{+} = e^{i\frac{\theta_{0}}{2}}\tilde{\pi}_{+}, \qquad \hat{\pi}^{2}_{+} = +e^{-i\frac{\theta_{0}}{2}} \Big(\hat{\kappa}_{\parallel}\tilde{\pi}_{+} + \hat{\kappa}_{\perp}\mathcal{F}\tilde{\pi}_{-} \Big),$$
(C.34)

where

$$\hat{\kappa}_{\parallel} = \frac{\sin 2\zeta \sin \theta \zeta_2}{\sqrt{\Xi_0}}, \qquad \hat{\kappa}_{\perp} = \sqrt{1 - \hat{\kappa}_{\parallel}^2},$$

$$\mathcal{F} = \frac{i}{\sqrt{\Xi_0 - \sin^2 2\zeta \sin^2 \theta \zeta_2^2}} \bigg(\sin \zeta \cos \theta \big(\zeta_2 \gamma^2 - \zeta_1 \gamma^3 \big) - \sin \theta \big(\cos 2\zeta \zeta_2 \gamma^1 - \gamma^5 - \zeta_3 \gamma^4 + \zeta_1 \gamma^6 \big) \bigg). \tag{C.35}$$

Clearly eq. (C.34) supports a U(1) of dynamical SU(2)-structures, as was the case in type-IIB, which we will not explicitly derive.

We are know ready to construct the two independent M-theory Killing spinors. These can be expressed in terms of the spinors in IIA as

$$\eta^1 = e^{-\Phi/6} (\epsilon_1 + \epsilon_2), \qquad \eta^2 = e^{-\Phi/6} (\tilde{\epsilon}_1 + \tilde{\epsilon}_2).$$
 (C.36)

In the conventions of [45] the M-theory spinors are

$$\eta^{i} = e^{\tilde{\Delta}/2} \left(\psi^{i}_{+} \otimes \chi_{i} + \left(\psi^{i}_{+}\right)^{c} \otimes \chi^{c}_{i} \right), \tag{C.37}$$

where $e^{\frac{\tilde{\Delta}}{2}} = e^{\frac{\tilde{A}}{2} - \frac{\Phi}{6}}$ and $e^{2\tilde{A}}$ is a modified warp factor of AdS_4 in IIA such that $Ricci(AdS_4) = -12g(AdS_4)$. Thus if we identify the AdS_4 spinors of IIA with those of eq. (C.37) we see that

$$\chi_1 = \frac{1}{\sqrt{2}} (\hat{\pi}^1_+ + \hat{\pi}^2_-), \qquad \chi_1^c = \frac{1}{\sqrt{2}} (\hat{\pi}^1_- + \hat{\pi}^2_+),$$

$$\chi_2 = \frac{1}{\sqrt{2}} (\hat{\pi}^1_+ + \hat{\pi}^2_-), \qquad \chi_2^c = \frac{1}{\sqrt{2}} (\hat{\pi}^1_- + \hat{\pi}^2_+), \qquad (C.38)$$

which clearly satisfy $\bar{\chi}_1 \chi_1 = \bar{\chi}_2 \chi_2 = 1$, and from these one can construct spinors of charge ± 2 under the U(1) R-symmetry

$$\chi_{\pm} = \frac{1}{\sqrt{2}} (\chi_1 \pm \chi_2). \tag{C.39}$$

It is then simply a matter of plugging the χ_{\pm} of this section into the spinor bi-linears in appendix B of [45], and rotating the frame to reproduce the results of section 4.2. Note that the frame used in this section needs to be rotated as in eq. (A.14) to reach the vielbein basis where flat directions 2456 may be identified with $\mathcal{G}^{1,2,3,4}$ of eq. (3.2) and the rest with eq. (A.3).

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- K. Sfetsos and D.C. Thompson, On non-abelian T-dual geometries with Ramond fluxes, Nucl. Phys. B 846 (2011) 21 [arXiv:1012.1320] [INSPIRE].
- [2] Y. Lozano, E.Ó. Colgáin, K. Sfetsos and D.C. Thompson, Non-abelian T-duality, Ramond Fields and Coset Geometries, JHEP 06 (2011) 106 [arXiv:1104.5196] [INSPIRE].
- [3] Y. Lozano, E.Ó. Colgáin, D. Rodríguez-Gómez and K. Sfetsos, Supersymmetric AdS₆ via T Duality, Phys. Rev. Lett. 110 (2013) 231601 [arXiv:1212.1043] [INSPIRE].
- [4] G. Itsios, C. Núñez, K. Sfetsos and D.C. Thompson, On Non-Abelian T-duality and new N = 1 backgrounds, Phys. Lett. B 721 (2013) 342 [arXiv:1212.4840] [INSPIRE].
- G. Itsios, C. Núñez, K. Sfetsos and D.C. Thompson, Non-Abelian T-duality and the AdS/CFT correspondence: new N = 1 backgrounds, Nucl. Phys. B 873 (2013) 1 [arXiv:1301.6755] [INSPIRE].
- [6] A. Barranco, J. Gaillard, N.T. Macpherson, C. Núñez and D.C. Thompson, G-structures and Flavouring non-Abelian T-duality, JHEP 08 (2013) 018 [arXiv:1305.7229] [INSPIRE].
- [7] N.T. Macpherson, Non-Abelian T-duality, G₂-structure rotation and holographic duals of N = 1 Chern-Simons theories, JHEP 11 (2013) 137 [arXiv:1310.1609] [INSPIRE].
- [8] Y. Lozano, E.O. Colgáin, D. Rodríguez-Gómez, Hints of 5d Fixed Point Theories from Non-Abelian T-duality, JHEP 05 (2014) 009 [arXiv:1311.4842] [INSPIRE].
- J. Gaillard, N.T. Macpherson, C. Núñez and D.C. Thompson, Dualising the Baryonic Branch: Dynamic SU(2) and confining backgrounds in IIA, Nucl. Phys. B 884 (2014) 696 [arXiv:1312.4945] [INSPIRE].
- [10] E. Caceres, N.T. Macpherson and C. Núñez, New Type IIB Backgrounds and Aspects of Their Field Theory Duals, JHEP 08 (2014) 107 [arXiv:1402.3294] [INSPIRE].
- [11] Y. Lozano and N.T. Macpherson, A new AdS₄/CFT₃ dual with extended SUSY and a spectral flow, JHEP 11 (2014) 115 [arXiv:1408.0912] [INSPIRE].
- [12] K. Sfetsos and D.C. Thompson, New $\mathcal{N} = 1$ supersymmetric AdS₅ backgrounds in Type IIA supergravity, JHEP **11** (2014) 006 [arXiv:1408.6545] [INSPIRE].
- [13] Ö. Kelekci, Y. Lozano, N.T. Macpherson and E.Ó. Colgáin, Supersymmetry and non-Abelian T-duality in type-II supergravity, Class. Quant. Grav. 32 (2015) 035014 [arXiv:1409.7406] [INSPIRE].
- [14] N.T. Macpherson, C. Núñez, L.A. Pando Zayas, V.G.J. Rodgers and C.A. Whiting, *Type IIB supergravity solutions with AdS₅ from Abelian and non-Abelian T dualities*, *JHEP* 02 (2015) 040 [arXiv:1410.2650] [INSPIRE].
- [15] K.S. Kooner and S. Zacarías, Non-Abelian T-Dualizing the Resolved Conifold with Regular and Fractional D3-branes, JHEP 08 (2015) 143 [arXiv:1411.7433] [INSPIRE].
- [16] T.R. Araujo and H. Nastase, N = 1 SUSY backgrounds with an AdS factor from non-Abelian T duality, Phys. Rev. D 91 (2015) 126015 [arXiv:1503.00553] [INSPIRE].

- [17] Y. Bea et al., Compactifications of the Klebanov-Witten CFT and new AdS₃ backgrounds, JHEP 05 (2015) 062 [arXiv:1503.07527] [INSPIRE].
- [18] Y. Lozano, N.T. Macpherson, J. Montero and E.Ó. Colgáin, New $AdS_3 \times S^2$ T-duals with $\mathcal{N} = (0, 4)$ supersymmetry, JHEP 08 (2015) 121 [arXiv:1507.02659] [INSPIRE].
- [19] A. Passias, A note on supersymmetric AdS₆ solutions of massive type IIA supergravity, JHEP 01 (2013) 113 [arXiv:1209.3267] [INSPIRE].
- [20] F. Apruzzi, M. Fazzi, A. Passias, D. Rosa and A. Tomasiello, AdS₆ solutions of type-II supergravity, JHEP **11** (2014) 099 [Erratum ibid. **05** (2015) 012] [arXiv:1406.0852] [INSPIRE].
- [21] H. Kim, N. Kim and M. Suh, Supersymmetric AdS₆ Solutions of Type IIB Supergravity, arXiv:1506.05480 [INSPIRE].
- [22] N. Seiberg, Five-dimensional SUSY field theories, nontrivial fixed points and string dynamics, Phys. Lett. B 388 (1996) 753 [hep-th/9608111] [INSPIRE].
- [23] D.R. Morrison and N. Seiberg, Extremal transitions and five-dimensional supersymmetric field theories, Nucl. Phys. B 483 (1997) 229 [hep-th/9609070] [INSPIRE].
- [24] K.A. Intriligator, D.R. Morrison and N. Seiberg, Five-dimensional supersymmetric gauge theories and degenerations of Calabi-Yau spaces, Nucl. Phys. B 497 (1997) 56 [hep-th/9702198] [INSPIRE].
- [25] A. Brandhuber and Y. Oz, The D4-D8 brane system and five-dimensional fixed points, Phys. Lett. B 460 (1999) 307 [hep-th/9905148] [INSPIRE].
- [26] O. Bergman and D. Rodriguez-Gomez, 5d quivers and their AdS₆ duals, JHEP 07 (2012) 171 [arXiv:1206.3503] [INSPIRE].
- [27] O. Aharony and A. Hanany, Branes, superpotentials and superconformal fixed points, Nucl. Phys. B 504 (1997) 239 [hep-th/9704170] [INSPIRE].
- [28] O. Aharony, A. Hanany and B. Kol, Webs of (p,q) five-branes, five-dimensional field theories and grid diagrams, JHEP 01 (1998) 002 [hep-th/9710116] [INSPIRE].
- [29] O. DeWolfe, A. Hanany, A. Iqbal and E. Katz, Five-branes, seven-branes and five-dimensional E_n field theories, JHEP 03 (1999) 006 [hep-th/9902179] [INSPIRE].
- [30] A. Rota and A. Tomasiello, AdS₄ compactifications of AdS₇ solutions in type-II supergravity, JHEP 07 (2015) 076 [arXiv:1502.06622] [INSPIRE].
- [31] D. Gaiotto and A. Tomasiello, Holography for (1,0) theories in six dimensions, JHEP 12 (2014) 003 [arXiv:1404.0711] [INSPIRE].
- [32] Y. Terashima and M. Yamazaki, SL(2, ℝ) Chern-Simons, Liouville and Gauge Theory on Duality Walls, JHEP 08 (2011) 135 [arXiv:1103.5748] [INSPIRE].
- [33] Y. Terashima and M. Yamazaki, Semiclassical Analysis of the 3d/3d Relation, Phys. Rev. D 88 (2013) 026011 [arXiv:1106.3066] [INSPIRE].
- [34] T. Dimofte, D. Gaiotto and S. Gukov, Gauge Theories Labelled by Three-Manifolds, Commun. Math. Phys. 325 (2014) 367 [arXiv:1108.4389] [INSPIRE].
- [35] T. Dimofte, D. Gaiotto and S. Gukov, 3-Manifolds and 3d Indices, Adv. Theor. Math. Phys. 17 (2013) 975 [arXiv:1112.5179] [INSPIRE].
- [36] J. Yagi, 3d TQFT from 6d SCFT, JHEP 08 (2013) 017 [arXiv:1305.0291] [INSPIRE].

- [37] S. Lee and M. Yamazaki, 3d Chern-Simons Theory from M5-branes, JHEP 12 (2013) 035 [arXiv:1305.2429] [INSPIRE].
- [38] C. Cordova and D.L. Jafferis, Complex Chern-Simons from M5-branes on the Squashed Three-Sphere, arXiv:1305.2891 [INSPIRE].
- [39] F. Benini, F. Canoura, S. Cremonesi, C. Núñez and A.V. Ramallo, Backreacting flavors in the Klebanov-Strassler background, JHEP 09 (2007) 109 [arXiv:0706.1238] [INSPIRE].
- [40] B.S. Acharya, J.P. Gauntlett and N. Kim, Five-branes wrapped on associative three cycles, Phys. Rev. D 63 (2001) 106003 [hep-th/0011190] [INSPIRE].
- [41] J.P. Gauntlett, N. Kim and D. Waldram, M Five-branes wrapped on supersymmetric cycles, Phys. Rev. D 63 (2001) 126001 [hep-th/0012195] [INSPIRE].
- [42] M. Pernici and E. Sezgin, Spontaneous Compactification of Seven-dimensional Supergravity Theories, Class. Quant. Grav. 2 (1985) 673 [INSPIRE].
- [43] J.P. Gauntlett, O.A.P. Mac Conamhna, T. Mateos and D. Waldram, AdS spacetimes from wrapped M5 branes, JHEP 11 (2006) 053 [hep-th/0605146] [INSPIRE].
- [44] O. Aharony, O. Bergman, D.L. Jafferis and J.M. Maldacena, N = 6 superconformal Chern-Simons-matter theories, M2-branes and their gravity duals, JHEP 10 (2008) 091 [arXiv:0806.1218] [INSPIRE].
- [45] M. Gabella, D. Martelli, A. Passias and J. Sparks, $\mathcal{N} = 2$ supersymmetric AdS_4 solutions of *M*-theory, Commun. Math. Phys. **325** (2014) 487 [arXiv:1207.3082] [INSPIRE].
- [46] Y. Lozano, Noncommutative branes from M-theory, Phys. Rev. D 64 (2001) 106011 [hep-th/0012137] [INSPIRE].
- [47] E. Bergshoeff, E. Eyras and Y. Lozano, The Massive Kaluza-Klein monopole, Phys. Lett. B 430 (1998) 77 [hep-th/9802199] [INSPIRE].
- [48] E. Bergshoeff, B. Janssen and T. Ortín, Kaluza-Klein monopoles and gauged σ-models, Phys. Lett. B 410 (1997) 131 [hep-th/9706117] [INSPIRE].
- [49] C.M. Hull, U duality and BPS spectrum of super Yang-Mills theory and M-theory, JHEP 07 (1998) 018 [hep-th/9712075] [INSPIRE].
- [50] N.A. Obers and B. Pioline, U duality and M-theory, Phys. Rept. 318 (1999) 113
 [hep-th/9809039] [INSPIRE].
- [51] E. Alvarez, L. Alvarez-Gaumé, J.L.F. Barbón and Y. Lozano, Some global aspects of duality in string theory, Nucl. Phys. B 415 (1994) 71 [hep-th/9309039] [INSPIRE].
- [52] I. Bah, M. Gabella and N. Halmagyi, BPS M5-branes as Defects for the 3d-3d Correspondence, JHEP 11 (2014) 112 [arXiv:1407.0403] [INSPIRE].
- [53] D. Gaiotto and J.M. Maldacena, The Gravity duals of N = 2 superconformal field theories, JHEP 10 (2012) 189 [arXiv:0904.4466] [INSPIRE].
- [54] H.-J. Chung, T. Dimofte, S. Gukov and P. Sulkowski, 3d-3d Correspondence Revisited, arXiv:1405.3663 [INSPIRE].
- [55] S.F. Hassan, T duality, space-time spinors and RR fields in curved backgrounds, Nucl. Phys. B 568 (2000) 145 [hep-th/9907152] [INSPIRE].
- [56] M. Graña, R. Minasian, M. Petrini and A. Tomasiello, Generalized structures of N = 1 vacua, JHEP 11 (2005) 020 [hep-th/0505212] [INSPIRE].

- [57] M. Graña, R. Minasian, M. Petrini and A. Tomasiello, A Scan for new N = 1 vacua on twisted tori, JHEP 05 (2007) 031 [hep-th/0609124] [INSPIRE].
- [58] D. Gaiotto and A. Tomasiello, Perturbing gauge/gravity duals by a Romans mass, J. Phys. A 42 (2009) 465205 [arXiv:0904.3959] [INSPIRE].
- [59] D. Andriot, New supersymmetric flux vacua with intermediate SU(2) structure, JHEP 08 (2008) 096 [arXiv:0804.1769] [INSPIRE].
- [60] I.R. Klebanov and M.J. Strassler, Supergravity and a confining gauge theory: Duality cascades and chi SB resolution of naked singularities, JHEP 08 (2000) 052 [hep-th/0007191] [INSPIRE].
- [61] J.P. Gauntlett, D. Martelli, J. Sparks and D. Waldram, Supersymmetric AdS₅ solutions of type IIB supergravity, Class. Quant. Grav. 23 (2006) 4693 [hep-th/0510125] [INSPIRE].
- [62] F. Apruzzi, M. Fazzi, A. Passias and A. Tomasiello, Supersymmetric AdS₅ solutions of massive IIA supergravity, JHEP 06 (2015) 195 [arXiv:1502.06620] [INSPIRE].
- [63] D. Andriot, String theory flux vacua on twisted tori and Generalized Complex Geometry, Ph.D. Thesis, Université Pierre et Marie Curie, Paris VI France (2010) [INSPIRE].
3.3 Large superconformal near-horizons from M-theory

PHYSICAL REVIEW D 93, 086010 (2016)

Large superconformal near-horizons from M-theory

Ö. Kelekci,^{1,2} Y. Lozano,³ J. Montero,³ E. Ó Colgáin,^{4,5,6} and M. Park⁷
 ¹Department of Mathematics, Istanbul Technical University, Maslak 34469, Istanbul, Turkey
 ²Department of Physics, Middle East Technical University, 06800 Ankara, Turkey
 ³Departamento de Física, Universidad de Oviedo, Oviedo 33007, Spain
 ⁴C. N. Yang Institute for Theoretical Physics, SUNY Stony Brook, New York 11794-3840, USA
 ⁵Department of Mathematics, University of Surrey, Guildford GU2 7XH, United Kingdom
 ⁶Asia Pacific Center for Theoretical Physics, Pohang 37673, Korea, and Department of Physics, Postech, Pohang 37673, Korea
 ⁷School of Physics, Korea Institute for Advanced Study, Seoul 130-722, Korea (Received 24 February 2016; published 27 April 2016)

We report on a classification of supersymmetric solutions to 11D supergravity with $SO(2, 2) \times SO(3)$ isometry, which are AdS/CFT dual to 2D CFTs with $\mathcal{N} = (0, 4)$ supersymmetry. We recover the Maldacena, Strominger, Witten near-horizon with small superconformal symmetry and identify a class of AdS₃ × $S^2 \times S^2 \times CY_2$ geometries with emergent large superconformal symmetry. This exhausts known compact geometries. Compactification of M-theory on CY_2 results in a vacuum of 7D supergravity with large superconformal symmetry, providing a candidate near-horizon for an extremal black hole and a potential new setting to address microstates.

DOI: 10.1103/PhysRevD.93.086010

I. INTRODUCTION

To obey the second law of thermodynamics, black holes must possess entropy, which Bekenstein and Hawking showed is proportional to the area of the event horizon [1]. This observation paved the way for the holographic principle and AdS/CFT [2]. One of the earliest AdS/CFT calculations (it predates the conjecture) shows that asymptotic symmetries of gravity in AdS3 correspond to the Virasoro algebra [3], a feature of 2D CFTs. This observation together with the Cardy formula [4] for the asymptotic growth of states for a CFT with central charge c is enough to provide a microscopic derivation for the Bekenstein-Hawking (BH) entropy [5,6]. For black holes with AdS₃ near-horizons, this methodology has been an incredible success, culminating in recent years in generalizations to extreme Kerr black holes [7], potential astrophysical bodies [8].

However, Einstein's gravity is at best an effective description [9], and the BH entropy is expected to be corrected in a candidate UV complete theory, such as M-theory. More concretely, compactifying the 6D M5-brane theory on a four-cycle in a Calabi-Yau three-manifold, CY_3 , gives rise to the Maldacena, Strominger, Witten (MSW) CFT [10], with $\mathcal{N} = (0, 4)$ supersymmetry at low energies. The corresponding black hole exhibits the near-horizon $AdS_3 \times S^2 \times CY_3$, and subleading corrections to the BH entropy have been shown to perfectly match corrections to the central charge [10,11].

The MSW CFT exhibits *small* superconformal symmetry [12] with an SU(2) R symmetry that is manifest in the twosphere in the dual geometry. Since other superconformal symmetries exist [13–15], a rich class of AdS/CFT geometries can be expected, e.g. [16]. In this paper, we identify a new class of M-theory vacua $AdS_3 \times S^2 \times S^2 \times CY_2$, implying the existence of a distinct class of $2D \mathcal{N} = (0, 4)$ CFTs with *large* superconformal symmetry and R symmetry $SU(2) \times SU(2)$. We recall that CFTs with large superconformal symmetry remain largely enigmatic. While constructions based on string theory, such as $AdS_3 \times S^3 \times S^3 \times S^1$ [17,18], exist, contrary to small superconformal CFTs, interpretation as a symmetric product CFT is problematic [19]. This issue continues to attract exciting new holographic proposals [20,21], against a backdrop where we have witnessed a deeper understanding of the role of integrability [22–24].

More concretely, we report the results of a complete classification of supersymmetric solutions to 11D supergravity, the low-energy description of M-theory, where we assume $SO(2,2) \times SO(3)$ isometry, i.e. warped AdS₃ × $S^2 \times \mathcal{M}_6$ spacetime, with \mathcal{M}_6 being an SU(2)-structure manifold. Since 2D $\mathcal{N} = (0, 4)$ superconformal field theories (SCFTs) are expected to exhibit at least an SU(2)isometry, corresponding to the R symmetry, this is a minimal requirement. One may contemplate a distinct class where the SU(2) R symmetry is realized as a squashed three-sphere, \tilde{S}^3 , but such an ansatz would preclude the MSW geometry. Indeed, noncompact $AdS_3 \times$ $\tilde{S}^3 \times S^2 \times T^3$ geometries, generated via non-Abelian Tduality [25,26], were identified recently in Ref. [27]. It is also an immediate corollary of this work that compact $AdS_3 \times \tilde{S}^3 \times \tilde{S}^3 \times T^2$ geometries with $\mathcal{N} = (0, 4)$ supersymmetry may be generated through T-duality-shift-Tduality (TsT) transformations [28].

2470-0010/2016/93(8)/086010(7)

086010-1

© 2016 American Physical Society

Ö. KELEKCI et al.

Our results pertain to general warped $AdS_3 \times S^2$ spacetimes and are not intended to apply to all M-theory geometries dual to 2D $\mathcal{N} = (0, 4)$ SCFTs. Within our assumptions, we prove that \mathcal{M}_6 is either CY_3 , thus recovering the MSW geometry, or it possesses an additional SU(2) R symmetry that emerges from the supersymmetry analysis. Truncating the emergent SU(2) to U(1), we recover a known class [29,30] of spacetimes with $SU(2) \times U(1)$ isometry [31].

The existence of a class of $AdS_3 \times S^2 \times S^2 \times CY_2$ solutions to 11D supergravity, with 2D $\mathcal{N} = (0, 4)$ SCFT duals, comes somewhat as a surprise. In the case where $CY_2 = T^4$, it was shown long ago that there are geometries related through T-duality to well-known $AdS_3 \times S^3 \times S^3 \times S^1$ solutions in 10D [17]. When $CY_2 = K_3$, the class appears new. It did not feature in a study of wrapped M5-brane geometries [29]. More recently, M-theory geometries dual to 2D $\mathcal{N} = (0, 2)$ SCFTs have been discussed, but where supersymmetry is enhanced to $\mathcal{N} = (0, 4)$, the geometry is either MSW [32,33], or no good AdS₃ vacuum exists [34,35]. Moreover, it is expected that M-theory on K_3 is dual to heterotic string theory on T^3 [36], a statement that can be made precise in the supergravity limit [37]. Despite this, in a recent classification of heterotic supergravity [38], the only compact, regular solutions with eight supersymmetries are shown to be $AdS_3 \times S^3 \times CY_2$ [39].

It can be expected our simply stated results will be of interest to anyone studying the holography of 2D $\mathcal{N} = (0, 4)$ CFTs.

II. $SO(2,2) \times SO(3)$ -INVARIANT SPACETIMES

We recall that bosonic sector of 11D supergravity consists of a metric, g, and a three-form potential, C, with four-form field strength, G = dC. The equations of motion follow from the action

$$S = \frac{1}{2\kappa^2} \int *R - \frac{1}{2}G \wedge *G - \frac{1}{6}C \wedge G \wedge G.$$
 (1)

Supersymmetric solutions satisfy the Killing spinor equation (KSE):

$$\nabla_M \eta + \frac{1}{288} [\Gamma_M^{NPQR} - 8\delta_M^N \Gamma^{PQR}] G_{NPQR} \eta = 0, \quad (2)$$

where $M, N = 0, ..., 10, \nabla_M \eta \equiv \partial_M \eta + \frac{1}{4} \omega_{MNP} \Gamma^{NP} \eta$, with spin connection ω , and η is a Majorana Killing spinor. It is well known that the Einstein equation is implied by the KSE once the Bianchi identity, dG = 0, and equation of motion for *C* hold [40].

2D $\mathcal{N} = (0,4)$ CFTs enjoy both SO(2,2) conformal symmetry and $SU(2) \simeq SO(3)$ R symmetry, which motivates the general ansatz

PHYSICAL REVIEW D 93, 086010 (2016)

$$ds^{2} = e^{2A} \left[\frac{1}{m^{2}} ds^{2} (AdS_{3}) + e^{2B} ds^{2} (S^{2}) + ds^{2} (\mathcal{M}_{6}) \right],$$

$$G = \frac{1}{m^{3}} \operatorname{vol}(AdS_{3}) \wedge \mathcal{A} + \operatorname{vol}(S^{2}) \wedge \mathcal{H} + \mathcal{G}, \qquad (3)$$

where *m* is the inverse AdS₃ radius, *A*, *B* denote scalar warp factors and \mathcal{A} , \mathcal{H} , \mathcal{G} are respectively closed one-, two- and four-forms. The curvatures of symmetric spaces are canonically normalized and fields depend only on the coordinates of the internal 6D Riemannian manifold \mathcal{M}_{6} .

In order to characterize the internal space and the fields, we decompose the 11D gamma matrices [41],

$$\Gamma_{\mu} = \tau_{\mu} \otimes \sigma_{3} \otimes \gamma_{7},$$

$$\Gamma_{\alpha} = 1_{2} \otimes \sigma_{\alpha} \otimes \gamma_{7},$$

$$\Gamma_{m} = 1_{2} \otimes 1_{2} \otimes \gamma_{m},$$
(4)

and 11D Killing spinor,

$$\eta = \psi \otimes e^{A/2} [\chi_+ \otimes \epsilon_+ + \chi_- \otimes \epsilon_-], \tag{5}$$

where $\mu = 0, 1, 2$ label AdS₃ directions, $\alpha = 1, 2$ denote those of $S^2, m = 1, ..., 6$ correspond to \mathcal{M}_6 and we define $\gamma_7 \equiv i\gamma_{123456}$. ψ is a solution to the AdS₃ KSE, $\nabla_{\mu}\psi = \frac{1}{2}\tau_{\mu}\psi$, resulting in Poincaré spinors of definite chirality, while χ_{\pm} denote an SU(2)-doublet satisfying the KSE on $S^2, \nabla_{\alpha}\chi_{\pm} = \pm \frac{i}{2}\sigma_{\alpha}\chi_{\pm}$, with $\chi_{-} = \sigma_3\chi_{+}$. It is a common feature of Refs. [30,41,42] that the Majorana condition is not manifest, however conjugate spinors, η^c , may easily be constructed e.g. [43]. Following the decomposition through, one determines the effective 6D KSE equations in terms of ε_{\pm} [41] and recasts them in terms of conditions on differential forms [44], which we illustrate later.

We stress that there is *a priori* no relation between ε_{\pm} , even if one is to be expected [45]. In related work, the authors of Ref. [42] simplified the problem by omitting a term in the four-form flux, which enabled a simplification of the KSE analysis, before showing that the omitted term could not be reconciled perturbatively. This term was later ruled out in general [46]. In the current setting, this simplification involves fixing $\mathcal{A} = \mathcal{G} = 0$. However, since geometries with nonzero \mathcal{A} , \mathcal{H} , \mathcal{G} can be generated via T-duality [27], this simplification is difficult to motivate.

III. SUPERSYMMETRY CONDITIONS

We review the salient conditions on bilinears, defined in the Appendix, which we construct from spinors ϵ_{\pm} [41], which encapsulate the local supersymmetry conditions we must solve. First, supersymmetry demands that the following bilinears vanish [41],

$$W^{-} = X^{+} = \operatorname{Re}(Y) = \tilde{Z} = 0.$$
 (6)

LARGE SUPERCONFORMAL NEAR-HORIZONS FROM M-THEORY

Moreover, the remaining bilinears are constrained:

$$2mV^{+} + e^{-B}\text{Im}(Y) = \frac{e^{-3A}}{2} \left[\frac{1}{2!} \text{Im}(L^{3})_{mn} (*_{6}\mathcal{G})^{mn} + K_{m}^{+} \mathcal{A}^{m} \right], \quad (7)$$

$$\tilde{Y} = -\frac{i}{2me^B}W^+, \qquad Z = -\frac{i}{2me^B}X^-.$$
(8)

Thus, there are only three real scalars, V^{\pm} , W^+ , and one complex scalar, X^- , which can be independent.

From the vector spinor bilinears, one can identify four *real* Killing vectors on \mathcal{M}_6 [41], three of which, $\operatorname{Im}(\tilde{K}^3)$, $\operatorname{Re}(K^4)$ and $\operatorname{Im}(K^4)$ extend to symmetries of the overall solution (3). In contrast, the S^2 warp factor (also \mathcal{H}) depends on \tilde{K}^+ , thus hinting at spacetimes with larger superisometry groups [47]. Thankfully, \tilde{K}^+ may be truncated out consistently provided $V^- = 0$, i.e. for 6D spinors e_{\pm} with equal norm. Henceforth, we consider $e_{\pm}^{\dagger}e_{+} = e_{\pm}^{\dagger}e_{-}$, so that $V^- = \tilde{K}^+ = 0$.

The scalars satisfy differential constraints [41],

$$\mathrm{d}V^+ = 0, \tag{9}$$

$$d[e^{-B}Im(Y)] = 0, \qquad (10)$$

$$e^{-3A}d[e^{3A}X^{-}] = -2m\tilde{K}^4, \qquad (11)$$

$$e^{-3A}d[e^{3A}W^+] = 2m\text{Re}(K^3),$$
 (12)

while the vectors must satisfy

$$d[e^{3A+B}K^{-}] = -e^{-B}Im(Y)\mathcal{H} + e^{3A}\tilde{L}^{1}, \qquad (13)$$

$$\begin{split} \mathrm{d}[e^{6A+B}\mathrm{Re}(\tilde{K}^3)] &= -e^{3A+B}\mathrm{Im}(Y)*_6\mathcal{G}\\ &\quad -e^{3A+B}\mathcal{A}\wedge K^- - e^{6A}\mathrm{Im}(L^3), \end{split}$$

$$d[e^{6A+2B}\text{Im}(\tilde{K}^{3})] = -e^{3A}W^{+}\mathcal{H} + 2me^{6A+2B}L^{1} + 2e^{6A+B}\text{Re}(L^{3}),$$
(15)

$$d[e^{6A+2B}K^4] = ie^{3A}X^-\mathcal{H} + 2me^{6A+2B}L^6 - 2ie^{6A+B}\tilde{L}^4.$$
(16)

We have removed all trivial bilinears and conditions that play no role in our analysis [48]. With \tilde{Y} pure imaginary from (8), and \tilde{K}^+ zero, \mathcal{A} and \mathcal{G} are fully determined in terms of bilinears:

$$\mathcal{A} = \frac{2me^{3A}}{V^+}K^+, \qquad \mathcal{G} = \frac{2me^{3A}}{V^+} *_6 \operatorname{Im}(L^3).$$
 (17)

This ends our review of the supersymmetry conditions of Ref. [41]. We will now solve the conditions by

PHYSICAL REVIEW D 93, 086010 (2016)

91

evoking *G*-structures to characterize the internal manifold \mathcal{M}_{6} .

 \mathcal{M}_{6} . We introduce two unit-norm, chiral spinors, ξ_{i} , which are orthogonal, $\xi_{i}^{\dagger}\xi_{j} = \delta_{ij}$. Each chiral spinor individually defines an SU(3)-structure. To see this, we introduce projection conditions,

$$\begin{aligned} \gamma_{12}\xi_1 &= \gamma_{34}\xi_1 = \gamma_{56}\xi_1 = i\xi_1 \qquad \Rightarrow \gamma_7\xi_1 = \xi_1, \\ &-\gamma_{135}\xi_1 = \xi_1^*, \end{aligned}$$
(18)

permitting us to specify the SU(3)-structure through a two-form $J_i^{(3)} = -\frac{i}{2}\xi_i^{\dagger}\gamma_{mn}\xi_i e^m \wedge e^n$ and (3,0)-form $\Omega_i^{(3)} = -\frac{1}{3!}\xi_i^T\gamma_{mnp}\xi_i e^m \wedge e^n \wedge e^p$. With the second spinor, $\gamma_5\xi_2^* = \xi_1$, whose projection conditions follow from (18), we can define two canonical SU(3)-structures, with forms $(J_1^{(3)}, \Omega_1^{(3)})$ and $(J_2^{(3)}, \Omega_2^{(3)})$, or equivalently, a canonical SU(2)-structure, which is specified by three two-forms and two one-forms:

$$J^{\alpha} = -\frac{i}{4} (\sigma^{\alpha})^{ij} \xi_i^{\dagger} \gamma_{mn} \xi_j e^m \wedge e^n, \qquad (19)$$

$$K^{1} + iK^{2} = -\frac{1}{2}\epsilon^{ij}\xi_{i}^{T}\gamma_{m}\xi_{j}e^{m},$$
 (20)

where σ^{α} , $\alpha = 1, 2, 3$ denote Pauli matrices. In general, we expand ϵ_{\pm} in terms of the chiral spinors and their conjugates

$$\begin{aligned} \epsilon_{+} &= \alpha_{1}\xi_{1} + \alpha_{2}\xi_{1}^{*} + \alpha_{3}\xi_{2} + \alpha_{4}\xi_{2}^{*}, \\ \epsilon_{-} &= \beta_{1}\xi_{1} + \beta_{2}\xi_{1}^{*} + \beta_{3}\xi_{2} + \beta_{4}\xi_{2}^{*}, \end{aligned} \tag{21}$$

where $\alpha_i, \beta_i \in \mathbb{C}$. Modulo phases of ξ_i , these are the most general spinors consistent with SU(2)-structure.

IV. SU(2)-STRUCTURE MANIFOLDS

As a warm-up, we consider SU(3)-structure manifolds by simply eliminating, α_3 , α_4 , β_3 , β_4 , so that only ξ_1 remains in (21). We recall that SU(3)-structure manifolds are classified according to five torsion classes W_i [49]. We will now demonstrate that all torsion classes vanish, so Calabi-Yau is the only \mathcal{M}_6 with SU(3)-structure.

One can use the constraints from vanishing scalars (6) to infer, $e_{-} = \pm ie_{+}$, where e_{+} need not be chiral. If it is chiral, the argument reverts to Ref. [41]; if nonchiral, since $K^{+} = \text{Im}(L^{3}) = 0$, \mathcal{A} and \mathcal{G} also are zero. Next, from (7), we deduce $2me^{B} = \mp 1$, and from (13), $\mathcal{H} = \pm e^{3A}/(2m)J^{(3)}(V^{+} = 1)$. Further differentiating (13)–(16), we can directly confirm that $dJ^{(3)} = d\Omega^{(3)} = 0$.

We now turn to the generic case. Evaluating the vector bilinears in terms of α_i , β_i using (21), we find that K^- and \tilde{K}^- are orthogonal allowing us without loss of generality to align them with the e^5 , e^6 axes of the internal space.

086010-3

(14)

Ö. KELEKCI et al.

Moreover, imposing (6), and $V^- = \tilde{K}^+ = 0$, we determine the following relations:

$$\begin{split} \beta_1 &= -\frac{\left[\alpha_3(\alpha_4^2 + \beta_4^2) + \alpha_1(\beta_2\beta_4 + \alpha_2\alpha_4)\right]}{(\beta_2\alpha_4 - \alpha_2\beta_4)},\\ \beta_3 &= \frac{\left[\alpha_1(\alpha_2^2 + \beta_2^2) + \alpha_3(\beta_2\beta_4 + \alpha_2\alpha_4)\right]}{(\beta_2\alpha_4 - \alpha_2\beta_4)},\\ if_1 &= \beta_2\alpha_2^* + \beta_4\alpha_4^*, \qquad f_2 &= \alpha_1^*\alpha_4 + \alpha_2^*\alpha_3, \end{split}$$

where $f_i \in \mathbb{R}$ are yet to be determined. With SU(2)structure, it follows from $\tilde{K}^+ = 0$ that $K^+ = 0$ and, as a result of (17), $\mathcal{A} = 0$, i.e. no electric flux. As another consequence of these relations, we discover $\operatorname{Re}(\tilde{K}^3) = 0$, which through (14) and (17) leads to the constraint

$$Y = -\frac{i}{2me^B}V^+.$$
 (23)

Since V^+ is a constant, so too is e^B through (10). We can now combine this with (8) to find that

$$\left[\beta_4 + \frac{i}{2me^B}\alpha_4\right](f_2 - \alpha_2^*\alpha_3 - \alpha_3^*\alpha_2) = 0.$$
 (24)

If we impose the vanishing of the first bracket, through the constraints it follows that $\mathcal{G} = 0$ and $\beta_i = -i\alpha_i$, i.e. $\epsilon_- = -i\epsilon_+$, so that we recover Calabi-Yau. To find something new, we impose the second condition, which implies $K^- = \tilde{K}^- = 0$. We recall that these are the original vectors that we aligned with the axes, so now we have the freedom to choose K^3 and \tilde{K}^3 , which are orthogonal, and rotate them to align with the axes. Doing so, we find it is possible to solve for all the spinor coefficients so that our constraints are satisfied:

$$\begin{aligned} \alpha_{1} &= \sqrt{V^{+}} \cos \frac{\zeta}{2} \cos \frac{\beta}{2} e^{i\varphi_{1}}, & \alpha_{2} &= \sqrt{V^{+}} \sin \frac{\zeta}{2} \cos \frac{\beta}{2} e^{i\varphi_{2}}, \\ \alpha_{3} &= \sqrt{V^{+}} \cos \frac{\zeta}{2} \sin \frac{\beta}{2} e^{i\varphi_{3}}, & \alpha_{4} &= \sqrt{V^{+}} \sin \frac{\zeta}{2} \sin \frac{\beta}{2} e^{i\varphi_{4}}, \\ \beta_{1} &= \frac{1}{2} \left(\frac{L}{R_{2}} \cot \frac{\zeta}{2} \alpha_{4} - i \frac{L}{R_{1}} \alpha_{1} \right), & \beta_{4} &= \frac{\beta_{2}}{\beta_{1}^{*}} \beta_{3}^{*}, \\ \beta_{3} &= -\frac{1}{2} \left(\frac{L}{R_{2}} \cot \frac{\zeta}{2} \alpha_{2} + i \frac{L}{R_{1}} \alpha_{3} \right), & \beta_{2} &= -\frac{\alpha_{2}}{\alpha_{1}^{*}} \beta_{1}^{*}, \end{aligned}$$
(25)

where $\varphi_1 + \varphi_2 = \varphi_3 + \varphi_4$ and we have redefined $m = L^{-1}$, $R_1 = e^B$, $R_2 = e^B/\sqrt{4m^2e^{2B}-1}$. With these expressions, we determine $W^+ = V^+ \cos \zeta$, $X^- = V^+ \sin \zeta e^{i\varphi_1+i\varphi_2}$ and solve (11) and (12) to show the warp factor e^A is a constant and

$$e^5 = -R_2 \mathrm{d}\zeta, \qquad e^6 = -R_2 \sin\zeta \mathrm{d}\chi, \qquad (26)$$

PHYSICAL REVIEW D 93, 086010 (2016)

where we have defined $d\chi = d(\varphi_1 + \varphi_2)$. This allows us to identify the one-forms dual to the Killing vectors,

$$\operatorname{Im}(\tilde{K}^{3}) = -\frac{LV^{+}}{2}\sin^{2}\zeta d\chi,$$

$$K^{4} = -\frac{LV^{+}}{2}e^{i\chi}(d\zeta + i\cos\zeta d\chi), \qquad (27)$$

which correspond to an emergent SU(2). We can ensure the Killing vectors are canonically normalized through the choice $V^+ = 2R_2^2/L$. Solving the remaining supersymmetry conditions, one arrives at the conclusion that χ aside, the other angular parameters are constant, with \mathcal{M}_6 being a direct product of S^2 and CY_2 , more concretely T^4 or K_3 . The final expression for the four-form flux reads

$$G = \frac{2e^{3A}}{LV^+} \left[-R_1^2 \tilde{L}^1 \wedge \operatorname{vol}(S^2) + *_6 \operatorname{Im}(L^3) \right].$$
(28)

It is easy to check that the equations of motion are satisfied, in line with expectations [40]. We also see that both ξ_1, ξ_2 and conjugates need to appear in the spinor. This may be contrasted with the spinor considered in Ref. [50], which is not the most general, and would appear to preclude this outcome. For this reason, setting $\beta_4 = \alpha_4 = 0$ in (24), one recovers the results of existing classifications [29,30]. Setting A = 0, since the overall warp factor is constant, we can confirm the radii satisfy

$$\frac{4}{L^2} = \frac{1}{R_1^2} + \frac{1}{R_2^2}.$$
(29)

The ratio between S^2 radii, α , corresponds to the supergroup $D(2, 1; \alpha)$, with bosonic subgroup $SL(2, \mathbb{R}) \times SU(2) \times SU(2)$.

To establish the connection to minimal ungauged supergravity in 7D [51], we exploit the following consistent Kaluza-Klein reduction ansatz:

$$ds_{11}^2 = e^{-\frac{8}{3}B}ds_7^2 + e^{2B}ds^2(CY_2),$$

$$G = F + \sum_{a=1}^3 F^a \wedge J^a,$$
(30)

where J^a denote the three self-dual harmonic two-forms of CY_2 , *B* is a scalar and *F* and F^a are respectively field strengths corresponding to a three-form and one-form potentials, F = dC, $F^a = dA^a$. The resulting action in Einstein frame in 7D is

$$\mathcal{L}_7 = R \mathrm{vol}_7 - \frac{36}{5} \mathrm{d}B \wedge *_7 \mathrm{d}B - \frac{1}{2} e^{\frac{24}{3}B} F \wedge *_7 F$$
$$- e^{-\frac{12}{5}B} F^a \wedge *_7 F^a - F \wedge F^a \wedge A^a. \tag{31}$$

To cast the action in the original notation of Ref. [51], one should employ the following redefinitions:

LARGE SUPERCONFORMAL NEAR-HORIZONS FROM M-THEORY

$$B = \frac{\sqrt{5}}{6}\phi, \qquad F_{\rm us} = \sqrt{2}F_{\rm them}, \qquad F_{\rm us}^a = \sqrt{2}F_{\rm them}^a. \tag{32}$$

V. DISCUSSION

We have initiated a classification of all solutions to 11D supergravity with $SO(2, 2) \times SO(3)$ isometry. This is the simplest geometric signature of a supergravity solution dual to a 2D CFT with $\mathcal{N} = (0, 4)$ supersymmetry, including the MSW CFT. In the process, we have identified a novel class of near-horizon geometries in M-theory with large superconformal symmetry. Compactifying M-theory on CY_2 , we identify a resulting $AdS_3 \times S^2 \times S^2$ vacuum to 7D supergravity, thus providing a candidate near-horizon for an extremal black hole and a potential new controlled setting to count black hole microstates.

The M-theory geometry provides a unifying description of well-known $AdS_3 \times S^3 \times S^3 \times S^1$ geometries of type II string theory through T-duality [27] and heterotic vacua via M-theory/heterotic duality [36]. A careful treatment of the central charge reveals the expected form of a large superconformal algebra [52]

$$c \sim \frac{k^+ k^-}{k^+ + k^-},$$
 (33)

with affine $SU(2)^{\pm}$ current algebras at levels k^{\pm} related to the quantized charges, yet where $c \sim N^2$, for large charge N, and not the more usual $c \sim N^3$ of geometries corresponding to M5-branes.

Our work has two interesting implications. First, it is striking that the $AdS_3 \times S^2 \times S^2 \times CY_2$ geometries are not identifiable as AdS₃ limits of wrapped M5-branes [29]. This suggests the M5-brane picture is novel and motivates further study to understand anomaly inflow [11]. Second, as we have shown, since 11D supergravity compactifies on CY_2 to 7D minimal supergravity, the $AdS_3 \times S^2 \times S^2$ solution hints at being the near-horizon of an extremal black hole. While such solutions have in principle been classified [53], we are not aware of a near-horizon uniqueness theorem in 7D, cf. [54]. Assuming a black hole exists, strong parallels to the MSW case, with M-theory compactified on Calabi-Yau, are expected to facilitate a microscopic derivation of the entropy. Since the small superconformal algebra is recovered from the large one through a decompactification of a two-sphere, it is tempting to speculate that contact with the MSW results may be made in the same limit.

Last, we remark that we have assumed SU(2)-structure, and more general solutions with identity structure are known to exist [27]. We hope to extend the classification to consider more general internal manifolds in future work [52].

ACKNOWLEDGMENTS

We have enjoyed discussions with D. Chow, D. Giataganas, K. M. Lee, S. Nam, G. Papadopoulos,

PHYSICAL REVIEW D 93, 086010 (2016)

C. Park, M. Roček, D. C. Thompson, H. Yavartanoo and P. Yi. O. K. is partially supported by TUBITAK Grant No. 114F321. Y. L. and J. M. are partially supported by the Spanish Ministry of Economy and Competitiveness Grant No. FPA2012-35043-C02-02 and the EU-COST Action MP1210. J. M. is supported by the FPI Fellowship No. BES-2013-064815 linked to the previous project, and is grateful for the warm hospitality extended by YITP, SUNY Stony Brook, where part of this work was done. E. Ó. C. is supported by the Marie Curie Award No. PIOF-2012-328625 T-DUALITIES and wishes to thank APCTP and KIAS for hospitality. M. Park is supported by TJ Park Science Fellowship of POSCO TJ Park Foundation.

APPENDIX: SPINOR BILINEARS

In our conventions, the 6D gamma matrices are Hermitian $\gamma_m^{\dagger} = \gamma_m$ and antisymmetric $\gamma_m^T = -\gamma_m$. Consistent with the symmetries of the gamma matrices, given ϵ_{\pm} , we can define an exhaustive set of scalar

$$\begin{split} V^{\pm} &= \frac{1}{2} (e^{\dagger}_{+} e_{+} \pm e^{\dagger}_{-} e_{-}), \\ W^{\pm} &= \frac{1}{2} (e^{\dagger}_{+} \gamma_{7} e_{+} \pm e^{\dagger}_{-} \gamma_{7} e_{-}), \\ X^{\pm} &= \frac{1}{2} (e^{T}_{+} e_{+} \pm e^{T}_{-} e_{-}), \\ Y &= e^{\dagger}_{+} e_{-}, \qquad \tilde{Y} = e^{\dagger}_{+} \gamma_{7} e_{-}, \\ Z &= e^{T}_{+} e_{-}, \qquad \tilde{Z} = e^{T}_{+} \gamma_{7} e_{-}, \end{split}$$
(A1)

and vector spinor bilinears:

$$\begin{split} K_m^{\pm} &= \frac{1}{2} (\epsilon_+^{\dagger} \gamma_m \epsilon_+ \pm \epsilon_-^{\dagger} \gamma_m \epsilon_-), \\ \tilde{K}_m^{\pm} &= \frac{i}{2} (\epsilon_+^{\dagger} \gamma_m \gamma_7 \epsilon_+ \pm \epsilon_-^{\dagger} \gamma_m \gamma_7 \epsilon_-), \\ K_m^3 &= \epsilon_+^{\dagger} \gamma_m \epsilon_-, \qquad \tilde{K}_m^3 &= \epsilon_+^{\dagger} \gamma_m \gamma_7 \epsilon_-, \\ K_m^4 &= \epsilon_+^{-} \gamma_m \epsilon_-, \qquad \tilde{K}_m^4 &= \epsilon_+^{-} \gamma_m \gamma_7 \epsilon_-, \end{split}$$
(A2)

where factors of i ensure vectors are real. We define the following two-forms:

$$L_{mn}^{1} = \frac{i}{2} (\epsilon_{+}^{\dagger} \gamma_{mn} \epsilon_{+} + \epsilon_{-}^{\dagger} \gamma_{mn} \epsilon_{-}),$$

$$\tilde{L}_{mn}^{1} = \frac{i}{2} (\epsilon_{+}^{\dagger} \gamma_{mn} \gamma_{7} \epsilon_{+} + \epsilon_{-}^{\dagger} \gamma_{mn} \gamma_{7} \epsilon_{-}),$$

$$L_{mn}^{3} = \epsilon_{+}^{\dagger} \gamma_{mn} \epsilon_{-}, \qquad \tilde{L}_{mn}^{4} = \epsilon_{+}^{T} \gamma_{mn} \gamma_{7} \epsilon_{-},$$

$$L_{mn}^{6} = \frac{1}{2} (\epsilon_{+}^{T} \gamma_{mn} \gamma_{7} \epsilon_{+} - \epsilon_{-}^{T} \gamma_{mn} \gamma_{7} \epsilon_{-}), \qquad (A3)$$

where notation follows Ref. [41].

Ö. KELEKCI et al.

- J. D. Bekenstein, Black holes, and the second law, Lett. Nuovo Cimento 4, 737 (1972); S. W. Hawking, Particle creation by black holes, Commun. Math. Phys. 43, 199 (1975); 46, 206 (1976).
- [2] J. M. Maldacena, The large N limit of superconformal field theories and supergravity, Int. J. Theor. Phys. 38, 1113 (1999); Adv. Theor. Math. Phys. 2, 231 (1998).
- [3] J. D. Brown and M. Henneaux, Central charges in the canonical realization of asymptotic symmetries: An example from three-dimensional gravity, Commun. Math. Phys. 104, 207 (1986).
- [4] J. L. Cardy, Operator content of two-dimensional conformally invariant theories, Nucl. Phys. B270, 186 (1986).
 [5] A. Strominger and C. Vafa, Microscopic origin of the
- [5] A. Strominger and C. Vafa, Microscopic origin of the Bekenstein-Hawking entropy, Phys. Lett. B 379, 99 (1996).
- [6] A. Strominger, Black hole entropy from near-horizon microstates, J. High Energy Phys. 02 (1998) 009.
- [7] M. Guica, T. Hartman, W. Song, and A. Strominger, The Kerr/CFT correspondence, Phys. Rev. D 80, 124008 (2009).
- [8] J. L. Blum, J. M. Miller, A. C. Fabian, M. C. Miller, J. Homan, M. van der Klis, E. M. Cackett, and R. C. Reis, Measuring the spin of GRS 1915 + 105 with relativistic disk reflection, Astrophys. J. 706, 60 (2009); J. M. Miller et al., NuSTAR spectroscopy of GRS 1915 + 105: Disk reflection, spin, and connections to jets, Astrophys. J. 775, L45 (2013); L. Gou, J. E. McClintock, R. A. Remillard, J. F. Steiner, M. J. Reid, J. A. Orosz, R. Narayan, M. Hanke, and J. García, Confirmation via the continuum-fitting method that the spin of the black hole in Cygnus X-1 is extreme, Astrophys. J. 790, 29 (2014).
- [9] J. F. Donoghue, General relativity as an effective field theory: The leading quantum corrections, Phys. Rev. D 50, 3874 (1994).
- [10] J. M. Maldacena, A. Strominger, and E. Witten, Black hole entropy in M theory, J. High Energy Phys. 12 (1997) 002.
- [11] D. Freed, J. A. Harvey, R. Minasian, and G. W. Moore, Gravitational anomaly cancellation for M theory fivebranes, Adv. Theor. Math. Phys. 2, 601 (1998); J. A. Harvey, R. Minasian, and G. W. Moore, Non-Abelian tensor multiplet anomalies, J. High Energy Phys. 09 (1998) 004.
- [12] M. Ademollo *et al.*, Supersymmetric strings and color confinement, Phys. Lett. B **62**, 105 (1976); M. Ademollo *et al.*, Dual string with U(1) color symmetry, Nucl. Phys. **B111**, 77 (1976).
- [13] K. Schoutens, O(n) extended superconformal field theory in superspace, Nucl. Phys. B295, 634 (1988).
- [14] A. Sevrin, W. Troost, A. Van Proeyen, and P. Spindel, Extended supersymmetric sigma models on group manifolds. 2. Current algebras, Nucl. Phys. B311, 465 (1988); A. Sevrin, W. Troost, and A. Van Proeyen, Superconformal algebras in two dimensions with N = 4, Phys. Lett. B 208, 447 (1988).
- [15] A. Ali, Classification of two-dimensional N = 4 superconformal symmetries, arXiv:hep-th/9906096.
- [16] E. D'Hoker, J. Estes, M. Gutperle, D. Krym, and P. Sorba, Half-BPS supergravity solutions and superalgebras, J. High Energy Phys. 12 (2008) 047.
- [17] H. J. Boonstra, B. Peeters, and K. Skenderis, Brane intersections, anti-de Sitter space-times and dual superconformal theories, Nucl. Phys. B533, 127 (1998).

PHYSICAL REVIEW D 93, 086010 (2016)

- [18] J. de Boer, A. Pasquinucci, and K. Skenderis, AdS/CFT dualities involving large 2-D N = 4 superconformal symmetry, Adv. Theor. Math. Phys. 3, 577 (1999).
- [19] S. Gukov, E. Martinec, G. W. Moore, and A. Strominger, The search for a holographic dual to $AdS_3 \times S^3 \times S^3 \times S^1$, Adv. Theor. Math. Phys. 9, 435 (2005).
- [20] M. R. Gaberdiel and R. Gopakumar, Large N = 4 holography, J. High Energy Phys. 09 (2013) 036.
- [21] D. Tong, The holographic dual of $AdS_3 \times S^3 \times S^3 \times S^1$, J. High Energy Phys. 04 (2014) 193.
- [22] A. Babichenko, B. Stefanski, Jr., and K. Zarembo, Integrability and the AdS(3)/CFT(2) correspondence, J. High Energy Phys. 03 (2010) 058.
- [23] O. Ohlsson Sax and B. Stefanski, Jr., Integrability, spinchains and the AdS3/CFT2 correspondence, J. High Energy Phys. 08 (2011) 029.
- [24] O. Ohlsson Sax, B. Stefanski, Jr., and A. Torrielli, On the massless modes of the AdS3/CFT2 integrable systems, J. High Energy Phys. 03 (2013) 109.
- [25] K. Sfetsos and D. C. Thompson, On non-Abelian T-dual geometries with Ramond fluxes, Nucl. Phys. B846, 21 (2011).
- [26] Y. Lozano, E. Ó. Colgáin, K. Sfetsos, and D. C. Thompson, Non-Abelian T-duality, Ramond fields and coset geometries, J. High Energy Phys. 06 (2011) 106.
- [27] Y. Lozano, N. T. Macpherson, J. Montero, and E. Ó. Colgáin, New $AdS_3 \times S^2$ T-duals with $\mathcal{N} = (0,4)$ supersymmetry, J. High Energy Phys. 08 (2015) 121.
- [28] O. Lunin and J. M. Maldacena, Deforming field theories with U(1) × U(1) global symmetry and their gravity duals, J. High Energy Phys. 05 (2005) 033.
- [29] J. P. Gauntlett, O. A. P. Mac Conamhna, T. Mateos, and D. Waldram, AdS spacetimes from wrapped M5 branes, J. High Energy Phys. 11 (2006) 053.
- [30] H. Kim, K. K. Kim, and N. Kim, 1/4-BPS M-theory bubbles with SO(3) × SO(4) symmetry, J. High Energy Phys. 08 (2007) 050.
- [31] Indeed, the existence of the $SU(2) \times U(1)$ isometry is a puzzle, since typically emergent isometries are R symmetries, but $SU(2) \times U(1)$ does not fit with a known $\mathcal{N} = 4$ superconformal algebra. Recently, the first explicit example of a geometry in this class was identified [27] and the emergent U(1) symmetry shown not to be an R symmetry, but rather the M-theory circle.
- [32] J. P. Gauntlett, O. A. P. Mac Conamhna, T. Mateos, and D. Waldram, New supersymmetric AdS(3) solutions, Phys. Rev. D 74, 106007 (2006).
- [33] I. Bah and V. Stylianou, Gravity duals of N = (0, 2) SCFTs from M5-branes, arXiv:1508.04135.
- [34] F. Benini and N. Bobev, Two-dimensional SCFTs from wrapped branes and c-extremization, J. High Energy Phys. 06 (2013) 005.
- [35] P. Karndumri and E. Ó. Colgáin, 3D supergravity from wrapped M5-branes, J. High Energy Phys. 03 (2016) 188.
- [36] C. M. Hull and P. K. Townsend, Unity of superstring dualities, Nucl. Phys. B438, 109 (1995); E. Witten, String theory dynamics in various dimensions, Nucl. Phys. B443, 85 (1995); P. Horava and E. Witten, Heterotic and type I string dynamics from eleven-dimensions, Nucl. Phys. B460,

LARGE SUPERCONFORMAL NEAR-HORIZONS FROM M-THEORY

506 (1996); Eleven-dimensional supergravity on a manifold with boundary, Nucl. Phys. **B475**, 94 (1996).

- [37] H. Lu, C. N. Pope, and K. S. Stelle, M theory/heterotic duality: A Kaluza-Klein perspective, Nucl. Phys. B548, 87 (1999).
- [38] S. W. Beck, J. B. Gutowski, and G. Papadopoulos, Geometry and supersymmetry of heterotic warped flux AdS backgrounds, J. High Energy Phys. 07 (2015) 152.
- [39] It has been suggested that the dilaton is singular in the nearhorizon of intersecting NS5-branes and this violates the partial integration argument of Ref. [38].
- [40] J. P. Gauntlett and S. Pakis, The geometry of D = 11 killing spinors, J. High Energy Phys. 04 (2003) 039.
- [41] E. Ó. Colgáin, J. B. Wu, and H. Yavartanoo, Supersymmetric AdS₃ × S² M-theory geometries with fluxes, J. High Energy Phys. 08 (2010) 114.
- [42] H. Lin, O. Lunin, and J. M. Maldacena, Bubbling AdS space and 1/2 BPS geometries, J. High Energy Phys. 10 (2004) 025.
- [43] B. Chen, E. Ó. Colgáin, J. B. Wu, and H. Yavartanoo, N = 2 SCFTs: An M5-brane perspective, J. High Energy Phys. 04 (2010) 078.
- [44] K. p. Tod, All metrics admitting supercovariantly constant spinors, Phys. Lett. B 121, 241 (1983).

PHYSICAL REVIEW D 93, 086010 (2016)

- [45] Naive supersymmetry counting works here: four AdS₃ spinors and two S^2 spinors already give eight supersymmetries, so we just expect one from \mathcal{M}_6 .
- [46] E. Ó. Colgáin, J. B. Wu, and H. Yavartanoo, On the generality of the LLM geometries in M-theory, J. High Energy Phys. 04 (2011) 002.
- [47] We have checked that \tilde{K}^+ is nonzero for maximally supersymmetric AdS₇ × S⁴.
- [48] We have corrected a sign typo in (14).
- [49] S. Chiossi and S. Salamon, The intrinsic torsion of SU(3) and G(2) structures, arXiv:math/0202282.
- [50] J. P. Gauntlett, D. Martelli, J. Sparks, and D. Waldram, Supersymmetric AdS(5) solutions of M theory, Classical Quantum Gravity 21, 4335 (2004).
- [51] P. K. Townsend and P. van Nieuwenhuizen, Gauged sevendimensional supergravity, Phys. Lett. B 125, 41 (1983).
- [52] Ö. Kelekci, Y. Lozano, J. Montero, E. Ó. Colgáin, and M. Park (to be published).
 [52] M. Carielia and O. A. P. Mag Conservation. Timelika Killing.
- [53] M. Cariglia and O. A. P. Mac Conamhna, Timelike Killing spinors in seven dimensions, Phys. Rev. D 70, 125009 (2004).
- [54] R. Emparan and H. S. Reall, Black holes in higher dimensions, Living Rev. Relativity 11, 6 (2008); H. K. Kunduri and J. Lucietti, Classification of near-horizon geometries of extremal black holes, Living Rev. Relativity 16, 8 (2013).

3.4 Three-dimensional $\mathcal{N} = 4$ linear quivers and non-Abelian T-duals



Published for SISSA by 2 Springer

RECEIVED: October 6, 2016 ACCEPTED: November 15, 2016 PUBLISHED: November 22, 2016

Three-dimensional $\mathcal{N} = 4$ linear quivers and non-Abelian T-duals

Yolanda Lozano,^a Niall T. Macpherson,^b Jesús Montero^a and Carlos Núñez^c

^aDepartment of Physics, University of Oviedo, Avda. Calvo Sotelo 18, Oviedo, 33007 Spain

^bDipartimento di Fisica, Università di Milano-Bicocca and

INFN, sezione di Milano-Bicocca,

Milano, I-20126 Italy

^cDepartment of Physics, Swansea University, Swansea, SA2 8PP United Kingdom

E-mail: ylozano@uniovi.es, niall.macpherson@mib.infn.it, monteroaragon@uniovi.es, c.nunez@swansea.ac.uk

ABSTRACT: In this paper we construct a new Type IIB background with an AdS_4 factor that preserves $\mathcal{N} = 4$ Supersymmetry. This solution is obtained using a non-Abelian T-duality transformation on the Type IIA reduction of the $AdS_4 \times S^7$ background. We interpret our configuration as a patch of a more general background with localised sources, dual to the renormalisation fixed point of a $T_{\rho}^{\hat{\rho}}(\mathrm{SU}(N))$ quiver field theory. This relates explicitly the AdS_4 geometry to a D3-D5-NS5 brane intersection, illuminating what seems to be a more general phenomenon, relating AdS_{p+1} backgrounds generated by non-Abelian T-duality to Dp-D(p + 2)-NS5 branes intersections.

KEYWORDS: AdS-CFT Correspondence, Gauge-gravity correspondence

ARXIV EPRINT: 1609.09061

OPEN ACCESS, \bigcirc The Authors. Article funded by SCOAP³.

doi:10.1007/JHEP11(2016)133

JHEP11 (2016) 13

С	ontents	
1	Introduction	1
2	The Type IIB $\mathcal{N} = 4 \ AdS_4$ solution	3
	2.1 Reduction of \mathbb{Z}_k orbifold to IIA	4
	2.2 The non-Abelian T-dual solution	5
3	Aspects of 3d $\mathcal{N}=4$ CFTs and their holographic duals	8
	3.1 3d $\mathcal{N} = 4$ CFTs	8
	3.2 The ABEG dual geometries	10
4	The Type IIB $\mathcal{N} = 4 \ AdS_4$ solution and CFT	12
5	The Type IIB $\mathcal{N}=4~AdS_4$ solution as a ABEG geometry	15
6	Free energy	18
7	Summary and conclusions	22
Α	The Abelian T-dual limit	24
	A.1 The solution	24
	A.2 Connection with ABEG geometries	26
	A.3 Free energy	28
в	Relating Abelian and non-Abelian T-duality	28

1 Introduction

The idea of duality is very old, perhaps dating back to the (self) duality of the Maxwell equations in the absence of charges and currents. The transformation of the fields describing a given dynamics into a *different* set of fields where particular phenomena become more apparent, is a recurrent idea in Theoretical Physics. Indeed, dualities like those proposed by Montonen and Olive [1], Seiberg and Witten [2, 3], Seiberg [4], or the U-duality web in String Theory (see for example [5, 6]) are examples of this. While these dualities are very hard to prove (hence initially conjectured), they have very far reaching consequences in Physics: the phenomena that in one description are highly fluctuating and hence eminently quantum mechanical, become semiclassical and characteristically weakly coupled in the dual set of variables. The AdS/CFT duality [7–9] relating gauge theories and String theories is a paradigmatic example of this.

Other dualities, like the Kramers-Wannier self duality of the two-dimensional Ising model [10], bosonisation in two dimensions [11, 12] or T-duality in the String Theory sigma model [13–15], are within the class of dualities that can be formally proven.

In 1993, Quevedo and de la Ossa [16], following ideas in [15], proposed a non-Abelian generalisation of T-duality, applicable to the Neveu-Schwarz sector of the string sigma model. This was later complemented by Sfetsos and Thompson, who showed how to transform the fields in the Ramond sector [17]. This important work opened the way for further study involving new backgrounds and illuminating some geometrical and dual field theoretic aspects of the non-Abelian T-duality [18–23]–[44]. These works have in turn motivated the search for new classes of supersymetric AdS solutions that were overlooked until recently [45]–[49].

Whilst the sigma-model procedure to calculate the non-Abelian T-dual of a given background is apparently straightforward, many interesting subtleties related to global aspects and invertibility of the duality arise. These subtle aspects were studied in the midnineties but not completely resolved, in spite of many serious attempts [50-52]-[55]. Some of such concrete problems are the (im)possibility of extending the non-Abelian duality procedure to all orders in string perturbation theory and α' , and the determination of the range of the coordinates and topology of the dual manifold. These issues cast doubts about the 'duality-character' of the non-Abelian T-duality transformation.

One goal of this paper — elaborating on ideas introduced in [44] — is to get information on some of the global problems mentioned above. The example we will consider here involves a Type IIB background with an AdS_4 factor, preserving $\mathcal{N} = 4$ Supersymmetry.

A second goal of this paper — of interest in a broader context — will be to produce a new analytic solution to the Type IIB Supergravity equations of motion with an AdS_4 factor, that can be interpreted as an intersection of D3-D5-NS5 branes. Our example illuminates what is surely a more general phenomenon, relating AdS_{p+1} geometries generated by non-Abelian T-duality with Dp-D(p + 2)-NS5 branes intersections — see for example [45, 56, 57] for other recent studies of such configurations.

Furthermore, our case-study provides an interesting arena where the CFT interpretation of non-Abelian T-duality put forward in [44] can be tested. Indeed, using the results in [58, 59] (see also [60]), which elaborate on certain limits of Type IIB Supergravity solutions discussed in [61, 62], it is possible to associate a concrete CFT dual to our AdS_4 solution. This will be a $\mathcal{N} = 4$, d=3 conformal field theory, arising as the Renormalisation Group fixed point of a $T_{\rho}^{\hat{\rho}}(\mathrm{SU}(N))$ quantum field theory that belongs to the general class introduced by Gaiotto and Witten in [63]. These conformal field theories can be described in terms of a linear quiver with bi-fundamental and fundamental matter or, equivalently, in terms of Hanany-Witten set ups [64] containing D3, NS5 and D5 branes.

This work extends the ideas in [44] to the AdS_4/CFT_3 case. The paper [44] deals with the *singular* background obtained by the application of non-Abelian T-duality on $AdS_5 \times S^5$ and its interpretation as a Gaiotto-Maldacena type of geometry [65]. Using the formal developments of [66, 67], a *completion* to the geometry generated by non-Abelian duality was proposed, with the following relevant properties:

- It is a smooth background, except at isolated points where brane sources are located.
- The dual CFT is known explicitly.

- The coordinates of the completed geometry have a definite range, determined by imposing the matching between observables calculated with the CFT and with the geometrical description.
- The original non-Abelian T-dual background (that is, the geometry before *comple-tion*) can be seen as a *zoom-in* on a patch of the completed manifold.

In this paper, we will use a combination of insights from three-dimensional $\mathcal{N} = 4$ CFTs and their dual geometries to obtain a similar understanding of an AdS_4 Type IIB background, obtained by the action of non-Abelian T-duality on the Type IIA reduction of $AdS_4 \times S^7$. An outline of this works goes as follows.

In section 2, we present our (new) background, analyse the amount of SUSY preserved and study the structure of its singularities. The calculation of the associated charges leads us to a proposal for the Hanany-Witten set-up [64]. In section 3 we discuss aspects of $\mathcal{N} = 4$ SCFTs in three dimensions. The associated backgrounds containing an AdS_4 sub-manifold are also discussed. In sections 4 and 5, we embed our non-abelian T-dual geometry into the formalism of [58] (ABEG hereafter). This leads us to a precise proposal for the CFT dual to our background. We interpret our singular solution as embedded in a more generic background (with the characteristics itemized above). Section 6 discusses the subtle calculation of the free energy for the CFT defined by the non-abelian T-dual geometry. Conclusions and some further directions to explore are collected in section 7. Appendix A summarises the main properties of the Abelian T-dual limit of the non-Abelian solution, of relevance for the interpretation of the free energy. Finally, appendix B contains an interesting general relation between Abelian and non-Abelian T-duals.

2 The Type IIB $\mathcal{N} = 4 \ AdS_4$ solution

In this section we present the new type IIB $\mathcal{N} = 4 \ AdS_4$ background where our ideas will be tested. It is generated from the maximally supersymmetric $AdS_4 \times S^7$ solution in M-theory (once reduced to Type IIA), through a non-Abelian T-duality transformation.

To begin we parametrise the M-theory solution such that we manifestly have two three-spheres S_1^3 and S_2^3 , as

$$ds_{11d}^{2} = ds^{2}(AdS_{4}) + 4L^{2} \left(\frac{1}{4}d\mu^{2} + \sin^{2}\left(\frac{\mu}{2}\right)ds^{2}\left(S_{1}^{3}\right) + \cos^{2}\left(\frac{\mu}{2}\right)ds^{2}\left(S_{2}^{3}\right)\right),$$

$$G_{4} = \frac{3\rho^{2}}{L^{3}}dt \wedge dx_{1} \wedge dx_{2} \wedge d\rho = \frac{3}{L}\operatorname{Vol}(AdS_{4}), \qquad ds^{2}(AdS_{4}) = \frac{\rho^{2}}{L^{2}}dx_{1,2}^{2} + L^{2}\frac{d\rho^{2}}{\rho^{2}},$$
(2.1)

where as usual for AdS_4 Freund-Rubin solutions the AdS and internal radii obey the relation $R_{S^7} = 2R_{AdS_4}$. We take the three-spheres to have unit radius, which means $\mu \in [0, \pi)$. With the above parametrisation there is enough symmetry to reduce to IIA within one of the three spheres and then perform a T-duality transformation on the other. Here we will focus on performing an SU(2) non-Abelian T-duality on the residual SU(2). We also give details of the Hopf fibre T-dual in appendix A. We want to reduce to Type IIA on the Hopf direction of S_2^3 by parametrising it as

$$4ds(S_2^3) = d\theta_2^2 + \sin^2\theta_2 d\phi_2^2 + (d\psi_2 + \cos\theta_2 d\phi_2)^2, \qquad (2.2)$$

with $\psi_2 \in [0, 4\pi]$. Since some supersymmetry will be broken in the process, as the isometry parametrised by ∂_{ψ_2} defines a U(1) subgroup of the full SO(8) R-symmetry, we briefly study the Killing spinor equations. To this end we introduce the manifestly U(1) ψ_2 invariant vielbein,

$$e^{xi} = \frac{R}{L} dx_i \quad (i = t, x_1, x_2), \quad e^R = \frac{L}{R} dR,$$

$$e^{\mu} = L d\mu, \quad e^1 = L \sin\left(\frac{\mu}{2}\right) \omega_1, \quad e^2 = L \sin\left(\frac{\mu}{2}\right) \omega_2, \quad e^3 = L \sin\left(\frac{\mu}{2}\right) \omega_3.$$

$$e^{\theta_2} = L \cos\left(\frac{\mu}{2}\right) d\theta_2, \quad e^{\phi_2} = L \cos\left(\frac{\mu}{2}\right) \sin\theta_2 d\phi_2, \quad e^{\psi_2} = L \cos\left(\frac{\mu}{2}\right) (d\psi_2 + \cos\theta_2 d\phi_2).$$

$$F_4 = \frac{3}{L} e^{tx_1 x_2 \rho}, \tag{2.3}$$

where

$$\omega_1 + i\omega_2 = e^{i\psi_1} \left(id\theta_1 + \sin\theta_1 d\phi_1 \right), \qquad \omega_3 = d\psi_1 + \cos\theta_1 d\phi_1, \tag{2.4}$$

which makes manifest an additional SU(2) isometry parametrised by S_1^3 . The gravitino variation on S^7 is given in flat indices by¹

$$\nabla_a \eta + \frac{1}{4L} \Gamma_a \hat{\gamma} \eta = 0, \qquad (2.5)$$

where $\hat{\gamma} = \Gamma_t \Gamma_{x_1} \Gamma_{x_2} \Gamma_{\rho}$. The number of preserved supercharges is determined by what fraction of the initial 32 SUSYs are consistent with setting $\partial_{\psi_2} \eta = 0$ in the frame of eq. (2.3). One can show by imposing that η is independent of ψ_2 that one is lead to a single projection that the Killing spinor must obey,

$$\Gamma_{\mu\theta_2\phi_2\psi_2}\eta = -\left(\cos\left(\frac{\mu}{2}\right) + \sin\left(\frac{\mu}{2}\right)\hat{\gamma}\Gamma_{\mu}\right)\eta,\tag{2.6}$$

which breaks supersymmetry by half, leaving 16 real supercharges preserved by the reduction to Type IIA. In fact the projection also makes the Killing spinor independent of $(\theta_1, \phi_1, \psi_1)$ in the frame of eq. (2.3) and independent of ψ_1 in any frame in which the Hopf isometry of S_1^3 is manifest. These are precisely the conditions for supersymmetry to be unbroken under SU(2) and U(1) T-duality transformations respectively [35, 68]. So 16 supercharges will remain in Type IIB after either of these duality transformations, enough for this background to be dual to a three-dimensional $\mathcal{N} = 4$ SCFT.

2.1 Reduction of \mathbb{Z}_k orbifold to IIA

Let us now proceed with the reduction on ψ_2 with a slight generalisation. Let us reduce the \mathbb{Z}_k orbifold of S_2^3 . This has the effect of generating a stack of k D6 branes in Type IIA while leaving the supersymmetry arguments unchanged.

¹That the AdS_4 directions solve is a standard exercise that we omit for brevity.

Taking the \mathbb{Z}_k orbifold, amounts to sending $S_2^3 \to S_2^3/\mathbb{Z}_k$ in eq. (2.1) with,

$$4ds(S_2^3/\mathbb{Z}_k) = d\theta_2^2 + \sin^2\theta_2 d\phi_2^2 + \frac{4}{k^2} \left(d\psi_2 + \frac{k}{2}\cos\theta_2 d\phi_2 \right)^2,$$
(2.7)

where ψ_2 now has period 2π . Setting $l_p = \alpha' = g_s = 1$ leads to the type IIA solution,

$$ds_{IIA}^{2} = e^{\frac{2}{3}\phi_{0}}\cos\left(\frac{\mu}{2}\right) \left[ds^{2}(AdS_{4}) + 4L^{2}\left(\frac{1}{4}d\mu^{2} + \sin^{2}\left(\frac{\mu}{2}\right)ds^{2}(S_{1}^{3}) + \frac{1}{4}\cos^{2}\left(\frac{\mu}{2}\right)ds^{2}(S_{2}^{2})\right) \right],$$

$$F_{4} = \frac{3}{L}\operatorname{Vol}(AdS_{4}), \qquad F_{2} = -\frac{k}{2}\operatorname{Vol}(S_{2}^{2}), \qquad e^{2\Phi_{0}} = e^{2\phi_{0}}\cos^{3}\left(\frac{\mu}{2}\right), \qquad e^{\frac{2}{3}\phi_{0}} = \frac{2L}{k}.$$
(2.8)

The reduction has generated a singularity at $\mu = \pi$, but this has a physical interpretation, it is due to the k D6 branes mentioned earlier. Indeed close to $\mu = \pi$ the metric has the form,

$$ds^{2} \sim \frac{e^{2\phi_{0}/3}}{2} \left[\sqrt{\nu} \left(ds^{2} (AdS_{4}) + 4L^{2} ds^{2} (S_{1}^{3}) \right) + \frac{L^{2}}{4\sqrt{\nu}} \left(d\nu^{2} + \nu^{2} ds^{2} (S_{2}^{2}) \right) \right], \quad e^{\Phi} \sim \frac{e^{\phi_{0}} \nu^{3/4}}{2\sqrt{2}}, \tag{2.9}$$

for $\nu = (\pi - \mu)^2$. We see that the reduction has generated D6 branes that extend along AdS_4 , wrap S_1^3 and are localised at $\mu = \pi$. Of course this was to be expected as D6 brane singularities are always generated anywhere the M-theory circle shrinks to zero size.

Before moving on, let us quote the D-brane charges,

$$N_{\rm D2} = \frac{1}{2\kappa_{10}^2 T_2} \int \star F_4 = \frac{2L^6}{k\pi^2}, \qquad Q_{\rm D6} = \frac{1}{2\kappa_{10}^2 T_4} \int_{S_2^2} F_2 = k.$$
(2.10)

In our conventions $2\kappa_{10}^2 T_{Dp} = (2\pi)^{7-p}$. We thus set

$$L^6 = \frac{k\pi^2 N_{\rm D2}}{2},\tag{2.11}$$

to have integer D2 brane charge. We find the expected number of D6 branes.

2.2 The non-Abelian T-dual solution

We now present the solution that will be the main focus of this work, which is the result of performing a non-Abelian T-dual transformation on the S_1^3 of eq. (2.8). Using the rules in [35], and parametrising the T-dual coordinates in terms of spherical coordinates (r, S_1^2) , we generate the NS sector,

$$ds_{\text{IIB}}^{2} = e^{\frac{2}{3}\phi_{0}}\cos\left(\frac{\mu}{2}\right) \left[ds^{2}(AdS_{4}) + L^{2}\left(d\mu^{2} + \frac{k^{2}}{L^{6}\sin^{2}(\mu)}dr^{2} + \cos^{2}\left(\frac{\mu}{2}\right)ds^{2}(S_{2}^{2})\right) \right] \\ + \frac{L^{6}}{k^{2}\Delta}r^{2}\sin^{2}\left(\frac{\mu}{2}\right)\sin^{2}(\mu)ds^{2}(S_{1}^{2}), \\ B_{2} = \frac{L^{3}}{k\Delta}r^{3}\sin\left(\frac{\mu}{2}\right)\sin(\mu)\operatorname{Vol}(S_{1}^{2}), \qquad e^{2\Phi} = \frac{1}{\Delta}e^{2\phi_{0}}\cos^{3}\left(\frac{\mu}{2}\right),$$
(2.12)

103

where we have introduced

$$\Delta = \frac{L^3}{k^3} \sin\left(\frac{\mu}{2}\right) \sin\left(\mu\right) \left(k^2 r^2 + L^6 \sin^2\left(\frac{\mu}{2}\right) \sin^2\left(\mu\right)\right).$$
(2.13)

The solution is completed with the RR fluxes,

$$F_{3} = \frac{1}{4} \operatorname{Vol}(S_{2}^{2}) \wedge d\left(kr^{2} - \frac{L^{6}}{k} \left(\cos^{2}(\mu) - 3\right) \cos(\mu)\right),$$

$$F_{5} = \operatorname{Vol}(AdS_{4}) \wedge d\left(\frac{L^{5}}{4k^{2}} \left(\cos(2\mu) - 4\cos(\mu)\right) - \frac{3}{2L}r^{2}\right)$$

$$- \frac{L^{9}}{4k^{2}\Delta}r^{2} \sin^{3}(\mu) \sin\left(\frac{\mu}{2}\right) \operatorname{Vol}(S_{1}^{2}) \wedge \operatorname{Vol}(S_{2}^{2}) \wedge \left(3r\sin(\mu)d\mu + 2\sin^{2}\left(\frac{\mu}{2}\right)dr\right).$$
(2.14)

We have explicitly checked that the background in eqs. (2.12)-(2.14) solves the Type IIB Supergravity equations of motion, which is also implied by the result of the paper [24].

As is common to all backgrounds generated through an SU(2) non-Abelian T-duality transformation, this solution incorporates a non-compact *r*-direction. Moreover, this solution has two singularities. The first lies at $\mu = \pi$ and is inherited from the stack of D6 branes in IIA. Indeed, close to $\mu = \pi$ one finds

$$ds^{2} \sim \frac{e^{2\phi_{0}/3}}{2} \left[\sqrt{\nu} \left(ds^{2} (AdS_{4}) + L^{2} ds^{2} (S_{1}^{2}) \right) + \frac{L^{2}}{4\sqrt{\nu}} \left(d\nu^{2} + d\tilde{r}^{2} + \nu^{2} ds^{2} (S_{2}^{2}) \right) \right], \ e^{\Phi} \sim \frac{2\sqrt{\nu}}{L^{3}\tilde{r}}$$

$$(2.15)$$

where we have defined $\tilde{r} = 2k/L^2r$ and $\nu = (\pi - \mu)^2$. This is almost the behaviour of the smeared D5 stack one would generate under Hopf fibre T-duality along ψ_1 . The *r*dependence of the dilaton however modifies this. Recalling that the dilaton is determined by a one loop effect in T-duality, which essentially amounts to imposing that $e^{-2\Phi} \operatorname{Vol}(\mathcal{M}_I)$ (where \mathcal{M}_I is the submanifold where the duality is performed) is duality invariant, the *r* factor has its origin in the different volumes of the original and non-Abelian T-dual submanifolds, which are respectively S^3 and \mathbb{R}^3 . This is manifest when we parametrise the volume of \mathbb{R}^3 in spherical coordinates (r, S_1^2) , where *r* is the radial direction. The second singularity at $\mu = 0$ is also unsurprising, since we have dualised on a sphere whose radius vanishes at this point. We indeed obtain the non-Abelian T-dual analogue of smeared NS5 branes, since close to $\mu \sim 0$ we have,

$$ds^{2} \sim e^{2\phi_{0}/3} \left[ds^{2} (AdS_{4}) + L^{2} ds^{2} (S_{2}^{2}) + \frac{L^{2}}{4\nu} \left(d\nu^{2} + d\tilde{r}^{2} + \nu^{2} ds^{2} (S_{1}^{2}) \right) \right], \qquad e^{\Phi} \sim \frac{8}{L^{3} \sqrt{\nu} \tilde{r}}$$
(2.16)

where now we have defined $\nu = \mu^2$ and once more it is the dependence of the dilaton on r that makes this deviate from the conventional $(\sqrt{\nu})^{-1}$ behaviour.

As previously discussed in other non-Abelian T-dual examples — see [27, 34, 36, 43], the behaviour of the solution close to the location of the NS5-branes brings in interesting information. Close to $\mu = 0$ we have $B_2 = r \operatorname{Vol}(S_1^2)$, with the metric spanned by (μ, S_1^2)



Figure 1. b_0 as a function of r.

becoming a singular cone, which defines a 2-cycle. This means that we must ensure that on this cycle S_1^2 , the quantity

$$b_0 = \frac{1}{4\pi^2} \int_{S_1^2} B_2 \tag{2.17}$$

satisfies $b_0 \in [0, 1]$ over the infinite range of r. This is achieved by performing a large gauge transformation $B_2 \to B_2 - n\pi \operatorname{Vol}(S_1^2)$ every time we cross $r = n\pi$ for $n = 0, 1, 2, \ldots$ so that b_0 is a piecewise linear periodic function as illustrated in figure 1. In this way r is naturally partitioned into intervals of length π , with different brane content in each one of them, as the study of the Page charges reveals. Indeed, there are two charges defined on compact sub manifolds,

$$N_{\rm D5} = \frac{1}{2\kappa_{10}^2 T_5} \int_{\Sigma_1} F_3 = \frac{L^6}{k\pi}, \qquad N_{\rm D3} = \frac{1}{2\kappa_{10}^2 T_3} \int_{\Sigma_2} \left(F_5 - B_2 \wedge F_3\right) = nN_{\rm D5}, \qquad (2.18)$$

where $\Sigma_1 = (\mu, S_2^2), \Sigma_2 = (\mu, S_2^1, S_2^2)$. Thus, we need to tune

$$L^{6} = k N_{\rm D5} \pi. \tag{2.19}$$

Notice that N_{D3} is not globally defined. Instead its value depends on which interval we consider. In addition to this there are three charges that are defined on the non compact sub-manifolds,

$$\tilde{\Sigma}_1 = (r, S_1^2), \qquad \tilde{\Sigma}_2 = (r, S_2^2), \qquad \tilde{\Sigma}_3 = (r, S_1^2, S_2^2).$$
 (2.20)

We take the non compact r to be indicative of an infinite linear quiver, as shown for a related AdS_5 example in [44]. We calculate the charges in the interval $r \in [n\pi, (n+1)\pi]$ and find,

$$N_{\rm NS5} = \frac{1}{2\kappa_{10}^2 T_{\rm NS5}} \int_{S_1^2} \int_{n\pi}^{(n+1)\pi} dr \, H_3 = 1,$$

$$k_{\rm D5} = -\frac{1}{2\kappa_{10}^2 T_5} \int_{S_2^2} \int_{n\pi}^{(n+1)\pi} dr \, F_3 = (1+2n) \frac{k\pi}{4} \equiv (2n+1)k_0.$$
(2.21)

Notice that the parameter k, originally quantised in the Type IIA solution needs to be re-quantised according to $k\pi = 4k_0$, after the non-Abelian T-duality. The same happens to the size of the space L as shown in eq. (2.19).



Figure 2. (NS5, D3, D5) brane set-up. The number of D3-branes is given in N_{D5} units and that of D5-branes in k_0 units.

We can also compute

$$k_{\text{D3}} = -\frac{1}{2\kappa_{10}^2 T_3} \int_{S_1^2 \times S_2^2} \int_{n\pi}^{(n+1)\pi} dr \ (F_5 - B_2 \wedge F_3) = (3n+2)\frac{k_0}{3},$$

but this last one will not be relevant in our analysis below. Notice that all these charges are integer provided $\frac{k\pi}{12} = \frac{1}{3}k_0$ is an integer.

The previous analysis suggests a (NS5, D3, D5) brane set-up in which NS5-branes wrapped on $AdS_4 \times S_2^2$ are located at $\mu = 0, r = \pi, 2\pi, \ldots, n\pi$, with *n* running to infinity, and there are nN_{D5} D3-branes, extended on ($\mathbb{R}^{1,2}, r$) stretched among the *n*'th and (n+1)'th NS5's. On top of this, $(2n+1)k_0$ D5-branes, wrapped on $AdS_4 \times S_1^2$ and located at $\mu = \pi$, lie between the *n*'th and (n + 1)'th NS5-branes. This brane set-up is depicted in figure 2. After we recall some basic properties of 3d $\mathcal{N} = 4$ CFTs and their holographic duals, following [58, 63], we will make a concrete proposal for the field theory living on this brane configuration.

3 Aspects of 3d $\mathcal{N} = 4$ CFTs and their holographic duals

In this section we recall the basic aspects of the three dimensional $\mathcal{N} = 4$ field theories studied in [63] and of their holographic duals, derived in [58, 60]. We start with the field theory description.

3.1 3d $\mathcal{N} = 4$ CFTs

The study of the moduli space of $\mathcal{N} = 4$ SYM in four dimensions defined on an interval with SUSY preserving boundary conditions, lead Gaiotto and Witten [63] to introduce a family of 3d quantum field theories — named $T_{\rho}^{\hat{\rho}}(\mathrm{SU}(N))$, characterised by an integer Nand two partitions of it, denoted ρ and $\hat{\rho}$. From $(N, \rho, \hat{\rho})$ it is possible to read the data defining the UV of these theories, namely, the gauge group $G = \mathrm{U}(N_1) \times \ldots \times \mathrm{U}(N_k)$, the bi-fundamental fields transforming in the (N_i, \bar{N}_{i+1}) representations, and the fundamental matter, transforming under $\mathrm{U}(M_i)$ for each gauge group.

	0	1	2	3	4	5	6	7	8	9
D3	х	х	х	x						
D5	х	х	х		x	х	х			
NS55	х	х	х					x	х	x

Table 1. Hanany-Witten brane set-up corresponding to the $\mathcal{N} = 4$ 3d theory.

Given a list of positive numbers $[l_1 \ge l_2 \ge \ldots \ge l_p]$, one can define a partition ρ of N by $N = \sum_{r=1}^p M_r l_r$. The numbers M_r , which indicate how many times the different integers l_r appear in the partition, give the ranks of the fundamental matter groups in the field theory. Similarly, one can define a second partition $\hat{\rho}$, consisting of the numbers $[\hat{l}_1 \ge \hat{l}_2 \ge \ldots \ge \hat{l}_{\hat{p}}]$, with multiplicities \hat{M}_r , such that $N = \sum_{r=1}^{\hat{p}} \hat{M}_r \hat{l}_r$. From these partitions the ranks of the different $U(N_i)$ gauge groups are computed from the expressions,

$$N_i = \sum_{s=1}^{i} (m_s - \hat{l}_s), \qquad (3.1)$$

where m_s denotes the number of terms that are equal or bigger than a given integer s in the decomposition $N = \sum_{r=1}^{p} M_r l_r$.

Gaiotto and Witten [63] conjectured that the condition for these three-dimensional field theories to flow to a conformal fixed point is (schematically) $\rho^T \ge \hat{\rho}$. More specifically, this condition means that

$$\sum_{s=1}^{i} m_s \ge \sum_{s=1}^{i} \hat{l}_s \qquad \forall i = 1, \dots \hat{p}.$$
(3.2)

Associating a Young tableau with rows of lengths $[l_1, \ldots, l_p]$ to the partition ρ and one with columns of lengths $[\hat{l}_1, \ldots, \hat{l}_{\hat{p}}]$ to the partition $\hat{\rho}$, this condition means that the number of boxes in the first i-rows of the Young tableau associated to ρ^T must be larger or equal than the corresponding number in the tableau associated to $\hat{\rho}$. In those cases in which the equality holds, that is,

$$\sum_{s=1}^{i} m_s = \sum_{s=1}^{i} \hat{l}_s \quad \text{for some} \quad i \,, \tag{3.3}$$

some gauge groups have zero rank, and the quiver becomes disconnected.

The quantum theory defined by $T^{\hat{\rho}}_{\rho}(\mathrm{SU}(N))$ has Coulomb and Higgs branches of vacua, while the theory defined by $T^{\rho}_{\hat{\rho}}(\mathrm{SU}(N))$ has the same moduli space, but with the Coulomb and Higgs vacua interchanged. Both theories are conjectured to flow to the same IR fixed point, which is a reflection of mirror symmetry. The three-dimensional CFT that appears at low energies is invariant under SO(2, 3)-reflecting the conformality in 3d, and SO(4) ~ SU(2)_L × SU(2)_R-reflecting the R-symmetry of $\mathcal{N} = 4$ SUSY in 3d. This field theory can be nicely realised through a Hanany-Witten [64] set-up consisting of p D5 branes and \hat{p} NS5 branes with D3 branes stretched between them. This brane set-up is shown in table 1.

The x_3 -direction on which D3 branes stretch is of finite size, thus giving rise at long distances to a three-dimensional QFT on [0, 1, 2]. The SU $(2)_L \times SU(2)_R$ R-symmetry is

associated with rotations in the [4, 5, 6] and [7, 8, 9] directions. In turn, the $l_1 \ge l_2 \ge \ldots \ge l_p$ and $\hat{l}_1 \ge \hat{l}_2 \ge \ldots \ge \hat{l}_{\hat{p}}$ numbers that define the partitions $(\rho, \hat{\rho})$ are respectively, the linking numbers associated to the p D5 and \hat{p} NS5 branes. These are defined by

$$l_{D5,a} = l_a = -n_a + R_a^{\text{NS5}}, \qquad a = 0, \dots, p,$$

 $\hat{l}_{NS5,b} = \hat{l}_b = n_b + L_b^{\text{D5}} \qquad b = 1, \dots, \hat{p},$

where n_a is the net number of D3 branes ending on the given five brane (number of D3 on the right — number of D3 on the left). In turn, R_a^{NS5} is the number of NS5 branes to the *right* of a given D5 brane, while L_b^{D5} is the number of D5 branes to the *left* of a given NS5 brane. The multiplicities of each linking number, M_r , \hat{M}_r , are thus the number of branes in the corresponding stack of D5 or NS5 branes.

3.2 The ABEG dual geometries

Following the formulation initiated in [61, 62], the authors of [58, 60] proposed that the supergravity solutions associated to the three dimensional $\mathcal{N} = 4$ CFTs that we just described are fibrations of $AdS_4 \times S^2 \times S^2$ over a Riemann surface Σ_2 . We will refer to these geometries as ABEG geometries for brevity. These solutions have manifest SO(2, 3) × SU(2)_L×SU(2)_R symmetry, and can be completely determined from two harmonic functions $h_1(z, \bar{z}), h_2(z, \bar{z})$, defined on the Riemann surface Σ_2 . From the functions h_1, h_2 , the background and fluxes are given by,

$$ds^{2} = \lambda^{2} ds^{2} (AdS_{4}) + \lambda_{1}^{2} ds^{2} (S_{1}^{2}) + \lambda_{2}^{2} ds^{2} (S_{2}^{2}) + ds^{2} (\Sigma_{2}),$$

$$H_{3} = d(b_{1}) \wedge \operatorname{Vol}(S_{1}^{2}), \qquad F_{3} = d(b_{2}) \wedge \operatorname{Vol}(S_{2}^{2}), \qquad ds^{2} (\Sigma_{2}) = 4\bar{\rho}^{2} |dz|^{2}$$

$$F_{5} = 4(1 + \star) f \wedge \operatorname{Vol}(S_{1}^{2}) \wedge \operatorname{Vol}(S_{2}^{2}), \qquad (3.4)$$

where $\tilde{\rho}^2, \lambda, \lambda_1, \lambda_2, b_1, b_2$ and the dilaton e^{Φ} are real functions and f denotes a 1-form on Σ_2 , explicitly written below. These functions can be written in a compact form from h_1 , h_2 using,

$$W = \partial_{z}h_{1}\partial_{\bar{z}}h_{2} + \partial_{\bar{z}}h_{1}\partial_{z}h_{2}, \qquad X = i(\partial_{z}h_{1}\partial_{\bar{z}}h_{2} - \partial_{\bar{z}}h_{1}\partial_{z}h_{2})$$

$$N_{1} = 2h_{1}h_{2}|\partial_{z}h_{1}|^{2} - h_{1}^{2}W, \qquad N_{2} = 2h_{1}h_{2}|\partial_{z}h_{2}|^{2} - h_{2}^{2}W, \qquad (3.5)$$

as,

$$\tilde{\rho}^{2} = \frac{\sqrt{N_{2}|W|}}{h_{1}h_{2}}, \qquad \lambda^{2} = 2\sqrt{\frac{N_{2}}{|W|}}, \qquad \lambda^{2}_{1} = 2e^{\Phi}h_{1}^{2}\sqrt{\frac{|W|}{N_{1}}}, \qquad \lambda^{2}_{2} = 2h_{2}^{2}\sqrt{\frac{|W|}{N_{2}}},$$
$$b_{1} = 2h_{2}^{D} + 2h_{1}^{2}h_{2}\frac{X}{N_{1}}, \qquad b_{2} = -2h_{1}^{D} + 2h_{1}h_{2}^{2}\frac{X}{N_{2}}, \qquad e^{2\Phi} = \frac{N_{2}}{N_{1}}.$$
(3.6)

Here h_1^D, h_2^D are the harmonic duals of h_1, h_2 , defined such that $h_1^D + ih_1$ and $h_2 - ih_2^D$ are holomorphic functions. Notice that we are working in string frame, hence some factors of the dilaton differ from [60, 62], which use Einstein frame. Finally, the 1-form f is given by,

$$f = 2 \operatorname{Im} \left(\left[3i \left(h_1 \partial_z h_2 - h_2 \partial_z h_1 \right) + \partial_z \left(h_1 h_2 \frac{X}{W} \right) \right] \frac{\lambda_1^2 \lambda_2^2}{\lambda^4} dz \right).$$
(3.7)



Figure 3. 5-branes distribution along the strip, parameterised by z = x + iy.

It was shown in [58] that the two harmonic functions h_1 , h_2 that encode the supergravity solution as shown above, can be determined from the (D5, NS5, D3) brane set-up associated to the $T^{\hat{\rho}}_{\rho}(\mathrm{SU}(N))$ theory. Defining the sets of numbers $[N^a_5, \delta_a]$ and $[\hat{N}^b_5, \hat{\delta}_b]$, denoting respectively the number of branes at each stack and the position of this stack, for D5 and NS5 branes, and taking Σ_2 as the strip defined by $-\infty < \operatorname{Re}[z] < \infty$ and $0 \leq \operatorname{Im}[z] \leq \frac{\pi}{2}$,² the h_1 , h_2 functions are given by,

$$h_1 = -\frac{1}{4} \sum_{a=1}^p N_5^a \log \tanh\left(\frac{i\frac{\pi}{2} + \delta_a - z}{2}\right) + cc, \qquad h_2 = -\frac{1}{4} \sum_{b=1}^{\hat{p}} \hat{N}_5^b \log \tanh\left(\frac{z - \hat{\delta}_b}{2}\right) + cc$$
(3.8)

These expressions exhibit logarithmic singularities at the locations of the stacks of D5branes, at $z = \delta_a + i\pi/2$, for h_1 , and at the locations of the NS5-branes $z = \hat{\delta}_b$, for h_2 . The brane distribution is depicted in figure 3. The Laplace problem that these functions solve must be complemented by conditions on the boundaries of the Riemann surface [61, 62],

$$h_1|_{\operatorname{Im}[z]=0} = \partial_{\perp} h_2|_{\operatorname{Im}[z]=0} = 0, \qquad h_2|_{\operatorname{Im}[z]=\frac{\pi}{2}} = \partial_{\perp} h_1|_{\operatorname{Im}[z]=\frac{\pi}{2}} = 0,$$
 (3.9)

where $\partial_{\perp} = \partial_z - \partial_{\bar{z}}$, which the h_1 and h_2 in eq. (3.8) satisfy.

From the expressions for h_1 , h_2 in eq. (3.8), the fluxes in eq. (3.4) can be calculated using eqs. (3.5) and (3.6). The associated charges are defined as,

$$N_5^a = \frac{1}{2\kappa_{10}^2 T_{\text{D5}}} \int_{I \times S_2^2} F_3, \qquad \hat{N}_5^b = \frac{1}{2\kappa_{10}^2 T_{\text{N55}}} \int_{\hat{I} \times S_1^2} H_3, \qquad (3.10)$$

where the 3-cycles, defined in [58], consist of a shrinking sphere times an interval I or \hat{I} , that semi-circles the position of the singularity at δ_a or $\hat{\delta}_b$. As we discussed, (N_5^a, \hat{N}_5^b) should be identified with the multiplicities (M_r, \hat{M}_r) in the two partitions $\rho, \hat{\rho}$.

Similarly, it is possible to define two Page charges associated to D3 branes, one being the S-dual of the other:

$$N_3^a = \frac{1}{2\kappa_{10}^2 T_{\text{D3}}} \int_{I \times S_1^2 \times S_2^2} [F_5 - B_2 \wedge F_3], \qquad \hat{N}_3^b = \frac{1}{2\kappa_{10}^2 T_{\text{D3}}} \int_{\hat{I} \times S_1^2 \times S_2^2} [F_5 + C_2 \wedge H_3].$$
(3.11)

These charges are well defined whenever the potential B_2 or C_2 entering in their expression is well-defined, that is, away from the positions where the NS5 or D5 branes are located.

²This choice of strip is consistent for linear quivers (see [58, 59]).

From these and the previous charges, the linking numbers associated to the D5 and NS5 branes can be determined as [58],

$$l_a = -\frac{N_a^3}{N_5^a} = \frac{2}{\pi} \sum_{b=1}^{\hat{p}} \hat{N}_5^b \arctan(e^{\hat{\delta}_b - \delta_a}), \qquad \hat{l}_b = \frac{\hat{N}_3^b}{\hat{N}_5^b} = \frac{2}{\pi} \sum_{a=1}^p N_5^a \arctan(e^{\hat{\delta}_b - \delta_a}).$$
(3.12)

As expected, they satisfy

$$N = \sum_{b=1}^{\hat{p}} \hat{N}_5^b \hat{l}_b = \sum_{a=1}^p N_5^a l_a.$$
(3.13)

Finally, in [69] a special limit of the general expressions for h_1 and h_2 given in eq. (3.8) was considered. In this limit, the NS5-branes and D5-branes located at the two boundaries of the strip, Im z = 0, Im $z = \pi/2$, are positioned at infinite values of Re z. This limit will be useful when we discuss the realisation of the non-Abelian T-dual solution as an ABEG geometry. Specifically, it was shown in [69] that if $\delta_a \to \infty$ and $\hat{\delta}_b \to -\infty$ one can approximate eq. (3.8) by,

$$h_{1} = \sin y \sum_{a=1}^{p} N_{5}^{a} e^{x-\delta_{a}} + \dots \quad \text{if} \quad x < \delta_{1},$$

$$= \sin y \sum_{a=i}^{p} N_{5}^{a} e^{x-\delta_{a}} + \dots \quad \text{if} \quad \delta_{i-1} < x < \delta_{i},$$

$$h_{2} = \cos y \sum_{\hat{b}=1}^{\hat{p}} \hat{N}_{5}^{b} e^{\hat{\delta}_{b}-x} + \dots \quad \text{if} \quad x > \hat{\delta}_{1},$$

$$= \cos y \sum_{\hat{b}=i}^{\hat{p}} \hat{N}_{5}^{b} e^{\hat{\delta}_{b}-x} + \dots \quad \text{if} \quad \hat{\delta}_{i-1} > x > \hat{\delta}_{i}, \qquad (3.14)$$

where the strip is parameterised by z = x + iy. Notice that these expressions still satisfy the boundary conditions in eq. (3.9).

4 The Type IIB $\mathcal{N} = 4 \ AdS_4$ solution and CFT

After we have discussed the basic ingredients of $3d \mathcal{N} = 4$ CFTs and their duals, we can go back to our brane configuration, discussed at the end of section 2, and make a concrete proposal for the CFT associated to the brane set-up depicted in figure 2.

Restricting the r direction to lie between zero and $r = (n + 1)\pi$, we have a total number of n + 1 NS5-branes (see figure 4). In order to have a field theory that flows to a non-trivial infrared fixed point (see below) we need to add $(n+1)N_{D5}$ D3-branes ending on the (n + 1)'th NS5-brane from the right. This is achieved inserting a stack of $(n + 1)N_{D5}$ D5-branes to the right of the (n + 1)'th NS5-brane, each one connected to this NS5-brane by a D3-brane. In turn, this is equivalent up to a Hanany-Witten move [64] to just taking the n'th stack of D5-branes with $(2n + 1)k_0 + (n + 1)N_{D5}$ branes. This field theoretical



Figure 4. Completed (NS5, D3, D5) brane set-up.

completion of the quiver has the geometric counterpart of making finite the range of the r-coordinate.

Thus, in the notation of ABEG, we have p = n + 1 and the multiplicity of D5 branes is,

$$N_5^a = (2a+1)k_0, \qquad a = 0, \dots, n-1 \qquad N_5^n = (2n+1)k_0 + (n+1)N_{\text{D5}}.$$
 (4.1)

We can now compute the linking numbers associated to the five branes in the Hanany-Witten set-up. These provide an invariant way of encoding the brane configuration, since they do not change under Hanany-Witten moves. The linking numbers associated to the D5-branes are given by ,

$$l_a = -n_a + R_a^{\text{NS5}},\tag{4.2}$$

where n_a denotes the net number of D3-branes ending on the *a*'th stack of D5-branes and R_a^{NS5} the number of NS5-branes located at its right. For our brane set-up we find,

$$l_a = n + 1 - a$$
 for $a = 0, 1, \dots, n.$ (4.3)

From here the total number of D3-branes N, reads

$$N = \sum_{a=0}^{n} l_a N_5^a = \sum_{a=0}^{n} (n+1-a)(2a+1)k_0 + (n+1)N_{\text{D5}}$$
$$= \frac{k_0}{6}(n+1)(n+2)(2n+3) + N_{\text{D5}}(n+1).$$
(4.4)

Alternatively, we can compute N using the NS5-branes stacks. In this case the linking numbers are computed from,

$$\hat{l}_b = n_b + L_b^{\rm D5},\tag{4.5}$$

where n_b denotes once more the net number of D3-branes ending on the *b*'th NS5-brane, and L_b^{D5} denotes the number of D5-branes to the left of the *b*'th NS5-brane. We find that,

$$\hat{l}_b = N_{\rm D5} + k_0 b^2, \qquad b = 1, \dots, n+1.$$
 (4.6)

Once can easily check that

$$N = \sum_{b=1}^{n+1} \hat{l}_b N_5^b = \sum_{b=1}^{n+1} \hat{l}_b = \frac{k_0}{6} (n+1)(n+2)(2n+3) + N_{\rm D5}(n+1), \qquad (4.7)$$

as in eq. (4.4). Thus, the ρ , $\hat{\rho}$ partitions associated to the brane configuration in figure 4 read

$$o: \qquad N = \underbrace{1 + \ldots + 1}_{(2n+1)k_0 + (n+1)N_{\text{D5}}} + \underbrace{2 + \ldots + 2}_{(2n-1)k_0} + \ldots + \underbrace{(n+1) + \ldots + (n+1)}_{k_0}, \qquad (4.8)$$

and

$$\hat{\rho}: \qquad N = \underbrace{N_{\text{D5}} + (n+1)^2 k_0}_{1} + \underbrace{N_{\text{D5}} + n^2 k_0}_{1} + \dots + \underbrace{N_{\text{D5}} + k_0}_{1}. \tag{4.9}$$

These two partitions define the $T^{\hat{\rho}}_{\rho}(\mathrm{SU}(N))$ field theory associated to our brane set-up. Following now the work of ABEG [58] we can read from eq. (4.8) the number of terms, m_l , that are equal or bigger than a given integer l,

$$m_1 = (n+1)N_{D5} + (n+1)^2 k_0, \qquad m_2 = n^2 k_0, \qquad \dots, \qquad m_n = 4k_0, \qquad m_{n+1} = k_0.$$

(4.10)

From these, the condition to have a field theory that flows to a non-trivial infrared fixed point, as was conjectured in [63], is,

$$\rho^T \ge \hat{\rho} \iff \sum_{s=1}^{i} m_s \ge \sum_{s=1}^{i} \hat{l}_s \qquad \forall i = 1, \dots, n+1,$$
(4.11)

where for this to hold the \hat{l}_i must be ordered such that $\hat{l}_1 \geq \hat{l}_2 \cdots \geq \hat{l}_i$. We will use this notation in the rest of this section. In the present case we have,

$$\sum_{s=1}^{i} m_s = (n+1)N_{\text{D5}} + \sum_{q=n-i+2}^{n+1} q^2 k_0, \qquad (4.12)$$

which is strictly larger than

$$\sum_{s=1}^{i} \hat{l}_s = iN_{\text{D5}} + \sum_{q=n-i+2}^{n+1} q^2 k_0, \qquad (4.13)$$

for $i = 1, \ldots, n$, while

$$\sum_{s=1}^{n+1} m_s = \sum_{s=1}^{n+1} \hat{l}_s.$$
(4.14)

The last condition is consistent with the fact that there are k_0 D5-branes in the $[0, \pi]$ interval that are disconnected from the rest of the branes, thus leading to a quiver that breaks into two disconnected components. We can also check that, consistently with our brane set-up in figure 4, the ranks of the gauge groups are given by

$$N_i = \sum_{s=1}^{i} (m_s - \hat{l}_s) = (n+1-i)N_{\text{D5}}, \qquad (4.15)$$

and the last gauge group is empty, in agreement with the fact that there are k_0 free hypermultiplets associated to the decoupled k_0 D5-branes. Each of these gauge groups has associated M_j hypermultiplets in the fundamental, with M_j given by,

$$\rho: \qquad N = \underbrace{1 + \ldots + 1}_{M_1} + \underbrace{2 + \ldots + 2}_{M_2} + \ldots + \underbrace{(n+1) + \ldots + (n+1)}_{M_{n+1}}. \tag{4.16}$$



Figure 5. Quiver associated to the (NS5, D3, D5) brane set-up in figure 4.

These can be read from eq. (4.8) in our case. The resulting quiver is represented in figure 5, and we can see that it is fully consistent with the brane configuration in figure 4. We can also check explicitly that

$$M_i + N_{i-1} + N_{i+1} > 2N_i, (4.17)$$

a condition for the quiver to flow towards a superconformal field theory in the infrared [63].

In summary, we have seen that ending the Hanany-Witten set-up in figure 2 and completing it with flavour branes as in figure 4, lead us to a concrete proposal for a quiver describing a $T^{\hat{\rho}}_{\rho}(\mathrm{SU}(N))$ theory. The charges of the non-Abelian T-dual solution calculated in eq. (2.21) were instrumental in identifying the quiver and its completion. We will now show that the metric and other fields in the non-Abelian T-dual background are also consistent with those associated to the quiver of figure 4. The non-Abelian T-dual geometry will arise as a zooming-in on a particular region of the ABEG [58] solution associated to the $T^{\hat{\rho}}_{\rho}(\mathrm{SU}(N))$ field theory.

5 The Type IIB $\mathcal{N} = 4 \ AdS_4$ solution as a ABEG geometry

Since the solution that we generated in section 2 preserves $\mathcal{N} = 4$ SUSY, contains an AdS_4 factor and has SO(4) isometry, it is natural to expect that it should fit within the formalism described in section 3.2. Below, we will prove this. We start by redefining,

$$\sigma = -\cos\mu, \qquad \beta^2 = \frac{k^2}{L^6}.$$
(5.1)

For the non-Abelian T-dual solution in eqs. (2.12)-(2.14), we can calculate,

$$\begin{split} \lambda_1^2 &= \frac{r^2}{2\beta^2 \Delta} (1-\sigma)(1+\sigma)^2, \quad \lambda_2^2 = \frac{(1-\sigma)^{3/2}}{\sqrt{2}\beta}, \quad \lambda^2 = \frac{\sqrt{2}}{\beta} \sqrt{1-\sigma}, \quad \frac{1}{\rho^2} = 2\sqrt{2}\beta\sqrt{1-\sigma}(1+\sigma), \\ b_1 &= \frac{r^3}{\sqrt{2}\beta\Delta} \sqrt{1-\sigma}(1+\sigma) - n\pi, \quad b_2 = c_0 + \frac{k}{4} \left(r^2 + \frac{\sigma(\sigma^2 - 3)}{\beta^2} \right), \quad z = \sigma + i\beta r, \\ e^{2\Phi} &= \frac{2\sqrt{2}}{k^2\beta\Delta} (1-\sigma)^{3/2}, \quad \Delta = \frac{1}{\sqrt{2}\beta^3} \sqrt{1-\sigma}(1+\sigma) \left(\beta^2 r^2 + \frac{(1-\sigma)(1+\sigma)^2}{2} \right), \end{split}$$
(5.2)

where $d(c_0) = 0$ and the $n\pi$ comes from the contribution to B_2 of n large gauge transformations. From this we find that the functions in eqs. (3.5)–(3.6) read,

$$N_{1} = \frac{rk^{3}\Delta}{64\sqrt{2-2\sigma}\beta^{3}}, \quad N_{2} = \frac{kr(1-\sigma)}{32\beta^{4}}, \quad W = -\frac{kr}{16\beta^{2}}, \quad X = -\frac{k}{16\beta^{3}}(1+\sigma), \quad (5.3)$$

$$h_1 = \frac{kr(1+\sigma)}{4\beta}, \quad h_2 = \frac{1-\sigma}{2\beta}, \quad h_1^D = -\frac{k(1-(2+\sigma)\sigma) + (4c_0 + kr^2)\beta^2}{8\beta^2}, \quad h_2^D = \frac{1}{2}(r-n\pi).$$

Notice that the functions h_1, h_2 are harmonic. As established in [62], this implies that the equations of motion of Type IIB Supergravity are satisfied. Also, note that the definition of h_2^D in each $n\pi < r < (n+1)\pi$ cell implies it is a piecewise periodic function such that $0 < h_2^D < \pi/2$.

Thus, we have shown that the solution generated by non-Abelian T-duality — eqs. (2.12)–(2.14), fits within the class of solutions discussed in eqs. (3.4)–(3.7). It is worth stressing nevertheless that it does not satisfy the boundary conditions in eq. (3.9) — nor does it show any of the *isolated* singularities than can be associated to the positions $(\delta, \hat{\delta})$, of the D5 and NS5 branes. This suggests that the solution generated by non-Abelian T-duality could be thought of as a *limit* of the generic solutions in eqs. (3.4)–(3.7), along the lines of eqs. (3.14). We next study this in detail.

Let us start by computing the positions of the D5 and NS5 brane stacks associated to our brane configuration in figure 4. As explained in [58] and summarised in section 3.2, these positions can be computed from,

$$l_a = \frac{2}{\pi} \sum_{b=1}^{\hat{p}} \hat{N}_5^b \arctan(e^{\hat{\delta}_b - \delta_a}) \qquad \hat{l}_b = \frac{2}{\pi} \sum_{a=1}^{p} N_5^a \arctan(e^{\hat{\delta}_b - \delta_a}).$$

These equations are simply solved by

$$e^{\hat{\delta}_b - \delta_a} = \tan\left(\frac{\pi}{2} \frac{l_a \hat{l}_b}{N}\right). \tag{5.4}$$

Using eqs. (4.3) and (4.6) this gives for our brane set-up

$$\hat{\delta}_b - \delta_a = \log \left[\tan \left(\frac{\pi}{2N} (n+1-a) (N_{\rm D5} + k_0 b^2) \right) \right],$$
 (5.5)

with N given by eq. (4.4).

Recalling that we read the charges of our brane configuration from the supergravity solution, we expect to find a sensible solution to eq. (5.5) in the supergravity limit $N_{D5} \rightarrow \infty$. Taking this limit we find,

$$\hat{\delta}_b - \delta_a = \log\left[\tan\left(\frac{\pi}{2}\left(1 - \frac{a}{n+1}\right)\right)\right] \tag{5.6}$$

which shows that in this limit all stacks of NS5-branes can be approximately taken at the same position $\hat{\delta}$. Equivalently, we can write eq. (5.6) as

$$\delta_a - \hat{\delta} = \log\left[\tan\left(\frac{\pi a}{2(n+1)}\right)\right]. \tag{5.7}$$



Figure 6. Positions of D5 and NS5 branes in the supergravity limit. In this limit the set-up becomes symmetric around $\hat{\delta} = \delta_{(n+1)/2}$, with the exception of the detached stack of D5-branes at $\delta_0 = -\infty$.

From here we see that the first stack of (detached) k_0 D5-branes lies strictly at $\delta_0 - \hat{\delta} = -\infty$, while the rest of stacks lie symmetrically at both sides of the NS5-branes, given that

$$\log\left[\tan\left(\frac{\pi c}{2(n+1)}\right)\right] = -\log\left[\tan\left(\frac{\pi a}{2(n+1)}\right)\right] \quad \text{for} \quad c = n+1-a.$$
(5.8)

Thus,

$$\delta_1 - \hat{\delta} = \hat{\delta} - \delta_n, \qquad \delta_2 - \hat{\delta} = \hat{\delta} - \delta_{n-1}, \qquad \dots, \qquad \hat{\delta} = \delta_{(n+1)/2}. \tag{5.9}$$

This brane distribution is depicted in figure 6. Let us now obtain the h_1 , h_2 functions associated to this configuration, following [58, 59].

In the supergravity limit the main contribution to h_1 in eq. (3.8) comes from the *n*'th stack, given that the number of branes in this stack goes with N_{D5} as shown by eq. (4.1). We can then approximate,

$$h_1 \sim -\frac{1}{4} N_5^n \log \tanh\left(\frac{i\frac{\pi}{2} + \delta_n - z}{2}\right) + cc.$$
 (5.10)

For h_2 we have in turn

$$h_2 \sim -\frac{1}{4}(n+1)\log \tanh\left(\frac{z-\hat{\delta}}{2}\right) + cc.$$
(5.11)

Choosing

$$\delta_n = -\hat{\delta} = -\frac{1}{2} \log \left[\tan \left(\frac{\pi}{2(n+1)} \right) \right],\tag{5.12}$$

the *n*'th stack of D5-branes lies approximately at plus infinity for large *n* while the stack of NS5-branes lies approximately at minus infinity. For finite *x* we can then use the approximate expressions for h_1 , h_2 in eq. (3.14) to produce,

$$h_{1} \sim \sin y \, N_{5}^{n} \, e^{x - \delta_{n}} \sim \sin y \, N_{\text{D5}} \, \sqrt{\frac{\pi(n+1)}{2}} e^{x},$$

$$h_{2} \sim \cos y \, (n+1) \, e^{\hat{\delta} - x} \sim \cos y \, \sqrt{\frac{\pi(n+1)}{2}} \, e^{-x},$$
 (5.13)

where we have approximated $\delta_n = -\hat{\delta} \sim -\frac{1}{2} \log \left[\left(\frac{\pi}{2(n+1)} \right) \right]$ for large *n*. Close to y = 0, x = 0 we have,

$$h_1 \sim y N_{\text{D5}} \sqrt{\frac{\pi(n+1)}{2}} (1+x),$$

 $h_2 \sim \sqrt{\frac{\pi(n+1)}{2}} (1-x).$ (5.14)

Let us now compare these expressions with those of our non-Abelian T-dual solution. Taking the functions h_1 , h_2 for this solution from eq. (5.3),

$$h_1 = \frac{kr(1+\sigma)}{4\beta},$$

$$h_2 = \frac{1-\sigma}{2\beta},$$
(5.15)

we find that they agree with eq. (5.14) if we identify $x = \sigma$ and

$$\frac{1}{2\beta} \sim \sqrt{\frac{\pi(n+1)}{2}}, \qquad r \sim \frac{2}{k} N_{\text{D5}} y.$$
 (5.16)

Taking into account that $\beta = k/L^3$, and $N_{D5} = L^6/(\pi k)$, these are equivalent to

$$N_{\rm D5} \sim 2k(n+1), \qquad r \sim 4(n+1)y.$$
 (5.17)

The output of this analysis is that the non-Abelian T-dual solution comes out when zooming into the region x = y = 0 of the ABEG solution associated to the brane set-up in section 4. Further, we have shown that in order to match these solutions n must go to infinity as N_{D5}/k , which is consistent with the fact that n is unbounded in the non-Abelian T-dual solution. Note however that this limit should be taken directly in equation (5.5) for consistency of the previous analysis. We have checked numerically that the matching between the non-Abelian T-dual solution and the ABEG geometry still holds in this limit in the region $x \sim 0$, $y \sim 0$. In this matching we must have $\sigma \sim x$, $r \sim 4(n+1)y$. Therefore, σ , which in the non-Abelian T-dual solution ranges in [-1, 1], must be small. The coordinate r in turn, may cover a finite region depending on how the $y \rightarrow 0$ limit is taken in the expression above, which is unspecified in our analysis.

The previous agreement suggests that we may see the ABEG solution as a *completion* of the non-Abelian T-dual geometry, that: i) Extends it to $-\infty < \sigma < \infty$, such that the singularities in $\sigma = \pm 1$ are moved to $\pm \infty$, and thus resolved, and ii) Delimits r to a bounded region. This is shown pictorially in figure 7. It is interesting that this completion makes explicit the ideas in [44], where a completion of the non-Abelian T-dual of $AdS_5 \times S^5$ as a superposition of Maldacena-Nunez geometries was outlined.

6 Free energy

The authors of reference [69] computed the free energy of some specific examples of $T^{\hat{\rho}}_{\rho}(\mathrm{SU}(N))$ field theories, both directly in the field theory as well as using holography.



Figure 7. The ABEG set-up in the limit in which the NS5-branes are placed at $-\infty$ and the D5branes at $+\infty$ (left). The non-Abelian T-dual set-up, with $\sigma \in [-1, 1]$, $r \in [0, +\infty)$, with smeared NS5-branes at $\sigma = -1$ and smeared D5-branes at $\sigma = +1$ (right). The matching of the solutions occurs locally around $x, y \sim 0$. The *completion* of the non-Abelian T-dual geometry is achieved extending σ to $\pm\infty$ and bounding r to an interval. In this completion the NS5-branes are localised at $-\infty$ and the D5-branes at $+\infty$.

This free energy was shown to exhibit a $\frac{1}{2}N^2 \log N$ behaviour at leading order. Further, it was argued that this value should provide an upper bound to the free energy of any, more general, $T_{\rho}^{\hat{\rho}}(\mathrm{SU}(N))$ field theory.

In this section we compute the free energy associated to the non-Abelian T-dual solution and compare it to that of the completed ABEG geometry. We show that, as expected, the free energies do not agree, consistently with the fact that the non-Abelian T-dual geometry approximates the ABEG geometry only in a small patch. On the contrary, this calculation shows explicitly how the completion of the non-Abelian T-dual geometry leads to a sensible value for the free energy of the dual $T^{\hat{\rho}}_{\rho}(\mathrm{SU}(N))$ field theory that is in consonance with previous results in the literature and satisfies the bound found in [69].

We will use the conventions in [69]. In this reference the free energy is computed from

$$S_{\rm eff} = \frac{1}{2^4 \pi^5} {\rm Vol}_6,$$
 (6.1)

where Vol₆ is the volume of the six dimensional internal space, which can be calculated from the functions h_1 , h_2 , defined in the 2d manifold Σ_2 as,

$$\operatorname{Vol}_{6} = 32(4\pi)^{2} \int_{\Sigma} d^{2}x(-W)h_{1}h_{2}.$$
(6.2)

We first use this expression to compute the free energy associated to the non-Abelian T-dual solution. In this case we find,

$$h_1 h_2 = \frac{k}{8\beta^2} r(1 - \sigma^2), \qquad W = -\frac{kr}{16\beta^2},$$
 (6.3)

and

$$Vol_6 = 32(4\pi)^2 \frac{k^2}{2^7 \beta^4} \int_{\Sigma} r^2 (1-\sigma^2)\beta dr d\sigma.$$
 (6.4)

Here we have used that the differential area of the strip is $d\Sigma = \beta dr d\sigma$. Integrating $r \in [0, (n+1)\pi]$ and $\sigma \in [-1, 1]$ we find

$$S_{\rm eff} = \frac{\pi^{3/2}}{9} \sqrt{k} N_{\rm D5}^{3/2} (n+1)^3.$$
 (6.5)

As in previous non-Abelian T-duals of AdS backgrounds, this free energy exhibits the same behaviour, in this case as $\sqrt{k}N^{3/2}$, of the original AdS background, multiplied by a power of (n + 1) coming from the NS5-branes.

Let us now analyse the "Abelian T-dual limit" of this expression. This limit was first discussed in [44] at the level of the central charges. It was shown that the central charge (and the free energy) of the SU(2) non-Abelian T-dual of $AdS_5 \times S^5$ and that of its Abelian T-dual counterpart³ exactly match if r is taken in a $r \in [n\pi, (n+1)\pi]$ interval and n is sent to infinity. In this limit both metrics do in fact fully agree. We have presented a detailed analysis of this limit in appendix A for the present AdS_4 non-Abelian T-dual solution and its Abelian T-dual counterpart (see also appendix B). Borrowing the result for the free energy of the Abelian T-dual solution we can show that it fully agrees with the free energy of the non-Abelian solution for $r \in [n\pi, (n+1)\pi]$ and $n \to \infty$.

Indeed, integrating $r \in [n\pi, (n+1)\pi]$, $\sigma \in [-1, 1]$ in eq. (6.4) and taking the $n \to \infty$ limit, we find

$$S_{\rm eff} = \frac{\pi^{3/2}}{3} \sqrt{k} N_{\rm D5}^{3/2} n^2.$$
 (6.6)

Using now that $N_{D3} = nN_{D5}$ for the non-Abelian solution and that in the large n limit $k_{D5} = nk\pi/2$, as implied by the second expression in eq. (2.21), eq. (6.6) can be rewritten as

$$S_{\rm eff} = \frac{\sqrt{2}\pi}{3} \sqrt{k_{\rm D5}} N_{\rm D3}^{3/2}.$$
 (6.7)

One can see that this result matches exactly the free energy of the Abelian T-dual background, given by eq. (A.12). As stressed in [44], this calculation shows that non-Abelian T-duality in an interval of length π corrects the Abelian T-duality calculation by 1/n terms. In our present set-up this provides a non-trivial check of the validity of expression (6.4).

Let us now compare the free energy of the full non-Abelian solution, given by eq. (6.5), to the free energy computed from the completed ABEG geometry. As shown in [69] the approximated expressions given by (3.14) are enough to capture the leading order behaviour. Using then these approximated expressions for h_1 , h_2 for our particular ABEG geometry, given by eqs. (5.13), we can write $h_1h_2 \sim \frac{1}{2}\sin 2yN_{\text{D5}}(n+1)^2e^{-2\delta_n}$ and

$$W = \frac{1}{4} \frac{\partial^2}{\partial y^2} (h_1 h_2) = -h_1 h_2.$$
(6.8)

The internal volume then reads,

$$\operatorname{Vol}_{6} = 16 \, (4\pi)^{2} N_{\mathrm{D5}}^{2} (n+1)^{4} e^{-4\delta_{n}} \int_{0}^{\frac{\pi}{2}} dy \sin^{2} 2y \int_{\hat{\delta}}^{\delta_{n}} dx, \tag{6.9}$$

which gives to leading order,

$$\operatorname{Vol}_{6} = 16\pi^{5} N_{\mathrm{D5}}^{2} (n+1)^{2} \log (n+1), \qquad (6.10)$$

³Namely, the result of T-dualising the original $AdS_5 \times S^5$ background along the Hopf fibre of the S^3 in the internal space.

and finally,

$$S_{\text{eff}} = N_{\text{D5}}^2 (n+1)^2 \log (n+1). \tag{6.11}$$

We thus see that the free energy of the $T_{\rho}^{\hat{\rho}}(N)$ theory associated to our configuration exhibits a similar logarithmic behaviour to that of the examples discussed in [69]. As in those examples, the logarithm comes holographically from the size of the configuration. In our case it depends however on the number of NS5-branes, rather than on the number of D3-branes. Interestingly, taking into account the relation between N and n, given by eq. (4.4), the free energy given by eq. (6.11) satisfies the $\frac{1}{2}N^2 \log N$ bound suggested in [69] for the free energy of general $T_{\rho}^{\hat{\rho}}(N)$ field theories. This is to our knowledge the first check in the literature of the conjecture in [69].

We would like to note that for our particular $T_{\rho}^{\hat{\rho}}(N)$ theory, there is no field theoretical computation in the literature, along the lines of [70, 71], with which we could compare our holographic result. Indeed, the scaling limit taken in the field theory computation in [69], given by

$$N_5^a = N^{1-\kappa_a} \gamma_a; \qquad l_a = N^{\kappa_a} \lambda_a, \tag{6.12}$$

with $0 \leq \kappa_a < 1$ and $\sum_{a=1}^{p} \gamma_a \lambda_a = 1$, is not fulfilled by our configuration, for which only $\kappa_n = 0$ is well-defined. The reason we avoid this scaling is that there is a further N dependence in the number p that appears in $N = \sum_{a=1}^{p} N_5^a l_a$, as compared to the situation considered in [69]. It would be interesting to extend the field theory calculation in [69] to cover the present, more general, set-up, and check if the result matches the holographic computation.

As we have previously mentioned, we can see quite explicitly from the calculation of the free energy how the non-Abelian T-dual solution is completed by the ABEG geometry. Indeed, taking into account the different parametrisation of the strip in the non-Abelian T-dual solution, $d\Sigma = \beta dr d\sigma$, and in the ABEG solution, $d\Sigma = dxdy$, and doing the completions

$$\int_{-1}^{1} (1 - \sigma^2) d\sigma \to \int_{\hat{\delta}}^{\delta_n} e^x e^{-x} dx, \qquad (6.13)$$

and

$$\beta \int_0^{(n+1)\pi} r^2 dr \to 2(n+1)^2 \int_0^{\pi/2} \sin^2 2y dy, \tag{6.14}$$

which extend in a particular way the relations $\sigma \sim x$, $r \sim 4(n+1)y$, valid in the $x, y \sim 0$ region, we can recover exactly the free energy associated to the ABEG solution, given by eq. (6.11), from that of the non-Abelian T-dual solution. In this completion the logarithm is associated to the infinite extension of the configuration in the x direction, which is what allows us to send the singularities in $\sigma = \pm 1$ to $\pm \infty$. Note that the completion changes as well, and quite dramatically, the $\sqrt{k}N_{\text{D5}}^{3/2}(n+1)^3$ scaling of the free energy of the non-Abelian T-dual solution into the $N_{\text{D5}}^2(n+1)^2$ scaling associated to the ABEG geometry. Interpreting the behaviour of the free energy of AdS backgrounds generated through non-Abelian T-duality has remained an interesting open problem in the non-Abelian T-dual iterature. Indeed, in all examples analysed so far the free energy of the non-Abelian T-dual was simply that of the original background corrected by a factor of (n + 1) to some power, associated to the NS5-branes. One was thus led to interpret that non-Abelian T-duality was not changing too much the field theory. Instead, the detailed calculation done in the present example shows that the completion needed to correctly define the dual CFT can change this behaviour quite significantly.

To summarise, we have shown that expressions (6.13), (6.14) inform us about the precise way in which the non-Abelian T-dual solution must be *completed* in order to describe holographically a $T_{\rho}^{\hat{\rho}}(SU(N))$ theory:

- Expression (6.13) shows that the interval $\sigma \in [-1,1]$ must be extended to $\sigma \in (-\infty,\infty)$. The two singularities at $\sigma = \pm 1$ are then moved to infinity such that a perfectly smooth background remains.
- Expression (6.14) informs us about how precisely the non-compact direction of the non-Abelian T-dual solution must be bounded.

Our AdS_4 example thus provides a new AdS background in which the CFT dual can be used to define the geometry, in complete analogy with the AdS_5 case discussed in [44]. It also shows that the completion can significantly change the scaling of the free energy, and thus the CFT. This may shed some light on the possible interpretation of the behaviour of the free energy under non-Abelian T-duality.

7 Summary and conclusions

Let us start by summarising the contents of this paper. Then we will present some ideas for future work and comment on open problems that our results suggest.

We started by constructing a new solution to the Type IIB equations of motion. This new background consists of an AdS_4 factor and two spheres S_1^2, S_2^2 , fibered on a Riemann surface $\Sigma(z, \bar{z})$. A dilaton, NS three form and Ramond three and five forms complete it. The system preserves sixteen supercharges and is obtained acting with non-Abelian T-duality on the dimensional reduction of $AdS_4 \times S^7$ to Type IIA. Both the original type IIA and its type IIB counterpart are singular. An important achievement of this paper is to understand the way of completing the geometry so that the only remaining *isolated* singularities are associated with brane sources. Global aspects of the geometry have also been understood thanks to this completion.

The procedure that we used to achieve these results can be summarised as follows. The study of the Page charges in section 2, suggested the brane distribution and Hanany-Witten set-up. The isometries of the background indicated the global symmetries of the dual field theory and the same goes for the amount of preserved SUSY. These data constrained our system in an important way, and suggested the way in which the Hanany-Witten set up, that in principle is *unbounded*, can be *completed* (hence *closed*) by the addition of flavour branes. This completion, shown explicitly comparing figure 4 with figure 2, is needed in order to define the partitions from which the $T^{\hat{\rho}}_{\rho}(SU(N))$ dual theory can be read. The position (in theory space) where this completion takes place is arbitrary and determines the parameters of the dual field theory. From here, the knowledge of the associated field

theory, that in this case flows to a conformal fixed point, is constraining enough to allow us to write a precise completed Type IIB background in terms of a couple of holomorphic functions defined on a Riemann surface. This background describes an intersection of D3-D5-NS5 branes, and is smooth, except at the isolated positions of the five brane sources. Then, we discussed how a particular zoom-in on a region of the completed background gives place to the original Type IIB solution obtained by non-Abelian T-duality. Finally, our calculation of the free energy showed explicitly that this completion produces a sensible result for the free energy of the associated $T_{\rho}^{\hat{\rho}}(SU(N))$ field theory, satisfying the upper bound $\frac{1}{2}N^2 \log N$ found in [69]. This result suggests that there could be a scaling in the field theory side that reproduces our gravitational result.

A couple of points are crucial in the previous summary. On the one hand we have assumed that the fluxes capture faithfully the brane distribution (except, of course, for the completion with flavour branes). This has allowed us to suggest a Hanany-Witten set-up and to calculate (after completing it) the linking numbers that select the particular CFT. On the other hand, the fact that we are using field theory knowledge to smooth out a supergravity background is quite original and key to our procedure.

Interestingly, our approach has also allowed us to find out about global properties, in particular about the range of the *r*-coordinate (which is one of the long-standing problems of the whole non-Abelian T-duality formalism). It also gives a clean way of resolving or interpreting singularities in terms of sources. This is particularly nice since the presence of these sources is a consequence of the flavour symmetry on the field theory side, that also reflects in the completed quiver. A circle of ideas closes nicely.

What remains to be done (for this particular system and more generally)?

The proposed picture of intersecting Dp-D(p + 2)-NS5 branes associated with an AdS_{p+1} background should be tested in detail. For this, a more complete case-by-case study is needed. Examples with different dimensionality might reveal new subtleties, that in the present study or in that of [44] do not show. In particular, it is clear that backgrounds with AdS_6 and AdS_3 factors should be studied following the ideas presented here. Progress should be possible in cases with less SUSY and smaller isometry groups.

In relation to the present AdS_4 /CFT₃ case, it would be interesting to investigate Wilson loops, vortex operators [72] and other subtle CFT aspects — see for example [73– 75], to understand, in particular, how our solutions capture these fine-points. The study of the spectrum of glueballs and mesons using our backgrounds (both the one obtained via non-Abelian T-duality and the completed one) is also of potential interest to learn about the nature of the duality. It would also be interesting to understand the geometric realisation of the decoupled flavour group in the quiver associated to our completed geometry

More generally, it would be very interesting to find out a precise answer for what is the effect of a non-Abelian T-duality transformation at the CFT level. In our example we started with a background dual to a CFT with one node and adjoint matter, to which we associated (after a non-Abelian T-duality transformation) a quiver containing a large number of colour and flavour groups. But, how precisely did we go from one quiver to the other? Is an 'unhiggsing' at work, or is the non-Abelian T-duality a genuine non-field theoretical operation? For a quite particular case of non-Abelian T-duality transformation, some progress was recently reported in [76].

Finally, it would be very nice if the ideas developed in this work could be used to answer deep questions about the nature of non-Abelian T-duality in String theory. For example, its invertibility, or the character of the genus and α' -expansions. We have given some evidence that the AdS/CFT correspondence can be very useful also in this regard.

Acknowledgments

Some colleagues contributed with discussions and clarifications that are reflected in this paper. For these, we thank Benjamin Assel, Georgios Itsios, Diego Rodríguez-Gomez, Daniel Thompson, Salomón Zacarías.

N.T.M. is supported by INFN and by the European Research Council under the European Union's Seventh Framework Program (FP/2007-2013) — ERC Grant Agreement n. 307286 (XD-STRING). Y.L. and J.M. are partially supported by the Spanish and Regional Government Research Grants FPA2015-63667-P and FC-15-GRUPIN-14-108. J.M. is supported by the FPI grant BES-2013-064815 of the Spanish MINECO. This work has been partially supported by the EU-COST Action MP1210, in particular through a Short Term Scientific Mission of J.M. to Swansea U. Y.L. and J.M. would like to thank the Physics Department of Swansea U. for the warm hospitality. C. Nunez is Wolfson Fellow of the Royal Society.

A The Abelian T-dual limit

In this appendix we summarise the key properties of the $\mathcal{N} = 4 \ AdS_4$ type IIB solution that is generated from the IIA solution in equation (2.8) by T-duality along the Hopf direction of S_1^3 . This solution, dual to a circular quiver, was discussed at length in [59]. Here we recall its more relevant properties in the notation used in this paper. We show that it emerges as the $r \to \infty$ limit of the non-Abelian T-dual solution in section 2 (see appendix B). In this limit the free energies of both solutions also agree, as shown in section 6.

A.1 The solution

The Abelian T-dual of the IIA solution in equation (2.8) along the Hopf direction of S_1^3 reads:

$$ds_{\text{IIB}}^{2} = e^{\frac{2}{3}\phi_{0}}\cos\left(\frac{\mu}{2}\right) \left[ds^{2}(AdS_{4}) + L^{2}\left(d\mu^{2} + \frac{k^{2}}{L^{6}\sin^{2}(\mu)}dr^{2} + \cos^{2}\left(\frac{\mu}{2}\right)ds^{2}(S_{2}^{2})\right) \right] \\ + \frac{L^{3}}{k}\sin\left(\frac{\mu}{2}\right)\sin(\mu)ds^{2}(S_{1}^{2}), \qquad B_{2} = \cos\theta_{1}d\phi_{1} \wedge dr, \qquad e^{2\Phi} = \frac{e^{4\phi_{0}/3}}{L^{2}\tan^{2}\left(\frac{\mu}{2}\right)}.$$
(A.1)

Since we dualise on the Hopf fibre $0 < \psi_1 < 4\pi$, we have $0 < \tilde{\psi}_1 < \pi$.⁴ To ease notation we choose to label $\tilde{\psi}_1 = r$, as we have a similar coordinate in the non-Abelian T-dual case

⁴Recall that the periodicity of the Abelian T-dual coordinate is fixed by the condition $\int d\psi_1 \wedge d\tilde{\psi}_1 = (2\pi)^2$.

of section 2, here though we stress that it is compact. Additionally this background is supported by the gauge invariant RR fluxes

$$F_3 = \frac{k}{2} dr \wedge \operatorname{Vol}(S_2^2), \qquad F_5 = -\frac{3}{L} dr \wedge \operatorname{Vol}(AdS_4) + \frac{3L^6}{4k} \sin^3(\mu) d\mu \wedge \operatorname{Vol}(S_2^1) \wedge \operatorname{Vol}(S_2^2).$$
(A.2)

As observed in [43, 44] for other AdS backgrounds (see appendix B for a general analysis), this solution arises in the $r \to \infty$ limit of the non-Abelian T-dual solution derived in section 2. This is straightforward for the metric and the NS-NS 2-form,⁵ while the dilatons differ by an r^2 factor that accounts for the different integration measures in the partition functions of the Abelian and non-Abelian T-dual σ -models, as explained in [44]. The RR sector, even if the fields are different (see appendix B), yields to the same quantised charges, as we show below. Finally, in order to match both solutions globally, r must live in an interval of length π , $r \in [n\pi, (n+1)\pi]$, with $n \to \infty$. It is indeed in this limit in which there is precise agreement between the corresponding free energies.

The Abelian T-dual solution is also $\mathcal{N} = 4$ supersymmetric, as discussed in section 2, and has two singularities. The first singularity at $\mu = \pi$ is inherited from the stack of D6 branes in IIA. Indeed, one finds that for $\mu \sim \pi$

$$ds^{2} \sim \frac{e^{2\phi_{0}/3}}{2} \bigg[\sqrt{\nu} \big(ds^{2} (AdS_{4}) + L^{2} ds^{2} (S_{1}^{2}) \big) + \frac{L^{2}}{4\sqrt{\nu}} \big(d\tilde{r}^{2} + d\nu^{2} + \nu^{2} ds^{2} (S_{2}^{2}) \big) \bigg], \ e^{\Phi} \sim \frac{e^{2\phi_{0}/3}}{2L} \sqrt{\nu}$$
(A.3)

which is the metric close to flat space smeared D5's, where we have defined $\tilde{r} = 4e^{-2\phi_0/3}/L^2r$ and $\nu = (\pi - \mu)^2$. The second singularity at $\mu = 0$ is caused by NS5 branes localised there, wrapping S_2^2 and smeared along r. This is a generic result of T-dualising on the Hopf fibre of a 3-sphere with vanishing radius. Close to $\mu = 0$ one finds the metric (now $\nu = \mu^2$)

$$ds^{2} \sim e^{2\phi_{0}/3} \left[ds^{2} (AdS_{4}) + L^{2} ds^{2} (S_{2}^{2}) + \frac{L^{2}}{4\nu} \left(d\tilde{r}^{2} + d\nu^{2} + \nu^{2} ds^{2} (S_{1}^{2}) \right) \right], \qquad e^{\Phi} \sim \frac{2e^{2\phi_{0}/3}}{L\sqrt{\nu}},$$
(A.4)

as expected.

The Page charges of this solution are given by

$$N_{\rm D3} = \frac{1}{2\kappa_{10}^2 T_3} \int_{\Xi_2} (F_5 - B_2 \wedge F_3) = \frac{N_{\rm D2}}{2},$$

$$k_{\rm D5} = \frac{1}{2\kappa_{10}^2 T_5} \int F_3 = \frac{k}{2},$$

$$N_{\rm NS5} = \frac{1}{2\kappa_{10}^2 T_{\rm NS5}} \int_{(r,S_1^2)} H_3 = 1,$$
(A.5)

where $\Xi_1 = (r, S_2^1, S_2^2)$, $\Xi_2 = (\mu, S_2^1, S_2^2)$ and we keep *L* defined as it was for IIA in eq (2.11). The factors of 2 in k_{D5} , N_{D3} originate from the different periodicities of the original and T-dual variables. They are usually absorbed through a redefinition of Newton's

⁵Note that B_2 arises in the gauge $B_2 = r \operatorname{Vol}(S_1^2)$ in the Abelian T-dual.



Figure 8. Brane set-up for the Abelian T-dual of the IIA reduction of $AdS_4 \times S^7/(\mathbb{Z}_k \times \mathbb{Z}_{k'})$. At each interval there are N_{D3} D3-branes stretched between the NS5-branes and $k_{D5} = k/2$ transverse D5-branes.

constant. Comparing these charges with those of the non-Abelian T-dual solution, given in expressions (2.18), (2.21), we find that

$$N_{\rm D3}^{\rm NATD} = n\pi N_{\rm D3}^{\rm ATD}, \qquad k_{\rm D5}^{\rm NATD} = n\pi k_{\rm D5}^{\rm ATD}.$$
 (A.6)

This same rescaling was found in [44] in the matching between the Abelian and non-Abelian T-dual AdS_5 spaces studied in that paper. As discussed there, the $n\pi$ factor can again be safely absorbed through a redefinition of Newton's constant. We give further details of the general relationship between T-dual and non-Abelian T-dual solutions in appendix B.

A simple generalisation is to allow $0 < r < k'\pi$, which is equivalent to taking the Tdual of the IIA reduction of the $AdS_4 \times S^7/(\mathbb{Z}_k \times \mathbb{Z}_{k'})$ orbifold. In that case $N_{\rm NS5} = k'$. The solution described in [59] corresponds to this situation. Its CFT dual consists of a circular quiver associated to a set of $N_{\rm D3}$ D3-branes, with $N_{\rm D3}$ as in (A.5), stretched between k'NS5-branes, as illustrated in figure 8. These D3-branes are thus winding D3-branes. At each interval of length π there are also $k_{\rm D5}$ D5-branes. The field theory associated to this brane configuration was studied in [59] and denoted as $C_{\rho}^{\hat{\rho}}(\mathrm{SU}(N), L)$, with the positive integer L referring to the number of winding D3-branes. These theories degenerate to the $T_{\rho}^{\hat{\rho}}(\mathrm{SU}(N))$ theories of [63] when L = 0. In the next subsection we illustrate the connection between the solution in [59] for k' = 1 and the Abelian T-dual solution under discussion. The value k' = 1 corresponds to the limiting case of $N_{\rm D3}$ D3-branes stretched between two NS5-branes that are identified.

A.2 Connection with ABEG geometries

As in the case of the non-Abelian T-dual solution, we expect that the Abelian T-dual, which also preserves $\mathcal{N} = 4$ SUSY, contains an AdS_4 factor and has SO(4) isometry, fits within the formalism described in section 3.2. Indeed, from eqs. (A.1), (A.2) we can read
off the values

$$\lambda_1^2 = \frac{\sqrt{(1-\sigma)}(1+\sigma)}{\sqrt{2}\beta} \quad \lambda_2^2 = \frac{(1-\sigma)^{3/2}}{\sqrt{2}\beta} \quad \lambda^2 = \frac{\sqrt{2}\sqrt{1-\sigma}}{\beta}, \quad \frac{1}{\tilde{\rho}^2} = 2\sqrt{2}\beta\sqrt{1-\sigma}(1+\sigma),$$

$$b_1 = r - n\pi, \quad b_2 = \frac{k}{2}r + c_0, \quad z = \sigma + i\beta r, \quad e^{2\Phi} = 4\frac{1-\sigma}{k^2(1+\sigma)}, \tag{A.7}$$

where $d(c_0) = 0$ and β and σ are defined as in section 5. In terms of the classification we find that the T-dual solution is given by

$$N_{1} = \frac{k^{3}}{128\beta^{4}}(1+\sigma), \qquad N_{2} = \frac{k}{32\beta^{4}}(1-\sigma), \qquad W = -\frac{k}{16\beta^{2}}, \qquad X = 0, \ (A.8)$$
$$h_{1} = \frac{k}{4\beta}(1+\sigma), \qquad h_{2} = \frac{1}{2\beta}(1-\sigma), \qquad h_{1}^{D} = -\frac{1}{2}\left(c_{0} + \frac{k}{2}r\right), \qquad h_{2}^{D} = \frac{1}{2}r.$$

As discussed in [59], this solution does not fit however within the Ansatz of [58]. Clearly, even if there are NS5 and D5 branes located at $\sigma = \pm 1$, which could be taken as boundaries of an infinite strip, the branes are smeared in r by construction, and h_1 , h_2 do not exhibit logarithmic singularities at the locations of the branes. The authors of [58] showed in [59] how to solve this problem. They considered a distribution $(\delta_a, \hat{\delta}_b)$ of 5-branes that is repeated infinitely-many times along the strip with a period 2t, such that $\delta_{a+p} - \delta_a = \hat{\delta}_{b+\hat{p}} - \hat{\delta}_b = 2t$, where p and \hat{p} are, correspondingly, the total numbers of D5 and NS5-branes stacks. The resulting h_1 , h_2 are, by construction, periodic under $z \to z + 2t$. This allows for the identification of points separated by the period 2t, thus turning the strip into an annulus (and thus the linear quiver into a circular quiver) in the $e^{i\pi z/t}$ plane, with NS5 (D5) brane stacks along the inner (outer) boundaries. The smearing of the branes comes out as a result of taking the limit $t \to 0$ combined with a far-from-the-boundaries approximation. h_1 and h_2 become then independent of r and non-singular.

The introduction of the period 2t on the gravity side induces the winding D3-branes on the dual quiver, in a way that we specify below. These branes do not end on the 5-branes and therefore do not contribute to the linking numbers. The corresponding circular quiver is then characterised by two partitions ρ and $\hat{\rho}$, together with the number of winding D3branes. The $t \to 0$ limit that yields the Abelian T-dual solution corresponds to a large number of these winding D3's, which are then identified with the N_{D3} in (A.5). In this approximation the number of D3-branes ending on 5-branes is negligible, and the brane picture depicted in figure 8 arises.⁶

For the sake of transparency, let us finally show that the IIB NS-sector derived in [59] for k' = 1 NS5-branes matches our Abelian T-dual solution.⁷ The Einstein frame metric and dilaton in [59] are

$$ds_{\text{IIB}}^2 = R^2 g(y)^{1/4} \left[ds_{AdS_4}^2 + y ds_{S_1^2}^2 + (1-y) ds_{S_2^2}^2 \right] + R^2 g(y)^{-3/4} \left[\frac{4t^2}{\pi^4} dx^2 + dy^2 \right] ,$$

$$e^{2\phi'} = \frac{k'}{k} \sqrt{\frac{1-y}{y}} , \qquad (A.9)$$

⁶Indeed, in this approximation the period 2t is simply related to N_{D3} as $N_{\text{D3}} = (\pi^2 k k')/(32t^2)$.

⁷Up to a scaling factor and an S-duality transformation, given e.g. in (6.1) of [59] for c = b = 0 and a = d = -1. Note also that [59] uses a non-standard form of the dilaton, $\phi' \equiv \Phi/2$.

where the AdS_4 space is taken to be of unit radius, $R^4 = \pi^4 k k'/t^2$ and g(y) = y(1-y). If expressed in string frame by multiplying with $e^{\phi'} = e^{\Phi/2}$ and allowing for the coordinate change

$$y = \sin^2\left(\frac{\mu}{2}\right), \qquad x = \frac{1}{4}r, \qquad (A.10)$$

the solution in equation (A.1) is reproduced for k' = 1, for which $r \in [0, \pi]$. It can also be easily checked that with this coordinate change the RR-sector in (A.2) corresponds to the one of [59], up to a gauge transformation.

A.3 Free energy

Using the results of the previous subsection, we can compute the free energy of the Abelian T-dual solution from W, h_1 and h_2 using expressions (6.1) and (6.2) (see [59]). Taking the differential area of the strip $d^2x = \beta dr d\sigma$ and integrating in $r \in [0, \pi]$, $\sigma \in [-1, 1]$, we find

$$S_{\text{eff}} = \frac{k^2}{3\pi^2\beta^3},\tag{A.11}$$

and, using the conserved charges in (A.5),

$$S_{\rm eff} = \frac{\sqrt{2}\pi}{3} \sqrt{k_{\rm D5}} N_{\rm D3}^{3/2}.$$
 (A.12)

It can easily be checked that this is the free energy of the IIA reduction of the $AdS_4 \times S^7/\mathbb{Z}_k$ orbifold, with $N_{D2} \to N_{D3}$ and $k \to k_{D5}$. It is shown in the main text that it agrees with the free energy of the non-Abelian T-dual solution in the $r \in [n\pi, (n+1)\pi]$ interval and $n \to \infty$.

B Relating Abelian and non-Abelian T-duality

In the previous appendix we discussed the relationship between the Abelian and non-Abelian T-dual AdS_4 spaces studied in this paper. In this appendix we complete this analysis and elucidate a general relationship between the geometries generated by acting on a round S^3 with Hopf fibre T-duality and SU(2) non-Abelian T-duality.

Consider a type II supergravity solution with global SO(4) isometry and NS sector that can be written as

$$ds^{2} = ds^{2}(\mathcal{M}_{7}) + 4e^{2C}ds^{2}(S^{3}), \qquad B = 0, \qquad e^{\Phi} = e^{\Phi_{0}}$$
(B.1)

where x are coordinates on \mathcal{M}_7 only. Non-Abelian T-duality acting on such solutions was considered at length in [18–23]. It will be useful to parametrise the 3-sphere in two different ways, making manifest the two dualisation isometries

$$4ds^{2}(S_{\mathrm{U}(1)}^{3}) = d\theta^{2} + \sin^{2}\theta d\phi^{2} + (d\psi + \cos\theta d\phi)^{2}, \qquad 4ds^{2}(S_{\mathrm{SU}(2)}^{3}) = \left(\omega_{1}^{2} + \omega_{2}^{2} + \omega_{3}^{3}\right),$$
(B.2)

where ω_i are SU(2) left invariant 1-forms. The first of these is suitable for T-duality on ψ which, following [35], results in the dual NS sector

$$ds_{\text{ATD}}^2 = ds^2(\mathcal{M}_7) + e^{-2C}dr^2 + e^{2C}ds^2(S^2),$$

$$B_2^{\text{ATD}} = r \operatorname{Vol}(S^2), \qquad e^{-\Phi_{\text{ATD}}} = e^{C-\Phi_0}, \qquad (B.3)$$

where we have performed a gauge transformation on B_2 to put it in this form, and S^2 is the unit norm 2-sphere spanned by θ, ϕ . The second sphere parametrisation is suitable for SU(2) non-Abelian T-duality and leads to the dual NS sector

$$ds_{\text{NATD}}^{2} = ds^{2}(\mathcal{M}_{7}) + e^{-2C}dr^{2} + \frac{e^{2C}r^{2}}{r^{2} + e^{4C}}ds^{2}(S^{2}),$$

$$B_{2}^{\text{NATD}} = \frac{r^{3}}{r^{2} + e^{4C}}\operatorname{Vol}(S^{2}), \qquad e^{-\Phi_{\text{NATD}}} = \sqrt{r^{2} + e^{4C}}e^{C-\Phi_{0}}.$$
(B.4)

Comparing eqs (B.3) and (B.4) one finds they obey the relation

$$\lim_{r \to \infty} ds_{\text{NATD}}^2 = ds_{\text{ATD}}^2, \qquad \lim_{r \to \infty} B_2^{\text{NATD}} = B_2^{\text{ATD}}, \qquad \lim_{r \to \infty} e^{-\Phi_{\text{NATD}}} = re^{-\Phi_{\text{ATD}}}.$$
(B.5)

This has been observed in the previous appendix and before, for instance in [44], but what has not been addressed is whether such a relation holds also for the RR fluxes. We now address this by considering the massive IIA fluxes

$$F_0 = m,$$

 $F_2 = G_2,$
 $F_4 = G_4 + 8G_1 \wedge \text{Vol}(S^3),$ (B.6)

however the following statements also hold when transforming from type IIB to IIA. Performing T-duality on the Hopf fibre as before leads to the dual fluxes

$$F_1^{\text{ATD}} = -mdr,$$

$$F_3^{\text{ATD}} = -dr \wedge G_2 - G_1 \wedge \text{Vol}(S^2)$$

$$F_5^{\text{ATD}} = -dr \wedge G_4 + e^{3C} \star_7 G_4 \wedge \text{Vol}(S^2)$$
(B.7)

while performing non-Abelian T-duality on the whole S^3 leads to

$$F_{1}^{\text{NATD}} = -G_{1} - mrdr,$$

$$F_{3}^{\text{NATD}} = e^{3C} \star_{7} G_{4} - rdr \wedge G_{2} - \frac{r^{3}}{r^{2} + e^{4C}} G_{1} \wedge \text{Vol}(S^{2}) + \frac{mr^{2}e^{4C}}{r^{2} + e^{4C}} dr \wedge \text{Vol}(S^{2})$$

$$F_{5}^{\text{NATD}} = -rdr \wedge G_{4} + \frac{r^{2}e^{4C}}{r^{2} + e^{4C}} dr \wedge G_{2} \wedge \text{Vol}(S^{2}) + \frac{r^{3}e^{3C}}{r^{2} + e^{4C}} \star_{7} G_{4} \wedge \text{Vol}(S^{2}) - e^{3C} \star_{7} G_{2}.$$
 (B.8)

Comparing eqs (B.7) and (B.8), one sees that there is indeed a relation between the flux polyforms, namely

$$\partial_r (\lim_{r \to \infty} F^{\text{NATD}}) = F^{\text{ATD}},$$
 (B.9)

which $e^{-\Phi_{\text{NATD}}}$ clearly also obeys. Notice that we can dispense with the derivative by weighting the flux polyform by the dilaton, namely

$$\lim_{r \to \infty} e^{\Phi_{\text{NATD}}} F^{\text{NATD}} = e^{\Phi_{\text{ATD}}} F^{\text{ATD}} .$$
(B.10)

That this holds is actually not so surprising. As shown in [17, 68], under T-duality the fluxes transform in the combination $e^{\Phi}F$. Specifically the fluxes and MW Killing spinors are transformed by the same matrix Ω as

$$\epsilon_1 = \epsilon_1^0,$$

$$\epsilon_2 = \Omega \epsilon_2^0,$$

$$e^{\Phi} F = e^{\Phi^0} F^0 \Omega^{-1},$$
(B.11)

where 0 hat denotes the seed solution. For SU(2) non-Abelian T-duality performed on a round 3-sphere there exists a frame in which

$$\Omega^{\text{NATD}} = \frac{1}{\sqrt{r^2 + e^{4C}}} \left(\Gamma_{r12} + r \Gamma_r \right) \tag{B.12}$$

where the flat directions 1, 2 span $e^{2C}S^2$ in the non-Abelian T-dual. Clearly

$$\lim_{r \to \infty} \Omega^{\text{NATD}} = \lim_{r \to \infty} \left(\Omega^{\text{NATD}} \right)^{-1} = \Gamma_r$$
(B.13)

which we recognise as Ω^{ATD} . This means that

$$\lim_{r \to \infty} e^{\Phi^0} F^0 (\Omega^{\text{NATD}})^{-1} = e^{\Phi^0} F^0 (\Omega^{\text{ATD}})^{-1}$$
(B.14)

and so eq (B.10) just reconciles this with the final expression in eq (B.11).

To conclude, we have observed that the Hopf fibre T-dual is related to the non-Abelian T-dual as

$$\lim_{r \to \infty} \begin{pmatrix} ds^2 \\ B_2 \\ e^{\Phi}F \\ \epsilon_{1,2} \end{pmatrix}_{\text{NATD}} = \begin{pmatrix} ds^2 \\ B_2 \\ e^{\Phi}F \\ \epsilon_{1,2} \end{pmatrix}_{\text{ATD}}$$
(B.15)

while the dilaton is related as

$$\lim_{r \to \infty} e^{-\Phi_{\text{NATD}}} = r e^{-\Phi_{\text{ATD}}}.$$
 (B.16)

As discussed below eq (2.15), it is easy to understand the r appearing in the dilaton at the level of the string frame supergravity actions, where this factor precisely cancels the change in the volume of the T-dual submanifold in the NS sector. In the RR sector, it is the combination $e^{\Phi}F$ that absorbs the volume change. The $r \to \infty$ limit of the non-Abelian T-dual thus reproduces the Abelian T-dual.

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- C. Montonen and D.I. Olive, Magnetic Monopoles as Gauge Particles?, Phys. Lett. B 72 (1977) 117 [INSPIRE].
- [2] N. Seiberg and E. Witten, Electric-magnetic duality, monopole condensation and confinement in N = 2 supersymmetric Yang-Mills theory, Nucl. Phys. B 426 (1994) 19
 [Erratum ibid. B 430 (1994) 485] [hep-th/9407087] [INSPIRE].
- [3] N. Seiberg and E. Witten, Monopoles, duality and chiral symmetry breaking in N = 2 supersymmetric QCD, Nucl. Phys. B 431 (1994) 484 [hep-th/9408099] [INSPIRE].
- [4] N. Seiberg, Electric-magnetic duality in supersymmetric non-Abelian gauge theories, Nucl. Phys. B 435 (1995) 129 [hep-th/9411149] [INSPIRE].
- [5] A. Sen, An Introduction to nonperturbative string theory, in proceedings of A Newton Institute Euroconference on Duality and Supersymmetric Theories, Cambridge, England, April 7–18 1997, D.I. Olive and P.C. West eds., Cambridge University Press, Cambridge U.K. (1999), pp. 297–413 [hep-th/9802051] [INSPIRE].
- [6] N.A. Obers and B. Pioline, U duality and M-theory, Phys. Rept. 318 (1999) 113
 [hep-th/9809039] [INSPIRE].
- J.M. Maldacena, The Large-N limit of superconformal field theories and supergravity, Int. J. Theor. Phys. 38 (1999) 1113 [Adv. Theor. Math. Phys. 2 (1998) 231] [hep-th/9711200]
 [INSPIRE].
- [8] S.S. Gubser, I.R. Klebanov and A.M. Polyakov, Gauge theory correlators from noncritical string theory, Phys. Lett. B 428 (1998) 105 [hep-th/9802109] [INSPIRE].
- [9] E. Witten, Anti-de Sitter space and holography, Adv. Theor. Math. Phys. 2 (1998) 253 [hep-th/9802150] [INSPIRE].
- [10] H.A. Kramers and G.H. Wannier, Statistics of the two-dimensional ferromagnet. Part I, Phys. Rev. 60 (1941) 252 [INSPIRE].
- S.R. Coleman, The Quantum sine-Gordon Equation as the Massive Thirring Model, Phys. Rev. D 11 (1975) 2088 [INSPIRE].
- [12] S. Mandelstam, Soliton Operators for the Quantized sine-Gordon Equation, Phys. Rev. D 11 (1975) 3026 [INSPIRE].
- T.H. Buscher, Path Integral Derivation of Quantum Duality in Nonlinear σ-models, Phys. Lett. B 201 (1988) 466 [INSPIRE].
- [14] T.H. Buscher, A Symmetry of the String Background Field Equations, Phys. Lett. B 194 (1987) 59 [INSPIRE].
- [15] M. Roček and E.P. Verlinde, Duality, quotients and currents, Nucl. Phys. B 373 (1992) 630 [hep-th/9110053] [INSPIRE].
- [16] X.C. de la Ossa and F. Quevedo, Duality symmetries from non-Abelian isometries in string theory, Nucl. Phys. B 403 (1993) 377 [hep-th/9210021] [INSPIRE].
- [17] K. Sfetsos and D.C. Thompson, On non-abelian T-dual geometries with Ramond fluxes, Nucl. Phys. B 846 (2011) 21 [arXiv:1012.1320] [INSPIRE].
- [18] Y. Lozano, E. Ó Colgáin, K. Sfetsos and D.C. Thompson, Non-abelian T-duality, Ramond Fields and Coset Geometries, JHEP 06 (2011) 106 [arXiv:1104.5196] [INSPIRE].

- [19] G. Itsios, Y. Lozano, E. Ó Colgáin and K. Sfetsos, Non-Abelian T-duality and consistent truncations in type-II supergravity, JHEP 08 (2012) 132 [arXiv:1205.2274] [INSPIRE].
- [20] G. Itsios, C. Núñez, K. Sfetsos and D.C. Thompson, Non-Abelian T-duality and the AdS/CFT correspondence: new N = 1 backgrounds, Nucl. Phys. B 873 (2013) 1 [arXiv:1301.6755] [INSPIRE].
- [21] Y. Lozano, E. Ó Colgáin, D. Rodríguez-Gómez and K. Sfetsos, Supersymmetric AdS₆ via T Duality, Phys. Rev. Lett. 110 (2013) 231601 [arXiv:1212.1043] [INSPIRE].
- [22] G. Itsios, C. Núñez, K. Sfetsos and D.C. Thompson, On Non-Abelian T-duality and new N = 1 backgrounds, Phys. Lett. B 721 (2013) 342 [arXiv:1212.4840] [INSPIRE].
- [23] J. Jeong, O. Kelekci and E. Ó Colgáin, An alternative IIB embedding of F(4) gauged supergravity, JHEP 05 (2013) 079 [arXiv:1302.2105] [INSPIRE].
- [24] G. Itsios, Y. Lozano, E. Ó Colgáin and K. Sfetsos, Non-Abelian T-duality and consistent truncations in type-II supergravity, JHEP 08 (2012) 132 [arXiv:1205.2274] [INSPIRE].
- [25] A. Barranco, J. Gaillard, N.T. Macpherson, C. Núñez and D.C. Thompson, G-structures and Flavouring non-Abelian T-duality, JHEP 08 (2013) 018 [arXiv:1305.7229] [INSPIRE].
- [26] N.T. Macpherson, Non-Abelian T-duality, G₂-structure rotation and holographic duals of N = 1 Chern-Simons theories, JHEP 11 (2013) 137 [arXiv:1310.1609] [INSPIRE].
- [27] Y. Lozano, E. O Colgáin and D. Rodríguez-Gómez, Hints of 5d Fixed Point Theories from Non-Abelian T-duality, JHEP 05 (2014) 009 [arXiv:1311.4842] [INSPIRE].
- [28] J. Gaillard, N.T. Macpherson, C. Núñez and D.C. Thompson, Dualising the Baryonic Branch: Dynamic SU(2) and confining backgrounds in IIA, Nucl. Phys. B 884 (2014) 696 [arXiv:1312.4945] [INSPIRE].
- [29] D. Elander, A.F. Faedo, C. Hoyos, D. Mateos and M. Piai, *Multiscale confining dynamics from holographic RG flows*, JHEP 05 (2014) 003 [arXiv:1312.7160] [INSPIRE].
- [30] S. Zacarías, Semiclassical strings and Non-Abelian T-duality, Phys. Lett. B 737 (2014) 90 [arXiv:1401.7618] [INSPIRE].
- [31] E. Caceres, N.T. Macpherson and C. Núñez, New Type IIB Backgrounds and Aspects of Their Field Theory Duals, JHEP 08 (2014) 107 [arXiv:1402.3294] [INSPIRE].
- [32] P.M. Pradhan, Oscillating Strings and Non-Abelian T-dual Klebanov-Witten Background, Phys. Rev. D 90 (2014) 046003 [arXiv:1406.2152] [INSPIRE].
- [33] K. Sfetsos and D.C. Thompson, New N = 1 supersymmetric AdS₅ backgrounds in Type IIA supergravity, JHEP 11 (2014) 006 [arXiv:1408.6545] [INSPIRE].
- [34] Y. Lozano and N.T. Macpherson, A new AdS₄/CFT₃ dual with extended SUSY and a spectral flow, JHEP 11 (2014) 115 [arXiv:1408.0912] [INSPIRE].
- [35] Ö. Kelekci, Y. Lozano, N.T. Macpherson and E. Ó Colgáin, Supersymmetry and non-Abelian T-duality in type-II supergravity, Class. Quant. Grav. 32 (2015) 035014 [arXiv:1409.7406]
 [INSPIRE].
- [36] N.T. Macpherson, C. Núñez, L.A. Pando Zayas, V.G.J. Rodgers and C.A. Whiting, *Type IIB supergravity solutions with AdS₅ from Abelian and non-Abelian T dualities*, *JHEP* 02 (2015) 040 [arXiv:1410.2650] [INSPIRE].
- [37] K.S. Kooner and S. Zacarías, Non-Abelian T-Dualizing the Resolved Conifold with Regular and Fractional D3-branes, JHEP 08 (2015) 143 [arXiv:1411.7433] [INSPIRE].

- [38] T.R. Araujo and H. Nastase, $\mathcal{N} = 1$ SUSY backgrounds with an AdS factor from non-Abelian T duality, Phys. Rev. **D** 91 (2015) 126015 [arXiv:1503.00553] [INSPIRE].
- [39] Y. Lozano, N.T. Macpherson, J. Montero and E. Ó Colgáin, New $AdS_3 \times S^2$ T-duals with $\mathcal{N} = (0, 4)$ supersymmetry, JHEP 08 (2015) 121 [arXiv:1507.02659] [INSPIRE].
- [40] Y. Lozano, N.T. Macpherson and J. Montero, A N = 2 supersymmetric AdS₄ solution in M-theory with purely magnetic flux, JHEP 10 (2015) 004 [arXiv:1507.02660] [INSPIRE].
- [41] T.R. Araujo and H. Nastase, Non-Abelian T-duality for nonrelativistic holographic duals, JHEP 11 (2015) 203 [arXiv:1508.06568] [INSPIRE].
- [42] L.A. Pando Zayas, V.G.J. Rodgers and C.A. Whiting, Supergravity solutions with AdS₄ from non-Abelian T-dualities, JHEP 02 (2016) 061 [arXiv:1511.05991] [INSPIRE].
- [43] N.T. Macpherson, C. Núñez, D.C. Thompson and S. Zacarías, Holographic Flows in non-Abelian T-dual Geometries, JHEP 11 (2015) 212 [arXiv:1509.04286] [INSPIRE].
- [44] Y. Lozano and C. Núñez, Field theory aspects of non-Abelian T-duality and N = 2 linear quivers, JHEP 05 (2016) 107 [arXiv:1603.04440] [INSPIRE].
- [45] F. Apruzzi, M. Fazzi, A. Passias, D. Rosa and A. Tomasiello, AdS₆ solutions of type-II supergravity, JHEP 11 (2014) 099 [Erratum ibid. 05 (2015) 012] [arXiv:1406.0852] [INSPIRE].
- [46] H. Kim, N. Kim and M. Suh, Supersymmetric AdS₆ Solutions of Type IIB Supergravity, Eur. Phys. J. C 75 (2015) 484 [arXiv:1506.05480] [INSPIRE].
- [47] Ö. Kelekci, Y. Lozano, J. Montero, E. Ó Colgáin and M. Park, Large superconformal near-horizons from M-theory, Phys. Rev. D 93 (2016) 086010 [arXiv:1602.02802] [INSPIRE].
- [48] E. D'Hoker, M. Gutperle, A. Karch and C.F. Uhlemann, Warped AdS₆ × S² in Type IIB supergravity I: Local solutions, JHEP 08 (2016) 046 [arXiv:1606.01254] [INSPIRE].
- [49] C. Couzens, Supersymmetric AdS₅ solutions of type IIB supergravity without D3 branes, arXiv:1609.05039 [INSPIRE].
- [50] E. Alvarez, L. Álvarez-Gaumé, J.L.F. Barbon and Y. Lozano, Some global aspects of duality in string theory, Nucl. Phys. B 415 (1994) 71 [hep-th/9309039] [INSPIRE].
- [51] E. Alvarez, L. Álvarez-Gaumé and Y. Lozano, On non-Abelian duality, Nucl. Phys. B 424 (1994) 155 [hep-th/9403155] [INSPIRE].
- Y. Lozano, Non-Abelian duality and canonical transformations, Phys. Lett. B 355 (1995) 165 [hep-th/9503045] [INSPIRE].
- [53] A. Giveon and M. Roček, On non-Abelian duality, Nucl. Phys. B 421 (1994) 173
 [hep-th/9308154] [INSPIRE].
- [54] S. Elitzur, A. Giveon, E. Rabinovici, A. Schwimmer and G. Veneziano, *Remarks on non-Abelian duality*, *Nucl. Phys.* B 435 (1995) 147 [hep-th/9409011] [INSPIRE].
- [55] C. Klimčík and P. Ševera, Dual non-Abelian duality and the Drinfeld double, Phys. Lett. B 351 (1995) 455 [hep-th/9502122] [INSPIRE].
- [56] D. Gaiotto and A. Tomasiello, Holography for (1,0) theories in six dimensions, JHEP 12 (2014) 003 [arXiv:1404.0711] [INSPIRE].
- [57] S. Cremonesi and A. Tomasiello, 6d holographic anomaly match as a continuum limit, JHEP 05 (2016) 031 [arXiv:1512.02225] [INSPIRE].

- [58] B. Assel, C. Bachas, J. Estes and J. Gomis, Holographic Duals of D = 3 $\mathcal{N} = 4$ Superconformal Field Theories, JHEP 08 (2011) 087 [arXiv:1106.4253] [INSPIRE].
- [59] B. Assel, C. Bachas, J. Estes and J. Gomis, *IIB Duals of* D = 3 $\mathcal{N} = 4$ *Circular Quivers*, *JHEP* **12** (2012) 044 [arXiv:1210.2590] [INSPIRE].
- [60] O. Aharony, L. Berdichevsky, M. Berkooz and I. Shamir, Near-horizon solutions for D3-branes ending on 5-branes, Phys. Rev. D 84 (2011) 126003 [arXiv:1106.1870] [INSPIRE].
- [61] E. D'Hoker, J. Estes and M. Gutperle, Exact half-BPS Type IIB interface solutions. II. Flux solutions and multi-Janus, JHEP 06 (2007) 022 [arXiv:0705.0024] [INSPIRE].
- [62] E. D'Hoker, J. Estes and M. Gutperle, Exact half-BPS Type IIB interface solutions. I. Local solution and supersymmetric Janus, JHEP 06 (2007) 021 [arXiv:0705.0022] [INSPIRE].
- [63] D. Gaiotto and E. Witten, S-duality of Boundary Conditions In N = 4 Super Yang-Mills Theory, Adv. Theor. Math. Phys. 13 (2009) 721 [arXiv:0807.3720] [INSPIRE].
- [64] A. Hanany and E. Witten, Type IIB superstrings, BPS monopoles and three-dimensional gauge dynamics, Nucl. Phys. B 492 (1997) 152 [hep-th/9611230] [INSPIRE].
- [65] D. Gaiotto and J.M. Maldacena, The Gravity duals of N = 2 superconformal field theories, JHEP 10 (2012) 189 [arXiv:0904.4466] [INSPIRE].
- [66] R.A. Reid-Edwards and B. Stefański Jr., On Type IIA geometries dual to N = 2 SCFTs, Nucl. Phys. B 849 (2011) 549 [arXiv:1011.0216] [INSPIRE].
- [67] O. Aharony, L. Berdichevsky and M. Berkooz, 4d N = 2 superconformal linear quivers with type IIA duals, JHEP 08 (2012) 131 [arXiv:1206.5916] [INSPIRE].
- [68] S.F. Hassan, T duality, space-time spinors and RR fields in curved backgrounds, Nucl. Phys. B 568 (2000) 145 [hep-th/9907152] [INSPIRE].
- [69] B. Assel, J. Estes and M. Yamazaki, Large-N Free Energy of 3d N = 4 SCFTs and AdS_4/CFT_3 , JHEP **09** (2012) 074 [arXiv:1206.2920] [INSPIRE].
- [70] S. Benvenuti and S. Pasquetti, 3D-partition functions on the sphere: exact evaluation and mirror symmetry, JHEP 05 (2012) 099 [arXiv:1105.2551] [INSPIRE].
- [71] T. Nishioka, Y. Tachikawa and M. Yamazaki, 3d Partition Function as Overlap of Wavefunctions, JHEP 08 (2011) 003 [arXiv:1105.4390] [INSPIRE].
- [72] B. Assel and J. Gomis, Mirror Symmetry And Loop Operators, JHEP 11 (2015) 055 [arXiv:1506.01718] [INSPIRE].
- [73] S. Cremonesi, A. Hanany, N. Mekareeya and A. Zaffaroni, T^σ_ρ(G) theories and their Hilbert series, JHEP 01 (2015) 150 [arXiv:1410.1548] [INSPIRE].
- [74] S. Cremonesi, A. Hanany and A. Zaffaroni, Monopole operators and Hilbert series of Coulomb branches of $3d \ \mathcal{N} = 4$ gauge theories, JHEP **01** (2014) 005 [arXiv:1309.2657] [INSPIRE].
- [75] A. Hanany and N. Mekareeya, Complete Intersection Moduli Spaces in N = 4 Gauge Theories in Three Dimensions, JHEP 01 (2012) 079 [arXiv:1110.6203] [INSPIRE].
- [76] B. Hoare and A.A. Tseytlin, Homogeneous Yang-Baxter deformations as non-abelian duals of the AdS₅ σ-model, arXiv:1609.02550 [INSPIRE].

3.5 The AdS_5 non-Abelian T-dual of Klebanov-Witten as a $\mathcal{N} = 1$ linear quiver from M5branes



Published for SISSA by 🖄 Springer

RECEIVED: June 23, 2017 REVISED: August 14, 2017 ACCEPTED: August 19, 2017 PUBLISHED: September 11, 2017

The AdS_5 non-Abelian T-dual of Klebanov-Witten as a $\mathcal{N} = 1$ linear quiver from M5-branes

Georgios Itsios,^a Yolanda Lozano,^b Jesús Montero^b and Carlos Núñez^c

^cDepartment of Physics, Swansea University,

Swansea SA2 8PP, United Kingdom

E-mail: gitsios@gmail.com, ylozano@uniovi.es, jesus.montero.a@gmail.com, c.nunez@swansea.ac.uk

ABSTRACT: In this paper we study an AdS_5 solution constructed using non-Abelian Tduality, acting on the Klebanov-Witten background. We show that this is dual to a linear quiver with two tails of gauge groups of increasing rank. The field theory dynamics arises from a D4-NS5-NS5' brane set-up, generalizing the constructions discussed by Bah and Bobev. These realize $\mathcal{N} = 1$ quiver gauge theories built out of $\mathcal{N} = 1$ and $\mathcal{N} = 2$ vector multiplets flowing to interacting fixed points in the infrared. We compute the central charge using *a*-maximization, and show its precise agreement with the holographic calculation. Our result exhibits n^3 scaling with the number of five-branes. This suggests an eleven-dimensional interpretation in terms of M5-branes, a generic feature of various AdS backgrounds obtained via non-Abelian T-duality.

KEYWORDS: String Duality, AdS-CFT Correspondence, Conformal Field Models in String Theory

ARXIV EPRINT: 1705.09661

OPEN ACCESS, C The Authors. Article funded by SCOAP³.

^aInstituto de Física Teórica, UNESP-Universidade Estadual Paulista,

R. Dr. Bento T. Ferraz 271, Bl. II, Sao Paulo 01140-070, SP, Brazil

^bDepartment of Physics, University of Oviedo,

Avda. Calvo Sotelo 18, 33007 Oviedo, Spain

Contents

1	Introduction	2
	1.1 General framework and organization of this paper	3
2	The non-Abelian T-dual of the Klebanov-Witten solution	5
	2.1 The $AdS_5 \times T^{1,1}$ solution	5
	2.2 The non-Abelian T-dual solution	6
	2.2.1 Asymptotics	8
	2.2.2 Quantized charges	9
	2.2.3 Central charge	11
3	Basics of Bah-Bobev 4d $\mathcal{N} = 1$ theories	12
	3.1 $\mathcal{N} = 1$ linear quivers	12
	3.2 IIA brane realization and M-theory uplift	14
4	The non-Abelian T-dual of Klebanov-Witten as a $\mathcal{N}=1$ linear quiver	15
	4.1 Proposed $\mathcal{N} = 1$ linear quiver	16
	4.2 β -functions and R-symmetry anomalies	17
	4.3 Field-theoretical central charge	19
5	Solving the INST-BBBW puzzle	22
6	Conclusions and future directions	2 4
Α	Connection with the GMSW classification	26
	A.1 Uplift of the non-Abelian T-dual solution	26
	A.2 Review of GMSW	27
	A.3 Recovering the non-Abelian T-dual from GMSW	28
В	The Abelian T-dual of the Klebanov-Witten solution	29
	B.1 Background	29
	B.2 Quantized charges and brane set-up	31
С	Some field theory elaborations	32
	C.1 A summary of the Klebanov-Witten CFT	32
	C.2. Control Charge of the $\mathcal{N} = 2$ UV CET	33
	0.2 Central Charge of the $N = 2.0$ V CF 1	00

1 Introduction

Non-Abelian T-duality [1], the generalization of the Abelian T-duality symmetry of String Theory to non-Abelian isometry groups, is a transformation between world-sheet field theories known since the nineties. Its extension to all orders in g_s and α' remains however a technically-hard open problem [2–8]. As a result, non-Abelian T-duality does not stand as a String Theory duality symmetry, as its Abelian counterpart does.

In the paper [9], Sfetsos and Thompson reignited the interest in this transformation by highlighting its potential powerful applications as a solution generating technique in supergravity. An interesting synergy between Holography (the Maldacena conjecture) [10] and non-Abelian T-duality was also pointed out. This connection was further exploited in [11]–[37]. These works have widely applied non-Abelian T-duality to generate new AdS backgrounds of relevance in different contexts. While some of the new solutions avoid previously existing classifications [11, 28, 31, 32], which has led to generalizations of existing families [38–41], some others provide the only known explicit solutions belonging to a given family [32, 33], which can be used to test certain conjectures, such as 3d-3d duality [42, 43]. Some of these works also put forward some ideas to define the associated holographic duals. Nevertheless, these initial attempts always encountered some technical or conceptual puzzle, rendering these proposals only partially satisfactory.

It was in the papers [44–46], where the field theoretical interpretation of non-Abelian T-duality (in the context of Holography) was first addressed in detail. One outcome of these works is that non-Abelian T-duality changes the dual field theory. In other words, that new AdS backgrounds generated through non-Abelian T-duality have dual CFTs different from those dual to the original backgrounds. This is possible because, contrary to its Abelian counterpart, non-Abelian T-duality has not been proven to be a String Theory symmetry.

The results in [44–46] open up an exciting new way to generate new quantum field theories in the context of Holography. In these examples the dual CFT arises in the low energy limit of a given Dp-NS5 brane intersection. This points to an interesting relation between AdS non-Abelian T-duals and M5-branes, that is confirmed by the n^3 scaling of the central charges.

Reversing the logic, the understanding of the field theoretical realization of non-Abelian T-duality brings in a surprising new way (using Holography!) to extract global information about the new backgrounds. Indeed, as discussed in the various papers [2–8], one of the long-standing open problems of non-Abelian T-duality is that it fails in determining global aspects of the dual background.

The idea proposed in [44] and further elaborated in [45, 46], relies on using the dual field theory to globally define (or *complete*) the background obtained by non-Abelian T-duality. In this way the Sfetsos-Thompson solution [9], constructed acting with non-Abelian Tduality on the $AdS_5 \times S^5$ background, was completed and understood as a superposition of Maldacena-Núñez solutions [47], dual to a four dimensional CFT. This provides a global definition of the background and also smoothes out its singularity. This idea was also put to work explicitly in [45] in the context of $\mathcal{N} = 4 \ AdS_4$ solutions. In this case the non-Abelian T-dual solution was shown to arise as a patch of a geometry discussed in [48–51], dual to the renormalization fixed point of a $T^{\hat{\rho}}_{\rho}(\mathrm{SU}(N))$ quiver field theory, belonging to the general class introduced by Gaiotto and Witten in [52].

In the two examples discussed in [44, 45] the non-Abelian T-dual solution arose as the result of zooming-in on a particular region of a completed and well-defined background. Remarkably, this process of zooming-in has recently been identified more precisely as a Penrose limit of a well-known solution. The particular example studied in the paper [46], a background with isometries $\mathbb{R} \times SO(3) \times SO(6)$, was shown to be the Penrose limit of a given Superstar solution [53]. This provides an explicit realization of the ideas in [44] that is clearly applicable in more generality.

In this paper we follow the methods in [44] to propose a CFT interpretation for the $\mathcal{N} = 1 \ AdS_5$ background obtained in [12, 13, 28], by acting with non-Abelian T-duality on a subspace of the Klebanov-Witten solution [54]. We show that, similarly to the examples in [44, 45], the dual CFT is given by a linear quiver with gauge groups of increasing rank. The dynamics of this quiver is shown to emerge from a D4-NS5-NS5' brane construction that generalizes the Type IIA brane set-ups discussed by Bah and Bobev in [55], realizing $\mathcal{N} = 1$ linear quivers built out of $\mathcal{N} = 1$ and $\mathcal{N} = 2$ vector multiplets that flow to interacting fixed points in the infrared. These quivers can be thought of as $\mathcal{N} = 1$ twisted compactifications of the six-dimensional (2,0) theory on a punctured sphere, thus providing a generalization to $\mathcal{N} = 1$ of the $\mathcal{N} = 2$ CFTs discussed in [56].

The results in this paper suggest that the non-Abelian T-dual solution under consideration could provide the first explicit gravity dual to an ordinary $\mathcal{N} = 1$ linear quiver associated to a D4-NS5 brane intersection [55]. In this construction, the $\mathcal{N} = 2$ SUSY D4-NS5 brane set-up associated to the Sfetsos-Thompson solution (see [44]) is reduced to $\mathcal{N} = 1$ SUSY through the addition of extra orthogonal NS5-branes, as in [55]. The quiver that we propose does not involve the T_N theories introduced by Gaiotto [57], and is in contrast with the classes of $\mathcal{N} = 1$ CFTs constructed in [58–61]. We support our proposal with the computation of the central charge associated to the quiver, which is shown to match exactly the holographic result. We also clarify a puzzle posed in [12, 13], where the non-Abelian T-dual background was treated as a solution in the general class constructed in [58, 59], involving the T_N theories, whose corresponding central charge was however in disagreement with the holographic result.

Before describing the plan of this paper, let us put the present work in a wider framework, discussing in some more detail the general ideas behind it.

1.1 General framework and organization of this paper

In the papers [12, 13], the non-Abelian T-dual of the Klebanov-Witten background was constructed. There, it was loosely suggested that the dual field theory could have some relation to the $\mathcal{N} = 1$ version of Gaiotto's CFTs. Indeed, following the ideas in [60], the non-Abelian T-dual of the Klebanov-Witten solution could be thought of as a mass deformation of the non-Abelian T-dual of $AdS_5 \times S^5/\mathbb{Z}_2$, as indicated in the following diagram,

$$\begin{array}{ccc} AdS_5 \times S^5/\mathbb{Z}_2 & \longrightarrow & \text{NATD of } AdS_5 \times S^5/\mathbb{Z}_2 \\ & & & & & \\ & & & & \\ & & & & \\ AdS_5 \times T^{1,1} & \longrightarrow & \text{NATD of } AdS_5 \times T^{1,1}. \end{array}$$

Nevertheless, there were many unknowns and not-understood subtle issues when the papers [12, 13] were written. To begin with, the dual CFT to the non-Abelian T-dual of $AdS_5 \times S^5$ was not known, the holographic central charge of such background was not expressed in a way facilitating the comparison with the CFT result, the important role played by large gauge transformations [19, 25] had not been identified, etc. In hindsight, the papers [12, 13] did open an interesting line of research, but left various uncertainties and loose ends.

This line of investigations evolved to culminate in the works [44–46], that gave a precise dual field theoretical description of different backgrounds obtained by non-Abelian T-duality. This led to a field-theory-inspired *completion* or *regularization* of the non-Abelian T-dual backgrounds. Different checks of this proposal have been performed. Most notably, the central charge is a quantity that nicely matches the field theory calculation with the holographic computation in the completed (regulated) background.

In this paper we will apply the ideas of [44–46] and the field theory methods of [55] to the non-Abelian T-dual of the Klebanov-Witten background. A summary of our results is:

- We perform a study of the background and its quantized charges, and deduce the Hanany-Witten [62] brane set-up, in terms of D4 branes and two types of five-branes NS5 and NS5'.
- We calculate the holographic central charge. This requires a regularization of the background, particularly in one of its coordinates. The regularization we adopt here is a hard-cutoff. Whilst geometrically unsatisfactory, previous experience in [44] shows that this leads to sensible results, easy to compare with a field theoretical calculation.
- Based on the brane set-up, we propose a precise linear quiver field theory. This, we conjecture, is dual to the *regulated* non-Abelian T-dual background. We check that the quiver is at a strongly coupled fixed point by calculating the beta functions and R-symmetry anomalies.
- The quiver that we propose is a generalization of those studied in [55]. It can be thought of as a mass deformation of the $\mathcal{N} = 2$ quiver dual to the non-Abelian Tdual of $AdS_5 \times S^5/\mathbb{Z}_2$, that is constructed following the ideas in [44]. It is the presence of a flavor group in the CFT that *regulates* the space generated by non-Abelian Tduality.
- We calculate the field theoretical central charge applying the methods in [55]. We find precise agreement with the central charge computed holographically for the regulated non-Abelian T-dual solution.

In more detail, the present paper is organized as follows. In section 2, we summarize the main properties of the solution constructed in [12, 13]. We perform a detailed study of the quantized charges, with special attention to the role played by large gauge transformations. Our analysis suggests a D4, NS5, NS5' brane set-up associated to the solution, similar to

that associated to the Abelian T-dual of Klebanov-Witten, studied in [63, 64]. In section 3 we summarize the brane set-up and $\mathcal{N} = 1$ linear quivers of [55], which we use in section 4 for the proposal of a linear quiver that, we conjecture, is dual to the *regulated* version of the non-Abelian T-dual solution of $AdS_5 \times T^{1,1}$. We provide support for our proposal with the detailed computation of the (field theoretical) central charge which we show to be in full agreement with the (regulated) holographic result. We give an interpretation for the field theory dual to our background in terms of a mass deformation of the $\mathcal{N}=2$ CFT associated to the non-Abelian T-dual of $AdS_5 \times S^5/\mathbb{Z}_2$. This suggests the geometrically sensible way of completing our background. Section 5 contains a discussion where we further elaborate on the relation between our proposal and previous results in [12, 13]. We also resolve a puzzle raised there regarding the relation between the non-Abelian T-dual solution and the solutions in [59]. Concluding remarks and future research directions are presented in section 6. Detailed appendices complement our presentation. In appendix A, we explicitly calculate the differential forms showing that the non-Abelian T-dual solution fits in the classification of [65], for $\mathcal{N} = 1$ SUSY spaces with an AdS_5 -factor. Appendix B studies in detail the relation between the non-Abelian T-dual solution and its (Abelian) T-dual counterpart. Finally in appendix C we present some field theory results relevant for the analysis in section 4.

2 The non-Abelian T-dual of the Klebanov-Witten solution

In this section we summarize the Type IIA supergravity solution obtained after a non-Abelian T-duality transformation acts on the $T^{1,1}$ of the Klebanov-Witten background [54]. This solution was first derived in [12, 13]. It was later studied in [28] where a more suitable set of coordinates was used. More general solutions in Type IIA were constructed in [26] as non-Abelian T-duals of $AdS_5 \times Y^{p,q}$ Sasaki-Einstein geometries. Following our paper, the study of their dual CFTs appears to be a natural next step to investigate.

We start by introducing our conventions for the background and by summarizing the calculation of the holographic central charge of the $AdS_5 \times T^{1,1}$ solution.

2.1 The $AdS_5 \times T^{1,1}$ solution

The metric is given by,

$$ds^{2} = ds^{2}_{AdS_{5}} + L^{2} ds^{2}_{T^{1,1}},$$

$$ds^{2}_{AdS_{5}} = \frac{r^{2}}{L^{2}} dx^{2}_{1,3} + \frac{L^{2}}{r^{2}} dr^{2},$$

$$ds^{2}_{T^{1,1}} = \lambda^{2}_{1}(\sigma^{2}_{1} + \sigma^{2}_{2}) + \lambda^{2}_{2}(\sigma^{2}_{1} + \sigma^{2}_{2}) + \lambda^{2}(\sigma_{3} + \cos\theta_{1} d\phi_{1})^{2},$$
(2.1)

where $\lambda^2 = \frac{1}{9}$, $\lambda_1^2 = \lambda_2^2 = \frac{1}{6}$ and

$$\sigma_{\hat{1}} = \sin \theta_1 \, d\phi_1, \qquad \qquad \sigma_{\hat{2}} = d\theta_1, \sigma_1 = \cos \psi \, \sin \theta_2 \, d\phi_2 - \sin \psi \, d\theta_2, \qquad \qquad \sigma_2 = \sin \psi \, \sin \theta_2 \, d\phi_2 + \cos \psi \, d\theta_2, \qquad (2.2) \sigma_3 = d\psi + \cos \theta_2 \, d\phi_2.$$

The background includes a constant dilaton and a self-dual RR five-form,

$$F_5 = \frac{4}{g_s L} \left[\text{Vol}(AdS_5) - L^5 \text{Vol}(T^{1,1}) \right].$$
(2.3)

The associated charge is given by

$$\frac{1}{2\kappa_{10}^2 T_{D3}} \int_{T^{1,1}} F_5 = N_3 \,. \tag{2.4}$$

Using that $2 \kappa_{10}^2 T_{Dp} = (2\pi)^{7-p} g_s {\alpha'}^{\frac{7-p}{2}}$ this leads to a quantization of the size of the space,

$$L^4 = \frac{27}{4} \pi g_s^2 \, \alpha'^2 \, N_3 \,. \tag{2.5}$$

To calculate the holographic central charge of this background, we use the formalism developed in [28, 66]. Indeed, for a generic background and dilaton of the form,

$$ds^{2} = a(r,\theta^{i}) \left[dx_{1,d}^{2} + b(r) dr^{2} \right] + g_{ij}(r,\theta^{i}) d\theta^{i} d\theta^{j}, \quad \Phi(r,\theta^{i}),$$
(2.6)

we define the quantities \hat{V}_{int} , \hat{H} as,

$$\hat{V}_{\text{int}} = \int d\theta^i \sqrt{\det[g_{ij}] e^{-4\Phi} a^d}, \qquad \hat{H} = \hat{V}_{\text{int}}^2.$$
(2.7)

The holographic central charge for the (d+1)-dimensional QFT is calculated as,

$$c = \pi d^{d} \frac{b^{d/2} \hat{H}^{\frac{2d+1}{2}}}{G_{N,10} \left(\hat{H}'\right)^{d}}, \quad G_{N,10} = 8 \pi^{6} g_{s}^{2} \alpha'^{4}.$$
(2.8)

Using these expressions for the background in eq. (2.1), we have

$$a = \frac{r^2}{L^2}, \quad b = \frac{L^4}{r^4}, \quad d = 3, \quad \sqrt{e^{-4\Phi} \,\det[g_{ij}] \,a^3} = g_s^{-2} L^2 r^3 \lambda \,\lambda_1^4 \,\sin\theta_1 \,\sin\theta_2 \,. \tag{2.9}$$

After some algebra, we obtain the well-known result [67],

$$c_{KW} = \pi \frac{L^8}{108\pi^3 g_s^4 \alpha'^4} = \frac{27}{64} N_3^2 \,. \tag{2.10}$$

We now study the action of non-Abelian T-duality on one of the SU(2) isometries displayed by the background in eq. (2.1). We use the notation and conventions in [28].

2.2 The non-Abelian T-dual solution

The NS-NS sector of the non-Abelian T-dual solution constructed in [12, 13, 28] is composed of a metric, a NS-NS two-form and a dilaton. Using the variables in [28], the metric reads,¹

$$d\hat{s}^{2} = \frac{r^{2}}{L^{2}}dx_{1,3}^{2} + \frac{L^{2}}{r^{2}}dr^{2} + L^{2}\lambda_{1}^{2}(d\theta_{1}^{2} + \sin^{2}\theta_{1}d\phi_{1}^{2}) + \frac{L^{2}}{Q} \Big[\lambda_{1}^{4} \left(\cos\chi \,d\rho - \rho\sin\chi \,d\chi\right)^{2} + \lambda^{2}\lambda_{1}^{2}\left(\sin\chi \,d\rho + \rho\cos\chi \,d\chi\right)^{2} + \lambda^{2}\lambda_{1}^{2}\rho^{2}\sin^{2}\chi \left(d\xi + \cos\theta_{1}d\phi_{1}\right)^{2} + \rho^{2}d\rho^{2}\Big].$$
(2.11)

¹Henceforth we use the rescaling $\rho \longrightarrow \frac{L^2}{\alpha'} \rho$ so that all factors in the internal metric scale with L^2 . We also substitute $\lambda_2 = \lambda_1$ for convenience.

The NS two-form is,

$$B_{2} = \frac{L^{2}\rho^{2}\sin\chi}{2Q} \Big[\left(\lambda^{2} - \lambda_{1}^{2}\right)\sin 2\chi \ d\xi \wedge d\rho + 2P \ \rho \ d\xi \wedge d\chi \Big] - \frac{L^{2}\lambda^{2}\cos\theta_{1}}{Q} \Big[\left(\lambda_{1}^{4} + \rho^{2}\right)\cos\chi d\rho \wedge d\phi_{1} - \lambda_{1}^{4} \ \rho\sin\chi \ d\chi \wedge d\phi_{1} \Big],$$

$$(2.12)$$

and the dilaton is given by, 2

$$e^{-2\hat{\Phi}} = \frac{L^6 Q}{g_{\delta}^2 \alpha'^3} . \tag{2.13}$$

For convenience we have defined the following functions,

$$Q = \lambda^2 \lambda_1^4 + \rho^2 P, \qquad P = \lambda^2 \cos^2 \chi + \lambda_1^2 \sin^2 \chi = \lambda_1^2 + (\lambda^2 - \lambda_1^2) \cos^2 \chi.$$
(2.14)

This solution is supported by a set of RR fluxes which read,

$$F_{2} = -\frac{4L^{4}\lambda\lambda_{1}^{4}}{g_{s}\alpha'^{3/2}}\sin\theta_{1}\,d\theta_{1}\wedge d\phi_{1},$$

$$F_{4} = -\frac{2L^{6}\lambda\lambda_{1}^{4}}{g_{s}\alpha'^{3/2}Q}\,\rho^{2}\sin\chi\,\sin\theta_{1}\,d\theta_{1}\wedge d\phi_{1}\wedge\left[\left(\lambda^{2}-\lambda_{1}^{2}\right)\sin2\chi\,d\xi\wedge d\rho+2\,P\,\rho\,d\xi\wedge d\chi\right]$$

$$= B_{2}\wedge F_{2}.$$
(2.15)

The higher rank RR fields which are related to the previous ones through $F_p = (-1)^{[p/2]} \star F_{10-p}$ read,

$$F_{6} = -\frac{4L^{3}}{g_{s} \alpha'^{3/2}} \rho \operatorname{Vol}_{AdS_{5}} \wedge d\rho,$$

$$F_{8} = -\frac{4L^{5}\lambda^{2}\lambda_{1}^{4}}{g_{s} \alpha'^{3/2}Q} \rho^{2} \sin \chi \operatorname{Vol}_{AdS_{5}} \wedge d\rho \wedge d\chi \wedge (d\xi + \cos \theta_{1} d\phi_{1}).$$
(2.16)

The associated RR potentials C_1 and C_3 , defined through the formulas $F_2 = dC_1$ and $F_4 = dC_3 - H_3 \wedge C_1$, are given by,

$$C_{1} = \frac{4L^{4}\lambda\lambda_{1}^{4}}{g_{s}\alpha'^{3/2}}\cos\theta_{1}\,d\phi_{1},$$

$$C_{3} = \frac{2L^{6}\lambda\lambda_{1}^{4}}{g_{s}\alpha'^{3/2}Q}\rho^{2}\cos\theta_{1}\,\sin\chi\left[\left(\lambda_{1}^{2}-\lambda^{2}\right)\sin2\chi\,d\rho\wedge d\xi - 2P\,\rho\,d\chi\wedge d\xi\right]\wedge d\phi_{1} \qquad (2.17)$$

$$= B_{2}\wedge C_{1}.$$

In the papers [12, 13] this solution of the Type IIA equations of motion was shown to preserve $\mathcal{N} = 1$ supersymmetry. In the coordinates used in this paper the Killing vector ∂_{ξ} is dual to the R-symmetry of the CFT.

In appendix A we promote the background in eqs. (2.11)–(2.17) to a solution of elevendimensional supergravity. We show that this background fits in the classification of $\mathcal{N} =$

 $^{^{2}}$ As in the original paper [1], the dilaton needs to transform as well in order to fulfil the equations of motion.

1 AdS_5 solutions in M-theory of [65]. We write in detail the forms satisfying a set of differential relations and define the SU(2)-structure. The eleven dimensional lift suggests that this solution is associated to M5-branes wrapped on a spherical 2d manifold. We discuss this picture further in section 5.

As indicated, one goal of this paper is to propose a conformal field theory dual to the Type IIA non-Abelian T-dual solution. We will do this by combining different insights coming from the large ρ -asymptotics, the quantized charges and the calculation of field theoretical observables using the background.

2.2.1 Asymptotics

In complicated systems, like those corresponding to intersections of branes, it is often illuminating to consider the asymptotic behavior of the background. In the case at hand, for the background in eqs. (2.11)–(2.17), we consider the leading-order behavior of the solution, when $\rho \to \infty$. This allows us to *read* the brane intersection that in the decoupling limit and for a very large number of branes generates the solution.

Indeed, for $\rho \to \infty$, the leading behavior of the NS-fields is

$$ds^{2} \approx ds_{AdS_{5}}^{2} + L^{2} \lambda_{1}^{2} \left[d\Omega_{2}^{2}(\theta_{1},\phi_{1}) + \left(d\chi^{2} + \frac{\lambda^{2} \sin^{2} \chi}{P(\chi)} \left(d\xi + \cos \theta_{1} d\phi_{1} \right)^{2} \right) + \frac{d\rho^{2}}{\lambda_{1}^{2} P(\chi)} \right],$$

$$B_{2} \approx -L^{2} \rho \left[d\Omega_{2}(\chi,\xi) + \frac{\lambda^{2} \cos \chi}{P(\chi)} d\Omega_{2}(\theta_{1},\phi_{1}) - \lambda^{2} \cos \theta_{1} \partial_{\chi} \left(\frac{\cos \chi}{P(\chi)} \right) d\chi \wedge d\phi_{1} \right]$$

$$+ \frac{L^{2} \sin \chi}{2 P(\chi)} \left(\lambda^{2} - \lambda_{1}^{2} \right) \sin 2\chi \, d\xi \wedge d\rho , \qquad (2.18)$$

$$e^{-2\phi} \approx \frac{L^{6}}{g_{s}^{2} \alpha'^{3}} P(\chi) \rho^{2} ,$$

where we have performed a gauge transformation in B_2 , of the form $B_2 + d\Lambda_1$, with

$$\Lambda_1 = L^2 \lambda^2 \rho \cos \theta_1 \left(\frac{\cos \chi}{P(\chi)} \right) d\phi_1 \; .$$

Intuitively, this result suggests that we have two different types of NS-five branes. One type of five-branes (which we refer to as NS) extend along $\mathbb{R}^{1,3} \times S^2(\theta_1, \phi_1)$. The second type of five branes (referred to as NS') extend along $\mathbb{R}^{1,3} \times \tilde{S}^2(\chi, \xi)$. To preserve SUSY, the spaces $S^2(\theta_1, \phi_1)$ and $\tilde{S}^2(\chi, \xi)$ are fibered by the monopole gauge field $A_1 = \cos \theta_1 d\phi_1$. This fibration is also reflected in the B_2 -field, that contains a term that mixes the spheres.

The asymptotics of the RR-fields can be easily read from eq. (2.16). Indeed, the expression $F_6 = dC_5$, generates asymptotically $C_5 \approx \rho r^4 dx_{1,3} \wedge d\rho$. This suggests an array of D4 branes extended along the directions $\mathbb{R}^{1,3} \times \rho$. D6 branes appear due to the presence of the B_2 -field, that blows up the D4 branes due to the Myers effect [68].

In summary, the asymptotic analysis suggests that the background in eqs. (2.11)-(2.17), is generated in the decoupling limit of an intersection of NS5-NS5'-D4 branes. This will be confirmed by the calculation of the quantized charges associated to this solution.

2.2.2 Quantized charges

In the papers [44, 45], the brane set-ups encoding the dynamics of the CFTs dual to the corresponding non-Abelian T-dual backgrounds were proposed after a careful analysis of the quantized charges. The charges that are relevant for the study of the non-Abelian T-dual of the Klebanov-Witten background are those related to D4, D6 and NS5 branes. Based on this analysis we will propose an array of branes, from which the dynamics of a linear quiver with gauge groups of increasing rank will be obtained.

For D6 branes the Page charge reads,

$$Q_{D6} = \frac{1}{2\kappa_{10}^2 T_{D6}} \int_{(\theta_1,\phi_1)} F_2 = \frac{8L^4\lambda\lambda_1^4}{g_s^2 \alpha'^2} = \frac{2}{27} \frac{L^4}{g_s^2 \alpha'^2} = N_6 , \qquad (2.19)$$

where we have absorbed an overall minus sign by choosing an orientation for the integrals. Imposing the quantization of the D6 charge, the AdS radius L is quantized in terms of N_6 ,

$$L^4 = \frac{27}{2} g_s^2 \, \alpha'^2 N_6 \; . \tag{2.20}$$

This relation differs from that for the original background, see eq. (2.5), which is a common feature already observed in the bibliography [19].

In turn, the Page charge associated to D4-branes vanishes,

$$Q_{D4} = \frac{1}{2\kappa_{10}^2 T_{D4}} \int_{M_4} \left(F_4 - F_2 \wedge B_2 \right) = 0 .$$
 (2.21)

This charge becomes however important in the presence of large gauge transformations,

$$B_2 \to B_2 + \Delta B_2 \,, \tag{2.22}$$

under which the Page charges transform as,

$$\Delta Q_{D4} = -\frac{1}{2\kappa_{10}^2 T_{D4}} \int_{M_4} F_2 \wedge \Delta B_2 \,, \qquad \Delta Q_{D6} = 0 \,. \tag{2.23}$$

Indeed, consider a four-manifold $M_4 = [\theta_1, \phi_1] \times \Sigma_2$, with the two-cycle given by $\Sigma_2 = [\chi, \xi]^3$ Under a large gauge transformation of the form,

$$\Delta B_2 = -n \,\pi \,\alpha' \sin \chi \, d\chi \wedge d\xi \,, \qquad d[\Delta B_2] = 0 \,, \tag{2.24}$$

the Page charges transform as

$$\Delta Q_{D4} = n N_6 , \qquad \Delta Q_{D6} = 0 . \tag{2.25}$$

The first relation shows that n units of D4-brane charge are induced in each D6-brane. Conversely, nN_6 D4-branes can expand in the presence of the B_2 field given by eq. (2.24) into N_6 D6-branes wrapped on Σ_2 , through Myers dielectric effect. Consider now the (conveniently normalized) integral of the B_2 field, given by eq. (2.12), along the non-trivial

³Note that this 2-cycle vanishes at $\rho \to 0$, while at $\rho \to \infty$ it is *almost* a two sphere of finite size.

2-cycle $\Sigma_2 = [\chi, \xi]$. Following the paper [25], this must take values in the interval [0, 1].⁴ Imposing this condition implies that $|b_0| \leq 1$ with,

$$b_0 = \frac{1}{4\pi^2 \alpha'} \int_{\Sigma_2} B_2 = -\frac{1}{\pi} \frac{L^2}{\alpha'} \left[\rho - \frac{\sqrt{2}}{6\sqrt{1+54\rho^2}} \tanh^{-1} \left(\frac{3\sqrt{2}\rho}{\sqrt{1+54\rho^2}} \right) \right].$$
(2.26)

The asymptotic behavior of b_0 for small and large values of ρ is given by,

$$b_0 = -\frac{48L^2\rho^3}{\pi\alpha'} + \mathcal{O}(\rho^5), \qquad \rho \ll 1,$$

$$b_0 = -\frac{L^2}{\alpha'}\frac{\rho}{\pi} + \mathcal{O}\left(\frac{1}{\rho}\right), \qquad \rho \gg 1.$$
(2.27)

The expression given by eq. (2.26) is monotonically increasing for all $\rho \in [0, \infty)$, and takes the value $|b_0| = 1$ only once. In order to bring the function $|b_0(\rho)|$ back to the interval [0, 1]we need to perform a large gauge transformation of the type defined in eq. (2.24), whenever $|b_0(\rho_n)| = n, n \in \mathbb{N}$. The number of D4-branes in the configuration then increases by a multiple of N_6 , as implied by eq. (2.25), each time we cross the position $\rho = \rho_n$.

The form of the B_2 potential in eq. (2.12) suggests that it is also possible to take a different 2-cycle,

$$\Sigma_2' = [\theta_1, \phi_1]_{\chi=0} , \qquad (2.28)$$

which is a rounded $S^2(\theta_1, \phi_1)$ at $\chi = 0$. As in the case analyzed above, large gauge transformations are needed as we move in ρ in order to render b_0 in the fundamental region, $b_0 \in [0, 1]$. This shift does not modify however the number of D4 or D6-branes, while it induces NS5-brane charge (we call these NS5' for later convenience) in the configuration.

Indeed, let us discuss the NS5-brane charges associated to the solution. Let us first consider the three-cycle,

$$\Sigma_3 = \left[\rho, \chi, \xi\right],\tag{2.29}$$

built out of the first 2-cycle $\Sigma_2 = [\chi, \xi]$ and the ρ -coordinate. Taking into account the expression for the B_2 field given by eq. (2.12) one finds,

$$H_3|_{\Sigma_3} = L^2 \left[\frac{(\lambda^2 - \lambda_1^2) \rho^2}{2} \partial_{\chi} \left(\frac{\sin \chi \sin 2\chi}{Q} \right) - P \sin \chi \ \partial_{\rho} \left(\frac{\rho^3}{Q} \right) \right] d\rho \wedge d\chi \wedge d\xi \ . \tag{2.30}$$

The first term does not contribute to the charge, which reads,

$$Q_{NS5} = \frac{1}{4\pi^2 \alpha'} \int_{(\rho,\chi,\xi)} H_3 = -\frac{1}{4\pi^2 \alpha'} 2\pi L^2 \rho_n^3 \int_0^{\pi} \frac{P}{Q} \sin \chi \, d\chi = b_0(\rho_n) = n \;. \tag{2.31}$$

This calculation shows that we have n NS5 branes for $\rho \in [0, \rho_n]$. If, on the other hand, we take the cycle defined by

$$\Sigma'_{3} = [\rho, \phi_{1}, \theta_{1}]_{\chi=0}, \qquad (2.32)$$

we find that $\rho'_n = n \pi \alpha'/L^2$ and that a new NS5' brane is created each time we cross these values ρ'_n for n = 1, 2, ...

⁴A physical interpretation of this condition in terms of a fundamental string action was presented in [35].





Figure 1. Brane set-up consistent with the quantized charges of the non-Abelian T-dual solution, consisting on $\alpha = 1, 2, ..., n+1$ NS5-branes (vertical black lines), $\beta = 1, 2, ..., n$ NS5'-branes (tilted red dashed lines) and $m N_6$ D4-branes (horizontal lines), where m = 1, 2, ..., n+1 changes by one each time a NS5-brane is crossed.

The conclusion of this analysis is that one can define *two types* of NS5-branes in the non-Abelian T-dual background: NS5-branes located at ρ_n and transverse to $\tilde{S}^2(\chi,\xi)$, and NS5'-branes located at $\rho'_n = n \pi \alpha'/L^2$ and transverse to $S^2(\theta_1, \phi_1)$. These branes are localized in the ρ direction, such that a NS5'-brane lies in between each pair of NS5-branes, as illustrated in figure 1. Further, as implied by eq. (2.25), N_6 D4-branes are created each time a NS5-brane is crossed. This brane set-up will be the basis of our proposed quiver in section 4, and will be instrumental in defining the dual CFT of the non-Abelian T-dual solution. As we will see, it will allow us to identify the global symmetries and the parameters characterizing the associated field theory.

Let us study now an important field theoretical quantity, calculated from the Type IIA solution, the central charge.

2.2.3 Central charge

In this section, we compute the holographic central charge associated to the non-Abelian T-dual solution in eqs. (2.11)–(2.17). This will be the main observable to check the validity of the $\mathcal{N} = 1$ quiver proposed in section 4.

We must be careful about the following subtle point. The calculation of the quantity \hat{V}_{int} in eq. (2.7), will involve an integral in the ρ -direction of the metric in eq. (2.11). The range of this coordinate is not determined by the process of non-Abelian T-duality (the global issues we referred to in the Introduction). If we take $0 \leq \rho < \infty$, we face the problem that the central charge will be strictly infinite. A process of *regularization* or *completion* of the background of eqs. (2.11)–(2.17) is needed. In this paper we choose to end the space with a hard cut-off, namely $0 \leq \rho \leq \rho_n$. We do know that this is geometrically unsatisfactory. Nevertheless, the field theoretical analysis of section 4 will teach us that a flavor group, represented by D6 branes added to the background of eqs. (2.11)–(2.17), should end the space in the correct fashion. Previous experience [44] tells us that the hard-cutoff used here does capture the result for the holographic central charge that is suitable to compare with the field theoretical one found in section 4.

We then proceed, by considering the metric in eq. (2.11), the dilaton in eq. (2.13) and eqs. (2.6)-(2.8). We obtain,

$$c_{\rm KWNATD} = \frac{9 \, L^6 \left(\rho_b^3 - \rho_a^3\right)}{64 \, \pi^3 \, \alpha'^3} \, N_6^2, \tag{2.33}$$

where we have integrated ρ between two arbitrary values $[\rho_a, \rho_b]$. We have also used the quantization condition of eq. (2.20). For $\rho \in [0, n \pi \alpha'/L^2)$ this gives

$$c_{\rm KWNATD}^{(0,n)} = \frac{9}{64} n^3 N_6^2 .$$
 (2.34)

On the other hand, for $\rho \in [n \pi \alpha'/L^2, (n+1) \pi \alpha'/L^2)$ we obtain,

$$c_{\text{KWNATD}}^{(n,n+1)} = \frac{9}{64} N_6^2 \left(3 n^2 + 3 n + 1\right).$$
(2.35)

This becomes $c_{\text{KWNATD}}^{(n,n+1)} = \frac{27}{64}N_4^2$ in the large *n* limit, where $\rho'_n = \rho_n$ and we can use that $N_4 = nN_6$ in the $[n \pi \alpha'/L^2, (n+1)\pi \alpha'/L^2)$ interval. Interestingly, this expression coincides with the central charge of the Abelian T-dual of the Klebanov-Witten background, that we discuss in detail in appendix B. This is that of the original background — see eq. (2.10), with N_3 replaced by N_4 ,

$$c_{\rm KWATD} = \frac{27}{64} N_4^2. \tag{2.36}$$

For completeness, we also reproduce in appendix C.3 this value of the central charge from the field theory, using *a*-maximization. This matching between the central charges of non-Abelian and Abelian T-duals was found in previous examples [44, 45].

Next, we review aspects of the $\mathcal{N} = 1$ quivers discussed in [55]. These will be the basis of the quiver proposed to describe the field theory associated to the non-Abelian T-dual solution. In section 4, the holographic result in eq. (2.34) will be found by purely field theoretical means.

3 Basics of Bah-Bobev 4d $\mathcal{N} = 1$ theories

In this section, we provide a summary of the results in [55], which will be instrumental for our proposal of a field theory dual to the background in eqs. (2.11)-(2.17).

3.1 $\mathcal{N} = 1$ linear quivers

In [55], Bah and Bobev introduced $\mathcal{N} = 1$ linear quiver gauge theories built out of $\mathcal{N} = 2$ and $\mathcal{N} = 1$ vector multiplets and ordinary matter multiplets. These theories were argued to flow to interacting 4d $\mathcal{N} = 1$ SCFTs in the infrared. They consist of products of $\ell - 1$ copies of SU(N) gauge groups, with either $\mathcal{N} = 1$ (shaded) or $\mathcal{N} = 2$ (unshaded) vector multiplets — see figure 2. Let n_1 be the number of $\mathcal{N} = 1$ vector multiplets and n_2 the number of $\mathcal{N} = 2$ vector multiplets. There are also $\ell - 2$ bifundamental hypermultiplets of SU(N) × SU(N), depicted in figure 2 as lines between the nodes, and two sets of N



Figure 2. General linear quiver in [55]. Shaded (unshaded) circles represent SU(N), $\mathcal{N} = 1$ ($\mathcal{N} = 2$) vector multiplets. Lines between them represent bifundamentals of $SU(N) \times SU(N)$. The boxes at the two ends represent SU(N) fundamentals.

hypermultiplets transforming in the fundamental of the two end SU(N) gauge groups. Thus, there are in total $\ell - 1 = n_1 + n_2$ gauge groups and ℓ matter multiplets. The total global symmetry is,

$$\mathrm{SU}(N) \times \mathrm{SU}(N) \times \mathrm{U}(1)^{\ell+n_2} \times \mathrm{U}(1)_R,$$

corresponding to the SU(N) flavor symmetries acting on the end hypermultiplets, the U(1) flavor symmetry acting on each of the ℓ hypermultiplets, the U(1) flavor acting on the chiral adjoint superfields (there are as many as $\mathcal{N} = 2$ vector multiplets) and the R-symmetry. Out of these U(1)'s only a certain non-anomalous linear combination will survive in the IR SCFT. Similarly, the fixed point R-charge is computed through a-maximization [69] as a non-anomalous linear combination of the U(1)'s and U(1)_R.

As shown in [55], it is convenient to assign a charge $\sigma_i = \pm 1$ to each matter hypermultiplet, with the rule that $\mathcal{N} = 1$ vector multiplets connect hypermultiplets with opposite sign, while $\mathcal{N} = 2$ vector multiplets connect hypermultiplets with the same sign. Let p be the number of hypermultiplets with $\sigma_i = +1$ and $q = \ell - p$ those with $\sigma_i = -1$, and let us introduce the twist parameter z,

$$z = \frac{p-q}{\ell} . \tag{3.1}$$

Thus, $z = \pm 1$ corresponds to a quiver with only $\mathcal{N} = 2$ nodes, involving hypermultiplets of the same charge. z = 0 corresponds in turn to a quiver with the same number of hypermultiplets of each type, so it includes the quiver with only $\mathcal{N} = 1$ nodes. We will focus on $0 \le z \le 1$ ($q \le p$) without loss of generality. We also introduce $\kappa = (\sigma_0 + \sigma_l)/2$, which can take values $\kappa = -1, 0, +1$. This will later be associated to the type of punctures on the Riemann surface on which M5-branes are wrapped.

In a superconformal fixed point the a and c central charges can be computed from the 't Hooft anomalies associated to the R-symmetry [70],

$$a = \frac{3}{32} \left(3 \operatorname{Tr} R^3 - \operatorname{Tr} R \right), \qquad c = \frac{1}{32} \left(9 \operatorname{Tr} R^3 - 5 \operatorname{Tr} R \right), \tag{3.2}$$

where the R-symmetry is given by

$$R_{\epsilon} = R_0 + \frac{1}{2} \epsilon \mathcal{F} \,, \tag{3.3}$$

and R_0 is the anomaly free R-symmetry, \mathcal{F} is the non-anomalous global U(1) symmetry and ϵ is a number that is determined by *a*-maximization [69]. This was used in [55] to compute the *a* and *c* central charges associated to the general quiver represented in figure 2. Their values were shown to depend only on the set of parameters $\{\kappa, z, \ell, N\}$. It was then conjectured that all quivers with the same $\{\kappa, z, \ell, N\}$ should be dual to each



Figure 3. The brane set-up associated to the Bah-Bobev $\mathcal{N} = 1$ linear quivers. Vertical lines represent NS5-branes extended along $\{x_4, x_5\}$, denoted in [55] as v-branes, while diagonal lines represent the NS5'-branes extended along $\{x_7, x_8\}$, denoted as w-branes. The same number of D4-branes extended along the x_6 direction stretch between adjacent 5-branes.

other and flow to the same SCFT in the infrared. Moreover, for $\ell \to \infty$ the two central charges were shown to agree. Therefore, in this limit the quivers can admit holographic AdS duals. In section 4 we will provide a variation of these $\mathcal{N} = 1$ quivers for which this condition is satisfied, and argue that it is associated to the AdS_5 non-Abelian T-dual solution presented in section 2.

3.2 IIA brane realization and M-theory uplift

Interestingly, it was shown in [55] that the linear quivers discussed above have a natural description in terms of D4, NS5, NS5' brane set-ups that generalize the $\mathcal{N} = 2$ brane constructions in [56], and allow for an M-theory interpretation. The two types of NS5-branes in this construction are taken to be orthogonal to each other, explicitly breaking $\mathcal{N} = 2$ supersymmetry to $\mathcal{N} = 1$. The specific locations of the branes involved are

- N coincident D4-branes extend along $\mathbb{R}^{1,3}$ and the x_6 direction.
- p non-coincident NS5-branes extend along $\mathbb{R}^{1,3} \times \{x_4, x_5\}$, and sit at $x_6 = x_6^{\alpha}$ for $\alpha = 1, \ldots, p$.
- q non-coincident NS5'-branes extend along $\mathbb{R}^{1,3} \times \{x_7, x_8\}$, and sit at $x_6 = x_6^\beta$ for $\beta = 1, \ldots, q$.

The corresponding brane set-up is depicted in figure 3, see also [55].

In this configuration, open strings connecting D4-branes stretched between two parallel NS5-branes are described at long distances and weak coupling by an $\mathcal{N} = 2 \text{ SU}(N)$ vector multiplet, while those connecting D4-branes stretched between perpendicular NS5 and NS5' branes are described by an $\mathcal{N} = 1 \text{ SU}(N)$ vector multiplet. In turn, open strings connecting adjacent D4-branes separated by a NS5-brane (NS5'-brane) are described at low energies by bifundamental hypermultiplets with charge $\sigma_i = 1$ ($\sigma_i = -1$). Finally, semi-infinite N D4-branes (or D6 branes) at both ends of the configuration yield two sets

of hypermultiplets in the fundamental representation of SU(N). The resulting field theory is effectively four dimensional at low energies compared to the inverse size of the D4 along x_6 . The effective gauge coupling behaves as $\frac{1}{g_4^2} \sim \frac{x_{6,n+1}-x_{6,n}}{g_s\sqrt{\alpha'}}$. Given that the 5-branes can be freely moved along the x_6 direction, the gauge couplings are marginal parameters. Rotations in the $v = x_4 + ix_5$ and $w = x_7 + ix_8$ planes of the NS5 and NS5' branes give a U(1)_v and a U(1)_w global symmetry, so that the IR fixed point R-symmetry and flavor U(1) are realized geometrically as linear combinations of them:

$$R_0 = U(1)_v + U(1)_w, \qquad \mathcal{F} = U(1)_v - U(1)_w.$$
(3.4)

Relying on similar $\mathcal{N} = 2$ constructions in [56], it is possible to describe the previous system of intersecting branes at strong coupling in M-theory. The x_6 direction is combined with the M-theory circle x_{11} to form a complex coordinate $s = (x_6 + ix_{11})/R_{11}$ describing a Riemann surface Σ_2 , which is a punctured sphere or, equivalently, a punctured cylinder. The uplift of this system yields,

- N M5-branes wrapping the cylinder, from the N D4-branes extended on x_6 .
- p simple punctures (in the language of [57]) on the cylinder, coming from the p transversal M5-branes with flavor charge $\sigma_i = 1$.
- q simple punctures on the cylinder, coming from the q transversal M5-branes with flavor charge $\sigma_i = -1$.
- Two maximal punctures, coming from the stacks of N transversal M5-branes at both ends of the cylinder. They are also assigned $\sigma_0, \sigma_\ell = \pm 1$, from which the additional parameter $\kappa = (\sigma_0 + \sigma_\ell)/2$ is defined, taking values $\kappa = -1, 0, +1$.

The cylinder or sphere the M5-branes wrap can be viewed as a Riemann surface $C_{g,n}$ of genus g = 0 and n = p + q + 2 punctures, so that $\Sigma_2 = C_{0,n}$. This Riemann surface can be deformed by bringing some of the punctures close to each other (which corresponds to certain weak and strong coupling limits of the dual 6d $\mathcal{N} = (0, 2) A_{N-1}$ field theory living on the M5-branes) to a collection of higher-genus and less-punctured surfaces. The κ parameter is associated to the type of punctures on the $C_{q,n}$ Riemann surface.

This closes our summary of the findings of the paper [55], that we will use in the next section. Let us now propose a dual CFT to our background in eqs. (2.11)-(2.17).

4 The non-Abelian T-dual of Klebanov-Witten as a $\mathcal{N} = 1$ linear quiver

As we showed in section 2.2.2, the analysis of the quantized charges of the non-Abelian T-dual solution is consistent with a D4, NS5, NS5' brane set-up in which the number of D4-branes stretched between the NS5 and NS5' branes increases by N_6 units every time a NS5-brane is crossed. This configuration thus generalizes the brane set-ups discussed in the previous section and in [55].

In this section, inspired by the previous analysis, we will use the brane set-up depicted in figure 1 to propose a linear quiver dual to the background in eqs. (2.11)-(2.17). As a consistency check we will compute its central charge using a-maximization and show that it is in perfect agreement with the holographic study in section 2.2.3 and the result of eq. (2.34), in particular. We will show that the central charge also satisfies the wellknown 27/32 ratio [72] with the central charge associated to the non-Abelian T-dual of $AdS_5 \times S^5/\mathbb{Z}_2$. This suggests defining our $\mathcal{N} = 1$ conformal field theory as the result of deforming by mass terms the $\mathcal{N} = 2$ CFT associated to the non-Abelian T-dual of $AdS_5 \times S^5/\mathbb{Z}_2$.

4.1 Proposed $\mathcal{N} = 1$ linear quiver

The quantized charges associated to the non-Abelian T-dual solution are consistent with a brane set-up, depicted in figure 1, in which D4-branes extend on $\mathbb{R}^{1,3} \times \{\rho\}$, NS5branes on $\mathbb{R}^{1,3} \times S^2(\theta_1, \phi_1)$ and NS5'-branes on $\mathbb{R}^{1,3} \times \tilde{S}^2(\chi, \xi)$. This produces for $\rho \in [n\pi\alpha'/L^2, (n+1)\pi\alpha'/L^2]$, $n \to \infty$ and upon compactification, the brane set-up, depicted in Figure 7 in appendix B, associated to the Abelian limit of the solution.

We conjecture that, in a similar fashion, the non-Abelian T-dual background in eqs. (2.11)-(2.17), arises as the decoupling limit of a D4, NS5, NS5' brane intersection. As opposed to its Abelian counterpart, the precise way in which D-branes transform under non-Abelian T-duality has not been worked out in the literature. This would require analysing the transformation of the boundary conditions at the level of the sigma model (see [71] for some preliminary steps in this direction). Still, as stressed in the previous works [44, 45], similar assumptions based on the analysis of the quantized charges of the supergravity background have produced consistent successful outcomes. Given that the precise D4, NS5, NS5' brane intersection is not known prior to the near horizon limit, it is unclear, on the other hand, how the original D3-brane configuration associated to the Klebanov-Witten solution would be recovered. In fact, even after taking the near horizon limit it is unclear how the Klebanov-Witten background would be recovered from the background defined by eqs. (2.11)-(2.17), given that the original SU(2) symmetry used to construct it is no longer present.⁵ These issues make non-Abelian T-duality substantially different from its Abelian counterpart, and underlie the fact that it can non-trivially change the dual CFT.

Coming back to our proposal, we would have an *infinite-length* quiver with (in the notation of section 3) p = n, q = n, $\ell = p + q = 2n$ and $z = (p-q)/\ell = 0$ with $n \to \infty$. The associated field theory would consist on (2n-1) $\mathcal{N} = 1$ vector multiplets and matter fields connecting them. However, this infinitely-long quiver *does not* describe a four dimensional field theory (its central charge is strictly infinite, among other problematic aspects). This is the same issue that we discussed when calculating the holographic central charge in section 2.2.3. Some regularization is needed and, as we will see, the field theory precisely provides the way to do this.

Elaborating on the ideas in [44], we propose to study this quiver for finite *n*, completing it as shown in figure 4. The proposed field theory has the following characteristics:

 $^{^{5}}$ This is related to the well-known non-invertibility of non-Abelian T-duality, noticed in the early works [2–8].



Figure 4. Linear quiver proposed as dual to the non-Abelian T-dual solution. There are two matter fields Q_j, \tilde{Q}_j in the bifundamental and anti-bifundamental of each pair of nodes, associated to a 5-brane connecting adjacent D4-stacks, with a total number of $j = 1, \ldots, n-1$ hypermultiplets $H_j = (Q_j, \tilde{Q}_j)$ at each side of the quiver. We label $r = 1, \ldots, [n/2]$ the $\sigma_j = +1$ hypermultiplets corresponding to NS5-branes and $s = 1, 2, \ldots, [n/2]$ the $\sigma_j = -1$ hypermultiplets from NS5'-branes, assuming an alternating distribution of both types of 5-branes. This configuration comes from a re-ordering of the branes in figure 1 that is consistent with Seiberg self-duality and the vanishing of the beta functions and R-symmetry anomalies. The squares in the middle of the quiver denote flavor groups corresponding either to semi-infinite D4-branes ending on the NS5 and NS5' branes or to D6-branes transversal to the D4-branes. They complete the quiver at finite n. We choose $\sigma_{f_1} = -\sigma_{f_2}$ for the corresponding fundamental hypermultiplets.

- It is strongly coupled. This is in correspondence with the fact that it should be dual to an AdS solution whose internal space is smooth in a large region and reduces to our non-Abelian T-dual background in eqs. (2.11)–(2.17) in some limit.
- The field theory is self-dual under Seiberg duality. This can be quickly seen, by observing that each node is at the self-dual point (with $N_f = 2N_c$).
- The beta function and the R-symmetry anomalies vanish, in correspondence with the SO(2, 4) isometry of the background and the number of preserved SUSYs.
- The central charge calculated by field theoretical means coincides (for long enough quivers) with the holographic result of eq. (2.34).
- The quiver can be thought of as a mass deformation of the N = 2 quiver dual to the non-Abelian T-dual of AdS₅ × S⁵/ℤ₂.

Below, we show that the field theory represented in figure 4 has all these characteristics. As it happens in the paper [44], the completion we propose with the flavor groups has the effect of ending the space at a given finite value in the ρ direction.

4.2 β -functions and R-symmetry anomalies

In this section we study the β -functions and the anomalies associated to the linear quiver proposed in figure 4. This analysis clarifies that the quantum field theory flows to a conformal fixed point in the infrared.

In a supersymmetric gauge theory, the β -function for a coupling constant g is given by the well-known Novikov-Shifman-Vainshtein-Zakharov (NSVZ) formula [73], which can be



Figure 5. The R-symmetry anomaly.

written in terms of the number of colors, N_c , the number of flavors, N_{fq} , and the anomalous dimensions for the matter fields, γ_q , as

$$\beta_g \sim 3 N_c - \sum_q N_{f_q} (1 - \gamma_q)$$
 (4.1)

Here, we considered the Wilsonian beta function. The denominator in the NSVZ formula is not relevant for us (see [74] for a nice explanation of this). Another important quantity is the R-symmetry anomaly, given by the correlation function of three currents and represented by the Feynman diagram in figure 5. The anomaly is given by the relation,

$$\langle \partial^{\mu} J^{5}_{\mu} \rangle \sim \Delta \Theta F \tilde{F}, \qquad \Delta \Theta = \sum R_{f} T(R_{f}), \qquad (4.2)$$

where R_f is the R-charge of the fermions in the multiplet. In the case of an SU(N) gauge group

$$T(R_f) = \begin{cases} 2N, & \text{for fermions in the adjoint representation} \\ 1, & \text{for fermions in the fundamental representation} \end{cases}$$
(4.3)

Moreover, at the conformal point, one should take into account the relation between the physical dimension of a gauge invariant operator \mathcal{O} (with engineering dimension $\Delta_{\mathcal{O}}$) and its R-charge $R_{\mathcal{O}}$,

$$\dim \mathcal{O} = \Delta_{\mathcal{O}} + \frac{\gamma_{\mathcal{O}}}{2} = \frac{3}{2} R_{\mathcal{O}} .$$
(4.4)

In the appendix C, we present details of these calculations for the well-known example of the Klebanov-Witten CFT. Readers unfamiliar with that example can study the details in appendix C and then come back to the more demanding calculation presented below.

Let us now analyze the quiver depicted in figure 4. We propose for the anomalous dimensions and R-charges of the matter fields and gauginos the same values as in the Klebanov-Witten CFT,

$$\gamma_Q = \gamma_{\bar{Q}} = -\frac{1}{2}, R_Q = R_{\bar{Q}} = \frac{1}{2}, \quad R(\lambda) = 1.$$
 (4.5)

Notice that in our proposal only one bifundamental field runs in each arrow. We call them Q or \tilde{Q} depending on the direction of the arrow. We find, substituting in eq. (4.1) for the nodes with rank kN_6 ,

$$\beta_k \sim 3 \, k \, N_6 - \left((k+1) \, N_6 + (k-1) \, N_6 \right) \left(1 + \frac{1}{2} \right) = 0 \,, \qquad k = 1, \dots, n \,. \tag{4.6}$$

The first term reflects the contribution of the gauge multiplets and the second that of the matter fields. For the anomaly we find,

$$\Delta \theta_k = 2 k N_6 + 2 \left((k+1) N_6 + (k-1) N_6 \right) \left(-\frac{1}{2} \right) = 0, \qquad k = 1, \dots, n.$$
(4.7)

The first term indicates the contribution of the gauginos and the second one the contribution of the fermions in the Q, \tilde{Q} multiplets.

These calculations indicate that both R-symmetry anomalies and beta functions are vanishing. Indeed, they belong to the same anomaly multiplet. Also, notice that the large anomalous dimensions indicate that the CFT is strongly coupled. With this numerology, we calculate that

$$\dim Q = \dim \tilde{Q} = \frac{3}{4}.$$
(4.8)

This allows for the presence of superpotential terms involving four matter multiplets, like the ones proposed in [55]. Let us move now to the calculation of the central charge.

4.3 Field-theoretical central charge

In this section we compute the central charge of the quiver depicted in figure 4 at the fixed point, using the a-maximization procedure [69].

As recalled in section 3, the *a* and *c* central charges can be computed from the $\mathcal{N} = 1$ R-symmetry t'Hooft anomalies of the fermionic degrees of freedom of the theory,

$$a(\epsilon) = \frac{3}{32} \left(3 \operatorname{Tr} R_{\epsilon}^3 - \operatorname{Tr} R_{\epsilon} \right), \quad c(\epsilon) = \frac{1}{32} \left(9 \operatorname{Tr} R_{\epsilon}^3 - 5 \operatorname{Tr} R_{\epsilon} \right).$$
(4.9)

The *R*-symmetry is given by $R_{\epsilon} = R_0 + \frac{1}{2} \epsilon \mathcal{F}$. Assigning charges $R_0(Q_j) = R_0(\tilde{Q}_j) = 1/2$ to the chiral multiplet scalars, we have that

$$R_{\epsilon}(Q_j) = R_{\epsilon}(\tilde{Q}_j) = \frac{1}{2} \left(1 + \epsilon \, \sigma_j \right),$$

and, for the fermions

$$R_{\epsilon}(\psi_j) = R_{\epsilon}(\tilde{\psi}_j) = \frac{1}{2} \left(-1 + \epsilon \sigma_j \right).$$

Now, we can compute the linear contribution to the anomaly coming from the hypermultiplet $H_j = (Q_j, \tilde{Q}_j)$, whose chiral fields transform in the fundamental of a gauge group with rank N_a and in the anti-fundamental of another gauge group with rank N_b , and vice-versa:

$$\operatorname{Tr} R_{\epsilon}(H_j) = N_a N_b \left(R_{\epsilon}(\psi_j) + R_{\epsilon}(\tilde{\psi}_j) \right) = N_a N_b(\epsilon \sigma_j - 1) \,. \tag{4.10}$$

The cubic contribution is

$$\operatorname{Tr} R_{\epsilon}^{3}(H_{j}) = N_{a} N_{b} \left(R_{\epsilon}^{3}(\psi_{j}) + R_{\epsilon}^{3}(\tilde{\psi}_{j}) \right) = 2N_{a} N_{b} \left[\frac{1}{2} (\epsilon \sigma_{j} - 1) \right]^{3}.$$
(4.11)

In turn, the linear and cubic anomaly contributions from an $\mathcal{N} = 1$ vector multiplet V_t are given by,

$$\operatorname{Tr} R_{\epsilon}(V_t) = \operatorname{Tr} R_{\epsilon}^3(V_t) = N_a^2 - 1, \qquad (4.12)$$

where we have used that $R_{\epsilon}(\lambda) = R_0(\lambda) = 1$ for the gaugino.

We now consider the completed quiver in figure 4. Hypermultiplets with $\sigma_j = +1$ and $\sigma_j = -1$ (transforming in the bifundamental of gauge groups of ranks N_j , N_{j+1}) alternate along the quiver, and $\sigma_{f_1} = -\sigma_{f_2}$. In this way all nodes are equipped with $\mathcal{N} = 1$ vector multiplets. Moreover, we have z = 0 exactly, as well as $\kappa = 0$. The total linear contribution of the hypermultiplets is then:

$$\operatorname{Tr} R_{\epsilon}(H) = \sum_{j=1,\text{left}}^{n-1} \operatorname{Tr} R_{\epsilon}(H_{j}) + \sum_{j=1,\text{right}}^{n-1} \operatorname{Tr} R_{\epsilon}(H_{j}) + \sum_{i=1}^{2} \operatorname{Tr} R_{\epsilon}(H_{f_{i}})$$

$$= N_{6}^{2} \left\{ \sum_{j=1}^{n-1} j \left(j+1 \right) \left(\epsilon \underbrace{(\sigma_{j,\text{left}} + \sigma_{j,\text{right}})^{-} 1 \right) + n \sum_{i=1}^{2} \left(\epsilon \sigma_{f_{i}} - 1 \right) \right\}$$

$$= N_{6}^{2} \left\{ -2 \sum_{j=1}^{n-1} j \left(j+1 \right) + n \left(\epsilon \underbrace{(\sigma_{f_{1}} + \sigma_{f_{2}})^{-} 2} \right) \right\}$$

$$= N_{6}^{2} \left\{ -\frac{2}{3} n \left(n^{2} - 1 \right) - 2n \right\} = N_{6}^{2} \left\{ -\frac{2}{3} n^{3} - \frac{4}{3}n \right\}$$

$$\approx -\frac{2}{3} n^{3} N_{6}^{2} + \mathcal{O}(n) . \qquad (4.13)$$

In the last line the approximation of a *long quiver* (large n) has been used. Similarly, the total cubic contribution of the hypermultiplets can be readily computed to be,

$$\operatorname{Tr} R_{\epsilon}^{3}(H) = \sum_{j=1,\text{left}}^{n-1} \operatorname{Tr} R_{\epsilon}^{3}(H_{j}) + \sum_{j=1,\text{right}}^{n-1} \operatorname{Tr} R_{\epsilon}^{3}(H_{j}) + \sum_{i=1}^{2} \operatorname{Tr} R_{\epsilon}^{3}(H_{f_{i}})$$
$$= N_{6}^{2} \left[\sum_{j=1}^{n-1} j \left(j+1 \right) \frac{1}{4} \left(\left(\epsilon \, \sigma_{j,\text{left}} - 1 \right)^{3} + \left(\epsilon \, \sigma_{j,\text{right}} - 1 \right)^{3} \right) + n \, N_{6}^{2} \sum_{i=1}^{2} \frac{1}{4} \left(\epsilon \, \sigma_{f_{i}} - 1 \right)^{3} \right]$$
$$= \frac{N_{6}^{2}}{4} \left[-2(1+3\epsilon^{2}) \sum_{j=1}^{n-1} j \left(j+1 \right) + n \sum_{i=1}^{2} \left(\epsilon \, \sigma_{f_{i}} - 1 \right)^{3} \right]$$
$$= \frac{N_{6}^{2}}{12} \left[-2(1+3\epsilon^{2})n^{3} - (12\epsilon^{2}+4)n \right] \approx -\frac{1}{6} n^{3} N_{6}^{2} \left(1+3\epsilon^{2} \right) + \mathcal{O}(n), \quad (4.14)$$

where long quivers have been considered in the last expression. In turn, recalling that each node appears twice in the quiver depicted in figure 4, with the exception of the central one, the trace anomaly coming from the $\mathcal{N} = 1$ vector multiplets becomes,

$$\operatorname{Tr} R_{\epsilon}(V) = \operatorname{Tr} R_{\epsilon}^{3}(V) = 2 \sum_{t=1}^{n-1} \operatorname{Tr} R_{\epsilon}(V_{t}) + \operatorname{Tr} R_{\epsilon}(V_{n}) = 2 \sum_{t=1}^{n-1} \left(t^{2} N_{6}^{2} - 1 \right) + \left(n^{2} N_{6}^{2} - 1 \right) \\ = \frac{N_{6}^{2}}{3} \left(2n^{3} + n \right) - 2(n-1) \approx \frac{2}{3} n^{3} N_{6}^{2} + \mathcal{O}(n) .$$

$$(4.15)$$

From this result we see that $\operatorname{Tr} R_{\epsilon}(V) \approx -\operatorname{Tr} R_{\epsilon}(H)$ in the large *n* limit, so that the overall linear trace anomaly is of order $n N_6^2$ at most. Putting all these expressions together we



Figure 6. Completed quiver associated to the non-Abelian T-dual of $AdS_5 \times S^5$. Each line represents a hypermultiplet of $\mathcal{N} = 2$ SUSY.

find, for the exact charges in eq. (4.9),

$$a(\epsilon) = \frac{3N_6^2}{64} \bigg\{ 3(1-\epsilon^2)n^3 + 2(1-3\epsilon^2)n - \frac{4}{N_6^2}(2n-1) \bigg\},$$

$$c(\epsilon) = \frac{N_6^2}{64} \bigg\{ 9(1-\epsilon^2)n^3 + 2(5-9\epsilon^2)n - \frac{8}{N_6^2}(2n-1) \bigg\}.$$
(4.16)

From these expressions we see that $a(\epsilon)$ is clearly maximized for $\epsilon = 0$, as expected for the $\mathcal{N} = 1$ fixed point [55]. The superconformal central charges are thus found to be

$$a_{\mathcal{N}=1} \equiv a(\epsilon = 0) = \frac{3}{64} \left\{ (3n^3 + 2n)N_6^2 - 4(2n - 1) \right\},\$$

$$c_{\mathcal{N}=1} \equiv c(\epsilon = 0) = \frac{1}{64} \left\{ (9n^3 + 10n)N_6^2 - 8(2n - 1) \right\}.$$
 (4.17)

They give, in the large n limit,

$$c_{\mathcal{N}=1} \approx a_{\mathcal{N}=1} \approx \frac{9}{64} n^3 N_6^2 + \mathcal{O}(n) \,.$$
 (4.18)

This final result matches the holographic calculation given by eq. (2.34). This provides a non-trivial check of the validity of the linear quiver in figure 4 as dual to the background in eqs. (2.11)–(2.17). It is noteworthy that the agreement with the holographic result occurs in the large number of nodes limit, $n \to \infty$.

A further non-trivial check of the validity of our proposed quiver is that the central charge given by (4.18) and that associated with the non-Abelian T-dual of $AdS_5 \times S^5/\mathbb{Z}_2$ satisfy the same 27/32 relation [72], that is,

$$c_{\mathcal{N}=1} = \frac{27}{32} c_{\mathcal{N}=2},\tag{4.19}$$

as the central charges of the corresponding theories prior to dualization. Indeed, the quiver associated to the non-Abelian T-dual of $AdS_5 \times S^5/\mathbb{Z}_2$ can be obtained by modding out by \mathbb{Z}_2 the quiver describing the non-Abelian T-dual of $AdS_5 \times S^5$, constructed in [44] and depicted in figure 6. This quiver was completed at finite *n* by a flavor group with gauge group SU(nN_6). It thus satisfies the condition to be conformal (preserving $\mathcal{N} = 2$ SUSY), i.e. that the number of flavors is twice the number of colors at each node. Modding out by \mathbb{Z}_2 results in the same quiver in figure 4, but built out of $2n \mathcal{N} = 2$ vector and matter multiplets. Taking the central charge, computed in [44], for the non-Abelian T-dual of $AdS_5 \times S^5$ and doubling it, we obtain the central charge of the non-Abelian T-dual of $AdS_5 \times S^5/\mathbb{Z}_2$

$$c_{NATD\,AdS_5 \times S^5/\mathbb{Z}_2} \approx 2 \times \frac{1}{12} n^3 N_6^2 + \mathcal{O}(n),$$
 (4.20)

and we find that eq. (4.19) indeed holds with $c_{\mathcal{N}=1}$ as in eq. (4.18) and $c_{\mathcal{N}=2}$ as in eq. (4.20). We have checked in appendix C.2 that the same result (4.20) is reproduced using *a*-maximization. The *a*-charge is maximized for $\epsilon = \frac{1}{3}$, as previously encountered in [55].

Further, one can check that also at finite n, $a_{\mathcal{N}=1}$ and $c_{\mathcal{N}=1}$ satisfy the relation [72],⁶

$$a_{\mathcal{N}=1} = \frac{9}{32} \left(4 \, a_{\mathcal{N}=2} - c_{\mathcal{N}=2} \right), \qquad c_{\mathcal{N}=1} = \frac{1}{32} \left(-12 \, a_{\mathcal{N}=2} + 39 \, c_{\mathcal{N}=2} \right) \tag{4.21}$$

with the $a_{\mathcal{N}=2}$, $c_{\mathcal{N}=2}$ exact central charges of the $\mathcal{N}=2$ quiver. The explicit expressions for $a_{\mathcal{N}=2}$ and $c_{\mathcal{N}=2}$ are given in eq. (C.11) in appendix C.2. This precisely defines our dual CFT as the result of deforming by mass terms the CFT dual to the Sfetsos-Thompson solution modded by \mathbb{Z}_2 .

The material presented in this section makes very precise the somewhat loose ideas proposed in the works [12, 13]. In particular, we have identified the concrete relation via a RG-flow between the non-Abelian T-dual of $AdS_5 \times S^2/\mathbb{Z}_2$ and the non-Abelian T-dual of the Klebanov-Witten solution. Notice that here, we are providing precisions about the CFT dual to the non-Abelian T-dual backgrounds. This more precise information is matched by the regularized form of the non-Abelian T-dual solution.

The diagram in the Introduction section summarizes the connections between the UV and IR field theories discussed in this section. We repeat it here for the perusal of the reader.

$$\begin{array}{ccc} AdS_5 \times S^5/\mathbb{Z}_2 & \longrightarrow & \text{NATD of } AdS_5 \times S^5/\mathbb{Z}_2 \\ & & & & & & \\ & & & & & \\ AdS_5 \times T^{1,1} & \longrightarrow & \text{NATD of } AdS_5 \times T^{1,1} \end{array}$$

As a closing remark, an explicit flow (triggered by a VEV) between the $\mathcal{N} = 1$ and the $\mathcal{N} = 2$ non-Abelian T-dual backgrounds was constructed in [35]. It should be interesting to use the detailed field theoretical picture developed above and in [44], to be more precise about various aspects of this RG-flow.

5 Solving the INST-BBBW puzzle

The non-Abelian T-dual of the Klebanov-Witten background was first written in [12, 13] (INST). Further, in that paper an attempt was made to match the non-Abelian T-dual background with a Bah, Beem, Bobev and Wecht (BBBW) solution [59]. This matching was feasible assuming a particular split of the metric into a seven-dimensional and a fourdimensional internal space (see below). The formula in [59] for the central charge of BBBW solutions led however to $a \sim c \sim 0 + O(N)$ for the non-Abelian T-dual solution, in blatant disagreement with the holographic result. This was the puzzle that the authors of [12, 13] pointed out. In this section we present its resolution. We start by summarizing the most relevant aspects of the work [59].

⁶We would like to thank Nikolay Bobev for suggesting this to us.

In the work of Bah, Beem, Bobev and Wecht new $\mathcal{N} = 1 \ AdS_5$ solutions in M-theory were constructed, describing the fixed points of new $\mathcal{N} = 1$ field theories associated to M5-branes wrapped on complex curves. The central charges of these SCFTs were computed using the six dimensional anomaly polynomial and *a*-maximization, and were shown to match, in the large number of M5-branes limit, the holographic results.

The solutions constructed in [59] were obtained by considering M-theory compactified on a deformed four-sphere. In principle, this compactification leads to an SO(5)-gauged supergravity in seven dimensions. Following the ideas in [47], BBBW searched for their solutions in the seven dimensional gravity theory (a U(1)² truncation of the full SO(5) theory) discussed in [75]. They proposed a background consisting of a metric, two gauge fields $A^{(i)}_{\mu}$ and two scalars $\lambda^{(i)}$, of the form

$$ds_7^2 = e^{2f(r)} [dx_{1,3}^2 + dr^2] + e^{2g(r)} d\Sigma_k(x_1, x_2),$$

$$F^{(1)} = \frac{p}{8q - 8} vol\Sigma_k, \quad F^{(2)} = \frac{q}{8q - 8} vol\Sigma_k, \quad \lambda^{(i)}(r).$$
(5.1)

They then searched for 'fixed point' solutions, namely, those where $\frac{d}{dr}\lambda^{(i)} = \frac{d}{dr}g = 0$ and $f \sim -\log r$, leading to backgrounds of the form $AdS_5 \times \Sigma_k$. They found general solutions depending on four parameters (N, κ, z, g) . For excitations with wavelength longer than the size of Σ_k , these are dual to four dimensional CFTs. In the dual CFT N is the number of M5-branes, $\kappa = \pm 1, 0$ is the curvature of the 2d Riemann surface that they are wrapping, g is its genus and z is the so-called 'twisting parameter', defined as $z = \frac{(p-q)}{2(g-1)}$ from the integer numbers p, q that indicate the twisting applied to the M5-branes. The holographic central charge computed in [59] depends on these parameters, and reads

$$c = a = N^{3}(1-g) \left[\frac{1-9z^{2} + \kappa(1+3z^{2})^{3/2}}{48z^{2}} \right].$$
 (5.2)

BBBW completed their analysis deriving various of their formulas, in particular the holographic central charge, using purely 4d CFT arguments. Their CFTs are combinations of Gaiotto's T_N -theories, conveniently gauged and connected with other T_N factors, with either $\mathcal{N} = 1$ or $\mathcal{N} = 2$ vector multiplets (shaded and unshaded T_N 's in the same line as what we explained in section 3).

The key-point to be kept in mind after this discussion is that these results were obtained in the context of a *compactification* of eleven-dimensional supergravity to seven dimensions.

Let us now come back to the paper [12, 13]. The matching of the non-Abelian Tdual solution with a BBBW geometry assumed that the seven dimensional part of the metric in (5.1) was $AdS_5 \times S^2(\theta_1, \phi_1)$ and that the internal space contained the coordinates $[\rho, \chi, \xi, x_{11}]$. Also, the authors of [12, 13] chose the parameters $\kappa = z = 1$ for such matching. Using the formula (5.2) in BBBW for the central charge they then found that at leading order the central charge vanished.

What was not-correct in the analysis of [12, 13] was the assumption that the non-Abelian T-dual solution could be obtained from a *compactification* of M-theory on a deformed four-sphere (and hence be in the BBBW class of solutions). In fact, inspecting the

BPS equations of BBBW — eq. (3.10) of [59] — one finds that a fixed point solution does not exist for the set of values $\kappa = |z| = 1$. Even more, the generic solution that BBBW wrote in their eq. (3.8) is troublesome for those same values.

A parallel argument can be made by comparing the BBBW and non-Abelian T-dual solutions in the language of the paper [76]. Indeed, the comparison in the appendix C of [76], shows that these solutions fit in their formalism in section 4.2 for values of parameters that are incompatible. Either BBBW is fit *or* the non-Abelian T-dual solution is, for a chosen set of parameters.

The resolution to this problem is that the non-Abelian T-dual background should instead be thought of as providing a non-compactification of eleven dimensional supergravity. Strictly speaking, our coordinate ρ runs in $[0, \infty]$, the four manifold is non-compact. In our calculation of the central charge, we assumed that the ρ -coordinate was bounded in $[0, n\pi \frac{\alpha'}{L^2}]$, but this hard cut-off, as we emphasized, is not a geometrically satisfactory way of bounding a coordinate. There should be another, more general solution, that contains our non-Abelian T-dual metric in a small patch of the space (for small values of ρ), and closes the ρ -coordinate at some large value $\rho_n = n\pi \frac{\alpha'}{L^2}$. But this putative new metric, especially its behaviour near ρ_n , will differ considerably from the one obtained via non-Abelian T-duality. Below, we will comment more about this putative solution.

Let us close with some field theoretical remarks. The class of CFTs studied by BBBW [59] are quite different from those studied by Bah and Bobev in [55]. Their central charges are different, and the first involve Gaiotto's T_N theories while the second do not. In the same line, our CFT discussed in section 4 is a generalization, but strictly different, of the theories in [55], and is certainly different from those in [59].

The quiver we presented in section 4 encodes the dynamics of a solution in Type IIA/M-theory where the ρ -coordinate is bounded in a geometrically sounding fashion. The addition of the flavor groups in our quiver encode the way in which the ρ -coordinate should be ended. Indeed, in analogy with what was observed in [44, 45], we expect the metric behaving like that of D6 branes close to the end of the space. In M-theory language, we expect to find a puncture on the Riemann surface, representing the presence of flavor groups in the dual CFT. We will be slightly more precise about this in the Conclusions section.

6 Conclusions and future directions

Let us briefly summarize the main achievements of this paper.

After discussing details of the Type IIA solution obtained by non-Abelian T-duality applied on the Klebanov-Witten background, we carefully studied its quantized charges and holographic central charge (section 2). We lifted the solution to M-theory and showed by explicit calculation of the relevant differential forms that the background has SU(2)structure and fits the classification of [65].

Based on the quantized charges, we proposed a brane set-up (section 4) and a precise quiver gauge theory, generalizing the class of theories discussed by Bah and Bobev in [55] (and summarized in our section 3). This quiver was used to calculate the central charge, one of the important observables of a conformal field theory at strong coupling. Indeed, in

section 4, we showed the precise agreement of this observable, computed by field theoretical means, with the holographic central charge. We also showed that the quiver has a strongly coupled IR-fixed point. Finally, section 5, solves a puzzle raised in previous bibliography. Various appendices discuss technical points in detail. In particular, relations of the non-Abelian T-dual of the Klebanov-Witten background and the more conventional T-dual, details about the dual field theory, etc, are carefully explained there.

To close this paper let us state the most obvious and natural continuation of our work. As we discussed, the holographic central charge calculation in section 2 was done for a *regulated* version of the Type IIA background. Indeed, the integral over the internal space was taken to range in a finite interval for the ρ -coordinate. We introduced a hard-cutoff, but emphasized that this form of regularization is not rigorous from a geometric viewpoint. Fortunately, the dual CFT provides a rationale to regulate the space. The flavor groups $SU(N_6)$ that end our quiver field theory (see figure 4), will be reflected in the Type IIA background by the presence of flavor branes that will backreact and end the geometry, solving the Einstein's equations. In eleven dimensions, the same effect will be captured by punctures on the S^2 that the M5 branes are wrapping. A phenomenon like this was at work in the papers [44, 45].

The formalism to backreact these flavor D6 branes is far-less straightforward in the present case, as the number of isometries and SUSY is less than in the cases of [44, 45]. Qualitatively one may think of defining the completed solution by deforming with mass terms the superposition of $\mathcal{N} = 2$ Maldacena-Nunez solutions [47] used in [44] to complete the Sfetsos-Thompson background. This would give rise to a superposition of $\mathcal{N} = 1$ MN solutions defining the completed non-Abelian T-dual solution. It is unclear however in which precise way this superposition would solve the (very non-linear) PDEs associated to $\mathcal{N} = 1$ solutions [65, 77]. We see two possible paths to follow:

- In the paper [77], Bah rewrote the general M-theory background of [65] in terms of a new set of coordinates that are more useful to discuss the addition of punctures on the Riemann surface. In the type IIA language the new solutions found using Bah's non-linear and coupled PDEs should represent the addition of the flavor D6 branes argued above. The equations need to be solved close to the singularity (the puncture or the flavor D6 brane) and then numerically matched with the rest of the non-Abelian T-dual background.
- In [76] generic backgrounds in massive Type IIA were found with an AdS₅ factor in the metric and preserving eight SUSYs. For the particular case in which the internal space contains a Riemann surface of constant curvature, the involved set of non-linear and coupled PDEs simplifies considerably. One of the solutions, for the case in which the massive parameter vanishes, is the one studied in this paper — named INST in [76]. Since the paper [76] and some follow-up works have discussed ways of ending these spaces by the addition of D6 and D8 branes, we could consider these technical developments together with the ideas discussed above.

Finding a *completed* or *regularized* solution would provide the first example for a back-

FHEP09 (2017) 03

00

ground dual to a CFT like that discussed in section 4. The natural following steps would be to extend the formalism to discuss the situations for a cascading QFT. In fact, the precise knowledge of the CFT we have achieved in this paper can be used to improve the understanding and cure the singularity structure of the backgrounds written in the first paper in [12, 13], in [20], etc. We reserve these problems to be discussed in forthcoming publications.

Acknowledgments

We would like to thank Fabio Apruzzi, Antoine Bourget, Noppadol Mekareeya, Achilleas Passias, Alessandro Tomasiello, Daniel Thompson, Salomón Zacarías and especially Nikolay Bobev and Niall Macpherson for very useful discussions. G.I. is supported by the FAPESP grants 2016/08972-0 and 2014/18634-9. Y.L. and J.M. are partially supported by the Spanish and Regional Government Research Grants FPA2015-63667-P and FC-15-GRUPIN-14-108. J.M. is supported by the FPI grant BES-2013-064815 of the Spanish MINECO, and the travel grant EEBB-I-17-12390 of the same institution. Y.L. would like to thank the Physics Department of Swansea U. and the Mainz Institute for Theoretical Physics (MITP) for the warm hospitality and support. J.M. is grateful to the Physics Department of Milano-Bicocca U. for the warm hospitality and exceptional working atmosphere. C.N. is Wolfson Fellow of the Royal Society.

A Connection with the GMSW classification

In this appendix we prove that the uplift of the non-Abelian T-dual of the Klebanov-Witten solution fits in the classification of $\mathcal{N} = 1 \ AdS_5$ backgrounds in M-theory of GMSW [65].

A.1 Uplift of the non-Abelian T-dual solution

The eleven dimensional uplift of the non-Abelian T-dual solution consists of metric and 4-form flux. The metric is given by

$$ds_{11}^2 = e^{-\frac{2\Phi}{3}} ds_{IIA}^2 + e^{\frac{4\Phi}{3}} \left(dx_{11} + C_1 \right)^2, \tag{A.1}$$

where x_{11} stands for the 11th coordinate, ds_{IIA}^2 is the ten dimensional metric, given by eq. (2.11), Φ is the dilaton, given by eq. (2.13), and C_1 is the RR potential given in eq. (2.17). The eleven dimensional four-form field, F_4^M , is derived from $F_4^M = dC_3^M$, where

$$C_3^M = C_3 + B_2 \wedge dx_{11}, \qquad (A.2)$$

and C_3 and B_2 are given by (2.17) and (2.12), respectively. The final expression for F_4^M is given by

$$F_4^M = -\frac{\alpha' \lambda^2 \left(L^4 \lambda_1^4 + \alpha'^2 \rho^2\right)}{Q} \cos \chi \, d\Omega_2(\theta_1, \phi_1) \wedge d\rho \wedge dx_{11} \\ + \frac{\alpha' L^4 \lambda_1^4 \lambda^2}{Q} \rho \sin \chi \, d\Omega_2(\theta_1, \phi_1) \wedge d\chi \wedge dx_{11}$$
$$+ \frac{4 \alpha'^{3/2} L^4 \lambda_1^4 \lambda \left(\lambda^2 - \lambda_1^2\right)}{g_s Q} \rho^2 \cos \chi \sin^2 \chi \, d\Omega_2(\theta_1, \phi_1) \wedge d\rho \wedge d\xi + \frac{4 \alpha'^{3/2} L^4 \lambda_1^4 \lambda P}{g_s Q} \rho^3 \, d\Omega_2(\theta_1, \phi_1) \wedge d\Omega_2(\chi, \xi) + \frac{4 \alpha'^{3/2} L^4 \lambda_1^4 \lambda^3 S}{Q^2} \rho^2 \cos \theta_1 \, d\Omega_2(\chi, \xi) \wedge d\rho \wedge d\phi_1 + \frac{\alpha'^3 \lambda^2 S}{Q^2} \rho^2 \sin \chi \left(d\xi + \cos \theta_1 \, d\phi_1\right) \wedge d\rho \wedge d\chi \wedge dx_{11},$$
(A.3)

where for the sake of clarity we have defined,

$$S \equiv Q - 2L^4 \,\lambda_1^4 \left(\lambda^2 + \lambda_1^2\right) - 2\,\alpha'^2 \,\rho^2 \,\lambda_1^2 \,. \tag{A.4}$$

In the large n limit this expression takes the simpler form $(\lambda^2 = 1/9, \lambda_1^2 = 1/6, \alpha' = g_s = 1),$

$$F_4^M \approx \frac{L^4}{27} (\rho - n\pi) \, d\Omega_2(\theta_1, \phi_1) \wedge d\Omega_2(\chi, \xi) \,, \tag{A.5}$$

which tells us that the M5-branes sourcing this flux are transversal to both squashed two-spheres $S^2(\theta_1, \phi_1)$ and $\tilde{S}^2(\chi, \xi)$. These are associated to the global isometries $U(1)_w$ and $U(1)_v$, whose product lies in the Cartan of both the local R-symmetry and the non-anomalous flavor symmetry.

A.2 Review of GMSW

Before matching the previous solution within the classification in [65], let us briefly review the most general $\mathcal{N} = 1$ eleven dimensional solutions with an AdS_5 factor found in that paper. These solutions are described by a metric of the form,

$$ds_{11}^2 = e^{2\Lambda} \left[ds_{AdS_5}^2 + ds_{M_4}^2 + \frac{e^{-6\Lambda}}{\cos^2 \zeta} \, dy^2 + \frac{\cos^2 \zeta}{9 \, m^2} \left(d\tilde{\psi} + \tilde{\rho} \right)^2 \right] \,. \tag{A.6}$$

Here $\tilde{\psi}$ is the R-symmetry direction, $\tilde{\rho}$ is a one-form defined on M_4 , whose components depend on both the M_4 coordinates and y, and Λ and ζ are functions also depending on the M_4 coordinates and y. The coordinate y is related to the warping factor Λ and the function ζ through,

$$2my = e^{3\Lambda} \sin\zeta, \qquad (A.7)$$

with m being the inverse radius of AdS_5 .

The four-dimensional manifold M_4 admits an SU(2) structure which is characterized by a (1, 1)-form J and a complex (2, 0)-form Ω . The SU(2) structure forms, together with the frame components K^1 and K^2 , defined as,

$$K^{1} \equiv \frac{e^{-3\Lambda}}{\cos\zeta} \, dy \,, \qquad K^{2} \equiv \frac{\cos\zeta}{3\,m} \left(d\tilde{\psi} + \tilde{\rho} \right), \tag{A.8}$$

must satisfy the following set of differential conditions dictated by supersymmetry,

$$e^{-3\Lambda} d(e^{3\Lambda} \sin \zeta) = 2m \cos \zeta K^1, \qquad (A.9)$$

$$e^{-6\Lambda} d(e^{6\Lambda} \cos \zeta \Omega) = 3 m \Omega \wedge \left(-\sin \zeta K^1 + i K^2\right), \tag{A.10}$$

$$e^{-6\Lambda} d\left(e^{6\Lambda} \cos\zeta K^2\right) = e^{-3\Lambda} \star G + 4m \left(J - \sin\zeta K^1 \wedge K^2\right), \tag{A.11}$$

$$e^{-6\Lambda} d(e^{6\Lambda} \cos \zeta J \wedge K^2) = e^{-3\Lambda} \sin \zeta G + m \left(J \wedge J - 2 \sin \zeta J \wedge K^1 \wedge K^2\right). \quad (A.12)$$

In the above formulas, \star stands for Hodge duality in the six-dimensional space spanned by M_4 and the one-forms K^1 and K^2 . G is an eleven-dimensional four-form whose components lie along the six-dimensional space that is transverse to AdS_5 ,⁷

$$G = -\partial_y e^{-6\Lambda} \widehat{\mathrm{vol}}_4 - \frac{e^{-9\Lambda}}{\cos\zeta} \left(\hat{\star}_4 d_4 e^{6\Lambda} \right) \wedge K^1 - \frac{\cos^3\zeta}{3m} \left(\hat{\star}_4 \partial_y \tilde{\rho} \right) \wedge K^2 - \left[\frac{e^{3\Lambda}}{3m} \cos^2\zeta \, \hat{\star}_4 d_4 \tilde{\rho} + 4m \, e^{-3\Lambda} \, \hat{J} \right] \wedge K^1 \wedge K^2 \,.$$
(A.13)

In this expression the hatted quantities are referred to the four-dimensional metric $\hat{g}^{(4)}_{\mu\nu} = e^{6\Lambda} g^{(4)}_{\mu\nu}$. Finally, d_4 is the exterior derivative on the four-dimensional space that is transverse to AdS_5 and K^1 , K^2 .

A.3 Recovering the non-Abelian T-dual from GMSW

Let us now find the explicit map between the GMSW geometry and the lifted non-Abelian T-dual geometry. In order to do this we first identify the functions Λ and ζ according to,

$$e^{6\Lambda} = 4y^2 + \frac{q}{9}, \qquad \cos\zeta = \sqrt{\frac{q}{36y^2 + q}},$$
 (A.14)

where q is a function of the coordinates on M_4 and y, determined below. We also take

$$\tilde{\rho} = -\frac{1}{6q} \, dw - \frac{1 - 12\,q}{12\,q} \, \cos\theta_1 \, d\phi_1 \; . \tag{A.15}$$

Then the one-forms K^1 and K^2 read,

$$K^{1} = \frac{3}{\sqrt{q}} dy, \qquad K^{2} = \frac{1}{3m} \sqrt{\frac{q}{36y^{2} + q}} \left[d\tilde{\psi} - \frac{1}{6q} dw - \frac{1 - 12q}{12q} \cos\theta_{1} d\phi_{1} \right].$$
(A.16)

Moreover, we define an orthogonal frame for the four-dimensional space M_4 ,

$$e^{1} = \frac{1}{\sqrt{6}} \sin \theta_{1} d\phi_{1}, \qquad e^{2} = \frac{1}{\sqrt{6}} d\theta_{1}$$

$$e^{3} = \frac{1}{18} \sqrt{\frac{36 q - 1}{36 y^{2} q + q^{2}}} dz, \qquad e^{4} = \frac{1}{18} \sqrt{\frac{36 q - 1}{36 y^{2} q + q^{2}}} \left(dw + \frac{1}{2} \cos \theta_{1} d\phi_{1} \right).$$
(A.17)

In the above expressions, q can be thought of as a function of z and y through the relation,

$$z - 162y^2 - \frac{36q - 1}{12} - \frac{1}{12}\ln(36q - 1) = 0.$$
 (A.18)

Solving this equation for q one finds,⁸

$$q = \frac{1}{36} \left[1 + \text{ProductLog} \left(e^{12(z-162y^2)} \right) \right].$$
(A.19)

⁷There is a sign difference between the first term in the second line of (A.13) and the corresponding term in eq. (2.50) of [65], that is due to our different conventions for Hodge duality.

⁸With ProductLog(\mathcal{Z}) we mean the solution of the equation $\mathcal{Z} = \mathcal{W} e^{\widetilde{\mathcal{W}}}$ in terms of \mathcal{W} .





Figure 7. Circular quiver associated to the Abelian T-dual solution and corresponding brane setup. There are N_4 D4-branes stretched between the NS5 and NS5' branes. NS5 and NS5'-branes are represented by transversal black and red dashed lines, respectively.

From the above frame one can construct the forms J and Ω of the SU(2) structure on M_4 as,

$$J = e^{1} \wedge e^{2} + e^{3} \wedge e^{4}, \qquad \Omega = e^{i\psi} \left(e^{1} + ie^{2} \right) \wedge \left(e^{3} + ie^{4} \right). \tag{A.20}$$

Both the metric and the 4-form flux associated to our solution are then obtained after identifying, 9

$$y = \frac{\rho \cos \chi}{6}, \qquad w = 9 x_{11} + \frac{\xi}{6}, \qquad \tilde{\psi} = \xi$$
 (A.21)

and

$$q = \frac{1}{36} + \frac{3}{2} \rho^2 \sin^2 \chi .$$
 (A.22)

One can also check that with the above definitions the constraints (A.9)-(A.12), proving that the solution of appendix A.1 fits into the class of solutions found in [65], are satisfied.

B The Abelian T-dual of the Klebanov-Witten solution

The Abelian T-dual, Type IIA description, of the Klebanov-Witten theory is particularly useful for the study of certain properties of this theory [63, 64]. One interesting aspect is that the field theory can directly be read from the D4, NS5, NS5' brane set-up associated to this solution. We have depicted both the brane set-up and the associated quiver in figure 7.

In this appendix we discuss some aspects of this description that are relevant for the understanding of the CFT interpretation of the non-Abelian T-dual solution, the main objective of this work.

B.1 Background

The paper [63] considered an Abelian T-duality transformation along the Hopf-fiber direction of the $T^{1,1}$. This dualization gives rise to a well-defined string theory background.

⁹We take $L = m = \alpha' = g_s = 1$ for convenience. There is a minus overall sign between G, from (A.13), and F_4 , from (A.3), due to our different conventions.

It is however a typical example of Supersymmetry without supersymmetry [78], being the low energy supergravity background non-supersymmetric. Since our ultimate goal in this section will be to compare with the non-Abelian T-dual solution, which is only guaranteed to be a well-defined string theory background at low energies, we will instead dualize along the ϕ_2 azimuthal direction of the $T^{1,1}$. This preserves the $\mathcal{N} = 1$ supersymmetry of the Klebanov-Witten solution, and can be matched directly with the non-Abelian T-dual solution in the large ρ limit.

We start by rewriting the Klebanov-Witten metric in terms of the T-duality preferred frame, in which ϕ_2 does only appear in the form $d\phi_2$ and just in one vielbein,

$$e^{x^{\mu}} = \frac{r}{L} dx^{\mu}, \qquad e^{r} = \frac{L}{r} dr, \qquad e^{1} = L \lambda_{1} d\theta_{1}, \qquad e^{2} = L \lambda_{1} \sin \theta_{1} d\phi_{1},$$
$$e^{\hat{1}} = L \lambda_{2} d\theta_{2}, \qquad e^{\hat{2}} = L \lambda \lambda_{2} \frac{\sin \theta_{2}}{\sqrt{P(\theta_{2})}} \left(d\psi + \cos \theta_{1} d\phi_{1} \right),$$
$$e^{3} = e^{C} \left(d\phi_{2} + \tilde{A}_{1} \right), \qquad (B.1)$$

where $e^{2C} = L^2 P(\theta_2)$ with $P(\theta_2) = \lambda^2 \cos^2 \theta_2 + \lambda_2^2 \sin^2 \theta_2$, and we have introduced the connection

$$\tilde{A}_1 = \frac{\lambda^2 \cos \theta_2}{P(\theta_2)} \left(d\psi + \cos \theta_1 \, d\phi_1 \right) \, .$$

The Klebanov-Witten metric thus reads

$$ds^{2} = ds_{AdS_{5}}^{2} + L^{2} \left[\lambda_{1}^{2} d\Omega_{2}^{2}(\theta_{1}, \phi_{1}) + \lambda_{2}^{2} \left(d\theta_{2}^{2} + \frac{\lambda^{2} \sin^{2} \theta_{2}}{P(\theta_{2})} \left(d\psi + \cos \theta_{1} d\phi_{1} \right)^{2} \right) + P(\theta_{2}) \left(d\phi_{2} + \frac{\lambda^{2} \cos \theta_{2}}{P(\theta_{2})} \left(d\psi + \cos \theta_{1} d\phi_{1} \right) \right)^{2} \right].$$
(B.2)

A U(1) T-duality performed on the ϕ_2 direction trades the vielbein e^3 for $\hat{e} = \alpha' e^{-C} d\phi_2$, and generates a NS-NS 2-form $B_2 = \alpha' \tilde{A}_1 \wedge d\phi_2$. The NS-NS sector for the dual solution is then given by:¹⁰

$$ds_{\text{ATD}}^{2} = ds_{AdS_{5}}^{2} + L^{2}\lambda_{1}^{2} \left[d\Omega_{2}^{2}(\theta_{1},\phi_{1}) + \left(d\theta_{2}^{2} + \frac{\lambda^{2} \sin^{2} \theta_{2}}{P(\theta_{2})} \left(d\psi + \cos \theta_{1} d\phi_{1} \right)^{2} \right) + \frac{d\phi_{2}^{2}}{\lambda_{1}^{2} P(\theta_{2})} \right],$$

$$B_{2}^{\text{ATD}} = -\frac{L^{2} \lambda^{2} \cos \theta_{2}}{P(\theta_{2})} \left(d\phi_{2} \wedge d\psi + \cos \theta_{1} d\phi_{2} \wedge d\phi_{1} \right),$$

$$e^{-2\Phi_{\text{ATD}}} = \frac{L^{2}}{g_{s}^{2} \alpha'} P(\theta_{2}).$$
(B.3)

We can see in the metric the geometrical realization of the U(1) R-symmetry in the ψ direction. We can also see that it agrees with the asymptotic form of the metric of the non-Abelian T-dual solution, given by the first equation in (2.18), under the replacements

$$\chi \to \theta_2, \qquad \xi \to \psi, \qquad \rho \to \phi_2.$$
 (B.4)

¹⁰We rescale $\phi_2 \rightarrow \frac{L^2}{\alpha'}\phi_2$, so that the metric of the internal space scales with L^2 . We also use that $\lambda_2 = \lambda_1$ for later comparison with the NATD solution.

The B_2 fields do also agree, once a gauge transformation of parameter

$$\Lambda = -L^2 \cos \theta_2 \,\phi_2 \Big(d\psi + \frac{\lambda^2 \cos \theta_1}{P(\theta_2)} d\phi_1 \Big) \tag{B.5}$$

is performed, giving rise to

$$B_{2} = -L^{2}\phi_{2}\left[d\Omega_{2}(\theta_{2},\psi) + \frac{\lambda^{2}\cos\theta_{2}}{P(\theta_{2})}d\Omega_{2}(\theta_{1},\phi_{1}) - \lambda^{2}\cos\theta_{1}\partial_{\theta_{2}}\left(\frac{\cos\theta_{2}}{P(\theta_{2})}\right)d\theta_{2} \wedge d\phi_{1}\right] + \frac{L^{2}\sin\theta_{2}}{2P(\theta_{2})}\left(\lambda^{2} - \lambda_{1}^{2}\right)\sin2\theta_{2}\,d\psi \wedge d\phi_{2}\,.$$
(B.6)

We will use this expression for the B_2 -field in the remaining of this section. As in [44, 45], the two dilatons satisfy $e^{-2\Phi_{NATD}} \approx \rho^2 e^{-2\Phi_{ATD}}$ for large ρ (after re-absorbing the scaling factors in $\rho \rightarrow \frac{\alpha'}{L^2}\rho$). As explained in [44, 45], this relation has its origin in the different measures in the partition functions of the non-Abelian and Abelian T-dual sigma models.

Finally, the RR fields are:

$$F_{4} = \frac{4L^{4}\lambda\lambda_{1}^{4}}{g_{s}\alpha'^{1/2}}\sin\theta_{1}\sin\theta_{2} \ d\theta_{1}\wedge d\phi_{1}\wedge d\theta_{2}\wedge d\psi,$$

$$F_{6} = \frac{4L}{q_{s}\alpha'^{1/2}}\operatorname{Vol}_{AdS_{5}}\wedge d\phi_{2}.$$
(B.7)

One can check that, as in [45], for large ρ the fluxes polyforms satisfy

$$e^{\Phi_{NATD}}F_{NATD} \approx e^{\Phi_{ATD}}F_{ATD}$$
. (B.8)

The previous relations show that the non-Abelian T-dual solution reduces in the $\rho \to \infty$ limit to the Abelian T-dual one. This connection between non-Abelian and Abelian Tduals was discussed previously in examples where the dualization took place on a round S^3 [44, 45]. Our results show that it extends more generally. It is worth stressing however that in this case the relation is more subtle globally. Indeed, the relations in eq. (B.4) identify $\xi \in [0, 2\pi]$ with $\psi \in [0, 4\pi]$. The reason for this apparent mismatch is that the dualization on ϕ_2 generates a bolt singularity in the metric, and this must be cured by setting $\psi \in [0, 2\pi]$, such that the bolt singularity reduces to the coordinate singularity of \mathbb{R}^2 written in polar coordinates. Once this is taken into account the ranges of both coordinates also agree. As encountered in [12, 13], the dualization has enforced a \mathbb{Z}_2 quotient on ψ . Our Abelian T-dual is thus describing the Klebanov-Witten theory modded by \mathbb{Z}_2 . This is consistent with the brane set-up that is implied by the quantized charges of the background, as we now show.

B.2 Quantized charges and brane set-up

The background fluxes of the Abelian T-dual solution support D4 and NS5-brane charges. The Page charge for the D4 branes is given by:

$$Q_{D4} = \frac{1}{2\kappa_{10}^2 T_{D4}} \int_{M_4} F_4 = \frac{2}{27} \frac{L^4}{\pi g_s^2 \alpha'^2} = N_4 .$$
(B.9)

Imposing the quantization of this charge we find that the radius L is related to the number of D4 branes through the formula:

$$L^4 = \frac{27}{2} \pi g_s^2 \, \alpha'^2 \, N_4 \; . \tag{B.10}$$

We find a factor of 2 of difference with respect to the original background. This is due to the change in the periodicity of the ψ direction from $[0, 4\pi]$ to $[0, 2\pi]$.

In turn, the charge of NS5 branes is calculated from:

$$Q_{NS5} = \frac{1}{4\pi^2 \,\alpha'} \int_{M_3} H_3 \,. \tag{B.11}$$

As in section 2.2.2, we can define two 3-cycles: $\Sigma_3 = [\phi_2, \theta_2, \psi]$ and $\Sigma'_3 = [\phi_2, \theta_1, \phi_1]_{\theta_2=0}$. Taking M_3 to be any of these cycles we find that there are two units of NS5, or NS5', charge. This is consistent with a brane picture of two alternating NS5, NS5' branes, transverse to either of the two 2-cycles $\tilde{S}^2(\theta_2, \psi)$, $S^2(\theta_1, \phi_1)$, located along the compact ϕ_2 -direction. This is the brane set-up discussed in [64], describing the Klebanov-Witten theory modded by \mathbb{Z}_2 in Type IIA. The general \mathbb{Z}_k case is depicted in figure 9 of appendix C.3. Note that, as discussed in [64], the positions of the branes in the ϕ_2 -circle are not specified by the geometry, so generically we can only think that they define four intervals in the ϕ_2 -circle.¹¹ The same number of D4-branes are stretched between each pair of NS5, NS5' branes since even if large gauge transformations are required as we pass the value $\phi_2 = \pi L^2/\alpha'$, the D4-brane charge does not change in the absence of F_2 -flux.

Coming back to section 2.2.3, the relation found there between the central charges of the non-Abelian and Abelian T-dual solutions helps us understand now the connection between ρ and ϕ_2 globally. The computation in that section showed that the central charges agree when $\rho \in [n\pi \frac{L^2}{\alpha'}, (n+1)\pi \frac{L^2}{\alpha'}]$ and n is sent to infinity. This is consistent with the $\rho \to \infty$ limit that must be taken at the level of the solutions. Furthermore, it clarifies why globally the ρ direction is identified, through the replacements in (B.4), with $\phi_2 \in [0, 2\pi \frac{L^2}{\alpha'}]$. This is just implied by the \mathbb{Z}_2 quotient enforced by the Abelian T-duality transformation.

C Some field theory elaborations

In this appendix we discuss some aspects of the field theory analysis presented in section 4. We start with the calculation of the beta functions and anomalies for the Klebanov-Witten CFT.

C.1 A summary of the Klebanov-Witten CFT

The field content of the Klebanov-Witten theory consists on a $SU(N) \times SU(N)$ gauge group with bifundamental matter fields A_1 , A_2 and B_1 , B_2 , transforming in the (N, \bar{N})

¹¹In [64] it was argued that the different orderings correspond to different phases in the Kähler moduli space of the orbifold singularity. This is interpreted in the field theory side in terms of Seiberg duality [79, 80], so the corresponding theories should flow to the same CFT in the infrared.



Figure 8. The KW quiver.

and (\bar{N}, N) representations of SU(N), respectively. This theory is represented by the quiver depicted in figure 8. The anomalous dimensions of the matter fields are,

$$\gamma_{A_i} = \gamma_{B_i} = -\frac{1}{2}, \qquad (C.1)$$

and thus the physical dimensions and the R-charges are given by,

$$\dim(A_i) = \dim(B_i) = 1 - \frac{1}{4} = \frac{3}{4},$$

$$R_{[A]} = R_{[B]} = \frac{1}{2}, \qquad R_{\Psi_A} = R_{\Psi_B} = -\frac{1}{2}.$$
(C.2)

Substituting in eq. (4.1) we see that the
$$\beta$$
-functions for the couplings g_1 and g_2 vanish:

$$\beta_i \sim 3N - 2N\left[1 - \left(-\frac{1}{2}\right)\right] = 0, \qquad i = 1, 2.$$
 (C.3)

We can also check the vanishing of the anomaly,

$$\Delta \theta_i = 2N + 2(2N)\left(-\frac{1}{2}\right) = 0, \qquad (C.4)$$

where we took into account that the R-charge of the gaugino is 1 while that of the two Weyl fermions is -1/2.

We hope that this has prepared the reader unfamiliar with these formalities to understand the material in our section 4.

C.2 Central Charge of the $\mathcal{N} = 2$ UV CFT

In this appendix we compute the central charge of the $\mathcal{N} = 2$ quiver associated to the non-Abelian T-dual of $AdS_5 \times S^5/\mathbb{Z}_2$, using *a*-maximization. We obtain that the central charge is maximized for $\epsilon = \frac{1}{3}$, as for the equal rank quivers considered in [55]. Furthermore, we show that the result of this calculation leads, consistently, to the holographic central charge given by eq. (4.20).

We consider the \mathbb{Z}_2 -reflection of the quiver of figure 6 and take $\sigma_i = +1$ for all hypermultiplets, including the ones associated with the flavor groups. We then find for the trace anomalies $(N \equiv N_6)$:

$$\operatorname{Tr} R_{\epsilon}(H) = 2 \sum_{j=1}^{n-1} \operatorname{Tr} R_{\epsilon}(H_j) + \sum_{i=1}^{2} \operatorname{Tr} R_{\epsilon}(H_{f_i})$$
$$= 2N^2 \sum_{j=1}^{n-1} j (j+1) (\epsilon \sigma_j - 1) + n N^2 \sum_{i=1}^{2} (\epsilon \sigma_{f_i} - 1)$$

$$= N^{2} \left[\frac{2}{3} (n^{3} - n) (\epsilon - 1) + 2n(\epsilon - 1) \right]$$

= $N^{2} \left[\frac{2}{3} n^{3} + \frac{4}{3} n \right] (\epsilon - 1) \approx \frac{2}{3} n^{3} N^{2} (\epsilon - 1) + \mathcal{O}(n),$ (C.5)

as well as

$$\operatorname{Tr} R_{\epsilon}^{3}(H) = 2 \sum_{j=1}^{n-1} \operatorname{Tr} R_{\epsilon}^{3}(H_{j}) + \sum_{i=1}^{2} \operatorname{Tr} R_{\epsilon}^{3}(H_{f_{i}})$$

$$= 2 N^{2} \sum_{j=1}^{n-1} j \left(j+1\right) \frac{\left(\epsilon-1\right)^{3}}{4} + n N^{2} \sum_{i=1}^{2} \frac{\left(\epsilon \sigma_{f_{i}}-1\right)^{3}}{4}$$

$$= \frac{N^{2}}{4} \left[\frac{2}{3} \left(n^{3}-n\right) \left(\epsilon-1\right)^{3} + 2n \left(\epsilon-1\right)^{3}\right]$$

$$= N^{2} \left[\frac{1}{6}n^{3} + \frac{1}{3}n\right] \left(\epsilon-1\right)^{3} \approx \frac{1}{6} n^{3} N^{2} \left(\epsilon-1\right)^{3} + \mathcal{O}(n) . \quad (C.6)$$

For the $\mathcal{N} = 2$ vector multiplets ($\mathcal{N} = 1$ vector + chiral adjoint) the non-anomalous Rcharge $R_{\epsilon} = R_0 + \epsilon \mathcal{F}/2$ is obtained from the R-charge for the gaugino, $R_0(\lambda) = 1$, plus the non-anomalous flavor charge of the fermion in the chiral adjoint $\mathcal{F}(\psi_j) = (-1)(\sigma_{j-1} + \sigma_j)$, being $R_0(\psi_j) = 0$. We thus have:

$$\operatorname{Tr} R_{\epsilon}(V_{j}) = \left(N_{j}^{2} - 1\right) \left(1 - \frac{1}{2} \epsilon \left(\sigma_{j-1} + \sigma_{j}\right)\right),$$

$$\operatorname{Tr} R_{\epsilon}^{3}(V_{j}) = \left(N_{j}^{2} - 1\right) \left(1 - \frac{1}{8} \epsilon^{3} \left(\sigma_{j-1} + \sigma_{j}\right)^{3}\right).$$
 (C.7)

These are summed up easily for all $\sigma_j = +1$:

$$\operatorname{Tr} R_{\epsilon}(V) = 2 \sum_{j=1}^{n-1} \operatorname{Tr} R_{\epsilon}(V_j) + \operatorname{Tr} R_{\epsilon}(V_n) = \left[2 \sum_{j=1}^{n-1} \left(j^2 N^2 - 1 \right) + \left(n^2 N^2 - 1 \right) \right] \left(1 - \epsilon \right)$$

$$= \left[\frac{1}{3}(2n^{3}+n)N^{2}-(2n-1)\right](1-\epsilon) \approx \frac{2}{3}n^{3}N^{2}(1-\epsilon) + \mathcal{O}(n).$$
 (C.8)

The cubic term follows most readily:

$$\operatorname{Tr} R^{3}_{\epsilon}(V) = \left[\frac{1}{3} \left(2n^{3} + n\right) N^{2} - (2n - 1)\right] \left(1 - \epsilon^{3}\right) \approx \frac{2}{3} n^{3} N^{2} \left(1 - \epsilon^{3}\right) + \mathcal{O}(n) . \quad (C.9)$$

We thus see that both linear contributions (C.5) and (C.8) from the hypermultiplets and vector multiplets cancel at leading order, so that

$$\operatorname{Tr} R_{\epsilon} \equiv \operatorname{Tr} R_{\epsilon}(H) + \operatorname{Tr} R_{\epsilon}(V) \approx \mathcal{O}(n) \; .$$

Now both $a(\epsilon)$ and $c(\epsilon)$ charges can be computed exactly,

$$a(\epsilon) = \frac{3}{64} (1-\epsilon) \left\{ \left[3n^3(1+\epsilon)^2 + 2(1+3\epsilon)n \right] N_6^2 - 2(2n-1) \left(2+3\epsilon(1+\epsilon) \right) \right\},\$$

$$c(\epsilon) = \frac{1}{64} (1-\epsilon) \left\{ \left[9n^3(1+\epsilon)^2 + 2(5+9\epsilon)n \right] N_6^2 - 2(2n-1) \left(4+9\epsilon(1+\epsilon) \right) \right\},$$
 (C.10)



Figure 9. Circular quiver associated to the KW theory modded by \mathbb{Z}_k , and corresponding brane set-up. There are N_4 D4-branes stretched between p = k NS5-branes, labeled by $r = 1, \ldots, k$ (as for the corresponding hypermultiplets) and q = k NS5'-branes labeled by $s = 1, \ldots, k$. NS5 and NS5'-branes are represented by transversal black and red dashed lines, respectively.

and $a(\epsilon)$ is maximized for $\epsilon = 1/3$, yielding the superconformal charges:

$$a_{\mathcal{N}=2} \equiv a(\epsilon = 1/3) = \frac{1}{24} \left\{ (4n^3 + 3n)N_6^2 - 10n + 5 \right\},\$$

$$c_{\mathcal{N}=2} \equiv c(\epsilon = 1/3) = \frac{1}{6} \left\{ (n^3 + n)N_6^2 - 2n + 1 \right\}.$$
 (C.11)

In the long quiver approximation, we recover the holographic result

$$c_{\mathcal{N}=2} \approx a_{\mathcal{N}=2} \approx \frac{1}{6} n^3 N_6^2 + \mathcal{O}(n) , \qquad (C.12)$$

as expected. It is noteworthy that $\epsilon = 1/3$ is the value of ϵ predicted in [55] for $\mathcal{N} = 2$ quivers with nodes of the same rank.

C.3 Central charge of the Klebanov-Witten theory modded by \mathbb{Z}_k

In this appendix we include, for completeness, the field theory calculation of the central charge of the Klebanov-Witten theory, using *a*-maximization. We will center in the more general case in which the theory is modded by \mathbb{Z}_k . The computation of the field theoretical central charge in this example is very illustrative of the *a*-maximization technique used throughout the paper.

In this case we have, in the Type IIA description, p = k NS5-branes and q = k NS5'branes, and $\ell = p + q = 2k$ hypermultiplets connecting $\ell \mathcal{N} = 1$ vector multiplets [64]. The first and the last nodes are made to coincide, as depicted in Figure 9. We closely follow the field-theoretical computation of the central charge for the linear quiver proposed in section 4.3. We just need to take $N_a = N_b = N_4$ for the bifundamentals in (4.10) for all the $\ell = 2k$ nodes. This yields the linear contribution for the hypermultiplets:

$$\operatorname{Tr} R_{\epsilon}(H) \equiv \sum_{j=1}^{\ell} \operatorname{Tr} R_{\epsilon}(H_j) = \sum_{r=1}^{k} \operatorname{Tr} R_{\epsilon}(H_r) + \sum_{s=1}^{k} \operatorname{Tr} R_{\epsilon}(H_s)$$
$$= \ell N_4^2 (z \epsilon - 1) \stackrel{z=0}{=} -2 k N_4^2 ,$$

$$\operatorname{Tr} R_{\epsilon}^{3}(H) = \frac{\ell}{4} N_{4}^{2} \left(z \left(\epsilon^{3} + 3 \epsilon \right) - 3 \epsilon^{2} - 1 \right) \Big|_{z=0} = -\frac{k}{2} N_{4}^{2} \left(3 \epsilon^{2} + 1 \right) \,.$$

Contributions from $\mathcal{N} = 1$ vector multiplets are computed straightforwardly to be:

Tr
$$R_{\epsilon}(V)$$
 = Tr $R_{\epsilon}^{3}(V) = \ell (N_{4}^{2} - 1) = 2 k (N_{4}^{2} - 1).$

We can now use (4.9) to get

$$c(\epsilon) = \frac{\ell}{128} \left(27 N_4^2 \left(1 - \epsilon^2 \right) - 16 \right),$$
 (C.13)

which, upon a-maximization for $\epsilon = 0$, yields the fixed point central charge (for large N_4):

$$c \approx \frac{27}{64} k N_4^2 \,,$$
 (C.14)

which coincides, as expected, with the holographic value (given by eq. (2.36) for k = 1). This expression is valid for any $k \ge 1$, i.e. no large $\ell = 2k$ limit has been assumed.

Note that in the absence of flavor groups it is not possible to define σ_0 , σ_ℓ , and neither $\kappa = \sigma_0 + \sigma_\ell$, as we have done for the linear quivers discussed in section 3. Still, the result in (C.14) agrees with the central charge of a Bah-Bobev type of linear quiver (see eq. (3.20) in [55]) for $\kappa = 0$ and large ℓ . Indeed, even if there is no clear definition for κ in this case, the uplift of the circular brane set-up is interpreted as M5-branes wrapping a torus ($\kappa = 0$) with minimal punctures, as the gauging of the end flavor groups of the linear quiver corresponds in M-theory to gluing the two left-over maximal punctures, closing up the Riemann surface.

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- X.C. de la Ossa and F. Quevedo, Duality symmetries from non-Abelian isometries in string theory, Nucl. Phys. B 403 (1993) 377 [hep-th/9210021] [INSPIRE].
- [2] E. Alvarez, L. Álvarez-Gaumé, J.L.F. Barbon and Y. Lozano, Some global aspects of duality in string theory, Nucl. Phys. B 415 (1994) 71 [hep-th/9309039] [INSPIRE].
- [3] E. Alvarez, L. Álvarez-Gaumé and Y. Lozano, On non-Abelian duality, Nucl. Phys. B 424 (1994) 155 [hep-th/9403155] [INSPIRE].
- [4] E. Alvarez, L. Álvarez-Gaumé and Y. Lozano, A canonical approach to duality transformations, Phys. Lett. B 336 (1994) 183 [hep-th/9406206] [INSPIRE].
- [5] S. Elitzur, A. Giveon, E. Rabinovici, A. Schwimmer and G. Veneziano, *Remarks on non-Abelian duality*, *Nucl. Phys.* B 435 (1995) 147 [hep-th/9409011] [INSPIRE].
- [6] C. Klimčík and P. Ševera, Dual non-Abelian duality and the Drinfeld double, Phys. Lett. B 351 (1995) 455 [hep-th/9502122] [INSPIRE].

- Y. Lozano, NonAbelian duality and canonical transformations, Phys. Lett. B 355 (1995) 165 [hep-th/9503045] [INSPIRE].
- [8] K. Sfetsos, Non-Abelian duality, parafermions and supersymmetry, Phys. Rev. D 54 (1996) 1682 [hep-th/9602179] [INSPIRE].
- K. Sfetsos and D.C. Thompson, On non-abelian T-dual geometries with Ramond fluxes, Nucl. Phys. B 846 (2011) 21 [arXiv:1012.1320] [INSPIRE].
- [10] J.M. Maldacena, The large-N limit of superconformal field theories and supergravity, Int. J. Theor. Phys. 38 (1999) 1113 [Adv. Theor. Math. Phys. 2 (1998) 231] [hep-th/9711200]
 [INSPIRE].
- [11] Y. Lozano, E.Ó Colgáin, D. Rodriguez-Gomez and K. Sfetsos, Supersymmetric AdS₆ via T Duality, Phys. Rev. Lett. 110 (2013) 231601 [arXiv:1212.1043] [INSPIRE].
- [12] G. Itsios, C. Núñez, K. Sfetsos and D.C. Thompson, Non-Abelian T-duality and the AdS/CFT correspondence:new N = 1 backgrounds, Nucl. Phys. B 873 (2013) 1 [arXiv:1301.6755] [INSPIRE].
- [13] G. Itsios, C. Núñez, K. Sfetsos and D.C. Thompson, On non-Abelian T-duality and new N = 1 backgrounds, Phys. Lett. B 721 (2013) 342 [arXiv:1212.4840] [INSPIRE].
- [14] Y. Lozano, E. O Colgain, K. Sfetsos and D.C. Thompson, Non-abelian T-duality, Ramond fields and coset geometries, JHEP 06 (2011) 106 [arXiv:1104.5196] [INSPIRE].
- [15] G. Itsios, Y. Lozano, E. O Colgain and K. Sfetsos, Non-abelian T-duality and consistent truncations in type-II supergravity, JHEP 08 (2012) 132 [arXiv:1205.2274] [INSPIRE].
- [16] J. Jeong, O. Kelekci and E.Ó Colgáin, An alternative IIB embedding of F(4) gauged supergravity, JHEP 05 (2013) 079 [arXiv:1302.2105] [INSPIRE].
- [17] A. Barranco et al., G-structures and flavouring non-Abelian T-duality, JHEP 08 (2013) 018
 [arXiv:1305.7229] [INSPIRE].
- [18] N.T. Macpherson, Non-Abelian T-duality, G₂-structure rotation and holographic duals of N = 1 Chern-Simons theories, JHEP 11 (2013) 137 [arXiv:1310.1609] [INSPIRE].
- [19] Y. Lozano, E.Ó Colgáin and D. Rodriguez-Gomez, Hints of 5d Fixed Point Theories from Non-Abelian T-duality, JHEP 05 (2014) 009 [arXiv:1311.4842] [INSPIRE].
- [20] J. Gaillard, N.T. Macpherson, C. Núñez and D.C. Thompson, Dualising the baryonic branch: dynamic SU(2) and confining backgrounds in IIA, Nucl. Phys. B 884 (2014) 696 [arXiv:1312.4945] [INSPIRE].
- [21] D. Elander, A.F. Faedo, C. Hoyos, D. Mateos and M. Piai, Multiscale confining dynamics from holographic RG flows, JHEP 05 (2014) 003 [arXiv:1312.7160] [INSPIRE].
- [22] S. Zacarias, Semiclassical strings and non-Abelian T-duality, Phys. Lett. B 737 (2014) 90 [arXiv:1401.7618] [INSPIRE].
- [23] E. Caceres, N.T. Macpherson and C. Núñez, New type IIB backgrounds and aspects of their field theory duals, JHEP 08 (2014) 107 [arXiv:1402.3294] [INSPIRE].
- [24] P.M. Pradhan, Oscillating strings and non-abelian T-dual Klebanov-Witten background, Phys. Rev. D 90 (2014) 046003 [arXiv:1406.2152] [INSPIRE].
- [25] Y. Lozano and N.T. Macpherson, A new AdS₄/CFT₃ dual with extended SUSY and a spectral flow, JHEP 11 (2014) 115 [arXiv:1408.0912] [INSPIRE].

- [26] K. Sfetsos and D.C. Thompson, New $\mathcal{N} = 1$ supersymmetric AdS₅ backgrounds in Type IIA supergravity, JHEP **11** (2014) 006 [arXiv:1408.6545] [INSPIRE].
- [27] O. Kelekci, Y. Lozano, N.T. Macpherson and E.Ó Colgáin, Supersymmetry and non-Abelian T-duality in type-II supergravity, Class. Quant. Grav. 32 (2015) 035014 [arXiv:1409.7406] [INSPIRE].
- [28] N.T. Macpherson et al., Type IIB supergravity solutions with AdS₅ from Abelian and non-Abelian T dualities, JHEP 02 (2015) 040 [arXiv:1410.2650] [INSPIRE].
- [29] K.S. Kooner and S. Zacarias, Non-abelian T-dualizing the resolved conifold with regular and fractional D3-branes, JHEP 08 (2015) 143 [arXiv:1411.7433] [INSPIRE].
- [30] T.R. Araujo and H. Nastase, $\mathcal{N} = 1$ SUSY backgrounds with an AdS factor from non-Abelian T duality, Phys. Rev. D 91 (2015) 126015 [arXiv:1503.00553] [INSPIRE].
- [31] Y. Bea et al., Compactifications of the Klebanov-Witten CFT and new AdS₃ backgrounds, JHEP 05 (2015) 062 [arXiv:1503.07527] [INSPIRE].
- [32] Y. Lozano, N.T. Macpherson, J. Montero and E.Ó Colgáin, New $AdS_3 \times S^2$ T-duals with $\mathcal{N} = (0, 4)$ supersymmetry, JHEP 08 (2015) 121 [arXiv:1507.02659] [INSPIRE].
- [33] Y. Lozano, N.T. Macpherson and J. Montero, $A \mathcal{N} = 2$ supersymmetric AdS_4 solution in *M*-theory with purely magnetic flux, *JHEP* **10** (2015) 004 [arXiv:1507.02660] [INSPIRE].
- [34] T.R. Araujo and H. Nastase, Non-Abelian T-duality for nonrelativistic holographic duals, JHEP 11 (2015) 203 [arXiv:1508.06568] [INSPIRE].
- [35] N.T. Macpherson, C. Núñez, D.C. Thompson and S. Zacarias, Holographic flows in non-Abelian T-dual geometries, JHEP 11 (2015) 212 [arXiv:1509.04286] [INSPIRE].
- [36] L.A. Pando Zayas, V.G.J. Rodgers and C.A. Whiting, Supergravity solutions with AdS₄ from non-Abelian T-dualities, JHEP 02 (2016) 061 [arXiv:1511.05991] [INSPIRE].
- [37] L.A. Pando Zayas, D. Tsimpis and C.A. Whiting, A supersymmetric IIB background with AdS₄ from massive IIA, Phys. Rev. D 96 (2017) 046013 [arXiv:1701.01643] [INSPIRE].
- [38] F. Apruzzi, M. Fazzi, A. Passias, D. Rosa and A. Tomasiello, AdS₆ solutions of type-II supergravity, JHEP 11 (2014) 099 [Erratum ibid. 05 (2015) 012] [arXiv:1406.0852] [INSPIRE].
- [39] I. Bah and V. Stylianou, Gravity duals of N = (0, 2) SCFTs from M5-branes, arXiv:1508.04135 [INSPIRE].
- [40] O. Kelekci, Y. Lozano, J. Montero, E.Ó Colgáin and M. Park, Large superconformal near-horizons from M-theory, Phys. Rev. D 93 (2016) 086010 [arXiv:1602.02802] [INSPIRE].
- [41] C. Couzens, Supersymmetric AdS₅ solutions of type IIB supergravity without D3 branes, JHEP 01 (2017) 041 [arXiv:1609.05039] [INSPIRE].
- [42] Y. Terashima and M. Yamazaki, SL(2, ℝ) Chern-Simons, Liouville and gauge theory on duality walls, JHEP 08 (2011) 135 [arXiv:1103.5748] [INSPIRE].
- [43] Y. Terashima and M. Yamazaki, Semiclassical analysis of the 3d/3d relation, Phys. Rev. D 88 (2013) 026011 [arXiv:1106.3066] [INSPIRE].
- [44] Y. Lozano and C. Núñez, Field theory aspects of non-Abelian T-duality and N = 2 linear quivers, JHEP 05 (2016) 107 [arXiv:1603.04440] [INSPIRE].

- [45] Y. Lozano, N.T. Macpherson, J. Montero and C. Núñez, Three-dimensional N = 4 linear quivers and non-Abelian T-duals, JHEP 11 (2016) 133 [arXiv:1609.09061] [INSPIRE].
- [46] Y. Lozano, C. Núñez and S. Zacarias, BMN vacua, superstars and non-Abelian T-duality, arXiv:1703.00417 [INSPIRE].
- [47] J.M. Maldacena and C. Núñez, Supergravity description of field theories on curved manifolds and a no go theorem, Int. J. Mod. Phys. A 16 (2001) 822 [hep-th/0007018] [INSPIRE].
- [48] B. Assel, C. Bachas, J. Estes and J. Gomis, Holographic duals of D = 3 N = 4 superconformal field theories, JHEP 08 (2011) 087 [arXiv:1106.4253] [INSPIRE].
- [49] B. Assel, C. Bachas, J. Estes and J. Gomis, *IIB duals of* D = 3 N = 4 *circular quivers*, *JHEP* **12** (2012) 044 [arXiv:1210.2590] [INSPIRE].
- [50] O. Aharony, L. Berdichevsky, M. Berkooz and I. Shamir, Near-horizon solutions for D3-branes ending on 5-branes, Phys. Rev. D 84 (2011) 126003 [arXiv:1106.1870] [INSPIRE].
- [51] E. D'Hoker, J. Estes, M. Gutperle and D. Krym, Exact half-BPS flux solutions in M-theory. I: local solutions, JHEP 08 (2008) 028 [arXiv:0806.0605] [INSPIRE].
- [52] D. Gaiotto and E. Witten, S-duality of boundary conditions in N = 4 super Yang-Mills theory, Adv. Theor. Math. Phys. 13 (2009) 721 [arXiv:0807.3720] [INSPIRE].
- [53] F. Leblond, R.C. Myers and D.C. Page, Superstars and giant gravitons in M-theory, JHEP 01 (2002) 026 [hep-th/0111178] [INSPIRE].
- [54] I.R. Klebanov and E. Witten, Superconformal field theory on three-branes at a Calabi-Yau singularity, Nucl. Phys. B 536 (1998) 199 [hep-th/9807080] [INSPIRE].
- [55] I. Bah and N. Bobev, Linear quivers and $\mathcal{N} = 1$ SCFTs from M5-branes, JHEP 08 (2014) 121 [arXiv:1307.7104] [INSPIRE].
- [56] E. Witten, Solutions of four-dimensional field theories via M-theory, Nucl. Phys. B 500 (1997) 3 [hep-th/9703166] [INSPIRE].
- [57] D. Gaiotto, N = 2 dualities, JHEP 08 (2012) 034 [arXiv:0904.2715] [INSPIRE].
- [58] I. Bah, C. Beem, N. Bobev and B. Wecht, AdS/CFT dual pairs from M5-branes on Riemann surfaces, Phys. Rev. D 85 (2012) 121901 [arXiv:1112.5487] [INSPIRE].
- [59] I. Bah, C. Beem, N. Bobev and B. Wecht, Four-dimensional SCFTs from M5-branes, JHEP 06 (2012) 005 [arXiv:1203.0303] [INSPIRE].
- [60] F. Benini, Y. Tachikawa and B. Wecht, Sicilian gauge theories and N = 1 dualities, JHEP 01 (2010) 088 [arXiv:0909.1327] [INSPIRE].
- [61] I. Bah and B. Wecht, New N = 1 superconformal field theories in four dimensions, JHEP 07 (2013) 107 [arXiv:1111.3402] [INSPIRE].
- [62] A. Hanany and E. Witten, Type IIB superstrings, BPS monopoles and three-dimensional gauge dynamics, Nucl. Phys. B 492 (1997) 152 [hep-th/9611230] [INSPIRE].
- [63] K. Dasgupta and S. Mukhi, Brane constructions, conifolds and M-theory, Nucl. Phys. B 551 (1999) 204 [hep-th/9811139] [INSPIRE].
- [64] A.M. Uranga, Brane configurations for branes at conifolds, JHEP 01 (1999) 022 [hep-th/9811004] [INSPIRE].
- [65] J.P. Gauntlett, D. Martelli, J. Sparks and D. Waldram, Supersymmetric AdS₅ solutions of M-theory, Class. Quant. Grav. 21 (2004) 4335 [hep-th/0402153] [INSPIRE].

- [66] I.R. Klebanov, D. Kutasov and A. Murugan, Entanglement as a probe of confinement, Nucl. Phys. B 796 (2008) 274 [arXiv:0709.2140] [INSPIRE].
- [67] S.S. Gubser, Einstein manifolds and conformal field theories, Phys. Rev. D 59 (1999) 025006 [hep-th/9807164] [INSPIRE].
- [68] R.C. Myers, Dielectric branes, JHEP 12 (1999) 022 [hep-th/9910053] [INSPIRE].
- [69] K.A. Intriligator and B. Wecht, The exact superconformal R symmetry maximizes a, Nucl. Phys. B 667 (2003) 183 [hep-th/0304128] [INSPIRE].
- [70] D. Anselmi, D.Z. Freedman, M.T. Grisaru and A.A. Johansen, Nonperturbative formulas for central functions of supersymmetric gauge theories, Nucl. Phys. B 526 (1998) 543
 [hep-th/9708042] [INSPIRE].
- [71] J. Borlaf and Y. Lozano, Aspects of T duality in open strings, Nucl. Phys. B 480 (1996) 239 [hep-th/9607051] [INSPIRE].
- [72] Y. Tachikawa and B. Wecht, Explanation of the central charge ratio 27/32 in four-dimensional renormalization group flows between superconformal theories, Phys. Rev. Lett. 103 (2009) 061601 [arXiv:0906.0965] [INSPIRE].
- [73] V.A. Novikov, M.A. Shifman, A.I. Vainshtein and V.I. Zakharov, Exact Gell-Mann-Low function of supersymmetric Yang-Mills theories from instanton calculus, Nucl. Phys. B 229 (1983) 381 [INSPIRE].
- [74] N. Arkani-Hamed and H. Murayama, Holomorphy, rescaling anomalies and exact β-functions in supersymmetric gauge theories, JHEP 06 (2000) 030 [hep-th/9707133] [INSPIRE].
- [75] M. Cvetič et al., Embedding AdS black holes in ten-dimensions and eleven-dimensions, Nucl. Phys. B 558 (1999) 96 [hep-th/9903214] [INSPIRE].
- [76] F. Apruzzi, M. Fazzi, A. Passias and A. Tomasiello, Supersymmetric AdS₅ solutions of massive IIA supergravity, JHEP 06 (2015) 195 [arXiv:1502.06620] [INSPIRE].
- [77] I. Bah, AdS₅ solutions from M5-branes on Riemann surface and D6-branes sources, JHEP
 09 (2015) 163 [arXiv:1501.06072] [INSPIRE].
- [78] M.J. Duff, H. Lü and C.N. Pope, Supersymmetry without supersymmetry, Phys. Lett. B 409 (1997) 136 [hep-th/9704186] [INSPIRE].
- [79] S. Elitzur, A. Giveon and D. Kutasov, Branes and N = 1 duality in string theory, Phys. Lett. B 400 (1997) 269 [hep-th/9702014] [INSPIRE].
- [80] J.T. Liu and R. Minasian, Black holes and membranes in AdS₇, Phys. Lett. B 457 (1999) 39 [hep-th/9903269] [INSPIRE].

3.6 $Mink_3 \times S^3$ solutions of type II supergravity



CLE IN PRESS JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.2 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-

and exhibiting different amounts of supersymmetry: consider the very incomplete list of [4–14] for $\mathcal{N} = 1$, [15–21] for $\mathcal{N} = 2$, [22] for $\mathcal{N} = 3$ and [23–26] for $\mathcal{N} = 4$. Solutions with $\mathcal{N} > 4$ where recently classified in [27], they are very restricted. One of the more prominent methods of finding AdS_d backgrounds is to find bosonic solutions with an AdS_d factor to the supersymmetry constraints, which also satisfy the Bianchi identities. As a consequence of various integrability theorems, such solutions automatically solve the equa-tions of motions. The Killing spinor equations reduce to constraints on the internal manifold, which can then be solved by means of G-structure and generalised geometrical techniques. The literature usually approaches this problem by assuming an AdS_d from the start. However we are also interested in solutions of relevance to flux compactifications and the broader definition of holography that includes non-conformal solutions. As such we shall consider assume Minkowski factor, in this case Mink₃, so that our results are more broadly applicable. Finding Minkowski solutions using G-structure techniques [28-31] or otherwise is by now quite a mature program, see [32–35] for some recent examples. Usually the aim is to preserve minimal or even no supersymmetry for phenomenological reason which makes the problem in general quite hard. We shall take inspiration from [36] and assume the existence of an S^3 factor in the metric. This will necessarily mean that we are dealing with at least $\mathcal{N} = 2$ which is of less phenomenological interest, however with these solutions classified it should then be possible to systematically break some (or even all) of this symmetry by deforming the S^3 . In this paper we classify all supersymmetric solutions of Type II supergravity on $\mathbb{R}^{1,2} \times S^3 \times$ M_4 , under the assumption that the seven-dimensional internal Killing spinors have equal norms and that the physical fields of the solution respect the $ISO(1, 2) \times SO(4)$ isometry subgroup. Our classification is quite detailed, going as far as to give explicit local expressions the metric, fluxes and dilaton in terms of simple (Laplace-like) PDE's. As we shall see, solutions in this class are generically $\mathcal{N} = 2$, from the Minkowski perspective, and support a SU(2) R-symmetry realised geometrically as one factor of the $SO(4) \simeq SU(2)_+ \times SU(2)_-$ isometry group manifold of S^3 - the remaining SU(2) factor is a "flavour" under which the Killing spinors are uncharged.¹ This may sound strange as there is no 3d superconformal algebra with $SU(2)_R$, but this only matters for solutions where $\mathbb{R}^{1,2}$ is part of a AdS₄ factor so that SO(2, 3) is realised. Ultimately our results end up side stepping this issue as in general their is either an enhancement of the R-symmetry to SO(4) via the emergence of an additional $S^{2,3}$ factor, or an enhancement of the Minkowski factor to dimensions where $SU(2)_R$ is a necessary part of the superconformal algebra. The classification recovers various well-known intersecting brane systems listed in [40] and some of their U-duals and some of their S-duals. New classes we find include a pure NS $\mathbb{R}^{1,2} \times S^3 \times S^2 \times \mathbb{R}$ vacuum, its U-dual in IIB, the cone over $\mathbb{R}^{1,2} \times S^3 \times S^3$, and a novel class of $\mathbb{R}^{1,2} \times S^2 \times S^3 \times \Sigma_2$ solutions in massless and massive IIA. One of our main results is that the only compact $AdS_4 \times S^3 \times M_3$ solution of type II su-pergravity is the known $\mathcal{N} = 4$ solution of type IIA on a foliation of AdS₄ × S³ × S² over an interval which is the near-horizon limit of the D2–D6 brane system. The required $SO(4)_R$ is realised with one SU(2) from each sphere, and not the S^3 alone. Indeed this is to be expected as if the 3-sphere does realise two SU(2) R-symmetries there would be two sets of $\mathcal{N} = 2$ spinors transforming in the (2, 1) and (1, 2) of SO(4) – there is no $\mathcal{N} = 4$ super-conformal algebra in 3d

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

¹ As such all classes we present are compatible with performing non-Abelian T-duality [45–47] on this SU(2) whilst preserving $SU(2)_R$ [48].

ARTICLE IN PRESS

 JID:NUPHB
 AID:14363 /FLA
 [m1+; v1.285; Prn:6/06/2018; 11:31] P.3 (1-49)

 N.T. Macpherson et al. / Nuclear Physics B ••• (••••)
 3

with Q-generators that transform in this fashion. So it seems likely that the only avenue left open for holographic duals of $\mathcal{N} = 4$ is to seek $AdS_4 \times S^2 \times S^2$ solution like [23,24], but in massive IIA. Our other main result is the discovery of a new class of $\mathcal{N} = 4$ solutions on $\mathbb{R}^{1,2} \times S^2 \times$ $S^3 \times \Sigma_2$ preserving an SO(4) R-symmetry but no AdS₄. These generically have all possible IIA fluxes turned on and can be divided into cases either in massless or massive IIA at which point solutions are in one to one correspondence with a single PDE on σ_2 . In particular the massless solutions are governed by a 3d cylindrical Laplace equation with axial symmetry. These classes look very promising both for finding compact Mink₃ solutions, but also possibly solutions that asymptote to AdS. Let us now describe the outline of the paper: in order to solve the supersymmetry constraints, we will make use of the reformulation of the Killing spinor equations in terms of so-called pure spinor equations. Such pure spinor equations were first used for backgrounds of the form $M_{10} = \mathbb{R}^{1,3} \times M_6$, where it was shown that they are related to integrability constraints of gen-eralised almost complex structures on the internal space M_6 [30]. For backgrounds of the form $M_{10} = \mathbb{R}^{1,2} \times M_7$, the pure spinor equations were constructed in [31] (see also [37]). Next, we decompose $M_7 = S^3 \times M_4$, leading to pure spinor equations on the internal M_4 . We explain this setup in detail in section 2. The resulting supersymmetry constraints vary significantly, de-pending on whether the theory at hand is type IIA or type IIB. We will solve the supersymmetry constraints as well as the Bianchi identities for IIB backgrounds in section 3 and for IIA back-grounds in section 4. In section 5, we then show that there is a unique solution with a warped AdS₄ factor, obtained from the D2–D6 system. In addition to the case where the internal Killing spinors have equivalent norm, in section 6 we examine all backgrounds in the case where one of the Killing spinors vanishes, i.e., $\epsilon_2 = 0$. In this case, there is no need to distinguish between IIA and IIB; we demonstrate that all such backgrounds are pure NSNS and give the solutions. In the appendix, we discuss conventions and identities used, a mild extension of the 3 + 7 pure spinor equation construction (including the non-equivalent norm case), and a discussion on sim-ilar backgrounds from an M-theory perspective.

³⁰ 2. Mink₃ with an S^3 factor

We are interested in solutions to type II with at least a three-dimensional external Minkowski component, with the fluxes respecting the three-dimensional Poincaré invariance:

$$ds^{2} = e^{2A}ds^{2}(\mathbb{R}^{1,2}) + ds^{2}(M_{7}), \quad F = f + e^{3A}\operatorname{Vol}_{3} \wedge \star_{7}\lambda(f), \quad (2.1)$$

where the RR flux f is a polyform on M_7 and the warp factor A and the dilaton Φ are functions on M_7 .² Moreover, we take the NSNS 3-form H to be internal as well. The Killing spinors for $\mathcal{N} = 1$ supersymmetric solutions decompose as

$$\epsilon_1 = \begin{pmatrix} 1 \\ -i \end{pmatrix} \otimes \zeta \otimes \chi_1, \quad \epsilon_2 = \begin{pmatrix} 1 \\ \pm i \end{pmatrix} \otimes \zeta \otimes \chi_2, \qquad (2.2)$$

where ζ is a Majorana spinor of Spin(1, 2) and $\chi_{1,2}$ are Majorana spinors of Spin(7) and where the upper (lower) signs are taken in IIA (IIB). Following [31], we define two real sevendimensional bispinors Φ_{\pm} in terms of $\chi_{1,2}$:

 $^{^{2}}$ We work in the democratic formalism. Other conventions can be found in appendix A.

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

		ARTICLE IN PRESS
JID:NUPHB	AID:14363 /FLA	[m1+; v1.285; Prn:6/06/2018; 11:31] P.4 (1-49)
4	N.T.	Macpherson et al. / Nuclear Physics $B \bullet \bullet \bullet (\bullet \bullet \bullet \bullet) \bullet \bullet \bullet - \bullet \bullet \bullet$

$\Phi_+ + i \Phi = 8e^{-A} \chi_1 \otimes \chi_2^{\dagger} ,$	(2.3)
where the subscript $+/-$ refers to the even/odd forms in the decomposition conditions for unbroken $\mathcal{N} = 1$ supersymmetry are equivalent to	n of the polyform. The
$d_H(e^{2A-\Phi}\Phi_{\pm})=0,$	(2.4a)
$d_H(e^{3A-\Phi}\Phi_{\mp}) + e^{3A} \star_7 \lambda(f) = 0$,	(2.4b)
$\left(\Phi_{\pm} \wedge \lambda(f)\right)\Big _{\mathrm{Top}} = 0 \ ,$	(2.4c)
as long as the norms of spinors $\chi_{1,2}$ are equal, ³ which leads to	
$ \chi^{1} ^{2} = \chi^{1} ^{2} = e^{A}$	(2.5)
The assumption of equal norm is a global requirement for AdS_4 (see for and a local requirement for the existence of calibrated D-branes or O-p however this is not a requirement in general – rather we view this as a well- assumption. Next, we require that the internal space can be decomposed locally as <i>M</i> order to ensure that compactification leads to an <i>SO</i> (4) global symmetry we respect the <i>SO</i> (4) isometry. As a result, the metric and fluxes decomposed	between the first section 5 is section 5 is between the first section 6 is motivated simplifying $M_7 = S^3 \times M_4$, and in we insist that the fluxes further as
$ds^2(M_z) = c^{2C} ds^2(S^3) + ds^2(M_z) = f - C = t c^{3C} Vol(S^3) \land C$	Turmer as
$H = H_3 + H_0 e^{3C} \operatorname{Vol}(S^3) .$, (2.6)
We decompose the 7d spinors in the same fashion in terms of a single ⁴ pso $-\xi$) Killing spinor ξ on S^3 , and two pseudoreal spinors $\eta_{1,2}$ on M_4 :	eudoreal (i.e., $(\xi^c)^c =$
$\chi_i = e^{\frac{A}{2}} (\xi \otimes \eta_i + \xi^c \otimes \eta_i^c) = e^{\frac{A}{2}} \xi^a \otimes \eta_i^a , \qquad i = 1, 2$	(2.7)
which is the most general parameterisation consistent with an $S^3 \times M_4$ procondition. ⁵ Note that we do not restrict the $Spin(4)$ spinors η_i to be chi $\eta_{1,2}^{\dagger}\eta_{1,2} = 1$. The Killing spinors on S^3 satisfy the Killing spinor equation	duct and the Majorana iral and we normalise
$\nabla_{\alpha}\xi = \frac{1}{2}i\nu\sigma_{\alpha}\xi , \nu = \pm 1 ,$	(2.8)
which preserves two supercharges for each of $\nu = \pm 1$. We will not make a establish whether any solutions are independent of this choice – the S^3 of preserve 4 supercharges. As explained in Appendix C a spinor on S^3 define	choice of v so we can such a solution would es a doublet
³ In [31], [37] an additional constraint that was imposed in order to derive (2.4) was that the NSNS 3-form flux is trivial; unlike in four dimensions, this is not enforced by Poincaré this second assumption is redundant though, as is shown in appendix D: if $ \chi_1 ^2 = \chi_2 ^2$ and cosmological constant, then supersymmetry enforces that the external NSNS flux vanishes.	the external component of invariance. It turns out that spacetime does not admit a
As explained in Appendix C, there are two independent types of Killing spinors on S^3 , cannot be mapped to each other using the $SO(4)$ invariants of the fluxes or the Killing spinor appears when one decompose $M_7 = S^3 \times M_4$, so if one were to include terms like $\xi_+ \otimes \eta_+$ the 7d spinor conditions to 4d ones you would find that η_{\pm} never mix. So setting one of η_{\pm} to in our analysis.	ξ_{+} and ξ_{-} – however they or equations. This is all that and $\xi_{-} \otimes \eta_{-}$ then reduced o zero excludes no solutions
⁵ One might imagine it was possible to construct a more general 7d spinor from two 4d sp But if one then adds the Majorana conjugate to this the resulting spinor can be put in the form	pinors like $\xi \otimes \eta + \xi^c \otimes \tilde{\eta}$. of (2.7) by redefining η , $\tilde{\eta}$.
Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II si	upergravity, Nucl. Phys. B

JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.5 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••)

$$\xi^a = \begin{pmatrix} \xi \\ \xi^c \end{pmatrix} \tag{2.9}$$

which is charged under one SU(2) factor of $SO(4) = SU(2)_+ \times SU(2)_-$, depending on the sign of $v - \xi^a$ is a singlet under the action of the second SU(2). As such, a generic solution with Mink₃ \times S³ will have an R-symmetry SU(2)_R and an additional global flavour symmetry $SU(2)_F$. Such solutions preserve at least $\mathcal{N} = 2$ supersymmetry from the 3d perspective, so 4 real supercharges - indeed, the 10d Killing spinors may be written as

$$\epsilon_1 = \begin{pmatrix} 1 \\ -i \end{pmatrix} \otimes \zeta^a \otimes (\xi^a \otimes \eta^1 + \xi^{ac} \otimes \eta^{1c}) , \quad \epsilon_2 = \begin{pmatrix} 1 \\ \pm i \end{pmatrix} \otimes \zeta^a \otimes (\xi^a \otimes \eta^2 + \xi^{ac} \otimes \eta^{2c}) ,$$

where ζ^a is a doublet of Killing spinors on $\mathbb{R}^{1,2}$, that allow the 10d spinors to be invariant under $SU(2)_R$ transformations. However we only need to solve an $\mathcal{N} = 1$ sub-sector, because the part of the Killing spinor which couples to ζ^1 is mapped to the part coupling to ζ^2 under the action of $SU(2)_R$ – so if you solve one part, the other is guaranteed. If a solution ends up being independent of v then there is a copy of (2.2) for each sign and supersymmetry is doubled to $\mathcal{N} = 4$ – there are two SU(2) R-symmetries, but they do not appear as a product so do not form $SO(4)_R$ – as we shall see, this only happen in a small number of special cases.

Using the gamma matrix decomposition (A.2), the seven-dimensional bispinor (2.3) decom-poses as

Here, $\hat{\gamma}$ is the four-dimensional chirality matrix and the \pm subscripts again refer to even and odd form components. We see that the components are in fact matrices and that the seven-dimensional bispinor is constructed as the trace of the product of the components.

The S^3 component leads to the bispinor matrix

$$\xi^{a} \otimes \xi^{b\dagger} = \frac{1}{2} \left((1 - ie^{3C} \operatorname{Vol}(S^{3})) + \frac{1}{2} \left(e^{C} K_{i} - \frac{\nu}{2} ie^{2C} dK_{i} \right) (\sigma^{i})_{ab} \right), \qquad (2.11)$$

where K_i is a vielbein defining a trivial structure on S^3 (see appendix A). The M_4 component leads to the bispinor matrix

$$(\eta_1^a \otimes \eta_2^{b\dagger})_{\pm} = \begin{pmatrix} \psi_{\pm}^1 & \psi_{\pm}^2 \\ \mp (\psi_{\pm}^2)_{*} & \pm (\psi_{\pm}^1)_{*} \end{pmatrix}, \quad (\hat{\gamma}\eta_1^a \otimes \eta_2^{b\dagger})_{\pm} = \begin{pmatrix} \psi_{\hat{\gamma}\pm}^1 & \psi_{\hat{\gamma}\pm}^2 \\ \mp (\psi_{\hat{\gamma}\pm}^2)_{*} & \pm (\psi_{\hat{\gamma}\pm}^1)_{*} \end{pmatrix}, \quad 34$$

$$(2.12) \quad 36$$

where

$$\psi^{1} = 4\eta^{1} \otimes \eta^{2\dagger}, \quad \psi^{2} = 4\eta^{1} \otimes \eta^{2c\dagger}, \quad \psi^{1}_{\hat{\gamma}} = 4\hat{\gamma}\eta^{1} \otimes \eta^{2\dagger}, \quad \psi^{2}_{\hat{\gamma}} = 4\hat{\gamma}\eta^{1} \otimes \eta^{2c\dagger} \tag{2.13}$$

Since the matrix entries are somewhat involved, we refer to appendix B for details. Plugging both components (2.11), (2.12) into the seven-dimensional bispinors (2.10), it follows that

$$\Phi_{+} = \operatorname{Re}\psi_{+}^{1} - e^{3C}\operatorname{Vol}(S^{3}) \wedge \operatorname{Im}\psi_{\hat{\gamma}-}^{1} + \frac{e^{C}}{2}(K_{1} \wedge \operatorname{Re}\psi_{\hat{\gamma}-}^{2} + K_{2} \wedge \operatorname{Im}\psi_{\hat{\gamma}-}^{2} + K_{3} \wedge \operatorname{Re}\psi_{\hat{\gamma}-}^{1}), \quad \overset{43}{44}$$

$$-\frac{e^{2C}}{4}(K_1 \wedge K_2 \wedge \mathrm{Im}\psi_+^1 + K_1 \wedge K_3 \wedge \mathrm{Re}\psi_+^2 + K_2 \wedge K_3 \wedge \mathrm{Im}\psi_+^2), \qquad (2.14) \qquad \overset{46}{47}$$

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

з

CLE IN PRESS

JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.6 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••

$$\Phi_{-} = \operatorname{Im}\psi_{-}^{1} - e^{3C}\operatorname{Vol}(S^{3}) \wedge \operatorname{Re}\psi_{\hat{\gamma}+}^{1} + \frac{e^{C}}{2}(K_{1} \wedge \operatorname{Im}\psi_{\hat{\gamma}+}^{2} - K_{2} \wedge \operatorname{Re}\psi_{\hat{\gamma}+}^{2} + K_{3} \wedge \operatorname{Im}\psi_{\hat{\gamma}+}^{1}), \quad \frac{1}{2}(K_{1} \wedge \operatorname{Im}\psi_{\hat{\gamma}+}^{2} - K_{2} \wedge \operatorname{Re}\psi_{\hat{\gamma}+}^{2} + K_{3} \wedge \operatorname{Im}\psi_{\hat{\gamma}+}^{1}),$$

$$+\frac{e^{2C}}{4}(K_1\wedge K_2\wedge \operatorname{Re}\psi_-^1-K_1\wedge K_3\wedge \operatorname{Im}\psi_-^2+K_2\wedge K_3\wedge \operatorname{Re}\psi_-^2).$$

At this point, the IIA and IIB supersymmetry equations diverge, and we shall relegate their explicit form to the relevant sections.

With our set up, a solution to the supersymmetry equations is a solution to the equation of motion if and only if it satisfies the Bianchi identities [4] [38] [39]. These are given by g $d_H F = dH = 0$ away from localised sources. By definition, a localised (magnetic) source man-ifests itself in the Bianchi identity of some field strength F as $dF = Q\delta^n(x)$ and hence in such cases F is discontinuous. Loosely speaking, a localised source corresponds physically to an extended object (such as a brane) located at a submanifold of the ten-dimensional spacetime $S \subset M_{10}$ which is pointlike in some of the local coordinates. The standard approach to obtaining backgrounds, which we follow as well, is to first solve the supersymmetry equations by introduc-ing local coordinates, and then afterwards determine the physically sensible range of these local coordinates by examining the obtained geometry and fluxes. The presence of localised sources is signified by discontinuities of not just the fluxes, but of the spacetime geometry as well, pre-cisely at the location of the sources. Therefore, it is possible to obtain solutions with localised sources even when making use of the Bianchi identities with no sources: one examines possible discontinuities in the geometry and fluxes and determines whether or not such discontinuities are associated with localised sources or not by comparing them with the divergent behaviour of known extended objects.

Making use of the flux decomposition (2.1), (2.6), the Bianchi identities thus reduce to

$$d_{H_3}(e^{3A+3C} \star_4 \lambda(G_{\pm})) = d_{H_3}(e^{3C} \star_4 \lambda(G_{\pm})) = 0, \qquad 25$$

$$d_{H_3}(G_{\pm}) = d_{H_3}(e^{3C}G_{\mp}) = 0, \qquad (2.15)$$

$$d_{H_3}(G_{\pm}) = d_{H_3}(e^{3C}G_{\mp}) = 0, \qquad (2.15)$$

- This is after imposing $H_0 = 0$, which turns out to be a requirement for every solution to the supersymmetry equations that we obtain.
- 2.1. Summary of obtained backgrounds

As the rest of the paper is somewhat technical, let us summarise our results here. We find a number of well-known backgrounds, as well as some new ones. In type IIB with internal Killing spinors of equal norm, we find: 1. The intersecting D3-D7 system with metric (3.27), fluxes (3.25) and scalar field constraints (3.26).2. The D5-brane with metric (3.37), fluxes (3.35) and scalar field constraints (3.36). 3. A generalization of the D5-brane generated by U-duality. The metric is given by (3.43), the

- fluxes by (3.41), scalar field constraints by (3.42). 4. A new background on the cone over $\mathbb{R}^{1,2} \times S^3 \times S^3_{sq}$, with S^3_{sq} a generically squashed three-
- sphere admitting an $SU(2) \times U(1)$ isometry group. For the unsquashed limit, the metric is given by (3.59), the fluxes by (3.57), the scalar field constraints by (3.58). In the generic
- squashed case, the metric and dilaton are given by (3.75), the fluxes by (3.74). We note that
- the more general squashed case can be obtained from the unsquashed case by a duality chain.

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; · N T. Macnherson et al. / Nuclear Physics B and (annu) and -annu	11:31] P.7 (1-49) 7
n.i. macpherson et al. 7 Machear 1 hystes D ()	,
In type IIA with internal Killing spinors of equal norm, we find:	
1. The intersecting D4–D8 system with metric (4.14), fluxes (4.12) and scalars (4.11), (4.13).	s constraints
2. The intersecting D2–D6 system with metric (4.24), fluxes (4.22) and scalar (4.21), (4.23).	r constraints
3. A generalization of the D4–D8 system generated by U-duality. The metric is giv the fluxes by (4.29), and scalar constraints by (4.27).	en by (4.30),
4. A class of new backgrounds. The metric contains an $\mathbb{R}^{1,2} \times S^3 \times S^2$ factor, warpings, and is given by (4.42). The warp factors are constrained by various	with various PDE, given
in (4.43). In general, all fluxes are turned on and are given by (4.44). This backgrounds contains a subset with a $U(1)$ isometry. In this case, T-dualisin	new class of ng along the
isometry direction leads to the new IIB backgrounds outlined above, with generi	ic squasning.
In addition, we find two more backgrounds when setting $\epsilon_2 = 0$. These backgrounds NS, and as such, can be found in both type IIA and type IIB. We find:	inds are pure
1. The NS5-brane, with metric (6.18), flux (6.16) and the scalar constraints (6.17) 2. A pure NS background on $\mathbb{R}^{1,2} \times \mathbb{R} \times S^3 \times S^3$ dual to the new (unscalashed) IIR	hackground
The metric is given by (6.22) , the flux by (6.21) . All scalars are determined up	to constant
factors.	
2 Mink with an 63 factor in HD	
5. Minks with an S ² factor in fib	
The type IIB supersymmetry equations are obtained by plugging the decomp dimensional bispinors (2.14) into the seven-dimensional supersymmetry constraint leads to the following constraints on the four-dimensional bispinors	bosed seven- is (2.4). This
$d_{H_3}(e^{2A-\Phi} \operatorname{Re} \psi^1_+) = 0$,	(3.1a)
$d_{H_3}(e^{3A+2C-\Phi}\psi^2) + 2i\nu e^{3A+C-\Phi}\psi_{\hat{\gamma}+}^2 = 0,$	(3.1b)
$d_{H_3}(e^{2A+2C-\Phi}\psi_+^2) + 2i\nu e^{2A+C-\Phi}\psi_{\hat{\gamma}-}^2 = 0,$	(3.1c)
$d_{H_3}(e^{3A+2C-\Phi} \text{Re}\psi^1) - 2\nu e^{3A+C-\Phi} \text{Im}\psi_{\hat{\gamma}+}^1 = 0 ,$	(3.1d)
$d_{H_3}(e^{2A+2C-\Phi} \mathrm{Im}\psi^1_+) + 2\nu e^{2A+C-\Phi} \mathrm{Re}\psi^1_{\hat{\gamma}-} = 0 ,$	(3.1e)
$d_{H_3}(e^{2A+3C-\Phi} \operatorname{Im} \psi_{\hat{\gamma}-}^1) + e^{2A+3C-\Phi} H_0 \operatorname{Re} \psi_+^1 = 0,$	(3.1f)
$d_{H_3}(e^{3A-\Phi} \text{Im}\psi^1) + e^{3A} \star_4 \lambda(G_+) = 0 ,$	(3.2a)
$d_{H_3}(e^{3A+3C-\Phi}\operatorname{Re}\psi^1_{\hat{\gamma}_+}) - e^{3A+3C-\Phi}H_0\operatorname{Im}\psi^1 + \nu e^{3A+3C} \star_4 \lambda(G) = 0$	(3.2b)
and must additionally satisfy the pairing equation	
$\left(\operatorname{Im} \psi_{\hat{\gamma}^{-}}^{1} \wedge \lambda(G_{-}) - \operatorname{Re} \psi_{+}^{1} \wedge \lambda(G_{+})\right)\Big _{4} = 0.$	(3.3)
Please cite this article in press as: N.T. Macpherson et al., Mink $_3 \times S^3$ solutions of type II supergravity.	Nucl. Phys. B

	ARTICLE IN PRESS
	JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.8 (1-49)
	8 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••
1	In order to solve these, we will first examine the 0-form conditions. These are given by
2	$(\psi_{\hat{\gamma}}^2)_0 = (\mathrm{Im}\psi_{\hat{\gamma}}^1)_0 = H_0(\mathrm{Re}\psi_{\hat{\gamma}}^1)_0 = 0. $ (3.4)
4	We solve the first two of these in Appendix B, which leads to a spinor ansatz depending on 6 real
5	functions with support on M_4
6	$\alpha, a_1, b_1, \lambda_1, \lambda_2, \lambda_3 \tag{3.5}$
7 8	subject to the constraint
9	$a_1^2 + b_1^2 + \lambda_1^2 + \lambda_2^2 + \lambda_3^2 = 1. $ (3.6)
10	The third 0-form constraint which is unique to IIB still needs to dealt with After making use of
11 12	(B.12), it reduces to
13	$H_0(a_1 \cos\frac{\alpha}{2} + b_1 \sin\frac{\alpha}{2}) = 0. $ (3.7)
14	
15	Here, as well as in IIA, the solutions depend drastically on the behaviour of α . We can distinguish between three different cases: $\alpha = 0$, $\alpha = \frac{1}{2}\pi$ and generic $\alpha \in (0, \pi)$, $\alpha \neq \frac{1}{2}\pi$. Let us reiterate
16	that we introduced α in (B.5) by defining
17 18	(2.9)
19	$\eta_1 = \cos\left(\frac{1}{2}\right)\eta + \sin\left(\frac{1}{2}\right)\gamma\eta, \qquad (3.8)$
20	where η is a locally defined non-chiral spinor, where the chiral components are normalised. Note
21	that the non-chirality is crucial: it ensures that η can be used to define the local trivial structure
22	(i.e., the vieldein) via (B.5). In the case that $\alpha = 0$, the 4d internal Killing spinors $\eta_1 = \eta$ are such that the chiral components of η_1 have equal norm. In the case that $\alpha = \pi/2$ we see that
23	such that the enhancements of η_1 have equal form. In the case that $\alpha = \pi/2$, we see that n_1 becomes chiral. It turns out that we can treat this case together with $\alpha \neq 0$, but find no such
24	solutions. Thus we separate our solutions into two branches.
26	Branch I: Here $\alpha = 0$. The only non-trivial zero form is $\alpha_1 H_0 = 0$, which a priori can be
27	solved in two ways. However, we shall see in the next section that only $H_0 = 0$ is consistent with
28	the higher form conditions. In order to solve (B.6) we parametrise
29 30	$a_1 = \sin \beta$, $b_2 = \cos \beta \sin \beta$, $b_3 = v_2 \cos \beta \cos \beta$, $b_4 = v_2 \cos \beta \cos \beta$, $b_5 = v_2 \cos \beta \cos \beta$
31	$u_1 = \sin p, \ v_1 = \cos p \sin v, \ v_1 = y_1 \cos p \cos v, \ v_1 = y_2 \cos p \cos v, \ v_3 = y_3 \cos p \cos v,$
32	(3.9)
33	with
34	$y_1 = \sin\theta\cos\phi, y_2 = \sin\theta\sin\phi, y_3 = \cos\theta.$ (3.10)
35 26	
37	
38	Branch II: Here $0 < \alpha < \pi$. Note that $\alpha = \pi$ is equivalent to $\alpha = 0$, which is easiest to see by sending $n \rightarrow \hat{u}n$ in (B.5). We choose to parametrise
39	by sending $\eta \rightarrow \gamma \eta$ in (b.5). We choose to parametrise
40	$a_1 = \cos\beta\sin(\delta - \frac{\alpha}{2}), b_1 = \cos\beta\cos(\delta - \frac{\alpha}{2})$
41 42	$\lambda_1 = -\cos\beta\cos\delta y_1$, $\lambda_2 = -\sin\beta y_3$
43	$\lambda_3 = -\sin\beta y_2$
44	where y_i is defined in (3.10). This ensures (B.6) and all but the first equation of (3.7) are solved.
45	which becomes
46	

 $H_0\cos\beta\sin\delta=0\,.$ (3.12)

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

	N.T. Macpherson et al. / Nuclear Physics $B \leftrightarrow (\bullet \bullet \bullet \bullet) \bullet \bullet \bullet - \bullet \bullet \bullet$	9
3.1. Brar	ach I: solutions with $\alpha = 0$	
In orde that follo	er to solve branch I, it is convenient to first examine a number of lower form co w from (3.1) . To do this it is useful to first rotate the canonical frame of (B.3) s	onditions such that
v_1	$\rightarrow \sin\phi w_2 + \cos\phi(\cos\theta v_1 + \sin\theta(\cos\delta w_1 - \sin\delta v_2))$	
v_2	$\rightarrow \cos \delta v_2 + \sin \delta w_1$	(2.12)
w_1	$\rightarrow \cos\theta(\cos\delta w_1 - \sin\delta v_2) - \sin\theta v_1$	(3.13)
w_2	$\rightarrow \cos\phi w_2 - \sin\phi(\cos\theta v_1 + \sin\theta(\cos\delta w_1 - \sin\delta v_2)) .$	
Making u	se of these, one finds that the supersymmetry equations imply	
sin	$\beta H_0 = 0$,	(3.14a)
d(e	$^{2A+2C-\Phi}\cos\beta\cos\delta) - 2\nu e^{2A+C-\Phi}\cos\beta\nu_2 = 0,$	(3.14b)
d(e	$^{2A+3C-\Phi}\sin\beta(\cos\delta v_2+\sin\delta w_1))-e^{2A+3C-\Phi}H_0\cos\beta\cos\delta v_1\wedge w_2=0,$	(3.14c)
cos	$\beta(e^C\cos\delta d\theta + 2\nu\sin\delta v_1) = \cos\beta(e^C\cos\delta\sin\theta d\phi - 2\nu\sin\delta w_2) = 0,$	(3.14d)
d(e	$^{3A+2C-\Phi}\sin\beta(\cos\delta w_1-\sin\delta v_2))+2\nu e^{3A+C-\Phi}(\sin\beta v_2\wedge w_1+\sin\delta\cos\beta v_2)$	$(1 \wedge w_2)$
$+ \epsilon$	$e^{3A+2C-\Phi}\cos\beta\cos\delta(d\theta\wedge w_2+\sin\theta d\phi\wedge v_1)=0,$	(3.14e)
which is r one needs The no $d\phi = 0, \phi$ ignoring ϕ 0. Let us r by introduct	hot a compete list. The first thing to establish is how to solve $(3.14a) - if$ we set s is to set $\cos \delta = 0$ to solve $(3.14c)$, but since $\nu = \pm 1$, $(3.14b)$ leads to a contradi- ext conditions we consider are $(3.14d)$. For $\cos \beta \neq 0$ we see that either $\sin \delta$ for $0 < \sin \delta < \frac{\pi}{2}$ in which case (θ, ϕ) define local coordinates on a 2-sphere $\cos \beta = 0$ because, as should be clear from, (3.9) , this is a subcase of $\sin \delta = d\theta$ how prove that $0 < \sin \delta < \frac{\pi}{2}$ is not possible: Since $H_0 = 0$ we can solve $(3.14b)$ ucing local coordinates x and $\rho = e^{2A+2C-\Phi} \cos \beta \cos \delta$ such that	in $\beta = 0$, ction. $= d\theta =$. We are $= d\phi =$ -(3.14c)
vv_2	$=\frac{1}{2}\sqrt{\frac{\cos\delta}{\rho\cos\beta}}e^{-A+\frac{\Phi}{2}}d\rho, \ w_1=\frac{\sqrt{\cos\beta}\cos^{3/2}\delta}{2\nu\rho^{3/2}\sin\delta}e^{A-\frac{\Phi}{2}}(2\nu\cot\beta dx-\sec\beta e^{-2A+\frac{\Phi}{2}})d\rho$	$^{+\Phi} ho d ho).$
We can a	lso rewrite (3.14e) as	
d(e	$^{3A+2C-\Phi}\sin\beta(\cos\delta w_1-\sin\delta v_2))+2\nu e^{3A+C-\Phi}(\sin\beta v_2\wedge w_1-\sin\delta\cos\beta v_1\wedge v_2)$	$w_2)=0,$
using (3. orthogona as there i that $\sin \delta$ specific v	14d). The key point here is that v_2 , w_1 only have legs in (ρ, x) while v_1 al to this with legs in (θ, ϕ) only. This means that the equation above, cannot be s a Vol (S^2) term whose coefficient is non-vanishing. Thus we can conclude in $= d\theta = d\phi = 0$. Plugging this back into (4.1), one finds that nothing dependence alues these parameters take so we can set	, w_2 sit be solved a general ls on the
H_0	$= heta=\phi=\delta=0$,	(3.15)
without le We are	biss of generality, leaving one undetermined function β . e now ready to write the supersymmetry conditions that follow when $\alpha = 0$, helpful to perform a second rotation of the canonical vielbein by considering	however

CLE IN PRESS JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.10 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) ••• $v_1 + iw_2 \rightarrow e^{-i\beta}w, \quad v_2 - iw_1 \rightarrow -iv,$ (3.16)to ease presentation. The necessary and sufficient conditions for supersymmetry in the $\alpha = 0$ branch are $H_0 = d(e^{2A-\Phi}\sin\beta) = d(e^{A+C-\frac{1}{2}\Phi}\sqrt{\cos\beta}) - \nu e^{A-\frac{\Phi}{2}}\sqrt{\cos\beta}\nu_2$ $= d(e^{2A+3C-\Phi}\sin\beta v_2) = 0,$ $d(e^{3A+2C-\Phi}w) + 2\nu e^{3A+C-\Phi}w \wedge v_2$ g $= d(e^{3A+2C-\Phi}\sin\beta v_1) + 2\nu e^{3A+C-\Phi}\sin\beta v_1 \wedge v_2 = 0,$ $d(e^{2A+2C-\Phi}v_1 \wedge w) - 2ve^{2A+C-\Phi}v_1 \wedge w \wedge v_2 = 0,$ $d(e^{2A+2C-\Phi}\sin\beta w_1 \wedge w_2) - 2\nu e^{2A+C-\Phi}\sin\beta w_1 \wedge w_2 \wedge v_2 + e^{2A+2C-\Phi}\cos\beta H_3 = 0,$ $H_3 + 2\beta \wedge w_1 \wedge w_2 = d(e^{4A-\Phi}\cos\beta) \wedge w_1 \wedge w_2 \wedge v_2 = 0,$ $d(e^{3A-\Phi}\cos\beta v_1) - d(e^{3A-\Phi}\sin\beta v_1 \wedge w_1 \wedge w_2) + e^{3A-\Phi}\cos\beta v_1 \wedge H_3 - e^{3A} \star_4 \lambda(G_+)$ = 0. $d(e^{3A+3C-\Phi}\cos\beta v_1\wedge v_2)-e^{3C+3A}\star_4\lambda(G_-)=0,$ $\left(\left(\sin\beta + \cos\beta w_1 \wedge w_2\right) \wedge v_2 \wedge \lambda(G_-) - \left(\sin\beta + \cos\beta w_1 \wedge w_2\right) \wedge \lambda(G_+)\right)\Big|_4 = 0.$ (3.17)We can simplify this system further, but not without making assumptions about β . We now proceed to study the systems that follow from different values of β , we find that the physical interpretation is quite different in each case. *3.1.1. Subcase:* $\beta = 0$ Upon setting $\beta = 0$ in (3.17) one can show that the supersymmetry conditions reduce to $H_3 = H_0 = d(e^{-\Phi}) \wedge w_1 \wedge w_2 = d(e^{-4A}) \wedge v_2 \wedge w_1 \wedge w_2 = 0,$ (3.18a) $d(e^{-A}v_1) \wedge w = d(e^{A+C-\frac{1}{2}\Phi}) - ve^{A-\frac{1}{2}\Phi}v_2 = d(e^Aw) = 0,$ (3.18b) $e^{3A} \star_4 \lambda(G_+) = d(e^{3A-\Phi}v_1), \quad e^{3A+3C} \star_4 \lambda(G_-) = d(e^{3A+3C-\Phi}v_1 \wedge v_2),$ (3.18c) $\lambda(G_{-}) \wedge v_2 \wedge w_1 \wedge w_2 - \lambda(G_{+}) \wedge w_1 \wedge w_2 = 0.$ (3.18d) We can solve (3.18b) by using it to define a vielbein in terms of local coordinates ψ , x_1 , x_2 and

$$\rho = e^{A + C - \frac{1}{2}\Phi} \tag{3.19}$$

such that

$$v_{1} = e^{A}(d\psi + V) , \quad v_{2} = ve^{-A + \frac{1}{2}\Phi}d\rho , \quad w = e^{-A}(dx_{1} + idx_{2}) ,$$

$$V = f_{1}(x_{1}, x_{2})dx_{1} + f_{2}(x_{1}, x_{2})dx_{2} .$$
(3.20)
(3.20)

Please cite this article in press as: N.T. Macpherson et al.,
$$Mink_3 \times S^3$$
 solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.11 (1-49)

N.T. Macpherson et al. / Nuclear Physics $B \bullet \bullet \bullet (\bullet \bullet \bullet \bullet) \bullet \bullet \bullet$

From (3.18a) we see there is no NSNS flux, ∂_{ψ} is an isometry and $A = A(\rho, x_1, x_2), \Phi =$ $\Phi(x_1, x_2)$. We now have enough information to calculate the fluxes. First we find $\star_4 G_{-} = -e^A \left(\partial_{x_1} (e^{-\Phi}) w_1 + \partial_{x_2} (e^{-\Phi}) w_2 \right) \wedge v_1 \wedge v_2 - e^{3A - \Phi} \left(\partial_{x_1} f_2 - \partial_{x_2} f_1 \right) v_1$ $\star_4 G_+ = -e^{3A} \left(\partial_{x_1} (e^{4A-\Phi}) w_1 \wedge v_2 + \partial_{x_2} (e^{4A-\Phi}) w_2 \wedge v_2 + v e^{-\frac{1}{2}\Phi} \partial_{\rho} e^{4A-\Phi} v_2 \wedge v_1 \right)$ $-e^{3A-\Phi}(\partial_1 f_2 - \partial_2 f_1)w_1 \wedge w_2$. (3.21)

We can then use coordinate dependence of the physical fields and local expression for the vielbein (3.20) to take the Hodge dual in (3.18c) arriving at

$$G_{-} = (\partial_{x_2} f_1 - \partial_{x_1} f_2) e^{4A - \Phi} (d\psi + V) + \partial_{x_2} (e^{-\Phi}) dx_1 - \partial_{x_1} (e^{-\Phi}) dx_2, \qquad (3.22)$$

$$G_{+} = -\nu e^{3A - \frac{3}{2}\Phi} \left(\partial_{x_{2}}(e^{-4A + \Phi}) dx_{1} \wedge d\rho - \partial_{x_{1}}(e^{-4A + \Phi}) dx_{2} \wedge d\rho \right)$$
15
16
17
17

$$-e^{-\Phi}\partial_{\rho}(e^{-4A+\Phi})dx_{1}\wedge dx_{2}-\nu e^{\Phi}(\partial_{x_{2}}f_{1}-\partial_{x_{1}}f_{2})(d\psi+V)\wedge d\rho\bigg).$$
(3.23) ¹⁸
₁₉
₂₀

Plugging this into (3.18d) we find $(\partial_{x_2} f_1 - \partial_{x_1} f_2) = 0$ which mean that V is closed and so we can locally fix

$$V = 0 , \qquad (3.24)$$

with a shift $\psi \to \psi - \eta$ for $d\eta = V$, without loss of generality. Taking this into account the ten-dimensional fluxes are

$$F_1 = \partial_{x_2}(e^{-\Phi})dx_1 - \partial_{x_1}(e^{-\Phi})dx_2, \quad F_5 = d\psi \wedge d(e^{4A-\Phi}) \wedge \text{Vol}_3 \tag{3.25}$$

$$-\nu\rho^{3}\left(\partial_{x_{2}}(e^{-4A+\Phi})dx_{1}\wedge d\rho-\partial_{x_{1}}(e^{-4A+\Phi})dx_{2}\wedge d\rho-e^{-\Phi}\partial_{\rho}(e^{-4A+\Phi})dx_{1}\wedge dx_{2}\right)$$

$$\wedge \operatorname{Vol}(S^3)$$

The final thing we need to do is impose the Bianchi identities, which away from localised sources rise to the PDEs

$$(\partial_{x_1}^2 + \partial_{x_2}^2)e^{-\Phi} = 0, \quad \frac{e^{-\Phi}}{\rho^3}\partial_{\rho}(\rho^3\partial_{\rho}e^{-4A+\Phi}) + (\partial_{x_1}^2 + \partial_{x_2}^2)(e^{-4A+\Phi}) = 0.$$
(3.26)

The local form of the metric is then

$$H = e^{-4A + \Phi} , \quad f = e^{-\Phi}$$

This corresponds to the intersecting D3–D7 brane system, where the D3-branes are embedded in the D7-branes [40].

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

 ARTICLE IN PRESS

 JID:NUPHB AID:14363 /FLA
 [m1+; v1.285; Prn:6/06/2018; 11:31] P.12 (1-49)

 12
 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••

3.1.2. Subcase: $\beta = \frac{\pi}{2}$ Setting $\beta = \frac{\pi}{2}$ in (3.17) leads to the following necessary and sufficient conditions for unbroken supersymmetry $d(e^{2A-\Phi}) = d(e^{2A+3C-\Phi}v_2) = H_3 = H_0 = G_- = 0,$ (3.28a) $d(e^{3A+2C-\Phi}u_i) + 2ve^{3A+C-\Phi}u_i \wedge v_2 = 0,$ (3.28b) $d(e^{2A+2C-\Phi}\epsilon_{ijk}u_i \wedge u_k) - 2\nu e^{2A+C-\Phi}\epsilon_{ijk}u_i \wedge u_k \wedge v_2 = 0,$ (3.28c)g $e^{3A} \star_4 \lambda(G_+) = d(e^{3A-\Phi}v_1 \wedge w_1 \wedge w_2).$ (3.28d) Here we have introduced the notation $u = (v_1, w_1, w_2),$ (3.29)both to ease notation and to stress that the vielbeine u_i obey a cyclic property. Exploiting this property will be very helpful in solving this system and other systems we shall encounter which mirror this behaviour, so we will be very explicit in our derivation here, but less so elsewhere. The first thing to note is that the combination $(3.28c)_i + e^{-A}\epsilon_{ijk}u_j \wedge (3.28b)_k$ leads to $\epsilon_{ijk} \left(d(e^{2A+C-\frac{1}{2}\Phi}) - \nu e^{2A-\frac{1}{2}\Phi} v_2 \right) \wedge u_j \wedge u_k = 0 \,.$ (3.30)This implies that the 1-form in large brackets is zero. This can be seen by writing it as $X_i u^j$ + X_0v_2 for some functions X_j , noting that the vielbeine $u_{1,2,3}$ are independent, and then consider-ing the resulting constraints for i = 1, 2, 3. Next by examining $(3.28c)_i + e^{-A}u_i \wedge (3.28b)_k$ and $(3.28c)_i - e^{-A} \wedge (3.28b)_i \wedge u_k$ for cyclic permutations of (i, j, k) = (1, 2, 3) one realises that $d(e^{-A}u_i) \wedge u_j = 0, \quad i \neq j,$ (3.31)which implies that $d(e^{-A}u_i)$ has no leg in u_i , it is then not hard to see that since (3.28b) has no $\epsilon_{ijk}u_j \wedge u_k$ term it is in fact zero. So we can conclude without loss of generality that $d(e^{-A}u_i) = d(e^{2A+C-\frac{1}{2}\Phi}) - \nu e^{2A-\frac{1}{2}\Phi}v_2 = 0,$ (3.32)which imply (3.28b)-(3.28c) without further constants. We can then solve these conditions by using them to define the vielbeine in terms of local coordinates as $24 \pm \frac{1}{2}$ $2A + C = \frac{1}{2} \sigma$

$$v_1 = e^A dx_1$$
, $w_1 = e^A dx_2$, $w_2 = e^A dx_3$, $v_2 = v e^{-2A + 2\Phi} d\rho$, $\rho = e^{2A + C - 2\Phi}$.
(3.33)

We can now solve (3.28a), which in fact just tells us that

$$e^{\Phi} = e^{2A},$$
 (3.34) ⁴⁰
₄₁

⁴² up to rescaling g_s and that e^{2A} is a function of ρ only, making ∂_{x_i} all isometry directions param-⁴³ eterising either \mathbb{R}^3 or T^3 locally. ⁴⁴ The only non-trivial flux is the RR 3-form

$$F_{3} = -\nu\rho^{3}\partial_{\rho}(e^{-4A})\operatorname{Vol}(S^{3})$$
(3.35) (3.35)

⁴⁷ and its Bianchi identity, $dF_3 = 0$, imposes that

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.13 (1-49) N.T. Macpherson et al. / Nuclear Physics $B \bullet \bullet \bullet$ (

 $e^{-4A} = c_1 + \frac{c_2}{\rho^2}, \quad dc_i = 0.$ (3.36)

This is the warp factor of a D5-brane or O5-hole, depending on the sign of c_2 (see for example [41]). Indeed the metric locally takes the form

$$ds^{2} = e^{2A} ds^{2}(\mathbb{R}^{1,5}) + e^{-2A} \left(d\rho^{2} + \rho^{2} ds^{2}(S^{3}) \right)$$
(3.37)
$$(3.37)$$

As we will see, this is a subcase of the solution in the next section.

3.1.3. Subcase: generic β

For generic $0 < \beta < \frac{\pi}{2}$ we are free to divide by the trigonometric functions in (3.17). Using $\sin \beta \neq 0$ it is possible to show that supersymmetry requires

$$d(e^{2A-\Phi}\sin\beta) = d(e^{-\Phi}\cos\beta) = de^A \wedge v_2 = 0,$$
(3.38a)
¹³
₁₄

$$d(e^{-A}v_1) = d(e^{2A+C-\frac{\Phi}{2}}\sqrt{\sin\beta}) - ve^{2A-\frac{\Phi}{2}}\sqrt{\sin\beta}v_2 = d(e^{-A}w\csc\beta) = 0, \qquad (3.38b)$$

$$H_{a} + 2d\theta \wedge w \wedge w - d\theta \wedge w \wedge w - 0 \qquad (3.38b)$$

$$H_3 + 2d\beta \wedge w_1 \wedge w_2 = d\beta \wedge v_2 \wedge w_1 \wedge w_2 = 0, \qquad (3.38c)$$

$$e^{3A} \star_4 \lambda(G_+) = d(e^{3A-\Phi}\cos\beta v_1) + d(e^{3A-\Phi}\sin\beta v_1 \wedge w_1 \wedge w_2) - e^{3A-\Phi}\cos\beta H_3 \wedge v_1, \qquad ^{19}_{20}$$

$$e^{3A+3C} \star_4 \lambda(G_-) = d(e^{3A+3C-\Phi} \cos\beta v_1 \wedge v_2), \qquad (3.38d) \qquad 21$$

$$\lambda(G_{-}) \wedge v_2 \wedge (\sin \alpha + \cos \alpha w_1 \wedge w_2) - \lambda(G_{+}) \wedge (\sin \alpha + \cos \alpha w_1 \wedge w_2) = 0, \quad (3.38e)$$

by following the same line of reasoning as in the previous subsection. First we solve (3.38b) by using it to define the vielbeine on M_4 locally

$$v_1 = e^A dx_1, \quad w = e^A \sin\beta (dx_2 + idx_3), \quad v_2 = v e^{-A} d\rho, \quad \rho = e^{A+C}, \quad (3.39) \quad 27$$

where we have used the first of (3.38a) to simplify these somewhat. Next (3.38a) is solved when

$$e^{\Phi} = e^{2A} \sin \beta, \quad \cot \beta = ce^{2A}, \quad dc = 0,$$
 (3.40) 30
31

with $A = A(\rho)$, $\beta = \beta(\rho)$. As a result, ∂_{x_i} are isometries. We then use (3.39) to take the Hodge dual of (3.38d), (3.38e) arriving at the fluxes

$$F_3 = -\nu\rho^3 \partial_\rho (e^{-4A}) \operatorname{Vol}(S^3), \quad H = 2e^{2A} \partial_\rho \beta \sin^2 \beta d\rho \wedge dx_2 \wedge dx_3, \tag{3.41}$$

$$F_5 = \operatorname{Vol}_3 \wedge dx_1 \wedge d(e^{2A} \cot \beta) + \nu e^{-2A} \rho^3 \left(\sin(2\beta) \partial_\rho A - \partial_\rho \beta \right) dx_2 \wedge dx_3 \wedge \operatorname{Vol}(S^3),$$

which solve (3.38e) without restriction. The Bianchi identities impose that

$$e^{-4A} = c_1 + \frac{c_2}{\rho^2}, \quad dc_i = 0,$$
(3.42) ³⁹
₄₀

which is again the warp factor of a D5-brane or O5-hole. However, in this case the metric takes the local form

⁴³
⁴⁴
⁴⁵

$$ds^{2} = e^{2A} ds^{2}(\mathbb{R}^{1,3}) + e^{2A} \sin\beta^{2} ds^{2}(T^{2}) + e^{-2A} \left(d\rho^{2} + \rho^{2} ds^{2}(S^{3}) \right)$$
⁴³
⁴³
⁴³
⁴⁴
⁴⁵
⁴³
⁴³
⁴⁵

where T^2 is spanned by (x_2, x_3) . This generalises the solution in the previous section by in-troducing an additional warping factor for a T^2 submanifold, thus breaking SO(1,5) Lorentz

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

	ARTICLE IN PRESS	
JID:NUPHB A	ID:14363 /FLA [m1+; v1.285; Prn:6/06/2018;	; 11:31] P.14 (1-49)
14	N.T. Macpherson et al. / Nuclear Physics $B \bullet \bullet \bullet (\bullet \bullet \bullet \bullet) \bullet \bullet \bullet - \bullet \bullet \bullet$	
		. 1.0 1
symmetry ar	id leading to more general fluxes. In fact, this solution can be gene	rated from the
D5-brane so	lution of the previous section via "G-structure rotation" [42] which	n is formally a
U-duality [4.	3].	
3.7 Branch	II: or non zero solutions	
J.2. Drunch	II. a non-zero solutions	
For the se	could branch with $0 < \alpha < \pi$, we begin by studying the lower form	conditions that
follow from	(3.1) Here we find it useful to rotate the canonical frame of (B.3) as	conditions that
Tonow nom		
$v \rightarrow -$	$-\sin w_1 + \cos \phi (\cos \theta v_1 - \sin \theta w_2) + i v_2,$	(3.44)
$w \to c$	$\cos\phi w_1 + \sin\phi(\cos\theta v_1 - \sin\theta w_2) + i(\cos\theta w_2 + \sin\theta v_1).$	(3.45)
We then find	the following necessary, but not sufficient, conditions for supersymm	netrv
$\cos\beta$ s	$\ln \delta H_0 = 0 ,$	(3.46a)
² C 005	$\alpha \sin \beta d\theta = 2u(\cos \beta \cos \delta u + \sin \beta \sin \alpha u) = 0$	(3.46b)
e cos	$u\sin pu v = 2v(\cos p\cos v v_1 + \sin p\sin u w_1) = 0,$	(3.400)
$e^C \cos$	$\alpha \sin \beta \sin \theta d\phi - 2\nu (\cos \beta \cos \delta w_1 - \sin \alpha \sin \beta v_1) = 0,$	(3.46c)
2.4		
$d(e^{2A})$	$-2ve^{2A+C-\Psi}(\cos\beta\cos\delta w_2-\sin\beta v_2),$	(3.46d)
$d(a^{2A+})$	$-3C-\Phi(\sin \alpha \sin \beta u_{2} + \sin (\alpha - \beta) \cos \beta u_{2})) + e^{2A+3C-\Phi}H_{0}\sin \beta \cos \beta u_{2}$	$\alpha w = 0$
u(e	$(\sin \alpha \sin p w_2 + \sin(\alpha - b)\cos p v_2)) + e$ $H_0 \sin p \cos b v_2$	$uv_1 \wedge w_1 = 0,$ (3.46e)
		(3.160)
$d(e^{3A+})$	$e^{-2C-\Phi}(\sin\beta\sin\alpha v_2 - \cos\beta\sin(\alpha - \delta)w_2))$	
$+2\nu e$	$^{3A+C-\Phi}\cos\beta(\sin\delta w_2 \wedge v_2 + \cos(\alpha - \delta)v_1 \wedge w_1)$	
$-e^{3A}$	$^{+2C-\Phi}(d\theta \wedge (\cos\beta\sin(\alpha-\delta)v_1+\sin\beta w_1))$	
	(1 + 1)	(2.466)
$+\sin \theta$	$\partial a \phi \wedge (\cos \beta \sin(\alpha - \delta) w_1 - \sin \beta v_1)) = 0.$	(3.401)
First we note	that if either θ or ϕ become constant or if $\cos \alpha = 0$ then (3.46b),	(3.46c) require
that $\sin \beta = 0$	$\cos \delta = 0$ which makes (θ, ϕ) drop out of (3.11) entirely and the final	line of $(3.46f)$
vanishes (set	ting $\sin \theta = 0$ leads to the same conclusion). In this case we can co	nclude that we
can set		
$H_0 = \theta$	$\theta = \phi = (\delta - \frac{\pi}{2}) = 0$	(3.47)
110	¢ (° 2 [']	(8117)
without loss	of generality, which we study in section 3.2.1.	
If we assu	time $\sin \theta$ and $\cos \alpha$ don't vanish, then (θ, ϕ) are local coordinates on	a 2-sphere and
we can take	$\rho = e^{2A+2C-\Phi} \cos \alpha \sin \beta$ as a local coordinate. We can then use (3.4)	(6b)-(3.46d) to
rewrite (3.46	f) as	
$d(e^{3A+})$	$e^{-2C-\Phi}(\sin\beta\sin\alpha v_2 - \cos\beta\sin(\alpha - \delta)v_2))$	
	$3A+C-\Phi \cos \theta \left(\sin \theta \cos \phi \sin (\cos \theta) \cos \phi \sin (\cos \theta) \right) = 0$	(2 10)
+ 2ve	$\cos p \left(\sin \delta w_2 \wedge v_2 - \cos(\alpha - \delta)v_1 \wedge w_1 \right) = 0,$	(3.48)
which we can	then use to fix some of the free functions. First we note that if we sol	ve (3.46a) with
$\sin \delta = 0$, the	$\sin (3.48)$ fixes $\cos \beta = 0$ – this is because $\sin \beta \sin \alpha v_2 - \cos \beta \sin \alpha u$	v_2 is parallel to
$d\rho$ in this lim	it and so cannot generate the Vol(S^2) factor that comes from $v_1 \wedge w_1$.	Next, for $H_0 =$

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.15 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••)

0 and for generic values of (α, β, δ) we can use (3.46b)–(3.46e) to locally define the vielbein on M_4 by introducing another local coordinate such that $dx = e^{2A+3C-\Phi}(\sin\alpha\sin\beta w_2 + \sin\alpha - \omega)$ $\delta \cos \beta v_2$), but then we must once more set the $v_1 \wedge w_1$ term in (3.48) to zero which fixes either $\cos \beta = 0$ or $\cos(\alpha - \delta) = 0$. Thus, for $H_0 = 0$ and a priori generic $(\theta, \phi, \alpha, \beta, \delta)$, we end up with just two cases. Firstly $\cos \beta = 0$, which solves (3.46a) and makes the δ dependence of (3.11) drop out such that we can set without loss of generality $\delta = 0$, $\beta = \frac{\pi}{2}$. (3.49)We shall examine this case in detail in section 3.2.2 where we find that it contains no solution. Secondly $\cos(\alpha - \delta) = 0$, such that we can set without loss of generality $\delta = \alpha + \frac{\pi}{2} ,$ (3.50)which we shall study in section 3.2.3, finding a new class of solution. There is one final option one can consider for $H_0 = 0$, by taking both $\cos \alpha$ and $\sin \theta$ non-vanishing – one can tune the values of (α, δ, β) such that $(\sin \alpha \sin \beta w_2 + \sin(\alpha - \delta) \cos \beta v_2)$ becomes parallel to $d\rho$ and so can no longer be used to introduce a local coordinate. This re-quires fixing $\tan \beta = \sqrt{\frac{\cos \delta \sin(\delta - \alpha)}{2}}$ (351)

It will then be
$$(3.48)$$
 that will be used to define the final vielbein direction, which will necessarily

be fibred over S^2 . We shall examine this possibility in section 3.2.4.

²⁴ 3.2.1. Subcase
$$\beta = 0$$
 24
²⁵ Ear $\beta = 0$ 25

For $\beta = 0$ the supersymmetry conditions reduce to 24120 4

$$d(e^{2A-\Phi}) = d(e^{2A+3C-\Phi}\cos\alpha v_2) = H_3 = H_0 = 0, \qquad (3.52a) \qquad 27$$

$$d(e^{3A+2C-\Phi}\cos\alpha u_i) + 2\nu e^{3A+C-\Phi}(u_i \wedge v_2 + \frac{1}{2}\sin\alpha\epsilon_{ijk}u_j \wedge u_k) = 0, \qquad (3.52b) \qquad 29$$

$$d(e^{2A+2C-\Phi}(\sin\alpha u_i \wedge v_2 + \frac{1}{2}\epsilon_{ijk}u_j \wedge u_k)) - ve^{2A+C-\Phi}\cos\alpha\epsilon_{ijk}u_j \wedge u_k \wedge v_2 = 0, \qquad 31$$
(3.52c)
(3.52c)

$$e^{3A} \star_4 \lambda(G_+) = d(e^{3A-\Phi} \cos \alpha v_1 \wedge w_1 \wedge w_2), \ e^{3A+3C} \star_4 \lambda(G_-) = -d(e^{3A+3C-\Phi} \sin \alpha), \tag{3.52d}$$

 $\left(\cos\alpha v_2 \wedge \lambda(G_-) + \sin\alpha v_1 \wedge v_2 \wedge w_1 \wedge w_2 \lambda(G_+)\right)\Big|_4 = 0,$ (3.52e)

where as usual $u = (v_1, w_1, w_2)$. Note that this system reduces to that of section 3.1.2 when $\sin \alpha = 0$, and that (3.52b) imposes that $\cos \alpha \neq 0$, so we can take $0 < 2\alpha < \pi$. By adding linear combinations of wedge products of (3.52a), (3.52b) and the vielbein to (3.52b), it is then possible to derive enough independent 2-form conditions to establish that

$$d(e^{A-C}\sin\alpha) \wedge v_2 = d\alpha \wedge v_2 = d(e^{A+C}\cos\alpha) - ve^A v_2 = d(e^{-A}\sec\alpha u_i) \wedge u_j = 0, \ i \neq j.$$
(3.53)

(3.53)

This is sufficient to establish that e^{2A} , e^{2C} and α are functions of a single local coordinate ρ , which v_2 is parallel to – specifically

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

з

CLE IN PRESS JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.16 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••

$$\rho = e^{A+C} \cos \alpha , \quad v_2 = v e^{-A} d\rho . \tag{3.54}$$

The final condition in (3.53) implies that $d(e^{-A} \sec \alpha u_1) \propto u_2 \wedge u_3$ and cyclic permutations, however plugging this into (3.52b) and (3.52b) we realise we can without loss of generality take

$$u_{i} = \frac{c_{1}}{2} e^{A} \cos \alpha \tilde{K}_{i} , \quad \tilde{K}_{i} + \frac{\nu}{2} \epsilon_{ijk} \tilde{K}^{j} \wedge \tilde{K}^{k} , \quad c_{1} = \frac{e^{-A+C}}{\sin \alpha} , \quad dc_{1} = 0 , \quad (3.55) \quad {}^{5}_{6}$$

where \tilde{K}^i are necessarily SU(2) invariant forms which furnish a frame for a round S^3 . We have now without loss of generality determined the vielbein on M_4 , which is a foliation of S^3 over an interval, and (3.52a)-(3.52c) are solved when

$$e^{-2A} = \frac{c_1 \sin \alpha \cos \alpha}{\rho} , \quad e^{2C} = e^{2A} c_1^2 \sin^2 \alpha , \quad e^{2A-\Phi} = c_2 , \quad dc_i = 0 . \tag{3.56}$$

The only non-trivial 10d flux can be extracted from (3.52d) and is given by

$$F_3 = 2c_1^2 c_2 \nu \left(\sin^2 \alpha - \rho \tan \alpha \partial_\rho \alpha\right) \operatorname{Vol}(S^3) + 2c_1^2 c_2 \nu \left(\cos^2 \alpha + \rho \cot \alpha \partial_\rho \alpha\right) \operatorname{Vol}(\tilde{S}^3) .$$
(3.57)

The pairing equation (3.52e) is equivalent to the Bianchi identity at this point; either one implies

$$d\alpha = 0$$
. (3.58) ₁₉
The metric is of the form

The metric is of the form

$$ds^{2} = e^{2A}ds^{2}(\mathbb{R}^{1,2}) + e^{-2A} \left[d\rho^{2} + \frac{\rho^{2}}{\cos^{2}\alpha} ds^{2}(S^{3}) + \frac{\rho^{2}}{\sin^{2}\alpha} ds^{2}(S^{3}) \right]$$
(3.59)

This solution has both an SO(4) R-symmetry and SO(4) flavour symmetry, and is S-dual to the one that we find in section 6, as will be explained in that section.

3.2.2. Subcase
$$\beta = \frac{\pi}{2}$$

Here one can show that supersymmetry implies

$$d(e^{2A+2C-\Phi}\cos\alpha) - 2\nu e^{2A+C-\Phi}\nu_2 = d(e^C\cot\alpha w_2) = 0, \qquad (3.60a) \qquad {}^{30}_{31}$$

$$e^{C}d\theta - 2\nu\tan\alpha w_{1} = e^{C}\sin\theta d\phi + 2\nu\tan\alpha v_{1} = 0, \qquad (3.60b) \quad 32$$

$$d(e^{-A-C}\sin\alpha) = d(e^{\Phi}\sin\alpha\tan\alpha) \wedge v_2 = d(e^{-A-C+\Phi}\tan\alpha) \wedge w_2 \wedge v_2 = 0, \quad (3.60c) \quad {}_{34}$$

$$d(e^{A+C-\frac{\Phi}{2}}\tan\alpha\sqrt{\cos\alpha}) \wedge w_2 + \frac{1}{2}\sqrt{\cos\alpha}e^{A+C-\frac{\Phi}{2}}H_0v_1 \wedge w_1 = 0, \qquad (3.60d)$$

$$d(e^{2A+2C-\Phi}\cos\alpha\cot^2\alpha) \wedge \text{Vol}(S^2) = 0, \qquad (3.60e) \qquad 38$$

$$H_3 + \frac{v}{2}e^C \cot \alpha w_2 \wedge \operatorname{Vol}(S^2) = d(e^{-2C} \tan \alpha) \wedge v_2 \wedge \operatorname{Vol}(S^2) = 0, \qquad (3.60f) \quad {}^{40}$$

$$e^{3A} \star_4 \lambda(G_+) - d(e^{3A - \Phi} w_2) + d(e^{3A - \Phi} \sin \alpha v_1 \wedge v_2 \wedge w_1) = 0, \qquad (3.60g) \quad {}^{42}_{43}$$

$$e^{3A+3C} \star_4 \lambda(G_-) + e^{3A+3C-\Phi} H_0 w_2 + d(e^{3A+3C-\Phi} \cos \alpha v_2 \wedge w_2) = 0, \qquad (3.60h) \quad 44$$

$$\begin{cases} 45 \\ 46 \\ 47 \end{cases} \left(\left(\sin \alpha w_2 - v_1 \wedge v_2 \wedge w_1 \right) \wedge \lambda(G_-) - \cos \alpha v_1 \wedge v_2 \wedge \lambda(G_+) \right) \Big|_4 = 0 . \\ (3.60i) \\ 47 \end{cases}$$

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

ARTICLE IN PRESS

JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.17 (1-49)

N.T. Macpherson et al. / Nuclear Physics $B \bullet \bullet \bullet (\bullet \bullet \bullet \bullet) \bullet \bullet \bullet - \bullet \bullet \bullet$

There is no solution to this set of constraints, as we will now show. First, we note that $H_0 = 0$ is imposed by (3.60d). Let $\rho = e^{2A+2C-\Phi} \cos \alpha$. Due to (3.60a), we have $w_2 \sim dx$, $v_2 \sim d\rho$. It is then possible to rewrite (3.60d), (3.60e) as $d(\sqrt{\rho}\tan\alpha) \wedge dx = 0$, $d(\rho\cot^2\alpha) \wedge d\theta \wedge d\phi = 0$. (3.61)The first equation implies $\tan \alpha = \rho^{-1/2} f(x)$, which is incompatible with the second equation – thus this putative class contains no solutions. 3.2.3. Subcase $\delta = \alpha + \frac{\pi}{2}$, $H_0 = 0$ As explained below (3.46f), here we necessarily have $\beta > 0$ and $0 < \alpha < \frac{\pi}{2}$ – for this reason it will turn out the case contains no solutions. As the proof is similar to that of the previous section we shall be brief, this time only quoting sufficient supersymmetry conditions to prove this. In addition to the rotation of (B.3) we find it useful to send $v_1 + iw_1 \rightarrow e^{-i\beta}(v_1 + iw_1)$, then a set necessary (but insufficient) conditions for supersymmetry are $d(e^{2A-\Phi}\cos\alpha\cos\beta) = 0,$ (3.62a) $d(e^C \cot \alpha w_2) = 0,$ (3.62b) $(v_1 + iw_1)\tan\alpha + \frac{e^C\sin\beta}{2\nu}(d\theta + i\sin\theta d\phi), \quad d(e^{2A+3C-\Phi}(\sin\alpha\sin\beta w_2 - \cos\beta v_2)) = 0,$ (3.62c) (3.62c) $d(e^{2A+2C-\Phi}\cos\alpha\sin\beta) - 2\nu e^{2A+C-\Phi}(\cos\beta\sin\alpha w_2 + \sin\beta v_2) = 0,$ (3.62d) $e^{2A-\Phi}\cos\alpha\cos\beta H_3 + d(e^{2A-\Phi}\cos\alpha\sin\beta v_1 \wedge w_1) = 0,$ (3.62e) $e^{2A+3C-\Phi}(\sin\alpha\sin\beta w_2-\cos\beta v_2)\wedge H_3$ $+ d(e^{2A+3C-\Phi}v_1 \wedge w_1 \wedge (\sin\beta v_2 - \cos\beta\sin\alpha w_2)) = 0.$ (3.62f) As elsewhere we can take (3.62c)-(3.62d) as a local definition of the vielbein without loss of generality $-v_1$, w_1 are clearly the local vielbeine of a round S^2 in terms of the local coordinates (θ, ϕ) . For v_2 , w_2 we introduce local coordinates x and $\rho = e^{A+C-\frac{1}{2}\Phi}\sqrt{\cos\alpha\sin\beta}$ such that $e^{2A+3C-\Phi}(\sin\alpha\sin\beta w_2 - \cos\beta v_2) = dx.$ (3.63)We can then use (3.62e) to define H_3 without loss of generality, which leaves (3.62a), (3.62b)and (3.62f) to solve – this turns out to be impossible. To see this, one needs to consider the combination $4(3.62f) + (f_1(3.62a) \land d\rho + f_2(3.62a)) \land Vol(S^2)$. When one tunes $f_1 = e^{-A+5C+\frac{\Phi}{2}} \nu \frac{\cos^{\frac{3}{2}} \alpha \sin^{\frac{5}{2}} \beta}{\sin^2 \alpha \cos \beta}, \quad f_2 = e^{2A+4C-\Phi} \cos \alpha \sin \beta \tan \beta,$ (3.64)this leads to $\cot\alpha \csc\alpha \sin\beta \tan\beta dx \wedge d\rho \wedge \operatorname{Vol}(S^2) = 0$ (3.65)which cannot be solved without violating the initial assumptions that lead to this case. We con-clude that there exist no solutions. Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

з

CLE IN PRESS \RT JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.18 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••

3.2.4. Subcase: special value of β , $H_0 = 0$

The final case in IIB requires us to tune $\tan \beta$ to a specific value. After redefining $\delta \rightarrow \delta + \alpha$, this value is

$$\tan \beta = \sqrt{\frac{\cos(\alpha + \delta)\sin\delta}{\sin\alpha}}.$$
(3.66)

In addition, we rotate the vielbein (with respect to (3.44)) as

$$v_1 + iw_1 \rightarrow -i \frac{\cos\beta\cos(\alpha + \delta) + i\sin\alpha\sin\beta}{\sqrt{\cos^2\beta\cos^2(\alpha + \delta) + \sin^2\alpha\sin^2\beta}} (v_1 + iw_1),$$
⁸
⁹
¹⁰
¹¹

$$v_2 + iw_2 \to -i \frac{\sin\alpha \sin\beta + i\cos\beta \sin\delta}{\sqrt{\sin^2\alpha \sin^2\beta + \cos^2\beta \sin^2\delta}} (v_2 + iw_2)$$
(3.67)
¹²
¹³
¹³
¹⁴

In what follows we assume that the undefined functions of the spinor ansatz are bounded as $0 < 2\alpha < \pi$ and $0 < \delta + \alpha < \frac{\pi}{2}$, as the upper and lower limits have been dealt with in the proceeding sections. It is then possible to show that the necessary and sufficient conditions for supersymmetry for this case are

.

$$d(e^{A-C}\sin\alpha) = d\left(e^{2A-\Phi}\sqrt{\frac{\sin\alpha\sin(\alpha+\delta)}{\cos\delta}}\right) = d\left(e^{-\Phi}\sqrt{\frac{\sin 2(\alpha+\delta)}{\sin 2\delta}}\right) = 0, \quad (3.68a) \quad \stackrel{19}{\underset{21}{22}}$$

$$2\nu e^{-C}\sqrt{\tan\alpha\cot\delta}(w_1 - i\nu_1) - (d\theta + i\sin\theta d\phi) = 0,$$
(3.68b)
(

$$2\nu d\left(e^{-C}\tan\alpha\sqrt{\frac{\cos\alpha\sin(\alpha+\delta)}{\sin\delta}}w_2\right) - \operatorname{Vol}(S^2) = d\left(e^{A+C}\sqrt{\frac{\cos\alpha\sin\delta}{\sin(\alpha+\delta)}}\right) - \nu e^A v_2 = 0, \quad 26$$
(3.68c) 27
(3.68c) 27

$$H_3 + \frac{1}{4}d\left(e^{2C}\frac{\cos^2\alpha\sin\delta}{\sin^2\alpha\sin(\alpha+\delta)\cos\delta}\sqrt{\cos(\alpha+\delta)\sin\alpha\sin\delta}\right) \wedge \operatorname{Vol}(S^2) = d\alpha \wedge v_2 = 0$$
(3.68d)
$$(3.68d)$$

$$e^{3A} \star_4 \lambda(G_+) - d_{H_3} \left(e^{3A - \Phi} \sqrt{\frac{\cos \alpha \cos(\alpha + \delta)}{\cos \delta}} \right)$$

$$-d\left(e^{3A-\Phi}\sqrt{\frac{\sin\alpha\cos\alpha\sin\delta}{\cos\delta}}v_1\wedge w_1\wedge w_2\right) = 0, \qquad (3.68e) \qquad \begin{array}{c} 36\\ 37\\ 38 \end{array}$$

$$-d\left(e^{3A+3C-\Phi}\cos\alpha\sqrt{\frac{\sin\delta\cos(\alpha+\delta)}{\cos\delta\sin(\alpha+\delta)}}v_2\wedge w_2\right) = 0,$$
43
44
45

$$\begin{array}{c} 45 \\ 46 \\ 47 \end{array} \qquad \left(\left(\sqrt{\frac{\sin\alpha\cos\alpha\sin\delta}{\cos\delta}} v_2 - \sqrt{\frac{\cos\alpha\cos(\alpha+\delta)}{\cos\delta}} v_1 \wedge w_1 \wedge v_2 \right) \wedge \lambda(G_-) \right) \\ (3.68f) \\ 47 \end{array}$$

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

ARTICLE IN PRESS

 JID:NUPHB
 AID:14363 /FLA
 [m1+; v1.285; Prn:6/06/2018; 11:31] P.19 (1-49)

 N.T. Macpherson et al. / Nuclear Physics B ••• (••••)
 19

 $\left(\frac{\sin\alpha\sin(\alpha+\delta)}{\sin\delta\cos(\alpha+\delta)}\right)$

$$-\left(\sqrt{\frac{1}{\cos\delta}-\cos\alpha}-\cos\alpha\sqrt{\frac{1}{\cos\delta}\sin(\alpha+\delta)}v_1\wedge w_1\right)$$

$$-\sqrt{\frac{\sin\alpha\cos\delta}{\sin(\alpha+\delta)}}v_1\wedge v_2\wedge w_1\wedge w_2\big)\wedge\lambda(G_+)\bigg)\bigg|_4=0,$$

⁷ where Vol(S^2) is the volume form on the S^2 spanned by (θ , ϕ). We solve these conditions by ⁸ first using (3.68b)–(3.68c) to define the vielbein locally without loss of generality as

$$w_1 = \frac{\nu}{2} e^C \sqrt{\frac{\cos\alpha\sin\delta}{\sin\alpha\cos\delta}} d\theta , \quad v_1 = -\frac{\nu}{2} e^C \sqrt{\frac{\cos\alpha\sin\delta}{\sin\alpha\cos\delta}} \sin\theta d\phi , \quad (3.69)$$

$$w_2 = -\frac{\nu}{2}e^C \cot\alpha \sqrt{\frac{\sin\delta}{\sin(\alpha+\delta)\cos\alpha}} (d\psi + \cos\theta d\phi) , \quad v_2 = \nu e^{-A}d\rho , \quad (3.70)$$

where we have taken θ , ϕ as local coordinates and introduced the additional coordinates ψ and

$$\rho = e^{A+C} \sqrt{\frac{\cos\alpha\sin\delta}{\sin(\alpha+\delta)}} \,. \tag{3.71}$$

We can invert this conditions then use (3.68a) to define A, C, Φ, δ in terms of α, ρ and some integration constants c_i as

$$e^{-4A} = \frac{c_1^2 \cos^2 \alpha \sin^2 \alpha}{\rho^2} - \frac{c_3 \sin^2 \alpha}{c_2^2}, \quad e^{2C} = c_1^2 \sin^2 \alpha e^{2A},$$

$$e^{-2\Phi} = e^{-4A} \left(c_2^2 + \frac{c_3 \rho^2}{c_1^2 \sin \alpha^2} \right), \quad \cot(\alpha + \delta) = \frac{c_3 \rho^2}{c_1^2 c_2^2 \sin \alpha \cos \alpha}.$$
(3.72)

The second equality in (3.68d) implies that α is itself a function of ρ only, so we realise that ∂_{ψ} is an isometry of the solution and that M_4 is foliation of a ($SU(2) \times U(1)$ preserving) squashed 3-sphere over an interval.

We now turn our attention to the fluxes. We have that (3.68d) simply defines the NSNS flux in such a way that it is automatically closed, while the RR fluxes are defined through the 4d fluxes that follow from (3.68e). We could use the definitions of the functions in (3.72) and vielbein in (3.69) to calculate the 10d fluxes immediately, however we already have enough information to first fix α . The 3-form component of $\star 4\lambda(G_{-})$ is necessarily parallel to $v_1 \wedge w_1 \wedge v_2$ from which it follows that the 10d flux F_1 is parallel to w_2 . As this vielbein is fibred over the S^2 we have that $dF_1 = 0$ iff $(\star 4\lambda(G_{-}))_3 = 0$. For $\cos(\alpha + \delta) \neq 0$ this imposes $d(c_1^2 c_2^2 \sin^2 \alpha + c_3 \rho^2) = 0$, which implies that

$$\sin^2 \alpha = c_4 - \frac{c_3 \rho^2}{c_1^2 c_2^2}, \quad dc_i = 0, \quad (3.73)$$

and as a result, every function has been solved in terms of ρ and the four integration constants c_i . We are now ready to calculate the fluxes. The non-trivial ones take the form

$$B = \sqrt{c_3} \frac{1 - c_4}{4c_4 c_2} \rho^2 \operatorname{Vol}(S^2) , \quad F_3 = \frac{1}{4} \nu c_1^2 c_2 (1 - c_4) d\psi \wedge \operatorname{Vol}(S^2) + 2\nu c_1^2 c_2 c_4 \operatorname{Vol}(S^3) , \quad \overset{45}{46}$$
(3.74)

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

з

ARTICLE IN PRESS

 JID:NUPHB
 AID:14363 /FLA
 [m1+; v1.285; Prn:6/06/2018; 11:31] P.20 (1-49)

 20
 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••

$$F_5 = B \wedge F_3 + c_1 c_2^2 \nu \sqrt{c_3} d\left(\frac{\rho^2}{2(c_3 \rho^2 - c_1^2 c_2^2 c_4)} (d\psi + \cos\theta d\phi)\right)$$

g

$$+\frac{\nu c_1^2 \sqrt{c_3 c_4}}{2} d\left(\rho^2 (d\psi + \cos\theta d\phi)\right) \wedge \operatorname{Vol}(S^3) ,$$

where dB = H. Clearly the Bianchi identities of the fluxes are implied automatically and one can show that this is true of (3.68f) also. So this case contains a single example, expressed in terms of 4 integration constants. The 10d metric, warp factor and dilation then take the form

$$ds^{2} = e^{2A}ds^{2}(\mathbb{R}^{1,2}) + e^{-2A} \bigg[d\rho^{2} + \frac{1}{1 - c_{4}}\rho^{2}ds^{2}(S^{3}) + \frac{\rho^{2}}{4(c_{4} - \frac{c_{3}}{c_{1}^{2}c_{2}^{2}}\rho^{2})} (d\psi + \cos\theta d\phi)^{2}$$

$$+\frac{\rho^2}{4c_4}ds^2(S^2)\bigg], \quad e^{-4A} = (1-c_4)\left(\frac{c_1c_4}{\rho^2} - \frac{c_3}{c_2^2}\right), \quad e^{-4A+2\Phi} = \frac{1}{c_2^2} - \frac{c_3}{c_1^2c_2^4c_4}\rho^2.$$
(3.75)

This solution preserves an SO(4) R-symmetry realised by one SU(2) factor of the round S^3 and the SU(2) of the squashed sphere – the residual symmetries of the spheres make up an $SU(2) \times U(1)$ flavour symmetry. Despite our assumption that $\alpha + \delta \neq \frac{\pi}{2}$ (which is when $c_3 =$ 0) when deriving (3.68a)–(3.68f) there is in fact no issue with taking this limit, which merely collapses this solution to that of section 3.2.1. There is good reason for this, as one can actually generate this solution from section 3.2.1 by first T-dualising on ∂_{ψ} then performing a formal U-duality⁶ on the Mink₃ followed by another T-duality on ∂_{ψ} . Additionally, this solution is also contained in section 4.2.2: it can be obtained by imposing that the coordinate x there (which should be identified with ψ in this section) is an isometry and then T-dualising it.

This concludes our IIB classification, we shall now turn our attention towards IIA.

4. Mink₃ with an S^3 factor in IIA

The type IIA supersymmetry conditions obtained from plugging (2.14) into the sevendimensional supersymmetry constraints (2.4) lead to the following constraints on the fourdimensional bispinors

$$d_{H_3}(e^{2A-\Phi}\mathrm{Im}\psi_-^1) = 0, \qquad (4.1a) \quad {}^{34}_{35}$$

 $d_{H_3}(e^{2A+2C-\Phi}\psi_-^2) + 2i\nu e^{2A+C-\Phi}\psi_{\hat{\gamma}+}^2 = 0, \qquad (4.1b) \qquad {}^{36}_{37}$

$$d_{H_3}(e^{3A+2C-\Phi}\psi_+^2) + 2i\nu e^{3A+C-\Phi}\psi_{\hat{\gamma}-}^2 = 0, \qquad (4.1c) \quad \frac{^{38}}{^{39}}$$

$$d_{H_3}(e^{2A+2C-\Phi}\operatorname{Re}\psi_-^1) - 2\nu e^{2A+C-\Phi}\operatorname{Im}\psi_{\hat{\gamma}+}^1 = 0, \qquad (4.1d) \quad {}^{40}_{41}$$

$$d_{H_3}(e^{2A+3C-\Phi}\operatorname{Re}\psi^1_{\hat{\gamma}_+}) - e^{2A+3C-\Phi}H_0\operatorname{Im}\psi^1_- = 0, \qquad (4.1e) \qquad 42$$

 $\overline{^{6}}$ T-dualities along the spatial components of Mink₃, then a lift to M-theory followed by a boost along the M-theory U(1) before reducing to IIA, and finally undoing the spatial T-dualities. This process needs to be supplemented by rescaling the coordinates along the way, which is why we refer to this as a formal U-duality.

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

NT Macherson et al / Nuclear Physics R ere (erea) erea	.əıj r .∠I (1-49) 21
	21
$d_{H_3}(e^{3A+2C-\Phi} \mathrm{Im}\psi^1_+) + 2\nu e^{3A+C-\Phi} \mathrm{Re}\psi^1_{\hat{\gamma}-} = 0 ,$	(4.1f)
while the fluxes are defined through	
$d_{H_3}(e^{3A-\Phi} \operatorname{Re} \psi^1_+) + e^{3A} \star_4 \lambda(G) = 0 ,$	(4.2a)
$d_{H_3}(e^{3A+3C-\Phi}\mathrm{Im}\psi^1_{\hat{\gamma}-}) + e^{3A+3C-\Phi}H_0\mathrm{Re}\psi^1_+ + \nu e^{3A+3C} \star_4 \lambda(G_+) = 0,$	(4.2b)
and must additionally satisfy	
$\left(\operatorname{Re}\psi_{\hat{\gamma}+}^{1} \wedge \lambda(G_{+}) + \operatorname{Im}\psi_{-}^{1} \wedge \lambda(G_{-})\right)\Big _{4} = 0$	(4.3)
As before, we will first examine the 0-form constraints. These are given by two constraints that were found for type IIB:	of the three
$(\psi_{\hat{\gamma}}^2)_0 = (\mathrm{Im}\psi_{\hat{\gamma}}^1)_0 = 0.$	(4.4)
Again, the solutions branch off similar to type IIB, with an $\alpha = 0$ and an $\alpha \neq 0$ parameterise Branch I as in (3.9) and Branch II as in (3.11).	branch. We
4.1. Branch I: solutions with $\alpha = 0$	
As was the case in IIB we first study the lower form conditions that follow from once more rotating the canonical frame of $(B.3)$ by (3.13) we extract the necess sufficient, supersymmetry conditions	(3.1). After ary, but not
$d(e^{2A+3C-\Phi}\cos\beta\sin\delta) + e^{2A+3C-\Phi}H_0\cos\beta w_1 = 0,$	(4.5a)
$e^{2A+3C-\Phi}H_0\sin\beta(\cos\delta w_1-\sin\delta v_2)\wedge v_1\wedge w_2+\cos\beta()=0,$	(4.5b)
$d(e^{2A-\Phi}\cos\beta w_1) = d(e^{3A+2C-\Phi}\cos\beta\cos\delta) - 2\nu e^{3A+C-\Phi}\cos\beta v_2 = 0,$	(4.5c)
$\cos\beta(\sin\delta v_1 + \frac{\nu}{2}e^C\cos\delta d\theta) = \cos\beta(\sin\delta w_2 - \frac{\nu}{2}e^C\cos\delta\sin\theta d\phi) = 0,$	(4.5d)
$d(e^{2A+2C-\Phi}\sin\beta(\cos\delta w_1-\sin\delta v_2))+2\nu e^{2A+C-\Phi}(\sin\beta w_1\wedge v_2-\cos\beta\sin\theta)$	$(\delta v_1 \wedge w_2)$
$-\cos\beta\cos\delta e^{2A+2C-\Phi}(d\theta\wedge w_2+\sin\theta d\phi\wedge v_1)=0,$	(4.5e)
where $\cos \beta()$ represents further terms which vanish when $\cos \beta = 0$. These are runcate the ansatz considerably. We note from (4.5a) that if $\sin \delta = 0$ then either $\cos \beta = 0$, however the latter also leads to $H_0 = 0$ because of (4.5b) – so $\sin \delta =$ $H_0 = 0$. We also observe that if $\sin \delta = 0$ then (4.5c) requires $d\theta = d\phi = 0$, or naive but this is a subcase of the former (see (3.9)), so $\sin \delta = 0$ also implies $d\theta = d\phi =$ now is to show that, as in IIB, $\sin \delta = 0$ is a necessary condition: first we note th $\cos \delta = 0$ then there is no solution as (4.5c) sets the vielbein to zero, thus we can considerations to $0 < \sin \delta < \frac{\pi}{2}$ where (w_1, v_2) must span an S^2 . However, as was IIB, (4.5e) can be rewritten as	sufficient to $H_0 = 0$ or = 0 implies $\log \cos \theta = 0$ 0. Our task that if we set restrict our s the case in

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021
CLE IN PRESS JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.22 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••

using (4.5d) which excludes this because $w_1 \wedge v_2$ gives rise to a Vol(S²) term that can not be cancelled by the parts involving (v_1, w_2) . Thus we can once more conclude that $H_0 = \delta = \theta = \phi = 0 \; .$ (4.7)Given this, we can write the necessary and sufficient solutions for supersymmetry in the $\alpha = 0$ limit in a relatively simple way. After rotating the canonical frame, this time as in (3.16), we find $H_0 = d(e^{2A-\Phi}\cos\beta v_1) = d(e^{\frac{3}{2}A+C-\frac{\Phi}{2}}\sqrt{\cos\beta}) - ve^{\frac{3}{2}A-\frac{\Phi}{2}}\sqrt{\cos\beta}v_2 = 0,$ $d(e^{2A+2C-\Phi}w) + 2\nu e^{2A+C-\Phi}w \wedge v_2$ $= d(e^{2A+2C-\Phi}\sin\beta v_1) + 2\nu e^{2A+C-\Phi}\sin\beta v_1 \wedge v_2 = 0,$

$$d(e^{3A+2C-\Phi}v_1 \wedge w) - 2ve^{3A+2C-\Phi}v_1 \wedge w \wedge v_2 = d(e^{3A-\Phi}\cos\beta) \wedge v_1 \wedge v_2 = 0,$$

$$d(e^{3A+2C-\Phi}\sin\beta w_1 \wedge w_2) - 2ve^{3A+2C-\Phi}\sin\beta w_1 \wedge w_2 \wedge v_2 + e^{3A+2C-\Phi}\cos\beta H_3 = 0,$$

$$e^{2A}H_3 \wedge v_1 + \cos^2\beta d(e^{2A}\tan\beta) \wedge v_1 \wedge w_1 \wedge w_2 = d(e^{-A+\Phi}\sin\beta) \wedge v_1 \wedge w_1 \wedge w_2$$
17

$$e^{2A}H_3 \wedge v_1 + \cos^2\beta d(e^{2A}\tan\beta) \wedge v_1 \wedge w_1 \wedge w_2 = d(e^{-A+\Phi}\sin\beta) \wedge v_1 \wedge w_1 \wedge w_2$$

= $w \wedge H_3 = 0$,

$$d(e^{3A-\Phi}\sin\beta) + d(e^{3A-\Phi}\cos\beta w_1 \wedge w_2) - e^{3A-\Phi}\sin\beta H_3 - e^{3A} \star_4 \lambda(G_-) = 0,$$

$$d(e^{3A+3C-\Phi}\sin\beta v_2) + d(e^{3A+3C-\Phi}\cos\beta w_1 \wedge w_2 \wedge v_2)$$
22
23
23

$$-e^{3A+3C-\Phi}\sin\beta H_3 \wedge v_2 - e^{3A+3C} \star_4 \lambda(G_+) = 0, \qquad 24$$

This is as far as we can go without making assumptions about β , which we now proceed to do. 4.1.1. Subcase: $\beta = 0$

Setting
$$\beta = 0$$
 in (4.8) immediately leads to $H_3 = 0$, the rest of the conditions are implied by
 $3 \pm \beta = 0$ in (4.8) immediately leads to $H_3 = 0$, the rest of the conditions are implied by

$$d(e^{\frac{3}{2}A+C-\frac{\Phi}{2}}) - ve^{\frac{3}{2}A-\frac{\Phi}{2}}v_2 = d(e^{2A-\Phi}v_1) = d(e^{-A}w) = 0, \qquad (4.9a)$$

$$d(e^{2A}) \wedge v_1 \wedge v_2 = d(e^{-A+\Phi}) \wedge w \wedge v_1 = 0, \qquad (4.9b) \qquad 35$$

$$e^{3A} \star_4 \lambda(G_-) - d(e^{3A - \Phi} w_1 \wedge w_2) = 0, \qquad (4.9c)$$

$$\left(v_1 \wedge v_2 \wedge \lambda(G_+) + v_1 \wedge \lambda(G_-) \right) \Big|_4 = 0.$$

$$(4.9e) \qquad (4.9e) \qquad 41$$

The first thing we note is that given (4.9c)-(4.9d), G_+ must be a 0-form and G_- a 1-form which means (4.9e) is solved automatically. Next, we solve (4.9a) by using it to define the vielbein

⁴⁵
₄₆
$$v_1 = e^{-2A + \Phi} g(x) dx, \quad w = e^A (d\psi_1 + id\psi_2), \quad v_2 = v e^{-3A/2 + \Phi/2} d\rho, \quad \rho = e^{\frac{3}{2}A + C - \frac{\Phi}{2}}, \quad \frac{^{45}}{^{46}}$$

₄₇ (4.10) 47

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.23 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •• where g is a function parametrising a potential coordinate transformation in x. From eq. (4.9c), we see that the combination $e^{A-\Phi}$ only depends on x and that e^A , e^{Φ} and e^C are functions of x, ρ only, so that ∂_{ψ_1} and ∂_{ψ_2} are isometries. We thus choose to parametrise $e^{A-\Phi} = f(x) , \quad g = -f ,$ (4.11)the latter of which is a convenient choice we make without loss of generality. For the fluxes, we use the v_1, v_2, w_1, w_2 vielbein on M_4 to compute the Hodge duals from eq. (4.9c) and (4.9d), arriving at the ten-dimensional fluxes $F_0 = \partial_x f , \quad F_4 = \nu \rho^3 \left(f \partial_\rho (f^{-1} e^{-4A}) dx - \partial_x (f^{-1} e^{-4A}) d\rho \right) \wedge \operatorname{Vol}(S^3) .$ (4.12)The Bianchi identities reduce to $dF_0 = dF_4 = 0$ which leads to $\partial_x^2 f = 0$, $\partial_x^2 (f^{-1} e^{-4A}) + f \frac{1}{\rho^3} \partial_\rho (\rho^3 \partial_\rho (f^{-1} e^{-4A})) = 0$, (4.13)the former of which can be immediately integrated as $f = (c + F_0 x)$, dc = 0. The metric takes the form $ds^{2} = \frac{1}{\sqrt{fH}} ds^{2}(\mathbb{R}^{1,4}) + \sqrt{\frac{H}{f}} \left(d\rho^{2} + \rho^{2} ds^{2}(S^{3}) \right) + \sqrt{fH} dx^{2}, \quad H = f^{-1} e^{-4A}.$ (4.14) This solution corresponds to an intersecting D4-D8 brane system, where the localised D4-branes are embedded in the D8-branes [40]. 4.1.2. Subcase: $\beta = \frac{\pi}{2}$ The $\beta = \frac{\pi}{2}$ limit of (4.8) leads to $H_3 \propto v_1 \wedge w_1 \wedge w_2$ with the remaining conditions implied by $d(e^{2A+2C-\Phi}u_i) + 2ve^{2A+C-\Phi}u_i \wedge v_2 = 0,$ (4.15a) $d(e^{3A+2C-\Phi}\epsilon_{ijk}u_j\wedge u_k)-2\nu e^{3A+C-\Phi}\epsilon_{ijk}u_j\wedge u_k\wedge \nu_2=0,$ (4.15b) $d(e^{3A-\Phi}) - e^{3A-\Phi}H_3 + e^{3A} \star_4 \lambda(G_-) = 0,$ (4.15c) $d(e^{3A+3C-\Phi}v_2) - e^{3A+3C-\Phi}H_3 \wedge v_2 + e^{3A+3C} \star_4 \lambda(G_+) = 0,$ (4.15d) $d(e^{-A-\Phi}) \wedge v_1 \wedge w_1 \wedge w_2 = v_1 \wedge w_1 \wedge w_2 \wedge \lambda(G_-) \Big|_{A} = 0,$ (4.15e)where we introduce (4.16) $u = (v_1, w_1, w_2),$ to ease notation, and to make clear the cyclic property of these vielbein. The first thing we note is that, given (4.15c), the second of (4.15c) reads $\star_4 H_3 \wedge v_1 \wedge w_1 \wedge w_2 = 0$, but since $H_3 \propto v_1 \wedge w_1 \wedge w_2$ we must set $H_3 = 0$. (4.17)Next one can show that both (4.15a) and (4.15b) together imply the useful identities $d(e^{A}u_{i}) \wedge u_{i} = \left[d(e^{A/2 + C - \Phi/2}) - ve^{A/2 - \Phi/2}v_{2} \right] \wedge u_{i} \wedge u_{i} = 0, \quad i \neq j,$ (4.18)

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

з

 ARTICLE IN PRESS

 JID:NUPHB AID:14363 /FLA
 [m1+; v1.285; Prn:6/06/2018; 11:31] P.24 (1-49)

 24
 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••

so we must have

$$d(e^{A}u_{i}) + \frac{c_{i}}{2}\epsilon_{ijk}u_{j} \wedge u_{k} = d(e^{A/2 + C - \Phi/2}) - \nu e^{A/2 - \Phi/2}v_{2} = dc_{i} = 0.$$
(4.19)

Consistency of the first of these with (4.15a) implies that $c_i = 0$, and with this fixed (4.15b) is also followed from (4.19). We can use the standard trick of taking (4.19) to define a vielbein in terms of local coordinates, namely

$$u_i = e^{-A} dx_i, \quad v_2 = v e^{-A/2 + \Phi/2} d\rho, \quad \rho = e^{A/2 + C - \Phi/2}.$$
(4.20) 8

Having defined the vielbein, it is then a simple matter to solve the first of (4.15e) by introducing a free function

$$e^{-A-\Phi} = f(x_1, x_2, x_3).$$
 (4.21) 12

All that remains is to calculate the fluxes, and impose their Bianchi identities. Using (4.20) to take the Hodge duals of 4.15c and 4.15d we find the 10d fluxes

$$F_6 = -\nu\rho^3 \left(\frac{1}{2}\epsilon_{ijk}\partial_{x_i}(f^{-1}e^{-4A})dx^j \wedge dx^k \wedge d\rho - f\partial_\rho(f^{-1}e^{-4A})dx_1 \wedge dx_2 \wedge dx_3\right)$$

$$\wedge$$
 Vol(S^3),

which clearly means the Bianchi identities, away from localised sources, follow from

$$\partial_{x_i}^2 f = 0 , \quad \partial_{x_i}^2 (f^{-1}e^{-4A}) + f \frac{1}{\rho^3} \partial_\rho (\rho^3 \partial_\rho (f^{-1}e^{-4A})) = 0 . \tag{4.23}$$

The metric takes the form

.

$$ds^{2} = \frac{1}{\sqrt{fH}} ds^{2}(\mathbb{R}^{1,2}) + \sqrt{\frac{H}{f}} \left(d\rho^{2} + \rho^{2} ds^{2}(S^{3}) \right) + \sqrt{fH} (dx_{1}^{2} + dx_{2}^{2} + dx_{3}^{3}), \quad (4.24)$$

$$H = f^{-1} e^{-4A}.$$

$$31$$

The solution corresponds to an intersecting D2–D6 brane system [44].

4.1.3. Subcase: generic β

For $0 < \beta < \frac{\pi}{2}$ one is able to divide by $\sin \beta$, $\cos \beta$ freely when simplifying (4.8). Assuming that $\cos \beta \neq 0$ the result is

$$d(e^{2A-\Phi}\cos\beta v_1) = d(e^{-A}\sec\beta w) = 0, \qquad (4.25a)$$

$$d(e^{\frac{3}{2}A+C-\frac{1}{2}\Phi}\sqrt{\cos\beta}) - \nu e^{\frac{3}{2}A-\frac{1}{2}\Phi}\sqrt{\cos\beta}v_2 = 0, \qquad (4.25b) \qquad 40$$

$$d(e^{-A+\Phi} \sec\beta) \wedge v_1 \wedge w = d(e^{2A}) \wedge v_1 \wedge v_2 = d(e^{-2A} \tan\beta) \wedge v_1 = 0, \qquad (4.25c)$$

$${}^{43}_{44} \qquad e^{2A}\cos^2\beta H_3 + d(e^{2A}\sin\beta\cos\beta) \wedge w_1 \wedge w_2 = 0, \qquad (4.25d) \qquad {}^{43}_{44}$$

45
$$d(e^{3A+3C-\Phi}\sin\beta v_2) + d(e^{3A+3C-\Phi}\cos\beta w_1 \wedge w_2 \wedge v_2)$$

46

$$-e^{3A+3C-\Phi}\sin\beta H_3 \wedge v_2 - e^{3A+3C} \star_4 \lambda(G_+) = 0,$$

JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.25 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) $d(e^{3A-\Phi}\sin\beta) + d(e^{3A-\Phi}\cos\beta w_1 \wedge w_2) - e^{3A-\Phi}\sin\beta H_3 - e^{3A} \star_4 \lambda(G_-) = 0,$ (4.25e) $\left(\cos\beta v_1 \wedge v_2 \wedge \lambda(G_+) + (\cos\beta - \sin\beta w_1 \wedge w_2) \wedge v_1 \wedge \lambda(G_-)\right)\Big|_{A_+} = 0,$ (4.25f) where H_3 is closed given (4.25a) and (4.25c). As usual we solve (4.25a) and (4.25b) by using them to define a vielbein in terms of local coordinates $v_1 = g(x)e^{-2A+\Phi} \sec\beta dx, \quad w = e^A \cos\beta (d\psi_1 + id\psi_2),$ $v_2 = v e^{-\frac{3}{2}A + \frac{1}{2}\Phi} \sqrt{\sec\beta} d\rho, \quad \rho = e^{\frac{3}{2}A + C - \frac{1}{2}\Phi} \sqrt{\cos\beta},$ (4.26)where $g(x_1)$ is a function parametrising a potential diffeomorphism in x_1 . With local coordinates introduced we can solve (4.25c) in terms of them as $e^{A-\Phi}\cos\beta = f(x)$, $A = A(\rho, x)$, $\tan\beta = c(x_1)e^{2A}$, g = -f, (4.27)so that ∂_{ψ_i} are necessarily isometry directions. We can then calculate the ten-dimensional fluxes as before - first we note that $F_0 = \partial_x f + \frac{\partial_x c f \tan^2 \beta}{2} ,$ (4.28)should be constant. We shall restrict ourselves to the case $\partial_x c = 0$. For generic β then the fluxes may be expressed as $F_0 = \partial_x f, \quad F_2 = F_0 B_2, \quad B_2 = -\frac{\sin^2 \beta}{c} d\psi_1 \wedge d\psi_2, \quad dc = 0$ (4.29) $F_4 = B_2 \wedge F_2 + \nu \rho^3 \left(f \partial_\rho (f^{-1} e^{-4A}) dx - \partial_x (f^{-1} e^{-4A}) d\rho \right) \wedge \operatorname{Vol}(S^3).$ We note that the Bianchi identities follow when anything not coupled to B_2 is closed, and since these terms reproduce (4.12) the Bianchi identities imply the PDEs of (4.13) once more. This is because the class of solutions in this section can be generated via U-duality, from the intersecting D4-D8 system in section 4.1.1. For completeness the metric takes the form $ds^{2} = \frac{1}{\sqrt{fH}} ds^{2}(\mathbb{R}^{1,2}) + \frac{\cos^{2}\beta}{\sqrt{fH}} ds^{2}(T^{2}) + \sqrt{\frac{H}{f}} \left(d\rho^{2} + \rho^{2} ds^{2}(S^{3}) \right) + \sqrt{fH} dx^{2},$ (4.30) $H = f^{-1}e^{-4A},$ where T^2 is spanned by (ψ_1, ψ_2)

43 4.2. Branch II: α non-zero

For the second branch with $0 < \alpha < \pi$, we begin by studying the lower form conditions that follow from (4.1). As in IIB we find it useful to rotate the canonical frame of (B.3) as (3.44). Necessary but insufficient conditions for supersymmetry are

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

	:31] P.26 (1-49)
26 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••	
$d(e^{2A+3C-\Phi}\cos\beta\cos(\alpha-\delta)) + e^{2A+3C-\Phi}H_0(\cos\beta\cos\delta v_2 + \sin\beta w_2) = 0,$	(4.31a)
$d(e^{2A-\Phi}(\cos\beta\cos\delta v_2+\sin\beta w_2))=0,$	(4.31b)
$d(e^{3A+2C-\Phi}\cos\alpha\sin\beta) + 2\nu e^{3A+C-\Phi}(\cos\beta\cos\delta w_2 - \sin\beta v_2) = 0,$	(4.31c)
$e^C \cos \alpha \sin \beta \sin \theta d\phi - 2\nu (\cos \beta \cos \delta w_1 - \sin \alpha \sin \beta v_1) = 0,$	(4.31d)
$e^C \cos \alpha \sin \beta d\theta - 2\nu(\cos \beta \cos \delta v_1 + \sin \alpha \sin \beta w_1) = 0,$	(4.31e)
$d(e^{2A+2C-\Phi}(\sin\alpha\sin\beta v_2 - \sin(\alpha - \delta)\cos\beta w_2))$	
$+ 2e^{2A+C-\Phi}v\cos\beta(\sin\delta w_2 \wedge v_2 + \cos(\alpha - \delta)v_1 \wedge w_1) = 0,$	(4.31f)
First from (4.31a) we observe that when $\cos(\alpha - \delta) = 0$ ($\cos \beta = 0$ is a subcase necessary have $H_0 = 0$. Next we observe that generically (4.31b)–(4.31e) can be use define the vielbein on M_4 , the only exception is when $\sin \beta = 0$ ($\cos \alpha = 0$ is a subcase Setting $\sin \beta = 0$ means that in order to solve (4.31d)–(4.31e) we must take $\delta =$ ally (θ, ϕ) drop out of the definition of the spinors so we can fix	of this) we ed to locally ase of this). $\frac{\pi}{2}$, addition-
$\beta = \theta = \phi = (\delta - \frac{\pi}{2}) = 0$	(4.32)
without loss of generality. Interestingly one doesn't need to set $H_0 = 0$, however as in section 4.2.1 this case actually contains no solutions. If one assumes $\sin \beta \neq 0$ we see that v_1 , w_1 must span S^2 while v_2 , w_2 can b in terms of local coordinates in such a way that they have no legs in S^2 . This is a (4.31f) which generically has an Vol(S^2) factor, due to the $v_1 \wedge w_1$ term which sits to everything else. Thus the only resolution is to fix $\cos(\alpha - \delta) = 0$ which leads to A	we shall see be expressed problem for s orthogonal $H_0 = 0$ also.
This actually leads to a novel class of solutions that we shall derive in section 4.2.2.	
 This actually leads to a novel class of solutions that we shall derive in section 4.2.2. 4.2.1. Subcase: β = 0 Upon setting β = 0 we are led to the following conditions for supersymmetry 	
This actually leads to a novel class of solutions that we shall derive in section 4.2.2. 4.2.1. Subcase: $\beta = 0$ Upon setting $\beta = 0$ we are led to the following conditions for supersymmetry $d(e^{2A+3C-\Phi}\sin\alpha) = d(e^{-A-\Phi}\cos^4\alpha) \wedge v_1 \wedge w_1 \wedge w_2 = 0$,	(4.33a)
This actually leads to a novel class of solutions that we shall derive in section 4.2.2. 4.2.1. Subcase: $\beta = 0$ Upon setting $\beta = 0$ we are led to the following conditions for supersymmetry $d(e^{2A+3C-\Phi}\sin\alpha) = d(e^{-A-\Phi}\cos^4\alpha) \wedge v_1 \wedge w_1 \wedge w_2 = 0$, $d(e^{2A+2C-\Phi}\cos\alpha u_i) + ve^{2A+C-\Phi}(\sin\alpha\epsilon_{ijk}u_j \wedge u_k + 2u_i \wedge v_2) = 0$,	(4.33a) (4.33b)
This actually leads to a novel class of solutions that we shall derive in section 4.2.2. 4.2.1. Subcase: $\beta = 0$ Upon setting $\beta = 0$ we are led to the following conditions for supersymmetry $d(e^{2A+3C-\Phi}\sin\alpha) = d(e^{-A-\Phi}\cos^4\alpha) \wedge v_1 \wedge w_1 \wedge w_2 = 0$, $d(e^{2A+2C-\Phi}\cos\alpha u_i) + ve^{2A+C-\Phi}(\sin\alpha\epsilon_{ijk}u_j \wedge u_k + 2u_i \wedge v_2) = 0$, $d(e^{3A+2C-\Phi}(\epsilon_{ijk}u_j \wedge u_k + 2\sin\alpha u_i \wedge v_2)) - 2ve^{3A+C-\Phi}\cos\alpha\epsilon_{ijk}u_j \wedge u_k \wedge v_k)$	(4.33a) (4.33b) $v_2 = 0$, (4.33c)
This actually leads to a novel class of solutions that we shall derive in section 4.2.2. 4.2.1. Subcase: $\beta = 0$ Upon setting $\beta = 0$ we are led to the following conditions for supersymmetry $d(e^{2A+3C-\Phi}\sin\alpha) = d(e^{-A-\Phi}\cos^4\alpha) \wedge v_1 \wedge w_1 \wedge w_2 = 0$, $d(e^{2A+2C-\Phi}\cos\alpha u_i) + ve^{2A+C-\Phi}(\sin\alpha\epsilon_{ijk}u_j \wedge u_k + 2u_i \wedge v_2) = 0$, $d(e^{3A+2C-\Phi}(\epsilon_{ijk}u_j \wedge u_k + 2\sin\alpha u_i \wedge v_2)) - 2ve^{3A+C-\Phi}\cos\alpha\epsilon_{ijk}u_j \wedge u_k \wedge \sin\alpha H_3 - \cos\alpha H_0v_1 \wedge w_1 \wedge w_2 = 0$,	$(4.33a) (4.33b) v_2 = 0, (4.33c) (4.33d)$

	N.T. Macpherson et al. / Nuclear Physics $B \leftrightarrow (\leftrightarrow \phi) \leftrightarrow \phi \to \phi$	27
e^{3A}	$^{+3C} \star_4 \lambda(G_+) + e^{3A+3C-\Phi} H_0 + d(e^{3A+3C-\Phi} \cos \alpha v_2)$	
— e	${}^{3A+3C-\Phi}H_0\csc\alpha v_1\wedge v_2\wedge w_1\wedge w_2=0,$	
	$\left. \left. \left. \cos \alpha v_1 \wedge w_1 \wedge w_2 \wedge \lambda(G) - (\sin \alpha - v_1 \wedge v_2 \wedge w_1 \wedge w_2) \wedge \lambda(G_+) \right) \right _4 = 0 \right.,$	(4.33f)
where her tion 3.1.2,	e as elsewhere $u = (v_1, w_1, w_2)$. Using the same techniques as are spelled on it is possible to establish that	ut in sec-
d(e)	$(\frac{1}{2}A+C-\frac{1}{2}\Phi) - \nu e^{\frac{1}{2}A-\frac{1}{2}\Phi}\cos\alpha v_2 = d(e^A u_i) \wedge u_j = 0, \text{ for } i \neq j$	(4.34)
which we $\rho = e^{\frac{1}{2}A+}$	can use as in section 3.2.1 to define the vielbein in terms of the local contract $C^{-\frac{1}{2}\Phi}$ and a set of left invariant 1-forms such that	oordinate
<i>v</i> ₂ =	$= v e^{-\frac{1}{2}A + \frac{1}{2}\Phi} d\rho, u_i = e^{-A} \cos \alpha c_i \tilde{K}_i, d\tilde{K}_i = \frac{1}{2} \tilde{K}_j \wedge \tilde{K}_k,$	(4.35)
under the	assumption that $\alpha \neq 0$. Plugging this back into (4.33b) fixes	
e^{2A}	$=\frac{c^4e^{-2\Phi}\sin^4\alpha}{\rho^4}, \sin\alpha=\frac{c_1}{\rho}, \ c_i=c$	(4.36)
however p	blugging this back into (4.33a) leads to	
d ho /	$\wedge \tilde{K}_1 \wedge \tilde{K}_2 \wedge \tilde{K}_3 = 0$	(4.37)
which can	not be solved.	
4.2.2. Sub The fin the rotation iw_1), then	bcase: $\delta = \alpha + \frac{\pi}{2}$ al case we consider is when $\delta = \alpha + \frac{\pi}{2}$ and contains $\beta = \frac{\pi}{2}$ as a subcase. In ac g the canonical frame (B.3) by (3.44) we find it useful to send $v_1 + iw_1 \rightarrow e$ a the necessary and sufficient conditions for supersymmetry are	Idition to $^{-i\beta}(v_1 +$
d(e)	$^{A-C}\sin\alpha) = H_0 = 0 , \qquad \qquad$	(4.38a)
$d(e^{2})$	$^{3A+2C-\Phi}\cos\alpha\sin\beta) - 2\nu e^{3A+C-\Phi}k_1 = d(e^{2A-\Phi}k_2) = 0$.,	(4.38b)
e^{C} c	$\cos\alpha\sin\beta d\theta + 2\nu\sin\alpha v_1 = e^C\cos\alpha\sin\beta\sin\theta d\phi + 2\nu\sin\alpha w_1 = 0,$	(4.38c)
H_3 :	$= \frac{1}{4} \left(d(e^{2C} \cot^2 \alpha \cos \beta \sin \beta) - 2e^C v \cos \alpha w_2 \right) \wedge \operatorname{Vol}(S^2),$	(4.38d)
$d\left(\cdot \right)$	$\frac{e^{-A+C}\sin\alpha}{\Delta}\right) \wedge k_1 - e^{2A+C-\Phi}\cos\alpha\sin\beta d\left(\frac{e^{-3A+\Phi}\cos\alpha\cos\beta}{\Delta}\right) \wedge k_2 = 0$, (4.38e)
$d\left(\left\{ {{}^{\prime }}\right\} ight)$	$\frac{e^{-\frac{3}{2}A+C+\frac{1}{2}\Phi}\cos^{\frac{3}{2}}\alpha\sqrt{\sin\beta}\cos\beta}{\Delta}\right)\wedge k_{1}$	
— e	$\frac{1}{2}A - \frac{1}{2}\Phi\sqrt{\cos\alpha\sin\beta}d\left(\frac{e^{-2A+C+\Phi}\cot\alpha\sin\beta}{\Delta}\right)\wedge k_2 = 0$,	

JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.28 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-••• $e^{3A} \star_4 \lambda(G_-) + d(e^{3A-\Phi}\cos\alpha\cos\beta) - d(e^{3A-\Phi}\cos\alpha\sin\beta v_1 \wedge w_1)$ $-e^{3A-\Phi}\cos\alpha\cos\beta H_3=0$, З $e^{3A+3C} \star_4 \lambda(G_+) + d(e^{3A+3C-\Phi}k_3) - d(e^{3A+3C-\Phi}v_1 \wedge w_1 \wedge k_1) + e^{3A+3C-\Phi}k_2 \wedge H_3,$ (4.38f) $\left(\cos\alpha\sin\beta(1+\cot\beta v_1\wedge w_1)\wedge v_2\wedge w_2\wedge\lambda(G_+)-(k_2-k_4\wedge v_1\wedge w_1)\wedge\lambda(G_-)\right)\Big|_4=0,$ g (4.38g) where we introduce $k_1 = (\sin \alpha \cos \beta w_2 + \sin \beta v_2), \quad k_2 = (\sin \alpha \cos \beta v_2 - \sin \beta w_2).$ $k_3 = (\cos \alpha v_2 - \sin \alpha \sin \beta w_2), \quad k_4 = (\cos \alpha w_2 + \sin \alpha \sin \beta w_1),$ $\Delta = \sin^2 \beta + \cos^2 \beta \sin^2 \alpha, \quad \operatorname{Vol}(S^2) = \sin \theta d\theta \wedge d\phi,$ (4.39) to ease presentation. We can use (4.38b)-(4.38d) to locally define the vielbein through $k_1 = \nu \sqrt{\cos \alpha \sin \beta} e^{-\frac{3}{2}A + \frac{1}{2}\Phi} d\rho, \quad k_2 = e^{-2A + \Phi} dx,$ (4.40) $v_1 = -\frac{1}{2\sin\alpha} v e^{-\frac{3}{2}A + \frac{1}{2}\Phi} \rho \sqrt{\cos\alpha \sin\beta} d\theta,$ $w_1 = -\frac{1}{2\sin\alpha} v e^{-\frac{3}{2}A + \frac{1}{2}\Phi} \rho \sqrt{\cos\alpha \sin\beta} \sin\theta d\phi,$ where (θ, ϕ, x) and $\rho = e^{\frac{3}{2}A + C - \frac{1}{2}\Phi} \sqrt{\cos \alpha \sin \beta}$ (4.41)are local coordinates on M_4 . The ten-dimensional metric then takes the form $ds^2 =$ (4.42) $e^{2A}ds^2(\mathbb{R}_{1,2}) + \frac{e^{-3A+\Phi}}{\cos\alpha\sin\beta}\rho^2 ds^2(S^3) + \frac{e^{-4A+2\Phi}dx^2}{\Lambda}$ $+\frac{e^{-3A+\Phi}\cos\alpha\sin\beta}{4\sin^2\alpha}\Big(\frac{4\sin^2\alpha d\rho^2}{\Delta}+\rho^2 ds^2(S^2)\Big).$ We can now turn our attention to (4.38a) and (4.38e) which lead to the PDEs $\partial_{\rho}(e^{A-C}\sin\alpha) = \partial_{x}(e^{A-C}\sin\alpha) = 0$, (4.43a) $\nu\rho\partial_{\rho}\left(\frac{e^{-3A+\Phi}\cos\alpha\cos\beta}{\Delta}\right) + \partial_{x}\left(\frac{e^{-A+C}\sin\alpha}{\Delta}\right) = 0,$ (4.43b) $\nu \partial_{\rho} \left(\frac{e^{-2A+C+\Phi} \cot \alpha \sin \beta}{\Delta} \right) + \partial_{x} \left(\frac{e^{-\frac{3}{2}A+C+\frac{1}{2}\Phi} \cos^{\frac{3}{2}} \alpha \sqrt{\sin \beta} \cos \beta}{\Delta} \right) = 0 ,$ (4.43c)and tell us that e^{2A} , e^{2C} , e^{Φ} , α , β are functions of (ρ, x) only, which means these solutions

support an additional SU(2) isometry due to round S^2 spanned by (θ, ϕ) . Actually this SU(2)Please cite this article in press as: N.T. Macpherson et al., Mink₃ × S^3 solutions of type II supergravity, Nucl. Phys. B

ARTICLE IN PRESS

JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.29 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-••• 29

is an additional part of an enhanced R-symmetry which together with the $SU(2)_R$ of S^3 gives SO(4)_R – there is also an SU(2) flavour symmetry. It is now a simple matter to calculate the Hodge dual of the fluxes from (4.38f) and then the fluxes themselves given (4.38d) and our local vielbein (4.40). At first G_{\pm} take a complicated form that we will not quote here, however, we are yet to impose also (4.38g): doing so and making extensive use of (4.43a)–(4.43c) one can express the ten-dimensional fluxes as:

$$H = \frac{1}{4} \left(d(e^{2C} \cot^2 \alpha \cos \beta \sin \beta) \right)$$

$$-\frac{2}{\Delta}(\rho e^{-3A+\Phi}\cos\alpha\cos\beta d\rho - \nu e^{-2A+C+\Phi}\cot\alpha\sin\beta dx))\wedge \operatorname{Vol}(S^2),$$

$$F_0 = 2 \frac{\Delta}{\sin \alpha} e^{-2\Phi} \partial_x (e^{2A} \sin \alpha) ,$$

$$F_2 = \frac{e^{-SA-\frac{1}{2}\Phi}\rho\sqrt{\cos\alpha}}{4\sin^2\alpha} \left(F_0 e^{\frac{3}{2}\Phi}\rho\sqrt{\cos\alpha}\cos\beta - 2\nu e^{\frac{3}{2}A}\sin\alpha\sqrt{\sin\beta}\right) \operatorname{Vol}(S^2) ,$$

$$F_4 = -\operatorname{Vol}_3 \wedge d(e^{3A-\Phi}\cos\alpha\sin\beta) + \frac{e^{-\frac{5}{2}A+\frac{1}{2}\Phi}\rho^2}{\sqrt{\cos\alpha\sin\beta}\sin\alpha} \left(d(e^{-2A}\cot\beta)\right)$$

$$-\frac{2e^{-\frac{5}{2}A}\sin\alpha}{\sin\beta\Delta}\left(\nu e^{\frac{1}{2}\Phi}\sqrt{\frac{\sin\beta}{\cos\alpha}}d\rho-e^{\frac{1}{2}A}\sin\alpha\cos\beta dx\right)\right)\wedge\operatorname{Vol}(S^3),$$

$$F_{6} = -\text{Vol}_{3} \wedge (d(e^{3A-\Phi}\cos\alpha\sin\beta v_{1}\wedge w_{1}) + e^{3A-\Phi}\cos\alpha\cos\beta H_{3})$$
(4.44) 25

$$+\frac{\nu e^{-10A+2\Phi}\rho^3}{4\sin^2\alpha}\bigg(d(e^{3A-\Phi}\cos\alpha\cos\beta)\wedge d\rho$$

$$-\frac{e^{-A+\Phi}}{\cos\alpha\sin\beta}d(e^{3A-\Phi}\cos\alpha\cos\beta)\wedge dx\right)\wedge\operatorname{Vol}(S^2)\wedge\operatorname{Vol}(S^3)$$

³² where (4.38g) can be expressed in terms of F_0 as

$$\rho\sqrt{\sin\beta}\partial_{\rho}(\rho e^{2A}\sin\alpha) + \frac{\nu}{2}F_0\frac{e^{\frac{1}{2}A + \frac{3}{2}\Phi}\sin^2\alpha\sqrt{\cos\alpha}\cos\beta}{\Delta} = 0.$$
(4.45)

Imposing that F_0 is constant together with (4.43a)–(4.43c) and (4.45) then implies dH = 0 and the Bianchi identities of the remaining fluxes. This system is quite complicated, however taking inspiration from section 4.1 of [36] (which re-derives [49]) we anticipate that the system can be further simplified if we treat the cases $F_0 = 0$ and $F_0 \neq 0$ separately. 4.2.3. Subcase $F_0 = 0$ If we set $F_0 = 0$ then (4.45) and (4.44) impose that $\rho \, e^{2A} \sin \alpha = L^2, \quad dL = 0$ (4.46)This leaves (4.43a)-(4.43c) to solve. We first integrate (4.43a) as $e^{A-C}\sin\alpha = c, \quad dc = 0,$ (4.47)

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

 ARTICLE IN PRESS

 JID:NUPHB AID:14363 /FLA
 [m1+; v1.285; Prn:6/06/2018; 11:31] P.30 (1-49)

 30
 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••

then use it to express (4.43b) as

$$c\,\partial_x\left(\frac{e^{-5A+\Phi}}{\cos\alpha\sin\beta\Delta}\right) + \nu\frac{1}{\rho}\partial_\rho\left(\frac{e^{-3A+\Phi}\cos\alpha\cos\beta}{\Delta}\right) = 0. \tag{4.48}$$

We note that this defines an integrability condition that we can solve by introducing an auxiliary function $h(\rho, x)$ such that

$$c\frac{e^{-5A+\Phi}}{\cos\alpha\sin\beta\Delta} = \frac{\nu}{\rho}\partial_{\rho}h, \quad \frac{e^{-3A+\Phi}\cos\alpha\cos\beta}{\Delta} = -\partial_{x}h. \tag{4.49}$$

Plugging these definitions into (4.43c) and making use of (4.46)–(4.47) we arrive at

$$c^{2}\partial_{x}^{2}h + \frac{1}{\rho}\partial_{\rho}(\rho\partial_{\rho}h) = 0.$$
 (4.50) ¹²
13

This is a 3d Laplacian expressed in axially symmetric cylindrical polar coordinates (up to rescaling x). Solution in this class are in one to one correspondence with solution to this Laplace equation. The physical data can be expressed in terms of h and the 2 constants (c, L), as

$$L^4 e^{-4A} = \rho^2 \sin^2 \alpha, \quad e^{-\Phi+5A} = \frac{\nu c \rho}{\cos \alpha \sin \beta \partial_\rho h \Delta}, \quad \Delta = 1 - \frac{c^3 \rho (\partial_\rho h)^2}{L^4 \partial_\rho h (1 - c \rho \partial_\rho h)},$$

$$\tan \alpha = \sqrt{\frac{L^4 (1 - c\rho \partial_\rho h)}{c^4 \rho^2 (\partial_x h)^2 + L^4 (1 - c\rho \partial_\rho h)^2} - 1},$$
21
22
23

$$\tan \beta = \frac{\nu \sqrt{1 - c\rho \partial_{\rho} h} \sqrt{L^4 \partial_{\rho} h (1 - c\rho \partial_{\rho} h) - c^3 \rho (\partial_x h)^2)}}{c^{\frac{3}{2}} \sqrt{\rho} \partial_x h}.$$
(4.51)
(5)

It is interesting that this class depends on axially symmetric Laplacian, indeed the same is true of the class of $AdS_5 \times S^2$ solutions in IIA [50] one obtains by dimensionally reducing the M-theory class of Lin-Lunin-Maldecena [51]. The M-theory class actually depend on a 3d Toda equation, which is equivalent to the Laplacian only when one imposes an additional U(1) isometry, which one then uses to reduce to IIA. As the class of this section is in massless IIA it can be lifted to M-theory, so an obvious question poses itself: Is there a class in M-theory governed by a 3d Toda from which the backgrounds in this section descend? It would be interesting to look into this and what connection, if any, this class has to $AdS_5 \times S^2$ or indeed any AdS class.

37 4.2.4. Subcase: $F_0 \neq 0$

We expect to be able to perform a similar simplification of the system of PDE's for $F_0 \neq 0$ case, however up to this point we have failed to do so in general. However there is a special case which is far more simple, namely $\beta = \frac{\pi}{2}$. Here (4.43a)–(4.43c) and (4.45) force

$$e^{2A} = \frac{\rho}{g(x)^{\frac{1}{4}} \sin \alpha}, \quad e^{\Phi - 5A} = \frac{\cos \alpha \sin^2 \alpha}{c^2 \rho^2}, \quad d\alpha = dc = 0, \tag{4.52}$$

all that remains is to ensure that $dF_0 = 0$ which is ensured as long as

$$g = \tilde{c} - \frac{2F_0 x \cos \alpha}{c_1^4}.$$
 (4.53) (4.53)

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.31 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••)

This solution has all the generic fluxes except the internal part of F_6 turned on, it bears some resemblance to D8-branes on some sort of cone, but the precise picture depends on what values the free constant α takes.

We shall come back to study the solutions that follow from these massless and massive systems in [71].

5. The unique type II $AdS_4 \times S^3$ background

We have classified all Mink₃ \times S³ with internal Killing spinors of equal norm, up to certain PDE determining various warp factors. As equal norms is a requirement⁷ for the existence of AdS₄ it is reasonable to ask whether such solutions are contained within our classification. Any AdS₄ solution can be expressed as a Mink₃ solution, one needs only parametrise AdS as the Poincaré patch. This comes about quite naturally in terms of the Mink₃ \times M₇ set up by imposing that

$$ds^{2} = e^{2A} ds^{2}(\mathbb{R}^{1,2}) + ds^{2}(M_{7}) = e^{2\tilde{A}} \left(r^{2} ds^{2}(\mathbb{R}^{1,2}) + \frac{dr^{2}}{r^{2}} \right) + ds^{2}(M_{6}) , \qquad (5.2) \qquad {}^{16}_{17}$$

with $e^{2\bar{A}}$ and M_6 independent of r. As we have local expressions on $M_7 = S^3 \times M_4$, this makes our task relatively easy. A quick scan through the various cases in section 3 and 4 indicates that the only class that are potentially compatible with AdS_4 are in sections 4.1.2 and 4.2.2 - the others manifestly cannot be put in the form (5.2). These are both in IIA, but closer inspection leads one to realise that the class of section 4.2.2 cannot work as (4.38a) would break the putative AdS isometry. This leaves only sections 4.1.2.

We will now show that there is a unique compact⁸ AdS₄ solution, at least locally, for the class of solutions of section 4.1.2. This background corresponds to a foliation of $AdS_4 \times S^3 \times S^2$ over an interval and is the near-horizon of a D2-D6 brane intersection, and can also be obtained by dimensionally reducing a certain \mathbb{Z}_k orbifold of $AdS_4 \times S^7$. Starting from M-theory one first parameterises S^7 as a foliation of $S^3 \times S^3$ over a closed interval, then performs both the orbifolding and reduction to IIA on the Hopf fibre of one of the S^{3} 's (see e.g. [52]) – there by preserving 16 supercharges. For this purpose, we take the metric (4.24), expressed in the form

$$ds^{2} = e^{2A} ds^{2}(\mathbb{R}^{1,2}) + e^{-A+\Phi} \left(d\rho^{2} + \rho^{2} ds^{2}(S^{3}) \right) + e^{-2A} (dx_{1}^{2} + dx_{2}^{2} + dx_{3}^{3}),$$

and assume an AdS₄ factor, which requires $e^{2A} = r^2 e^{2\tilde{A}}$, with the rescaled warp factor \tilde{A} as well as the dilaton Φ undetermined functions independent of the AdS₄ radial coordinate r. Further-

⁷ It is established in Appendix D that the 7d spinors χ_1 , χ_2 must obey the relation $|\chi_1|^2 \pm |\chi_2|^2 = c_{\pm}e^{\pm A}$ where c_{\pm} are constants. We can, without loss of generality solve these conditions in terms of unit norm spinors χ_i^0 and an angle ζ

$$\chi_1 = \frac{1}{\sqrt{2}} e^{\frac{A}{2}} \sqrt{1 + \sin\zeta} \chi_1^0, \quad \chi_2 = \frac{1}{\sqrt{2}} e^{\frac{A}{2}} \sqrt{1 - \sin\zeta} \chi_2^0, \quad c_+ = 1, \quad c_- = e^{2A} \sin\zeta$$
(5.1)

To make a Mink₃ solution AdS₄ requires us to fix the dependence of e^{2A} on the AdS radius, but since $e^{2A} \sin \zeta$ is constant, we must either set $\zeta = c_{-} = 0$ or fix ζ such that it also depends on the AdS radius. The latter contradicts the assumption of an SO(2, 3) isometry, so we conclude that AdS_4 requires $c_- = 0$; i.e. equal 7d spinor norms.

⁸ Strictly speaking section 3.1.1 contains $AdS_5 \times S^5$ which can be expressed as a non-compact AdS_4 solution. We will disregard such higher dimensional AdS solutions.

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.32 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) ••• more, the background is only SO(2,3) invariant if the internal metric is independent of r, which fixes ρ and the x_i to scale as $\rho \sim r^{1/2}$ and $x_i \sim r$. Keeping this in mind, we parametrise $x_1 = r q(\mu) \sin \theta \cos \phi,$ $x_2 = r q(\mu) \sin \theta \sin \phi,$ $x_3 = r q(\mu) \cos \theta \,,$ $\rho = r^{1/2} h(\mu) \,,$ g where $q(\mu)$ and $h(\mu)$ are undetermined functions of some coordinate μ , and the (θ, ϕ) directions parametrise a 2-sphere such that the \mathbb{R}^3 spanned by x_i is written in polar coordinates with radius $r q(\mu)$. Now we have to ensure that the metric is diagonal with respect to the r-direction, i.e. set $g_{r\mu} = 0$, and that it shows the $1/r^2$ behaviour for g_{rr} , which amounts to imposing $g_{rr} = e^{2\tilde{A}}/r^2$. These two conditions lead to the following expressions for \tilde{A} and Φ in terms of $q(\mu), h(\mu)$ and independent of (θ, ϕ) : $e^{4\tilde{A}} = q(\mu) \left(q(\mu) - h(\mu) \frac{q'(\mu)}{2h'(\mu)} \right),$ $e^{\Phi} = -2e^{-\tilde{A}} \frac{q(\mu) q'(\mu)}{h(\mu) h'(\mu)}.$ These expressions imply, once inserted in the first eq. of (4.15e), the following ODE for the q and h functions: $q'(\mu) \left[h'(\mu)^2 + h(\mu) h''(\mu) \right] = h(\mu) h'(\mu) q''(\mu),$ which can be solved in closed form as $h = h(q(\mu))$ and also implies the Bianchi identities of the fluxes. As h is a function of q, rather than μ , we can use diffeomorphism invariance to fix q such that *h* is simple, without loss of generality we choose $q(\mu) = \frac{2L^3}{k} \cos^2\left(\frac{\mu}{2}\right),$ where L and k are constants. This leads to $h(\mu) = -2L^{3/2}\sin\left(\frac{\mu}{2}\right).$ The resulting metric is of the form $ds^{2} = \frac{2L}{k}\cos\left(\frac{\mu}{2}\right) \left[ds^{2}(\text{AdS}_{4}) + L^{2}\left(d\mu^{2} + 4\sin^{2}\left(\frac{\mu}{2}\right)ds^{2}(S^{3}) + \cos^{2}\left(\frac{\mu}{2}\right)ds^{2}(S^{2})\right) \right]$ (5.3)with fluxes $F_2 = -\frac{k}{2} \text{Vol}(S^2)$, $F_4 = \frac{3}{L} \text{Vol}(\text{AdS}_4)$. (5.4)This is the IIA reduction of $AdS_4 \times S^7/\mathbb{Z}_k$ with length scale L and k D6-branes, as in eq. (2.8) of [52]. The fact that (θ, ϕ) are isometry directions of this solution means that there is an additional

⁴⁶ $SU(2)_{S^2}$ symmetry due to the round S^2 factor in the metric and fluxes. The spinors of this ⁴⁷ solution are then charged under $SU(2)_{S^2}$ and just one of the SU(2)'s of S^3 (see the ν dependence ⁴⁷

JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.33 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••

of (4.22)), $SU(2)_+$ say. Since S^2 and S^3 appear as a product the spinors are actually charged under $SU(2)_+ \times SU(2)_{S^2}$ which realises an enhanced SO(4) R-symmetry as required by the $\mathcal{N} = 4$ super-conformal algebra in 3d – $SU(2)_{-}$, under which the spinors are not charged, is a flavour symmetry. 6. Type II with a single Killing spinor In the previous two sections, we have worked out the supersymmetry conditions making use of the pure spinor equations (2.4), which are valid only in case $|\chi_1|^2 = |\chi_2|^2$. Note that this is a necessary condition for the existence of D-branes which do not break background super-symmetry. The supersymmetry condition for a D_p -brane is given by $\Gamma^{(p)}\epsilon_1 = \epsilon_2$. Since $\Gamma^{(p)}$ is unitary, squaring this equation leads to the conclusion that left- and right-handside must have equal norm.9 We will examine the simplest non-equal norm case, namely the one where $\epsilon_2 = 0$. (6.1)We could either make use of the generalised geometrical reformulation of supersymmetry which incorporates $|\chi_1|^2 - |\chi_2|^2 \neq 0$ as deduced in appendix D, or use the actual Killing spinor equa-tions. Considering the simplicity of this case, we will use the latter. Much of the work has however already been done: the conditions for seven-dimensional pure

NSNS solutions have been deduced in [29,56,57] up to some ansätze. We will merely show that the ansätze made in [29,56,57] (no warp factor, no external NSNS flux) are in fact enforced by supersymmetry, and then proceed to plug in the decomposition resulting from $M_7 = S^3 \times M_4$. This leads to a pair of explicit pure NS backgrounds: the NS5-brane and the U-dual to the IIB conical backgrounds of section 3.2.1. We will also analyse the seven-dimensional RR-sector, which is new, but the conclusion is that all RR-fluxes vanish.

6.1. Seven-dimensional decomposition

Our starting point are the democratic supersymmetry equations, which read as follows for $\epsilon_2 = 0$:

$$\left(\partial \!\!\!/ \phi - \frac{1}{2} \not\!\!\!/ H\right) \epsilon_1 = \left(\nabla_M - \frac{1}{4} \not\!\!\!/ M_M\right) \epsilon_1 = 0, \quad \lambda \not\!\!\!/ \Gamma_M \epsilon_1 = 0. \tag{6.2}$$

As can be seen, the NSNS and RR sectors decouple. We impose a similar 3 + 7 decomposition as before: the metric and RR flux is given by (2.1), while the Killing spinor ϵ_1 is given by (2.2) and $\epsilon_2 = 0$. We generalize the NSNS 3-form flux by allowing a term $h e^{3A} \text{Vol}_3$.

Using the convention $\gamma_{\mu\nu\rho} = \epsilon_{\mu\nu\rho}$, plugging the above decompositions into (6.2) leads to the following 7d equations:

$$\left(\partial_{m}\phi\gamma^{m} - \frac{1}{12}H_{mnp}\gamma^{mnp} + \frac{1}{2}ih\right)\chi = \left(\nabla_{m}^{(7)} - \frac{1}{8}H_{mnp}\gamma^{np}\right)\chi = 0$$
(6.3a)
(6.3a)
(6.3a)

⁹ The argument is slightly more complicated due to the fact that Spin(1,9) spinors do not admit a non-trivial norm, and hence one should decompose to Spin(9) first. See [53] for details. Also note that strictly speaking, the norms need only be equivalent on the brane.

CLE IN PRESS JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.34 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-••• $\lambda f \chi = \lambda f \gamma_m \chi = 0 .$ (6.3c) From (6.3b) it follows that (6.4) $\partial_m A = h = 0$. As can be seen, the NSNS and RR sector split and can thus be analysed independently. The existence of a globally defined nowhere-vanishing Spin(7) Majorana spinor χ reduces the structure group of M_7 to G_2 . More concretely, the following bilinears can be defined: $ix^{\dagger}x + x + (+-\infty) = x^{\dagger}x + x$ (6.5) where we have normalized χ . The other bilinears, i.e., the 1-, 2-, 5- and 6-form vanish. As has

$$\varphi_{mnp} = -\iota \chi \gamma_{mnp} \chi , \quad (\star_7 \varphi)_{mnpq} = \chi \gamma_{mnpq} \chi , \tag{6.3}$$

been deduced in [29,56,57], (6.3a) can be rewritten in terms of the G_2 -structure as

$$d\varphi \wedge \varphi = d(e^{-2\Phi} \star_7 \varphi) = d(e^{-2\Phi}\varphi) - e^{-2\Phi} \star_7 H = 0.$$
(6.6)

We will analyse the solution to the NSNS sector by requiring a further splitting of $M_7 = S^3 \times M_4$. On the other hand, we will show that the RR-fluxes vanish for any M_7 .

6.2. NSNS sector

Considering the case $M_7 = S^3 \times M_4$, we further decompose the spinor as

$$\chi = \xi \otimes \left(\sin(\alpha/2)\eta + \cos(\alpha/2)\hat{\gamma}\eta \right) + \text{m.c.}, \quad \eta^{\dagger}\eta = 1, \quad \eta^{\dagger}\hat{\gamma}\eta = 0 , \quad (6.7)$$

with m.c. the Majorana conjugate. This leads to a further reduction of the structure group. Since S^3 is parallelisable, it has trivial structure group, leading to a Spin(4) structure group on M_4 . Generically, the structure group need not reduce on M_4 .¹⁰ In the case where either η_+ or η_- is nowhere vanishing, the structure group reduces to SU(2), in case both are nowhere-vanishing, the structure group is trivial. As everywhere else, our analysis is purely local and we will work with a local trivial structure, parametrising possible vanishing of either chiral spinor by the an-gle α .

First, as in [58], we make use of an auxiliary SU(3)-structure (J, Ω) to express the G_2 -structure as

$$\varphi = -v_2 \wedge J - \operatorname{Im}\Omega, \quad \star_7 \varphi = \frac{1}{2}J \wedge J + \operatorname{Re}\Omega \wedge v_2.$$
 (6.8) ³⁵
₃₆

Next, we decompose the SU(3)-structure in terms of the vielbeine as

$$J = -\frac{1}{2}(K_1 \wedge w_1 + K_2 \wedge w_2 + K_3 \wedge v_1), \quad \Omega = e^{i\alpha}(\frac{K_1}{2} + iw_1) \wedge (\frac{K_2}{2} + iw_2) \wedge (\frac{K_3}{2} + iv_1)$$
(6.9)
(6.9)

Inserting this into (6.6), one finds

$$10$$
 Note that on any M_{π} with a coin structure on $SU(2)$ structure can be found [54] [55]. However, this is not necessarily

the structure group defined by the spinors we are making use of, and so even when splitting $M_7 = M_3 \times M_4$ with M_3

parallelisable, M_4 need not admit a globally well-defined SU(2)-structure; consider for example $M_7 = S^3 \times S^4$.

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

N.T. Macpherson et al. / Nuclear Physics $B \leftrightarrow (\bullet \bullet \bullet \bullet) \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet$	35
$d(e^{3C-2\Phi}\cos\alpha v_2) = d(e^{2C-2\Phi}u_i) + ve^{C-2\Phi}(2u_i \wedge v_2 + \sin\alpha\epsilon_{ijk}u_j \wedge u_k) = 0,$	(6.10a)
$d(e^{2C-2\Phi}(\sin\alpha u_i\wedge v_2+\frac{1}{2}\epsilon_{ijk}u_j\wedge u_k))-\nu\epsilon_{ijk}e^{C-2\Phi}\cos\alpha u_j\wedge u_k\wedge v_2=0,$	(6.10b)
$d\alpha \wedge v_1 \wedge w_1 \wedge w_2 = 0, u = (v_1, w_1, w_2),$	(6.10c)
$e^{-2\Phi} \star_7 H = -d(e^{-2\Phi}\cos\alpha v_1 \wedge w_1 \wedge w_2) - \nu \operatorname{Vol}(S^3) \wedge d(e^{3C-2\Phi}\sin\alpha).$	(6.10d)
When $\alpha \neq \frac{\pi}{2}$, by taking linear combinations, exterior derivatives and wedge products vielbein of the equations in (6.6), one can derive	s with the
$\epsilon_{ijk} \left[d(e^{C-\Phi} \cos \alpha) - v e^{-\Phi} v_2 \right] \wedge u_j \wedge u_k + e^{5C+12\Delta} \cos^4 \alpha \left[d(e^{-3C+2\Phi} \sec \alpha \tan \alpha) \wedge v_2 \right] \wedge u_i = 0,$	
where, since u_i form a basis of independent 1-forms, the terms in square parentheses mu This is sufficient to conclude that	st vanish.
$d\alpha = 0$, $C = C(\rho)$, $\Phi = \Phi(\rho)$, $\rho = e^{C-\Phi}$.	(6.11)
It is then not hard to establish that	
$du_i \wedge u_j = 0, i \neq j,$	(6.12)
in a similar fashion. This means we can locally parametrise	
$v_2 = v \sec \alpha e^{\Phi} d\rho$, $du_i = c_i \epsilon_{ijk} u_j \wedge u_k$, $dc_i = 0$.	(6.13)
Plugging this back into (6.6) we find that	
$c_i = e^{-C} v \tan \alpha,$	(6.14)
so either $dC = 0$ or $c = \alpha = 0$.	
Case 1 : When $\alpha = 0$, u_i are the vielbeine of T^3 so we can simply take	
$u_i = dx_i$.	(6.15)
All that is left to do is calculate H and impose its Bianchi identity. We find	
$H = \nu \partial_{\rho} (e^{2\Phi}) \rho^{3} \operatorname{Vol}(S^{3}) .$	(6.16)
Closure of the flux then implies that $e^{2\Phi}$ is harmonic, leading to	
$e^{2\Phi}=g_s^2\left(1+rac{c}{ ho^2} ight),\qquad g_s,c\in\mathbb{R},$	(6.17)
which is consistent with the definition of ρ . Finally, we note that the metric is given by	•
$ds^{2} = ds^{2}(\mathbb{R}^{1,5}) + e^{2\Phi} \left(d\rho^{2} + \rho^{2} ds^{2}(S^{3}) \right) .$	(6.18)
This is an NS5-brane, dual to the D5-brane solution in section 3.1.2 [41].	
Case 2 : When $0 < \alpha < \frac{\pi}{2}$, u_i span the vielbeine of another S^3 so we take	
$u = e^{C} \tilde{K}_{i} = d\tilde{K}_{i} + \tilde{v}_{civ} d\tilde{K}_{i} \wedge d\tilde{K}_{i} = d\tilde{C} = 0$	(6.10)

31D:NOPHB AID:14363 /FLA [m1+; V1.285; Pfn:6/06/2018; 11:31] P.36 (1-4) 36 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••	49)
where consistency requires that	
$\tilde{\nu}e^C\cos\alpha + e^{\tilde{C}}\nu\sin\alpha = 0, \qquad (6.2)$	20)
and find	
$H = -2e^{3C}\partial_{\rho}(e^{-\Phi})\cos^{2}\alpha \operatorname{Vol}(S^{3}) - 2e^{3\tilde{C}}\partial_{\rho}(e^{-\Phi})\cos\alpha\sin\alpha \operatorname{Vol}(\tilde{S}^{3}). $ (6.2)	21)
By definition of ρ , the Bianchi identity is satisfied. After redefining $r = \exp(e^{-C} \cos \alpha \rho)$, follows that the metric is given by	it
$ds^{2} = ds^{2}(\mathbb{R}^{1,2}) + dr^{2} + e^{2C}ds^{2}(S^{3}) + \cot^{2}\alpha e^{2C}ds^{2}(\tilde{S}^{3}). $ (6.2)	2)
Note that in IIB, this solution can be obtained from the solution of section 3.2.1 by means of the following S-duality transformation (up to redefining some constants):	he
$\Phi \to -\Phi$, $ds^2 \to e^{-\Phi} ds^2$, $F_3 \to -H$. (6.2)	:3)
6.3. RR-sector	
The NSNS sector has been analysed by imposing a further decomposition $Spin(7)$ - $Spin(3) \times Spin(4)$ on the spinor. On the other hand, we will analyse the RR-sector in full generality	→ n-
Let us consider the RR-flux constraints equations (6.3c), repeated here for convenience:	
$\lambda f \chi = \lambda f \gamma_m \chi = 0 . \tag{6.2}$:4)
For type IIA, we have that	
$\lambda f = f_0 + -f_2 - if_3 - if_1 , \qquad (6.2)$:5)
where we have defined $f_3 = \star_7 f_4$, $f_1 = \star_7 f_6$. For type IIB, one finds	
$\lambda f = f_1 - f_3 - if_2 - if_0 , \qquad (6.2)$:6)
with $f_2 = \star_7 f_5$, $f_0 = \star_7 f_7$, hence up to some field redefinitions, the supersymmetry constrain are identical. The fluxes, a priori irreducible representations of $SO(7)$, decompose into representations of G_2 as follows: $7 \rightarrow 7$, $21 \rightarrow 7 + 14$, $35 \rightarrow 1 + 7 + 27$. Concretely, we parametrise	ıts n-
$ \begin{aligned} & f_{2 mn} = \varphi_{mnp} f_2^2 + f_{2 mn} \\ & f_{3 mnp} = f_3 \varphi_{mnp} + \psi_{mnpq} f_3^q + f_{3 q[m} \varphi_{np]}^{\ \ q} , \end{aligned} $ (6.2)	:7)
where the 14 satisfies $f_{2 mn} = \frac{1}{2}\psi_{mnpq}f_2^{pq}$ and the 27 is corresponded to a symmetric tracele 2-tensor. Furthermore, we have introduced the notation $\psi = \star_7 \varphi$ for convenience. Making use the G_2 -structure identities (A.5), (A.6), we find	ess of
$f_{1}\chi = f_{1 m}\gamma^{m}\chi \qquad f_{1}\gamma_{m}\chi = f_{1 m}\chi - i\varphi_{mnp}f_{1}^{n}\gamma^{p}\chi f_{2}\chi = 3if_{2 m}\gamma^{m}\chi \qquad f_{2}\gamma_{m}\chi = 3if_{2 m}\chi - \varphi_{mnp}f_{2}^{n}\gamma^{p}\chi - 2f_{2 mn}\gamma^{n}\chi $ (6.2)	:8)
$f_3\chi = 7if_3\chi - 4f_{3 m}\gamma^m\chi$ $f_3\gamma_m\chi = -if_3\gamma_m\chi + 4f_{3 m}\chi + 6if_{3 mn}\gamma^n\chi$. Inserting the above into (6.24) and comparing representation by representation, it follows that a RR fluxes vanish.	all

	ARTICLE IN PRESS	07 (1 40)
JIL	$\frac{1131}{14303}$	37 (1-49) 37
		51
A	cknowledgements	
	We would like to thank Alessandro Tomasiello for collaboration in the early stages	of this
pr	oject. D.P. would like to thank Noppadol Mekareeya for discussion. J.M. is gratefu	l to the
Pł	sysics Department of Milano-Bicocca U., the ITP at KU Leuven and SISSA for th	e warm
hc	spitality. Throughout this project N.M. has been variously funded by INFN; the Europ	ean Re-
se	arch Council under the European Union's Seventh Framework Program (FP/2007-2013	3); ERC
G	ant Agreement n. 307286 (XD-STRING); and the Italian Ministry of Education, Univ	versities
an	d Research under the Prin project "Non-Perturbative Aspects of Gauge Theories and S	Strings"
(2	015MP2CX4). J.M. is supported by the FPI grant BES-2013-064815 of the Spanish MI	NECO,
an	d the travel grant EEBB-1-17-12390 of the same institution. He also acknowledges	partial
su Ec	pport through the Spanish and Regional Government Research Grants FPA2015-0300	/-P and NC
г	-13-OKOPIN-14-108. D.P. IS funded by EKC Ofant Agreement II. 30/280 (AD-51 KI	NG).
	mandin A. Commentions & identities	
A	openaix A. Conventions & identities	
	W 1	
	we decompose ten-dimensional gamma matrices as	
	$\Gamma = \sigma \otimes e^A u \otimes \Gamma = \sigma \otimes \Gamma \otimes u$	(A 1)
	$\Gamma_{\mu} = O_3 \otimes e \gamma_{\mu} \otimes \mathbb{I} , \Gamma_m = O_1 \otimes \mathbb{I} \otimes \gamma_m ,$	(A.1)
an	d seven-dimensional gamma matrices as	
	$\gamma_{\alpha}^{(1)} = e^{\mathbb{C}} \sigma_{\alpha} \otimes \hat{\gamma} , \gamma_{a}^{(1)} = \mathbb{I} \otimes \gamma_{a}, B_{7} = \sigma_{2} \otimes B_{4}, B_{4} B_{4}^{*} = -\mathbb{I} .$	(A.2)
Α.	1. M ₇	
	We consider gamma matrices satisfying	
	$(1)\frac{1}{2}n(n-1)$	$(\Lambda 2)$
	$\gamma_{mnpqrst} = \iota \epsilon_{mnpqrst}$, $\gamma_{(7-n)} = (-1)^{2\beta_{1}(n-1)} \iota \star_{7} \gamma_{(n)}$.	(A.3)
A	nowhere-vanishing $Spin(7)$ spinor defines a G_2 -structure on M_7 by means of the bilin	near
	$\varphi_{mnp} = -i \chi^{\dagger} \gamma_{mnp} \chi \; .$	(A.4)
Б		
D	elimiting $\psi = \star_7 \phi = \chi^+ \gamma_{mnpq} \chi$, the G ₂ -structure satisfies the following identities [59]:	
	$r_{s} = 2r_{s} = 2r_{s} = 1.48 + 8.48 + 8.58$	
	$\psi_{mnrs}\psi_{pq} = -2\psi_{mnpq} + 4\phi_{mp}\phi_{nq} - 4\phi_{mq}\phi_{np} ,$	· · - ·
	$\psi_{mnrs}\varphi_p^{\ \ s}=-4\varphi_{mnp}\;,$	(A.5)
	$\varphi_{mrs}\varphi^{nrs}=6\delta_m^n$,	
26	well as	
as	wen as	
	$\gamma_{mn}\chi = i\varphi_{mnp}\gamma^p\chi ,$	
	$\gamma_{mnn} \chi \equiv i \varphi_{mnn} \chi - \psi_{mnnn} \chi^q \chi$	(A.6)

ARTICLE IN PRESS	018: 11:31] P.38 (1-49)
38 N.T. Macpherson et al. / Nuclear Physics B ••• (••••)	010, 11.01,1.00(1.10)
A.2. S^3	
We consider Pauli-matrices playing the role of gamma-matrices. They satis	fy
$\sigma_{\alpha\beta\gamma} = i\epsilon_{\alpha\beta\gamma} , \sigma_{(3-n)} = (-1)^{\frac{1}{2}n(n-1)} \star_3 \sigma_{(n)} ,$	(A.7)
with α , β , $\gamma = 1, 2, 3$ indices on S^3 . The non-vanishing spinor bilinears of S^3	are given by
$\xi^{c\dagger}\sigma^{\alpha}\xi = \frac{1}{2}\left(K_1^{\alpha} + iK_2^{\alpha}\right), \xi^{\dagger}\sigma^{\alpha}\xi = \frac{1}{2}K_3^{\alpha}.$	(A.8)
The real 1-forms K_i , $i = 1, 2, 3$, define a trivial structure on S^3 (i.e., a vielbe sation). Note that S^3 is parallelisable and hence the trivial structure is globally will always normalise the volume form as	in, up to normali- well-defined. We
$K_1 \wedge K_2 \wedge K_3 = -8 \operatorname{Vol}(S^3) ,$	(A.9)
regardless of which specific vielbein is used.	
Appendix B. The bispinors of M_4	
We consider gamma matrices satisfying	
$\gamma_{abcd} = \epsilon_{abcd}$, $\gamma_{(4-n)} = (-1)^{\frac{1}{2}n(n+1)} \star_4 \gamma_{(n)} \hat{\gamma}$,	(B.1)
with $\hat{\gamma} = \gamma_{1234}$ the chirality matrix. Given a globally well-defined nowhere spinor η_+ , one can construct the bilinears	e vanishing chiral
$J_{ab} = i \eta^{\dagger}_+ \gamma_{ab} \eta_+ \ , \omega_{ab} = i \eta^{c\dagger}_+ \gamma_{ab} \eta_+$	(B.2)
which furnish an $SU(2)$ -structure. Given two globally well-defined nowhere spinors of opposite chirality, the structure group reduces to a trivial structure supersymmetry requires a nowhere vanishing spinor η , which can admit a ensures that, although the structure group of M_4 cannot be globally reduced, i duce the structure group of the generalised cotangent bundle $TM_4 \oplus T^*M_4$ to completely analogously to the well-known situation of $SU(3)$ -structures [30] symmetry constraints are local, we will always work with the vielbeine deter trivial structure. Using the conventions of [61] with $\eta = (\eta_+, \eta)$, we set	e vanishing chiral [60]. Generically, chiral locus. This t is possible to re- $SU(2) \times SU(2)$, l. Since the super- ermining the local
$v = v_1 + iv_2 = \eta^{\dagger} \gamma_a \eta_+ dx^a, w = w_1 + iw_2 = \eta^{c\dagger} \gamma_a \eta_+ dx^a.$	(B.3)
Although some care must be taken on the chiral locus, where the above 1-for turns out that no solutions exist on the chiral locus, as discussed in sections expand the locally defined 4d components of the Killing spinors $\eta_{1,2}$ in terms	forms all vanish, it is 3 and 4. We can of η as
$\eta^{\dagger}\eta = 1$, $\eta^{\dagger}\hat{\gamma}\eta = 0$	(B.4)
$\eta_1 = \cos\left(\frac{\alpha}{2}\right)\eta + \sin\left(\frac{\alpha}{2}\right)\hat{\gamma}\eta , \eta_2 = a\eta + b\hat{\gamma}\eta + c\eta^c + d\hat{\gamma}\eta^c$	(B.5)
where a, b, c and d are subject to	
$ a ^{2} + b ^{2} + c ^{2} + d ^{2} = 1$.	(B.6)

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:NUPHB AID:14363 /FLA [m1+; v1.285; Pm:6/06/2018; 11:	31] P.39 (1-49
N.T. Macpherson et al. / Nuclear Physics $B \leftrightarrow (\leftrightarrow \bullet \bullet) \leftrightarrow \bullet \bullet$	39
We can then calculate the 4d bispinors appearing in $(3.1a)$ – $(3.1f)$ and $(4.1a)$ – $(4.1f)$, we so we find it useful to parametrise	where to do
$a = a_1 + ia_2$, $b = b_1 + ib_2$, $c = c_1 + ic_2$, $d = d_1 + id_2$,	(B.7)
for a_i, b_i, c_i and d_i real. However we first note that in both IIA and IIB we must solve constraints	the 0-form
$(\psi_{\hat{\gamma}}^2)_0 = (\mathrm{Im}\psi_{\hat{\gamma}}^1)_0 = 0$,	(B.8
which reduce to	
$b_2 \cos \frac{\alpha}{2} + a_2 \sin \frac{\alpha}{2} = d_2 \cos \frac{\alpha}{2} + c_2 \sin \frac{\alpha}{2} = d_1 \cos \frac{\alpha}{2} + c_1 \sin \frac{\alpha}{2} = 0$.	(B.9)
We can solve these in general by fixing	
$a_2 = \lambda_1 \cos(\frac{\alpha}{2})$ $b_2 = -\lambda_1 \sin(\frac{\alpha}{2})$ $c_1 = \lambda_2 \cos(\frac{\alpha}{2})$,	(B 10
$c_2 = -\lambda_3 \cos(\frac{\alpha}{2})$ $d_1 = -\lambda_3 \sin(\frac{\alpha}{2})$ $d_2 = -\lambda_1 \sin(\frac{\alpha}{2})$,	(2110)
which turns (B.6) into	
$a_1^2 + b_1^2 + \lambda_1^2 + \lambda_2^2 + \lambda_3^2 = 1.$	(B.11)
In terms of this parametrisation the 4d bispinors are given by	
$\psi_{1}^{1} = a_{1} - i\lambda_{1} - ib_{1}v_{1} \wedge v_{2} - (\lambda_{2} - i\lambda_{3})v_{1} \wedge (w_{1} - iw_{2}) + (ia_{1} + \lambda_{1})w_{1}$	$\wedge w_2$
$+b_1v_1 \wedge v_2 \wedge w_1 \wedge w_2,$	
$\psi_{-} = (a_{1} - i \lambda_{1})v_{1} - i b_{1}v_{2} - (\lambda_{2} - i \lambda_{3})(w_{1} - i w_{2}) + (i a_{1} + \lambda_{1})v_{1} \wedge w_{1} \wedge w_{1} \wedge w_{2}$	w_2
$+ b_1 b_2 \wedge w_1 \wedge w_2$,	
$\psi_{+}^{2} = -(\lambda_{2} + i\lambda_{3}) - (a_{1} + i\lambda_{1})v_{1} \wedge (w_{1} - iw_{2}) - ib_{1}v_{2} \wedge (w_{1} - iw_{2})$	
$-i(\lambda_2+i\lambda_3)w_1\wedge w_2$,	
$\psi_{-}^{2} = -(\lambda_{2} + i \lambda_{3})v_{1} - (a_{1} + i \lambda_{1})(w_{1} - i w_{2}) - i b_{1}v_{1} \wedge v_{2} \wedge (w_{1} - i w_{2})$	
$+ (\lambda_3 - i \lambda_2) v_1 \wedge w_1 \wedge w_2$,	(B.12)
$ak^{1} = b$ $(ia + 1) = b$ $(ia + 1) = b$	
$\psi_{\hat{\gamma}+} = b_1 - (l u_1 + \lambda_1) v_1 \wedge v_2 - l (\lambda_2 - l \lambda_3) v_2 \wedge (w_1 - l w_2) + l b_1 w_1 \wedge w_2$	
$+ (a_1 - i \lambda_1) v_1 \wedge v_2 \wedge w_1 \wedge w_2,$	
$\psi_{\hat{\gamma}-}^1 = -b_1v_1 + (ia_1 + \lambda_1)v_2 + (\lambda_3 + i\lambda_2)v_1 \wedge v_2 \wedge (w_1 - iw_2) - ib_1v_1 \wedge u_2$	$w_1 \wedge w_2$
$- (a_1 - i\lambda_1)v_2 \wedge w_1 \wedge w_2,$	
$\psi_{2,1}^2 = (i\lambda_2 - \lambda_3)v_1 \wedge v_2 - b_1v_1 \wedge (w_1 - iw_2) - i(a_1 + i\lambda_1)v_2 \wedge (w_1 - iw_2)$)
$-(\lambda_2 + i\lambda_3)v_1 \wedge v_2 \wedge w_1 \wedge w_2 = i(\omega_1 + i\lambda_3)v_2 \wedge (\omega_1 + i\omega_2)$	/
(<i>A</i> ₂ + <i>i A</i> ₃) <i>v</i> ₁ / <i>v</i> ₂ / <i>w</i> ₁ / <i>w</i> ₂ ,	
$\psi_{\hat{\gamma}-}^{2} = (\lambda_{3} - i\lambda_{2})v_{2} + b_{1}(w_{1} - iw_{2}) + i(a_{1} + i\lambda_{1})v_{1} \wedge v_{2} \wedge (w_{1} - iw_{2})$	
$+ (\lambda_2 + i \lambda_3) v_2 \wedge w_1 \wedge w_2$.	

.IID·NUPHR	ARTICLE IN PRESS	8: 11:31] P40(1-49
40	<i>N.T. Macpherson et al. / Nuclear Physics B</i> ••• (••••) •••–•••	6, 11.51] F.40 (1-49
Appendix	C. The $SU(2)$ doublets of S^3	
There e	exist two independent spinors on S^3 that obey the Killing spinor relation	ons
$\nabla_a \xi$	$_{\pm} = \pm \frac{i}{2} \gamma_a \xi_{\pm},$	(C.1)
each of wh decompose $K^i_{\pm}, i = 1$,	hich preserves two supercharges. Additionally the global isometry group data $SO(4) = SU(2)_+ \times SU(2)$, so S^3 supports two sets of $SU(2)_+$, 2, 3, that are dual to one forms that obey	oup of <i>S</i> ³ can be) Killing vectors
dK_i^{\exists}	$^{\pm} \pm \frac{1}{2} \epsilon_{ijk} K_j^{\pm} \wedge K_k^{\pm},$	(C.2)
i.e. the rig $SU(2)_{\pm}$ de	ht-/left-invariant forms of $SU(2)$. It is possible to use the spinors on oublets. Consider the following vector with spinor entries	S^3 to construct
$\xi^{\alpha}_{\pm} =$	$=\left(egin{array}{c} \xi_{\pm} \ \xi_{\pm}^c \end{array} ight)^{lpha}.$	(C.3)
These tran	sform under the action of the spinorial Lie derivative as ¹¹	
$\mathcal{L}_{K_i^{\pm}}$	$\xi_{\pm}^{a} = \pm \frac{i}{2} (\sigma_{i})^{a}{}_{b} \xi_{\pm}^{\beta}, \mathcal{L}_{K_{i}^{\pm}} \xi_{\mp}^{a} = 0,$	(C.5)
for σ_i the transformation	Pauli matrices, which means that ξ^a_{\pm} transforms as a doublet under ations and a singlet under $SU(2)_{\mp}$.	er local $SU(2)_{\pm}$
Appendix	D. Supersymmetry conditions for three-dimensional external spa	acetimes
In [31], of bispino conditions	supersymmetry conditions for 3+7 dimensional compactifications ar rs. The repackaging of the supersymmetry conditions was done und :	e given in terms er the following
• The ex	sternal space is Minkowski.	
• The Sp • The N	SNS flux H does not have an external component.	
In this sec tions. Our constraints	tion, we will look at relaxing the latter two conditions to obtain mo starting point will be the ten-dimensional bispinor description of the s, as described in [37]:	ere general solu- supersymmetry
¹¹ The spino	orial Lie derivative along a Killing vector K is defined as	
$\mathcal{L}_{K}\epsilon$	$= K^{\mu} \nabla_{\mu} \epsilon + \frac{1}{8} (dK)_{\mu\nu} \gamma^{\mu\nu} \epsilon.$	(C.4)
The easiest v $e^1 = \frac{1}{2}d\theta$,	way to see that this leads to the claimed transformation property, is to parametrise th $e^2 = \frac{1}{2}\sin\theta d\phi$, $e^3 = \frac{1}{2}(d\psi + \cos\theta d\phi)$ and take the flat space gamma-matrices to b	the vielbein on S^3 as the pauli matrices
σ_i . Then (C.) then precisely	1) is solved by $\xi_{+} = e^{\frac{i}{2}\phi\sigma_{1}}e^{\frac{i}{2}\phi\sigma_{3}}\xi_{+}^{0}, \xi_{-} = e^{-\frac{i}{2}\psi\sigma_{3}}\xi_{-}^{0}$ for ξ_{\pm}^{0} constant 2d spinors. The y $K_{i}^{+} = -i\mathrm{Tr}(\sigma_{i}dgg^{-1}), K_{i}^{-} = -i\mathrm{Tr}(\sigma_{i}g^{-1}dg)$ for $g = e^{\frac{i}{2}\phi\sigma_{3}}e^{\frac{i}{2}\phi\sigma_{2}}e^{\frac{i}{2}\psi\sigma_{3}}$. The re	e $SU(2)_{\pm}$ forms are sult is then not hard

JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.41 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-••• $d_H \left(e^{-\Phi} \Psi \right) + \tilde{K} \wedge F + \iota_K F = 0$ (D.1a) $d\tilde{K} = \iota_{\kappa} H$ (D.1b) $\mathcal{L}_K g_{10} = 0$ (D.1c) $\left(e_{+1}\cdot\Psi\cdot e_{+2},\Gamma^{MN}\left(\pm d_H\left(e^{-\Phi}\Psi\cdot e_{+2}\right)+\frac{1}{2}e^{\Phi}d^{\dagger}\left(e^{-2\Phi}e_{+2}\right)\Psi-F\right)\right)=0,$ (D.1d) $\left(e_{+1}\cdot\Psi\cdot e_{+2}, \left(d_H\left(e^{-\Phi}\Psi\cdot e_{+1}\right) - \frac{1}{2}e^{\Phi}d^{\dagger}\left(e^{-2\Phi}e_{+1}\right)\Psi - F\right)\Gamma^{MN}\right) = 0,$ (D.1e) with $\Psi = \epsilon_1 \otimes \bar{\epsilon_2} , \quad K_M = \frac{1}{32} \left(\bar{\epsilon_1} \Gamma_M \epsilon_1 + \bar{\epsilon_2} \Gamma_M \epsilon_2 \right) , \quad \tilde{K}_M = \frac{1}{32} \left(\bar{\epsilon_1} \Gamma_M \epsilon_1 - \bar{\epsilon_2} \Gamma_M \epsilon_2 \right) .$ (D.2) The final two equations are known as the pairing equation; we refer to [37] for more details, and will follow along the lines of section 4.2 in the following. We consider the case where the Killing spinors are given by (2.2). Due to the properties of Spin(1, 2), we can define $\frac{1}{2}\bar{\zeta}\gamma_{\mu}\zeta = v_{\mu} , \quad \frac{1}{2}\bar{\zeta}\gamma_{\mu\nu}\zeta = (\star_{3}v)_{\mu\nu} ,$ (D.3) with the other bilinears vanishing. Since we are considering flat space, ζ are covariantly constant, hence dv = 0. Making use of the spinor decomposition, it follows that $8e^{-A}\Psi = v \wedge \Phi_{\mp} - \star_3 v \wedge \Phi_{\pm} , \quad K = \frac{1}{8}e^A(|\chi_1|^2 + |\chi_2|^2)v , \quad \tilde{K} = \frac{1}{8}e^A(|\chi_1|^2 - |\chi_2|^2)v ,$ (D.4) where we have defined $\Phi_+ + i \Phi_- = 8e^{-A}\chi_1 \otimes \chi_2^{\dagger}$ and K, \tilde{K} should be read as 1-forms in ten dimensions. Using the flux decomposition $F = f + e^{3A} \operatorname{Vol}_3 \wedge \star_7 \lambda(f)$, $H = H_3 + e^{3A} h \operatorname{Vol}_3$. (D.5) We will first solve (D.1c): since by construction, v, K are Killing vectors, we must have $|\chi_1|^2 + |\chi_2|^2 = c_+ e^A$. (D.6) Next let's consider (D.1b), which leads to $c_+e^{3A}h=0$, $|\chi_1|^2-|\chi_2|^2=c_-e^{-A}$. (D.7) Next, let us consider (D.1a). We find that $\mathrm{d}_{H_3}\left(e^{3A-\Phi}\Phi_{\pm}\right) = c_+ e^{3A} \star_7 \lambda(f)$ (D.8) $\mathrm{d}_{H_3}\left(e^{2A-\Phi}\Phi_{\mp}\right)=c_-f\;.$ In addition, the fact that $c_{-} \neq 0$ does not change the argument of [37], so the pairing equations remain unchanged, leading to $(f, \Phi_{\pm}) = 0$. (D.9)

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

 JID:NUPHB
 AID:14363 /FLA
 [m1+; v1.285; Prn:6/06/2018; 11:31] P.42 (1-49)

 42
 N.T. Macpherson et al. / Nuclear Physics B ••• (••••)
 •••• -•••

Appendix E. M-theory

The focus of this paper are backgrounds in type IIA and type IIB. In this appendix, we will discuss M-theory backgrounds on $\mathbb{R}^{1,2} \times S^3 \times M_5$. Given equivalent internal spinor norms, our (massive) IIA classification is complete (up to finding solutions to PDE). Therefore, a significant number of backgrounds one would obtain from a similar analysis of M-theory are those which one can obtain from uplifting our massless IIA backgrounds. Novel solutions from a complete M-theory analysis would be backgrounds satisfying one of the two conditions: either M_5 does g not admit an S^1 factor to be integrated out to perform the dimensional reduction to IIA, or the internal component of the Killing spinor on $M_8 = S^3 \times M_5$ is such that after the reduction, the resulting seven-dimensional internal components of the IIA spinors are not of equal norm. Such a full M-theory classification is beyond the scope of this paper. Instead, we aim to make

contact with the literature of M-theory on $\mathbb{R}^{1,2} \times M_8$, which is much studied (see for example [62–67]). We will derive the decomposed supersymmetry conditions, and give several simple classes of solutions. For $\mathcal{N} = 1$ solutions to the supersymmetry constraints on $\mathbb{R}^{1,2} \times M_8$, the Killing spinor ϵ

For $\mathcal{N} = 1$ solutions to the supersymmetry constraints on $\mathbb{R}^{1,2} \times M_8$, the Killing spinor ϵ decomposes as

$$\epsilon = \xi \otimes (\chi_+ + \chi_-) , \qquad (E.1)$$

where ξ is a Majorana spinor of Spin(1, 2) and χ_{\pm} are chiral Majorana spinors of Spin(8). Generically, χ_{\pm} can have zeroes, and the structure group of M_8 is SO(8), although a Spin(7)-structure can be defined on the auxiliary space $M_8 \times S^1$ [66]. In the case where one of the two does not vanish, the structure group reduces to Spin(7) [65]. If both chiral spinors have no zeroes, both of them define a Spin(7)-structure: the intersection of the two leads to a reduction of the structure group to G_2 .¹² The reduction of the structure group leads to the exis-tence of globally defined invariant tensors. Instead, we will work locally, and consider patches where either one or both are non-zero.

E.1. Spin(7) holonomy

 $\epsilon = \zeta \otimes \chi_+ \,. \tag{E.2}$

Following the conventions of [65], the general solution to the M-theory supersymmetry constraints with these ansätze is that

$$ds^{2} = e^{2\Delta} ds^{2}(\mathbb{R}^{1,2}) + e^{-\Delta} ds^{2}(M_{8}), \quad G = \text{Vol}_{3} \wedge d(e^{3\Delta}) + F , \quad (E.3) \quad 37$$

where $ds^2(M_8)$ a metric of Spin(7) holonomy. The four-form F lies in the **27** of Spin(7), i.e., it satisfies

$$F^{pqr}_{\ m}\Psi_{npqr} = 0 , \qquad (E.4)$$

with $\Psi_{mnpq} = \bar{\chi} \gamma_{mnpq} \chi$ the invariant four-form defining the Spin(7)-structure. In addition, the Bianchi identity and equation of motion for *F* require that *F* is harmonic and satisfies

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

⁴⁶ $\frac{12}{12}$ Note that this is only the case for two *Spin*(7)-structures defined in terms of opposite chirality spinors. Given two ⁴⁷ same chirality globally well-defined nowhere vanishing spinors, the structure group instead reduces to *Spin*(6) $\simeq SU(4)$.

ARTICLE IN PRESS

 JID:NUPHB
 AID:14363 /FLA
 [m1+; v1.285; Prn:6/06/2018; 11:31] P.43 (1-49)

 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••
 43

$$d(\star_8 d(e^{-3\Delta})) + \frac{1}{2}F \wedge F = 0$$
(E.5)

3 away from M2-brane sources.

Let us now impose $M_8 = S^3 \times M_5$. The internal metric and Killing spinor decompose as

$$ds^{2}(M_{8}) = e^{2C} ds^{2}(S^{3}) + ds^{2}(M_{5}), \quad \chi_{+} = \begin{pmatrix} 1\\ 0 \end{pmatrix} \otimes (\xi \otimes \eta + \xi^{c} \otimes \eta^{c}), \quad (E.6) \quad \stackrel{5}{6}$$

and the gamma matrices decompose as

٦õ

$$\gamma_{\alpha}^{(8)} = \sigma_1 \otimes \sigma_{\alpha} \otimes \mathbb{I} , \quad \gamma_a^{(8)} = \sigma_2 \otimes \mathbb{I} \otimes \gamma_a , \quad B^{(8)} = \sigma_3 \otimes \sigma_2 \otimes B_5 , \quad (E.7)$$

with the charge conjugation matrix satisfying $B_{5*} = -B_5$, α an index on S^3 and a an index on M_5 . The pseudoreal Spin(5) spinor η of unit norm gives rise to an SU(2)-structure on M_5 , via [68]

$$\eta \otimes \eta^{\dagger} = \frac{1}{4}(1+V) \wedge e^{-iJ}, \quad \eta \otimes \eta^{c\dagger} = \frac{1}{4}(1+V) \wedge \omega.$$
(E.8)

The (local) SU(2)-structure consists of a real one-form V, real two-form J and a complex twoform ω such that $J \wedge J = \frac{1}{2}\omega \wedge \omega^*$ and $\iota_V J = \iota_V \omega = J \wedge \omega = \omega \wedge \omega = 0$.

By making use of this decomposition, the Spin(7) four-form decomposes in terms of the $\mathbb{I} \times SU(2)$ -structure as

$$\Psi = -\frac{1}{2}J \wedge J + \frac{e^{\tilde{C}}}{2}V \wedge (K_1 \wedge \operatorname{Re}\omega + K_2 \wedge \operatorname{Im}\omega + K_3 \wedge J)$$
(E.9)
(E.9)

$$-\nu \frac{e^{2C}}{4} (dK_1 \wedge \operatorname{Re}\omega + dK_2 \wedge \operatorname{Im}\omega + dK_3 \wedge J) - \nu e^{3\tilde{C}} V \wedge \operatorname{Vol}(S^3) .$$
(E.10)

²⁶ Using the decomposed Ψ , we examine the supersymmetry conditions. First, we consider the flux ²⁷ component *F*, which we decompose as

$$F = e^{3C} \operatorname{Vol}(S^3) \wedge F_1 + \star_5 \tilde{F}_1 , \qquad (E.11) \quad 29$$

with F_1 , $\tilde{F_1}$ one-forms on M_5 . Inserting this and (E.9) into (E.4), it follows from (m, n) = (a, b)that $F_1 \sim \tilde{F_1}$, $F_1^a V_a = 0$ and from $(m, n) = (\alpha, a)$ that $F_1^a \omega_{ab} = F_1^a \omega_{ab}^* = F_1^a J_{ab} = 0$. Hence $F_1 = \tilde{F_1} = 0$, hence F = 0. The equation of motion for the flux (E.5) thus reduces to the following constraint on the warp factors:

$$d(e^{3C} \star_5 d(e^{-3\Delta})) = 0.$$
 (E.12) ³⁵
₃₆

Next, the requirement that M_8 is of Spin(7) holonomy is equivalent to the closure of Ψ , which is equivalent to

³⁹
₄₀
$$d(e^{3\tilde{C}}V) = d(e^{2\tilde{C}}J) + 2\nu e^{\tilde{C}}V \wedge J = d(e^{2\tilde{C}}\omega) + 2\nu e^{\tilde{C}}V \wedge \omega = d(J \wedge J) = 0.$$
 (E.13) ³⁹
₄₀

In general, this means that locally $V = e^{-3C} d\tau$, and we will write

$$ds^{2}(M_{8}) = e^{2C} ds^{2}(S^{3}) + e^{-6C} d\tau^{2} + ds_{4}^{2}$$
(E.14)
$$(E.14)$$

Let us give some simple classes of examples for which the above conditions are solved.

• In the case that
$$C = C(\tau)$$
, the metric ds_4^2 is conformally Calabi–Yau. Let $J = e^{2(W-C)}\tilde{J}$, 46
 $\omega = e^{2(W-C)}\tilde{\omega}$. Then provided that we define $W(\tau)$ to satisfy 47

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

з

 JID:NUPHB
 AID:14363 /FLA
 [m1+; v1.285; Prn:6/06/2018; 11:31] P.44 (1-49)

 44
 N.T. Macpherson et al. / Nuclear Physics B ••• (••••)
 •••• -•••

$$W' + \nu e^{-4C} = 0 \,, \tag{E.15}$$

we find that $d\tilde{J} = d\tilde{\omega} = 0$.

• By taking $C = -\frac{1}{4} \log(4\tau)$, we find that $dJ = d\omega = 0$, hence ds_4^2 is a Calabi–Yau metric.¹³ Introducing $\rho = (4\tau)^{1/4}$, the metric reduces to

$$ds^{2}(M_{8}) = d\rho^{2} + \rho^{2}e^{2C}ds^{2}(S^{3}) + ds_{4}^{2}, \qquad (E.16)$$

and thus $M_8 = \mathbb{R}^4 \times Y_2$, where Y_2 is a Calabi–Yau two-fold.

• Next, let us examine Sasaki–Einstein structures, as well as a class of generalizations. It can be shown that any five-dimensional Sasaki–Einstein can be defined by means of a set of real forms (\tilde{V}, ω_j) , j = 1, 2, 3 with \tilde{V} a one-form and ω_j two-forms. These satisfy [15]

$$d\omega_1 = d(\omega_2 + i\omega_3) + 3i\tilde{V} \wedge (\omega_2 + i\omega_3) = 0.$$
(E.17)

¹⁶ A more general class of spaces are the so-called hypo manifolds [69], satisfying $d\omega_1 = d(\tilde{V} \wedge \omega_2) = d(\tilde{V} \wedge \omega_3) = 0$, which themselves are a subclass of balanced manifolds [70], satisfying ¹⁸

$$d(\omega_1 \wedge \omega_1) = d(\tilde{V} \wedge \omega_2) = d(\tilde{V} \wedge \omega_3) = 0.$$
(E.18) ₁₉

By setting $J = \omega_1$, $e^c V = \tilde{V}$, $\text{Re}\omega = \omega_2$, $\text{Im}\omega = \omega_3$, it follows that any solution to the supersymmetry constraints is a balanced metric. On the other hand, any solution to the supersymmetry constraints which is hypo automatically is such that ds_4^2 is Calabi–Yau. This leads to the conclusion that the spinors do not define a Sasaki–Einstein on M_5 , as the base space of a Sasaki–Einstein manifold is not Ricci-flat. Another way to see this is to note that Sasaki–Einstein metrics can be written as a fibration over a Kähler–Einstein base, but it is clear that since the supersymmetry constraints are invariant under permutations of $(J, \text{Re}\omega, \text{Im}\omega)$, ds_4^2 cannot be non-Calabi–Yau Kähler.

E.2.
$$G_2$$
-structure

Next, let us examine the case where both internal chiral Killing spinors are (locally) non-vanishing. Again following [65], the Killing spinor is given by

$$\epsilon = e^{-\Delta}\theta \otimes (\chi_{+} + \chi_{-}) , \qquad (E.19) \quad {}^{34}$$

leading to the solution

$$ds^{2} = e^{2\Delta} \left(ds^{2}(\mathbb{R}^{1,2}) + ds^{2}(M_{8}) \right), \quad G = e^{3\Delta}(\text{Vol}_{3} \wedge f + F).$$
(E.20)

This time, the metric is not of special holonomy. Instead, it allows a G_2 -structure with non-trivial torsion. The norms of χ_{\pm} can be parametrised as

$$|\chi_{\pm}|^2 = 1 \pm \sin \zeta$$
, (E.21) ⁴²

with $\sin \zeta$ a function of M_8 such that the norms of χ_{\pm} are non-vanishing. The bilinears of χ_{\pm} defining the G_2 -structure are given by

¹³ We have redefined τ to absorb the sign ν .

g

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.45 (1-49) N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-••• $K_m = \frac{1}{\cos\zeta} \chi^{\dagger}_+ \gamma^{(8)}_m \chi_-, \quad \varphi_{mnp} = \frac{1}{\cos\zeta} \chi^{\dagger}_+ \gamma^{(8)}_{mnp} \chi_-.$ (E.22) $d(e^{3\Delta}\cos\zeta K) = 0,$ $K \wedge d(e^{6\Delta}\iota_K \star_8 \varphi) = 0,$ $\omega_0 \omega = 0,$ In terms of these, the constraints (E.23a) (E.23b) $d(e^{12\Delta}\cos\zeta\,\varphi\wedge\iota_K\star_8\varphi)=0\,,$ (E.23c) $\cos \zeta d(\varphi) \wedge \varphi + 4 \star_8 d\zeta - 2 \cos \zeta \star_8 f = 0,$ (E.23d) $d(e^{3\Delta}\sin\zeta) - e^{3\Delta}f = d(e^{6\Delta}\cos\zeta\varphi) + e^{6\Delta}\star_8F - e^{6\Delta}\sin\zeta F = 0,$ (E.23e) are locally equivalent to the supersymmetry conditions. Next, we split $M_8 = S^3 \times M_5$, leading to the following decomposition of the metric and flux: $ds^{2}(M_{8}) = e^{2C} ds^{2}(S^{3}) + ds^{2}(M_{5})$, $F = e^{3C} F_{1} \wedge \operatorname{Vol}(S^{3}) + F_{4}$. (E.24) The spinors decompose as $\chi_{+} = \sqrt{1 + \sin \zeta} \begin{pmatrix} 1 \\ 0 \end{pmatrix} \otimes (\xi \otimes \eta_{1} + \xi^{c} \otimes \eta_{1}^{c}) , \quad \chi_{-} = \sqrt{1 - \sin \zeta} \begin{pmatrix} 0 \\ 1 \end{pmatrix} \otimes (\xi \otimes \eta_{2} + \xi^{c} \otimes \eta_{2}^{c})$ (E.25) and the gamma matrices again decompose as (E.7). The Spin(5) spinors can be expanded in a common basis as $\eta^{1} = \eta, \quad \eta^{2} = a_{0}\eta + a\eta^{c} + \frac{b}{2}\overline{w}\eta, \quad |a_{0}|^{2} + |a|^{2} + |b|^{2} = 1$ (E.26) where b can be made real by rotating the 1-form w and η is unit norm. We assume $w = w_1 + iw_2$ is locally non-vanishing. As a result, a second locally non-vanishing 1-form can be defined as $u = \frac{1}{2}\iota_{w^*}\omega ,$ (E.27) with ω defined as in (E.8). We thus see that the local SU(2)-structure defined on M_5 by η reduces further to a trivial structure, with the local vielbein defined by (V, w_1, w_2, u_1, u_2) . We now express the G_2 -structure (E.22) in terms of the trivial structure of $S^3 \times M_5$. The first bilinear we calculate is $K = \frac{e^C}{2} (\operatorname{Im} a_0 K_1 + \operatorname{Re} a_0 K_2 - \operatorname{Im} a K_3) - bu_2 - \operatorname{Re} a V,$ (E.28) and the only way to make this compatible with (E.23a) is to set $a_0 = \text{Im}a = 0$ (E.29) so that the spinors $\eta_{1,2}$ are nowhere parallel. We are now free to parametrise (E.30) $b = \cos \alpha$, $\text{Re}a = \sin \alpha$ and rotate to a frame where

ANTIOLL IN FILOS JID:NUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:	31] P.46 (1-49)
46 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••	-] -(-/
K = V.	(E.31)
Having done this the other bilinears take the form	
-2C	
$\varphi = -\cos\alpha e^1 \wedge e^2 \wedge e^3 - e^{3C}\sin\alpha \operatorname{Vol}(S^3) - \nu \frac{e^{-C}}{4}\cos\alpha e^i \wedge dK_i,$	
4	
$-\frac{e^{C}}{K_{i}} \wedge (u_{1} \wedge e^{i} + \frac{1}{\epsilon_{i,ik}} \sin \alpha e^{j} \wedge e^{k}).$	
$2^{-i_1 + i_2 + i_1 + i_2 + i_2 + i_3 + \dots + i_2 + i_3 + \dots + i_3}$	
$\iota_K \star_8 \varphi = u_1 \wedge (\sin \alpha e^1 \wedge e^2 \wedge e^3 - e^{3C} \cos \alpha \wedge \operatorname{Vol}(S^3))$	
$+\nu \frac{e^{2C}}{2}(\sin \alpha u_1 \wedge e^i + \frac{1}{2}\epsilon_{ijk}e^j \wedge e^k) \wedge dK_i - \frac{e^C}{2}\cos \alpha \epsilon_{ijk}u_1 \wedge e^j \wedge e^k$	$e^k \wedge K_i$,
4 2 3 4 4 3 4	(E 32)
	(1.52)
where we have defined	
$e = (w_1, w_2, -u_2),$	(E.33)
for ease of presentation. Remark that	
for ease of presentation. Remark that	
$\varphi \wedge \iota_K \star_8 \varphi = 7\nu \operatorname{Vol}_7,$	(E.34)
where Vol7 is the volume form of the manifold spanned by the warped left-invariant	forms of S^3
and the vielbein, with orientation	
$\begin{bmatrix} e^C & e^C & e^C \end{bmatrix}$	
$\left\{\frac{1}{2}K_1, \frac{1}{2}K_2, \frac{1}{2}K_3, u_1, e^1, e^2, e^3\right\}.$	
Inserting these definitions for w and $w \neq w$ into (E 23a) (E 23a) lead to the 5d cond	itions
Inserting these definitions for φ and $v_K \star_S \varphi$ into (E.2.54)–(E.2.54) fead to the 5d cond	ittions
$d(e^{3\Delta}\cos\zeta V) = d(e^{6\Delta+3C}\cos\alpha u_1) \wedge V = d(\frac{e^{-6\Delta-3C}}{2}u_1) \wedge e^1 \wedge e^2 \wedge e^2$	$r^3 = 0$.
$\cos^3 \alpha \cos^2 \zeta$	(E 35a)
	(E.55a)
$d(e^{6\Delta+2C}\cos\zeta\cos\alpha e^i) + v e^{6\Delta+C}\cos\zeta(2u_1 \wedge e^i + \sin\alpha\epsilon_{ijk}e^j \wedge e^k) = 0,$	(E.35b)
$\left(d(e^{6\Delta+2C}(\sin\alpha u_1\wedge e^i+\frac{1}{2}\epsilon_{ijk}e^j\wedge e^k))+\nu e^{6\Delta+C}\cos\alpha\epsilon_{ijk}u_1\wedge e^j\wedge e^k\right)\wedge$	V=0,
(2)	(E.35c)
1/-2C is a key transition in key of	(E 25 I)
$a(e^{-\epsilon}\cos\alpha\epsilon_{ijk}e^{s})\wedge e^{\epsilon}\wedge u_1\wedge v=\epsilon_{ijk}u_1\wedge au_1\wedge e^{s}\wedge e^{\epsilon}=0,$	(E.35d)
$-2\cos\zeta d\alpha \wedge e^1 \wedge e^2 \wedge e^3 + 2 \star_5 d\zeta - \cos\zeta \star_5 f = d(e^{3\Delta}\sin\zeta) - e^{3\Delta}f =$	0,
	(E.35e)
$d(e^{6\Delta+3C}\cos z\sin \alpha) + e^{6\Delta+3C}(z+E) + \sin zE) = 0$	(E 255
$a(e \cos\zeta \sin a) + e (- \star_5 r_4 + \sin\zeta r_1) = 0,$	(E.331)
$d(e^{6\Delta}\cos\zeta\cos\alpha e^1\wedge e^2\wedge e^3)+e^{6\Delta}(\star_5F_1-\sin\zeta F_4)=0,$	(E.35g)
where we have used that are 1 () and are any show that are up () is incompletent	with suman

46 where we have used that $\cos \zeta \neq 0$ and one can show that $\cos \alpha = 0$ is inconsistent with super-47 symmetry. In addition, the Bianchi identities and equations of motion for the flux reduce to 47

JID:NUPHB AID:14363 /FLA [m1+: v1.285: Prn:6/06/2018: 11:31] P.47 (1-49)

N.T. Macpherson et al. / Nuclear Physics B ••• (••••)

 $d\left(e^{3\Delta}f\right) = d\left(e^{3\Delta+3C}F_1\right) = d\left(e^{3\Delta}F_4\right) = 0$ $d\left(e^{6\Delta+3C}\star_5 F_4\right) - e^{6\Delta+3C}f \wedge F_1 = 0$ з (E.36) $d\left(e^{6\Delta}\star_5 F_1\right) - e^{6\Delta}f \wedge F_4 = 0$ $d\left(e^{6\Delta+3C}\star_5 f\right)+e^{6\Delta+3C}F_1\wedge F_4=0.$

Note that the signs are such that supersymmetry together with the Bianchi identities imply the first two equations of motions. All Hodge duals in the above are with respect to the unwarped five-dimensional metric.

References

[1] B.E.W. Nilsson, C.N. Pope, Hopf fibration of eleven-dimensional supergravity, Class. Quantum Gravity 1 (1984) 499. https://doi.org/10.1088/0264-9381/1/5/005. [2] J. Bagger, N. Lambert, Modeling multiple M2's, Phys. Rev. D 75 (2007) 045020, https://doi.org/10.1103/PhysRevD. 75.045020, hep-th/0611108; J. Bagger, N. Lambert, Gauge symmetry and supersymmetry of multiple M2-branes, Phys. Rev. D 77 (2008) 065008, https://doi.org/10.1103/PhysRevD.77.065008, 0711.0955; J. Bagger, N. Lambert, Comments on multiple M2-branes, J. High Energy Phys. 0802 (2008) 105, https://doi.org/ 10.1088/1126-6708/2008/02/105, 0712.3738. [3] O. Aharony, O. Bergman, D.L. Jafferis, J. Maldacena, N=6 superconformal Chern-Simons-matter theories, M2-branes and their gravity duals, J. High Energy Phys. 0810 (2008) 091, https://doi.org/10.1088/1126-6708/2008/ 10/091, 0806.1218. [4] D. Lust, D. Tsimpis, Supersymmetric AdS(4) compactifications of IIA supergravity, J. High Energy Phys. 0502 (2005) 027, https://doi.org/10.1088/1126-6708/2005/02/027, hep-th/0412250. [5] J. Bovy, D. Lust, D. Tsimpis, N = 1, 2 supersymmetric vacua of IIA supergravity and SU(2) structures, J. High Energy Phys. 0508 (2005) 056, https://doi.org/10.1088/1126-6708/2005/08/056, hep-th/0506160. [6] C. Kounnas, D. Lust, P.M. Petropoulos, D. Tsimpis, AdS4 flux vacua in type II superstrings and their domain-wall solutions, J. High Energy Phys. 0709 (2007) 051, https://doi.org/10.1088/1126-6708/2007/09/051, 0707.4270. [7] A. Tomasiello, New string vacua from twistor spaces, Phys. Rev. D 78 (2008) 046007, https://doi.org/10.1103/ PhysRevD.78.046007, 0712.1396. [8] P. Koerber, D. Lust, D. Tsimpis, Type IIA AdS(4) compactifications on cosets, interpolations and domain walls, J. High Energy Phys. 0807 (2008) 017, https://doi.org/10.1088/1126-6708/2008/07/017, 0804.0614. [9] C. Caviezel, P. Koerber, S. Kors, D. Lust, D. Tsimpis, M. Zagermann, The effective theory of type IIA AdS(4) compactifications on nilmanifolds and cosets, Class. Quantum Gravity 26 (2009) 025014, https://doi.org/10.1088/ 0264-9381/26/2/025014.0806.3458. [10] D. Lust, D. Tsimpis, Classes of AdS(4) type IIA/IIB compactifications with SU(3) x SU(3) structure, J. High Energy Phys. 0904 (2009) 111, https://doi.org/10.1088/1126-6708/2009/04/111, 0901.4474. [11] K. Behrndt, M. Cvetic, General N=1 supersymmetric fluxes in massive type IIA string theory, Nucl. Phys. B 708 (2005) 45, https://doi.org/10.1016/j.nuclphysb.2004.12.004, hep-th/0407263. [12] A. Rota, A. Tomasiello, AdS₄ compactifications of AdS₇ solutions in type II supergravity, J. High Energy Phys. 1507 (2015) 076, https://doi.org/10.1007/JHEP07(2015)076, 1502.06622. [13] O. Varela, AdS₄ solutions of massive IIA from dyonic ISO(7) supergravity, J. High Energy Phys. 1603 (2016) 071, https://doi.org/10.1007/JHEP03(2016)071, 1509.07117. [14] S. Grigorian, G2-structures for N=1 supersymmetric AdS4 solutions of M-theory, 1711.10619. [15] D. Lust, D. Tsimpis, New supersymmetric AdS(4) type II vacua, J. High Energy Phys. 0909 (2009) 098, https:// doi.org/10.1088/1126-6708/2009/09/098, 0906.2561. [16] M. Petrini, A. Zaffaroni, N=2 solutions of massive type IIA and their Chern–Simons duals, J. High Energy Phys. 0909 (2009) 107, https://doi.org/10.1088/1126-6708/2009/09/107, 0904.4915. [17] Y. Lozano, N.T. Macpherson, A new AdS₄/CFT₃ dual with extended SUSY and a spectral flow, J. High Energy Phys. 1411 (2014) 115, https://doi.org/10.1007/JHEP11(2014)115, 1408.0912; Y. Lozano, N.T. Macpherson, J. Montero, A $\mathcal{N} = 2$ supersymmetric AdS₄ solution in M-theory with purely mag-netic flux, J. High Energy Phys. 1510 (2015) 004, https://doi.org/10.1007/JHEP10(2015)004, 1507.02660.

Please cite this article in press as: N.T. Macpherson et al., $Mink_3 \times S^3$ solutions of type II supergravity, Nucl. Phys. B (2018), https://doi.org/10.1016/j.nuclphysb.2018.05.021

JID:	JUPHB AID:14363 /FLA [m1+; v1.285; Prn:6/06/2018; 11:31] P.48 (1-49)
48	N.T. Macpherson et al. / Nuclear Physics $B \leftrightarrow (\bullet \bullet \bullet \bullet) \bullet \bullet \bullet - \bullet \bullet \bullet$
[18]	A. Guarino, D.L. Jafferis, O. Varela, String Theory origin of dyonic N=8 supergravity and its Chern–Simons duals,
[10]	Phys. Rev. Lett. 115 (9) (2015) 091601, https://doi.org/10.1103/PhysRevLett.115.091601, 1504.08009.
[20]	A. Passias, G. Solard, A. Toliasieno, $N = 2$ supersymmetric AdS4 solutions of type IIB supergravity, 1709.09009. M Gabella D Martelli A Passias I Sparks $N = 2$ supersymmetric AdS4 solutions of M-theory Commun Math
[=0]	Phys. 325 (2014) 487, https://doi.org/10.1007/s00220-013-1865-0, 1207.3082.
[21]	L.A. Pando Zayas, D. Tsimpis, C.A. Whiting, Supersymmetric IIB background with AdS4 vacua from massive IIA
	supergravity, Phys. Rev. D 96 (4) (2017) 046013, https://doi.org/10.1103/PhysRevD.96.046013, 1701.01643.
[22]	Y. Pang, J. Rong, N=3 solution in dyonic ISO(7) gauged maximal supergravity and its uplift to massive type IIA
[22]	supergravity, Phys. Rev. D 92 (8) (2015) 085037, https://doi.org/10.1103/PhysRevD.92.085037, 1508.05376.
[23]	Ianus I High Energy Phys 0706 (2007) 021 https://doi.org/10.1088/1126-6708/2007/06/021_0705.0022
[24]	E. D'Hoker, J. Estes, M. Gutperle, Exact half-BPS type IIB interface solutions. II. Flux solutions and multi-Janus,
	J. High Energy Phys. 0706 (2007) 022, https://doi.org/10.1088/1126-6708/2007/06/022, 0705.0024.
[25]	B. Assel, C. Bachas, J. Estes, J. Gomis, Holographic duals of D=3 N=4 superconformal field theories, J. High
10/2	Energy Phys. 1108 (2011) 087, https://doi.org/10.1007/JHEP08(2011)087, 1106.4253.
[26]	B. Assel, C. Bachas, J. Estes, J. Gomis, IIB duals of $D=5$ N=4 circular quivers, J. High Energy Phys. 1212 (2012) 044 https://doi.org/10.1007/IIIED12(2012)044, 1210.2500
[27]	A.S. Haupt, S. Lautz, G. Papadopoulos, AdS4 backgrounds with N>16 supersymmetries in 10 and 11 dimensions.
1-11	1711.08280 [hep-th].
[28]	J.P. Gauntlett, D. Martelli, D. Waldram, Superstrings with intrinsic torsion, Phys. Rev. D 69 (2004) 086002, https://
	doi.org/10.1103/PhysRevD.69.086002, hep-th/0302158.
[29]	J.P. Gauntlett, N. Kim, D. Martelli, D. Waldram, Five-branes wrapped on SLAG three cycles and related geometry, J. High Engage Phys. 0111 (2001) 018. https://doi.org/10.1088/1124 (2082)001(11/018.html)
[30]	J. High Energy Phys. 0111 (2001) 018, https://doi.org/10.1088/1120-6/08/2001/11/018, hep-th/0110034. M. Grana R. Minasian M. Petrini, A. Tomasiello, Generalized structures of N=1 vacua. I. High Energy Phys. 0511.
[50]	(2005) 020. https://doi.org/10.1088/1126-6708/2005/11/020. hep-th/0505212.
[31]	M. Haack, D. Lust, L. Martucci, A. Tomasiello, Domain walls from ten dimensions, J. High Energy Phys. 0910
	(2009) 089, https://doi.org/10.1088/1126-6708/2009/10/089, 0905.1582.
[32]	D. Andriot, New supersymmetric vacua on solvmanifolds, J. High Energy Phys. 1602 (2016) 112, https://doi.org/
[22]	10.1007/JHEP02(2016)112, 1507.00014. D. Andriot I. Plåbäck T. Ven Piot. Minkowski flux vegus of two II supergravities. Days. Pay. Lett. 118 (1) (2017)
[33]	D. Andriot, J. Bladack, 1. van Kiel, Minkowski nux vacua of type if supergravities, rhys. Rev. Lett. 118 (1) (2017) 011603. https://doi.org/10.1103/PhysRevI ett. 118.011603. 1609.00729
[34]	N. Halmagyi, D. Israel, M. Sarkis, E.E. Svanes, Heterotic hyper-Kähler flux backgrounds, J. High Energy Phys.
	1708 (2017) 138, https://doi.org/10.1007/JHEP08(2017)138, 1706.01725.
[35]	D. Andriot, On classical de Sitter and Minkowski solutions with intersecting branes, 1710.08886.
[36]	N.T. Macpherson, A. Tomasiello, Minimal flux Minkowski classification, 1612.06885.
[3/]	A. Iomasieno, Generalized structures of ten-dimensional supersymmetric solutions, J. High Energy Phys. 1205 (2012) 073 https://doi.org/10.1007/IHEP03(2012)073_1109.2603
[38]	U. Gran, J. Gutowski, G. Papadopoulos, D. Roest, Systematics of IIB spinorial geometry, Class. Quantum Gravity
	23 (2006) 1617, https://doi.org/10.1088/0264-9381/23/5/012, hep-th/0507087.
[39]	J.P. Gauntlett, D. Martelli, J. Sparks, D. Waldram, Supersymmetric AdS(5) solutions of M theory, Class. Quantum
F 401	Gravity 21 (2004) 4335, https://doi.org/10.1088/0264-9381/21/18/005, hep-th/0402153.
[40]	D. Youm, Partially localized intersecting BPS branes, Nucl. Phys. B 556 (1999) 222, https://doi.org/10.1016/S0550- 2213(00)00384 3, hep-th/0002208
[41]	C.V. Johnson, D-brane primer, https://doi.org/10.1142/9789812799630_0002_hep-th/0007170
[42]	J. Gaillard, D. Martelli, C. Nunez, I. Papadimitriou, The warped, resolved, deformed conifold gets flavoured, Nucl.
	Phys. B 843 (2011) 1, https://doi.org/10.1016/j.nuclphysb.2010.09.011, 1004.4638.
[43]	J. Maldacena, D. Martelli, The unwarped, resolved, deformed conifold: fivebranes and the baryonic branch of the
	Klebanov–Strassler theory, J. High Energy Phys. 1001 (2010) 104, https://doi.org/10.1007/JHEP01(2010)104, 0906.
[44]	U091. N Itzhaki A A Teautin S Vankielowicz Supergravity solutions for branes localized within branes Days Latt D
[44]	432 (1998) 298 https://doi.org/10.1016/S0370-2693(98)00670-4. hep-th/9803103
[45]	X.C. de la Ossa, F. Quevedo, Duality symmetries from nonAbelian isometries in string theory, Nucl. Phys. B 403
	(1993) 377, https://doi.org/10.1016/0550-3213(93)90041-M, hep-th/9210021.
[46]	K. Sfetsos, D.C. Thompson, On non-abelian T-dual geometries with Ramond fluxes, Nucl. Phys. B 846 (2011) 21,
	https://doi.org/10.1016/j.nuclphysb.2010.12.013, 1012.1320.
1471	G. Itsios, C. Nunez, K. Stetsos, D.C. Thompson, Non-Abelian T-duality and the AdS/CFT correspondence: new

ARTICLE IN PRESS

 JID:NUPHB
 AID:14363 /FLA
 [m1+; v1.285; Pm:6/06/2018; 11:31] P.49 (1-49)

 N.T. Macpherson et al. / Nuclear Physics B ••• (••••) •••-•••
 49

[48] Ö. Kelekci, Y. Lozano, N.T. Macpherson, E.Ó. Colgáin, Supersymmetry and non-Abelian T-duality in type II super-gravity, Class. Quantum Gravity 32 (3) (2015) 035014, https://doi.org/10.1088/0264-9381/32/3/035014, 1409.7406. [49] Y. Imamura, 1/4 BPS solutions in massive IIA supergravity, Prog. Theor. Phys. 106 (2001) 653, https://doi.org/10. 1143/PTP.106.653, hep-th/0105263. [50] H. Lin, J.M. Maldacena, Fivebranes from gauge theory, Phys. Rev. D 74 (2006) 084014, https://doi.org/10.1103/ PhysRevD.74.084014, hep-th/0509235. [51] H. Lin, O. Lunin, J.M. Maldacena, Bubbling AdS space and 1/2 BPS geometries, J. High Energy Phys. 0410 (2004) 025, https://doi.org/10.1088/1126-6708/2004/10/025, hep-th/0409174. [52] Y. Lozano, N.T. Macpherson, J. Montero, C. Nunez, Three-dimensional $\mathcal{N} = 4$ linear quivers and non-Abelian T-duals, J. High Energy Phys. 1611 (2016) 133, https://doi.org/10.1007/JHEP11(2016)133, 1609.09061. [53] P. Koerber, D. Tsimpis, Supersymmetric sources, integrability and generalized-structure compactifications, J. High Energy Phys. 0708 (2007) 082, https://doi.org/10.1088/1126-6708/2007/08/082, 0706.1244. [54] T. Friedrich, I. Kath, A. Moroianu, U. Semmelmann, On nearly parallel G(2) structures, SFB-288-162, hal-[55] P. Kaste, R. Minasian, A. Tomasiello, Supersymmetric M theory compactifications with fluxes on seven-manifolds and G structures, J. High Energy Phys. 0307 (2003) 004, https://doi.org/10.1088/1126-6708/2003/07/004, hep-th/ 0303127. [56] T. Friedrich, S. Ivanov, Parallel spinors and connections with skew symmetric torsion in string theory, Asian J. Math. 6 (2002) 303, math.dg/0102142. [57] T. Friedrich, S. Ivanov, Killing spinor equations in dimension 7 and geometry of integrable G(2) manifolds, J. Geom. Phys. 48 (2003) 1, https://doi.org/10.1016/S0393-0440(03)00005-6, math.dg/0112201. [58] J. Gaillard, D. Martelli, Fivebranes and resolved deformed G_2 manifolds, J. High Energy Phys. 1105 (2011) 109, https://doi.org/10.1007/JHEP05(2011)109.1008.0640. [59] A. Bilal, J.P. Derendinger, K. Sfetsos, (Weak) g(2) holonomy from selfduality, flux and supersymmetry, Nucl. Phys. B 628 (2002) 112, https://doi.org/10.1016/S0550-3213(02)00042-1, hep-th/0111274. [60] D. Lust, P. Patalong, D. Tsimpis, Generalized geometry, calibrations and supersymmetry in diverse dimensions, J. High Energy Phys. 1101 (2011) 063, https://doi.org/10.1007/JHEP01(2011)063, 1010.5789. F. Apruzzi, M. Fazzi, A. Passias, D. Rosa, A. Tomasiello, AdS₆ solutions of type II supergravity, J. High Energy Phys. 1411 (2014) 099, https://doi.org/10.1007/JHEP11(2014)099, arXiv:1406.0852; Erratum: J. High Energy Phys. 1505 (2015) 012, https://doi.org/10.1007/JHEP05(2015)012. [62] K. Becker, M. Becker, M theory on eight manifolds, Nucl. Phys. B 477 (1996) 155, https://doi.org/10.1016/0550-3213(96)00367-7, hep-th/9605053. [63] K. Dasgupta, G. Rajesh, S. Sethi, M theory, orientifolds and G-flux, J. High Energy Phys. 9908 (1999) 023, https:// doi.org/10.1088/1126-6708/1999/08/023, hep-th/9908088. [64] K. Becker, A note on compactifications on spin(7)-holonomy manifolds, J. High Energy Phys. 0105 (2001) 003, https://doi.org/10.1088/1126-6708/2001/05/003, hep-th/0011114. [65] D. Martelli, J. Sparks, G structures, fluxes and calibrations in M theory, Phys. Rev. D 68 (2003) 085014, https:// doi.org/10.1103/PhysRevD.68.085014, hep-th/0306225. [66] D. Tsimpis, M-theory on eight-manifolds revisited: N=1 supersymmetry and generalized spin(7) structures, J. High Energy Phys. 0604 (2006) 027, https://doi.org/10.1088/1126-6708/2006/04/027, hep-th/0511047. [67] D. Prins, D. Tsimpis, Type IIA supergravity and M -theory on manifolds with SU(4) structure, Phys. Rev. D 89 (2014) 064030, https://doi.org/10.1103/PhysRevD.89.064030, 1312.1692. [68] F. Apruzzi, M. Fazzi, A. Passias, A. Tomasiello, Supersymmetric AdS₅ solutions of massive IIA supergravity, J. High Energy Phys. 1506 (2015) 195, https://doi.org/10.1007/JHEP06(2015)195, 1502.06620. [69] D. Conti, S. Salamon, Generalized Killing spinors in dimension 5, Trans. Am. Math. Soc. 359 (11) (2007) 5319-5343, math.dg/0508375. [70] M. Fernandez, A. Tomassini, L. Ugarte, R. Villacampa, Balanced Hermitian metrics from SU(2)-structures, J. Math. Phys. 50 (2009) 033507, https://doi.org/10.1063/1.3086834, 0808.1201. [71] N.T. Macpherson, J. Montero, D. Prins, to appear.

4. Results

In this thesis we have studied a number of AdS solutions to type II supergravity, obtained applying NATD to some relevant backgrounds, in addition to classifying certain Mink₃ solutions. For the newly generated AdS geometries, we systematically extracted information from the supergravity fields, intending to provide some basic hints about the putative dual SCFT. This included the derivation of the quantised charges and holographic central charge¹ or free energy.

In the examples considered, a non-flat Kalb-Ramond B_2 field was generated, carrying NS5-brane charge along non-trivial 2-cycles in the new background. In some cases, the 3-sphere on which NATD was applied shrank to a point in the seed geometry, which was then mapped to a singularity in the dual background. This was interpreted as localized NS5-brane sources putting an end to the space, with the NS-NS sector behaving as that of a smeared NS5 near-horizon geometry close to the singularity, as happens also under ATD.

Advancing along the non-compact direction ρ arising from NATD required applying large gauge transformations repeatedly to the Kalb-Ramond field B_2 , as we moved from a certain interval in this direction to the next². Each LGT has the effect of inducing one unit of NS5-brane charge and also shifting the Page charges associated to the RR-fluxes, which are the ones being quantized in the presence of B_2 . In particular, if the original background had colour D(p-1)-branes, in the new solution we typically found colour Dp- and D(p+2)-branes wrapping the same directions, up to an extra two-sphere for the higher-dimensional branes. Their charges were related as $N_p = nN_{p+2}$, where n is the number of LGTs applied. Indeed, these two different kind of branes are thought to be associated to the same degrees

¹The prescriptions used are explained in section 4.2 of [42] and the references therein.

²Large gauge transformations are introduced in order to enforce the $0 \leq \left| \int_{\Sigma_2} B_2 \right| \leq 4\pi^2 \alpha'$ condition, where Σ_2 is any non-trivial two cycle for which the holonomy of B_2 is non-vanishing. This is a requirement for the partition function of an Euclidean string winding around Σ_2 to be well-defined.

of freedom, in the sense that the charge of the Dp-branes are dissolved in D(p+2)-branes. For this reason, we picked up only the lower-dimensional realization of the colour branes for the holographic interpretation of the new backgrounds.

As a result, brane set-ups read-off from these charges were not only infinite³ but also presented an increasing number of colour branes as we move along ρ , realizing an infinite linear quiver with nodes of increasing rank.

Precisely, it was for the IIA $\mathcal{N} = 1 \text{ AdS}_5$ and the IIB $\mathcal{N} = 4 \text{ AdS}_4$ solutions, that we also managed to regularize the brane set-ups by an adequate introduction of flavour branes. We could therefore propose consistent, finite dual quiver field theories.

In particular, for the background obtained in [13, 12] applying NATD on $S^3 \hookrightarrow T^{1,1} \to S^2$ of the $\operatorname{AdS}_5 \times T^{1,1}$ gravity dual of the Klebanov-Witten theory [36], we obtained in [24] (section 3.5) a D4-NS5-NS5' brane set-up, with NS5 and NS5'-branes mutually transversal and also perpendicular to the D4-branes, whose number augments every time an NS5 is crossed. This generalises [72], where the number of D4's is constant, but is still expected to realize a mass deformation of some $\mathcal{N} = 2$ theory realized by a D4-NS5 system with all NS5-branes parallel among them⁴.

The regularisation of this brane set-up and the corresponding linear quiver was achieved through the addition of flavour D6-branes, but only after a specific brane re-ordering, compatible with the fixed point gravity dual. This was necessary for the quiver description to be compatible with a mass deformation⁵, taking as UV theory the $\mathcal{N} = 2$ quiver theories of [25]. We relied on the exact field theoretical central charges computed via a-maximization at both UV and IR fixed points [73], which were related by the Tachikawa-Wecht relations [74], as expected for an RG flow triggered by a mass deformation. By construction, the famous 27/32 ratio was recovered in the holographic limit (this is implied in the Tachikawa-Wecht result), when also the field theoretical central charges coincided with their holographic counterparts⁶.

³The non-compact $\rho \in \mathbf{R}^+$ direction realizes a new field-theory direction along which the colour branes are extended. Not bounding ρ therefore implies an infinite brane set-up.

⁴The mass deformation is realized adding mass terms to the superpotential of the $\mathcal{N} = 2$ UV theory, triggering an RG-flow to a $\mathcal{N} = 1$ IR fixed-point. In the dual brane picture, one starts with a D4-NS5 intersection with at least two mutually-parallel 5-branes orthogonal to the D4-branes, and rotates one of the NS5's w.r.t. the other (the angle of rotation being related to the mass-deformation parameter) until they end up being perpendicular between them and also to the D4-branes [37].

⁵In particular, a hard cut-off for the brane set-up (introducing flavours at fixed ρ position) was not compatible with a $\mathcal{N} = 1$ theory with a quartic superpotential, as expected for the mass deformation of the cubic superpotential of the $\mathcal{N} = 2$ theory.

⁶The holographic constrain $a \approx c$ for the 4D central charges holds in the long quiver

It is worth stressing that conjecturing the supergravity background found directly after the application of NATD to be related to a D4-NS5-NS5' brane set-up, allowed us to propose a consistent dual field theory, even in this $\mathcal{N} = 1$ scenario with less (super)isometries. It was indeed the relation with $\mathcal{N} = 2$ Gaiotto-Maldacena geometries, exploited through the $\mathcal{N} = 2$ theories proposed in [25], that provided the information we needed for the proposal of the $\mathcal{N} = 1$ field theory.

Regarding the previously mentioned IIB $\mathcal{N} = 4$ AdS₄ solution, this was obtained in [23] (section 3.4) applying NATD on the 3-sphere of the AdS₄ × S³ × S² background arising as the near-horizon limit of a D2-D6 intersection. Supersymmetry was fully preserved by the transformation in this case⁷. The new AdS₄×S²×S² geometry fitted in the classification of [32], and the Page charges were compatible with a D3-D5-NS5 Hanany-Witten brane set-up. The holographic dictionary of [34] eased the interpretation of the worldvolume theory of the former brane configuration as a three-dimensional Gaitto-Witten $T_{\rho}^{\hat{\rho}}(SU(N))$ theory, the strong coupled fixed-point of which would yield the $\mathcal{N} = 4$ SCFT dual. The regularisation of the infinite array of branes was achieved in this case as a hard cut-off by the addition of a certain number of semi-inifinite D5-branes, acting as flavours, in such a way that the linking numbers (read off from the brane set-up as described in [34]) were consistent with the conditions for the field theory to flow to a non-trivial fixed point in the IR.

Still considering AdS₄ spaces, a new $\mathcal{N} = 2$ solution in 11D supergravity supported by magnetic G_4 flux was discovered in [43] (section 3.2) applying a chain of a NATD followed by an ATD to the $\mathcal{N} = 6$ AdS₄ × \mathbb{CP}^3 background, the IIA reduction of the M-theory gravity dual of ABJM [58], followed by uplitf to 11D. The new solution is found to be the only explicit example of this class and holographic candidate for the 3d-3d duality, besides the uplift of the Pernici-Sezgin solution. We remark that, unlike the aforementioned cases, supersymmetry is not fully preserved, but reduced from $\mathcal{N} = 6$ to $\mathcal{N} = 2$ because of the NATD transformation⁸. The new background possesses a $U(1) \approx SO(2)$ isometry interpreted as the geometrical realization of the Rsymmetry of the dual SCFT. Indeed, it rotates the pure spinors inducing a U(1)-worth of the corresponding G-structure in $\mathcal{N} = 1$ language.

limit, i.e. for a large number of nodes or, equivalently, NS5-branes [24] (section 3.5).

⁷The reduction to IIA of the $AdS_4 \times S^7/\mathbf{Z}_k$ solution already reduced supersymmetry from $\mathcal{N} = 8$ to $\mathcal{N} = 4$. The resulting spinors were independent of the NATD-directions, so that no further supersymmetry was broken by the transformation.

⁸It was possible to apply the second transformation, in this case an ATD, on an isometric circle of the intermediate background found in [68], under which the Killing spinors were uncharged, thus preserving the remaining supersymmetry.

This novel $\mathcal{N} = 2 \text{ AdS}_4$ solution belongs to the class of [45], which describes $\mathcal{N} = 2 \text{ AdS}_4$ backgrounds with magnetic flux in 11D supergravity whose Killing spinors satisfy the same projection conditions as M5-branes wrapping calibrated 3-cycles (a more general class of AdS₄ spaces was engineered in [46], where electric flux was also allowed). The classification could therefore include also AdS₄ geometries which are not identifiable as nearhorizon limits of wrapped M5-branes. This could be the case for our $\mathcal{N} = 2$ AdS₄ solution, as the holographic free energy does not exhibit the expected N^3 scaling, but the $N^{3/2}$ inherited from ABJM⁹.

Regarding $\operatorname{AdS}_3 \times S^2$ geometries preserving $\mathcal{N} = (0, 4)$ supersymmetry, an extension of the existing classification [55, 45] was motivated by the solution found in [54] (section 3.1) applying NATD on one of the two 3-spheres of the type IIB realization of the $\operatorname{AdS}_3 \times S^3 \times S^3 \times S^1$ background, supported by D1 and D5-brane fluxes [75, 76]. Even if the new background belongs to massive IIA, after Abelian T-dualizing twice (on different U(1) isometries), a massless IIA solution could be reached, which allowed for and uplift to 11D supergravity, providing a solution lying outside the Kim class [55].

Taking as an Ansatz 1/4-BPS $SO(2,2) \times SO(3)$ invariant solutions in 11D supergravity, with the internal spinors realizing an SU(2)-structure, three classes were identified in [56] (section 3.3): the six-dimensional internal manifold either has an SU(3)-holonomy, i.e. Calabi-Yau, or corresponds to an SU(2)-structure manifold with an emergent U(1) or SU(2) isometry. The first case belongs to the Kim class, included in [45], with magnetic G_4 flux only along the S^2 and where the extra U(1) isometry does not correspond to an R-symmetry but to the M-theory circle. The latter case represents the extension of the classification, allowing for a general, but still magnetic G_4 flux supporting an $AdS_3 \times S^2 \times S^2 \times CY_2$ geometry. In this case, the extra SU(2) isometry realizes an emergent R-symmetry, corresponding to a large superconformal algebra [53]. This is in agreement with the holographic central charge, whose scaling with the quantized charges (or levels of the affine SU(2) algebras) does not change under (non-)Abelian T-dualities, nor upon uplift to 11D. A solution within this novel class was found in [54] applying a chain of Abelian T-dualities to the type IIB $AdS_3 \times S^3 \times S^3 \times S^1$. It is worth highlighting that the central charge scales like $c \sim N^2$ instead of N^3 as expected for M5-brane geometries. Furthermore, this solution was not considered in the wrapped M5-brane geometries of [45]. These facts point to an unclear M5-brane origin, similarly to the case of the $\mathcal{N} = 2 \text{ AdS}_4$ solution.

Remark however that our extension of the Kim class still lacks the gener-

⁹NATD does not seem to alter the scaling with the number of colour branes N of holographic quantities computed from the volume of the internal space.

ality required to accommodate the uplift to 11D of the non-Abelian T-dual of $AdS_3 \times S^3 \times S^3 \times S^1$, also found in [54], which does include both electric and magnetic components for the G_4 flux.

Concerning the aforementioned issues arising from the discrepancy between the expected and observed scaling for the central charges/free energies with the quantized charges, it was observed that these holographic observables depend on the volume of the internal space. Therefore, the way the regularization is imposed to the NATD-generated background may influence drastically their scaling behaviour. Most typically, a hard cutoff was applied upon computation of such an observable c, with the following effect: if the scaling in the original background was like $c \sim N_p^q$ with the N_p the number of colour Dp-branes, then for the new geometry it exhibited the behaviour $c \sim n^3 N_{p+2}^q \sim n^{3-q} N_p^q$. Recall that $N_p = n N_{p+2}$ with n the number of NS5-branes, due to the Myers dielectric effect relating the Dp and D(p+2) colour branes, as explained at the beginning of this section.

However, a regularization for the geometry other than a hard cut-off may be necessary in order to get a consistent dual field theory. An explicit example of this was given for the new $\mathcal{N} = 4 \text{ AdS}_4$ background in [23] (section 3.4). Getting the holographic free-energy into the form of the field-theoretical computation of [77], required quite a non-trivial (though physically motivated) continuation of the local geometry obtained directly from NATD.

Let us remark that this regularization-scheme dependence might be causing also the discrepancy of the holographic observables of the new $\mathcal{N} = (0, 4)$ $\mathrm{AdS}_3 \times S^2$ and $\mathcal{N} = 2$ AdS_4 solutions w.r.t. the N^3 behaviour expected for geometries arising from wrapped M5-branes [45]. On the other hand, for $c \sim n^3 N^q$ with *n* roughly giving the number of NS5-branes, one might still wonder about the possibility of the corresponding geometry arising as a near-horizon limit of some configuration of wrapped M5-branes.

The most general class of solutions considered in this thesis were type II $\mathcal{N} > 1$ backgrounds on Mink₃ × S^3 , for which a classification was initiated in [67] (section 3.6). Several backgrounds arising as near-horizons of well-known brane intersections were recovered in both type IIA and IIB, assuming the two internal spinors to be of equal norm (what is a requirement for AdS spaces), apart from pure NS backgrounds found when setting one of the internal spinors to vanish.

Our main result is that the only type II $\operatorname{AdS}_4 \times S^3$ compact solution is the $\mathcal{N} = 4$ AdS_4 obtained from the near-horizon limit of intersecting D2–D6 branes (also as a IIA reduction of $\operatorname{AdS}_4 \times S^7/\mathbb{Z}_k$ preserving $\mathcal{N} = 4$ supersymmetry). Besides, we report on a novel IIA class of $\mathcal{N} = 4$ solutions on $\operatorname{Mink}_3 \times S^2 \times S^3$ preserving an SO(4) R-symmetry (with no emergent AdS_4 factor), with *a priori* all fluxes turned on. Assuming a further U(1) isometry in this new class, ATD can be applied on this circle to get a new IIB background with $\operatorname{Mink}_3 \times S^3 \times S^3_{sq}$, where the squashed 3-sphere admits in general an $SU(2) \times U(1)$ isometry. For the unsquashed case, this solution is connected by S-duality to a new pure NS background on $\operatorname{Mink}_3 \times S^3 \times S^3$. These new solutions were found separately in our classification.

5. Conclusions

Throughout the papers in which this thesis is based, we have investigated the power of non-Abelian T-duality as a solution generating technique in type II supergravity, exploring the AdS/CFT correspondence for the new solutions. These were also used to either probe known classifications, providing explicit geometries, or challenge their generality, motivating extensions thereof.

We focused on NATD applied on SU(2) isometries acting without isotropy in the target space. One of the main strengths of this incarnation of the procedure is that new solutions generated this way may still preserve some supersymmetry, if not all. On the other hand, the NATD transformation produces by construction non-compact spaces even if the isometry group is compact: it exchanges the coordinates associated to this isometry group, which realize a compact manifold¹ in target space, for Lagrange multipliers living in the Lie algebra of the group.

Given that a well-defined SCFT cannot be dual to a geometry with an infinite internal space, trying to assign such a theory to an AdS solution generated with NATD reduces to the question of whether a consistent, physicallymotivated regularization can be found for this new, uncompact geometry. We partially answered this question with the particular proposals of [23, 24], see sections 3.4 and 3.5, based on the seminal work of [25].

One of the most important open questions concerning the solution generating technique aspect of NATD in the context of holography is the effect of the transformation on the field theory side. Only systematics known to date is the transformation rule for the RR-fluxes presented in [6]. Already the identification of the relevant d.o.f.s in the putative dual theory is far from straightforward, due to e.g. the scaling behaviour of the central charge/free energy depending on how the regularization of the geometry is performed. An extension of the results so far on quiver completions will probably be necessary before some general pattern might be derived.

In this direction, a straightforward generalization of our results concerning

 $^{^1\}mathrm{In}$ this context, "compact manifold" is used to denote a manifold with a finite volume integral.
the compatibility of mass deformations and NATD might be achieved. We will study the effect of the latter on the \mathbb{CP}^1 -worth of fixed points that may be reached from the $\mathcal{N} = 2 \text{ AdS}_5 \times S^5/\mathbb{Z}_2$ solution applying different mass deformations. This "moduli space" of fixed points include as particular cases a version of the Pilch-Warner solution [78], gravity dual to the Leigh-Strassler CFT [79], apart from the Klebanov-Witten background studied in this thesis.

Large gauge transformations played a central role in the holographic interpretation of the new gravity backgrounds. They were responsible for the rise in the number of colour and flavour branes at each stack of the brane set-up, introducing also almost all NS5-branes separating these stacks.

Abelian T-duality arises for large values of the radial ρ direction generated with NATD. In this large- ρ limit - equivalent to a large number n of LGTs for a background generated with NATD we recover the background generated with ATD from the same seed solution, as explained in appendix B of [23], included in section 3.4. In this limit, the NATD central charge/free energy differs from its ATD counterpart by $\mathcal{O}(1/n)$ "corrections". In the spirit of [57], the linear quivers generated with NATD could be seen as some "open up" limit of the circular quivers arising with ATD. This connection has been made clearer for the NATD and ATD backgrounds obtained from the $\mathcal{N} = 4$ near-horizon limit of the D2-D6 system. It was shown in [80] that both geometries could be seen as arising in different limits of the same circular quiver construction. It would be interesting to see if this result underlies a more general phenomenon, extensible to AdS/CFT pairs in other dimensions.

We remarked in the previous section that the classification of $\mathcal{N} = (0, 4)$ AdS₃ × S² geometries achieved in [56] (section 3.3) still could not accommodate the uplift of the non-Abelian T-dual of the AdS₃ × S³ × S¹ solution. We require the internal spinors to realize an identity structure, which is beyond our SU(2)-structure Ansatz, in order to be able to reproduce both electric and magnetic components for the G_4 flux. A relevant question is whether geometries described by this would-be classification could realize some M2-M5 brane intersection unknown to date. Furthermore, the possible M5-brane origin of new geometries with purely magnetic flux remains unclear. A better understanding thereof could allow for the exploitation of the anomaly inflow [81] to calculate corrections to the entropy of the putative black holes, which could then be compared to results from other methods.

Finally, explicit solutions to the IIA class of $\mathcal{N} = 4$ Mink₃ × $S^3 \times S^2$ backgrounds were left to be found in [67] (section 3.6), including, through U-dualities, type IIB and pure NS realizations. There are both massive and massless IIA candidate solutions compatible with supersymmetry, which look very promising both for finding Mink₃ solutions with compact internal space, but also possibly geometries that asymptote to AdS.

Conclusiones

A lo largo de los artículos en los que se basa esta tesis, hemos investigado el potencial de T-dualidad no abeliana (NATD) como técnica para generar soluciones en supergravedad de tipo II, explorando la correspondencia AdS/CFT para las nuevas soluciones. Éstas también fueron empleadas para sondar las clasificaciones conocidas, proporcionando geometrías explícitas, o poniendo a prueba su generalidad, motivando así la extensión de las mismas.

Nos centramos en NATD aplicada a isometrías SU(2) actuando sin isotropía en el espacio ambiente. Una de las principales ventajas de esta encarnación del procedimiento es que las nuevas soluciones así generadas pueden preservar algo de supersimetría, si no toda. Por otro lado, la transformación NATD produce espacios no compactos por construcción, incluso si el grupo de isometría es compacto: intercambia las coordenadas asociadas al grupo de isometría, las cuales dan lugar a una variedad compacta² en el espacio ambiente, por los multiplicadores de Lagrange que son elementos del álgebra del grupo.

Dado que una SCFT bien definida no puede ser dual a una geometría con un espacio interno infinito, la cuestión de intentar asignar una teoría de este tipo a una solución AdS generada con NATD se reduce a si es posible encontrar una regularización consistente y físicamente motivada para esta nueva geometría no compacta. Respondimos, al menos parcialmente, a esta cuestión con las propuestas particulares hechas en [23, 24], véase las secciones 3.4 y 3.5, basándonos en el trabajo seminal de [25].

Una de los problemas abiertos de mayor relevancia concerniente a NATD, como técnica para generar soluciones en el contexto de holografía, es el efecto de la transformación en el lado de la teoría de campos. A día de hoy, sólo las reglas de transformación del sector RR, presentadas en [6], se conocen de forma sistemática. Incluso la identificación de los grados de libertad relevantes en la supuesta teoría dual no es en absoluto evidente, debido p.ej. a la dependencia del escaleo de la carga central o la energía libre en la forma

 $^{^2 \}mathrm{En}$ este contexto, por "variedad compacta" que remos decir que su integral de volumen es finita.

en que se regulariza la geometría. Probablemente sea necesario extender los resultados actuales en compleciones de *quivers* para poder llegar a derivar algún patrón general.

En esta dirección, una generalización directa de nuestros resultados en lo concerniente a la compatibilidad de deformaciones masivas y NATD se podría conseguir estudiando el efecto de esta última en el conjunto de puntos fijos, de los que hay tantos como elementos en \mathbb{CP}^1 , que se puede alcanzar aplicando diferentes deformaciones masivas a la solución $\mathcal{N} = 2 \text{ AdS}_5 \times S^5/\mathbb{Z}_2$. Este "espacio de móduli" de puntos fijos incluye como casos particulares una versión de la solución de Pilch-Warner [78], dual gravitatorio de la CFT de Leigh-Strassler [79], aparte del espacio de Klebanov-Witten estudiado en esta tesis.

Las transformaciones de *large gauge* (LGT) han jugado un papel fundamental en la interpretación holográfica de las nuevas soluciones gravitatorias, ya que son responsables del aumento en el número de branas de color (y de sabor, si las hubiera) en cada pila de la configuración de branas, así como de la introducción de casi todas las branas NS5 que separaban las distintas pilas.

T-dualidad abeliana (ATD) surge para valores altos de la dirección radial ρ generada con NATD. En este límite de ρ alto - equivalente a un gran número n de LGT - para una espacio generado con NATD, se recupera el espacio generado con ATD a partir de la misma solución original, tal y como se explica en el apéndice B de [23], incluido en la sección 3.4. En este límite, la carga central o energía libre derivadas de la aplicación de NATD difiere de su opuesta abeliana por "correciones" de orden $\mathcal{O}(1/n)$. En la línea de razonamiento de [57], se puede ver los quivers lineales generados con NATD como un límite en el que los quivers circulares dados por ATD se "abren". Esta conexión se hizo más patente para las soluciones NATD y ATD obtenidas del límite near-horizon $\mathcal{N} = 4$ del sistema D2-D6, mostrándose en [80] que ambas geometrías se podían contemplar como distintos límites de una misma construcción de quiver circular. Sería intersante comprobar si este resultado se debe a un fenómeno mas general, extrapolable a pares AdS/CFT en otras dimensiones.

En la sección anterior señalábamos que la clasificación de geometrías $\mathcal{N} = (0, 4) \operatorname{AdS}_3 \times S^2$, conseguida en [56] (sección 3.3), aún no podía acomodar la elevación a 11D del T-dual no abeliano de la solución $\operatorname{AdS}_3 \times S^3 \times S^3 \times S^1$. Se require que los espinores internos den lugar a una estructura identidad, la cual va más allá de nuestro Ansatz con una estructura SU(2), para poder reproducir tanto una componente eléctrica como una magnética para el flujo G_4 . Una pregunta de interés es si las geometrías descritas por una tal clasificación podrían provenir de alguna intersección de M2-M5 aún desconocida.

De hecho, el posible origen en branas M5 de las nuevas geometrías con flujo puramente magnético sigue sin estar claro. Una mejor comprensión del mismo podría permitir que se aprovechase la afluencia de anomalías [81] para calcular correciones a la entropía de los presuntos agujeros negros, las cuales se podrían comparar entonces a los resultados obtenidos con otros métodos.

Por último, en [67] (sección 3.6) no se pudieron encontrar soluciones explícitas de la clase IIA de espacios $\mathcal{N} = 4 \operatorname{Mink}_3 \times S^3 \times S^2$, que incluye, mediante U-dualidades, versiones de tipo IIB y puramente NS de esas mismas geometrías. Las posibles soluciones, masivas y no masivas de tipo IIA, podrían ser compatibles con supersimetría, y resultan muy prometedoras de cara a encontar tanto geometrías Mink₃ con espacio interno compacto, como posiblemente soluciones que asintoten a AdS.

Bibliography

- J. M. Maldacena, The Large N limit of superconformal field theories and supergravity, Int. J. Theor. Phys. 38 (1999) 1113–1133, [hep-th/9711200].
- [2] T. H. Buscher, Path Integral Derivation of Quantum Duality in Nonlinear Sigma Models, Phys. Lett. B201 (1988) 466–472.
- [3] E. Bergshoeff, C. M. Hull and T. Ortin, Duality in the type II superstring effective action, Nucl. Phys. B451 (1995) 547–578, [hep-th/9504081].
- [4] M. Rocek and E. P. Verlinde, Duality, quotients, and currents, Nucl. Phys. B373 (1992) 630-646, [hep-th/9110053].
- [5] X. C. de la Ossa and F. Quevedo, Duality symmetries from nonAbelian isometries in string theory, Nucl. Phys. B403 (1993) 377-394, [hep-th/9210021].
- [6] K. Sfetsos and D. C. Thompson, On non-abelian T-dual geometries with Ramond fluxes, Nucl. Phys. B846 (2011) 21–42, [1012.1320].
- S. F. Hassan, T duality, space-time spinors and RR fields in curved backgrounds, Nucl. Phys. B568 (2000) 145–161, [hep-th/9907152].
- [8] M. Cvetic, H. Lu, C. N. Pope and J. F. Vazquez-Poritz, AdS in warped space-times, Phys. Rev. D62 (2000) 122003, [hep-th/0005246].
- Y. Lozano, E. Ó Colgáin, D. Rodríguez-Gómez and K. Sfetsos, Supersymmetric AdS₆ via T Duality, Phys. Rev. Lett. **110** (2013) 231601, [1212.1043].
- [10] A. Brandhuber and Y. Oz, The D-4 D-8 brane system and five-dimensional fixed points, Phys. Lett. B460 (1999) 307-312, [hep-th/9905148].
- [11] Y. Lozano, E. O. O Colgáin and D. Rodríguez-Gómez, *Hints of 5d Fixed Point Theories from Non-Abelian T-duality*, *JHEP* 05 (2014) 009, [1311.4842].

- [12] G. Itsios, C. Nunez, K. Sfetsos and D. C. Thompson, Non-Abelian T-duality and the AdS/CFT correspondence:new N=1 backgrounds, Nucl. Phys. B873 (2013) 1–64, [1301.6755].
- [13] G. Itsios, C. Nunez, K. Sfetsos and D. C. Thompson, On Non-Abelian T-Duality and new N=1 backgrounds, Phys. Lett. B721 (2013) 342–346, [1212.4840].
- [14] K. Sfetsos and D. C. Thompson, New $\mathcal{N} = 1$ supersymmetric AdS₅ backgrounds in Type IIA supergravity, JHEP **11** (2014) 006, [1408.6545].
- [15] A. Giveon and M. Rocek, On nonAbelian duality, Nucl. Phys. B421 (1994) 173-190, [hep-th/9308154].
- [16] E. Alvarez, L. Alvarez-Gaume, J. L. F. Barbon and Y. Lozano, Some global aspects of duality in string theory, Nucl. Phys. B415 (1994) 71–100, [hep-th/9309039].
- [17] M. J. Duff, H. Lu and C. N. Pope, Supersymmetry without supersymmetry, Phys. Lett. B409 (1997) 136-144, [hep-th/9704186].
- [18] M. J. Duff, H. Lu and C. N. Pope, AdS(5) x S**5 untwisted, Nucl. Phys. B532 (1998) 181-209, [hep-th/9803061].
- [19] Y. Lozano, E. O Colgain, K. Sfetsos and D. C. Thompson, Non-abelian T-duality, Ramond Fields and Coset Geometries, JHEP 06 (2011) 106, [1104.5196].
- [20] E. Gevorgyan and G. Sarkissian, Defects, Non-abelian T-duality, and the Fourier-Mukai transform of the Ramond-Ramond fields, JHEP 03 (2014) 035, [1310.1264].
- [21] O. Kelekci, Y. Lozano, N. T. Macpherson and E. O. Colgain, Supersymmetry and non-Abelian T-duality in type II supergravity, Class. Quant. Grav. 32 (2015) 035014, [1409.7406].
- [22] M. Hong, Y. Kim and E. Ó Colgáin, On non-Abelian T-duality for non-semisimple groups, 1801.09567.
- Y. Lozano, N. T. Macpherson, J. Montero and C. Nunez, *Three-dimensional* N = 4 linear quivers and non-Abelian T-duals, JHEP 11 (2016) 133, [1609.09061].
- [24] G. Itsios, Y. Lozano, J. Montero and C. Nunez, The AdS₅ non-Abelian T-dual of Klebanov-Witten as a $\mathcal{N} = 1$ linear quiver from M5-branes, JHEP **09** (2017) 038, [1705.09661].
- [25] Y. Lozano and C. Núñez, Field theory aspects of non-Abelian T-duality and N = 2 linear quivers, JHEP 05 (2016) 107, [1603.04440].

- [26] Y. Lozano, C. Nunez and S. Zacarias, BMN Vacua, Superstars and Non-Abelian T-duality, JHEP 09 (2017) 000, [1703.00417].
- [27] G. Itsios, C. Nunez and D. Zoakos, Mesons from (non) Abelian T-dual backgrounds, JHEP 01 (2017) 011, [1611.03490].
- [28] J. van Gorsel and S. Zacarías, A Type IIB Matrix Model via non-Abelian T-dualities, JHEP 12 (2017) 101, [1711.03419].
- [29] G. Itsios, H. Nastase, C. Núñez, K. Sfetsos and S. Zacarías, Penrose limits of Abelian and non-Abelian T-duals of AdS₅ × S⁵ and their field theory duals, JHEP **01** (2018) 071, [1711.09911].
- [30] D. Gaiotto and J. Maldacena, The Gravity duals of N=2 superconformal field theories, JHEP 10 (2012) 189, [0904.4466].
- [31] J. M. Maldacena and C. Nunez, Supergravity description of field theories on curved manifolds and a no go theorem, Int. J. Mod. Phys. A16 (2001) 822-855, [hep-th/0007018].
- [32] E. D'Hoker, J. Estes and M. Gutperle, Exact half-BPS Type IIB interface solutions. I. Local solution and supersymmetric Janus, JHEP 06 (2007) 021, [0705.0022].
- [33] E. D'Hoker, J. Estes and M. Gutperle, Exact half-BPS Type IIB interface solutions. II. Flux solutions and multi-Janus, JHEP 06 (2007) 022, [0705.0024].
- [34] B. Assel, C. Bachas, J. Estes and J. Gomis, Holographic Duals of D=3 N=4 Superconformal Field Theories, JHEP 08 (2011) 087, [1106.4253].
- [35] D. Gaiotto and E. Witten, S-Duality of Boundary Conditions In N=4 Super Yang-Mills Theory, Adv. Theor. Math. Phys. 13 (2009) 721–896, [0807.3720].
- [36] I. R. Klebanov and E. Witten, Superconformal field theory on three-branes at a Calabi-Yau singularity, Nucl. Phys. B536 (1998) 199-218, [hep-th/9807080].
- [37] A. M. Uranga, Brane configurations for branes at conifolds, JHEP 01 (1999) 022, [hep-th/9811004].
- [38] M. Grana, R. Minasian, M. Petrini and A. Tomasiello, Generalized structures of N=1 vacua, JHEP 11 (2005) 020, [hep-th/0505212].
- [39] M. Grana, R. Minasian, M. Petrini and A. Tomasiello, A Scan for new N=1 vacua on twisted tori, JHEP 05 (2007) 031, [hep-th/0609124].
- [40] M. Grana, Flux compactifications in string theory: A Comprehensive review, Phys. Rept. 423 (2006) 91–158, [hep-th/0509003].

- [41] A. Barranco, J. Gaillard, N. T. Macpherson, C. Núñez and D. C. Thompson, *G-structures and Flavouring non-Abelian T-duality*, *JHEP* 08 (2013) 018, [1305.7229].
- [42] N. T. Macpherson, C. Núñez, L. A. Pando Zayas, V. G. J. Rodgers and C. A. Whiting, *Type IIB supergravity solutions with AdS₅ from Abelian and non-Abelian T dualities*, *JHEP* 02 (2015) 040, [1410.2650].
- [43] Y. Lozano, N. T. Macpherson and J. Montero, $A \mathcal{N} = 2$ supersymmetric AdS_4 solution in M-theory with purely magnetic flux, JHEP **10** (2015) 004, [1507.02660].
- [44] N. T. Macpherson, Non-Abelian T-duality, G₂-structure rotation and holographic duals of N = 1 Chern-Simons theories, JHEP **11** (2013) 137, [1310.1609].
- [45] J. P. Gauntlett, O. A. P. Mac Conamhna, T. Mateos and D. Waldram, AdS spacetimes from wrapped M5 branes, JHEP 11 (2006) 053, [hep-th/0605146].
- [46] M. Gabella, D. Martelli, A. Passias and J. Sparks, $\mathcal{N} = 2$ supersymmetric AdS₄ solutions of M-theory, Commun. Math. Phys. **325** (2014) 487–525, [1207.3082].
- [47] B. S. Acharya, J. P. Gauntlett and N. Kim, Five-branes wrapped on associative three cycles, Phys. Rev. D63 (2001) 106003, [hep-th/0011190].
- [48] J. P. Gauntlett, N. Kim and D. Waldram, M Five-branes wrapped on supersymmetric cycles, Phys. Rev. D63 (2001) 126001, [hep-th/0012195].
- [49] M. Pernici and E. Sezgin, Spontaneous Compactification of Seven-dimensional Supergravity Theories, Class. Quant. Grav. 2 (1985) 673.
- [50] J. M. Maldacena, A. Strominger and E. Witten, Black hole entropy in M theory, JHEP 12 (1997) 002, [hep-th/9711053].
- [51] A. Strominger and C. Vafa, Microscopic origin of the Bekenstein-Hawking entropy, Phys. Lett. B379 (1996) 99–104, [hep-th/9601029].
- [52] A. Strominger, Black hole entropy from near horizon microstates, JHEP 02 (1998) 009, [hep-th/9712251].
- [53] A. Ali, Classification of two-dimensional N=4 superconformal symmetries, hep-th/9906096.
- [54] Y. Lozano, N. T. Macpherson, J. Montero and E. O. Colgáin, New $AdS_3 \times S^2$ T-duals with $\mathcal{N} = (0, 4)$ supersymmetry, JHEP **08** (2015) 121, [1507.02659].

- [55] H. Kim, K. K. Kim and N. Kim, 1/4-BPS M-theory bubbles with SO(3) x SO(4) symmetry, JHEP 08 (2007) 050, [0706.2042].
- [56] O. Kelekci, Y. Lozano, J. Montero, E. O. Colgáin and M. Park, Large superconformal near-horizons from M-theory, Phys. Rev. D93 (2016) 086010, [1602.02802].
- [57] B. Assel, C. Bachas, J. Estes and J. Gomis, *IIB Duals of D=3 N=4 Circular Quivers*, *JHEP* **12** (2012) 044, [1210.2590].
- [58] O. Aharony, O. Bergman, D. L. Jafferis and J. Maldacena, N=6 superconformal Chern-Simons-matter theories, M2-branes and their gravity duals, JHEP 10 (2008) 091, [0806.1218].
- [59] J. P. Gauntlett, D. Martelli and D. Waldram, Superstrings with intrinsic torsion, Phys. Rev. D69 (2004) 086002, [hep-th/0302158].
- [60] J. P. Gauntlett, N. Kim, D. Martelli and D. Waldram, Five-branes wrapped on SLAG three cycles and related geometry, JHEP 11 (2001) 018, [hep-th/0110034].
- [61] M. Haack, D. Lust, L. Martucci and A. Tomasiello, Domain walls from ten dimensions, JHEP 10 (2009) 089, [0905.1582].
- [62] D. Andriot, New supersymmetric vacua on solvmanifolds, JHEP 02 (2016) 112, [1507.00014].
- [63] D. Andriot, J. Blaback and T. Van Riet, Minkowski flux vacua of type II supergravities, Phys. Rev. Lett. 118 (2017) 011603, [1609.00729].
- [64] N. Halmagyi, D. Israel, M. Sarkis and E. E. Svanes, *Heterotic Hyper-Kähler flux backgrounds*, JHEP 08 (2017) 138, [1706.01725].
- [65] D. Andriot, On classical de Sitter and Minkowski solutions with intersecting branes, JHEP 03 (2018) 054, [1710.08886].
- [66] N. T. Macpherson and A. Tomasiello, Minimal flux Minkowski classification, JHEP 09 (2017) 126, [1612.06885].
- [67] N. T. Macpherson, J. Montero and D. Prins, $Mink_3 \times S^3$ solutions of type II supergravity, 1712.00851.
- [68] Y. Lozano and N. T. Macpherson, A new AdS₄/CFT₃ dual with extended SUSY and a spectral flow, JHEP 11 (2014) 115, [1408.0912].
- [69] T. Curtright and C. K. Zachos, Currents, charges, and canonical structure of pseudodual chiral models, Phys. Rev. D49 (1994) 5408-5421, [hep-th/9401006].

- Y. Lozano, NonAbelian duality and canonical transformations, Phys. Lett. B355 (1995) 165–170, [hep-th/9503045].
- [71] E. Bergshoeff, R. Kallosh, T. Ortin, D. Roest and A. Van Proeyen, New formulations of D = 10 supersymmetry and D8 - O8 domain walls, Class. Quant. Grav. 18 (2001) 3359–3382, [hep-th/0103233].
- [72] I. Bah and N. Bobev, Linear quivers and N = 1 SCFTs from M5-branes, JHEP 08 (2014) 121, [1307.7104].
- [73] K. A. Intriligator and B. Wecht, The Exact superconformal R symmetry maximizes a, Nucl. Phys. B667 (2003) 183–200, [hep-th/0304128].
- [74] Y. Tachikawa and B. Wecht, Explanation of the Central Charge Ratio 27/32 in Four-Dimensional Renormalization Group Flows between Superconformal Theories, Phys. Rev. Lett. 103 (2009) 061601, [0906.0965].
- [75] J. de Boer, A. Pasquinucci and K. Skenderis, AdS / CFT dualities involving large 2-D N=4 superconformal symmetry, Adv. Theor. Math. Phys. 3 (1999) 577-614, [hep-th/9904073].
- [76] D. Tong, The holographic dual of $AdS_3 \times S^3 \times S^3 \times S^1$, JHEP **04** (2014) 193, [1402.5135].
- [77] B. Assel, J. Estes and M. Yamazaki, *Large N Free Energy of 3d N=4* SCFTs and AdS₄/CFT₃, JHEP **09** (2012) 074, [1206.2920].
- [78] K. Pilch and N. P. Warner, N=1 supersymmetric renormalization group flows from IIB supergravity, Adv. Theor. Math. Phys. 4 (2002) 627-677, [hep-th/0006066].
- [79] R. G. Leigh and M. J. Strassler, Exactly marginal operators and duality in four-dimensional N=1 supersymmetric gauge theory, Nucl. Phys. B447 (1995) 95-136, [hep-th/9503121].
- [80] B. Assel and A. Tomasiello, *Holographic duals of 3d S-fold CFTs*, *JHEP* 06 (2018) 019, [1804.06419].
- [81] F. Benini and N. Bobev, Two-dimensional SCFTs from wrapped branes and c-extremization, JHEP 06 (2013) 005, [1302.4451].
- [82] H. Lin, O. Lunin and J. M. Maldacena, Bubbling AdS space and 1/2 BPS geometries, JHEP 10 (2004) 025, [hep-th/0409174].
- [83] N. Gaddam, A. Gnecchi, S. Vandoren and O. Varela, *Rholography, Black Holes and Scherk-Schwarz, JHEP* 06 (2015) 058, [1412.7325].
- [84] D. Gaiotto, N=2 dualities, JHEP 08 (2012) 034, [0904.2715].
- [85] D. Joyce, Lectures on special Lagrangian geometry, math/0111111.

- [86] A. Fayyazuddin and D. J. Smith, Localized intersections of M5-branes and four-dimensional superconformal field theories, JHEP 04 (1999) 030, [hep-th/9902210].
- [87] E. O Colgain and B. Stefanski, Jr., A search for AdS5 X S2 IIB supergravity solutions dual to N = 2 SCFTs, JHEP **10** (2011) 061, [1107.5763].
- [88] R. A. Reid-Edwards and j. Stefanski, B., On Type IIA geometries dual to N = 2 SCFTs, Nucl. Phys. **B849** (2011) 549–572, [1011.0216].
- [89] A. Donos and J. Simon, The electrostatic view on M-theory LLM geometries, JHEP 01 (2011) 067, [1010.3101].
- [90] O. Aharony, L. Berdichevsky and M. Berkooz, 4d N=2 superconformal linear quivers with type IIA duals, JHEP 08 (2012) 131, [1206.5916].
- [91] P. M. Petropoulos, K. Sfetsos and K. Siampos, Gravity duals of $\mathcal{N} = 2$ superconformal field theories with no electrostatic description, JHEP **11** (2013) 118, [1308.6583].
- [92] P. M. Petropoulos, K. Sfetsos and K. Siampos, Gravity duals of N = 2 SCFTs and asymptotic emergence of the electrostatic description, JHEP 09 (2014) 057, [1406.0853].
- [93] O. A. P. Mac Conamhna and E. O Colgain, Supersymmetric wrapped membranes, AdS(2) spaces, and bubbling geometries, JHEP 03 (2007) 115, [hep-th/0612196].
- [94] E. O Colgain, J.-B. Wu and H. Yavartanoo, On the generality of the LLM geometries in M-theory, JHEP 04 (2011) 002, [1010.5982].
- [95] E. O Colgain, Beyond LLM in M-theory, JHEP 12 (2012) 023, [1208.5979].
- [96] T. Ortiz, H. Samtleben and D. Tsimpis, *Matrix model holography*, *JHEP* 12 (2014) 096, [1410.0487].
- [97] E. O Colgain, J.-B. Wu and H. Yavartanoo, Supersymmetric AdS3 X S2 M-theory geometries with fluxes, JHEP 08 (2010) 114, [1005.4527].
- [98] Y. Bea, J. D. Edelstein, G. Itsios, K. S. Kooner, C. Nunez, D. Schofield et al., Compactifications of the Klebanov-Witten CFT and new AdS₃ backgrounds, JHEP 05 (2015) 062, [1503.07527].
- [99] J. Gaillard, N. T. Macpherson, C. Núñez and D. C. Thompson, Dualising the Baryonic Branch: Dynamic SU(2) and confining backgrounds in IIA, Nucl. Phys. B884 (2014) 696–740, [1312.4945].

- [100] E. Caceres, N. T. Macpherson and C. Núñez, New Type IIB Backgrounds and Aspects of Their Field Theory Duals, JHEP 08 (2014) 107, [1402.3294].
- [101] K. S. Kooner and S. Zacarías, Non-Abelian T-Dualizing the Resolved Conifold with Regular and Fractional D3-Branes, JHEP 08 (2015) 143, [1411.7433].
- [102] S. Zacarías, Semiclassical strings and Non-Abelian T-duality, Phys. Lett. B737 (2014) 90–97, [1401.7618].
- [103] P. M. Pradhan, Oscillating Strings and Non-Abelian T-dual Klebanov-Witten Background, Phys. Rev. D90 (2014) 046003, [1406.2152].
- [104] A. Passias, A note on supersymmetric AdS₆ solutions of massive type IIA supergravity, JHEP 01 (2013) 113, [1209.3267].
- [105] F. Apruzzi, M. Fazzi, A. Passias, D. Rosa and A. Tomasiello, AdS₆ solutions of type II supergravity, JHEP **11** (2014) 099, [1406.0852].
- [106] H. Kim, N. Kim and M. Suh, Supersymmetric AdS₆ Solutions of Type IIB Supergravity, Eur. Phys. J. C75 (2015) 484, [1506.05480].
- [107] O. DeWolfe, A. Hanany, A. Iqbal and E. Katz, Five-branes, seven-branes and five-dimensional E(n) field theories, JHEP 03 (1999) 006, [hep-th/9902179].
- [108] T. Dimofte, D. Gaiotto and S. Gukov, Gauge Theories Labelled by Three-Manifolds, Commun. Math. Phys. 325 (2014) 367–419, [1108.4389].
- [109] T. R. Araujo and H. Nastase, $\mathcal{N} = 1$ SUSY backgrounds with an AdS factor from non-Abelian T duality, Phys. Rev. **D91** (2015) 126015, [1503.00553].
- [110] I. Bah, C. Beem, N. Bobev and B. Wecht, AdS/CFT Dual Pairs from M5-Branes on Riemann Surfaces, Phys. Rev. D85 (2012) 121901, [1112.5487].
- [111] I. Bah, C. Beem, N. Bobev and B. Wecht, Four-Dimensional SCFTs from M5-Branes, JHEP 06 (2012) 005, [1203.0303].
- [112] F. Benini, Y. Tachikawa and B. Wecht, Sicilian gauge theories and N=1 dualities, JHEP 01 (2010) 088, [0909.1327].
- [113] J. Murugan and H. Nastase, A nonabelian particle-vortex duality, Phys. Lett. B753 (2016) 401–405, [1506.04090].
- [114] P. Karndumri and E. O. Colgáin, 3D Supergravity from wrapped D3-branes, JHEP 10 (2013) 094, [1307.2086].

- [115] J. P. Gauntlett, D. Martelli, J. Sparks and D. Waldram, Supersymmetric AdS(5) solutions of M theory, Class. Quant. Grav. 21 (2004) 4335–4366, [hep-th/0402153].
- [116] P. M. Cowdall and P. K. Townsend, Gauged supergravity vacua from intersecting branes, Phys. Lett. B429 (1998) 281–288, [hep-th/9801165].
- [117] S. Gukov, E. Martinec, G. W. Moore and A. Strominger, The Search for a holographic dual to AdS(3) x S**3 x S**3 x S**1, Adv. Theor. Math. Phys. 9 (2005) 435-525, [hep-th/0403090].
- [118] H. J. Boonstra, B. Peeters and K. Skenderis, Brane intersections, anti-de Sitter space-times and dual superconformal theories, Nucl. Phys. B533 (1998) 127-162, [hep-th/9803231].
- [119] J. P. Gauntlett, R. C. Myers and P. K. Townsend, Supersymmetry of rotating branes, Phys. Rev. D59 (1998) 025001, [hep-th/9809065].
- [120] S. Elitzur, O. Feinerman, A. Giveon and D. Tsabar, String theory on AdS(3) x S^{**3} x S^{**3} x S^{**1}, Phys. Lett. B449 (1999) 180-186, [hep-th/9811245].
- [121] J. D. Brown and M. Henneaux, Central Charges in the Canonical Realization of Asymptotic Symmetries: An Example from Three-Dimensional Gravity, Commun. Math. Phys. 104 (1986) 207–226.
- [122] A. Sevrin, W. Troost and A. Van Proeyen, Superconformal Algebras in Two-Dimensions with N=4, Phys. Lett. B208 (1988) 447–450.
- [123] G. Itsios, Y. Lozano, E. O Colgain and K. Sfetsos, Non-Abelian T-duality and consistent truncations in type-II supergravity, JHEP 08 (2012) 132, [1205.2274].
- [124] N. Seiberg, Electric magnetic duality in supersymmetric nonAbelian gauge theories, Nucl. Phys. B435 (1995) 129–146, [hep-th/9411149].
- [125] I. R. Klebanov, D. Kutasov and A. Murugan, Entanglement as a probe of confinement, Nucl. Phys. B796 (2008) 274–293, [0709.2140].
- [126] A. Donos and J. P. Gauntlett, Flowing from AdS_5 to AdS_3 with $T^{1,1}$, JHEP **08** (2014) 006, [1404.7133].
- [127] J. Jeong, O. Kelekci and E. O Colgain, An alternative IIB embedding of F(4) gauged supergravity, JHEP **05** (2013) 079, [1302.2105].
- [128] F. Apruzzi, M. Fazzi, D. Rosa and A. Tomasiello, All AdS₇ solutions of type II supergravity, JHEP 04 (2014) 064, [1309.2949].
- [129] F. Apruzzi, M. Fazzi, A. Passias and A. Tomasiello, Supersymmetric AdS₅ solutions of massive IIA supergravity, JHEP 06 (2015) 195, [1502.06620].

- [130] C. Bachas, E. D'Hoker, J. Estes and D. Krym, *M-theory Solutions Invariant under D*(2, 1; γ) ⊕ D(2, 1; γ), *Fortsch. Phys.* 62 (2014) 207–254, [1312.5477].
- [131] M. J. Duff, H. Lu and C. N. Pope, AdS(3) x S**3 (un)twisted and squashed, and an O(2,2,Z) multiplet of dyonic strings, Nucl. Phys. B544 (1999) 145–180, [hep-th/9807173].
- [132] N. Seiberg, Five-dimensional SUSY field theories, nontrivial fixed points and string dynamics, Phys. Lett. B388 (1996) 753-760, [hep-th/9608111].
- [133] D. R. Morrison and N. Seiberg, Extremal transitions and five-dimensional supersymmetric field theories, Nucl. Phys. B483 (1997) 229-247, [hep-th/9609070].
- [134] K. A. Intriligator, D. R. Morrison and N. Seiberg, Five-dimensional supersymmetric gauge theories and degenerations of Calabi-Yau spaces, Nucl. Phys. B497 (1997) 56-100, [hep-th/9702198].
- [135] O. Bergman and D. Rodriguez-Gomez, 5d quivers and their AdS(6) duals, JHEP 07 (2012) 171, [1206.3503].
- [136] O. Aharony and A. Hanany, Branes, superpotentials and superconformal fixed points, Nucl. Phys. B504 (1997) 239–271, [hep-th/9704170].
- [137] O. Aharony, A. Hanany and B. Kol, Webs of (p,q) five-branes, five-dimensional field theories and grid diagrams, JHEP 01 (1998) 002, [hep-th/9710116].
- [138] A. Rota and A. Tomasiello, AdS_4 compactifications of AdS_7 solutions in type II supergravity, JHEP **07** (2015) 076, [1502.06622].
- [139] D. Gaiotto and A. Tomasiello, Holography for (1,0) theories in six dimensions, JHEP 12 (2014) 003, [1404.0711].
- [140] Y. Terashima and M. Yamazaki, SL(2,R) Chern-Simons, Liouville, and Gauge Theory on Duality Walls, JHEP 08 (2011) 135, [1103.5748].
- [141] F. Benini, F. Canoura, S. Cremonesi, C. Nunez and A. V. Ramallo, Backreacting flavors in the Klebanov-Strassler background, JHEP 09 (2007) 109, [0706.1238].
- [142] C. M. Hull, U duality and BPS spectrum of superYang-Mills theory and M theory, JHEP 07 (1998) 018, [hep-th/9712075].
- [143] N. A. Obers and B. Pioline, U duality and M theory, Phys. Rept. 318 (1999) 113-225, [hep-th/9809039].
- [144] Y. Lozano, Noncommutative branes from M-theory, Phys. Rev. D64 (2001) 106011, [hep-th/0012137].

- [145] E. Bergshoeff, B. Janssen and T. Ortin, Kaluza-Klein monopoles and gauged sigma models, Phys. Lett. B410 (1997) 131-141, [hep-th/9706117].
- [146] E. Bergshoeff, E. Eyras and Y. Lozano, The Massive Kaluza-Klein monopole, Phys. Lett. B430 (1998) 77–86, [hep-th/9802199].
- [147] I. Bah, M. Gabella and N. Halmagyi, BPS M5-branes as Defects for the 3d-3d Correspondence, JHEP 11 (2014) 112, [1407.0403].
- [148] H.-J. Chung, T. Dimofte, S. Gukov and P. Sulkowski, 3d-3d Correspondence Revisited, JHEP 04 (2016) 140, [1405.3663].
- [149] J. P. Gauntlett, D. Martelli, J. Sparks and D. Waldram, Supersymmetric AdS(5) solutions of type IIB supergravity, Class. Quant. Grav. 23 (2006) 4693-4718, [hep-th/0510125].
- [150] D. Andriot, New supersymmetric flux vacua with intermediate SU(2) structure, JHEP 08 (2008) 096, [0804.1769].
- [151] D. Andriot, String theory flux vacua on twisted tori and Generalized Complex Geometry. PhD thesis, Paris, LPTHE, 2010.
- [152] D. Gaiotto and A. Tomasiello, Perturbing gauge/gravity duals by a Romans mass, J. Phys. A42 (2009) 465205, [0904.3959].
- [153] J. D. Bekenstein, Black holes and the second law, Lett. Nuovo Cim. 4 (1972) 737–740.
- [154] J. L. Cardy, Operator Content of Two-Dimensional Conformally Invariant Theories, Nucl. Phys. B270 (1986) 186–204.
- [155] M. Guica, T. Hartman, W. Song and A. Strominger, *The Kerr/CFT Correspondence*, *Phys. Rev.* D80 (2009) 124008, [0809.4266].
- [156] J. F. Donoghue, General relativity as an effective field theory: The leading quantum corrections, Phys. Rev. D50 (1994) 3874–3888, [gr-qc/9405057].
- [157] D. Freed, J. A. Harvey, R. Minasian and G. W. Moore, Gravitational anomaly cancellation for M theory five-branes, Adv. Theor. Math. Phys. 2 (1998) 601–618, [hep-th/9803205].
- [158] M. Ademollo et al., Supersymmetric Strings and Color Confinement, Phys. Lett. 62B (1976) 105–110.
- [159] K. Schoutens, O(n) Extended Superconformal Field Theory in Superspace, Nucl. Phys. B295 (1988) 634–652.
- [160] E. D'Hoker, J. Estes, M. Gutperle, D. Krym and P. Sorba, *Half-BPS supergravity solutions and superalgebras*, *JHEP* 12 (2008) 047, [0810.1484].

- [161] M. R. Gaberdiel and R. Gopakumar, Large N=4 Holography, JHEP 09 (2013) 036, [1305.4181].
- [162] A. Babichenko, B. Stefanski, Jr. and K. Zarembo, Integrability and the AdS(3)/CFT(2) correspondence, JHEP 03 (2010) 058, [0912.1723].
- [163] O. Ohlsson Sax and B. Stefanski, Jr., Integrability, spin-chains and the AdS3/CFT2 correspondence, JHEP 08 (2011) 029, [1106.2558].
- [164] O. Ohlsson Sax, j. Stefanski, Bogdan and A. Torrielli, On the massless modes of the AdS3/CFT2 integrable systems, JHEP 03 (2013) 109, [1211.1952].
- [165] O. Lunin and J. M. Maldacena, Deforming field theories with U(1) x U(1) global symmetry and their gravity duals, JHEP 05 (2005) 033, [hep-th/0502086].
- [166] J. P. Gauntlett, O. A. P. Mac Conamhna, T. Mateos and D. Waldram, New supersymmetric AdS(3) solutions, Phys. Rev. D74 (2006) 106007, [hep-th/0608055].
- [167] I. Bah and V. Stylianou, Gravity duals of N=(0,2) SCFTs from M5-branes, 1508.04135.
- [168] P. Karndumri and E. O. Colgáin, 3D supergravity from wrapped M5-branes, JHEP 03 (2016) 188, [1508.00963].
- [169] C. M. Hull and P. K. Townsend, Unity of superstring dualities, Nucl. Phys. B438 (1995) 109–137, [hep-th/9410167].
- [170] H. Lu, C. N. Pope and K. S. Stelle, M theory / heterotic duality: A Kaluza-Klein perspective, Nucl. Phys. B548 (1999) 87-138, [hep-th/9810159].
- [171] S. W. Beck, J. B. Gutowski and G. Papadopoulos, Geometry and supersymmetry of heterotic warped flux AdS backgrounds, JHEP 07 (2015) 152, [1505.01693].
- [172] J. P. Gauntlett and S. Pakis, The Geometry of D = 11 killing spinors, JHEP 04 (2003) 039, [hep-th/0212008].
- [173] B. Chen, E. O Colgain, J.-B. Wu and H. Yavartanoo, N = 2 SCFTs: An M5-brane perspective, JHEP 04 (2010) 078, [1001.0906].
- [174] K. p. Tod, All Metrics Admitting Supercovariantly Constant Spinors, Phys. Lett. 121B (1983) 241–244.
- [175] S. Chiossi and S. Salamon, The Intrinsic torsion of SU(3) and G(2) structures, in International Conference on Differential Geometry held in honor of the 60th Birthday of A.M. Naveira Valencia, Spain, May 8-14, 2001, 2002. math/0202282.

- [176] P. K. Townsend and P. van Nieuwenhuizen, GAUGED SEVEN-DIMENSIONAL SUPERGRAVITY, Phys. Lett. B125 (1983) 41-46.
- [177] M. Cariglia and O. A. P. Mac Conamhna, Timelike Killing spinors in seven dimensions, Phys. Rev. D70 (2004) 125009, [hep-th/0407127].
- [178] R. Emparan and H. S. Reall, Black Holes in Higher Dimensions, Living Rev. Rel. 11 (2008) 6, [0801.3471].
- [179] C. Montonen and D. I. Olive, Magnetic Monopoles as Gauge Particles?, Phys. Lett. 72B (1977) 117–120.
- [180] N. Seiberg and E. Witten, Electric magnetic duality, monopole condensation, and confinement in N=2 supersymmetric Yang-Mills theory, Nucl. Phys. B426 (1994) 19–52, [hep-th/9407087].
- [181] A. Sen, An Introduction to nonperturbative string theory, in Duality and supersymmetric theories. Proceedings, Easter School, Newton Institute, Euroconference, Cambridge, UK, April 7-18, 1997, pp. 297–413, 1998. hep-th/9802051.
- [182] H. A. Kramers and G. H. Wannier, Statistics of the two-dimensional ferromagnet. Part 1., Phys. Rev. 60 (1941) 252–262.
- [183] S. R. Coleman, The Quantum Sine-Gordon Equation as the Massive Thirring Model, Phys. Rev. D11 (1975) 2088.
- [184] C. Couzens, Supersymmetric AdS₅ solutions of type IIB supergravity without D3 branes, JHEP **01** (2017) 041, [1609.05039].
- [185] E. Alvarez, L. Alvarez-Gaume and Y. Lozano, On nonAbelian duality, Nucl. Phys. B424 (1994) 155–183, [hep-th/9403155].
- [186] S. Elitzur, A. Giveon, E. Rabinovici, A. Schwimmer and G. Veneziano, *Remarks on nonAbelian duality, Nucl. Phys.* B435 (1995) 147–171, [hep-th/9409011].
- [187] C. Klimcik and P. Severa, Dual nonAbelian duality and the Drinfeld double, Phys. Lett. B351 (1995) 455-462, [hep-th/9502122].
- [188] S. Cremonesi and A. Tomasiello, 6d holographic anomaly match as a continuum limit, JHEP 05 (2016) 031, [1512.02225].
- [189] O. Aharony, L. Berdichevsky, M. Berkooz and I. Shamir, Near-horizon solutions for D3-branes ending on 5-branes, Phys. Rev. D84 (2011) 126003, [1106.1870].

- [190] A. Hanany and E. Witten, Type IIB superstrings, BPS monopoles, and three-dimensional gauge dynamics, Nucl. Phys. B492 (1997) 152–190, [hep-th/9611230].
- [191] N. T. Macpherson, C. Nunez, D. C. Thompson and S. Zacarias, *Holographic Flows in non-Abelian T-dual Geometries*, *JHEP* **11** (2015) 212, [1509.04286].
- [192] S. Benvenuti and S. Pasquetti, 3D-partition functions on the sphere: exact evaluation and mirror symmetry, JHEP 05 (2012) 099, [1105.2551].
- [193] T. Nishioka, Y. Tachikawa and M. Yamazaki, 3d Partition Function as Overlap of Wavefunctions, JHEP 08 (2011) 003, [1105.4390].
- [194] B. Assel and J. Gomis, Mirror Symmetry And Loop Operators, JHEP 11 (2015) 055, [1506.01718].
- [195] S. Cremonesi, A. Hanany, N. Mekareeya and A. Zaffaroni, T^{σ}_{ρ} (G) theories and their Hilbert series, JHEP **01** (2015) 150, [1410.1548].
- [196] B. Hoare and A. A. Tseytlin, Homogeneous Yang-Baxter deformations as non-abelian duals of the AdS₅ sigma-model, J. Phys. A49 (2016) 494001, [1609.02550].
- [197] F. Leblond, R. C. Myers and D. C. Page, Superstars and giant gravitons in M theory, JHEP 01 (2002) 026, [hep-th/0111178].
- [198] E. Witten, Solutions of four-dimensional field theories via M theory, Nucl. Phys. B500 (1997) 3-42, [hep-th/9703166].
- [199] I. Bah and B. Wecht, New N=1 Superconformal Field Theories In Four Dimensions, JHEP 07 (2013) 107, [1111.3402].
- [200] K. Dasgupta and S. Mukhi, Brane constructions, conifolds and M theory, Nucl. Phys. B551 (1999) 204–228, [hep-th/9811139].
- [201] S. S. Gubser, Einstein manifolds and conformal field theories, Phys. Rev. D59 (1999) 025006, [hep-th/9807164].
- [202] R. C. Myers, *Dielectric branes*, *JHEP* **12** (1999) 022, [hep-th/9910053].
- [203] D. Anselmi, D. Z. Freedman, M. T. Grisaru and A. A. Johansen, Nonperturbative formulas for central functions of supersymmetric gauge theories, Nucl. Phys. B526 (1998) 543-571, [hep-th/9708042].
- [204] V. A. Novikov, M. A. Shifman, A. I. Vainshtein and V. I. Zakharov, Exact Gell-Mann-Low Function of Supersymmetric Yang-Mills Theories from Instanton Calculus, Nucl. Phys. B229 (1983) 381–393.

- [205] N. Arkani-Hamed and H. Murayama, Holomorphy, rescaling anomalies and exact beta functions in supersymmetric gauge theories, JHEP 06 (2000) 030, [hep-th/9707133].
- [206] M. Cvetic, M. J. Duff, P. Hoxha, J. T. Liu, H. Lu, J. X. Lu et al., Embedding AdS black holes in ten-dimensions and eleven-dimensions, Nucl. Phys. B558 (1999) 96-126, [hep-th/9903214].
- [207] I. Bah, AdS5 solutions from M5-branes on Riemann surface and D6-branes sources, JHEP 09 (2015) 163, [1501.06072].
- [208] S. Elitzur, A. Giveon and D. Kutasov, Branes and N=1 duality in string theory, Phys. Lett. B400 (1997) 269-274, [hep-th/9702014].
- [209] A. H. Chamseddine and M. S. Volkov, NonAbelian BPS monopoles in N=4 gauged supergravity, Phys. Rev. Lett. 79 (1997) 3343-3346, [hep-th/9707176].
- [210] A. H. Chamseddine and W. A. Sabra, D = 7 SU(2) gauged supergravity from D = 10 supergravity, Phys. Lett. **B476** (2000) 415–419, [hep-th/9911180].
- [211] M. Cvetic, H. Lu and C. N. Pope, Consistent Kaluza-Klein sphere reductions, Phys. Rev. D62 (2000) 064028, [hep-th/0003286].